

Sarah Walsh Director, Regulatory Affairs

Gas Regulatory Affairs Correspondence Email: gas.regulatory.affairs@fortisbc.com

Electric Regulatory Affairs Correspondence Email: <u>electricity.regulatory.affairs@fortisbc.com</u> FortisBC 16705 Fraser Highway Surrey, B.C. V4N 0E8 Tel: (778) 578-3861 Cell: (604) 230-7874 Fax: (604) 576-7074 www.fortisbc.com

October 24, 2024

British Columbia Utilities Commission Suite 410, 900 Howe Street Vancouver, BC V6Z 2N3

Attention: Patrick Wruck, Commission Secretary

Dear Patrick Wruck:

Re: FortisBC Energy Inc. (FEI)

Application for a Certificate of Public Convenience and Necessity (CPCN) for the Tilbury Liquefied Natural Gas (LNG) Storage Expansion Project (Application)

Supplemental Evidence

On December 29, 2020, FEI filed the above referenced Application in relation to the Tilbury Liquefied Natural Gas (LNG) Storage Expansion (TLSE) Project. In its decision of March 23, 2023 (Decision and Order G-62-23), the British Columbia Utilities Commission (BCUC) adjourned the proceeding, identified additional analysis it required before determining whether to grant a CPCN, and invited FEI to file further evidence (Supplemental Evidence).

FEI appreciates the opportunity to provide additional evidence in the context of this proceeding and has taken the time necessary to complete a significant amount of additional analysis that addresses the BCUC's comments. FEI hereby submits the public version of its Supplemental Evidence, dated October 24, 2024.

Request for Confidential Treatment of Certain Information

The Supplemental Evidence includes the following Appendices, which FEI is filing confidentially in accordance with Section 18 of the BCUC's Rules of Practice and Procedure regarding confidential documents, as set out in Order G-72-23.

- Appendix D CB&I Report
- Appendix E WSP Report
- Appendix G TLSE Detailed Cost Estimate Summary 3 BCF
- Appendix I Validation Estimating Contingency Report
- Appendix J Validation Estimating Escalation Report
- Appendix K Financial Schedules (Preferred Alternative)

FEI respectfully requests that the BCUC hold the above listed documents confidential, and believes that such information should remain confidential in perpetuity. FEI outlines below the



reasons for keeping the information confidential, and making it available only to those individuals who represent interveners and have signed undertakings.

Appendices D and E

Appendices D and E are engineering documents related to FEI's Base Plant. They should be kept confidential on the basis that they contain operationally sensitive information pertaining to FEI's assets. The disclosure of this information could impede FEI's ability to work safely and reliably operate its gas system assets and could risk the safety of both its workers and the public.

Appendices G, I, J and K

Appendices G, I, J and K include cost estimates, containing capital cost estimates for the TLSE Project. These appendices should be kept confidential on the basis that FEI may be going to the market to seek competitive bids for the materials and construction work for the Project. If the estimated costs for the material and construction work are disclosed, FEI reasonably expects that its negotiating position may be prejudiced. For instance, the bidding parties with knowledge about the estimated costs may use the estimate costs as a reference for their bidding.

Access to Confidential Application for Interveners

A number of representatives of interveners in the TLSE Project CPCN Application proceeding have already signed confidentiality undertakings. FEI is continuing to rely on those undertakings. That is, they would continue to entitle the signatories to access the Confidential Supplemental Evidence, as well as confidential information already in the evidentiary record that was previously provided to interveners. Should other individual representatives of those interveners require access to the Confidential Supplemental Evidence, FEI provided an Undertaking of Confidentiality in Appendix T-3 to the Application (Exhibit B-1-4). It would have to be executed before confidential information may be released to registered parties under the terms of the undertaking.

FEI requests that the BCUC provide it with the opportunity to file comments on any objections or concerns that it may have, should any other registered parties seek access to confidential information.

If further information is required, please contact the undersigned.

Sincerely,

FORTISBC ENERGY INC.

Original signed:

Sarah Walsh

Attachments

cc (email only): Registered Interveners



FORTISBC ENERGY INC.

Application for a Certificate of Public Convenience and Necessity for the Tilbury Liquefied Natural Gas Storage Expansion Project

Supplemental Evidence

October 24, 2024



Table of Contents

1.	INTR	DDUCTION	1
	1.1	Introduction	1
	1.2	Initial Comment on Application's Importance for Public Safety	3
	1.3	Executive Summary and Organization of the Evidence	5
		1.3.1 Summary of the Process and Evidence to date, and FEI's Response to the Adjournment Decision (Section 2)	5
		1.3.2 Project Need (Section 3)	5
		1.3.3 Expanded Alternatives Analysis (Section 4)	.14
		1.3.4 Project Description (Section 5)	.23
		1.3.5 Financial Analysis for Preferred Alternative (Section 6)	.23
		1.3.6 Environmental and Archaeological (Section 7)	.24
		1.3.7 Consultation and Engagement (Section 8)	.24
		1.3.8 BC Energy Objectives and Long-Term Gas Resource Plan (Section 9)	.25
	1.4	Proposed Regulatory Review Process	.25
2.	SUMI RESF	IARY OF THE PROCESS AND EVIDENCE TO DATE, AND FEI'S ONSE TO THE ADJOURNMENT DECISION	27
	2.1	The Regulatory Process To-Date	.27
	2.2	References to Previously Filed Evidence	.28
	2.3	FEI Has Addressed the Adjournment Decision Findings	.28
3.	PROJ	ECT NEED	31
	3.1	Introduction	.31
	3.2	Loss of T-South Supply in Winter is, by Far, FEI's Greatest Resiliency Risk and Should be Mitigated	.35
		3.2.1 2024 Resiliency Plan Is a Sound Basis for Determining Project Need	.35
		3.2.2 Catastrophic Consequences of T-South Winter No-Flow Event	.40
		3.2.3 High Cumulative Probability of T-South Winter No-Flow Event	.57
		3.2.4 Exponent's Unmitigated Probability-Adjusted Risk Results	.61
		3.2.5 Prioritizing Mitigation of T-South Risk Reflects Sound Risk Management Practices	.65
	3.3	Base Plant End-of-Life Jeopardizes Ability to Serve Customers in Normal Conditions	.68
		3.3.1 Equipment Is Obsolete, Is Experiencing Increasing Failure Rates and Is Increasingly Difficult to Maintain or Repair	.68



		3.3.2	Seismic, Environmental and Flooding Issues Are Inherent in the Base Plant Design	75
		3.3.3	Long Replacement Project Lead Time Increases Risk	80
		3.3.4	Losing Access to Tilbury Supply Would Challenge FEI's Ability to	04
			Meet Peak Load	81
	3.4	Concl	usion	99
4.	EXP/	ANDED	ALTERNATIVES ANALYSIS	100
	4.1	Introd	luction	100
	4.2	Sumn the Pr	nary of Supplemental Alternatives and Methodology to Select referred Alternative	104
		4.2.1	FEI Considered 13 Supplemental Alternatives, Contingent Scenarios, Plus a Southern Crossing Extension	104
		4.2.2	FEI Employed a Structured Process to Identify the Preferred Alternative	106
	4.3	Resul Non-V	ts of Step 1: FEI Eliminated Technically and Commercially /iable Alternatives	113
	4.4	Resul FEI's	ts of Step 2: FEI Eliminated Alternatives that Do Not Retain Existing On-System Firm Peaking Gas Supply	114
		4.4.1	Aging Base Plant Tank and Expected Increase in RS 46 Sales Make Supplemental Alternatives 2 & 3 Non-Viable	114
		4.4.2	Absence of a Gas Supply Reserve Make Supplemental Alternatives 5, 6, and 7 Non-Viable	116
		4.4.3	Four Supplemental Alternatives Remain for Step 3 Evaluation	116
	4.5	Resul	ts of Step 3 Alternatives Analysis	117
		4.5.1	Resiliency Benefit Scoring Criterion	117
		4.5.2	Availability of Dependable Gas Supply During Peak Demand Scoring Criterion	127
		4.5.3	Resolves Age-Related Base Plant Challenges Scoring Criterion	129
		4.5.4	Levelized Total Rate Impact Scoring Criterion	129
		4.5.5	Future Use Scoring Criterion	142
	4.6	Summ	nary of Alternatives Analysis Results	158
	4.7	Additi Benef	onal Information on the Preferred Alternative's Resiliency its	163
		4.7.1	Cumulative Risk Reduction in Respect of a Winter T-South No- Flow Event	163
		4.7.2	Preferred Alternative Offers Potential for Staged Shutdown to Maintain Service to Some Customers	170
		4.7.3	Preferred Alternative Has Resiliency Value Even Where T-South No-Flow Event Is Too Long to Bridge	175
		4.7.4	Preferred Alternative Mitigates Risk of 13 Other Assessed Vulnerabilities	178



		4.7.5	Preferred Alternative Protects Customers from Off-System Storage Disruptions	
	4.8	Concl	usion	186
5.	PRO.	IECT D	ESCRIPTION	187
•	5.1	Introd		187
	5.2	Drojoc	t Components Romain the Same	197
	5.2	Geote	chnical Requirements Have Changed Due to Seismic Design	107
		Standa	ard Changes	189
	5.4	Appro	ach to Updated TLSE Project Cost Estimate	189
		5.4.1	Project Cost Estimate Updated Based on AACE with New Contingency and Escalation Reports	
		5.4.2	Basis of Estimate Remains the Same Except for Timing	
		5.4.3	Risk Analysis Reviewed	190
	5.5	Update	ed TLSE Project Schedule	190
	5.6	Concl	usion	192
6.	FINA		ANALYSIS FOR PREFERRED ALTERNATIVE	193
	6.1	Update	ed TLSE Project Cost Estimate	
	•	6.1.1	Approach to Updated Project Costs Estimate, Contingency and Escalation.	
		6.1.2	Summary of Updated Project Cost Estimate	
	6.2	Update	ed Financial Analysis – 67 Year Analysis Period	198
		6.2.1	Summary of Results – 67-Year Analysis Period	
		6.2.2	Basis for 67-Year Analysis Period	
		6.2.3	Importance of Recognizing Avoided Gas Supply Costs	200
		6.2.4	Assumptions Used	200
	6.3	Update	ed Rate Impact for the Preferred alternative	202
	6.4	FEI Pe Period	rformed A Sensitivity with a Shorter (27-year) Amortization I to Address BCUC Commentary	203
	6.5	Concl	usion	204
7.	ENVI	RONME	ENT AND ARCHAEOLOGY	206
	7.1	Enviro	onmental Assessment	206
	7.2	Archa	eological Assessment	207
	7.3	Concl	usion	208
8.	CONS	SULTA	TION AND ENGAGEMENT	209
	8.1	Introd	uction	209



	8.2	FEI Ha	as Continued to Engage with Indigenous Groups	209
		8.2.1	Summary of FEI's Approach to Engagement with Indigenous Groups	209
		8.2.2	Indigenous Groups Identified for TLSE Project	210
		8.2.3	TLSE Project-Specific Engagement with Indigenous groups	211
		8.2.4	FEI Has Responded to Issues, Concerns and Opportunities Raised by Indigenous Groups	212
		8.2.5	Agreements with Indigenous Groups	213
		8.2.6	Engagement with Indigenous Groups Will Continue	214
	8.3	FEI Ha Stakel	as Continued to Engage with the Public, Governments, and nolders	214
		8.3.1	Summary of FEI's Approach to Engagement with the Public, Government and Other Stakeholders	214
		8.3.2	Engagement with the Public and Other Stakeholders	215
		8.3.3	Engagement with Government	216
		8.3.4	FEI Has Responded to Issues, Concerns and Opportunities Raised by the Public, Governments and Other Stakeholders	216
		8.3.5	Engagement with the Public, Governments and Other Stakeholders Will Continue	219
	8.4	Concl	usion	219
9.	BC EI	NERG	OBJECTIVES AND LONG TERM GAS RESOURCE PLAN	220
	9.1	Introd	uction	220
	9.2	Britisł	n Columbia's Energy Objectives	220
	9.3	Long ⁻	Term Gas Resource Plan	224
	9.4	Concl	usion	225
10.	CONC	LUSIC	DN	226



List of Appendices

- Appendix A Reference to Previous FEI Evidence by Topic
- **Appendix B** Concordance with BCUC Adjournment Commentary
- Appendix C Detailed Analysis of Supplemental Alternatives
 - C-1 OGEE Development Inc. Report
- Appendix D CB&I Report CONFIDENTIAL
- Appendix E WSP Report CONFIDENTIAL
- Appendix F Raymond Mason Report
 - A Letter of Instruction
 - B Work Experience
 - **C** Supplemental Energy Market Information
- Appendix G TLSE Detailed Cost Estimate Summary 3 BCF CONFIDENTIAL
- Appendix H Detailed Project Schedule
- Appendix I Validation Estimating Contingency Report CONFIDENTIAL
- Appendix J Validation Estimating Escalation Report CONFIDENTIAL
- Appendix K Financial Schedules (Preferred Alternative) CONFIDENTIAL
- Appendix L Notification Letter to Potentially Affected Indigenous Groups
- Appendix M Draft Orders
 - M-1 Draft Procedural Order
 - M-2 Draft Final Order Revised



Index of Tables and Figures

Table 1-1:	Time to Failure Following T-South Winter No-flow Event – Status Quo (150 MMcf/d, Regardless of LNG Volumes Available)	8
Table 1-2:	Quantitative Metrics Related to Severity of a T-South No-Flow Event at Average	
	Winter Temperatures by Incident Location	9
Table 1-3:	Exponent's Calculated Cumulative Probability Range of T-South No-Flow Event in Winter	11
Table 1-4:	Summary of Supplemental Alternatives	15
Table 1-5:	Proposed Regulatory Timetable	25
Table 2-1:	Summary of Regulatory Process	27
Table 2-2:	Summary of Where FEI Addresses Adjournment Decision Findings	28
Table 2-3:	Summary of Where FEI Addresses the BCUC's Commentary on Resiliency Plan Content	
Table 3-1:	Quantitative Metrics Related to Severity of a T-South No-Flow Event at Average Winter Temperatures by Incident Location	41
Table 3-2:	Time to Failure Following T-South Winter No-Flow Event – Status Quo (150 MMcf/d, Regardless of Available LNG volumes)	44
Table 3-3:	Exponent's Calculated Cumulative Probability of T-South No-Flow Event in Winter	58
Table 3-4:	Illustrative Commodity Costs to Replace 150 MMcf/d and 0.6 Bcf Peaking Supply at Winter 2022/23 Sumas Prices	96
Table 4-1:	Summary of Supplemental Alternatives	105
Table 4-2:	Supplemental Alternative Scoring System	109
Table 4-3:	Supplemental Alternative Evaluation Criteria	109
Table 4-4:	Step 3 Evaluation Criteria Weighting	111
Table 4-5:	Summary of Non-Viable Alternatives	113
Table 4-6:	Summary of Step 2 Screen	114
Table 4-7:	Supplemental Alternatives that Passed Step 2	116
Table 4-8:	Load Support Duration in Days Provided by Supplemental Alternatives 8 and 9 at Different Temperatures	124
Table 4-9:	Summary of Capital Costs, Cost of Service, Gas Supply Costs/Savings, and	
	Levelized Total Rate Impacts for Feasible Supplemental Alternatives	130
Table 4-10	: Alternative Resource Capacity Required to Replace 1 Bcf / 200 MMcf/d of On- System Peaking Supply	135
Table 4-11	: Avoided Annual Gas Supply Costs for Supplemental Alternatives 4, 4A, 8, and 9 (\$ millions)	139
Table 4-12	: Risk Reduction per Dollar of Rate Impact	142
Table 4-13	: Resiliency Reserve Support Under Two Hypothetical Customer Loss Sensitivities	150
Table 4-14	: LNG Volume Required Under the 2050 mDEP (2% and 5%) Load – Resiliency Retention Approach (i.e., Maintain Equivalent Support Duration as Current	
	Year)	154



Table 4-15: Results from Three-Step Process to Identify the Preferred Alternative	158
Table 4-16: Step 3 Scoring Results	160
Table 4-17: FEI's Response Options to No-Flow Event with Preferred Alternative	170
Table 4-18: Preferred Alternative Support of Lower Mainland Under Various Winter Temperatures	171
Table 4-19: Preferred Alternative (2 Bcf Resiliency Reserve & 800 MMcf/d) Lower Mainland Support Durations Under Various Staged Shutdown Scenarios	174
Table 4-20: Preferred Alternative Support Durations Under Average Winter Conditions for Non-T-South AVs Only	181
Table 5-1: Overview of Project Components	188
Table 5-2: TLSE Project Schedule and Milestones	191
Table 6-1: Breakdown of the TLSE Project Cost Estimate (\$ millions)	195
Table 6-2: Breakdown of Project Capital Costs by In-Service Year (2025-2030)	196
Table 6-3: Forecast Application and Preliminary Stage Development Costs Deferral	197
Table 6-4: Financial Analysis of the Project – 67-year Analysis Period	
Table 6-5: Summary of Delivery Rate Impact for the TLSE Project	
Table 6-6: Financial Analysis of the Project over 27 Years Compared to 67 Years (20 Years vs. 60 Years Expected Service Life)	204
Table 8-1: Indigenous Groups Potentially Affected by the TLSE Project	210
Table 8-2: Indigenous Groups Identified as Potentially Affected by the BC EAO	211
Table 9-1: British Columbia's Energy Objectives	221

Figure 1-1:	Resiliency Plan Approach	7
Figure 1-2:	PwC Economic Harm Calculation for Winter T-South No-Flow Event (Low, Median, High)	10
Figure 1-3:	Expected 23-year Winter-Only Customer Outage-Days for the Combination of	12
Figure 1-4:	Results from Structured Process to Identify the Preferred Alternative	
Figure 1-5:	Step 3 Alternatives Evaluation Results	
Figure 1-6:	T-South at Avg. Winter Temp – Expected Annual Loss Reduction	19
Figure 1-7:	Duration of Lower Mainland Load Support Provided by Supplemental Alternatives at Various Temperatures	20
Figure 1-8:	Step 3 – Scoring Results	22
Figure 3-1:	Approach to the 2024 Resiliency Plan	
Figure 3-2:	Impact to Lower Mainland at -10°C due to Loss of T-South Supply, with Mitigation from Existing Tilbury Facilities (150 MMcf/d Regas and at Least 0.35 Bcf LNG)	46
Figure 3-3:	Impact to Lower Mainland at -1.4°C due to Loss of T-South Supply, with Mitigation from Existing Tilbury Facilities (150 MMcf/d Regas and at Least 0.35 Bcf LNG)	46



Figure 3-4:	Impact to Lower Mainland at +4°C due to Loss of T-South Supply, with	
	Mitigation from Existing Tilbury Facilities (150 MMcf/d Regas and at Least	
<u>-</u> :	0.35 Bct LNG)	
Figure 3-5:	Tilbury Base Plant and Tilbury 1A Facilities – 2024 Configuration	
Figure 3-6:	Duration of Lower Mainland Load Support Provided by Tilbury Does Not	40
Figure 2.7.	Timeline for AV 2 Customer Service Restoration (with AMI)	
Figure 2 9:	Timeline for AV-2 Customer Service Restoration (with AMI)	
Figure 2 0:	PwC Economic Harm Calculation for Winter T South No Flow Event (Low	
Figure 5-9.	Median High)	54
Figure 3-10	· T-South Pipe Partially Submerged in Coguiballa River	61
Figure 3-11	Expected 23-year Winter-only GDP Loss for the Combination of AV-1, -2, -3.	
	and -54 and for Other AVs for the Tilbury Baseline Scenarios	62
Figure 3-12	: Expected 23-year Winter-only Customer Outage-days for the Combination of	
-	AV-1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios	62
Figure 3-13	: Expected 23-year Winter-only Customer Outages for the Combination of AV-	
	1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios	63
Figure 3-14	: Expected 67-year Winter-only GDP Loss for the Combination of AV-1, -2, -3,	
	and -54 and for Other AVs for the Tilbury Baseline Scenarios	63
Figure 3-15	Expected 67-year Winter-only Customer Outage-days for the Combination of AV-1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios	64
Figure 3-16	: Expected 67-year Winter-only Customer Outages for the Combination of AV-	
-	1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios	64
Figure 3-17	: Severe Corrosion in Vaporizer Stack A	72
Figure 3-18	: Typical Coating Failure on Floor of Vaporizer Bath	72
Figure 3-19	: Example of Coating Failure on Walls of Vaporizer	73
Figure 3-20	: Discoloration and Bowing of the Bath Walls	73
Figure 3-21	: 2024/25 Design and Peak Day Load vs. Recommended Supply Portfolio	83
Figure 3-22	: LNG Peak Shaving Facilities in Western Canada and the United States	85
Figure 3-23	: Holding Pipeline Capacity to Meet Peak Demand Results in Under-utilization	88
Figure 3-24	: Natural Gas for Power Generation on Northwest Pipeline (Winter Averages in	
	Bcf/Day)	
Figure 3-25	: FEI's Mist Recall (Storage Deliverability)	92
Figure 3-26	: Regional Settled Daily Prices for Winter 2022/23	95
Figure 4-1:	Three-Step Process to Identify the Preferred Alternative	107
Figure 4-2:	T-South at Average Winter – Expected Annual Loss Reduction	119
Figure 4-3:	Expected Annual Winter-only Customer Outage-days Loss for Different	
-	Supplemental Alternatives for T-South (AV-1, -2, -3, and -54) at -1.4°C	122
Figure 4-4:	Lower Mainland Support Duration Provided by Supplemental Alternatives at Various Temperatures	123
Figure 4-5:	Distribution of Average Daily Winter (December through February)	
	Temperature in the Lower Mainland (Vancouver Airport) between January	
	2013 and December 2022	125



Figure 4-6:	Graphical Illustration of Economies of Scale by Tank Capacity (from 0.6 Bcf to 3.0 Bcf)	
Figure 4-7:	Annual Cost (Post Mitigation) of Using Expanded Regional Infrastructure to Supply Equivalent of 1 Bcf, 200 MMcf/d	
Figure 4-8:	Annual Cost (Post Mitigation) of Using Expanded Regional Infrastructure to Supply Equivalent of 0.6 Bcf, 150 MMcf/d	
Figure 4-9:	Lower Mainland 2050 Customers at 2 Percent Customer Decrease Per Year (mDEP 2% Adverse Sensitivity)	145
Figure 4-10:	Lower Mainland 2050 Customers at 5 Percent Customer Decrease Per Year (mDEP 5% Adverse Sensitivity)	146
Figure 4-11:	Lower Mainland 2050 Gas Load at 2 Percent Customer Decrease Per Year (mDEP 2% Adverse Sensitivity)	147
Figure 4-12:	Lower Mainland 2050 Gas Load at 5 Percent Customer Decrease Per Year (mDEP 5% Adverse Sensitivity)	147
Figure 4-13:	Lower Mainland Load Duration Curve 2030 & 2050 Showing Peak Day Demand Assuming 2% Per Year Customer Decrease (mDEP 2% Adverse Sensitivity)	
Figure 4-14:	Load Duration Curve 2030 & 2050 Showing Peak Day Demand Assuming 5% Per Year Customer Decrease (mDEP 5% Adverse Sensitivity)	
Figure 4-15:	Impact to Lower Mainland due to Loss of T-South Supply with 2% per Year Customer Loss (mDEP 2% Adverse Sensitivity)	151
Figure 4-16:	Impact to Lower Mainland due to Loss of T-South Supply with 5% per Year Customer Loss (mDEP 2% Adverse Sensitivity)	151
Figure 4-17:	Expected Annual Customer Outage-Days With and Without Mitigation from TLSE Project – Current and Hypothetical Future Adverse Load Sensitivities (mDEP 2% and 5%)	
Figure 4-18:	Gas Supply Cost Mitigation Scenarios – Selling Peaking Supply in the Market – 5-Year Cumulative Value	
Figure 4-19:	Results from Structured Process to Identify the Preferred Alternative	158
Figure 4-20:	Step 3 Scoring Results	161
Figure 4-21:	Preferred Alternative Annual T-South-Only Expected Customer Outage Days Reduction	164
Figure 4-22:	Preferred Alternative Annual T-South-Only Expected Customer Outages Reduction	
Figure 4-23:	Preferred Alternative Annual T-South-Only Expected GDP Loss Reduction	165
Figure 4-24:	Preferred Alternative 67-Year T-South-Only Expected Customer Outage Days Reduction	
Figure 4-25:	Preferred Alternative 67-Year T-South-Only Expected Customer Outages Reduction	
Figure 4-26:	Preferred Alternative 67-Year T-South-Only Expected GDP Loss Reduction	167
Figure 4-27:	Preferred Alternative 23-Year T-South-Only Expected Customer Outage Days Reduction	
Figure 4-28:	Preferred Alternative 23-Year T-South-Only Expected Customer Outages Reduction	



Figure 4-29:	Preferred Alternative 23-Year T-South-Only Expected GDP Loss Reduction	169
Figure 4-30:	Impact to Lower Mainland Due to T-South No-Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification at -10°C	172
Figure 4-31:	Impact to Lower Mainland Due to T-South No-Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification at -1.4°C	172
Figure 4-32:	Impact to Lower Mainland Due to T-South No Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification) at Average Winter Conditions (+4°C)	173
Figure 4-33:	Annual Winter Only Expected Customer Outage Days Reduction – All AVs Except T-South	179
Figure 4-34:	Annual Winter Only Expected Customer Outages Reduction – All AVs Except T-South	
Figure 4-35:	Annual Winter-Only Expected GDP Loss Reduction – All AVs Except T-South	
Figure 4-36:	Total (All AVs) Expected Annual Winter-Only Customer Outage Days Reduction	
Figure 4-37:	Total (All AVs) Expected Annual Winter-Only Customer Outages Reduction	
Figure 4-38:	Total (All AVs) Expected Annual Winter-Only GDP Loss Reduction	



1 **1. INTRODUCTION**

2 **1.1 INTRODUCTION**

In December 2020, FortisBC Energy Inc. (FEI) applied for a Certificate of Public Convenience and Necessity (CPCN or Application) for approval of the Tilbury Liquefied Natural Gas (LNG) Storage Expansion Project (TLSE Project or the Project). The TLSE Project will enhance the resiliency of FEI's system by providing immediate backup gas supply to FEI customers, primarily in the Lower Mainland, in the event of a supply emergency, while also replacing the aging Tilbury Base Plant to maintain the essential gas supply and operational functions it has provided since 1971. The Project is therefore needed to:

- Mitigate the significant resiliency risk that hundreds of thousands of customers in the Lower Mainland will lose service for many weeks following a winter no-flow event¹ on the Westcoast Energy Inc. (WEI) T-South pipeline (T-South); and
- Ensure that FEI continues to have access to sufficient dependable peaking supply to be
 able to serve firm customers during normal operations.

15 In its decision of March 23, 2023 (Adjournment Decision),² the British Columbia Utilities Commission (BCUC) identified additional analysis required before determining whether to grant 16 17 a CPCN. The BCUC acknowledged the need for resilient utility infrastructure and the importance 18 of resiliency in the provision of safe and reliable service,³ FEI's vulnerability to a supply interruption to the Lower Mainland (a no-flow event),⁴ and that the TLSE Project will mitigate that 19 20 risk.⁵ However, the BCUC indicated that system vulnerabilities should be assessed holistically by 21 comparing various resiliency options and prioritizing and planning against various outage 22 scenarios, and then developing a comprehensive resiliency plan.⁶ The BCUC also stated that it 23 was unable to assess the cost effectiveness of the TLSE Project as a replacement for the Base

¹ As in the Application, FEI uses the term "no-flow event" to describe a total cessation of physical gas flows to a portion of FEI's service area. A no-flow event can have many different causes and will depend on the pipeline or other asset.

² Decision and Order G-62-23.

³ Adjournment Decision, p. 12: "The Panel accepts the need for resilient utility infrastructure and the importance of resiliency in the provision of safe and reliable service."

⁴ Adjournment Decision, p. 16: "The existing system currently has limited ability to mitigate a three day no-flow event. While it may be able to do so in July, it would be very challenged to do so in cooler months and not at all likely to be able to do so in a typical December or January."

⁵ Adjournment Decision, p. 16: "The Panel is satisfied with the evidence provided by FEI on the limitations of the system's ability to mitigate a 3 day no-flow event and finds that the TLSE Project will mitigate a 3-day no-flow event, provided the no-flow event does not occur simultaneously with the design peak day. In that latter circumstance, there would be insufficient regasification capacity. However, we accept FEI's assertion that even in this circumstance, the TLSE Project would provide FEI more time to conduct a more orderly shutdown than it otherwise would be able to conduct within the limits of the existing infrastructure."

⁶ Adjournment Decision, p. 12: "A robust resiliency plan should consider multiple credible threats to the FEI system, along with an assessment of the likelihood and consequence of each threat. Proposed solutions to mitigate those threats should consider the ability of a solution to mitigate one or more of the threats and a cost benefit analysis of that solution." The BCUC confirmed the need for holistic analysis in the FEI 2022 Long Term Gas Resource Plan (LTGRP) Decision: see Decision and Order G-78-24, p. 40.



- 1 Plant due to the lack of evidence regarding the TLSE Project as a "like-for-like" replacement of
- 2 the Base Plant and the ability of the TLSE Project to replace the functions or capacity currently
- 3 provided by the Base Plant with regard to dependable peaking supply.⁷

The BCUC invited FEI to provide more information about alternatives that would allow it to find "that the TLSE Project is the preferred alternative to address the need for resiliency more generally on FEI's system."⁸ In particular, the BCUC sought additional information on the remaining life of the Base Plant,⁹ identified specific Tilbury and non-Tilbury alternatives for further evaluation,¹⁰ and invited further consideration of how potential future developments, such as changes in load, might impact the alternatives assessment.¹¹

FEI appreciates the opportunity to provide additional evidence in the context of this proceeding and believes it has addressed the BCUC's comments. FEI has taken the necessary time to complete a significant amount of additional technical analysis, with the assistance of external experts, on matters including:

- A comprehensive and holistic 2024 Resiliency Plan (2024 Resiliency Plan), which includes
 FEI's risk exposure to widespread customer outages anywhere on its system due to
 supply disruptions;
- The condition of the now-53-year-old Tilbury Base Plant and independent engineering
 analysis on the potential for refurbishment;
- The availability of sufficient dependable peaking supply in the market to replace the supply
 functions of the Base Plant;
- Updated capital costs; and
- An expanded alternatives analysis which encompasses 13 Supplemental Alternatives,
 including continuing to rely on the existing Base Plant with no capital upgrades, new
 regasification only, new LNG facilities of various sizes and allocations between a resiliency
 reserve¹² and gas supply functions, and other non-Tilbury options that the BCUC had
 identified in its Adjournment Decision.

The 2024 Resiliency Plan confirms, through quantitative (probability x consequence) risk analysis, that a T-South winter no-flow event is, by far, FEI's single largest customer outage risk and should be mitigated. Irrespective of resiliency, new analysis has confirmed that the Base Plant has reached end-of-life, and a sizable capital investment is required simply to ensure that FEI can continue to dependably serve firm customers in cold periods as FEI has done for decades. While

⁷ Adjournment Decision, p. 14.

⁸ Adjournment Decision, p. 25.

⁹ Adjournment Decision, pp. 14 and 51.

¹⁰ Adjournment Decision, pp. 25-33.

¹¹ Adjournment Decision, pp. 39-40.

¹² As in the Application, FEI uses the term "resiliency reserve" to describe an amount of tank capacity that is set aside for use in a supply emergency only, i.e., it would not be used in the ordinary course of business for any other purpose such as gas supply.



- 1 FEI is relying on the facility continuing to function (albeit at reduced reliability) until construction
- 2 of the TLSE Project is complete, there is no feasible option to extend the life of the Base Plant.
- 3 FEI would be unable to replace in the market the peaking gas supply that Tilbury has provided 4 since 1971.
- 5 FEI's additional analysis confirms that, as originally proposed in the Applicaton, FEI customers 6 will obtain the greatest value from a new facility with 800 MMcf/d of regasification and a 3 Bcf tank 7 that is allocated between a 2 Bcf "resiliency reserve" and a third Bcf for gas supply (Preferred 8 Alternative). This is due in particular to a combination of: (1) the significant customer outage risk 9 mitigation it provides in respect of a winter T-South no-flow event and other known vulnerabilities; 10 (2) its peaking supply capabilities, which not only maintain FEI's Tilbury peaking supply but also 11 give FEI the flexibility to avoid annual gas supply costs by displacing other resources in its 12 portfolio; and (3) the very significant economies of scale associated with constructing a new LNG 13 facility. FEI has used hypothetical adverse load loss sensitivities to demonstrate that any of the LNG facility alternatives, regardless of size, can be expected to remain fully utilized in the future; 14
- 15 on-system LNG is a uniquely flexible asset in this regard.
- 16 FEI considers that this additional evidence reinforces that the Preferred Alternative is in the public
- 17 interest and should be approved as proposed, along with the associated deferral account and
- 18 depreciation rate approvals requested in Section 6 of the Application.

1.2 INITIAL COMMENT ON APPLICATION'S IMPORTANCE FOR PUBLIC SAFETY 19

Phrases like "adding resiliency" and "maintain peaking supply", and the associated technical 20 21 discussion about probability adjusted risk and avoided gas supply costs, tend to understate the 22 practical significance of this Application. The TLSE Project is, in essence, about ensuring that 23 hundreds of thousands of customers have gas for heat in the winter in both normal operations and in the type of supply emergency that was already experienced in 2018 (2018 T-South 24 Incident).¹³ Since FEI filed this Application in 2020, FEI has experienced another supply disruption 25 26 on T-South that necessitated reliance on Tilbury LNG, and a river flood that submerged one of 27 the T-South pipelines.

28 FEI needs the Tilbury LNG facility to be able to serve firm customers – i.e., those customers who 29 are relying on having gas service at all times - in cold winter periods in normal operations. The Base Plant facility, which houses the only regasification equipment at Tilbury and its only reserved 30 31 peaking supply, has now reached end-of-life. It can no longer be counted on to reliably provide 32 that peaking supply when it is needed. If unmitigated, thousands of firm customers in the Lower Mainland are at risk of being curtailed - having their gas turned off by FEI as a last resort to 33 34 maintain system pressure – every winter in normal operations.¹⁴ This is something that has never 35 happened in FEI's decades of operation and it is vital that it does not happen now. Losing heat in the winter is a serious health risk, particularly for the vulnerable.¹⁵ It is essential that, at an 36

¹³ See Application, Exhibit B-1-4, Section 3.4.2.2.

¹⁴ See Supplemental Evidence, Section 3.3.4.2.

¹⁵ See Supplemental Evidence, Section 3.2.2.3.



absolute minimum, FEI has continued access to enough peaking supply to meet firm load at all
 times of the year.

The 2024 Resiliency Plan, and the expert analysis that went into it, shows that a winter T-South no-flow event will, if it goes unmitigated, result in the entire Lower Mainland losing service for many weeks. That means no heat for well over half a million homes and businesses for many weeks, the effects of which would be significant, and could potentially include the deaths of vulnerable people or the suffering of serious health effects, the closing of businesses, the disruption of industrial production, and the failure of greenhouse crops. A winter T-South no-flow event, if unmitigated, is expected to cause cascading economic harm to the province.¹⁶

NERC noted recently that hundreds of people died as a result of an electric outage in Texas that
 left people without heat for only four days:¹⁷

12 More than 4.5 million people in Texas lost power during the Event, and some went 13 without power for as long as four days, while exposed to below-freezing 14 temperatures for over six days. At least 210 people died during the Event, with 15 most of the deaths connected to the power outages, of causes including 16 hypothermia, carbon monoxide poisoning, and medical conditions exacerbated by 17 freezing conditions. Among the deaths were a mother and her seven-year-old 18 daughter, and an 11- year-old boy who died in his bed, who all died of carbon 19 monoxide poisoning, and a 60-year-old disabled man who died of hypothermia. A 20 grandmother and three children trying to keep warm using a wood-burning 21 fireplace died in a house fire. In cities including Austin, Houston and San Antonio. 22 over 14 million people were ordered to boil drinking and cooking water, and 23 multiple cities ordered water conservation measures, due to broken pipes and 24 power outages (which lowered water pressure). After the city of Denton, Texas, 25 lost its gas supply, it was forced to cut power to nursing homes and water pumping 26 stations. [Footnotes removed from original]

NERC determined that the February 2021 incident in Texas was precipitated by a cold weather
 event in which critical gas and electric infrastructure failed to operate due to lack of investment in
 resiliency (i.e., weather hardening to withstand cold weather conditions):¹⁸

A confluence of two causes, both triggered by cold weather, led to the Event, part of a recurring pattern for the last ten years. First, generating units unprepared for cold weather failed in large numbers. Second, in the wake of massive natural gas production declines, and to a lesser extent, declines in natural gas processing, the natural gas fuel supply struggled to meet both residential heating load and

¹⁶ See Supplemental Evidence, Section 3.2 for a discussion of consequences of a winter T-South no-flow event.

¹⁷ FERC - NERC - Regional Entity Staff Report: The February 2021 Cold Weather Outages in Texas and the South Central United States (2021), pp. 9-10:

https://www.nerc.com/pa/rrm/ea/Documents/February 2021 Cold Weather Report.pdf. See also Supplemental Evidence, Section 3.2.2.3.

¹⁸ FERC - NERC - Regional Entity Staff Report, pp. 11-12.



generating unit demand for natural gas, exacerbated by the increasing reliance by
 generating units on natural gas. Natural gas pipeline capacity is for the most part
 designed, certificated and constructed to accommodate firm transportation
 commitments, while many natural gas-fired generating units rely on non-firm
 commodity and/or pipeline transportation contracts.

6 The energy systems were unable to meet firm peak winter demand and were forced to initiate the 7 largest controlled firm load shedding event in US history, not only causing approximately 210 8 deaths, but also an estimated \$80 to \$130 billion in direct and indirect losses to the Texas

9 economy.¹⁹

10 FEI is very cognizant that this Project involves significant capital costs and will increase customer

bills. However, as demonstrated in Texas, the implications of not making an investment in a new facility to serve load in normal operations and a likely supply emergency are too significant to

13 ignore.

14 **1.3 EXECUTIVE SUMMARY AND ORGANIZATION OF THE EVIDENCE**

15 The Supplemental Evidence is organized as follows.

16 1.3.1 Summary of the Process and Evidence to date, and FEI's Response to 17 the Adjournment Decision (Section 2)

18 This section provides an overview of the proceeding to date, provides references to prior 19 evidence, and explains how FEI has addressed the BCUC's Adjournment Decision commentary. 20 Appendix A to this Supplemental Evidence provides references to key evidence in the record to 21 date. Appendix B is a table of concordance, aligning the Adjournment Decision commentary with 22 sections in this Supplemental Evidence.

23 **1.3.2 Project Need (Section 3)**

In this section, FEI provides the additional information on Project need identified in the
 Adjournment Decision as well as updated information given the passage of time since the
 Application was filed.

27 The Application described the Project need as enhancing the resiliency of FEI's system by

28 providing immediate backup gas supply to FEI customers, primarily in the Lower Mainland, in the

29 event of a supply emergency.²⁰ FEI further clarified the Project need as including the replacement

- 30 of the existing Tilbury Base Plant in its Final Argument, as follows:²¹
- 31 Although the TLSE Project is properly characterized as a resiliency project, it would 32 be incorrect to conceptualize the full project cost as the cost of increasing

¹⁹ FERC - NERC - Regional Entity Staff Report, pp. 9-10.

²⁰ Exhibit B-1-4, Application, p. 19.

²¹ FEI Final Argument (Public Version), p. 9.



resiliency. As explained in Part Five, Section D, the TLSE Project also replaces
 the existing Tilbury Base Plant tank, which is now over 50 years old – well-beyond
 its expected service life. In the absence of the TLSE Project, FEI would still need
 to maintain the current gas supply and operational benefits provided by the Base
 Plant.

In response to the BCUC's findings and commentary in the Adjournment Decision, FEI has
undertaken extensive additional analysis to support the Project need, including the
comprehensive 2024 Resiliency Plan which is supported by the evidence of external experts. FEI
has filed the 2024 Resiliency Plan concurrently with this Supplemental Evidence.

- 10 Based on the additional evidence, FEI confirms that the TLSE Project is needed to:
- mitigate the significant resiliency risk that hundreds of thousands of customers in the
 Lower Mainland will lose service for many weeks following a winter no-flow event on T South; and
- ensure that FEI continues to have access to sufficient dependable peaking supply to be
 able to serve firm customers during normal operations.

16 The *status quo*, in the absence of significant capital investment, leaves customers exposed to 17 loss of service in normal operations due to the Base Plant having reached end-of-life and no 18 longer being a dependable source of critical peaking supply. In a widespread and lengthy gas 19 outage, which will occur following a winter T-South no-flow event, British Columbians will be 20 exposed to serious health and mortality risk and significant social and economic consequences. 21 This would be, in FEI's view, an unacceptable basis for future system planning.

22 **1.3.2.1** Mitigating FEI's Largest Customer Outage Risk (Section 3.2)

The 2024 Resiliency Plan includes a risk assessment prepared with the assistance of the expert consulting firm Exponent, Inc. (Exponent) that accounts for both probability and consequence. The risk assessment has reconfirmed that a total loss of T-South supply during winter is, and (unless mitigated) will remain, by far FEI's single greatest risk of a widespread and prolonged service disruption. The customer outage risk is so substantial that FEI needs to invest to mitigate the risk, wholly apart from the need to invest to continue dependably serving firm load in peak winter periods.

- 30 Section 3.2 is organized as follows:
- 2024 Resiliency Plan Is a Sound Basis for Determining Project Need (Section 3.2.1):
 The 2024 Resiliency Plan is a holistic and comprehensive resiliency assessment. The
 analytical approach, which was developed with Exponent's advice, is depicted in the
 following flow chart and is explained in Section 3 of the 2024 Resiliency Plan.





- Four other AVs have been identified as warranting further investigation, but FEI is not recommending at this time any additional investment where the primary driver is resiliency.
- 12Based on FEI's current analysis, the AVs other than those noted above13are already managed in a reasonable manner accounting for the14magnitude of the risk of customer outages and the cost of mitigation,15recognizing it is not feasible to fully mitigate every outage risk on a natural16gas system.
- For all Assessed Vulnerabilities, FEI will consider further risk mitigation in
 sustainment capital planning, as assets come due for replacement, and in
 the context of potential projects that also have other non-resiliency drivers.

11

²² 2024 Resiliency Plan, pp. 3-4.



- Catastrophic Consequences of Winter T-South No-Flow Event (Section 3.2.2): FEI 1 • 2 used its standard hydraulic system modelling to determine the extent of the customer 3 outages for each of the 58 AVs. A winter T-South no-flow event lasting only a matter of 4 hours will, without question, have catastrophic consequences. By virtue of the regasification constraint at Tilbury, FEI's Lower Mainland gas system will fail on the first 5 day of such an event regardless of AMI being in place.²³ and regardless of how much LNG 6 7 is assumed to be present in the Base Plant or Tilbury 1A on the day the no-flow event 8 occurs (see Table 1-1 below). In most scenarios, the depressurization will occur in an 9 uncontrolled manner that entails public safety risks. The outage will last many weeks, even 10 assuming that a material portion of customers are relighting their own appliances and that a full complement of technicians is available from the contracting community and other 11 utilities to speed customer restoration.²⁴ It will cause billions of dollars of economic harm, 12 and can be expected to result in adverse health impacts and deaths. 13
- 14

15

Table 1-1: Time to Failure Following T-South Winter No-flow Event – Status Quo (150
MMcf/d, Regardless of LNG Volumes Available)

Temperature Condition	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ²⁵
-10.0°C (very cold Lower Mainland winter day) ²⁶	2 hours
-1.4°C (warmest Lower Mainland winter in 10 years) 27	5 hours
+4.0°C (average Lower Mainland winter) ²⁸	7 hours

The Table 1-2 below summarizes direct customer impacts at average winter temperatures (+4°C) for the four anonymized segments of T-South.²⁹ The number of 17 18 affected customers increases from the values shown as temperatures drop below

²³ As the BCUC has approved the AMI Project, the entirety of the analysis in this Supplementary Evidence and the 2024 Resiliency Plan assumes that residential and small commercial AMI is in place. The extent of the mitigation provided by AMI, in terms of reducing the extent and duration of an outage, was discussed in FEI's Rebuttal Evidence.

²⁴ The basis of the outage duration estimates was discussed in detail in FEI's Rebuttal Evidence.

²⁵ This represents the approximate duration of full firm load support from all on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event.

²⁶ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are 4°C.

²⁷ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from the Vancouver International Airport (YVR).

²⁸ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.

²⁹ See Section 3.2.2.1.1 of this Supplemental Evidence for a summary of the calculation parameters. The deanonymized description of the four Assessed Vulnerabilities (AVs) comprising T-South are found in Appendices RP 4-01, 4-02 4-03 and 4-54 to the 2024 Resiliency Plan, which have been kept confidential for security reasons.



- 1 average, and the outage duration increases (other things equal) when the depressurization occurs in an uncontrolled manner.³⁰ 2
- 3 4

Table 1-2: Quantitative Metrics Related to Severity of a T-South No-Flow Event at Average Winter **Temperatures by Incident Location**

Type of Impact	Quantitative Metric	AV-1 Value	AV-2 Value	AV-3 Value	AV-54 Value
Direct customer service impact	Number of firm customers losing service on Day 1	640,100	600,400	640,400	600,400
Direct customer service impact	Total Outage Duration ³¹	63.3 days	60.2 days	71.9 days	66.3 days
Direct customer service impact	Total firm customer- outage-days	24 million	21 million	32 million	28 million
Direct customer service impact	Firm customers losing service on Day 1 as a percentage of total FEI customers	60%	56%	60%	56%

FEI retained PricewaterhouseCoopers (PwC) to provide independent analysis of 5 6 economic harm to the Province (i.e., GDP loss). PwC estimated that a single incident 7 anywhere on T-South, even at average winter temperature in the Lower Mainland $(+4^{\circ}C)$, 8 would result in catastrophic economic harm well in excess of the cost of the Preferred 9 Alternative (see Figure 1-2 below, which was prepared by PwC). PwC's analysis incorporated conservatism in a number of respects such that these GDP losses are likely 10 understated.32 11

³⁰ An uncontrolled shutdown extends the outage duration as now any air entrained in the system must be purged prior to commencing relights.

³¹ The Total Outage duration is defined as the estimated period (in days) starting on Day 1 of the customer outage to the day when service is finally restored to the last customer. Reported as the mean from Exponent's Monte Carlo analysis. Refer to Section 3.4.1.2 of the 2024 Resiliency Plan for additional information.

³² See Section 3.2.2.2.1 of the Supplemental Evidence. See also e.g., Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, Table 3 (pp. 8-9).





Figure 1-2: PwC Economic Harm Calculation for Winter T-South No-Flow Event (Low, Median, High)³³



 High Cumulative Probability of Winter T-South No-Flow Event (Section 3.2.3): Exponent used engineering analysis to estimate the probability of failure for the 58 AVs.
 Exponent's winter-only probability calculations for T-South are summarized in the Table 1-3 below. Exponent calculated a very high cumulative probability of a winter no-flow event on the T-South system. These already-high cumulative probability results are still understated, since as they do not reflect the potential for cyberattacks or other malicious action.

³³ PwC's original economic harm report, filed with the Application, was based on a hypothetical major system wide disruption over a 120-day period at a time when such an outage is anticipated to have the maximum impact (peak demand during a cold winter period). The disruption in the original PwC report was assumed to affect the entire natural gas system, outages are immediate and impact 80-90 percent of demand, and there was limited time for preparation or mitigation measures. As such, it was necessary to interpolate to estimate losses with a winter T-South no-flow event. PwC's latest report, by contrast, is based on actual outage characteristics determined by system modelling in respect of each AV. The economic harm impacts shown in this figure are related only to a T-South no-flow incident occurring at average winter temperature, and which affects customers in the Lower Mainland region.



Table 1-3: Exponent's Calculated Cumulative Probability Range of T-South No-Flow Event in Winter34

Calculation Horizon	Basis for Horizon	Exponent's Calculated Cumulative Winter Only (90 Days) Probability	
67 years	Expected Life of TLSE Project	95% - 100%	
23 years	Adverse sensitivity included in response to BCUC commentary assumes no further use of Tilbury facility for resiliency or gas supply after 2050	65% - 97%	

3 4

5

6

7

8

Just as the 2018 T-South Incident highlighted FEI's exposure to a T-South supply interruption, there have been two more concerning events on T-South since FEI filed this Application in 2020. FEI has experienced another supply disruption on T-South that necessitated reliance on Tilbury LNG and a river flood that submerged one of the T-South pipelines.

Exponent's Unmitigated Probability-Adjusted Risk Results (Section 3.2.4): Exponent 9 • 10 has calculated the winter-only risk for the 58 AVs over various time horizons (1 year, 23-11 years and 67-years), using various consequence measures (customers lost, customer-12 outage-days, and GDP losses). The same pattern shown in Figure 1-3 below is evident 13 regardless of the time horizon or consequence metric: the probability-adjusted risk posed 14 by a winter T-South no-flow event exceeds the risk associated with any other AV by a wide 15 margin. For instance: (1) the expected annual winter-only loss associated with T-South is 16 over eight times greater than the combined expected annual winter-only loss of all 54 of 17 the other AVs; and (2) the expected 23-year winter-only GDP loss for T-South is approximately 14 times greater than the next largest loss (AV-18³⁵). 18

³⁴ Reported as the lower bound and upper bound.

³⁵ AVs are anonymized for security reasons. The de-anonymized description of AV-18 is found in Confidential Appendix RP 4-18 to the 2024 Resiliency Plan.





4 5 6

7

8

3

Note: Vulnerabilities assessed by FEI and Exponent have been anonymized as "Assessed Vulnerability" numbers or "AV-#" throughout this evidence and the 2024 Resiliency Plan for security considerations. The de-anonymized information is confined to specific confidential appendices of the 2024 Resiliency Plan and expert reports.

Note: Exponent figures based on average winter temperatures exclude certain AVs where the system modelling ultimately determined that no losses would occur at average winter temperatures. See Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 173.

³⁶ The de-anonymized description of AVs-1, 2, 3 and 54 are found in the associated AV-specific appendices to the 2024 Resiliency Plan, which has been kept confidential for security reasons.

³⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figure 24.



Prioritizing Mitigation of T-South Risk Reflects Sound Risk Management Practices (Section 3.2.5): Exponent recommends prioritizing mitigation of the highest risk customer outage vulnerability, which is T-South (AV's 1, 2, 3, and 54):³⁸ "Given the probability of failure in combination with the consequences associated with AV-1, -2, -3, and -54, Exponent would expect the risk on these AVs to be prioritized for mitigation." FEI concurs with this approach.

The known consequences of a winter T-South no-flow event are sufficiently severe that,
 according to Exponent, established risk management principles would support a material
 investment to reduce the harm even at much lower probabilities than those calculated by
 Exponent. In this regard, Exponent's recommendation aligns with the views of all of the
 other independent experts that have previously provided evidence in this proceeding on
 this topic – JANA Corporation, Guidehouse and PwC.

13 1.3.2.2 Base Plant End-of-Life Jeopardizes Ability to Serve Customers in Normal 14 Conditions (Section 3.3)

The Base Plant has played a critical role in FEI's supply portfolio since 1971, including providing critical peaking supply in cold winter periods. However, additional engineering analysis, and the Base Plant's deteriorating performance despite further investment, indicate that the Base Plant has reached end-of-life. There is no feasible option to extend the life of the Base Plant or replace its peaking supply in the market. A sizable capital investment is required, irrespective of resiliency considerations, to ensure that FEI can continue to serve firm load in normal operating conditions.

- 21 Section 3.3 is organized as follows:
- Equipment is Obsolete, is Experiencing Increasing Failure Rates and is Increasingly
 Difficult to Maintain or Repair (Section 3.3.1): The Base Plant houses the only
 regasification equipment at Tilbury. It is obsolete and is experiencing increasing rates of
 failure and reliability issues, which have rendered the facility unavailable when called
 upon. The facility is also increasingly difficult to maintain and repair, with repairs taking
 weeks to months to complete. FEI is already decommissioning the Base Plant liquefaction
 equipment.
- Seismic, Environmental and Flooding Issues Are Inherent in Base Plant Design (Section 3.3.2): The Base Plant was constructed to lower engineering and safety standards that were in place at the time of construction. Even if the regasification equipment was to be replaced, there are seismic, environmental and flooding issues inherent in the original Base Plant design. FEI now operates the tank well-below its design capabilities for seismic reasons (i.e., at 0.35 Bcf instead of 0.6 Bcf), and experts have recently advised against tank retrofits to restore those original capabilities.

³⁸ Appendix RP 2, Exponent Report, para. 245.



- Long Replacement Project Lead Time Increases Risk (Section 3.3.3): In light of the
 long lead time to permit and construct a new tank and regasification, investment decisions
 must be made well in advance. FEI already faces a number of years of continuing
 equipment deterioration while a replacement facility is built, and prolonging that time
 increases the potential for a permanent failure or further reduction to tank fill levels to
 occur before a solution is in place.
- 7 Losing Access to Tilbury Supply Would Challenge FEI's Ability to Meet Peak Load 8 (Section 3.3.4): Losing partial or full access to LNG at Tilbury, which would occur (e.g.) if 9 the Base Plant regasification fails, would impair FEI's ability to provide uninterrupted service to customers in a typical winter. Even inferior fallback options for market-10 11 dependent peaking supply do not exist with today's highly constrained regional pipeline 12 and storage infrastructure. In the absence of peaking supply from Tilbury, firm customers would face curtailments in normal operations unless and until sufficiently large third-party 13 14 regional infrastructure upgrades were completed to make up the shortfall.

15 **1.3.3 Expanded Alternatives Analysis (Section 4)**

In the Application, FEI had considered a variety of ways to improve FEI's ability to withstand a
 winter T-South no-flow event.³⁹ In the Adjournment Decision, the BCUC indicated FEI should
 further consider:

- "...alternatives that offer different resiliency benefits from those that the TLSE Project
 purports to provide whether such alternatives offer greater, lesser, or qualitatively
 different levels of resiliency benefits relative to TLSE.";⁴⁰
- the possibility of using a portion of the Tilbury 1A tank for resiliency;⁴¹
- whether FEI could extend the life of the Base Plant;⁴²
- three other non-Tilbury alternatives, and revisit the resiliency provided by a Southern
 Crossing Pipeline expansion;⁴³ and
- the potential for the transition towards a lower carbon future to affect the appropriate sizing
 of the TLSE Project.⁴⁴

- ⁴² Adjournment Decision, p. 14.
- ⁴³ Adjournment Decision, pp. 25-30.

³⁹ Exhibit B-1-4, Application, Section 4. The first stage of that original alternatives analysis considered load management approaches, on- and off-system storage, and four different regional pipeline solutions. In the second stage of the original alternatives analysis, FEI assessed different tank sizing and regasification capacities that can support Lower Mainland load for at least three days during winter.

⁴⁰ Adjournment Decision, p. 25.

⁴¹ Adjournment Decision, pp. 29.

⁴⁴ Adjournment Decision, p. 52: "Further, if the throughput of natural gas is reduced due to a decrease in demand, the size of a tank and the amount of regasification required would likely be reduced." And p. 22: "The larger tank provides flexibility to accommodate future load growth that may occur. However, given the current emphasis on electrification and decarbonization in BC, it is unclear whether FEI will experience significant, or even any, future



- 1 The expanded alternatives analysis in Section 4 and Appendix C of this Supplemental Evidence
- 2 includes the alternatives and considerations identified by the BCUC in the Adjournment Decision.
- 3 Section 4 focuses primarily on those four Supplemental Alternatives that are both feasible and
- 4 can avoid curtailments of firm load in normal operations, while providing varying amounts of
- 5 resiliency. Appendix C includes the detailed analysis for every Supplemental Alternative.
- 6 Supplemental Alternative 9 is the Preferred Alternative, as it provides superior value to customers7 having regard to various considerations.

8 1.3.3.1 Summary of Supplemental Alternatives and Methodology to Select the 9 Preferred Alternative (Section 4.2)

10 The 13 Supplemental Alternatives are described in summary form in Table 1-4 below. Expanded

11 descriptions are included in Section 4.2 of this Supplemental Evidence, with full details of

12 modelling parameters provided in Table C-1 in Section 2.1 of Appendix C.

13

Table 1-4: Summary of Supplemental Alternatives

Supp. Alt. #	Name				
Alternatives Reliant on Existing Facilities ⁴⁵					
Alt 1	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)				
Alt 2	New Regasification Only – 400 MMcf/d (No Resiliency Reserve)				
Alt 3	New Regasification Only – 600 MMcf/d (No Resiliency Reserve)				
New Facility with Gas Supply But No Resiliency Reserve ⁴⁶					
Alt 4	Like-for-Like (No Resiliency Reserve)				
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification				
New Facility with Resiliency Reserve But No Gas Supply ⁴⁷					
Alt 5	Like-for-Like (Full Resiliency Reserve)				
Alt 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification				
Alt 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification				
New Facility with Both Resiliency Reserve and Gas Supply					
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification				
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification (the Preferred Alternative)				

natural gas load growth. The larger tank means greater risk of a stranded, or partially stranded, asset in the event that FEI's increased load does not emerge or decreases beyond the current load."

⁴⁵ These alternatives include prolonged reliance on the Base Plant tank with no dependable resiliency reserve, declining reliability, and a high likelihood of relying on the market for some replacement gas supply.

⁴⁶ These alternatives do not include a dependable resiliency reserve but provide different amounts of peaking gas supply and improved reliability.

⁴⁷ These alternatives include a full resiliency reserve but still rely on the market for replacement of the gas supply functions.



Supp. Alt. #	Name			
Non-Tilbury Alternatives ⁴⁸				
Alt 10	Alt 1 plus Vancouver Island Transmission System (VITS) Reverse Flow			
Alt 11	LNG from Woodfibre LNG			
Alt 12	Floating LNG			

2

3

4

5

6

7

FEI also evaluated sensitivities for Supplemental Alternatives 1, 2, 3, 4, 4A and 5 (referred to as **"contingent" scenarios**) to determine whether, optimistically assuming additional LNG volumes would be available at Tilbury on the day of a winter T-South no-flow event, would materially mitigate customer outage risk (they do not). Although FEI has provided these sensitivities in response to the Adjournment Decision, FEI remains of the view that resiliency planning should instead follow typical utility planning principles premised on sizing infrastructure and gas supply

8 assets to be able to meet firm customer requirements consistently.⁴⁹

9 FEI employed the following structured three step alternatives analysis process to ultimately

10 identify Supplemental Alternative 9 as the Preferred Alternative.



Figure 1-4: Results from Structured Process to Identify the Preferred Alternative



12

1.3.3.2 Results of Step 1: FEI Eliminated Technically and Commercially Non Viable Alternatives (Section 4.3)

In Step 1, FEI determined that all of the non-Tilbury LNG options involving the use of floating
storage, Woodfibre LNG and upgrades to the Vancouver Island Transmission System (i.e.,
Supplemental Alternatives 10, 11 and 12) are not technically and commercially viable. Continuing

⁴⁸ FEI considers these alternatives to be non-viable approaches to providing winter resiliency and peaking gas supply.

⁴⁹ In Appendix C to this Supplemental Evidence, FEI explains why contingent modelling scenarios are not dependable and do not materially reduce risk.



to rely on the Tilbury Base Plant without capital upgrades was also ruled out because the Base
 Plant has reached end-of-life. Nine supplemental alternatives (2-9) passed the Step 1 screen.

1.3.3.3 Results of Step 2: FEI Eliminated Alternatives that Do Not Retain FEI's Existing On-System Firm Peaking Gas Supply (Section 4.4)

5 Five options that would fail to retain FEI's existing peaking supply are unacceptable because they 6 would result in firm customers being curtailed (i.e., losing service) in peak winter periods under 7 normal operating conditions. The four supplemental alternatives that passed this secondary 8 screen – Supplemental Alternatives 4, 4A, 8, and 9 – would all involve a full replacement of the 9 existing Base Plant with a new facility that will, at a minimum, restore the Base Plant's original 10 design capabilities for gas supply (150 MMcf/d and 0.6 Bcf).

11 *1.3.3.4* Results of Step 3 (Section 4.5)

In Step 3, FEI numerically scored Supplemental Alternatives 4, 4A, 8, and 9 (and, for information only, Supplemental Alternatives 2, 3, 5, 6, and 7⁵⁰) having regard to the following criteria:⁵¹

- Resiliency Benefit: Considering the alternative's ability to mitigate the risk associated with a winter T-South no-flow event.
- Gas Supply: Considering the alternative's impact on the availability of dependable gas
 supply during peak demand.
- Base Plant Challenges: Considering the alternative's impact on the age-related Base
 Plant challenges (send-out reliability, seismic design, flooding, and tank venting).
- **Rate Impact:** Considering the alternative's levelized total rate impact due to both the capital costs on delivery rates and gas supply impacts/benefits on commodity rates.
- Future Use: Considering if the alternative would be useful or underutilized under two adverse future load sensitivities, in which FEI's Diversified Energy (Planning) Scenario is modified to assume higher rates of customer and load loss (2% and 5% annually) between the in-service date and 2050.

As summarized in Figure 1-5 and the bullets below, Supplemental Alternative 9 scored materially higher than the other Supplemental Alternatives. The scoring is relative to Supplemental Alternative 1 – No Capital Upgrades, which defines the financial outcome if the Application is not approved and no capital investment is made.

⁵⁰ Please refer to Appendix C to this Supplemental Evidence for the detailed analysis and scoring.

⁵¹ Further discussion regarding the scoring of viable alternatives and associated weighting of the scoring criteria can be found in Section 4 of this Supplemental Evidence.



Figure 1-5: Step 3 Alternatives Evaluation Results

Evaluation Criterion	Criterion Weightin	Alternative 4	Alternative 4A	Alternative 8	Alternative 9
Resiliency Benefit	30%	No Impact	No Impact	Medium Positive Impact	High Positive Impact
Availability of Dependable Gas Supply During Peak Demand	20%	Medium Positive Impact	High Positive Impact	Medium Positive Impact	High Positive Impact
Resolves Age Related Base Plant Challenges	20%	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact
Levelized Total Rate Impact	20%	Low Negative Impact	Low Negative Impact	Medium Negative Impact	Medium Negative Impact
Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP) Between the In- Service Date and 2050	10%	No Impact	No Impact	No Impact	No Impact
	Total Weighted Score:	1.4	1.8	1.9	2.9
					Preferred

2

3

4

5

6 7 • Step 3 "Resiliency Benefit" Scoring Criterion (Section 4.5.1): Supplemental Alternative 9 is superior to any other alternative from a risk mitigation standpoint because, in addition to significantly increasing regasification capacity, it provides a large, dedicated resiliency reserve that provides a longer load support duration at a range of temperatures present during much of the winter.

8 As shown in Figure 1-6 below, at average winter temperatures (+4°C) only Supplemental 9 Alternatives 8 and 9 provide a material improvement in risk mitigation relative to the 10 existing Base Plant with no capital upgrades. This is true even accounting for the other 11 alternatives' contingent scenarios, in which FEI is assumed to have access to non-12 dependable LNG supply on the day of a no-flow event.



Figure 1-6: T-South at Avg. Winter Temp – Expected Annual Loss Reduction



2

1

Supplemental Alternative 9 will provide superior risk mitigation relative to Supplemental Alternative 8 at below average winter temperatures that the Lower Mainland experiences during much of a typical winter. This can be seen, for instance, in Figure 1-7 below, which provides load support durations determined by transient modelling. Based on historical temperature data, nearly one-quarter of Lower Mainland winter days fall in the range of temperatures in which Supplemental Alternative 9 can bridge a three day no-flow event but Supplemental Alternative 8 could not (-6.8°C to +1.7°C).









- Step 3 "Availability of Dependable Gas Supply During Peak Demand" Scoring
 Criterion (Section 4.5.2): Supplemental Alternatives 4A and 9 score higher on this
 criterion because, while Supplemental Alternatives 4 and 8 replace the existing Tilbury
 peaking gas (150 MMcf/d and 0.6 Bcf), FEI's requirements exceed that amount. FEI
 requires 200 MMcf/d and 1.0 Bcf of peaking supply. Supplemental Alternative 4A and 9
 provide enough peaking supply to meet FEI's full requirements and thus optimize the gas
 supply portfolio.
- 8 Step 3 "Resolves Age-Related Base Plant Challenges" Scoring Criterion (Section
 9 4.5.3): Supplemental Alternatives 4, 4A, 8, and 9 all score highly because they all include
 10 the installation of a new tank and new regasification equipment built to current standards.
- Step 3 "Levelized Total Rate Impact" Scoring Criterion (Section 4.5.4): Supplemental Alternative 9 will provide an additional 0.4 Bcf of peaking gas supply relative to the smaller Supplemental Alternatives 4 and 8), which avoids the need for FEI to continue procuring peaking supply in the market. The avoided annual peaking gas costs, which would otherwise have to be incurred to obtain capacity on upgraded regional infrastructure, offset incremental capital costs to the point where the levelized total rate impact for Supplemental Alternative 9 is less than Supplemental Alternative 8.
- Step 3 "Future Use" Scoring Criterion (Section 4.5.5): All of the viable alternatives (Supplemental Alternatives 4, 4A, 8, and 9) would be fully utilized for resiliency and/or gas supply in 2050 under two adverse load loss sensitivities. On-system LNG is a unique asset when it comes to the flexibility afforded in response to changing load:
- The resiliency value (other things equal) increases if load declines because the same volume of LNG can support less load for longer. Exponent's calculations show that, since there is still residual risk with Supplemental Alternative 9 (the largest option) at current load, the additional risk mitigation that would be provided under a hypothetical reduced load scenario would be valuable for the remaining customers.
- 28 o Alternatively, FEI (with input from the BCUC) could reallocate some of the 29 resiliency reserve to gas supply to generate gas supply benefits for customers. 30 FEI's overall energy requirements in 2050 under the adverse load loss sensitivities will still far exceed 3 Bcf. The inherent characteristics of on-system LNG (e.g., its 31 32 value as a "backstop" for other resources, rapid send out, and the available 33 mitigation opportunities) mean that, in a hypothetical declining load scenario, it 34 would make more sense to optimize the portfolio by reducing other gas supply 35 assets.

36 Figure 1-8 is an alternate presentation of the results of the structured three-step alternative 37 analysis process.







FEI's strong view is that Supplemental Alternative 9 will provide the best value in meeting the Project objectives by providing the optimum peaking gas supply and providing material customer outage risk reduction. Comparing the 67-year expected GDP loss reduction and the levelized total rate impact of Supplemental Alternative 4A to Supplemental Alternatives 8 and 9 leads to the following conclusions:

For an additional 1.3 percent and 1.2 percent levelized total rate impact, Supplemental Alternatives 8 and 9 would both provide significant loss reduction against a T-South no-flow event when compared to Supplemental Alternative 4A, which would optimize peaking supply only. This is true for direct customer impacts (customer outage-days) and GDP impacts.



- The increase in levelized total rate impact between Supplemental Alternative 4A (which would optimize gas supply only) and Supplemental Alternatives 8 and 9 (which would have varying amounts of gas supply plus resiliency reserves) is small compared to the significant loss reduction that would be provided by Supplemental Alternatives 8 and 9.
- 5 Supplemental Alternatives 8 and 9 would provide similar levels of risk mitigation against a 6 T-South winter no-flow event at average winter temperature. However, Supplemental 7 Alternative 9 will provide superior risk mitigation at temperatures below the average winter 8 temperature (i.e., the expected annual customer outage-day loss at -1.4°C is lower under 9 Supplemental Alternative 9 than under Supplemental Alternative 8) for a lower levelized 10 total rate impact. Based on historical temperature data, nearly one-quarter of Lower Mainland winter days fall in the range (-6.8°C to +1.7°C) in which Supplemental Alternative 11 12 9 can bridge the regulatory shutdown period but Supplemental Alternative 8 could not. 13 This shows the benefit of the Supplemental Alternative 9's incremental resiliency reserve 14 (i.e., an additional 0.6 Bcf relative to Supplemental Alternative 8) and incremental gas 15 supply capabilities (i.e., an additional 0.4 Bcf), which come at a relatively small incremental 16 cost.

17 *1.3.3.5* Additional Information on the Preferred Alternative's Resiliency Benefits 18 (Section 4.7)

19 Supplemental Alternative 9 (the Preferred Alternative) will provide significant resiliency benefits 20 relative to today, over and above what are accounted for in the Step 3 "Resiliency Benefit" 21 criterion. These benefits include: providing the potential option of a staged shutdown that 22 maintains service to some portion of FEI's customers; avoiding uncontrolled depressurization and 23 the attendant safety risks; providing valuable time for customers, governments and social / health 24 services to prepare for an outage; mitigating the calculated customer outage risk for 13 other 25 customer outage vulnerabilities across FEI's system; and, backstopping off-system storage 26 supply in the event of a disruption at those facilities.

27 **1.3.4 Project Description (Section 5)**

This section updates certain information contained in Section 5 of the Application. The description of the TLSE Project has not changed since the filing of the Application, as the preferred alternative continues to be to replace the existing Base Plant with 3 Bcf of storage and 800 MMcf/d of regasification capacity. However, since the Application was filed, geotechnical requirements have changed due to changes in seismic design standards. FEI has updated the AACE Class 3 cost estimate with new contingency and escalation reports and has updated the Project schedule to reflect an updated in-service date at the end of 2030.

35 **1.3.5 Financial Analysis for Preferred Alternative (Section 6)**

This section provides the updated financial analysis for the Preferred Alternative (Supplemental Alternative 9), a new LNG facility with 3 Bcf of LNG storage capacity and 800 MMcf/d of regasification. At a high level, the financial analysis includes an updated AACE Class 3 capital


cost estimate, delivery rate impact and total rate impact analyses. FEI has made changes to the
 analysis to address specific commentary in the Adjournment Decision, and to better reflect the

3 overall customer bill impact.

- Updated TLSE Project Cost Estimate (Section 6.1): FEI has updated the AACE Class
 3 cost estimate, including to account for industry-wide inflationary pressures, a larger
 contingency, and P70-based escalation (in the Application, FEI used a P50). The updated
 total cost estimate for the TLSE Project is \$1,143.889 million in as-spent dollars, including
 AFUDC.
- Updated Financial Analysis 67-year Analysis Period (Section 6.2): FEI has undertaken its primary financial evaluation of the Preferred Alternative in a similar manner as in the Application. The exception is that FEI has now also recognized offsetting reductions in the cost of gas due to the Preferred Alternative's incremental gas supply benefits (avoided costs) relative to Supplemental Alternative 1 – No Capital Upgrades.
- Updated Rate Impact for the Preferred Alternative (Section 6.3): The Preferred Alternative results in a levelized total rate (i.e., bill) impact of approximately \$21 per year over the 67-year analysis period relative to Supplemental Alternative 1. FEI's bill impact analysis reflects the incremental delivery rate impact associated with the Project, partially offset by gas supply benefits (reflected in a lower cost of gas rate) when compared to current 2024 Approved rates.
- FEI Performed a Sensitivity with a Shorter (27-year) Amortization Period to Address
 BCUC Commentary (Section 6.4): In recognition of the BCUC's commentary in the
 Adjournment Decision, FEI has also provided sensitivities based on a shorter amortization
 period (i.e., full depreciation by 2050). However, FEI considers this period is too short,
 given the ongoing usefulness of the Preferred Alternative.

25 **1.3.6 Environmental and Archaeological (Section 7)**

This section provides an update of the Environmental Assessment for the Tilbury Phase 2 LNG Expansion Project (Phase 2) which has involved additional environmental and archaeological activities. Based on the additional assessments undertaken, FEI's assessment of the premitigation potential environmental and archaeological impacts of the Project have downgraded. FEI expects to mitigate potential impacts to negligible levels through additional assessments, permitting, and standard protection and mitigation measures.

32 **1.3.7 Consultation and Engagement (Section 8)**

This section provides an update on consultation and engagement with Indigenous groups, the public, government and other stakeholders. These activities are consistent with the BCUC's CPCN Guidelines, are in alignment with other FEI applications approved by the BCUC, and reflect the TLSE Project's current stage of development.



1 1.3.8 BC Energy Objectives and Long-Term Gas Resource Plan (Section 9)

2 This section discusses the factors that section 46(3.1) of the Utilities Commission Act states the 3 BCUC must consider when determining whether to issue a CPCN. The applicable of British 4 Columbia's energy objectives support the TLSE Project. FEI anticipates the Project will deliver 5 socio-economic benefits. Otherwise, British Columbia's energy objectives are generally neutral in 6 relation to the Project. Further, this Application considers and is consistent with the outcome of 7 the 2022 LTGRP proceeding, which identified the need for FEI to revise and resubmit its resiliency plan. FEI has performed that work, and the 2024 Resiliency Plan provides the basis for this 8 9 Application.

10 **1.4 PROPOSED REGULATORY REVIEW PROCESS**

11 Due to the technical nature, size of the evidentiary record, and multiple layers of confidentiality,

12 FEI considers that a written hearing process which includes a technical workshop on Exponent's

13 risk and consequence determination method and two rounds of information requests (IRs) from

14 the BCUC and interveners will provide for an appropriate and efficient review of the Supplemental

15 Evidence.

16 FEI proposes the regulatory timetable set out in Table 1-5 below for reviewing the Supplemental

17 Evidence and relevant portions of the 2024 Resiliency Plan. FEI notes that interveners have

18 already registered and have actively participated in the proceeding to date, which has involved a

19 workshop, multiple rounds of IRs, and intervener evidence and rebuttal evidence over the course

20 of multiple years until the adjournment. Existing interveners are receiving prompt notice of this

21 evidence filing.

22 The proposed regulatory timetable contemplates that the BCUC issue a procedural order related

to this submission by the week of November 25, 2024. A draft procedural order is attached as

- 24 Appendix M-1 to the Supplemental Evidence.
- 25

Table 1-5: Proposed Regulatory Timetable

ACTION	DATE (2024)
FEI provides notice of Supplemental Evidence and procedural order to registered Interveners	Friday, December 6
FEI provides confirmation of notice requirements	Friday, December 13
ACTION	DATE (2025)
Technical Workshop on Exponent's Risk and Consequence Determination Method	Wednesday, January 29
BCUC Information Request (IR) No. 1	Thursday, February 20
Intervener IR No. 1	Thursday, February 27
FEI responses to IR No. 1	Thursday, April 3

FORTISBC ENERGY INC. FEI TLSE CPCN APPLICATION – SUPPLEMENTAL EVIDENCE



ACTION	DATE (2025)
BCUC IR No. 2	Thursday, April 17
Intervener IR No. 2	Thursday, April 24
FEI responses to IR No. 2	Thursday, May 15
Letters of comment deadline	Thursday, May 22
FEI final argument	Thursday, June 12
Intervener final argument	Thursday, July 3
FEI reply argument	Thursday, July 24

1

12.SUMMARY OF THE PROCESS AND EVIDENCE TO DATE, AND2FEI'S RESPONSE TO THE ADJOURNMENT DECISION

The purpose of this Supplementary Evidence is to respond to the BCUC's findings in the Adjournment Decision and to update the evidence where needed given the passage of time since the Application was filed. There is a significant body of evidence already filed in this proceeding, and the prior evidence remains important. Accordingly, FEI has sought to avoid unduly repeating previous evidence.

8 The following sections summarize the regulatory process undertaken to-date, provide references
9 to the previously filed evidence, and identifies where in the Supplemental Evidence FEI has
10 addressed the BCUC's findings from the Adjournment Decision.

11 2.1 THE REGULATORY PROCESS TO-DATE

12 FEI filed the Application on December 29, 2020. The proceeding commenced with a workshop in

13 March 2021 and was adjourned approximately two years later, pending the filing of additional

- 14 evidence. The regulatory process to-date is summarized in the following table.
- 15

Table 2-1: Summary of Regulatory Process

Steps	Date
FEI filed Application	December 29, 2020
Workshop	March 11, 2021
In-camera Technical Session	April 7, 2021
BCUC & Intervener Information Request (IR) No. 1	June 17 & July 9, 2021
FEI Response to IR No. 1	September 8, 2021
BCUC & Intervener IR No. 2	October 6, 2021
FEI Response to IR No. 2	November 10, 2021
BCUC Panel IR No. 1	January 25, 2022
FEI Response to BCUC Panel IR No. 1	March 1, 2022
Intervener Evidence Written & Oral	March 15 & April 6, 2022
IR No. 1 on Intervener Written & Oral Evidence	April 21 & May 2, 2022
Intervener Response to IR No. 1 on Written & Oral Evidence	May 11 & May 30, 2022
FEI Rebuttal Evidence	June 2, 2022
IR No. 1 on FEI Rebuttal Evidence	June 23, 2022
FEI Response to IR No. 1 on Rebuttal Evidence	July 14, 2022
BCUC & Intervener IRs on Exhibit A2-1	August 25 & September 1, 2022
FEI Response to IRs on Exhibit A2-1	September 16, 2022
FEI Written Final Argument	October 24, 2022
Intervener Oral & Written Argument	November 3 or 4 & November 24, 2022
FEI Written Reply Argument	December 12, 2022
BCUC Adjournment Decision	March 23, 2023

FORTIS BC^{*}



1 2.2 REFERENCES TO PREVIOUSLY FILED EVIDENCE

- 2 FEI has already filed a significant amount of evidence in this proceeding, including the Application,
- 3 responses to five rounds of IRs, and rebuttal evidence. While FEI has avoided repeating that
- 4 information within the Supplemental Evidence, it remains important. Appendix A identifies key
- 5 topics discussed in previous evidence, with citations to the applicable evidence. FEI's previously
- 6 filed written Final and Reply Arguments also include extensive citations to the relevant evidence.

7 2.3 FEI HAS ADDRESSED THE ADJOURNMENT DECISION FINDINGS

8 FEI has addressed the BCUC's findings regarding further evidence needed to evaluate the TLSE

9 Project and the need for a comprehensive Resiliency Plan. The following table, which summarizes

- 10 the more detailed Table of Concordance provided in Appendix B, identifies the BCUC's findings
- 11 and commentary in the Adjournment Decision and where these points are addressed in the
- 12 Supplemental Evidence.
- 13

Adjournment Decision Finding	Where Finding is Addressed
Need for comprehensive	FEI has prepared a comprehensive Resiliency Plan, filed concurrently with this Supplemental Evidence (2024 Resiliency Plan).
Resiliency Plan (p. 12)	As part of the 2024 Resiliency Plan, FEI examined the entire regional system for single point of failure risk and performed a detailed assessment of 58 AVs (see Section 5 of the 2024 Resiliency Plan).
Consequence of T- South no-flow event (p. 11)	FEI has used its system modelling to determine how many customers would lose service following a no-flow event, and how long those customers are likely to be without service. PwC has performed an economic impacts analysis.
	(See Section 5 of the 2024 Resiliency Plan and Section 3.2.2 of this Supplemental Evidence)
Probability of T- South no-flow event	Exponent has reviewed conditions specific to each AV and determined the probability of a no-flow event occurring during the three winter months.
(time of year) (p. 9)	(See Section 5 of the 2024 Resiliency Plan and Section 3.2.3 of this Supplemental Evidence)
Probability of T- South no-flow event (time horizon) (pp. 8-9)	The 67-year expected life of the TLSE Project is an appropriate time scale for assessing risk and rate impacts, given FEI's expectation that the facility will remain useful. Exponent has nevertheless also calculated the probability for failure over a 23-year time horizon. The cumulative probability of a winter no-flow event on T-South is still significant over 23 years.
	(See Section 5 of the 2024 Resiliency Plan and Section 3.2.4 of this Supplemental Evidence)

Table 2-2: Summary of Where FEI Addresses Adjournment Decision Findings



Adjournment Decision Finding	Where Finding is Addressed						
Duration of T-South no-flow event (residual risk with TLSE Project) (pp. 9, 11, 16)	Exponent has considered the potential duration of a no-flow event having regard to the potential causes it identified. Exponent also prepared a Monte Carlo analysis to account for the potential for different no-flow event durations. Exponent determined that the TLSE Project significantly reduces the risk associated with a T-South no-flow event, both in terms of reducing probability of a customer outage and the consequences of any customer outage.						
	(See Section 5.1 of the 2024 Resiliency Plan)						
Other causes of customer outages	The 2024 Resiliency Plan evaluated 58 AVs, including a number in the Lower Mainland. The TLSE Project addresses the AVs that give rise to the largest risks.						
(p. 17)	(See Section 7 of the 2024 Resiliency Plan)						
Future developments (impact on	FEI has assessed the impact of future developments on consequence (e.g., impact of electrification of Lower Mainland gas load) and probability (e.g., climate change increasing potential of no-flow events).						
probability and consequence of T- South no-flow event) (pp. 8-9)	(See Section 6 of the 2024 Resiliency Plan) (Section 4.5.5 of this Supplemental Evidence provides adverse load loss sensitivities)						
	(Section 5.2 of the 2024 Resiliency Plan addresses the Federal Government's assessment of increasing cyber risk for utilities)						
Gas supply benefits (p. 14)	The TLSE Project, by replacing the end-of-life Base Plant that is no longer able to reliably perform its critical gas supply function, will ensure that FEI is able to continue serving peak winter loads as it has for decades. As part of the Supplemental Alternatives analysis, FEI assessed the financial value for customers of LNG located in the Lower Mainland based on the avoided cost of acquiring peaking supply on upgraded regional infrastructure.						
	(See Section 4.5.4 of this Supplemental Evidence)						
Other potential alternatives to the TLSE Project	FEI has evaluated all of the alternatives identified by the BCUC, including options that provide varying levels of customer outage risk mitigation and peaking gas supply.						
(pp. 25-32)	(See Section 4.2 of this Supplemental Evidence)						
Stranded asset risk (p. 53)	FEI discusses the future use for the TLSE Project. FEI started with its accepted planning scenario from the 2022 LTGRP, the Diversified Energy Planning (DEP) Scenario, modified into two hypothetical adverse load sensitivities to reflect hypothetical customer losses in the Lower Mainland of 2 percent and 5 percent per year until 2050.						
	(See Section 4.5.5 of this Supplemental Evidence)						

The following table provides references to where the 2024 Resiliency Plan addresses the BCUC

3 Panel's specific comments in the Adjournment Decision on the expected content of a4 comprehensive resiliency plan.



	Content							
Adjournment Decision Commentary	Where Commentary is Addressed in 2024 Resiliency Plan							
Identification of vulnerabilities	FEI identified 87 system outage vulnerabilities through a holistic review. Exponent quantified the risk associated with the 58 AVs.							
(p. 51)	(See Section 4.1 of the 2024 Resiliency Plan)							
Current risk assessment / resiliency gap	FEI identified 87 system outage vulnerabilities through a holistic review. Exponent's quantitative risk analysis of the 58 AVs accounts for available supply assets, infrastructure (including AMI), and personnel.							
analysis (p. 51)	(See Section 5 of the 2024 Resiliency Plan)							
Current risk assessment / gap	Exponent has quantified risk for all 58 AVs based on probability x consequence, as follows:							
(consequences / probability) (pp. 11, 51)	 a variety of modes of failure were considered, as applicable; Exponent only considered the risk during a three-month winter period; Exponent used hydraulic modelling that accounts for available alternative supply and uses average winter temperatures in the relevant area to determine customer outage breadth; customer outage duration is based on the BCUC-approved System Preservation and Restoration Plan⁵²; and Exponent used a Monte Carlo analysis to address uncertainty in the duration of the no-flow event. (See Section 3 and Appendix RP 3 to the 2024 Resiliency Plan (PwC Report)) 							
Impact of potential future developments on	FEI has assessed the impact of future developments on consequence (e.g., impact of electrification of Lower Mainland gas load) and probability (e.g., climate change increasing potential of no-flow events).							
risk (pp. 52-53)	(See Section 6 of the 2024 Resiliency Plan)							
	(Section 5.2 of the 2024 Resiliency Plan addresses the Federal Government's assessment of increasing cyber risk for utilities)							
Options to address risks (p. 51)	Based on the risk assessments of the 58 AVs, FEI is only recommending resiliency-specific investments to address a no-flow event on T-South at this time.							
	(See Section 7.1 of the 2024 Resiliency Plan)							

Table 2-3: Summary of Where FEI Addresses the BCUC's Commentary on Resiliency Plan

3

⁵² These assumptions were discussed in detail in FEI's Rebuttal Evidence to RCIA (Exhibit B-46).



1 3. PROJECT NEED

2 **3.1** INTRODUCTION

3 In this section, FEI provides the additional information on Project need identified in the 4 Adjournment Decision as well as updated information given the passage of time since the 5 Application was filed.

6 The Application described the Project need as enhancing the resiliency of FEI's system by 7 providing immediate backup gas supply to FEI customers, primarily in the Lower Mainland, in the 8 event of a supply emergency.⁵³

9 FEI further clarified the Project need as including the replacement of the existing Tilbury Base
 10 Plant, stating in its Final Submissions:⁵⁴

Although the TLSE Project is properly characterized as a resiliency project, it would be incorrect to conceptualize the full project cost as the cost of increasing resiliency. As explained in Part Five, Section D, the TLSE Project also replaces the existing Tilbury Base Plant tank, which is now over 50 years old – well-beyond its expected service life. In the absence of the TLSE Project, FEI would still need to maintain the current gas supply and operational benefits provided by the Base Plant.

18 In the Adjournment Decision, the BCUC accepted the need for resilient utility infrastructure and 19 the importance of resiliency in the provision of safe and reliable service. However, the BCUC 20 stated that resiliency objectives are best assessed on a holistic level by comparing various 21 resiliency options and prioritizing and planning against various outage scenarios, and then developing a comprehensive resiliency plan.⁵⁵ Regarding the need to replace the Base Plant, the 22 23 BCUC stated that it was unable to assess the cost effectiveness of the TLSE Project as a 24 replacement for the Base Plant due to the lack of evidence regarding the TLSE Project as a "likefor-like" replacement of the Base Plant and the ability of the TLSE Project to replace the functions 25 or capacity currently provided by the Base Plant with regard to peaking supply.⁵⁶ 26

Accordingly, and in response to the BCUC's findings and commentary in the Adjournment Decision, FEI has undertaken extensive additional analysis to support the Project need, including the comprehensive 2024 Resiliency Plan which is supported by the evidence of external experts.

30 Based on the additional evidence, FEI confirms that the TLSE Project is needed to:

⁵³ Exhibit B-1-4, p. 19.

⁵⁴ FEI Final Argument (Public Version), p. 9.

⁵⁵ Adjournment Decision, p. 12.

⁵⁶ Adjournment Decision, p. 14.



- mitigate the significant resiliency risk that hundreds of thousands of customers in the
 Lower Mainland will lose service for many weeks following a winter no-flow event on T South; and
- ensure that FEI continues to have access to sufficient dependable peaking supply to be
 able to serve firm customers during normal operations.

6 The status quo, in the absence of significant capital investment, leaves customers exposed to 7 loss of service in normal operations due to the Base Plant having reached end-of-life and no 8 longer being able to reliably provide critical peaking supply. In a widespread and lengthy gas 9 outage, which will occur following a winter T-South no-flow event, British Columbians are left 10 exposed to serious health and mortality risk and significant social and economic consequences. 11 This would be, in FEI's view, an unacceptable basis for future system planning.

- 12 Section 3 is organized around the following points:
- Section 3.2 Mitigating FEI's Largest Customer Outage Risk: The 2024 Resiliency
 Plan analysis confirms that a winter T-South no-flow event is FEI's single largest customer
 outage risk, and that the risk is too significant to leave unmitigated.
- Section 3.2.1 2024 Resiliency Plan Is a Sound Basis for Determining Project
 Need: The 2024 Resiliency Plan, filed in tandem with this Supplemental Evidence,
 provides a sound basis for the BCUC to conclude that FEI is appropriately targeting
 its resiliency investments towards its highest risk. It is a holistic and comprehensive
 resiliency assessment, undertaken with considerable input from independent
 experts, and reflects the BCUC's guidance and direction in the Adjournment
 Decision.
- 23 Section 3.2.2 – Catastrophic Consequences of T-South Winter No-flow 0 Event: FEI used its standard hydraulic system modelling to determine the extent 24 25 of the customer outages for each Assessed Vulnerability. A winter no-flow event 26 on T-South lasting only a matter of hours will, without question, have catastrophic 27 consequences. By virtue of the regasification constraint at Tilbury, FEI's Lower 28 Mainland gas system will fail on Day 1 regardless of how much LNG is assumed 29 to be present in the Base Plant or Tilbury 1A on the day the no-flow event occurs. 30 The outage will last many weeks. It will cause billions of dollars of economic harm, 31 and can be expected to result in adverse health impacts and death for vulnerable 32 populations.
- Section 3.2.3 High Cumulative Probability of T-South Winter No-flow Event:
 Exponent calculated a very high cumulative probability of a winter no-flow event
 on the T-South system. This is true regardless of whether the time horizon is the
 67-year expected service life of the TLSE Project or the 23-year shortened life (i.e.,
 to 2050) that FEI included as a sensitivity to be responsive to the Adjournment
 Decision. The already high results are still understated as they do not reflect the

2

27

31

34

35



potential for cyberattacks or other malicous action. There have been two more near-miss events on T-South since FEI filed the Application in 2020.

- 3 Section 3.2.4 – Exponent's Unmitigated Probability-Adjusted Risk Results: 0 4 Exponent has used a consistent methodology to quantify risk for all AVs, having 5 regard to both probability and consequences. The probability-adjusted risk posed 6 by a T-South winter no-flow event exceeds the risk associated with any other AV 7 by a wide margin, regardless of the consequence measures or time horizon used.
- 8 Section 3.2.5 – Prioritizing Mitigation of T-South Risk Reflects Sound Risk 0 9 Management Practices: Exponent recommends prioritizing mitigation of the 10 highest risk customer outage vulnerability, which is T-South. The known 11 consequences of a T-South winter no-flow event are so severe that, according to 12 Exponent (and consistent with prior expert evidence in this proceeding), 13 established risk management principles would support a material investment to 14 reduce the harm even at much lower probabilities than those calculated by 15 Exponent.
- Section 3.3 Ensuring Sufficient Peaking Supply to Maintain Service in Normal 16 17 **Operations:** The Tilbury Base Plant has played a critical role in normal operations since 1971, including providing critical winter peaking supply, and support during less severe 18 19 supply constraints and system work to sustain service to customers. However, it is clear 20 based on additional operating experience and engineering studies since filing the Application that, despite FEI's investment in the Base Plant in recent years, the Base Plant 21 22 has reached end-of-life. Continuing to operate this critical supply portfolio asset longer 23 than is necessary to construct a replacement facility would jeopardize FEI's ability to meet peak loads in normal operations. There is no feasible option to extend the life of the Base 24 25 Plant or replace its peaking supply in the market.
- 26 Section 3.3.1 – Equipment is Obsolete, is Experiencing Increasing Failure 0 Rates and is Increasingly Difficult to Maintain or Repair: The Base Plant 28 houses the only regasification equipment at Tilbury. It is obsolete and is 29 experiencing increasing rates of failure and reliability issues, which have rendered 30 the facility unavailable when called upon. The facility is also increasingly difficult to maintain and repair, with repairs taking weeks to months to complete. FEI is 32 already decommissioning the Base Plant liquefaction equipment. The declining 33 regasification equipment reliability is a matter of significant concern to FEI, since the equipment is only called on to function at times where the supply is necessary to meet customer demand and avoid curtailments.
- 36 Section 3.3.2 – Seismic, Environmental and Flooding Issues Are Inherent in 37 Base Plant Design: The Base Plant was constructed to lower engineering and 38 safety standards that were in place at the time of construction. Even if the 39 regasification equipment was to be replaced, there are seismic, environmental and 40 flooding issues inherent in the original Base Plant design. FEI now operates the



- tank well-below its design capabilities for seismic reasons, and experts have
 recently advised against tank retrofits to restore those original capabilities.
- Section 3.3.3 Long Replacement Project Lead Time Increases Risk: In light
 of the long lead time to permit and construct a new tank and regasification
 equipment, investment decisions must be made well in advance. FEI already faces
 a number of years of continuing equipment deterioration while a replacement
 facility is built, and prolonging that time increases the potential for a permanent
 failure or further reduction to tank fill levels to occur before a solution is in place.
- 9 Section 3.3.4 – Losing Access to Tilbury Supply Would Challenge FEI's Ability to Meet Peak Load: Losing partial or full access to LNG at Tilbury, which 10 11 would occur, for example, if the Base Plant regasification fails, would impair FEI's 12 ability to provide uninterrupted service to customers in a typical winter. Even 13 inferior fallback options for market-dependent peaking supply do not exist with 14 today's highly constrained regional pipeline and storage infrastructure. In the 15 absence of peaking supply from Tilbury, firm customers would face curtailments in 16 normal operations unless and until regional upgrades could be completed to make 17 up the shortfall.

FORTIS BC^{**}

3.2 Loss of T-South Supply in Winter is, by Far, FEI's Greatest Resiliency Risk and Should be Mitigated

3 As described below, the 2024 Resiliency Plan, with its probability x consequence analysis that reflects actual location-specific causes of supply disruptions, confirms that a total loss of T-South 4 5 supply during winter is FEI's single greatest customer outage risk by a large margin. FEI's Lower 6 Mainland gas system will fail on Day 1, resulting in hundreds of thousands of customers losing 7 service for many weeks, regardless of how much LNG is assumed to be present in the Base Plant 8 or Tilbury 1A on the day the no-flow event occurs. Exponent calculated a high cumulative 9 probability of a T-South winter no-flow event. Exponent endorsed targeting the T-South risk for 10 mitigation, and suggested that established risk management principles would support a material 11 investment to reduce catastrophic harm even at much lower probabilities.

12 3.2.1 2024 Resiliency Plan Is a Sound Basis for Determining Project Need

- 13 This section provides a high-level summary of the approach and content of the 2024 Resiliency
- 14 Plan, with a focus on the considerable input of independent experts and how it provides the type
- 15 of holistic assessment suggested in the Adjournment Decision. The 2024 Resiliency Plan reflects
- 16 substantial analysis and provides a sound basis for the BCUC to conclude that FEI is appropriately
- 17 targeting its resiliency investments towards its highest risk.
- 18 The primary recommendation of the 2024 Resiliency Plan is:⁵⁷

FEI has confirmed the need for resiliency-driven investment to mitigate FEI's greatest customer outage risk that is due to a winter T-South no-flow event (Assessed Vulnerabilities (AV) 1, 2, 3 and 54), as well as risk associated with AV 18. New and larger on-system LNG at Tilbury will mitigate both risks.

Four other AVs have been identified as warranting further investigation, but FEI is not recommending at this time any additional investment where the primary driver is resiliency. Based on FEI's current analysis, the AVs other than those noted above are already managed in a reasonable manner accounting for the magnitude of the risk of customer outages and the cost of mitigation, recognizing it is not feasible to fully mitigate every outage risk on a natural gas system.

For all Assessed Vulnerabilities, FEI will consider further risk mitigation in sustainment capital planning, as assets come due for replacement, and in the context of potential projects that also have other non-resiliency drivers.

19

The 2024 Resiliency Plan provides a more comprehensive explanation of the approach and results. Section 2 and Appendix B to this Supplemental Evidence provide tables of concordance summarizing how FEI has addressed the BCUC's commentary regarding: (1) further evidence needed to evaluate the TLSE Project; and (2) specific content of FEI's holistic resiliency plan.

⁵⁷ 2024 Resiliency Plan, Section 1.3.



1 *3.2.1.1* Summary of 2024 Resiliency Plan Approach and Content

The 2024 Resiliency Plan is the output of a structured process developed with input from
Exponent. Exponent's assessment of the process is: "...Exponent agrees with FEI's Resiliency
Risk Assessment Plan and its methodology for identifying, assessing, and quantifying system
risk."⁵⁸ Among other things, the 2024 Resiliency Plan incorporates:

- a holistic scan of vulnerabilities, both on FEI's system and upstream of FEI's system, that
 can interrupt supply and expose FEI to a material customer outage (87 potential resiliency
 vulnerabilities);
- a consequence and probability-based risk assessment of the 58 AVs associated with the
 most significant potential customer outages;
- discussion of cyber-related risk for energy infrastructure, sourced from government
 publications, which would be incremental to the quantified assessed risk for AVs;
- consideration of whether the risk assessment might change in the future having regard to
 future developments such as aging infrastructure, climate change impact on natural
 hazards, changes in load or anticipated projects in the region or on FEI's own system;
- identification of the vulnerabilities for which the assessed risk is sufficiently great to
 warrant resiliency-driven investment;
- identification of the vulnerabilities for which the assessed risk warrants further
 investigation to determine if a resiliency-driven investment is required;
- identification of potential resiliency investments to mitigate unacceptable risks; and
- prioritization of the AVs that warrant resiliency-driven investment.

Section 3 of the 2024 Resiliency Plan sets out the analytical process that FEI followed in
developing the Plan, which is depicted in Figure 3-1 below. Sections 4 to 8 of the 2024 Resiliency
Plan present the results of each analytical step.

⁵⁸ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 47.





Figure 3-1: Approach to the 2024 Resiliency Plan

2

In Step 1, FEI used its standard system modelling techniques to identify the off-system supply vulnerabilities and vulnerabilities on FEI's own system where an interruption in gas flows could result in material customer outages. The modelling allowed FEI to determine where the system would depressurize following a no-flow event and determine the number of customers that would experience an outage. At this initial stage of identifying potential vulnerabilities for assessment, FEI conducted its modelling at design temperatures (i.e., the coldest expected weather condition), so as to be as inclusive as possible. See Section 3.2 of the 2024 Resiliency Plan.

10 In Step 2, FEI focused its review on vulnerabilities where a no-flow event would impact at least 10,000 customers or otherwise could cause a lesser number of customers to experience an 11 12 outage of at least 14 days. As described in Section 3.3.2 of the 2024 Resiliency Plan, the objective 13 in selecting screens was to balance two competing considerations: (i) ensuring that the Plan 14 considered FEI's most significant potential outages; and (ii) avoiding unnecessary complexity and 15 delay associated with assessing smaller outages that are inherent in the gas system configuration and adequately addressed in the ordinary course of business through FEI's sustainment capital 16 17 planning and an effective Emergency Response Plan. Exponent endorsed the screening approach.59 18

⁵⁹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, p. 2: "FEI has undertaken a two-step consequencebased screening to identify vulnerabilities with the potential to result in significant customer outages that were then subject to further detailed quantification. Exponent considers FEI's overall approach for its 2024 Resiliency Plan to identify and screen system vulnerabilities (for subsequent detailed assessment and quantification) to be reasonable and along the lines of good industry risk assessment practices."



Step 3, described in Section 3.4 of the 2024 Resiliency Plan, entailed quantitative risk analysis of the 58 AVs that passed these initial screens. Exponent identified potential causes of failure and conducted a quantitative risk analysis accounting for both probability and consequence. The objective of the risk analysis was to identify where significant investments to mitigate the risk are potentially required. Exponent summarized:

6 Exponent has performed quantitative risk analyses to determine the 90-day winter-7 only annual rates of failure, as well as the probability of failure in 23 and 67 years, 8 for pipelines, compressor stations, control stations, valve assemblies, and bridges 9 carrying pipelines on FEI's transmission system that are vulnerabilities with the 10 potential to result in significant customer outages. Exponent's analysis typically makes use of the principles of performance-based engineering, which is a 11 12 sophisticated and well recognized method of quantitative risk analysis. The 13 frequency of hazards of a particular intensity are determined, in conjunction with 14 developing relationships between the hazard intensity and the probability of 15 unwanted outcomes (e.g., pipe rupture). These relationships are integrated to 16 determine annual rates of the unwanted outcome, which can be used to determine 17 the probability of occurrence over some time period. There is uncertainty in the 18 assets characteristics and hazards; thus, lower and upper bounds were 19 determined. The performance-based engineering methodology is described in 20 more detail in Section 4.2.60

21

. . .

22 Exponent has calculated the overall risk based on the consequences of a failure 23 at average winter temperatures in terms of gross domestic product ("GDP") loss, 24 customer outage days ("CODs"), and customer outages based on input 25 parameters related to the number of customers impacted, outage durations, 26 existing gas supply sources for resiliency, and the system configuration provided 27 by FEI, and based on economic losses estimated by PwC. This analysis was 28 conducted for three baseline scenarios using the Monte Carlo simulation of the 29 impact of a failure on each AV, as described in Section 6 of this report.⁶¹

30 Step 4 considered qualitatively how future events or developments may impact the current risk
 31 assessment for the AVs. FEI considered developments (including those identified in the
 32 Adjournment Decision) that could affect the probability of failure and the potential consequences.

The **Steps 5 and 6** identification and prioritization of resiliency gaps (i.e., AVs that warrant resiliency-driven investment) were informed by Exponent's advice on risk management approaches. As discussed in Section 3.2.5 of this Supplemental Evidence, Exponent recommended prioritizing the highest risk AVs. Exponent also advised that it is reasonable to

⁶⁰ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 8.

⁶¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 13.



- 1 perform mitigation activities for foreseeable events when consequences are known to be severe,
- 2 even at lower probabilities.⁶²

3 *3.2.1.2* External Experts Had Significant Input on the Plan

- The 2024 Resiliency Plan reflects input and analysis from independent experts. In particular, FEI
 has relied on experts in the following areas:
- The overall design of the structured process for developing the 2024 Resiliency Plan (Exponent);
- Appropriate criteria for identifying vulnerabilities for detailed analysis as AVs in the Plan (Exponent);
- Estimating the economic (GDP) consequences associated with a customer outage for the
 AVs (PwC);
- Estimating the probability of failure (Exponent, based on its own analysis of external risks and some prior engineering analysis of FEI's transmission system undertaken by JANA in the ordinary course of business);⁶³
- The calculation of risk based on probability and consequences (Exponent); and
- The recommended approach to risk management in terms of prioritizing risk-mitigation
 investment in light of catastrophic and non-catastrophic risks (Exponent, which re-affirmed
 previously filed advice from Guidehouse, JANA, and PwC).
- FEI expands on the roles of the respective independent experts in Section 3.1 of the 2024Resiliency Plan.

3.2.1.3 Exponent's Risk Methodology Accounts for Various Time Horizons, Various Consequence Metrics and Uncertainty

23 Exponent describes its approach to calculating risk in Section 6 of the Exponent Report:⁶⁴

24 The risk associated with each AV is guantified by the expected annual winter-only 25 loss, which is the product of combining annual winter failure rates of gas 26 infrastructure, as determined by Exponent and JANA considering a variety of 27 external and internal hazards, and the associated consequence. FEI identified 28 three consequence metrics as part of its analysis: (1) gross domestic product 29 ("GDP") loss; (2) customer outage-days ("CODs"); and (3) customer outages. 30 These are standard consequence metrics for risk assessments. For each AV, the 31 results of the risk assessment are therefore reported in terms of: (1) expected 32 annual winter-only GDP loss; (2) expected annual winter-only CODs; (3) and

⁶² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, paras. 42 and 46.

⁶³ This engineering work by JANA is distinct from the JANA "white paper" filed in the TLSE Project CPCN proceeding assessing cumulative probability of rupture for T-South based on industry rupture rates.

⁶⁴ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 155.



expected annual winter-only customer outages. The expected 23-year and 67-year
 winter-only GDP loss, CODs, and customer outages are also provided. All hazards
 (non-earthquake, earthquake, and internal mechanisms) are considered in this
 analysis.

5 The time horizons reflect the expected life of the TLSE Project (67 years), and a 2050 adverse 6 sensitivity (23 years) to address the BCUC's comments in the Adjournment Decision.

- 7 Exponent accounted for uncertainty in probability and consequences of failure using Monte Carlo
 8 analysis:⁶⁵
- 9 When an annual winter-only failure rate is multiplied by the consequences of said failure (in terms of GDP loss, customer outage-days, or customer outages), the 10 result is the expected annual loss, which is a measure of the risk associated with 11 12 an AV. The expected annual loss accounts for both the likelihood of a failure 13 occurring as well as the magnitude of its consequences. Expected loss values can 14 provide a basis for developing mitigation strategies or prioritizing interventions 15 within the system of pipelines, compressor stations, control stations, and valve 16 assemblies.
- 17 The probability of an asset failing due to a particular hazard is uncertain, i.e., is not 18 known exactly. The consequence given that a failure occurs - that is, the 19 conditional consequence or conditional loss - is also uncertain. Additionally, the 20 losses that follow a failure are conditional on the influence that parallel segments, 21 regulatory shutdowns, and other mitigation measures (e.g., the Tilbury Alternatives 22 or FEI's Mount Hayes LNG facility) have in bridging gaps in supply. To account for 23 uncertainties in both probabilities of failure and consequences of failure, Exponent 24 has performed a Monte Carlo analysis to determine the expected GDP loss, 25 expected CODs, and expected customer outages for each AV.

26 3.2.2 Catastrophic Consequences of T-South Winter No-Flow Event

27 FEI used its standard hydraulic system modelling to determine the extent of the customer outages 28 for each AV. The 2024 Resiliency Plan analysis confirmed that the loss of T-South supply lasting 29 only a matter of hours, even under average winter conditions,⁶⁶ will without question have 30 catastrophic consequences. By virtue of the regasification constraint at Tilbury, FEI's Lower 31 Mainland gas system will fail on Day 1 regardless of how much LNG is assumed to be present in 32 the Base Plant or Tilbury 1A on the day the no-flow event occurs. The outage will last many 33 weeks. It will cause billions of dollars of economic harm and can be expected to result in adverse 34 health impacts and deaths in vulnerable populations. FEI discusses each type of expected 35 consequence below.

⁶⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 160.

⁶⁶ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from the Vancouver International Airport (YVR).

3.2.2.1 Direct Customer Impact: Widespread and Lengthy Loss of Gas Service in Lower Mainland and Potentially Vancouver Island and the Interior

As discussed below, at average winter temperatures of +4°C, between 600,000 and 640,000 customers will lose service on Day 1 of a T-South no-flow event, with the number in that range depending on the location of the disruption on T-South. With this number of customer outages, and assuming FEI's Advanced Metering Infrastructure (AMI) is in place, it will take FEI between 8 and 10 weeks to restore service. The number of customers losing service on Day 1 will be higher than the numbers noted above in below-average winter temperatures, which (all else equal) has the effect of further extending customer service restoration time.^{67 68}

10 3.2.2.1.1 QUANTITATIVE METRICS REGARDING SEVERITY OF DIRECT CUSTOMER IMPACTS

The 2024 Resiliency Plan presents various metrics to assess the severity of direct customer impacts associated with a no-flow event and facilitates comparisons among all AVs. The following table provides the quantitative metrics for a T-South winter no-flow event at average winter temperatures. The 2024 Resiliency Plan evaluates T-South in four segments, which are anonymized as AV-1, AV-2, AV-3 and AV-54, recognizing that the impacts can differ depending on where on T-South the disruption occurs.⁶⁹ The impacts are severe for all segments, even at average winter temperatures.

- 18 At below average winter temperatures, the number of affected customers can increase markedly.
- 19 Near the upper end of potential consequences, under design day conditions, a prolonged failure
- 20 in one T-South AV (and at a different failure location than assumed in the 2024 Resiliency Plan)
- 21 would result in approximately 955,000 customer outages.

Table 3-1: Quantitative Metrics Related to Severity of a T-South No-Flow Event at Average Winter Temperatures by Incident Location

Type of Impact	Quantitative Metric	AV-1 Value	AV-2 Value	AV-3 Value	AV-54 Value
Direct customer service impact	Number of firm customers losing service on Day 1	640,100	600,400	640,400	600,400
Direct customer service impact	Total Outage Duration ⁷⁰	63.3 days	60.2 days	71.9 days	66.3 days
Direct customer service impact	Total firm customer- outage-days	24 million	21 million	32 million	28 million

⁶⁷ The reported customer outage numbers are from the risk assessment performed as part of the 2024 Resiliency Plan. As noted in the 2024 Resiliency Plan, the four AVs which represent T-South are based on specific failure locations and temperature conditions.

FORTIS BC^{*}

⁶⁸ The stated customer outage numbers are based on the duration of the T-South no-flow event being shorter than the duration that FEI's Mt. Hayes LNG Facility can support the Vancouver Island Transmission System (VITS). Under average winter conditions, Mt. Hayes can support the VITS for approximately 13 days. If the no-flow duration were to exceed the Mt. Hayes supply duration, then an additional 159,500 customers would lose service within FEI's VITS.

⁶⁹ These segments have been anonymized for security reasons using AV designations from the 2024 Resiliency Plan. Each of these AVs has a separate Confidential Appendix in the 2024 Resiliency Plan with restricted access.

⁷⁰ The Total Outage duration is defined as the estimated period (in days) starting on Day 1 of the customer outage to the day when service is finally restored to the last customer. Reported as the mean from Exponent's Monte Carlo analysis. Refer to Section 3.4.1.2 of the 2024 Resiliency Plan for additional information.



Type of Impact	Quantitative Metric	AV-1 Value	AV-2 Value	AV-3 Value	AV-54 Value
Direct customer service impact	Firm customers losing service on Day 1 as a percentage of total FEI customers	60%	56%	60%	56%

3

4 5

6

- 2 FEI summarizes below how these values were calculated:
 - FEI's Standard System Modelling Shows Number of Customers Losing Service: FEI
 used its standard steady state hydraulic modelling (which evaluates system pressure in
 response to fixed supply and demand conditions) to determine the extent of a customer
 outage following a no-flow event at average winter temperatures for all of the AVs.
- 7 FEI Assumed a Failure Occurred at the Downstream End of a Pipeline Segment: The 8 approach taken in the 2024 Resiliency Plan was to base consequences on the assumption 9 that a pipeline failure occurred at the far downstream end of an AV's pipeline segment. 10 This was done for consistency and to facilitate analysis. However, this approach has the 11 effect of under-reporting consequences if a pipeline fails anywhere else on the segment. 12 In many cases (e.g., sparsely populated rural areas and/or a short AV pipeline segment), 13 the difference in customer outages may be relatively small, regardless of where on the AV 14 segment the failure is assumed to occur. However, along with the temperature condition 15 at the time of the failure event, the failure location makes a significant difference for some AVs. Notably, a prolonged failure further upstream on AV-1 and at design day conditions 16 results in approximately 955,000 customers losing service. 17
- Purging, Regasification and Relight Times Are Based on System Preservation and Restoration Plan: FEI's Rebuttal Evidence discussed in detail the timeline for restoring service to customers once gas flows resume on FEI's system.⁷¹ It was based on FEI's BCUC-approved System Preservation and Restoration Plan and forms the basis of the current calculations.
- Exponent's Monte Carlo Analysis Accounted for Uncertainty in Length of No-Flow 23 • 24 **Event:** Exponent's overall risk assessment used Monte Carlo analysis to capture the 25 uncertainty in how long a no-flow event will last, given the various potential causes and 26 uncertainties. Given how little load support is provided by Tilbury at present, for all practical 27 purposes any T-South no-flow event in winter would result in the type of widespread 28 customer outages described earlier in this section. However, Exponent's use of Monte 29 Carlo analysis in its risk calculations becomes more important in the context of assessing 30 project alternatives. It addresses the BCUC's Adjournment Decision commentary about 31 whether a no-flow event could exceed the duration of support provided by the TLSE Project. Exponent's overall risk calculations are adjusted for this. 32

⁷¹ Exhibit B-46-1, Rebuttal Evidence to RCIA, Figures 3 and 4 (pp. 21-22).



1 3.2.2.1.2 All Lower Mainland Customers Will Lose Service on Day 1 Due to Regasification CONSTRAINT AT TILBURY

FEI's system modelling shows that, at minimum, a T-South no-flow event at average Lower Mainland winter temperatures (+4°C) will result in approximately 600,000 Lower Mainland customers losing service on Day 1. The modelling also reconfirmed that Tilbury's limited regasification capacity, and not the amount of LNG present, is currently the governing constraint on Tilbury's ability to support load following a winter no-flow event. In other words, the system will

- 8 fail in the same short amount of time regardless of whether Tilbury 1A is full or effectively empty
- 9 on the day the no-flow event occurs. These points are further discussed below.

10 Detailed Transient System Modelling Results

- 11 FEI used its transient hydraulic modelling tool to determine the length of time that the Tilbury Base
- 12 Plant (assuming it functions as intended despite its condition) and the Supplemental Alternatives

13 (further described in Section 4 of this Supplemental Evidence) can support the Lower Mainland

- 14 following a T-South no-flow event.
- 15 FEI's transient system modelling shows the relationship between gas supply and the pressure in
- 16 FEI's transmission system. An unmitigated supply disruption will result in a supply/demand
- 17 imbalance which causes the pressure to drop on FEI's system. FEI's transmission system must
- 18 operate above a minimum pressure to serve customers. If the pressure drops below this minimum
- 19 threshold, then customers in the affected area lose service. This relationship between gas supply
- 20 and pressure allows FEI to model the impact that a supply disruption has on system pressure
- 21 over time and, by extension, how quickly customers will lose service.

The following table summarizes how long the existing Tilbury LNG facilities, in combination with FEI's other existing capabilities, will be able to sustain the Lower Mainland load following a winter no-flow event. FEI has consistently modelled three temperatures that are reflective of local conditions, since FEI's load increases as temperatures decrease. The modelling shows that all firm customers will lose service on Day 1. Given how fast the system depressurizes, FEI expects that the system will likely depressurize in an uncontrolled manner, creating safety risks and service restoration challenges.⁷²

⁷² For a discussion of the implications of an uncontrolled shut-down, please see Exhibit B-46-1, Rebuttal Evidence to RCIA, pp. 8-9, and Exhibit B-50, RCIA IR3 43.1.



Table 3-2: Time to Failure Following T-South Winter No-Flow Event – Status Quo (150 MMcf/d, Regardless of Available LNG volumes)

Temperature Condition	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ⁷³	Expectation of Controlled or Uncontrolled Shut Down
-10°C (very cold Lower Mainland winter day) ⁷⁴	2 hours	AV-1: Uncontrolled AV-2: Uncontrolled AV-3: Uncontrolled AV-54: Uncontrolled
-1.4°C (warmest winter in 10 years) ⁷⁵	5 hours	AV-1: Uncontrolled AV-2: Uncontrolled AV-3: Uncontrolled AV-54: Uncontrolled
+4.0°C (average Lower Mainland winter) ⁷⁶	7 hours	AV-1: Controlled AV-2: Controlled AV-3: Uncontrolled AV-54: Uncontrolled

3

4 Figures 3-2 to 3-4 below are the graphical outputs of the transient modelling results for the 5 temperature scenarios in Table 3-2 above.

6 Each graphical presentation of FEI's transient modelling throughout the Supplemental Evidence7 contains the following information:

- The gas supply entering the CTS from T-South, which is represented by the grey dotted
 line titled "Huntingdon InFlow";
- The gas pressure in the CTS, which is represented by the solid orange line titled "Fraser
 Inlet Pressure"; and
- The gas supply entering the CTS from the Tilbury Facility for the alternative being modelled, which is represented by the solid yellow line titled "Tilbury Facility InFlow".

14 Each graphical presentation of FEI's transient modelling throughout the Supplemental Evidence

15 follows the timeline described below. Each event in the timeline is indicated by a red circle in the

16 graph:

⁷³ This represents the approximate duration of full firm load support from on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁷⁴ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁷⁵ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁷⁶ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



1 • At 12:00 AM:

2

3

- A no-flow event occurs off-system on the T-South pipeline.
- FEI is made aware of the event and begins preparing Tilbury for sendout.
- At 2:00 AM:
- 5 O Gas supply to the CTS ceases (i.e., between 12:00 AM and 2:00 AM it is assumed 6 that the CTS receives gas supply via T-South linepack). This can be seen in the 7 transient modelling graphs by observing the Huntingdon InFlow (grey dotted line) 8 decreasing to zero.
- 9 o LNG sendout from the Tilbury Facility begins. This can be seen in the transient
 10 modelling graphs by observing the Tilbury Facility InFlow (solid yellow line)
 11 increasing from zero.
- At 4:00 AM:
- 13 o Interruptible customers are offline.⁷⁷

The final event in the timeline is the point at which the CTS pressure (solid orange line) drops below the minimum pressure threshold and customer outages start to occur.⁷⁸ The time at which this event occurs varies based on the following parameters:

- The temperature condition at which the supply disruption is assumed to occur (i.e., the temperature condition drives the demand on the system, which impacts the rate at which the system pressure drops under the upset condition); and
- The design parameters (i.e., storage volume and regasification capacity) of the on-system
 LNG facility that is intended to mitigate the supply disruption (i.e., the Supplemental
 Alternatives).

The constraining design parameter of a given Tilbury Facility can be determined by observing the status of the LNG sendout (yellow solid line) at the time when customer outages begin. For example, if the LNG sendout goes to zero before customer outages occur, it indicates that the tank volume is the constraining factor. If customer outages occur before the LNG sendout goes to zero (indicating there is still volume left in the tank), it indicates that the regasification capacity is the constraint. The latter is true in the *status quo* modelling, indicating that the 150 MMcf/d regasification capacity at Tilbury is currently the governing limitation.

⁷⁷ Under normal operating conditions, if the Lower Mainland temperature is -10°C, interruptible customers would be offline. Therefore, the graphs showing modelling results for the -10°C temperature condition do not show interruptible customers going offline at 4:00 AM, as they would not be online due to the cold temperature condition.

⁷⁸ The CTS pressure that indicates when customer outages have started varies based on the temperature condition used in the analysis. FEI undertook a detailed iterative study to estimate the threshold pressure at each temperature condition.



- 1 The mitigation provided by the Preferred Alternative or other Supplemental Alternatives can be
- 2 viewed as the delay (relative to the *status quo*) in the time it takes for the CTS pressure to decay
- 3 to the point where customers lose service.

4Figure 3-2: Impact to Lower Mainland at -10°C due to Loss of T-South Supply, with Mitigation from5Existing Tilbury Facilities (150 MMcf/d Regas and at Least 0.35 Bcf LNG)



6

7 8

Figure 3-3: Impact to Lower Mainland at -1.4°C due to Loss of T-South Supply, with Mitigation from Existing Tilbury Facilities (150 MMcf/d Regas and at Least 0.35 Bcf LNG)



9

Figure 3-4: Impact to Lower Mainland at +4°C due to Loss of T-South Supply, with Mitigation from Existing Tilbury Facilities (150 MMcf/d Regas and at Least 0.35 Bcf LNG)



3

Regasification Constraint Means a Day 1 Full Customer Outage Will Occur Regardless of Assumed LNG Volume

As FEI described in the Application, the rapid depressurization of the Lower Mainland system
occurs because LNG from Tilbury is the only potential source of supply in winter, and Tilbury is
too small to support the daily Lower Mainland load in winter. The governing resiliency limitation
at Tilbury is the current regasification capacity of 150 MMcf/d, all of which is part of the Base Plant
(see facility schematic in Figure 3-5 below).

The modelling confirms that, by virtue of the regasification constraint, Lower Mainland customers will lose service in the same short amount of time regardless of whether Tilbury 1A is full or effectively empty on the day the no-flow event occurs. In other words, the results do not improve

even in the best-case assumption that there is 1.35 Bcf available at Tilbury (i.e., 1.0 Bcf at Tilbury

15 1A plus 0.35 Bcf in the Base Plant) when the no-flow event occurs.

FORTIS BC^{*}



Figure 3-5: Tilbury Base Plant and Tilbury 1A Facilities – 2024 Configuration⁷⁹



2

3 A regasification capacity of 150 MMcf/d means that the amount of gas FEI can re-inject in the

4 Lower Mainland system each day is limited to 150 MMcf (or 0.15 Bcf). This regasification capacity

5 is only a fraction of the daily Lower Mainland load in a typical winter. In such circumstances, the

6 amount of gas liquefied, or LNG stored, is a moot point from a resiliency standpoint. Regardless

7 of how much LNG is assumed to be on-hand at Tilbury at the time of a no-flow event, the system

8 quickly depressurizes because FEI cannot regasify the LNG fast enough keep up with the Day 1

9 demand.

10 Figure 3-6 below compares the load support duration at various winter temperatures in the Lower 11 Mainland assuming 150 MMcf/d of existing regasification, but changing the assumptions about 12 the available LNG so as to simulate different volumes being present in Tilbury 1A on the day of 13 the no-flow event (0.75 Bcf, labelled as "Alternative 1 (Contingent w/T1A)") and 0.35 Bcf (labelled as "Alternative 1 (Contingent)"80). Note that none of the LNG at Tilbury is set aside as a designated 14 15 "resiliency reserve"; rather, it is dedicated to other purposes, such that it may or may not be 16 present on the day of a no-flow event. Even with more available LNG, the durations remain 17 identical for a given temperature condition. The only time the regasification constraint would not determine the load support duration is if the temperature was high enough that daily load declines 18 19 markedly, which would typically only occur in summer months.

⁷⁹ The figure shows the reduced operating level of the Base Plant tank due to seismic reasons.

⁸⁰ This labelling reflects the fact that Supplemental Alternative 1, discussed in Section 4 of the Supplemental Evidence, contemplates a continuation of the status quo by virtue of not performing any capital upgrades. The label "Contingent" denotes that none of the LNG at Tilbury is set aside as a designated "resiliency reserve"; rather, it is dedicated to other purposes, such that it may or may not be present on the day of a no-flow event.



Figure 3-6: Duration of Lower Mainland Load Support Provided by Tilbury Does Not Change Under Different LNG Volume Assumptions



3

Although the governing constraint at Tilbury is the regasification capacity, the Base Plant tank is 4 5 also undersized based on current load for both resiliency and gas supply. The limited LNG storage 6 would quickly become the resiliency constraint if the regasification capacity was increased 7 significantly without increasing the amount of LNG storage. This is discussed further in the context 8 of Supplemental Alternatives 2 and 3 (see Section 4.4.1 and Appendix C of this Supplemental 9 Evidence), both of which involve adding regasification capacity while retaining the Base Plant 10 tank. Neither of those alternatives are effective at avoiding a Lower Mainland-wide customer outage following a winter T-South no-flow event, even at average winter temperatures. 11

12 3.2.2.1.3 RESTORING SERVICE AFTER T-SOUTH GAS FLOW RESUMES WILL TAKE WEEKS

FEI's Rebuttal Evidence explained the service restoration process and timeline in detail. This
section summarizes how the process and timeline applies to the four Avs that comprise T-South
(AV-1, AV-2, AV-3 and AV-54).

16 The number of customers losing service due to a T-South no-flow event depends on both the 17 location of the failure as well as the temperature condition at which the failure occurs. Under 18 average winter conditions and even with AMI in place,⁸¹ it would take between 57 and 70 days – 19 an average of approximately 9 weeks – to restore service to all affected customers following the

20 prompt resumption of flows on T-South. For simplicity, the durations presented here are based

⁸¹ As the BCUC has approved the AMI Project, the entirety of the analysis in this Supplemental Evidence and the 2024 Resiliency Plan assumes that residential and small commercial AMI is in place. The extent of the mitigation provided by AMI, in terms of reducing the extent and duration of an outage, is discussed in FEI's Rebuttal Evidence.



- 1 on an assumed 3-day no-flow event that does not delay restoration efforts. Exponent's Monte
- 2 Carlo-based risk analysis accounts for the uncertainty in the no-flow duration, such that its risk
- 3 calculations consider scenarios where the duration to restore service exceeds this range.
- The duration is approximately 57 days if the location of the failure on T-South is such that (a) the
 Lower Mainland alone is affected and (b) FEI is able to initiate a controlled shutdown before the
 system depressurized in an uncontrolled manner, thereby saving some restoration time (AV-2).
 The 70-day duration is applicable where the failure location is such that additional customers lose
- 8 service, and the system is shutdown in an uncontrolled manner (AV-3). The remaining two T-
- 9 South AVs (AV-1 and AV-54) result in durations of 61 days and 66 days, respectively.
- 10 Figure 3-7 below shows the contributors to the 57-day timeline for AV-2, which is the bottom end
- 11 of the time range.

1	2

Figure 3-7: Timeline for AV-2 Customer Service Restoration (with AMI)

With AMI			Weeks											
			2	3	4	5	6	7	8	9	10	11	12	13
	Shutdown	-•												
urize	DP System Regassification	•												
ressi	Purge													
Rep	Leak Survey													
	75% Relight									•				

- 14 With respect to the various stages of the AV-2 timelines:
- 4.0
- Shutdown: FEI would first need to visit approximately 50,000 large commercial and industrial premises over approximately 3-4 days to manually turn off meter valves. AMI avoids the need to do this for hundreds of thousands of residential and small customer premises, such that the shut-down timeline would otherwise be much longer.⁸²
- Regasification, purge and leak surveys: The time for these steps differs significantly depending on whether FEI has enough time to react to prevent an uncontrolled depressurization. Figure 3-7 above reflects the favourable situation where FEI has enough time to execute a controlled shutdown. Provided that FEI has enough time to assess the situation, the ability with AMI to close customer valves remotely can allow FEI to maintain pressure in portions of the system.⁸³ Purging and extensive leak surveys will be largely

⁸² Exhibit B-46-1, Rebuttal Evidence to RCIA, pp. 23-24.

⁸³ This is not universally true across all portions of the system because customers with manual meters would continue to consume gas. However, the assumption for AV-2 is that, since the failure location is reasonably far from the demand centre and does not disrupt FEI's Kingsvale Supply from reaching the Lower Mainland, there would be sufficient time for FEI to close the manual meter valves before an uncontrolled shutdown occurs. Refer to FEI's 2024 Resiliency Plan for assumptions on controlled vs. uncontrolled shutdowns.



1 unnecessary in those areas, and thus the time required for these activities is avoided and 2 the overall restoration timeline is lower relative to an uncontrolled shutdown.

3 **Relighting customer appliances:** Most of the restoration time is spent manually 4 relighting customer appliances. FEI's Rebuttal Evidence, and the two timelines shown in 5 this section (Figures 3-7 and 3-8), reflect a number of favourable assumptions that (other 6 things being equal) may tend to understate the total restoration duration. For example, it 7 contemplates only relighting essential appliances within a premises to save time and 8 assumes that 25 percent of customers relight their own appliances. It also assumes that 9 FEI has full access not only to its own workforce, but also to the entire Lower Mainland 10 gas contracting community and a large complement of mutual aid personnel from other utilities in the region. To the extent that any of these additional personnel fail to materialize, 11 12 the duration of the relight process could be materially longer.

13 Figure 3-8 below shows the timeline for restoring service to customers following a no-flow event on AV-3 at average winter temperatures (+4°C). The restoration timeline shown in Figure 3-8 is 14 longer than the above timeline primarily because, due to the AV-3 failure location, FEI expects 15 that it will be unable to initiate a controlled shutdown before the system depressurizes in an 16 17 uncontrolled manner. Additionally, AV-3 results in approximately 40,000 more customer outages 18 than AV-2.

19

Figure 3-8: Timeline for AV-3 Service Restoration (with AMI)

With AMI			Weeks											
			2	3	4	5	6	7	8	9	10	11	12	13
	Shutdown	-•												
urize	DP System Regassification	-			-									
ressu	Purge	-			-									
Rep	Leak Survey	-			-									
	75% Relight										_			

20

21 With respect to the various stages of the AV-3 restoration timeline:

22 Shutdown: The shutdown stage in the uncontrolled scenario is the same as for the • 23 controlled scenario (i.e., AMI will be used to shut down the majority of meter valves; 24 however, approximately 50,000 meters will need to be closed manually).

25 Regasification, purge and leak surveys: As AV-3 is expected to result in an uncontrolled 26 shutdown, this stage of the timeline is significantly longer than the Figure 3-7 timeline 27 (wherein the shutdown is controlled). In an uncontrolled shutdown, purging will be required 28 and thus the overall timeline will be longer. At present, there would be very little time for 29 FEI to react to an AV-3 event before the system depressurizes in an uncontrolled manner. 30 While in theory FEI could immediately shut all residential and small commercial customer 31 meter valves remotely, at present FEI would be unlikely to have the necessary information



to make that decision until it is too late. The timeline of the 2018 Incident⁸⁴ illustrated how
 long it can take before FEI has actionable information from Westcoast Energy.
 Disconnecting customers at the first notice of an incident based on limited information
 would give rise to the risk of causing unnecessary outages based on incomplete
 information. Further, even if all AMI valves were instantly shut off, the remaining 50,000
 valves requiring manual shutoff would lead to an uncontrolled shutdown.

7 Relighting customer appliances: One key difference in the relight stage for a controlled 8 shutdown vs. an uncontrolled shutdown is that, in an uncontrolled shutdown, the relight 9 rate is slightly lower than for a controlled shutdown. Due to the purging that is required in 10 an uncontrolled shutdown, resources that could otherwise be relighting customers must 11 be allocated to purging the distribution system. The result is a lower average relight rate 12 for an uncontrolled shutdown when compared to a controlled shutdown. Further, the relight 13 times for AV-3 shown in Figure 3-8 above could be materially understated for the same 14 reasons stated above regarding AV-2.

15 In response to the suggestion from REL Engineering (REL) (retained by RCIA) that the overall

16 duration to restore service to customers could be reduced, FEI explained in its Rebuttal Evidence

17 that the potential for variances in the time to fully restore service to customers is asymmetrical:⁸⁵

- 18 FEI recognizes that an actual event would vary somewhat from the assumptions 19 used; however, the potential for time variances is asymmetrical. That is, although 20 unforeseen events (e.g., identification of major leaks, bad weather, competing 21 demands limiting mutual aid assistance) could cause significant delays in the 22 restoration work, it is much less likely that opportunities for time savings would 23 meaningfully shorten the time required. FEI has performed its own sensitivity 24 testing of the working model (refer to the response to Q36) to test the assumptions 25 and does not foresee any realistic scenario where there could be time savings of the magnitude hypothesized by REL.⁸⁶ 26
- FEI believes that the timelines above are unlikely to be materially shorter than shown and couldbe materially longer.

29 3.2.2.1.4 A T-South Winter No-Flow Event Will Also Cause a Large Outage in the Interior in Some Cases

A disruption on T-South can also result in a large outage in the Interior, depending on where the incident occurs on T-South. AV-1 is the segment of T-South that, if disrupted, results in an outage in both the Lower Mainland and Interior. The result will be that, at average winter temperatures, approximately 640,000 customers will lose service for almost 9 weeks after accounting for AMI and assuming a controlled shutdown can take place. FEI's comments above regarding the potential for service restoration to take longer than estimated apply here as well.

⁸⁴ Exhibit B-1-4, Application, p. 41.

⁸⁵ Exhibit B-46-1, Rebuttal Evidence to RCIA, p. 19.

⁸⁶ Exhibit B-46-1, Rebuttal Evidence to RCIA, p. 19.



- 1 As previously explained, the modelling assumes that a failure occurs at the downstream end of
- 2 an AV, which has the effect of understating the number of customers effected if the failure occurs
- further upstream. AV-1 is an instance where the location can make a material difference. For
 instance, changing the AV-1 failure location and assuming design day conditions, a prolonged
- failure on AV-1 would result in approximately 955,000 customer outages.

6 3.2.2.2 Economic Harm: New Outage-Specific PwC Analysis Confirms Severe 7 GDP Impacts

- As part of preparing the 2024 Resiliency Plan, FEI retained PwC to estimate the economic impacts
 of a winter outage for all of the AVs based on their specific circumstances. PwC's 2024 report is
 Appendix RP 3 to the 2024 Resiliency Plan. It confirms severe GDP impacts associated with a TSouth winter no-flow event.
- 12 The Application included another PwC report (Original PwC Report), in which PwC modelled three 13 hypothetical outage scenarios using different assumptions regarding (among other things) 14 temperature, outage area, economic activity, and customer numbers. The Original PwC Report 15 was prepared for another purpose, and the hypothetical scenarios were intended to book-end 16 economic harm from a widespread customer outage. On one end of the spectrum was a localized 17 outage, while on the other end of the spectrum was an outage affecting the entire province 18 including areas beyond FEI's own service territory. Neither of these book-ends matched the 19 specific outage characteristics that FEI would experience following a T-South no-flow event, 20 making it necessary to interpolate. Thus, as part of preparing the 2024 Resiliency Plan, FEI 21 retained PwC to estimate the economic impacts of a winter outage for all of the AVs based on 22 their specific circumstances.87
- 23 PwC has used AV-specific inputs obtained from FEI, including the number of affected customers,
- 24 load profile, outage area-specific GDP and the applicable average winter temperatures. PwC also
- conducted interviews with FEI customers in various economic sectors to understand the impact a gas disruption would have on the customers' businesses, including their preparedness in the
- 27 event of a loss of gas service due to a no-flow event.⁸⁸
- PwC's key findings in respect of the impacts associated with a T-South no-flow event are explained in Section 2.2 of the 2024 Resiliency Plan. In short, PwC has estimated that a single incident on any segment of T-South (AV-1, AV-2, AV-3 or AV-54) during an average winter in the Lower Mainland would result in catastrophic economic harm well in excess of the cost of the Preferred Alternative. Figure 3-9 below was prepared by PwC at FEI's request and shows the estimated range of economic harm that would result from a single winter T-South no-flow event.⁸⁹

⁸⁷ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, Section 1.5.

⁸⁸ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, Section 2 (pp. 4-9).

⁸⁹ See Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, Figure 5 (p. 10) for summary of the highest estimated economic impacts.



1 Figure 3-9: PwC Economic Harm Calculation for Winter T-South No-Flow Event (Low, Median, 2 High)



4 3.2.2.2.1 PWC ECONOMIC IMPACT CALCULATIONS MAY BE SIGNIFICANTLY UNDERSTATED

5 PwC's calculations of economic harm, although already significant, are potentially understated. 6 For instance, PwC assumed there are no impacts from economic sectors where they lacked 7 certain data. PwC also excluded the impact of consequential outages on the Lower Mainland 8 electric system. Technical discussions between FEI and BC Hydro following the 2018 T-South 9 Incident concluded that a widespread gas outage could require rotating electric feeder outages (i.e., brown outs).⁹⁰ PwC provided a qualitative evaluation of how the loss of electric service would 11 affect the GDP impact results:⁹¹

12 Natural gas supply outages in B.C. may also place a strain on the electrical grid 13 as many households and businesses may seek to substitute the energy provided 14 by gas to that from electricity. At peak hourly demand, B.C. consumes 65 TJ of 15 natural gas, compared to only 37 TJ of electricity, so the ability of the electrical grid 16 to make up for the loss of natural gas is likely to be limited, and attempts to do so 17 may lead to infrastructure damage or the need for mitigation actions such as 18 managed power brownouts to protect the grid. In Appendix 4 we have reviewed 19 literature on other utility outage events with a focus on electrical outages to give

3

⁹⁰ Exhibit B-22-1, RCIA IR1 10.1.2.

⁹¹ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, p. 14.



- 1 insights into the possible consequences of any knock-on effects on the grid. In 2 summary our literature review suggest that:
- 3 Economic impacts estimated from full power blackouts tend to be more • 4 acute than those estimated in this report for a natural gas outage. The 5 natural gas outage scenarios in this report estimate an impact in the region 6 of 5% to 20% of GDP for the duration of the outage; in the case of a 7 complete loss of electricity, the literature provides examples where economic losses can be in the 25% to 50% range, or higher in some cases. 8
- 9 [Emphasis in original and references omitted]

3.2.2.3 Serious Public Health, Mortality and Safety Implications of Uncontrolled 10 11 Shut-Down and Prolonged Winter Outage

12 A winter outage in the Lower Mainland can also reasonably be expected to have a negative impact 13 on public health and safety, despite extensive emergency planning by FEI, municipal, regional

14 and provincial agencies. While it would be very difficult to quantify health and safety impacts, it is 15

- possible to describe potential impacts qualitatively.
- 16 3.2.2.3.1 **UNCONTROLLED DEPRESSURIZATION CREATES HAZARDOUS CONDITIONS**
- 17 An uncontrolled depressurization of the gas system, which is a most likely outcome at present,
- 18 can present hazardous conditions for the public, including the risk of fire and explosions.⁹²

19 3.2.2.3.2 LOSS OF HEAT IN WINTER LINKED TO POOR HEALTH AND MORTALITY

20 There is a known link between cold residences and workplaces and incidence of poor health and 21 mortality. PwC stated, for example:93

22 Health and safety may also be impacted, as cold residences and workplaces 23 would likely lead to an increase in the incidence of poor health such as respiratory 24 illnesses. There is extensive evidence of the link between temperature and poor 25 health. For example, mortality rates in Canada are 11% higher in winter than in 26 summer on a like-for-like basis, with death rates amongst the elderly rising by 1-27 2% for every 1°C drop in external temperature. This statistic does not incorporate 28 any loss of heating capacity; thus, it is reasonable to assume that with loss of some 29 heating capacity excess deaths would be higher than a typical winter.

30 Linked to the above point, whilst main hospitals are required to have at least three 31 days of backup heating on-site (often in the form of fuel oil), this is often not the 32 case at smaller medical facilities such as family doctors' offices, which may close 33 in the event of a natural gas outage. [Emphasis in original and references omitted]

⁹² Exhibit B-46-1, Rebuttal Evidence to RCIA, pp. 8-9; Exhibit B-50, RCIA IR3 43.1.

Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, p. 14.



Vulnerable populations (e.g., residents of long-term care facilities and nursing homes, individuals
 with disabilities, and persons requiring home-health services) are at elevated risk.

NERC noted recently that hundreds of people died as a result of an electric outage in Texas that
 left people without heat for only four days (vs. a 9-week outage following a T-South no-flow event):

5 More than 4.5 million people in Texas lost power during the Event, and some went 6 without power for as long as four days, while exposed to below-freezing 7 temperatures for over six days. At least 210 people died during the Event, with 8 most of the deaths connected to the power outages, of causes including 9 hypothermia, carbon monoxide poisoning, and medical conditions exacerbated by 10 freezing conditions. Among the deaths were a mother and her seven-year-old 11 daughter, and an 11- year-old boy who died in his bed, who all died of carbon 12 monoxide poisoning, and a 60-year-old disabled man who died of hypothermia. A grandmother and three children trying to keep warm using a wood-burning 13 14 fireplace died in a house fire. In cities including Austin, Houston and San Antonio, over 14 million people were ordered to boil drinking and cooking water, and 15 16 multiple cities ordered water conservation measures, due to broken pipes and 17 power outages (which lowered water pressure). After the city of Denton, Texas, 18 lost its gas supply, it was forced to cut power to nursing homes and water pumping stations.94 19

As Exponent discusses in its report,⁹⁵ measures intended to protect vulnerable populations 20 require time to properly implement. Providers of services to vulnerable populations would benefit 21 22 from having more than a few hours to plan for a pending outage. Examples include: relocating 23 vulnerable members of the population; setting up warming shelters for evacuated individuals who 24 cannot shelter in place at home; supplying government buildings, emergency centres, hospitals, 25 and long-term facilities with electricity-, gasoline- or propane-powered heaters, backup 26 generators, and sufficient supply of fuel; pre-positioning medical personnel and supplies at shelter 27 locations; and establishing out-of-hospital care centres to respond to cold weather related 28 conditions (e.g., hypothermia).

Natural gas outages may place strain on the Lower Mainland electrical grid as gas users may
 seek to substitute the energy provided by gas with electricity. PwC explained:⁹⁶

- 31 Natural gas supply outages in B.C. may also place a strain on the electrical grid
- 32 as many households and businesses may seek to substitute the energy provided
- 33 by gas to that from electricity. At peak hourly demand, B.C. consumes 65 TJ of
- 34 natural gas, compared to only 37 TJ of electricity, so the ability of the electrical grid
- 35 to make up for the loss of natural gas is likely to be limited, and attempts to do so

⁹⁴ <u>https://www.nerc.com/pa/rrm/ea/Documents/February_2021_Cold_Weather_Report.pdf.</u>

⁹⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, paras. 248-250.

⁹⁶ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, p. 14.



may lead to infrastructure damage or the need for mitigation actions such as
 managed power brownouts to protect the grid.

3 In the event of consequential outages on the electric system, adverse health impacts can increase 4 significantly. Loss of heat becomes a more widespread issue. There can be logistical challenges 5 for communities with municipal and regional emergency support services, which frequently 6 establish warming and reception centres in extreme cold weather conditions. These centres often 7 rely on energy such as natural gas to provide emergencies services, like heating and cooking, to 8 displaced and vulnerable people. Additionally, backup generation capacity at facilities, such as 9 hospitals, police stations, fire halls, and schools are temporary and based on the availability of 10 fuels, and have limited ability to bridge outages.

PwC explained that such non-GDP impacts can be conceptualized as "consumer surplus" effects
 and that PwC would expect them to be material:⁹⁷

13 The literature also measures "consumer surplus" effects on residential customers associated with electrical outages, which can be defined as effects on consumer 14 15 wellbeing that are not measured by GDP such as inconvenience, health impacts, 16 leisure and other factors. At a high level, studies have placed this cost at around 17 US\$1,750 per household for a one-month blackout. This cost would equate to 18 US\$3.5 billion if applied to the 2 million households in B.C. While our study has not 19 measured consumer surplus effects of a natural gas outage, we would also expect 20 these to be material given the impacts on health, education, reduced ability to heat 21 homes and other disruptions that residents would likely experience.

22 3.2.3 High Cumulative Probability of T-South Winter No-Flow Event

23 The 2024 Resiliency Plan reflects Exponent's independent expert assessment of the probability 24 of failure for the AVs based on various integrity-related (internal) and external causes. Exponent 25 calculated a very high cumulative probability of a winter no-flow event on the T-South system. 26 This is true regardless of whether the time horizon is the 67-year expected service life of the TLSE 27 Project or the 23-year shortened life (i.e., to 2050) that FEI included as a sensitivity to be 28 responsive to the Adjournment Decision. The 23-year winter-only lower and upper bound failure 29 probabilities of a T-South failure are 65 percent and 97 percent, respectively. These already-high 30 cumulative probability results are still understated as they do not reflect the potential for cyberattacks or other malicous action. Just as the 2018 T-South Incident highlighted FEI's 31 32 exposure to a T-South supply interruption, there have been two more near-miss events on T-33 South since FEI filed this Application in 2020.

⁹⁷ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report, pp. 14-15.



1 *3.2.3.1* Exponent's Calculated Cumulative Probabilities

- FEI instructed Exponent to perform cumulative probability calculations for a winter-only disruption
 on all AVs based on two horizons:
- The 67-year expected life of the TLSE Project, which FEI regards as the appropriate horizon; and
- A 23-year sensitivity based on an assumed facility retirement in 2050, which FEI included
 to address the BCUC's commentary regarding the energy transition potentially shortening
 the useful life of a new LNG facility.

9 FEI explains in Section 4.5.5 of this Supplemental Evidence why it regards 23 years as being too

10 short.⁹⁸ However, the cumulative probability is very high regardless, as shown in the table below.

11 Table 3-3: Exponent's Calculated Cumulative Probability of T-South No-Flow Event in Winter⁹⁹

Calculation Horizon	Basis for Horizon	Exponent's Calculated Cumulative Winter Only (90 Days) Probability
67 years	Expected Life of TLSE Project	95% - 100%
23 years	Hypothetical adverse sensitivity assumes no further use of Tilbury facility for resiliency or gas supply after 2050	65% - 97%

12

The Exponent Report explains these results.¹⁰⁰ A key factor driving the high probability of failure 13 14 is the length of the T-South pipeline. Due to its length, T-South has increased exposure to all 15 hazards. One cause of failure (also referred to as a "mode of failure"), that drives the high probability of a winter T-South outage, is the internal per kilometre rupture and ignited rupture 16 17 rates when applied to a very lengthy pipeline like T-South. As occurred in the 2018 T-South Incident, rupture or ignited rupture of one of the two T-South lines could be expected to prompt a 18 19 regulatory / precautionary shut-down of the other line in the same right-of-way, resulting in a no-20 flow event. In the 2018 T-South Incident, the regulatory shut-down/no-flow event lasted 2 days, 21 followed by a long period of reduced flow. At present, Tilbury, regardless of the potential volumes 22 in the Base Plant and Tilbury 1A, could not support Lower Mainland load for a single day in winter, 23 such that the regulatory shutdown would cause Lower Mainland customers to lose service very 24 quickly.

25 3.2.3.1.1 EXPONENT'S APPROACH TO CALCULATING CUMULATIVE PROBABILITY

26 Exponent's calculated overall cumulative probability of a winter T-South no-flow event reflects:

⁹⁸ If FEI's load declines in the future, FEI would expect to optimize the size of its resiliency reserve and allocate more of the tank to gas supply. The unique attributes of on-system LNG make it possible to optimize the gas supply portfolio by shedding other contracted elements of the gas supply portfolio, and generate gas supply mitigation revenue in the normal course.

⁹⁹ Reported as the lower bound and upper bound.

¹⁰⁰ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 138.



- The combination of the annual failure rate associated with all of the modes of failure on
 the T-South system capable of being assigned a failure rate (cyberattacks and sabotage
 are not included, such that the cumulative probabilities are understated);
- The annual failure rate is translated into a winter-only rate by pro-rating to 90 days; and
- The winter-only annual failure rate is translated to a cumulative probability over a defined
 period. Exponent was instructed to assume 67 years (coinciding with the expected service
 life of the TLSE Project) and a 23-year shortened life sensitivity to 2050.

8 The cumulative probability calculations for a T-South no-flow event combine engineering work 9 undertaken previously by JANA for integrity projects and new work undertaken by Exponent. The 10 Exponent Report provides more information on the basis for Exponent's probability 11 calculations.¹⁰¹

3.2.3.2 Two Incidents Since the 2018 T-South Incident Highlight FEI's Exposure to Upstream Supply Disruptions

14 Two incidents have occurred since the 2018 T-South Incident that highlight FEI's risk exposure 15 associated with being dependent on T-South for much of its supply. Given the recency of these

16 events, this is new information in this proceeding.

17 3.2.3.2.1 <u>THE 2023 T-SOUTH "SWAMP GAS" INCIDENT</u>

18 On January 31, 2023, Westcoast identified a potential leak on its T-South system NPS 36¹⁰² pipeline, and as a result, shut in the pipeline as a precaution to investigate. The shut-in resulted 19 20 in flows on the T-South system being reduced to approximately 65 percent of firm service. The 21 proximity between the incident and FEI's system meant that, had the triggering incident resulted 22 in a no-flow event, the majority of the T-South linepack would not be accessible to FEI customers 23 in the Lower Mainland. Instead, T-South linepack would be limited to the volume of gas stored in 24 the relatively short pipeline segment between the incident location and FEI's system. Less 25 linepack results in a shorter support duration, which in turn results in FEI having less time to react 26 to the incident.

Westcoast shut in its NPS 36 T-South pipeline due to safety concerns related to gas "bubbling" that was observed in the vicinity of their pipeline. With only limited field resources in the area, Enbridge accepted assistance from FEI to investigate the potential leak. FEI provided leak surveyors and equipment to investigate the gas emissions, and several samples were taken, analyzed, and eventually determined to be swamp gas.¹⁰³ Once the pipeline integrity was confirmed, Enbridge resumed full service of the Westcoast T-South system, restoring full capacity and supply to FEI's CTS approximately 24 hours after the initial shut in.

¹⁰¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Section 4.2 (pp. 23-28).

¹⁰² NPS refers to the nominal pipe size outside diameter, measured in inches.

¹⁰³ Swamp gas is a naturally occurring gas (composed primarily of methane) that results from the anaerobic decomposition of organic matter in moisture-laden ground conditions.


- As a result of the incident, FEI experienced an unexpected reduction of gas supply to its downstream CTS over approximately a 24-hour period. FEI relied on its Mt. Hayes and Tilbury facilities to make up for the supply shortfall, which limited the amount of Sumas supply FEI had to purchase on the day when the incident occurred. In addition, the actual temperature on January 31, 2023 was warmer than previously forecast. Had the weather been colder, FEI would have used more LNG supply and risked not having sufficient peaking supply for the rest of the winter. FEI notes that, despite the triggering event ultimately being a false alarm, it still took 24 hours to
- 8 restore full capacity and supply to the CTS after the initial shut in. Once AMI is in place, FEI would 9 theoretically have the ability to shut off all residential and small commercial customers in the 10 Lower Mainland so as to avoid a worst-case uncontrolled shut-down. However, doing so without 11 the benefit of full information risks unnecessarily causing hundreds of thousands of customers to 12 lose service for many weeks. Having enough on-system LNG on hand to support the Lower 13 Mainland load while FEI assesses the situation avoids the potential for false alarms to trigger an 14
- unnecessary shut-down.

1

2

3

4

5

6

7

15 3.2.3.2.2 THE NOVEMBER 2021 FLOOD

- In 2021, a flooding river left a portion of the T-South pipeline submerged and undercut. Enbridge 16
- restricted the flow on T-South, resulting in FEI losing 175 TJ of supply for the Lower Mainland.¹⁰⁴ 17
- 18 FEI relied on market area storage and on-system LNG to make up for the T-South supply loss.
- 19 As the figure below shows, the pipeline was at risk. Exponent has identified a number of locations
- 20 on T-South that are susceptible to natural hazards and has factored those into its risk assessment.

¹⁰⁴ 175 TJ is approximately 30 percent of the supply required on November 16, 2021.





Figure 3-10: T-South Pipe Partially Submerged in Coquihalla River

2

1

3 3.2.4 Exponent's Unmitigated Probability-Adjusted Risk Results

The following figures from the Exponent Report show probability-adjusted risk for all AVs (the T-South AVs 1, 2, 3 and 54 are shown combined) using standard consequence measures and the 67-year and 23-year horizons.¹⁰⁵ The probability-adjusted risk associated with a winter no-flow event on T-South is very significant and exceeds the risk associated with any other AV by a wide margin, regardless of the consequence measure or time horizon. For instance:

- 9 The expected 23-year winter-only GDP loss for T-South is approximately 14 times greater
 10 than the next largest loss (AV-18);
- The expected annual winter-only loss associated with T-South is over eight times greater
 than the <u>combined</u> expected annual winter-only loss of all other AVs;¹⁰⁶ and

¹⁰⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 23-28 (pp. 93-98).

¹⁰⁶ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 245.



If the next largest AV (i.e., AV-18) is excluded, the expected annual winter-only loss on the combined AV-1, -2, -3, and -54 (i.e., T-South) is more than 30 times the <u>combined</u> expected loss on the remaining AVs.

These results do not capture cybersecurity risk, nor do they reflect the potential health and safety impacts, such as any increased incidence of poor health or mortality associated with cold residences and workplaces. They also do not account for the potential for an unexpected shift of space and water heating from natural gas to electricity during winter conditions to cause an electric system outage. While these factors are not readily quantifiable, they are real and suggest that the calculated risk is understated.

10Figure 3-11: Expected 23-year Winter-only GDP Loss for the Combination of AV-1, -2, -3, and -5411and for Other AVs for the Tilbury Baseline Scenarios



13Figure 3-12: Expected 23-year Winter-only Customer Outage-days for the Combination of AV-1, -2,14-3, and -54 and for Other AVs for the Tilbury Baseline Scenarios



12



Figure 3-13: Expected 23-year Winter-only Customer Outages for the Combination of AV-1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios



4 Figure 3-14: Expected 67-year Winter-only GDP Loss for the Combination of AV-1, -2, -3, and -54 5 and for Other AVs for the Tilbury Baseline Scenarios



6

1

2

FORTIS BC

Figure 3-15: Expected 67-year Winter-only Customer Outage-days for the Combination of AV-1, -2, -3, and -54 and for Other AVs for the Tilbury Baseline Scenarios



Figure 3-16: Expected 67-year Winter-only Customer Outages for the Combination of AV-1, -2, -3,
 and -54 and for Other AVs for the Tilbury Baseline Scenarios





3.2.5 Prioritizing Mitigation of T-South Risk Reflects Sound Risk Management Practices

Exponent discussed risk management considerations in Section 10 of its report. Its expert advice
regarding standard risk management practices supports FEI's decision to invest in mitigating
FEI's largest outage risk – the risk associated with a winter no-flow event on T-South or
unexpected supply loss due to extreme weather events.

7 3.2.5.1 Exponent Recommends Targeting Resiliency Investment at Largest Risks

8 Exponent recommended prioritizing mitigation of the highest risk AVs, and T-South in particular:

9 It is a good industry practice for a Resiliency Risk Assessment Plan to address the 10 System Level Risks in a Top Down manner. In order to conserve and prioritize 11 allocation of risk assessment effort resources to the most critical areas, a good risk 12 assessment often begins with a prioritized screening of high-interest 13 vulnerabilities. Such an effort to identify high-interest vulnerabilities and 14 performance of a prioritized screening allows the identification of areas where 15 subsequent and detailed risk assessment efforts should be focused.¹⁰⁷

16 ...

Per general good industry practices, subsequent to a proactive risk assessment
 evaluation, the high-risk scenarios should be reduced to acceptable levels.¹⁰⁸

19

20 As shown in Chapter 7, AV-1 has the highest associated risk at the status guo, i.e., 21 with the Tilbury Baseline Scenarios, followed by AV-54, AV-3, and AV-2. When 22 considered in combination, AV-1, -2, -3, and -54 have the highest associated risk 23 at the status quo. The risk of AV-1 alone and the risk of AV-1, -2, -3, and -54 in 24 combination are significantly higher than the risk associated with the other AVs 25 included in this study: the expected annual winter-only loss on the combined AV-1, -2, -3, and -54 is more than eight times the combined expected annual winter-26 27 only loss of all other AVs in the status quo scenario. Therefore, prioritizing these 28 AVs for mitigation would be reasonable. In general, mitigating assets in 29 descending order of risk - that is, mitigating the highest-risk assets first - is considered an effective approach to reducing risk. Given the probability of failure 30 in combination with the consequences associated with AV-1, -2, -3, and -54, 31 32 Exponent would expect the risk on these AVs to be prioritized for mitigation.¹⁰⁹

¹⁰⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 42.

¹⁰⁸ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 244.

¹⁰⁹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 245.



- Further, regarding T-South, Exponent stated: "...that there are numerous hazards that can lead to such outages [outages with substantial (multi-billion CAD) losses],
- 3 and that mitigation scenarios are hugely beneficial in reducing the risk."¹¹⁰

3.2.5.2 Mitigating the Known Catastrophic Consequences Would Reflect Sound Risk Management Even at Lower Calculated Probabilities

Exponent expressed the opinion that the known consequences of a winter T-South no-flow event
are sufficiently severe that established risk management principles would support investment to
reduce the harm even at much lower probabilities than those calculated by Exponent.

9 In respect of the T-South AVs, Exponent stated:¹¹¹

10 While there is uncertainty in the determination of failure probabilities, based on its analysis, Exponent does not consider the hazards and subsequent consequences 11 12 that can impact FEI's system and the customers it serves to be "low probability" with respect to certain AVs (-1, -2, -3, -18, and -54). There is significant benefit to 13 mitigating the consequences of a failure – this would be true even if the hazards 14 were considered to be low probability. Scenario-based analysis considers the 15 16 expected impacts of a failure, independent of the likelihood of the failure. PwC's 17 consequence analysis indicates that there are significant losses if certain AVs fail 18 (the scenario), and Exponent's analysis indicates that this loss can be largely 19 mitigated if it stems from certain hazards. It is well established that scenario-based 20 analysis is a valid approach for making mitigation decisions when the consequences of a loss are substantial, independent of the likelihood of the failure. 21 Additionally, it is common and most productive to address the largest risk first and 22 those with the highest benefit relative to mitigation cost first. Further discussion is 23 24 provided in Section 10 of this report. [Emphasis added.]

25 Exponent also stated:

1

2

26 The analysis performed in this report has revealed that there are a number of AVs that have failure modes with relatively high probabilities of occurrence, e.g., non-27 earthquake induced landslides on AV-1. As results have been bounded, it is 28 29 possible that the hazard-specific failure rate for an AV is towards the lower end of the bounds. However, based on its analysis, Exponent does not consider the 30 31 hazards present on FEI's system to be "low probability." Furthermore, AVs are 32 typically subjected to multiple hazards, and the cumulative rate of failure of an AV 33 (i.e., the rate of failure considering all applicable hazards) will naturally be higher than the rate of failure due to any one of the applicable hazards. When the rates 34 35 of failure for different AVs are considered collectively, the cumulative rate of failure of the combination of AVs may be even higher – as is the case for the combination 36

¹¹⁰ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 243.

¹¹¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 22.



- 1 of AV-1, AV-2, AV-3, and AV-54. <u>Thus, there is reason to perform mitigation</u> 2 <u>activities, as it is foreseeable that a failure could occur, and the consequences of</u> 3 <u>a failure are substantial. This holds true even in cases where the rate of failure of</u> 4 <u>an AV is low. ¹¹² [Emphasis added.]</u>
- 5 Exponent has referenced academic literature and provided industry examples in support of this6 view in Section 10 of its report, including the following:
- 7 There is large uncertainty in predicting low-probability, high-consequence events 8 because observations of such events are very sparse, and the observational time 9 span is typically not long enough. Therefore, the distributions fitted to the observed 10 data tend to fit the central tendencies of the data, but may underestimate the rare 11 tail events, whereas distributions fitted to the extreme tail events must contend with 12 a very small number of available observations. The large uncertainty in the occurrence of the tail events may lead to hazard and risk estimates that are highly 13 14 sensitive to the distribution parameters and modeling assumptions.¹¹³
- 15

. . .

- 16 Standard probabilistic risk assessment may underestimate the impacts of rare events because of the difficulty to estimate their probabilities and quantify their 17 18 impacts, and the sensitivity of the risk to the variables of the hazard and modeling 19 decisions. In addition, some studies suggest that risk analysis based on fat-tailed 20 power laws may still underestimate rare risk events. Specifically, the presence of 21 outliers (defined as extreme events which may be significantly larger than the 22 predictions of power-law distributions) has been documented. Those are events 23 that are sometimes referred to as Black Swans or Dragon Kings, and have been identified in nuclear accident datasets, and the magnitude-frequency distribution 24 of earthquakes in localized regions in southern California.¹¹⁴ [Emphasis added and 25 26 references omitted.]
- Exponent's recommendations regarding the approach to high consequence events is aligned with
 those of the three other experts who provided evidence on this point JANA, PwC and
 Guidehouse. JANA stated, for instance:¹¹⁵
- When we land in Quadrant IV [limited knowledge, unpredictable timing and location of event, high consequences], what we must do is 1.) Accept that we cannot predict what will happen, or when; 2.) Reject all narratives and projections that try to tell us what will happen and when; and 3) Work towards mitigating the consequence of such an occurrence.
- 35 Please refer to Section 3.7 of the 2024 Resiliency Plan for a summary of the other experts' views.

¹¹² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 231.

¹¹³ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 234.

¹¹⁴ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 239.

¹¹⁵ Exhibit B-32, BCOAPO IR2 2.3.



1 3.3 BASE PLANT END-OF-LIFE JEOPARDIZES ABILITY TO SERVE CUSTOMERS 2 IN NORMAL CONDITIONS

The Base Plant has played a critical role in FEI's supply portfolio since 1971, including providing peaking supply in normal operations. FEI, like other utilities throughout western North America, relies on on-system LNG because it has a shorter response time than off-system capacity resources and deliverability does not depend on third-party infrastructure. FEI has also relied on it during less severe supply disruptions and constraints, and for routine operations support, to sustain service to customers.¹¹⁶

9 The Adjournment Decision sought additional information on the remaining life of the Base Plant,¹¹⁷ 10 which FEI provides below. Additional engineering analysis on the Base Plant, and its recent 11 deteriorating performance despite further investment, indicate that the Base Plant has reached 12 end-of-life. Continuing to operate this critical supply portfolio asset longer than the time it takes to 13 construct a replacement facility would jeopardize FEI's ability to meet peak loads in normal 14 operations. There is no feasible option to extend the life of the Base Plant or replace its peaking 15 supply in the market. A sizable capital investment is required, irrespective of resiliency 16 considerations.

17 3.3.1 Equipment Is Obsolete, Is Experiencing Increasing Failure Rates and Is 18 Increasingly Difficult to Maintain or Repair

19 The Base Plant houses the only regasification capacity at Tilbury (there is no regasification 20 equipment in Tilbury 1A). It is connected to the Base Plant storage tank and FEI's transmission system through interconnection piping.¹¹⁸ The Base Plant equipment is only called on to function 21 22 at times where the supply is necessary to meet customer demand, such that its reliability is 23 critically important to avoid curtailment of firm load. As discussed below, despite investment in 24 recent years, the Base Plant equipment has been experiencing unpredictable failures consistent 25 with equipment that is end of life. The regasification equipment is obsolete, has been experiencing increasing rates of failure and reliability issues, and is difficult to maintain or repair, as further 26 27 explained below. Its deteriorating condition is compromising both existing peaking gas supply and 28 resiliency. FEI is already decommissioning the Base Plant liquefaction equipment.

¹¹⁶ Please refer to Section 4.4.1.5.4 of the Application. Additionally, Tilbury provides operations support for unplanned activities. For example, both Tilbury and Mt. Hayes were used to send out gas on January 24, 2024 in response to an East Kootenay Exchange (EKE) compressor outage. FEI's LNG facilities helped make up supply that was lost due to this unforeseen compressor outage given FEI could not source more gas from Trans Canada pipeline.

¹¹⁷ Adjournment Decision, p. 14. "FEI states that "even with significant additional capital investment, the extent of additional operational life that FEI would be able to achieve is unclear". Given this uncertainty regarding the cost of extending the life of the existing tank and the amount of extended life that can be achieved, the Panel is unable to assess the cost effectiveness of the TLSE Project as a replacement for the Base Plant." See also p. 51.

¹¹⁸ FEI constructed the interconnection piping in response to the 2018 T-South Incident and the subsequent 12 to 18 months of supply concern from T-South (pressure restriction on the failed line to enable subsequent integrity work). The interconnect enables FEI to access the Tilbury 1A storage and liquefaction capacity (when capacity is available) at the time of an emergency. Additionally, as explained in the Application, FEI determined that utilizing Tilbury 1A liquefaction capacity to fill the Base Plant tank would be more efficient and cost effective.



1 *3.3.1.1* Regasification Equipment Technical and Reliability Concerns

- 2 The Base Plant regasification equipment is comprised primarily of send-out pumps (pumps that
- 3 move the LNG from the tank to the vaporizers), vaporizers (units that convert the liquid to a gas)
- 4 and ancillary equipment such as power supply and utilities. The discussion below focusses on
- 5 the issues with these two primary components (i.e., send-out pumps and vaporizers).

6 3.3.1.1.1 SEND OUT PUMPS ARE EXPERIENCING ABOVE NORMAL FAILURE RATES

- 7 The send-out pumps and motors are a key component of the regasification system and have been8 experiencing unplanned outages more frequently.
- 9 The Base Plant is designed with four send-out pumps; however, currently only the "A", "B" and 10 "C" pumps can run in parallel. The "D" pump does not have the same performance curve, and 11 therefore it does not function as a spare for the other three pumps. Each send-out pump consists 12 of a motor and a pump with common suction piping, all of which is tied to a common pipe. These 13 pumps pull LNG from the Base Plant tank, and push it into the vaporizers, where it is gasified and 14 pumps diate the distribution are pipeling. Functioning and out pumps on a second to the page.
- 14 pushed into the distribution gas pipeline. Functioning send-out pumps are essential to the Base
- 15 Plant being able to send out gas into the system.
- FEI is currently working to upgrade pump "D" so that it can run in parallel with the other three pumps and function as a standby spare with roughly 90 percent of the capacity of the other three pumps. This upgrade is expected to be completed in the fall/winter of 2025. Currently there is no redundancy, which elevates reliability risk. For example, without the partial redundancy provided by the fourth send-out pump, the loss of a single pump would reduce the send-out capacity provided by the Base Plant from 150 MMcf/d to approximately 100 MMcf/d, and even lower if multiple failures occur simultaneously.
- The three functioning send-out pumps (Pumps "A", "B" and "C") suffer from various failure modes, including process seal failures, freezing in place during cool down, and high vibrations due to their long narrow shafts and lack of pump bearings. A more modern pump of similar capacity would benefit from a shorter, wider shaft, higher efficiency impellers achieved through modern manufacturing processes, and modern process seal technology.
- The three functioning send-out pumps have experienced a higher-than-normal failure rate over the past three years, including the following failures:
- August 2020: Pump "C" was removed, as it was seized. The pump was repaired and reinstalled.
- November 2021: Pump "A" seized during cool down. The pump was warmed up, and subsequently cooled down again. The pump unseized and was restarted.
- December 2022: The Pump "A" motor experienced high vibrations and subsequent arcing
 of the windings. As a result, FEI overhauled the pump motor, including rewinding the stator
 and machining the shaft.



• **August 2023:** Pump "B" had a seal failure which was repaired by the Original Equipment Manufacturer (OEM).

3 • November 2023:

1

2

- Pump "A" had a higher-than-normal oil leak into the pump. In order to remain
 operational, the oiler required top up approximately every 4 hours to keep the
 pump running. This has subsequently been repaired.
- Pump "B" had high bearing temperatures. The pump could only be operated for 15
 minutes before it had to be shut down. Repairs to this pump have been completed.
- 9 o Pump "C" seized on cool down due to a faulty temperature probe indicating a temperature higher than actual. In particular, the pump cooled down too quickly, resulting in the pump freezing in place. The pump worked normally after warming and cooling down again.

Given the prevalence of send out pump failures over the past three years, even with plans for partial redundancy of a fourth send out pump, the reliability of the regasification equipment increases the risk to FEI's customers when relying on the Base Plant for peaking supply and resiliency purposes.

17 3.3.1.1.2 VAPOURIZERS ARE INCREASINGLY UNRELIABLE AND INCLUDE OBSOLETE TECHNOLOGY THAT 18 CANNOT BE REPLACED

There are four 50 MMcf/d vapourizers at the Base Plant, including three operating vapourizers and one spare vapourizer (i.e., a maximum of three can operate at any one time -3×50 MMcf/d = 150 MMcf/d). While FEI cycles through the vapourizers, and plans maintenance cycles accordingly, the vaporizers have nonetheless become appreciably unreliable. There are several concerns regarding the condition of the vapourizers and FEI's ability to continue maintaining them.

- Over the past five years it has become common to take multiple attempts to start the vapourizers when there is a demand for regasification. At times, this has delayed the ability to send out by up to 4 to 8 hours. FEI has identified a few potential causes for this lack of reliability and has taken steps to try to address them (e.g., upgrading the ignitors, electrodes, transformers, and bench testing them annually). FEI will continue to monitor these steps taken to improve the reliability of the vapourizers.
- Three of the four vaporizers are manufactured from carbon steel and suffer from corrosion
 in the water/glycol bath. FEI monitors the corrosion and completes spot repairs over time.
 As the corrosion continues and affects more of the vapourizer metal, FEI expects more
 unplanned repairs. The corrosion is affecting many aspects of the vapourizers. For
 instance:
- In the most recent inspection, three of the four vapourizers had significant coating
 failures on the bath walls and floor, which requires removal via blasting, followed
 by recoating;



1 2	0	Seven of the 16 wall and floor supports on Vapourizers "A" and "C" have broken and have been weld repaired;
3	0	The walls of all four vapourizers are bowed out of shape due to thermal cycling;
4 5	0	The corner vertical seam on one vapourizer has cracked, requiring a 4-foot vertical weld repair; and
6 7 8 9	0	The outlet nozzle on one vapourizer has corroded through to atmosphere and currently contains a temporary patch. Two of the other vapourizers have outlet nozzles that are nearly corroded through. A permanent weld repair is required on these three vapourizers at the next thorough inspection.
10	The issues outlined above are indicative of the types of failures that can be expected as the	

11 vaporizers are operated well-beyond their design life. As they age, the vaporizers will very likely

12 see increased rates of these types of issues. Even with ongoing monitoring, predicting when a

13 failure might occur is difficult and it is inevitable that reliability will decrease over time.

The existing vapourizers are tuned each year to ensure an adequate fuel-to-air ratio. Whereas modern vapourizers are tuned via control valve positioners and Programmable Logic Controllers (PLCs), the existing vapourizers require tuning via a mechanical linkage connecting the air and fuel inlet valves. The mechanical linkage makes tuning the vapourizers difficult as very small changes to the linkage orientation (e.g., loosening joints or changing valve friction while stroking) have adverse effects on the combustion efficiency.

20 The following photos show examples of corrosion. As noted above, as the vaporizers age, FEI

21 expects the frequency of these incidents to increase as the cumulative corrosion increases and

22 distortion of the bath walls continues. As the frequency of failures increases so too does the need

23 to replace the vapourizers to improve reliability.





Figure 3-17: Severe Corrosion in Vaporizer Stack A

Figure 3-18: Typical Coating Failure on Floor of Vaporizer Bath







Figure 3-19: Example of Coating Failure on Walls of Vaporizer

Figure 3-20: Discoloration and Bowing of the Bath Walls



4

- 5 3.3.1.1.3 PLANNED AND UNPLANNED MAINTENANCE RESULTS IN EXTENDED DOWNTIME
- 6 The problem of more frequent failures of the send-out pumps is compounded by the challenges
- 7 with performing emergency repairs or maintenance quickly.



1 The Base Plant has limited isolation valves, making it difficult to complete emergency repairs and 2 maintenance tasks quickly. When compared to a modern plant, the Base Plant has large sections 3 of piping and equipment which must be warmed up to achieve a safe isolation on any piece of 4 equipment. This warming up and subsequent cooling down adds 6-10 days to the repair duration. 5 The lengthy time to conduct repairs applies to nearly all of the regasification system, but especially 6 to the pumps, motors, piping and valves that interconnect the tanks with the regasification 7 equipment. Further, if a failure or breakdown occurs during or just prior to a resiliency event, this 8 effectively renders the Base Plant useless for the event given the antiguated design and 9 challenges in completing repairs in a timely manner. FEI provides examples of the impact of

- 10 limited isolation valves below:
- 11 In the case of the send-out systems, the limited isolation valves result in long timeframes 12 and significant costs to complete emergency maintenance or repairs. In the absence of isolation valves, performing maintenance on any of the pumps or motors (including 13 14 bearing, seal, wear ring replacements or impeller/inducer inspection/repairs), necessitates 15 first warming the entire send-out system and the interconnected piping, requiring purging 16 of LNG gas vapours. Cooling down the interconnect and send-out systems following 17 maintenance requires bringing in a liquid nitrogen injection truck at an average cost of \$150,000.¹¹⁹ The full cycle to warm up the equipment and then cool it back down to put it 18 19 back into service can take up to 10 days, (excluding the time to repair the damaged 20 equipment, which may take longer than 10 days, depending on the repair scope).
- 21 Inadequate isolation points for each valve make it difficult for FEI to remove and recertify the Base Plant's 44 Pressure Safety Valves (PSVs) that form part of the regasification 22 23 system without venting natural gas to atmosphere. FEI removes and recertifies the PSVs 24 approximately every 24 months (and no longer than every 30 months), as per CSA Z276. This work is necessary to avoid failures and the risks of leaks, which can cause the Base 25 26 Plant to be out of service for 10 days. A typical new facility includes two isolation valves 27 immediately upstream of each PSV being removed, allowing for quick isolation, and 28 minimal venting.

FEI has evaluated the potential to install additional isolation valves to reduce the down time required for maintenance activities on the regasification system. Given the age of the Base Plant, the piping around the send-out system does not have sufficient room to allow installation of any additional valving. Retrofitting the system would require reconfiguring the regasification piping and surrounding infrastructure. Given the space limitations and the challenges with the inherent design, this would require a very substantial outage and is not a practical solution.

35 *3.3.1.2* Base Plant Liquefaction Equipment is Being Decommissioned

The Base Plant was constructed in 1971 with 5 MMcf/d of liquefaction capacity. The equipment is obsolete, and its reliability declined appreciably over the past decade. In order to address the issue, FEI interconnected the Base Plant and Tilbury 1A tanks so that FEI can use 5 MMcf/d of

¹¹⁹ The cost depends on a number of variables including weather conditions and duration.



1 Tilbury 1A liquefaction to fill the Base Plant tank.¹²⁰ Sustaining the obsolete Base Plant 2 liquefaction equipment is no longer practical, and FEI recently ceased to use the equipment 3 altogether

- 3 altogether.
- 4 In 2022, FEI began the process of decommissioning the Base Plant liquefaction equipment and
- 5 laying it up for safe long-term storage and eventual demolition. This development has not eroded
- 6 FEI's gas supply and resiliency, given FEI's ability to rely on Tilbury 1A liquefaction. However, the
- 7 deterioration of the Base Plant liquefaction equipment is symptomatic of the age of the Base Plant.

8 3.3.2 Seismic, Environmental and Flooding Issues Are Inherent in the Base 9 Plant Design

As described below, the Base Plant was constructed to lower engineering and safety standards that were in place at the time of construction. Even if the regasification equipment was to be replaced, there are seismic, environmental and flooding issues inherent in the original Base Plant design. FEI now operates the tank well-below its design capabilities for seismic reasons, and experts have advised against tank retrofits to restore those original capabilities.

3.3.2.1 The Base Plant Tank Was Designed to Lower Seismic Standards and *Issues Cannot Be Solved with Retrofits*

17 The Base Plant tank was constructed beginning in 1969, at a time when seismic standards were 18 much lower than today. Over time, changes in standards have prompted improvements, as well 19 as multiple reductions in the tank fill elevation to mitigate identified seismic risks. This section 20 discusses the latest seismic studies that caused FEI to operate the tank at 0.35 Bcf, and the new 21 engineering advice against attempting tank retrofits to restore its design capabilities.

22 3.3.2.1.1 RECENT SEISMIC ASSESSMENTS HAVE PROMPTED REDUCED LNG FILL ELEVATION

At the time of construction in 1969, 1,600 timber piles were installed under the tank in order to improve the ground it was located on and guard against unwanted movements and displacements during a seismic event. The tank was constructed as a single wall vessel and an earthen dike was constructed to act as secondary containment for the LNG in the event of a breach of the tank. Since that time, understanding of seismic risk has advanced, resulting in more stringent design requirements and prompting actions by FEI.

- 1981: A new study was completed based on the 1978 version of the CSA Z276 standard for LNG facilities. As a result of this study, in 1983 additional ground improvements were completed around the tank to help mitigate soil liquefaction during seismic events. In addition, the earthen dike was abandoned and a new concrete containment wall was built around the tank.
- **1996:** A further code review was completed based on the CSA Z276-1994 standard. As a result of this review, the maximum liquid level in the tank was reduced to 26 metres (or 95

¹²⁰ See Exhibit B-1-4, p. 62 for further discussion.

7



- percent of the tank level) to prevent spillage and protect the integrity of the tank during a
 seismic event.
- 2018: FEI operated the tank at the reduced 95 percent level until 2018 when further studies were initiated to check the tank performance against the 2015 version of the CSA Z276 standard and to review the performance of the tank related to any potential breach (in particular the fire suppression system). Those reviews identified the following:
 - The firewater foam system was inadequate to combat a fire within the containment walls from a full breach of the tank.
- 9 o The original ground improvements completed at the time of the tank construction 10 and subsequently in 1983 would be inadequate based on current day code to 11 prevent large differential settlements between the tank and the connecting piping 12 and equipment. As a result, even with operating the tank at the reduced liquid 13 levels there remains a loss of containment risk in the event of a seismic event.
- 2020 / 2023: In light of the findings in the 2018 reports, FEI completed updated studies in
 2020 and 2023 (by CB&I, an industry expert in the design and construction of tanks) to
 assess the performance of the Base Plant tank against current engineering standards for
 seismic design. Similar to the situation in 1996, when FEI reduced the fill level of the tank
 to 95 percent to reflect then current day seismic standards, these engineering studies
 based in current standards have prompted FEI to operate the Base Plant tank at 58
 percent capacity.
- CB&I issued its first report in 2020 (2020 CB&I Report), noting that a fill elevation
 in the tank of 16 metres (versus design level of 26 metres) is required to limit
 stresses to acceptable levels for the inner tank, prevent the ring wall foundation
 from unloading, and to comply with current day seismic requirements.¹²¹ A 16
 metre fill elevation equates to 58 percent capacity, or 0.35 Bcf.
- A further study completed in 2023, also by CB&I, included a more detailed assessment of the tank (utilizing finite element analysis). This study confirmed that operating the tank above 16 metres could cause the tank's ring wall to unload under the CSA Z276-2015 geotechnical and seismic standards.¹²² The report also noted that based on even more recent code changes, as well as improved understanding of how the ground at the Tilbury site would act during a seismic

¹²¹ Confidential Appendix D, p. 3: "CB&I is in the process of performing a more rigorous evaluation of the singlecontainment BPT. In 2020, CB&I performed a review of the BPT using current-day seismic levels and code requirements which indicated that it should only be operated at a capacity of ~59% (16 m product height) of the original tank design capacity."

¹²² Confidential Appendix D, p. 6: "Under the 16.0 meters of product with 100% horizontal SSE and 40% vertical SSE, the bearing pressures beneath the ringwall based on the ANSYS analysis and original seismic spectra are comparable with hand calculations. In this condition, the foundation is on the verge of unloading as previously stated in 246376-000-PS-RP-00001."



1 2 event, that unloading of the foundation may occur even at the 16 metre liquid elevation.¹²³

Improved understanding of how the tank would react, combined with the tank's old single wall technology (meaning any breach of the tank would result in LNG being discharged into the environment), has led FEI to continue to operate the tank at a fill elevation of 16 metres (equivalent of 0.35 Bcf) as initially recommended in the 2020 CB&I Report.

- FEI would need to perform further risk / engineering evaluation before considering operating the
 tank at design volumes with additional risk mitigation measures. However, CB&I has suggested
 that it expects further analysis could instead result in further reductions in the recommended fill
- 10 elevation (i.e., the tank capacity could be reduced below 0.35 Bcf):
- 11 The preliminary evaluation considering the 2023 WSP response spectra indicates 12 that unloading of the ring wall is likely, even at the 16.0 m product level. As 13 discussed in 246376-000-PS-RP-00001, unloading of the foundation is not 14 covered by the API Standards <u>and is expected to worsen the results from the</u> 15 Standards-based hand calculations. An FEA would be required to determine the
- 16 full extent of the impact of the 2023 WSP response spectra.¹²⁴ [Emphasis added.]

17 3.3.2.1.2 RETROFITTING TO RESTORE THE TANK DESIGN CAPACITY IS NOT FEASIBLE

FEI retained CB&I and WSP (previously Golder, an expert in geotechnical matters) in 2023 to assess the feasibility of refurbishing the Base Plant tank to withstand minimum seismic requirements and return it to its original design capacity of 0.6 Bcf. CB&I's 2023 review focused on the tank itself, while WSP focused on the foundations. Those reports show that refurbishment of the Base Plant tank would be impractical and risky, such that FEI has concluded tank refurbishment is not feasible.

24 2023 CB&I Report Recommended Against Attempting to Remediate the Deficiencies in 25 <u>the Tank Itself</u>

- CB&I's 2023 report (2023 CB&I Report) is included in Confidential Appendix D. The section on
 Remediation Methodology describes what refurbishment would involve, as well as the associated
 ricks OP8//s hear findings included
- risks. CB&I's key findings include:
- Numerous elements of the tank would not comply with the most recent edition of API 620,
 Design and Construction of Low-Pressure Storage Tanks.

One of the keys to restoring the Base Plant tank to its design capacity is its foundations.
 The only way to resolve one of the key failure modes (unloading of the foundation) is to

¹²³ Confidential Appendix D, p. 8: "The preliminary evaluation considering the 2023 WSP response spectra indicates that unloading of the ring wall is likely, even at the 16.0 m product level. As discussed in 246376-000-PS-RP-00001, unloading of the foundation is not covered by the API Standards and is expected to worsen the results from the Standards-based hand calculations. An FEA would be required to determine the full extent of the impact of the 2023 WSP response spectra."

¹²⁴ Confidential Appendix D, p. 8.



- replace the existing foundation with one designed to accommodate the higher loads. CB&I
 did not comment on how to replace the foundation, as FEI retained WSP for that purpose.
- The existing tank also has deficiencies with the anchor straps holding the inner tank, the
 compressive strength of the outer tank concrete wall and the outer tank anchor strap
 attachments.
- 6 The above is not an exhaustive list of components that would require upgrading.
- CB&I identified a potential scenario to repair the tank components (not the foundations) but noted
 it would involve significant risk. It recommended against undertaking the repairs. CB&I stated the
- 9 following in the 2023 CB&I Report (Section 2.0 (Contractor Opinion)):
- 10 CB&I believe that performing the modifications and repairs is possibly achievable,
- however, the repairs and modification process is fraught with significant risk. 11 Safety and environmental risks aside, the consequences of noted, but not limited 12 13 to risks would likely jeopardise the intended purpose of the repairs / modification, 14 which is to bring the tank up to current day standards and seismic conditions and 15 provide full capacity utilization. Moreover, even if possible, and in the unlikely event that all risks are mitigated to inconsequentiality, after the tank modifications are 16 17 carried out the tank may still not be up to current day standards for siting of a single 18 containment tank at this location (to be determined by Owner / Regulator / Permit 19 issuer).
- 20 It should be noted that the following discussion about refurbishment of the Base 21 Plant Tank assumes that all tank components are in as-new condition. 22 Components being in as-new condition is unlikely given that tank was placed into 23 service ~52 years ago, circa 1971. Any component/s that are discovered to have 24 deteriorated will need to be repaired or replaced accordingly. Any deteriorated 25 components that are not discovered may potentially affect the BPT longevity and 26 operability of the tank. Further, after any potential remediation CB&I would not 27 warrant the tank or operability due to remediated condition or pre-existing or latent 28 unknown conditions in the existing structure.
- With the risks of the repair and unknowns CB&I would suggest that this path not
 be followed. [Emphasis added]

31 <u>WSP 2023 Report Confirmed That the Base Plant Tank Will Fail Under Current Seismic</u> 32 <u>Loads Without Ground Improvements</u>

33 WSP's 2023 report on the foundation (2023 WSP Report) is included as Confidential Appendix E.

Even if all the tank repairs identified by CB&I could be completed as planned, the foundation

would have to be replaced or the tank would still fail under seismic loading derived from recent

36 codes and standards. The potential options that WSP identified would not be cost-effective or

37 feasible.



1 WSP found that based on previous reports and engineering efforts, the ground below the existing

timber piles supporting the Base Plant tank would likely experience soil liquefaction and as a
 result, post-earthquake differential settlements of between 400 and 600 millimetres.

WSP considered two potential options to address this differential settlement: (1) completing ground improvements with the Base Plant tank remaining in place; or (2) completing ground improvements with the Base Plant tank being decommissioned, removed and then replaced after ground improvements were complete. WSP found that with Option 1 (leaving the Base Plant tank in place), some soil improvements could be affected around the perimeter of the tank; however, it would not be possible to improve the ground directly below the timber piles. As a result, Option 1 would not mitigate soil liquefaction potential under the tank.

11 With respect to WSP's second option, even if it were possible to safely remove and reinstall the 12 Base Plant tank, this option still presents risks because the existing condition of the timber piles 13 is unknown. This could be mitigated by installing new stone columns instead; however, this 14 installation would still present significant challenges in working around the existing timber 15 foundations. FEI does not consider decommissioning, temporarily removing, and then reinstalling 16 the Base Plant tank to be a cost effective or feasible solution given its age and the complexity and 17 risk that would be involved with this approach. The re-installed tank would still need to be operated 18 at its reduced capacity, and the Base Plant would still be exposed to the other challenges outlined 19 above.

FEI will continue to operate the tank at its reduced capacity of 0.35 Bcf with ongoing evaluation to determine if further derating is necessary depending on the tank condition and any future changes to seismic codes.

23 3.3.2.2 The Base Plant Tank's Design Entails Higher Environmental Risk

From an environmental perspective, the Base Plant tank design does not meet current design expectations as it relates to emissions.

26 The Base Plant tank was constructed with one boil off gas compressor to manage pressure build 27 up within the tank (boil off gases). In addition, the tank was designed to operate under a very 28 narrow pressure range. Under normal operations, this compressor manages the pressure in the 29 tank within the design ranges and captures the boil off gases and sends them back to the pipeline. 30 However, even under minor upset conditions or during periods of maintenance for the 31 compressor, the pressure can build up in the tank beyond the design range and is released to 32 atmosphere through a vent at the top of the tank. This design configuration was common practice 33 for tanks built in the 1970s; however, current day standards require multiple boil off gas 34 compressors (to provide redundancy), and include a wider range of design pressures to avoid 35 venting boil off gases to atmosphere.

Similarly, in the event of a breach of the tank, the current standard is to have full secondary containment built within the tank itself, unlike the Base Plant tank which has open air secondary containment. If a leak were to occur from the Base Plant tank, the LNG would collect within the



- 1 open-air secondary containment and would vent to atmosphere as it changes from a liquid to a
- 2 gas. Conversely, a new TLSE Project tank would have full secondary containment (as does the
- 3 existing Tilbury 1A tank), preventing methane venting to atmosphere in the event of a breach.

4 3.3.2.3 Plant Was Built at an Elevation that is Susceptible to Flooding

- 5 The Base Plant was constructed at an elevation that makes it susceptible to significant damage 6 and service disruption in the event of any flooding. Retrofits to address this risk are not practical.
- Flood modeling completed in this section of the Fraser River indicates that the Base Plant's elevation makes it susceptible to flooding. Specifically, if a flood event were to occur, there is a very high risk that the Base Plant control room and other process buildings would be damaged. This includes the electrical transformer and its connections to the process equipment. When constructing the new Tilbury 1A facility, the City of Delta required FEI to raise the elevation of the site by 3 to 3.5 metres to ensure that in the event of a flood the new equipment would remain operational.
- 14 The damage from a flood affecting the control room and process equipment would likely interrupt
- 15 FEI's ability to rely on the Base Plant for months to allow for damage assessment and delivery of
- 16 replacement parts for the distributed control system (DCS), uninterrupted power supply (UPS)
- 17 and motor control center (MCC). During that period, FEI would have no ability to send out gas
- 18 from Tilbury since the Base Plant houses the only regasification equipment at Tilbury.
- 19 It would not be possible to raise the elevation of the ground under the Base Plant; therefore, the 20 Base Plant equipment would remain susceptible to damage from flooding. Moreover, trying to 21 raise the aging equipment would be complex, risky and very costly. Keeping the Base Plant 22 operational during the work would also be impractical. It would require constructing new 23 temporary facilities at a different location to maintain the functionality of the plant. The existing 24 facilities would then need to be removed, the elevation in that area raised to the new height, and 25 then the existing facilities would need to be reinstalled. The temporary facilities could then be 26 removed.

27 **3.3.3 Long Replacement Project Lead Time Increases Risk**

- In light of the long lead time to permit and construct a new tank and regasification, investment
 decisions must be made well in advance. FEI already faces a number of years of continuing
 equipment deterioration due to the lead time for the TLSE Project. Prolonging that time increases
- 31 the potential for a permanent failure or further reduction to tank fill levels to occur before a solution
- 32 is in place.
- 33 FEI estimates that it would take 6 to 8 years to permit and construct a new tank and regasification
- 34 equipment after deciding to build a new facility. FEI's 6- to 8-year estimate of the potential lead
- 35 time is based on its experience. The execution phase for the TLSE Project is estimated to be 5
- 36 years from project approval. Following approvals of a Base Plant replacement project, it would



- 1 take over 4 years to engineer, procure and construct a 0.6 Bcf tank and 4×50 MMcfd vaporization.
- 2 This is contingent on regulatory approvals, market conditions and other external factors.

3.3.4 Losing Access to Tilbury Supply Would Challenge FEI's Ability to Meet Peak Load

5 FEI's supply portolio (i.e., the BCUC-accepted Annual Contracting Plan (ACP)) has, since 1971 6 when the Base Plant was commissioned, included 0.6 Bcf of energy (storage) and 150 MMcf/d of 7 capacity (regasification) from Tilbury in its planning to meet customer demand. FEI explains below 8 why losing access to some or all of the peaking capacity and energy currently provided by Tilbury, 9 which would occur, for example, if the Base Plant regasification fails, would impair FEI's ability to 10 provide uninterrupted service to customers in winter. With today's highly constrained regional 11 pipeline and storage infrastructure, even inferior fall-back options for market-dependent peaking 12 supply no longer exist. FEI could not replace 150 MMcf/d and 0.6 Bcf of peaking supply in the 13 market. FEI does not have control over regional infrastructure expansions, and if those 14 expansions were to occur in the future they would entail very significant annual gas supply costs 15 for FEI customers.

16 3.3.4.1 On-System LNG Is a Critical Part of FEI's Gas Supply Portfolio

On-system LNG has played an important, long-standing role in FEI's overall supply portfolio (as
it does for utilities throughout the Pacific Northwest), providing critical dependable peaking supply
on cold winter days, while also backstopping other resources within the portfolio.

20 3.3.4.1.1 <u>TILBURY LNG PROVIDES PEAKING SUPPLY IN FEI'S ANNUAL CONTRACTING PLAN</u>

- FEI described how it designs its gas portfolio in the Application.¹²⁵ That evidence is summarized below.
- FEI's gas supply strategy has been to rely on physical assets (i.e., pipeline capacity and regional storage capacity) to procure supply with limited exposure to market risks such as counterparty default risk and commodity price spikes on high demand days. The physical supply available at Sumas is the Station 2 gas delivered through T-South, and the costs of Sumas winter supply are much higher than the costs of holding T-South plus Station 2 commodity. FEI currently sources the majority of its gas supply from Station 2 and AECO, and only includes limited Sumas supply as contingency resources in the ACP portfolio.
- As the figure below illustrates,¹²⁶ FEI's resource portfolio includes pipeline, market area storage and on-system LNG storage for peaking. Each asset or resource, with different supply durations (i.e., energy) and daily deliverability (i.e., capacity), provides supply at different times of the year to meet seasonal gas demand while having different attributes. Each year, FEI conducts a portfolio optimization exercise using standard gas portfolio planning software to seek an optimal

¹²⁵ Exhibit B-1-4, Application, pp. 30-31 and 79-80.

¹²⁶ The specific volumes are commercially sensitive but are available in Figure 3-4 from the 2023/24 Annual Contracting Plan.



- portfolio to meet requirements based on updated inputs (i.e., forecast demand). The ACP sets
 out the annual requirements of each type of supply resource to optimally match FEI's load profile.
- 3 In general:
- Since pipeline capacity must generally be purchased for longer durations (typically, 365 or 151 days), FEI tends to use pipeline supply for loads that require service year-round and the majority of the winter. FEI's pipeline capacity holdings are comprised of many individual contracts with different volumes and renewal dates, which gives FEI significant flexibility to recontract or de-contract resources as load changes over time.
- Market area storage provides supplemental supply during the cold winter periods (i.e., 10 to 60 days of demand). FEI employs the same contracting strategy as with pipelines, entering into multiple storage agreements with different volumes and renewal dates.
 These assets are also used to balance the system by injecting and withdrawing gas to and from storage throughout the year.
- On-system LNG peaking supply is typically reserved for peak day (i.e., 1-10 days of demand) and cold winter days because it provides additional <u>capacity</u> (i.e., daily deliverability, determined by regasification output) on system which reduces the needs for pipeline supply during cold winter days. This capacity is backed by <u>energy</u> (i.e., LNG in storage) with the on-system <u>energy</u> from LNG being a small, but crucial, supply to meet overall energy requirements in peak periods. Tilbury LNG has been a critical peaking supply asset to serve Lower Mainland gas customers since it was built in 1971.

As shown in the Figure 3-21 below, FEI develops a mix of resources in the gas supply portfolio to meet forecast demand. Through the development of the ACP, FEI contracts for resources to meet daily demand, taking into consideration security, diversity and reliability of supply while minimizing the overall cost of the portfolio. FEI's ability to meet the objectives within the ACP is advanced by continuing to have access to, and flexibility around, each of the three resource types listed above.







2

1

3 3.3.4.1.2 ON-SYSTEM LNG'S UNIQUE ATTRIBUTES

On-system LNG has several unique attributes that make it a critical part of FEI's supply portfolio,
as well as the supply portfolios of other regional utilities.

- 6 On-system LNG provides not only energy (peaking supply, measured in Bcf of LNG) but 7 also supplemental capacity (measured in MMcf/d and determined by regasification output) 8 which does not require upstream pipeline capacity that comes with annual tolls when the capacity is only required for a short duration. Holding pipeline capacity that is only required 9 10 during a few cold winter days each year necessitates significant mitigation activities during 11 the remainder of the year to attempt to offset fixed demand costs. FEI's gas supply 12 portfolio is already subject to considerable cost mitigation risk given the load profile of the 13 customer base (i.e., RS 1 has an approximately 30 percent load factor, meaning the profile is very "peaky"). Further, as discussed below, additional pipeline and regional storage 14 15 capacity are no longer available given that facilities are fully contracted.
- On-system LNG allows FEI to access energy on short notice, responding to rapid weather
 changes or urgent operational needs. By contrast, upstream pipeline supply requires
 scheduling 24-48 hours in advance due to commercial transactions or business rules on
 pipelines, and storage supply requires scheduling and may be constrained by the
 operating conditions of storage and the associated pipelines.
- On-system LNG storage is commercially valuable because the gas distribution utility (in this case, FEI) has full control of the supply without relying on counterparties and third-



party assets (e.g., T-South) for delivery during cold weather when marketplace conditions
 are volatile. In FEI's experience, transacting commercial deals to receive a significant
 amount of supply required on cold days is not only costly but also risky given the market
 that FEI operates within. The key risks are:

- 5 1. The difficulty of securing these types of deal structures with a long-term 6 commitment given the constrained regional gas supply resources (i.e., term of the 7 deal); and
- 8
- 2. Execution failure, in which parties are not able to deliver their commitments.¹²⁷

9 On-system LNG resources can backstop other supply resources that may be subject to 10 planned maintenance or unplanned outages. It is not uncommon to have multiple 11 operational issues during cold weather events. For example, during a recent cold snap in 12 January 2024 in the Pacific Northwest region, both TransCanada (i.e., Foothills) and an outage at the Jackson Prairie Storage (JPS) facility required additional natural gas to be 13 14 sent out from Tilbury and Mt. Hayes during the cold event. Because of the unexpected 15 JPS outage, mutual aid across the pacific Northwest was activated, and FEI, given the on-16 system LNG resources, was able to send out extra supply from the Mt. Hayes LNG facility 17 and Tilbury to help maintain the pressure on the T-South system so gas could flow south. 18 This ensures that downstream utilities received enough supply to meet their load 19 requirements. On-system LNG provides additional supply and operational flexibility to the 20 regional gas system.

For many years, through FEI's ACPs, FEI has received the BCUC's support for the strategy of holding on-system LNG and other physical assets to ensure reliable supply – which is critical during cold and extreme weather events when there is considerable market uncertainty.

24 This approach of using on-system physical assets to de-risk and optimize the portfolio is common 25 in the Pacific Northwest region. As shown in Figure 3-22 below, there are a number of LNG facilities¹²⁸ in the region, including FEI's Mt. Hayes facility and Tilbury, with all but one owned by 26 27 local distribution companies (LDC).¹²⁹ These other LNG facilities are designed to serve the last 28 measure of demand on the very coldest days of the year, and LDCs typically reserve LNG 29 capacity to serve their gas and/or electric customers. Mr. Raymond Mason, an expert in the 30 regional gas supply market who FEI retained to opine on the role of on-system LNG storage in a supply portfolio, explains:¹³⁰ 31

LNG peak-shaving plants ensure that adequate supplies of natural gas are readily
 available when demand is at its peak. Throughout North America, natural gas
 transmission pipeline operators and/or utilities, use these facilities to liquefy natural

¹²⁷ In the early 2000s, FEI experienced a counterparty failing to deliver gas during a cold weather event, which was transacted through a commercial arrangement. FEI used LNG from Tilbury to replace the lost supply.

¹²⁸ Northwest Gas Association 2022 Outlook, p. 16: <u>054dfe_207b3155de904ebb8d4513ef2790cfb9.pdf (nwga.org)</u>.

¹²⁹ i.e., NW Natural, Puget Sound Energy, Intermountain Gas and FEI.

¹³⁰ Appendix F, Raymond Mason Report, pp. 19-20.



- gas for storage. When demand is lower (i.e., typically in the summer), the operator
 can liquefy natural gas for above ground storage and then regasify the LNG when
 demand is high.
- 4 These facilities typically provide reliable supply in areas where pipeline/distribution
- 5 capacity limitations and/or weather conditions tend to cause supply and demand
- 6 discrepancies. [Footnote omitted]

Figure 3-22, which is reproduced from Mr. Mason's report, shows where LNG facilities are
 located.¹³¹

9

Figure 3-22: LNG Peak Shaving Facilities in Western Canada and the United States



10

11 3.3.4.2 FEI's Peaking Supply Requirements Already Exceed Tilbury's Capabilities

12 As discussed below, the 150 MMcf/d regasification (i.e., daily deliverability) and 0.6 Bcf at Tilbury

13 reflected in FEI's ACPs are already insufficient based on current peak load. In recent years, the

14 BCUC-accepted ACPs have augmented the Tilbury peaking supply with temporary and

15 suboptimal measures. FEI's portfolio optimization suggests that FEI requires 200 MMcf/d x 5 days

16 (1.0 Bcf) of peaking supply.

17 3.3.4.2.1 PEAK LOADS HAVE INCREASED SINCE 1971, AND MATERIALLY IN RECENT YEARS

- The original design capacity of the Base Plant when it was constructed in 1971 was 150 MMcf/d
 of regasification and 0.6 Bcf, which means the Base Plant was designed to provide 150 MMcf/d
- 20 (i.e., 0.15 Bcf/d) of daily deliverability for 4 days (150 MMcf/d x 4d = 0.6 Bcf). Over the past five
- 21 decades, FEI's customer demand has increased significantly. The number of gas customers in
- the Lower Mainland has increased from approximately 200,000 in 1971 to 630,000 in 2023. Since
- 23 2016/2017 alone, FEI's peak day demand has increased by 125 MMcf/d,¹³² which is attributed to:
- 24 (1) customer growth; and (2) Transportation Service customers (i.e., RS 23 and 25) returning to

¹³¹ Appendix F, Raymond Mason Report, p. 20.

¹³² RS 1 to 7 customers.



- bundled service (i.e., RS 3 and 5). The demand growth has increased the need for gas supply
 resources within the portfolio. FEI's peaking capacity requirements now exceed 150 MMcf/d
- 3 regasification capacity of the Base Plant, while the energy requirements now exceed 0.6 Bcf.

4 3.3.4.2.2 PEAK LOADS NOW EXCEED EVEN THE TEMPORARY AND SUBOPTIMAL MEASURES FEI HAS BEEN USING TO AUGMENT THE UNDERSIZED BASE PLANT

In recent years, the Annual Contracting Plans have augmented the 150 MMcf/d and 0.6 Bcf of
Tilbury peaking supply with additional pipeline capacity on T-South. This was the only resource
FEI was able to contract in a constrained market to meet the growing demand. This has been
effective, though sub-optimal from a gas supply planning perspective; however, load growth is
now exceeding the capacity of FEI's additional T-South holdings as well.

- As discussed in the ACPs, relying on additional capacity on T-South for peaking supply is sub-
- optimal because it results in lower asset utilization (i.e., pipeline capacity is only being used on a
 few days each year). It requires FEI to mitigate unused capacity during most of the year,
- 14 increasing portfolio cost risk for customers.

Operating the Base Plant at the reduced tank capacity of 0.35 Bcf for seismic reasons (discussed in Section 3.3.2 above) has only exacerbated the situation. FEI has had to start relying on 0.25 Bcf of peaking energy from Tilbury 1A just to restore the ACP requirement for LNG and serve customers during peak winter periods. For example, FEI needed to draw on this energy from Tilbury 1A during the 2021/2022 winter, when BC experienced prolonged cold weather. This is a temporary and limited solution.

- 21 Tilbury 1A was built pursuant to Direction No. 5 to the BCUC on the basis that it is to be • 22 used for serving natural gas to the transportation sector to reduce GHG emissions by 23 using cleaner fuels. Using Tilbury 1A tank volumes for peaking supply is only possible at 24 present because LNG sales growth has been slower than anticipated to date; however, 25 the recent provincial and federal approvals of the Tilbury Jetty are a significant 26 development because delays in the jetty approval had represented a significant sales 27 constraint. FEI now expects RS 46 LNG sales to increase significantly and sell out Tilbury 28 1A as early as 2028;
- As noted in Section 3.3.2 above, CB&I has suggested that it expects further analysis could
 result in further reductions in the recommended fill elevation (i.e., the tank capacity could
 be reduced below 0.35 Bcf); and
- As explained in further detail below, there is no further pipeline capacity available to serve
 as a substitute for this peaking energy.

34 *3.3.4.3* FEI Could Not Replace Existing Dependable Tilbury Peaking Supply

As discussed below, FEI's assessment, which is supported by the report of Mr. Mason, is that
 fully-contracted regional infrastructure would preclude replacing Tilbury's existing peaking
 capabilities with high-capacity long-term commercial contracts for peaking supply arrangements.
 In the absence of on-system LNG at Tilbury, FEI's ability to secure replacement peaking supply



- 1 would depend on regional pipeline or storage infrastructure being upgraded beyond any currently
- 2 planned expansions. FEI would not control the timing or size of any upgrades, it would take years
- 3 to construct an expansion, and the costs for customers would be significant. In the absence of 4 adequate regional expansions, FEI's firm customers would likely face curtailments in normal
- 5 operations.

6 3.3.4.3.1 HIGH-CAPACITY LONG-TERM CONTRACTS ON REGIONAL INFRASTRUCTURE ARE NO LONGER AVAILABLE

- 8 As noted above, FEI requires 200 MMcf/d x 5 days (1.0 Bcf) of peaking supply and the majority
- 9 of that is provided by Tilbury LNG. Current gas supply resources in the Pacific Northwest region,
- including pipeline capacity on the major regional pipelines and regional storage facilities, are fully
 contracted.¹³³ As described below, the regional demands on the infrastructure are increasing,
- 12 notably due to natural gas power production in the Pacific Northwest and the pending completion
- 13 of Woodfibre LNG. The fully-contracted state of regional infrastructure also means that some of
- FEI's existing holdings are exposed to non-renewal risk.

15 *Pipeline is 365 Days and Storage Also Requires US Pipeline Capacity*

- 16 At a high level, when FEI refers to acquiring regional pipeline or storage capacity, it means:
- Meeting FEI's peaking supply requirements with pipeline capacity would mean securing daily 200 MMcf/d for 365 days per year, and then looking to mitigate (i.e., resell) unused capacity during the remainder of the year when peak loads do not occur. Pipeline capacity held for peaking supply is not required for most of the year (see illustrative Figure 3-23 below).

¹³³ Appendix F, Raymond Mason Report, pp. 15 and 34.



Figure 3-23: Holding Pipeline Capacity to Meet Peak Demand Results in Under-utilization



2

3

4

5

6

7

1

• Replacing Tilbury peaking supply with off-system storage would require off-system storage itself (either at JPS or Mist), plus additional capacity on US pipeline infrastructure between the facility and FEI's system to facilitate storage injection and withdraw. That pipeline capacity, assuming it could be obtained, would be sold on a 365-day basis and would have to be mitigated in a similar fashion.

8 These approaches have been considered as one of the Supplemental Alternatives evaluated in
9 Section 4 of this Supplemental Evidence (Supplemental Alternative 1 – No Capital Upgrades).

10 FEI determined that option was not viable given the issues outlined below.

11 Increasing Gas-Fired Power Production Competes for Capacity

12 As detailed in the 2024/25 Annual Contracting Plan, increased reliance on natural gas-fired power generation in both the I-5 Corridor and the broader Western energy markets has developed into 13 14 a gas supply portfolio risk in the past few years. Government policies aimed at reducing GHG 15 emissions have incentivized development of renewable resources, including hydroelectric, solar 16 and wind, to offset the loss of supply from the coal-fired generation retirements across western 17 North America. However, because renewable resources generally provide intermittent generation 18 (i.e., not firm, dependable generation), natural gas power plants are increasingly needed to meet 19 the region's electricity demand.

Over the past decade, FEI has monitored the daily demand from the natural gas-fired generators
 that are directly tied to Williams' Northwest Pipeline, which is illustrated in Figure 3-24 below.



1 Historically, these gas-fired generators would typically purchase their gas supply at the 2 Huntingdon/Sumas market and would only run on colder-than-normal days during the winter with 3 the ability to shut off when no longer needed to meet demand on peak (cold) days. This practice 4 has fundamentally changed. Figure 3-24 shows that for the past two winter seasons, demand 5 from these natural gas-fired generators averaged 220 MMcf/d higher than the 5-year average 6 between 2016 and 2020, and that they operated at or near maximum capacity on a daily basis 7 throughout the entire winter period. The incremental 220 MMcf/d coming from the natural gas-8 fired generation is approximately 33 Bcf over a 151-day winter season. This is a larger volume 9 than what Jackson Prairie (25 Bcf) or Mist (19 Bcf) storage facilities could offer on their own. 10 These developments have not only put a strain on peaking resources, but also on baseload and 11 storage resources during the winter period. The issue has been compounded by the lack of 12 development of additional infrastructure to meet this incremental demand.

13Figure 3-24: Natural Gas for Power Generation on Northwest Pipeline (Winter Averages in14Bcf/Day)



15

FEI does not expect the reliance on natural gas power generation to change any time soon. A recent North American Electric Reliability Corporation (NERC) report concluded that Western North America is currently at an elevated risk of having insufficient capacity available and energy from resources during extreme and prolonged weather events.¹³⁴ Around the same time that this report was released, an assessment from the Western Electricity Coordinating Council (WECC) concluded that Western North America was not prepared to meet the rapidly increasing demand

¹³⁴ NERC (December 2023). "2023 Long-Term Reliability Assessment."



in the region over the next 10 years.¹³⁵ Currently, the only additions of new capacity are from
 renewable resources, which, as noted-above, are intermittent sources of energy.

3 Woodfibre LNG Completion Will Remove 15 Percent of T-South Capacity

In addition to increased daily demand from the natural gas-fired generators since FEI filed the
 Application, development of the Woodfibre LNG facility has progressed and will significantly
 impact the regional market when it comes online.

- In November 2021, Woodfibre LNG Ltd. (Woodfibre) signalled its intent¹³⁶ to construct an LNG
 export terminal, with a capacity of 2.1 MTPA (approximately 275 MMcf/d) located near Squamish.
 Woodfibre has since commenced construction on its facilities and FEI is constructing the
 infrastructure to bring gas to Woodfibre. Woodfibre has already contracted approximately 15
 percent of the existing T-South capacity to supply its facilities, and for the moment is reselling it.
- 12 However, when the Woodfibre LNG facility commences operations, the capacity will no longer be
- 13 available to the market. This will represent a very significant tightening of the market.

14 The expected loss of Woodfibre capacity has led to proposals for pipeline expansions to help solve existing market dynamics. In particular, Enbridge Inc. (Enbridge) conducted an "open 15 16 season" in Q2 2022 to develop an expansion of T-South, specifically to replace the capacity that 17 will no longer be available to the market when Woodfibre enters service. In November 2022, after 18 this open season was over-subscribed, Enbridge announced that it is moving ahead with plans to 19 expand T-South with up to 300 MMcf/d of additional capacity (i.e., T-South Sunrise Expansion).¹³⁷ 20 Since the open season, Enbridge has provided two updates regarding Sunrise's capital cost. The 21 first update occurred during the open season process, with the capital cost estimated at \$3.6 22 billion. Subsequently, the capital cost estimate was revised to \$4 billion. These cost increases 23 reflect a refinement of the scope of the project and a more detailed build-up of project 24 requirements.

The T-South Sunrise Expansion does not add capacity beyond what will be lost to Woodfibre LNG, and it could thus not serve as a replacement for peaking supply from Tilbury. Moreover, the regional gas supply markets (i.e., Huntingdon/Sumas) are expected to remain high and volatile during winter seasons. As noted by Mr. Mason:¹³⁸

29 ...even assuming the proposed WEI Sunrise Expansion Program is successfully
 30 implemented and constructed, this project would only replace Woodfibre LNG
 31 supplies, leaving the conditions at the Huntingdon/Sumas market in a similar

 ¹³⁵ Western Energy Electricity Coordinating Council (November 2023). "Western Assessment of Resource Adequacy."
 ¹³⁶ Woodfibre announced the award of an Engineering, Procurement, Fabrication, and Construction contract for the construction of the facility to McDermott International. Woodfibre LNG had announced that pre-installation work beginning in 2022 and provided a Notice to Proceed in April 2022 to McDermott International, which has led to major construction starting in September 2023. This Notice to Proceed is effectively an FID in signaling that Woodfibre will move forward with the project.

¹³⁷ https://www.enbridge.com/projects-and-infrastructure/projects/sunrise-expansion-program.

¹³⁸ Appendix F, Raymond Mason Report, p. 49.



- supply/demand and pricing position as today (i.e., 300-400 MMcf/d of market
 absorption potential).
- 3 FEI remains concerned that, in the absence of a further regional infrastructure expansion beyond
- the T-South Sunrise Expansion, infrastructure constraints will drive high Sumas prices with
 significant price volatility.
- 6 Over the last two years, FEI conducted a comprehensive evaluation of potential alternative 7 pipeline routes. The findings suggest that collaborating with other regional market participants on 8 an integrated solution could be beneficial for the region and FEI's customers. However, no further 9 upgrades are imminent, and any upgrade would come with increased annual costs for FEI 10 customers. The current tolls FEI must pay on regional infrastructure for existing and incremental 11 capacity will increase regardless of who triggers an upgrade, due to the rolled-in tolling of federally 12 regulated transmission pipelines in Canada. The tolls are cost of service based, such that once 13 the upgrades are completed, the associated cost will be recovered from all shippers. Since FEI 14 holds the most firm-service capacity on T-South, the cost increases to FEI's supply portfolio will 15 be significant when the system undergoes major expansions.

16 FEI's Existing Regional Supply Assets Are Exposed to Non-Renewal Risk

17 In addition to the above considerations, FEI's Annual Contracting Plan includes resources that do

18 not have renewal rights within the existing contracts with counterparties and are at risk of not

19 being renewed because of the market conditions throughout the region. The loss of access to

20 Tilbury would exacerbate an already significant gas supply issue for FEI.

21 For instance, FEI currently holds approximately 3 Bcf of storage and approximately 110 MMcf/d 22 of deliverability from the Mist facility. However, Northwest Natural, the owner of Mist and a large 23 utility in the Pacific Northwest, has already indicated in public documents that it is intending to 24 place greater reliance on Mist capacity for its own purposes as part of its own response to regional 25 market forces described above (the Mist facility is on-system storage for Northwest Natural). Once 26 the Woodfibre facility is in-service, the regional gas flow and prices for all customers that rely on 27 the Huntingdon/Sumas market hub may be impacted, as the hub is expected to become more 28 volatile in the absence of new infrastructure. Northwest Natural has indicated that their strategy 29 to reduce this Huntingdon/Sumas supply exposure for its utility customers is to recall Mist storage 30 capacity for its own use. Northwest Natural's 2022 Integrated Resource Plan (IRP) discussed 31 customer exposure to the Huntingdon/Sumas market, and how this exposure will be "further 32 exacerbated in 2027 when the Woodfibre LNG facility is expected to come online."139

FEI has had ongoing discussions with Northwest Natural to understand the materiality of the expected capacity being recalled in the near future. The exact amount to be recalled is subject to change based on updates to NW Natural's IRP and/or changes to the expected in-service date of the Woodfibre facility. However, based on the most recent discussions, FEI expects that as much as 50 percent of FEI's currently contracted capacity could be recalled by winter 2027/2028. The

¹³⁹ 2022 NW Natural Integrated Resource Plan (September 2022), "<u>Exposure to Sumas</u>", p. 209 online at: <u>https://edocs.puc.state.or.us/efdocs/HAA/Ic79haa174551.pdf</u>.



- 1 results of this for FEI's holdings are shown in Figure 3-25 below. FEI filed an application with the
- 2 BCUC in respect of underwriting an expansion of the Mist facility to restore those amounts, which
- 3 was recently approved. The effect of that expansion is shown in the yellow bars in Figure 3-25).



Figure 3-25: FEI's Mist Recall (Storage Deliverability)¹⁴⁰

5

4

6 FEI continues to be exposed to non-renewal rights on some of its other existing regional supply 7 resources. The recall of FEI's capacity in the Mist facility due to the requirements of its owner is 8 a demonstration of how difficult it would be to secure alternative storage in the event that on-9 system LNG at Tilbury were to cease to be available.

10 3.3.4.3.2 FEI Could Not Replace Tilbury Peaking Supply with Short-Term Ad Hoc Commercial Agreements

As discussed below, adopting a strategy of replacing this dependable peaking supply on an *ad hoc* basis without long-term contracts would be very detrimental to customers. First, under current market conditions, it would not be realistic to expect that FEI could replace that much peaking supply at the Sumas market; significant curtailments of firm load in normal operations are a likely outcome. To the extent that some *ad hoc* contracts were available, FEI would be exposed to nondelivery risk and very significant prices.

¹⁴⁰ 1 Dth = 1 Mcf = 0.001 MMcf.



Mr. Mason, an expert in the regional gas market, offered the following assessment which aligns
 with FEI's experience discussed in this section:¹⁴¹

3 In my opinion, FEI would not be able to contract for peaking capacity resources 4 that I would consider dependable, beyond those that are currently committed, as 5 an alternative to on-system LNG storage. The deployment duration and dispatch 6 capability of a resource remain the main criteria for security of supply from a 7 peaking capacity resource. Due to the nature of the regional demands of FEI's 8 customers, third-party off-system storage and companies offering peaking gas supply arrangements, are constrained by the ability to transport the underlying 9 10 supply of natural gas, as well as the potential for unplanned outages. Furthermore, based on the results of my market research evaluating third-party arrangements, 11 12 the costs for peaking resources are extremely expensive, do not support a 13 consistent long term supply resource, and would require a portfolio of participants 14 to be able to meet FEI winter demand. To put this in perspective, FEI would be competing/accessing the Huntingdon/Sumas gas supply market, on the coldest 15 16 days of the winter, for significant volumes historically destined to the PNW (rapidly 17 escalating pricing throughout daily trading hours). These factors, when taken together, are not conducive to contract for dependable peaking supply resources. 18

19 In the absence of readily available dependable peaking resources (including 20 sufficient existing on-system storage), FEI would need to make significant 21 investments (including capital investments and/or contractual capacity 22 commitments) to meet its peaking capacity needs. The types of investments could 23 include expanding mainline transportation, and the associated interconnected off-24 system storage facilities, and/or developing on-system peak shaving LNG.

25 <u>The Amount of Capacity that Tilbury Provides is Too Great to Replace Through Ad Hoc</u> 26 <u>Commercial Arrangements</u>

It would not be realistic to assume FEI could replace the existing Tilbury peaking supply withcommercial deals on a short-term, *ad hoc* basis to meet peak day demand.

The existing 150 MMcf/d capacity at Tilbury represents approximately 8 percent¹⁴² of the current firm service WEI T-South capacity, which is a large amount from a market perspective.

The winter flow on T-South is already at its maximum capacity, and replacing the peaking supply provided by the Base Plant means parties who hold existing T-South capacity will have to sell their capacity to FEI or sell the delivered Station 2 supply at Sumas in the form of commercial arrangements such as call options. Brokering out physical assets to the secondary market would only be transacted if it involves limited risk. FEI holds more than one third of the T-South capacity, and other utilities in the region are facing the same challenges during cold winter periods.

¹⁴¹ Appendix F, Raymond Mason Report, p. 4.

¹⁴² MMcf/d / 1,800 MMcf/d (i.e., 1,800 MMcf/d is Westcoast Energy firm service to Huntington or Sumas).



- 1 On a short-term basis, it may be possible to transact a peaking call option to receive some volume
- 2 of gas delivered at Huntingdon; however, this is not dependable over the long term and would still
- 3 leave FEI well-short of being able to meet its full peak day requirements. Moreover, peaking call
- options of this nature would require FEI to pay a call option premium upfront (i.e., demand charge)
 to have the rights to purchase the gas at prevailing Sumas prices. FEI would have to pay Sumas
- 6 daily prices which have been volatile and expensive during peak periods. Further, when significant
- 7 physical disruptions occur on regional infrastructure, commercial counterparties may not be able
- 8 to deliver on commercial arrangements.

9 <u>Counterparties May Not Provide Gas When Called Upon</u>

- 10 FEI has relatively few commercial contracts included in the current portfolio. FEI's gas supply
- 11 strategy has long been to rely on long-term firm services through contracting pipeline and storage
- 12 capacities to meet customer demand. In addition to exposing FEI to non-renewal risk and potential
- 13 cost escalation, commercial contracts hold the potential for non-performance.
- Commercial peaking deals are associated with volatile commodity prices and physical delivery on peak day. Unlike long-term contracts with storage and pipeline owners directly, the counterparties on short-commercial transactions may or may not hold physical capacity on regional infrastructure. If the counterparty failed to provide the gas when called, it would be challenging for
- 18 FEI to replace the required volume on a cold/peak day when other market competitors could be
- 19 seeking gas at the same time.
- 20 To mitigate this non-performance risk, FEI contracts for the underlying firm service with the
- pipeline and/or storage resource. Given the current resource constraints in the region, transacting
 commercial deals without holding the underlying firm resources to provide large amount of
- 23 peaking supply is not prudent or practical.
- Further, FEI notes that over time there is considerable turnover in the counterparties that participate in the Sumas marketplace, which highlight some of the challenges associated with the Sumas market.
- 27 Any Gas Successfully Procured on the Spot Market Will Be Very Costly
- To the extent that FEI was successful in partially replacing the lost on-system peaking gas supply from Tilbury with *ad hoc* contractual arrangements on the day, FEI customers would be very exposed to price risk.
- The lack of existing regional infrastructure already results in extreme price volatility and price spikes during peak demand, including increased power demand. These periods of pricing volatility usually occur when increased demand in the Pacific Northwest region and the western US exceeds the delivery capacity of pipelines into the region, which causes prices to increase significantly above other market prices. In recent years this has become more pronounced. It was evident during the winter months of 2022/23, for instance, as is illustrated in the following figure.





Figure 3-26: Regional Settled Daily Prices for Winter 2022/23

2

1

3 As Figure 3-26 above shows, during winter 2022/23 the prices at Sumas, Malin and Rockies, 4 which are market hubs for the Pacific Northwest region and the western US, became 5 disconnected from AECO/NIT, the main market hub for western Canada. The price spikes at the 6 market hubs were due to cold weather in the Pacific Northwest region at the same time as frigid 7 temperatures for much of the US midcontinent, which increased natural gas demand, including 8 increased natural gas demand for power generation. These periods of pricing volatility usually 9 occur when increased demand in the region exceeds the delivery capacity of pipelines into the 10 region.

11 Table 3-4 below illustrates the magnitude of potential costs customers could face by relying on 12 the Sumas market on a spot (non-firm) basis. The table provides the total cost that FEI would 13 have incurred assuming that it had been necessary for FEI to procure its peaking supply from the 14 Sumas market during the 2022/23 winter. This illustrative calculation assumes that the supply 15 would have actually been available in the market on the cold days, which would be in considerable 16 doubt given the large volumes required. It is also understated because it does not account for the 17 upward price impacts that could be expected from FEI trying to buy such large volumes in an 18 already tight market, as discussed below.


Table 3-4: Illustrative Commodity Costs to Replace 150 MMcf/d and 0.6 Bcf Peaking Supply at Winter 2022/23 Sumas Prices¹⁴³

		Sumas Peaking Supply Commodity Costs (150 MMcf/d x 4 days = 0.6 Bcf)			Annual Costs	Annual Costs	
		Day 1 Day 2 Day 3 Day 4		4 Days	4 Days		
		150000 Mcf	150000 Mcf	150000 Mcf	150000 Mcf	(US\$/year)	(CAD\$/year)
	\$10	\$1,500,000	\$1,500,000	\$1,500,000	\$1,500,000	\$6,000,000	\$8,108,108
Sumas Daily	\$20	\$3,000,000	\$3,000,000	\$3,000,000	\$3,000,000	\$12,000,000	\$16,216,216
Price*	\$30	\$4,500,000	\$4,500,000	\$4,500,000	\$4,500,000	\$18,000,000	\$24,324,324
(US\$/Mcf)	\$40	\$6,000,000	\$6,000,000	\$6,000,000	\$6,000,000	\$24,000,000	\$32,432,432
	\$50	\$7,500,000	\$7,500,000	\$7,500,000	\$7,500,000	\$30,000,000	\$40,540,541

3

* 1 Mcf = 1 MMBtu

4 These historical market prices occurred in an environment when FEI has had limited need to

5 participate in buying Sumas gas during winter periods due to the buying strategies within the ACP.

6 That would no longer be possible if FEI were to cease to have access to on-system LNG at Tilbury.

7 FEI is one of the major market participants in the local gas markets, such that any significant

8 changes FEI makes in its supply portfolio can be expected to have an impact on the local price

9 setting during high demand periods. Telegraphing to the market that FEI was facing having to

10 replace the existing 150 MMcf/d of LNG supply from Tilbury with commercial arrangements - an

11 amount representing approximately 8 percent of the daily capacity of T-South – could be expected

12 to put upward pressure on the monthly and daily prices that FEI's customers would have to pay.

13 That is, the daily Sumas price shown above would have been higher if FEI was participating in

14 the Sumas trading, as the market would be further constrained.

3.3.4.4 FEI Customers Would Face Curtailments Pending a Third Party Upgrading *Regional Infrastructure Sufficiently*

A strategy of relying on uncertain regional infrastructure expansions to replace FEI's existing
critical on-system peaking supply would contradict the strategy that has been inherent in FEI's
Annual Contracting Plans for many years. It would be very detrimental to customers, as it would
involve the likelihood that FEI would have to curtail firm load in peak winter periods.

FEI would not control the timing or size of any upgrades. FEI does not own T-South or the regional storage facilities and is no longer pursuing the RGSD Project as an FEI project.

23 Even once triggered, it would take years to develop a pipeline or regional storage expansion. In

the meantime (i.e., between the time when FEI loses access to Tilbury and completion of the

25 regional infrastructure upgrades), FEI would be reliant on short-term contractual arrangements

26 with the attendant availability, pricing and counterparty risks outlined above.¹⁴⁴

¹⁴³ The table illustrates potential commodity costs of Sumas peaking supply, which does not include the call option premium if FEI were to transact for firm supply to be delivered at the Sumas/Huntingdon area.

¹⁴⁴ To be conservative, FEI did not include the costs for these short-term contractual gas supply contracts in its financial analyses for any of the alternatives analyzed.



1 Mr. Mason highlighted the timing risk in his Report:¹⁴⁵

2 In the absence of readily available dependable peaking resources (including 3 sufficient existing on-system storage), FEI would need to make significant 4 (including capital investments and/or contractual capacity investments 5 commitments) to meet its peaking capacity needs. The types of investments could 6 include expanding mainline transportation, and the associated interconnected off-7 system storage facilities, and/or developing on-system peak shaving LNG. 8 Building these types of resources takes time, especially capital projects that 9 require consultation with multiple parties throughout the planning and construction 10 phases. For example, increasing WEI mainline capacity and associated interconnected off-system storage facility (i.e., Aitken Creek) would likely require 11 12 longer development periods than increasing off-system storage displacement in 13 the Pacific Northwest and/or increasing on-system peak shaving LNG.

14 In my opinion, the development of a proprietary on-system asset, such as the 15 proposed TLSE Project, has advantages over the alternative infrastructure 16 investments when it comes to designing a gas supply portfolio. Constructing a 17 mainline transportation resource, for the purposes of meeting winter demand, 18 would also result in inefficient utilization due to underutilized capacity in non-winter 19 months. Any investments considered by FEI should consider the assets' long-term 20 utilization parameters and the timing in which the resource could be deployed.

As discussed in Section 4 of this Supplemental Evidence, FEI has screened out Supplemental Alternatives that do not maintain FEI's existing peaking gas capabilities in light of the risks associated with procuring replacement peaking supply in the spot market or as part of undefined regional infrastructure upgrades. The options ruled out on this basis include those Supplemental Alternatives that continue to rely on the aging Base Plant (Supplemental Alternatives 1-3), as well as those that replace the Base Plant but allocate the entirety of the tank to a resiliency reserve (Supplemental Alternatives 5-7).¹⁴⁶

3.3.4.5 Relying on Upgraded Regional Infrastructure for Peaking Supply Would Involve Significant Gas Supply Costs for FEI Customers

If and when those regional infrastructure upgrades were to occur and FEI was able to secure
 replacement capacity on that infrastructure, customers would face a significant annual cost
 recovered through cost of gas.

Any regional pipeline or storage expansions capable of replacing or supplementing FEI's existing on-system peaking supply would involve the owners of that infrastructure incurring significant capital costs. These capital costs get reflected in higher tolls or demand charges, which are

¹⁴⁵ Appendix F, Raymond Mason Report, pp. 4-5.

¹⁴⁶ Although FEI screened out these alternatives, FEI nonetheless completed a financial evaluation that accounts for the expected gas supply costs associated with holding capacity on expanded regional infrastructure at some point in the future.



- 1 passed along to FEI customers through cost of gas. The cost of JPS or Mist expansions are borne
- 2 by the party that triggers them, and the higher tolls are passed on to FEI customers through cost
- of gas. In the case of pipeline expansions, due to rolled-in tolling design FEI would pay higher
 tolls than today for <u>all</u> of its capacity (existing holdings plus new incremental holdings). That
- 5 means FEI's existing capacity holdings on the regional infrastructure would also become more
- 6 expensive for customers.
- 7 As part of the alternatives analysis in Section 4, FEI has estimated the annual gas supply costs
- 8 that FEI customers would incur for peaking supply in a scenario where regional infrastructure was
- 9 upgraded at a future date. The feasible Supplemental Alternatives avoid these costs to varying
- 10 degrees, representing a direct supply benefit for customers. Any such supply benefit is accounted
- 11 for as a reduction in the levelized customer bill impacts in the financial analysis for the feasible
- 12 Supplemental Alternatives.



1 **3.4 CONCLUSION**

2 The need to mitigate the significant resiliency risk facing FEI's customers and ensure that FEI

3 continues to have access to sufficient dependable peaking supply is clear. FEI must now invest

4 in solutions to address these needs as continuing to rely on the *status quo* will only further expose

5 FEI's customers to these risks and is therefore unacceptable.

6 The customer outage risk assessment undertaken by Exponent as part of the 2024 Resiliency 7 Plan has re-confirmed that a winter T-South no-flow event presents a very significant risk to FEI 8 customers and BC generally. There is a high probability that hundreds of thousands of customers 9 will, at some point, experience a lengthy loss of gas supply during winter that will expose 10 vulnerable populations to serious health and mortality risks and have widespread social and 11 economic consequences for British Columbians. This risk is too significant to leave unmitigated.

12 Further, based on the additional time that has elapsed since the Application proceeding and the additional work performed by FEI, it is clear that the 53-year-old Base Plant needs to be replaced 13 14 irrespective of resiliency. The Tilbury Base Plant has reached end-of-life and is unable to reliably 15 perform its critical gas supply function. A strategy of prolonged reliance on the the Base Plant for 16 critical peaking supply, given its age-related issues and the absence of any feasible contingency 17 plan for peaking supply, leaves firm customers at serious risk of losing service in normal 18 operations. Allowing this to occur would contradict the gas supply planning approach that has 19 been in place for decades.

- 20 Section 4 explains why this need is best addressed with a larger facility that will both provide a
- 21 designated resiliency reserve and meet FEI's full peaking supply requirements.



1 4. EXPANDED ALTERNATIVES ANALYSIS

2 **4.1** INTRODUCTION

This section addresses FEI's expanded alternatives analysis. FEI describes the 13 options considered as part of the expanded analysis (Supplemental Alternatives), the process used to screen and assess them, and the overall evaluation results. Appendix C to this Supplemental Evidence provides the detailed supporting information and analysis on each of the options.

In the Application, FEI considered a variety of ways to improve its ability to withstand a winter T-South no-flow event. The first stage of that original alternatives analysis considered load management approaches, on- and off-system storage, and four different regional pipeline solutions. In the second stage of that original alternatives analysis, FEI assessed different tank sizing and regasification capacities that can support Lower Mainland load for at least three days during winter.¹⁴⁷

- 13 In the Adjournment Decision, the BCUC identified a need for FEI to further consider:
- "...alternatives that offer different resiliency benefits from those that the TLSE Project
 purports to provide whether such alternatives offer greater, lesser, or qualitatively
 different levels of resiliency benefits relative to TLSE";¹⁴⁸
- the possibility of using a portion of the Tilbury 1A tank for resiliency;¹⁴⁹
- whether FEI could extend the life of the Base Plant;¹⁵⁰
- three other non-Tilbury alternatives, and revisit the resiliency provided by a Southern
 Crossing Pipeline expansion;¹⁵¹ and
- the potential for the transition towards a lower carbon future to affect the appropriate sizing
 of the TLSE Project.¹⁵²
- The expanded alternatives analysis includes all of the alternatives and considerations identified
 by the BCUC in the Adjournment Decision.¹⁵³
- FEI's expanded alternatives analysis confirms that, among the viable options, Supplemental Alternative 9 (replacing the Base Plant with 800 MMcf/d new regasification and 3 Bcf tank with 2

¹⁴⁷ Exhibit B-1-4, Application, Section 4.

¹⁴⁸ Adjournment Decision, p. 25.

¹⁴⁹ Adjournment Decision, p. 29.

¹⁵⁰ Adjournment Decision, p. 14.

¹⁵¹ Adjournment Decision, pp. 25-31.

¹⁵² Adjournment Decision, p. 52. "Further, if the throughput of natural gas is reduced due to a decrease in demand, the size of a tank and the amount of regasification required would likely be reduced." And p. 22: "The larger tank provides flexibility to accommodate future load growth that may occur. However, given the current emphasis on electrification and decarbonization in BC, it is unclear whether FEI will experience significant, or even any, future natural gas load growth. The larger tank means greater risk of a stranded, or partially stranded, asset in the event that FEI's increased load does not emerge or decreases beyond the current load."

¹⁵³ Adjournment Decision, Section 4.



Bcf set aside as a "resiliency reserve") is the Preferred Alternative. Supplemental Alternative 9 delivers on the Project objectives in a way that provides superior customer value. In particular, Supplemental Alternative 9: (1) significantly mitigates FEI's largest customer outage risk; (2) meets FEI's full peaking supply requirements in an optimal manner from a gas supply portfolio design perspective; (3) fully addresses age-related Base Plant challenges; (4) compares favourably to smaller replacement facilities in terms of levelized total rate impact; and (5) is expected to be fully utilized throughout its expected service life.

8 The section is organized as follows:

- Section 4.2 Summary of Supplemental Alternatives and Methodology to Select the
 Preferred Alternative: FEI describes the 13 Supplemental Alternatives, the structured
 three-step process, and the scoring approach that FEI used to evaluate the Supplemental
 Alternatives and select the Preferred Alternative.
- Section 4.3 Results of Step 1: FEI Eliminated Technically and Commercially Non-Viable Alternatives: FEI screened out four Supplemental Alternatives as non-viable, including continuing to rely on the end-of-life Tilbury Base Plant with no capital upgrades, as well as non-Tilbury options.
- 17 Section 4.4 – Results of Step 2: FEI Eliminated Alternatives that Do Not Retain FEI's 18 Existing On-System Firm Peaking Gas Supply: Five options that would fail to retain 19 FEI's existing peaking supply are unacceptable because they would result in firm 20 customers losing service in peak winter periods under normal operating conditions. The 21 four Supplemental Alternatives that passed this secondary screen - Supplemental 22 Alternatives 4, 4A, 8, and 9 – would all involve a full replacement of the existing Base Plant with a new facility that will, at a minimum, restore the Base Plant's original design 23 24 capabilities for gas supply.
- Section 4.5 Step 3: FEI describes the scoring criteria applied to the Supplemental
 Alternatives
- Section 4.5.1 Step 3 "Resiliency Benefit" Scoring Criterion: Only two
 Supplemental Alternatives those with 800 MMcf/d of regasification and a
 resiliency reserve of 1 Bcf or more (Supplemental Alternatives 8 and 9) would
 materially improve FEI's ability to withstand a winter T-South no-flow event at
 average winter temperatures. Supplemental Alternative 9 will provide considerably
 more outage risk mitigation than any other option. Smaller replacement facilities,
 for all practical purposes, would only serve a gas supply function.
- Section 4.5.2 Step 3 "Availability of Dependable Gas Supply During Peak
 Demand" Scoring Criterion: Only Supplemental Alternatives 4A and 9 would
 provide the optimum amount of peaking gas supply (i.e., sufficient peaking supply
 to meet FEI's entire requirements, such that it both replaces the supply provided
 by Tilbury and avoids having to augment it with year-round pipeline capacity). The



- former is sized solely for optimum gas supply, while Supplemental Alternative 9 is a larger facility that combines that optimum gas supply with resiliency.
- Section 4.5.3 Step 3 "Resolves Age-Related Base Plant Challenges"
 Scoring Criterion: Supplemental Alternatives 4, 4A, 8, and 9 would all involve the
 installation of a new tank and new regasification equipment built to current
 standards, such that FEI assigned a "High Positive Impact" score to each of these
 options.
- 8 Section 4.5.4 – Step 3 "Levelized Total Rate Impact" Scoring Criterion: FEI 0 9 considered the combined impact of capital and operating costs and the cost of 10 procuring the necessary peaking supply (200 MMcf/d and 1.0 Bcf), since there is 11 no zero-cost option available for customers when it comes to peaking gas supply. 12 The relative levelized total rate impact for the Supplemental Alternatives is 13 significantly influenced by: (1) economies of scale in the construction of the facility: 14 and (2) the extent to which an alternative allocates a portion of the tank to peaking 15 gas supply, thereby avoiding the cost of using longer duration off-system storage 16 or pipeline capacity for peaking supply if / when regional infrastructure is upgraded 17 sufficiently. Supplemental Alternative 4A, which is sized based on the optimal 18 amount of peaking gas supply only¹⁵⁴, would have the lowest levelized rate impact 19 among Supplemental Alternatives 4, 4A, 8, and 9. The Preferred Alternative's 20 levelized rate impact will fall between Supplemental Alternatives 4, 4A and 8, which 21 translates to much more risk reduction per dollar of rate impact than any of those 22 other options.
- Section 4.5.5 Step 3 "Future Use" Scoring Criterion: In response to BCUC commentary in the Adjournment Decision, FEI assessed the risk that the Supplemental Alternatives would be underutilized in the future under hypothetical load loss sensitivities. The hypothetical sensitivities illustrate how all of the viable options (Supplemental Alternatives 4, 4A, 8, and 9) would be fully utilized for resiliency and/or gas supply in 2050. On-system LNG is a unique asset when it comes to the flexibility afforded in response to changing load.
- **Section 4.6 Summary of Alternatives Analysis Results**
- Section 4.7 Additional Information on the Preferred Alternative's Resiliency
 Benefits: Supplemental Alternative 9 (the Preferred Alternative) will provide significant
 resiliency benefits over and above what are accounted for in the Step 3 "Resiliency
 Benefit" criterion. These benefits include: (1) providing the potential option of a staged
 shutdown that maintains service to some portion of FEI's customers; (2) avoiding
 uncontrolled depressurization and the attendant safety risks; (3) providing valuable time
 for customers, governments and social / health services to prepare for an outage; (4)

¹⁵⁴ It provides sufficient peaking supply to meet FEI's entire requirements, such that it both replaces the supply provided by Tilbury and avoids having to augment it with year-round pipeline capacity.



mitigating the calculated customer outage risk for 13 other customer outage vulnerabilities
 across FEI's system; and (5) backstopping off-system storage supply in the event of a
 disruption at those facilities.



14.2SUMMARY OF SUPPLEMENTAL ALTERNATIVES AND METHODOLOGY TO2SELECT THE PREFERRED ALTERNATIVE

3 This section summarizes the Supplemental Alternatives and FEI's three-step assessment 4 process. FEI selected the Preferred Alternative based on considerations of technical and 5 commercial viability, the ability to retain existing firm peaking supply, the ability to mitigate 6 customer outage risk, age related Base Plant challenges, levelized total rate impact, and future 7 use.

4.2.1 FEI Considered 13 Supplemental Alternatives, Contingent Scenarios, Plus a Southern Crossing Extension

Table 4-1 below provides summary descriptions of each of the Supplemental Alternatives, which include the options identified in the Adjournment Decision. More detailed descriptions of each Supplemental Alternative are included in Table C.4 in Appendix C. Section 2.4.

12 Supplemental Alternative are included in Table C-1 in Appendix C, Section 2.1.

13 All of the Supplemental Alternatives listed in the table below reflect a **planning view**, which means

14 that they treat stored LNG as being available on a dependable basis for a single planned purpose,

15 since it will not be dependable for any purpose if it is allocated to (or planned for) multiple

16 purposes. This is the typical basis of utility planning, and FEI considers it should be used for the

17 reasons described in Section 4 of Appendix C to this Supplemental Evidence.

18 However, in response to the BCUC's commentary in the Adjournment Decision, FEI also 19 investigated "contingent" scenarios for viable Supplemental Alternatives that would involve 20 either only replacing the regasification equipment or a new facility with a tank less than 2 Bcf (i.e., Supplemental Alternatives 1, 2, 3, 4, 4A, and 5). In general, the contingent scenarios for a given 21 22 alternative consider the case where some volume of LNG that is not set aside as a "resiliency 23 reserve" is available on the day of a no-flow event. In other words, some volume of LNG from 24 Tilbury 1A or the Base Plant (as applicable) that was intended to be used for RS 46 LNG sales or 25 gas supply remains in the tank on the day of the no-flow event and is then used for resiliency 26 instead of its intended purpose. While FEI does not endorse using this approach for planning 27 purposes (because FEI cannot rely on these LNG volumes being present on Day 1 of a no-flow 28 event), the scenarios address comments in the Adjournment Decision about the potential 29 resiliency value of such volumes. Each contingent scenario is explained in the Appendix C 30 sections addressing Supplemental Alternatives 1, 2, 3, 4, 4A, and 5,



Table 4-1: Summary of Supplemental Alternatives

Supp Alt #	Name	Description			
	Alternatives Reliant on Existing Facilities ¹⁵⁵				
Alt 1	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)	Run the Base Plant until it is no longer usable with no resiliency reserve.			
Alt 2	New Regasification Only – 400 MMcf/d (No Resiliency Reserve)	Replace the Base Plant regasification with 400 MMcf/d of new capacity, but continue to rely on a non-refurbished Base Plant tank until it is no longer usable. There is no resiliency reserve.			
Alt 3	New Regasification Only – 600 MMcf/d (No Resiliency Reserve)	Replace Base Plant regasification with 600 MMcf/d of new capacity but continue to rely on a non-refurbished Base Plant tank until it is no longer usable. There is no resiliency reserve.			
	New Fa	cility with Gas Supply But No Resiliency Reserve ¹⁵⁶			
Alt 4	Like-for-Like (No Resiliency Reserve)	Replace the Base Plant like-for-like to restore the 1971 design capacity (150 MMcf/d regasification and 0.6 Bcf tank) and continue using Tilbury as a supply peaking resource, without a resiliency reserve.			
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification	Replace the Base Plant with the smallest new facility capable of providing FEI's optimum peaking gas supply, including a 1 Bcf tank and 200 MMcf/d regasification (with an additional 200 MMcf/d for redundancy). Continue using it as a supply peaking resource, without a resiliency reserve.			
	New Fa	cility with Resiliency Reserve But No Gas Supply ¹⁵⁷			
Alt 5	Like-for-Like (Full Resiliency Reserve)	Replace the Base Plant like-for-like to restore the 1971 design capacity (150 MMcf/d regasification and 0.6 Bcf tank) and allocate the entire tank as a resiliency reserve.			
Alt 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	Replace the Base Plant with a 1.0 Bcf tank and 800 MMcf/d regasification. Allocate the entire tank as a resiliency reserve.			
Alt 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	Replace the Base Plant with a 2.0 Bcf tank and 800 MMcf/d regasification. Allocate the entire tank as a resiliency reserve.			
	New Facility wit	h Both Resiliency Reserve and Replacement of Gas Supply			
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct the smallest facility that allows FEI to both avoid curtailments of firm peak load in normal operations and provide a certain level of resiliency reserve. This includes replacing the Base Plant with a 2 Bcf tank and 800 MMcf/d regasification, of which 1.4 Bcf is allocated as a resiliency reserve, and 0.6 Bcf is allocated to replace the existing gas supply functions at Tilbury.			

¹⁵⁵ These alternatives include prolonged reliance on the Base Plant tank with no dependable resiliency reserve, declining reliability, and a high likelihood of relying on the market for some replacement gas supply.

¹⁵⁶ These alternatives do not include a dependable resiliency reserve but provide different amounts of peaking gas supply and improved reliability.

¹⁵⁷ These alternatives include a full resiliency reserve but still rely on the market for replacement of the gas supply functions.



Supp Alt #	Name	Description
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct a facility that both significantly mitigates FEI's largest customer outage risks and meets FEI's required peaking gas supply in an optimal manner (since it no longer requires relying on year-round pipeline capacity to provide short-term peaking supply). Replace the Base Plant with a 3 Bcf tank and 800 MMcf/d regasification and allocate 2 Bcf as a resiliency reserve, and 1 Bcf to gas supply.
		Non-Tilbury Alternatives ¹⁵⁸
Alt 10	Alt 1 plus VITS Reverse Flow	FEI would retain the existing Tilbury facilities with no capital upgrades (i.e., Supplemental Alternative 1). FEI would also construct the necessary facilities to allow significant reverse flows on the Vancouver Island Transmission System (VITS) at all times during the year, such that the combined daily delivery is at least 550 MMcf/d.
Alt 11	LNG from Woodfibre LNG	Use the existing Tilbury facilities with no capital upgrades (i.e., Supplemental Alternative 1). FEI would also contract with Woodfibre LNG for a long-term firm supply of LNG.
Alt 12	Floating LNG	Purchase a vessel to provide floating LNG storage. Acquire a water lot that would allow for permanent mooring. Add more regasification capacity, either as an integrated component of the LNG storage vessel or on the adjacent shoreline. Construct onshore facilities, including a jetty and interconnecting pipe.

2 In response to the BCUC's commentary in the Adjournment Decision, Appendix C to this 3 Supplemental Evidence also addresses a Southern Crossing Pipeline extension (e.g., the RGSD 4 Project). The RGSD Project was one of the alternatives that FEI originally evaluated in the TLSE 5 Application. FEI has recently determined not to pursue RGSD on its own, although FEI has not 6 foreclosed participating with others in a similar pipeline project. Regardless, FEI explained in the 7 Application why it regarded a Southern Crossing Pipeline extension as complementary to, rather 8 than a substitute for, on-system LNG when it comes to mitigating the risk associated with a winter 9 T-South no-flow event. That conclusion is unchanged following FEI's additional analysis. FEI's 10 transient modelling shows that in the winter months the Lower Mainland system will depressurize 11 before any supply could be obtained from a Southern Crossing Pipeline extension, regardless of 12 size or end-point.

4.2.2 FEI Employed a Structured Process to Identify the Preferred Alternative

FEI employed a structured process to evaluate the 13 Supplemental Alternatives and identify a
 preferred alternative. It is depicted in Figure 4-1 and further described below.

16 To inform the steps in the process, FEI conducted extensive analysis and evaluation of the 17 Supplemental Alternatives with respect to technical and commercial viability, resiliency, gas 18 supply, reliability, levelized total rate impact, and future use. This involved conducting both internal 19 and external studies and analyses to understand the performance of the Supplemental 20 Alternatives (e.g., risk analysis, gas supply market studies, capital cost estimates, and levelized

¹⁵⁸ FEI considers these alternatives to be non-viable approaches to providing winter resiliency and peaking gas supply.



- 1 total rate impact analysis). Appendix C to this Supplemental Evidence provides the full results of
- 2 the scoring analysis for all viable Supplemental Alternatives that passed Step 1.
- 3

Figure 4-1: Three-Step Process to Identify the Preferred Alternative



- 5 **<u>Step 1</u>**: Screen out alternatives that are not technically or commercially viable.
- 6 <u>Step 2</u>: Screen out alternatives that would not retain FEI's existing on-system firm peaking gas
 7 supply capabilities.
- 8 Step 3: Score the remaining¹⁵⁹ Supplemental Alternatives to select a preferred alternative,
 9 having regard to resiliency, gas supply, age-related Base Plant challenges, levelized
 10 total rate impact, and future use.

4.2.2.1 Step 1 Methodology: Screening of Technically and Commercially Non Viable Alternatives

13 Step 1 in the structured process was to screen out Supplemental Alternatives that, after 14 evaluation, are determined to be technically or commercially non-viable. FEI considered the 15 balance of technical and commercial challenges associated with an alternative and then 16 considered viability holistically.

4.2.2.2 Step 2 Methodology: Screening of Alternatives That Would Not Retain FEI's Existing On-System Firm Peaking Gas Supply Capabilities

In Step 2, FEI screened out Supplemental Alternatives that would not retain FEI's existing on system firm peaking gas supply capabilities of 0.6 Bcf¹⁶⁰ of LNG and 150 MMcf/d of regasification
 capacity.

- 22 FEI's gas supply and system planning has always been based on being able to meet firm load at
- all times, and this remains the appropriate basis for planning. As discussed in Section 3.3.4 of
- 24 this Supplemental Evidence, since 1971 FEI has been relying on on-system LNG in its BCUC-

¹⁵⁹ For completeness, FEI has still scored any alternatives screened out at Step 2. See Appendix C for this analysis.

¹⁶⁰ Consisting of 0.35 Bcf from the Tilbury Base Plant plus 0.25 Bcf from Tilbury 1A.



- 1 approved Annual Contracting Plans (ACP) to serve customers during periods of peak demand.
- 2 This approach to supply portfolio design has recognized the important attributes of on-system
- 3 LNG, including: providing supplemental capacity (i.e., MMcf/d versus only energy measured in
- 4 Bcf); providing deliverability on short notice; being within FEI's full control; and the ability to
- 5 backstop other resources.
- As discussed in Section 3.3.4.2, FEI's current requirements for peaking supply are well in excess
 of what the Base Plant can provide, and there are no available substitutes in the market to replace

8 the capacity and energy provided by it. In those circumstances, losing access to the 0.6 Bcf / 150

9 MMcf/d peaking supply in the ACP would mean likely curtailments for firm customers in normal

- 10 operations during the coldest times of the year.
- In practice, there are two potential ways that a Supplemental Alternative might fail to retain FEI's
 existing on-system firm peaking gas supply capabilities:
- the Supplemental Alternative removes FEI's existing on-system peaking resource without replacement; or
- the Supplemental Alternative relies on a resource that is expected to be unavailable in the
 future due to market conditions (e.g., no capacity available on pipelines or regional
 storage) or aging FEI infrastructure.
- FEI considers that any alternative that fails to retain FEI's existing on-system peaking resources is not viable, because of the high risk of FEI not being able to meet firm load. However, FEI has analyzed and scored these options in Appendix C to this Supplemental Evidence so as to provide
- 21 the BCUC with complete information.

22 *4.2.2.3* Step 3 Methodology: Scoring of Viable Alternatives

Step 3 involves scoring the viable Supplemental Alternatives (i.e., those that are technically and commercially viable and that, at minimum, retain FEI's existing firm peaking supply capabilities) against a set of weighted criteria to determine the preferred alternative. Step 3 consists of the following three elements, which are explained in the following sections:

- 27 1. Scoring System and Approach;
- 28 2. Evaluation Criteria; and
- 29 3. Evaluation Criteria Weighting.

30 4.2.2.3.1 SCORING SYSTEM AND APPROACH IN STEP 3

- 31 FEI's approach to scoring the Supplemental Alternatives in Step 3 was to assess the impact the
- 32 alternative would have on the given evaluation criterion relative to retaining the end-of-life Base
- 33 Plant in its current state with no capital upgrades (Supplemental Alternative 1).



- 1 As an alternative may have a positive, negative, or no impact on a given criterion, FEI used the
- 2 scoring system shown in the table below.
- 3

Table 4-2:	Supplemental	Alternative	Scoring	System
------------	--------------	-------------	---------	--------

Score	Corresponding Numerical Score (un-weighted)
High Negative Impact	-5
Medium Negative Impact	-3
Low Negative Impact	-1
No Impact	0
Low Positive Impact	1
Medium Positive Impact	3
High Positive Impact	5

5 After FEI assigned a score to a Supplemental Alternative for a given criterion, the corresponding

6 un-weighted numerical score was multiplied by the criterion's weighting (discussed below) to

7 determine the weighted numerical score. The weighted numerical scores from each criterion were

8 then summed to determine the total weighted numerical score for each alternative. The

9 Supplemental Alternative with the highest weighted numerical score is the preferred alternative.

10 4.2.2.3.2 EVALUATION CRITERIA, METRICS AND SCORING GUIDELINES IN STEP 3

11 The Step 3 evaluation criteria used for scoring is presented below.

12

Table 4-3: Supplemental Alternative Evaluation Criteria

Category	Evaluation Criterion	Evaluation Criterion Description
Resiliency	Resiliency Benefit	The ability to mitigate the risk associated with a winter T-South no-flow event.
Gas Supply	Availability of Dependable Gas Supply During Peak Demand	The impact on the availability of dependable gas supply during peak demand.
Base Plant Challenges	Resolves Age Related Base Plant Challenges	The impact on the age-related Base Plant challenges (send-out reliability, seismic design, flooding, and tank venting).
Rate Impact	Levelized Total Rate Impact	The levelized total rate impact due to both the capital costs on delivery rates and gas supply impacts/benefits on commodity rates.
Future Use	Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP 2% and 5%) Between the In-Service Date and 2050	Considers if the alternative is useful or underutilized under two adverse future load sensitivities, in which FEI's Diversified Energy (Planning) Scenario is modified to assume higher rates of customer and load loss (2% and 5% annually) between the in-service date and 2050.



1 Due to the broad similarities in the viable alternatives (i.e., all involve on-system LNG at Tilbury), some criteria that FEI typically considers, such as constructability,¹⁶¹ have been excluded from 2 3 this evaluation. Environmental considerations are reflected in "Base Plant Challenges", since the 4 primary environmental considerations that assist in distinguishing amongst the viable options are 5 related to the increased environmental standards required for newer construction including (but 6 not limited to) fully enclosed secondary containment and improved tank venting configurations.¹⁶² 7 For each criterion, FEI identified a metric for evaluating the alternatives, as well as scoring 8 guidelines to ensure consistent scoring across the alternatives: 9 1. <u>Resiliency Benefit</u> 10 **Evaluation Metric:** The level of risk mitigation provided by the alternative against • 11 a winter T-South no-flow event as determined via a quantitative risk assessment 12 performed by Exponent (i.e., risk without the alternative – risk with the alternative). 13 Scoring Guidelines: Relative comparison of the level of risk mitigation each 14 alternative provides. 15 2. Availability of Dependable Gas Supply During Peak Demand 16 Evaluation Metric: Qualitative evaluation of the alternative's impact on the • availability of dependable gas supply during peak demand conditions. 17 18 Scoring Guidelines: Relative comparison of the impact the alternative has to the • 19 availability of dependable gas supply. 3. Resolves Age Related Base Plant Challenges 20 21 Evaluation Metric: Qualitative evaluation of the alternative's impact on the age-• related Base Plant challenges discussed in Section 3 of this Supplemental 22 23 Evidence (i.e., send-out reliability, seismic design, flooding, and tank venting). 24 • Scoring Guidelines: Relative comparison of the impact the viable alternative has on the age-related Base Plant challenges. 25 26 4. Levelized Total Rate Impact 27 Evaluation Metric: The levelized total rate impact of the viable alternative • (delivery rate impacts associated with the capital costs of the Project net of gas 28 supply benefits or costs). 29

¹⁶¹ In the Application, FEI assessed that all of the options involving a new facility at Tilbury up to 3 Bcf were constructable.

¹⁶² See Section 3.3 of this Supplemental Evidence for a discussion of environmental considerations inherent in the Base Plant design.



• Scoring Guidelines: Comparison of levelized total rate impact between each supplemental alternative relative to Supplemental Alternative 1 - No Capital Upgrades.

4 5. <u>Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP 2% and</u> 5 <u>5%) Between the In-Service Date and 2050</u>

- 6 Evaluation Metric: Qualitative evaluation to assess if the viable alternative will be 7 useful for resiliency and/or FEI's gas supply portfolio (to serve load or generate 8 mitigation revenue), and its potential to be underutilized between the in-service 9 date and 2050 based on two adverse future load sensitivities referred to as "mDEP 2% and 5%". These sensitivities start with FEI's 2022 LTGRP Diversified Energy 10 11 (Planning) Scenario (DEP Scenario), with modifications to assume higher rates of 12 customer and load loss (2% and 5% annually) between the in-service date and 13 2050.
- Scoring Guidelines: Relative comparison of each alternative's usefulness/underutilization between the in-service date and 2050 based on FEI's modified DEP Scenario adverse future load sensitivities (mDEP 2% and 5%).
- To inform the alternative evaluation and to establish usefulness versus underutilization, FEI applied the following guidelines:
- 191. An alternative remains useful and not underutilized for resiliency if, over20the course of the evaluation period (i.e., from the in-service date to 2050),21the load support duration provided by the resiliency reserve does not22exceed what is reasonably required to avoid an outage.
 - An alternative remains useful and not underutilized for gas supply if, over the course of the evaluation period (i.e., from the in-service date to 2050), the gas supply allocation is less than the volume that can be utilized for FEI gas supply and/or be monetized via FEI's ordinary course gas supply mitigation activities.
- 28 4.2.2.3.3 EVALUATION CRITERIA WEIGHTING IN STEP 3
- As shown in Table 4-4 below, FEI assigned a weighting to each evaluation criterion.
- 30

23

24

25

26

27

1

2

3

Table 4-4: Step 3 Evaluation Criteria Weighting

Category	Evaluation Criterion	Weighting
Resiliency	Resiliency Benefit	30%
Gas Supply	Availability of Dependable Gas Supply During Peak Demand	20%



Category	Evaluation Criterion	Weighting
Base Plant Challenges	Resolves Age Related Base Plant Challenges	20%
Rate Impact	Levelized Total Rate Impact	20%
Future Use	Used and Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP 2% and 5%) Between the In-Service Date and 2050	10%

To determine the weighting of each criterion, FEI Subject Matter Leads (SMLs) conducted a qualitative review considering the relative importance of each criterion. All of the criteria included in the evaluation process are important to FEI, such that the assigned weightings only indicate relative importance. FEI provides the following rationale for the assigned weightings:

- 6 **Resiliency** is assigned the most weight as FEI is currently exposed to a significant 7 customer outage risk that FEI believes should be mitigated. As discussed in Section 3 of 8 this Supplemental Evidence, the consequences are catastrophic and the cumulative 9 probability calculated by Exponent is high. The known consequences of a winter T-South 10 no-flow event are so severe that, according to Exponent (and consistent with prior expert 11 evidence in this proceeding), established risk management principles would support a 12 material investment to reduce the harm even at much lower probabilities than those 13 calculated by Exponent.
- Gas Supply, Base Plant Challenges, and Levelized Total Rate Impact are weighted equally in the Step 3 scoring. Consideration of gas supply at Step 2 means that all of the alternatives being scored in Step 3 provide, at minimum, sufficient dependable peaking supply to allow FEI to continue meeting its peak firm load. Once that minimum standard is met, FEI considered it appropriate to give equal weight to differences in gas supply, levelized total rate impact and the extent to which an alternative addresses the Base Plant's inherent challenges (reliability, seismic, flooding and venting).
- **Future Use** is given relatively less weight due to the known current needs and the uncertainty surrounding the long-term implications of the energy transition.

FEI describes below the results of applying the three-step process. The structured process confirmed Supplemental Alternative 9, which is the originally proposed 3 Bcf tank (2 Bcf resiliency reserve, 1 Bcf gas supply) with 800 MMcf/d regasification capacity, as the preferred alternative.



1 4.3 RESULTS OF STEP 1: FEI ELIMINATED TECHNICALLY AND COMMERCIALLY 2 NON-VIABLE ALTERNATIVES

3 FEI investigated all 13 Supplemental Alternatives to assess technical and commercial viability, as

- 4 defined in Section 4.2.1 above. Through this step, FEI determined that Supplemental Alternatives
- 5 1, 10, 11, and 12 are not viable.
- 6 The following table provides a high-level summary of why each Supplemental Alternative was
- 7 found to be non-viable. A more detailed discussion of these Supplemental Alternatives and why
- 8 they were found to be non-viable is provided in Appendix C to this Supplemental Evidence.
- 9

Table 4-5:	Summary	of Non-Viable	Alternatives
------------	---------	---------------	--------------

No.	Description	Summary of Why Non-Viable	Reference
Alt 1	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)	Supplemental Alternative 1 is technically non-viable due to the age-related challenges associated with the Tilbury Base Plant. As discussed in Section 3, the Tilbury Base Plant has reached its end of life and can no longer reliably perform its intended function. As a result, continuing to rely on the Base Plant without capital upgrades is not technically viable.	Section 3.3 and Appendix C, Section 5.1.1
Alt 10	Alt 1 plus VITS Reverse Flow	Even with significant upgrades, a hydraulic constraint exists which limits the amount of reverse flow that is possible through the VITS. Also significantly more costly compared to Tilbury-based alternatives, given this alternative would involve looping significant portions of FEI's VITS and completing multiple compressor station upgrades.	Appendix C, Section 5.5.1
Alt 11	LNG from Woodfibre	Inconsistent with Woodfibre's (WFLNG) business model and LNG markets. WFLNG's business model requires that any LNG storage WFLNG has on the site be inventoried to ensure the next customer vessel can be filled on schedule. Further, even assuming FEI were able to contract for LNG supply from WFLNG, FEI would then need significant new infrontructure to make use of the LNC. Noither of the two	Appendix C, Section 5.5.2
		infrastructure to make use of the LNG. Neither of the two infrastructure options are viable.	
Alt 12	Floating LNG	There are no appropriate sites. All options also have issues with technical feasibility either due to the tie-in point pressure rating and pipeline capacity, or execution difficulties caused by the location of the tie-in point in the Fraser River, difficulty and uncertainty of additional regulatory approvals associated with a water-based option, and high probability of an Environmental Assessment given the size of the storage capacity or based on a discretionary mandate by the Province.	Appendix C, Section 5.5.3

10



14.4Results of Step 2: FEI Eliminated Alternatives that Do Not2Retain FEI's Existing On-System Firm Peaking Gas Supply

In Step 2, FEI evaluated the technically and commercially viable Supplemental Alternatives and
eliminated those that would not retain FEI's existing on-system firm gas supply capabilities of 0.6
Bcf of on-system storage and 150 MMcf/d of regasification capacity, as loss of the existing firm
gas supply capabilities creates a high risk of material firm load curtailments in normal operations
during peak winter periods. Table 4-6 below summarizes the results of the Step 2 screening.
Further detail is provided in the sections below.

9

Table 4-6: Summary of Step 2 Screen

Supp Alt	Name	On-System Modelling P	Gas Supply arameters ¹⁶³	Retains FEI's Existing On-System Firm
#		То 2030	After 2030	Capabilities?
Alt 2	New Regasification Only - 400 (No Resiliency Reserve)	0.6 Bcf 400 MMcf/d	< 0.6 Bcf 400 MMcf/d	×
Alt 3	New Regasification Only - 600 (No Resiliency Reserve)	0.6 Bcf 600 MMcf/d	< 0.6 Bcf 600 MMcf/d	×
Alt 4	Like-for-Like (No Resiliency Reserve)	0.6 Bcf 150 MMcf/d	0.6 Bcf 150 MMcf/d	~
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 Regasification	1 Bcf 400 MMcf/d	1 Bcf 400 MMcf/d	~
Alt 5	Like-for-Like (Full Resiliency Reserve)	0 Bcf 150 MMcf/d	0 Bcf 150 MMcf/d	×
Alt 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 Regasification	0 Bcf 800 MMcf/d	0 Bcf 800 MMcf/d	×
Alt 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 Regasification	0 Bcf 800 MMcf/d	0 Bcf 800 MMcf/d	×
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 Regasification	0.6 Bcf 800 MMcf/d	0.6 Bcf 800 MMcf/d	~
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 Regasification	1.0 Bcf 800 MMcf/d	1.0 Bcf 800 MMcf/d	~

10

4.4.1 Aging Base Plant Tank and Expected Increase in RS 46 Sales Make Supplemental Alternatives 2 & 3 Non-Viable

13 Supplemental Alternatives 2 and 3 would replace the regasification equipment at Tilbury that has

14 reached end-of-life, and would thus solve one potential cause of losing the Tilbury Base Plant as

15 a peaking resource. However, Supplemental Alternatives 2 and 3 would still leave FEI without full

¹⁶³ The modelling parameters for each supplemental alternative are discussed in greater detail in Appendix C to this Supplemental Evidence.



- access to the dependable peaking supply from the storage that is currently in FEI's ACP. FEI
 would be at risk of no longer being able to meet firm peak load in normal operations.
- 3 There are three interrelated reasons why this is the case:
- As discussed in Section 3.3.2.1 of this Supplemental Evidence, seismic challenges are
 inherent in the original Base Plant design. FEI now operates the tank well-below its design
 capabilities for seismic reasons (at a reduced volume of 0.35 Bcf), and experts have
 advised against tank retrofits to restore it to 0.6 Bcf. If the operating level of tank were
 further reduced due to the seismic design challenges, then other things being equal FEI
 would be left with less than 0.6 Bcf of on-system firm peaking supply.
- Changing market conditions, along with the approval of the Tilbury Jetty and the anticipated delivery of an LNG marine bunker vessel to service the Port of Vancouver, have resulted in an expected increase in RS 46 sales. Therefore, the 0.25 Bcf of LNG from Tilbury 1A that FEI currently relies on for peaking supply will no longer be available.
- 14 3. As discussed in Section 3.3.4.3 of this Supplemental Evidence, there is no back-up option for dependable peaking supply if the 150 MMcf/d and 0.6 Bcf provided by Tilbury is no 15 longer available. Relying on the market for all of FEI's peaking supply would be a 16 17 significant deviation from FEI's longstanding practice of relying on on-system resources in the BCUC-approved ACP, and would not be viable in the current market context. Regional 18 19 infrastructure is fully-contracted, such that any access to long-term replacement peaking 20 supply would depend on regional infrastructure upgrades. FEI would not control the timing 21 or size of any upgrades, it would take years to construct an expansion, and the impact on 22 gas supply (storage and transportation) costs for customers would be significant. 23 Upstream resources are not as dependable as on-system LNG.
- 24 In the meantime, FEI would be faced with the highly undesirable position of relying on the 25 Sumas market for short-term, non-dependable supply contracts with significant price 26 exposure. There is no certainty that FEI would be able to buy at Sumas. This is supported 27 by Mr. Mason, an expert in the regional gas market, who concluded: "In my opinion, FEI would not be able to contract for peaking capacity resources that I would consider 28 29 dependable, beyond those that are currently committed, as an alternative to on-system LNG storage."¹⁶⁴ Considering that T-South capacity is fully contracted during winter, FEI 30 31 would be attempting to buy the required supply, most likely at a premium price, from 32 parties who hold T-South capacity to deliver gas at Sumas/Huntingdon. FEI customers 33 would bear additional supply costs assuming FEI was able to transact the deal on the day. 34 If unable to transact the deal, then FEI would be unable to serve all customers and would 35 have to start curtailing.

¹⁶⁴ Appendix F, Raymond Mason Report, pp. 4 and 22.



4.4.2 Absence of a Gas Supply Reserve Make Supplemental Alternatives 5, 6, and 7 Non-Viable

Supplemental Alternatives 5, 6, and 7 would replace the existing Tilbury Base Plant with a new
storage tank and regasification. However, these supplemental alternatives would not allocate any
of the new storage volume to gas supply. In each case, the entirety of the tank would be allocated
to a resiliency reserve, leaving FEI without its existing 0.6 Bcf and 150 MMcf/d of on-system
peaking resources.

As a result, these options would potentially reduce the risk exposure to a T-South winter no-flow
event and would address reliability considerations and age-related issues with the Base Plant.
However, these alternatives create an unacceptable risk to FEI's ability to continue serving firm
load during peak winter periods in normal operations. As discussed in the previous section above
and Section 3.3.4 of this Supplemental Evidence, making up the lost supply on the spot market

13 or through regional resources would not be viable.

14 **4.4.3** Four Supplemental Alternatives Remain for Step 3 Evaluation

15 Supplemental Alternatives 4, 4A, 8, and 9 are the four alternatives that are technically and 16 commercially viable and retain FEI's existing on-system firm peaking supply.

17

unu	rotain	Chisting	011 0	yotom	 pean	'9 '	Sabl	Jiy.

New Facility with Gas Supply But No Resiliency Reserve ¹⁶⁵							
Alt 4	Like-for-Like (No Resiliency Reserve)	Replace the Base Plant like-for-like to restore the 1971 design capacity (150 MMcf/d regasification and 0.6 Bcf tank) and continue using Tilbury as a supply peaking resource, without a resiliency reserve.					
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification	Replace the Base Plant with the smallest new facility capable of providing FEI's optimum peaking gas supply, including a 1 Bcf tank and 200 MMcf/d regasification (with an additional 200 MMcf/d for redundancy). Continue using it as a supply peaking resource, without a resiliency reserve.					
New Facility with Both Resiliency Reserve and Replacement of Gas Supply							
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct the smallest facility that allows FEI to both avoid curtailments of firm peak load in normal operations (albeit in a suboptimal manner, since it still requires relying on pipeline capacity to provide short-term peaking supply) and provide a certain level of resiliency reserve. This includes replacing the Base Plant with a 2 Bcf tank and 800 MMcf/d regasification, of which 1.4 Bcf is allocated as a resiliency reserve, and 0.6 Bcf is allocated to replace the existing gas supply functions at Tilbury.					

 Table 4-7:
 Supplemental Alternatives that Passed Step 2

¹⁶⁵ These alternatives do not include a dependable resiliency reserve but provide different amounts of peaking gas supply and improved reliability.



New Facility with Both Resiliency Reserve and Replacement of Gas Supply						
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct a facility that both significantly mitigates FEI's largest customer outage risks and meets FEI's required peaking gas supply in an optimal manner (since it no longer requires relying on pipeline capacity to provide short-term peaking supply). Replace the Base Plant with a 3 Bcf tank and 800 MMcf/d regasification and allocate 2 Bcf as a resiliency reserve, and 1 Bcf to gas supply.				

2 4.5 RESULTS OF STEP 3 ALTERNATIVES ANALYSIS

This section sets out the results of the Step 3 analysis, focusing on the four options that passed
Step 2 (Supplemental Alternatives 4, 4A, 8 and 9). Please see Appendix C for the Step 3 results
for Supplemental Alternatives that did not pass the Step 2 screen.

6 4.5.1 Resiliency Benefit Scoring Criterion

7 The "Resiliency Benefit" criterion considers an alternative's ability to mitigate the risk associated 8 with a winter T-South no-flow event, which is FEI's largest customer outage risk exposure. As 9 described below, the evaluation is based on multiple quantitative analyses performed by 10 Exponent that consider various winter temperature conditions and the results of FEI's system modelling. On this criterion, Supplemental Alternatives 4 and 4A scored the lowest ("No Impact"), 11 12 while Supplemental Alternatives 8 and 9 scored as "Medium Positive Impact" and "High Positive 13 Impact", respectively. Supplemental Alternative 9 is therefore superior when it comes to mitigating 14 the customer outage risk posed by a T-South winter no-flow event. In particular:

- Supplemental Alternatives 4 and 4A would provide no risk mitigation on a planning basis
 as they do not include a resiliency reserve.
- While Supplemental Alternatives 8 and 9 would provide similar and significant risk mitigation under average winter temperatures (refer to Figure 4-2 below), the larger resiliency reserve provided by Supplemental Alternative 9 (2.0 Bcf vs. 1.4 Bcf) will make it far superior at below average temperatures prevalent for significant portions of a typical winter.
- 22 The longer load support duration provided by Supplemental Alternative 9 relative to 23 Supplemental Alternative 8 means that there would be a greater likelihood that any Lower 24 Mainland shutdown would occur in a controlled, versus uncontrolled, manner. This is 25 particularly true when the temperature is between -6.8°C to +1.7°C, as in that temperature 26 range Supplemental Alternative 9 will be able to provide at least a 3-day¹⁶⁶ support 27 duration but Supplemental Alternative 8 would not. Refer to Section 3.2.2 of this 28 Supplemental Evidence for details on why a controlled shutdown is preferred over an 29 uncontrolled shutdown.

¹⁶⁶ Per FEI's 2024 Resiliency Plan, three days is enough time to implement a controlled shutdown.



- The resiliency benefits provided by Supplemental Alternative 9 will extend beyond the risk mitigation discussed in this section. The risk mitigation calculations presented in this section relate specifically to mitigation of FEI's largest risk (i.e., T-South winter outage risk) which is how the "Resiliency Benefit" criterion in Step 3 is defined. Supplemental Alternative 9 will provide additional quantifiable risk mitigation for other Assessed Vulnerabilities and off-system storage disruptions. The cumulative risk reduction benefits of Supplemental Alternative 9, accounting for
- 7 all Assessed Vulnerabilities, are discussed in greater detail in Section 4.7 below.

4.5.1.1 Exponent Accounted for Load Support Duration and Uncertainty in Duration of No-Flow Period

10 The assessment under the Resiliency Benefit criterion is based on risk calculations performed by 11 Exponent in respect of a winter T-South no-flow event.¹⁶⁷ Exponent performed three analyses to 12 determine the risk mitigation provided by the Supplemental Alternatives at different temperature 13 conditions that can occur in a typical Lower Mainland winter:

- 13 conditions that can occur in a typical Lower Mainland winter:
- Risk reduction provided by Supplemental Alternatives 4, 4A, 8 and 9 at average winter
 temperatures in the Lower Mainland (+4°C) against a T-South winter no-flow event.
- Expected annual customer-outage-day risk under Supplemental Alternatives 8 and 9 at a
 fixed temperature of -1.4°C.
- 18 3. Load support duration provided by Supplemental Alternatives 8 and 9 at variable colder 19 temperatures (i.e., support duration as a function of temperature), including identifying the 20 temperature range in which the alternatives provide different levels of risk mitigation in the 21 context of a 3-day regulatory shutdown. The analysis also considers the load support 22 duration as a function of temperature with respect to historical temperature data and in the 23 context of a 3-day regulatory shutdown (i.e., based on historical temperature data, it 24 identifies the portion of winter days in which Supplemental Alternative 9 can bridge a 3-25 day regulatory shutdown but Supplemental Alternative 8 cannot).

26 Exponent's calculated risk reduction reflects the results of FEI's transient modelling, which 27 determines the length of time the Supplemental Alternatives could support the Lower Mainland in 28 the event of a loss of supply from T-South. Exponent's Monte Carlo-based risk analysis accounts 29 for multiple modes of failure and the potential for no-flow events of varying durations, (i.e., the risk 30 mitigation shown in all of Exponent's risk mitigation figures accounts for the possibility that the noflow duration may exceed the duration of support provided by Supplemental Alternative 9). 31 32 Exponent provides a detailed description of its Monte Carlo-based risk analysis in Appendix RP-2 to the 2024 Resiliency Plan, Exponent Report, Appendix U. 33

¹⁶⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report.



1 4.5.1.2 Exponent's Risk Mitigation Results – Average Winter Temperature (+4°C)

2 This section addresses the results of Exponent's analysis at average winter temperature in the

3 Lower Mainland (+4°C).

Figure 4-2 below summarizes the annual expected GDP loss reduction associated with each of Supplemental Alternatives 4, 4A, 8 and 9, relative to Supplemental Alternative 1 (Planning) scenario that involves no capital upgrades.¹⁶⁸ The pattern is similar regardless of the time horizon or whether the consequence metric used is cumulative GDP loss reduction or customer-outagedays.



Figure 4-2: T-South at Average Winter – Expected Annual Loss Reduction



¹⁶⁸ Refer to Section 2.5 of Appendix C to this Supplemental Evidence for a detailed discussion on the resiliency performance of all Supplemental Alternatives.



Supplemental Alternatives 4 and 4A received a score of "No Impact" for the Resiliency Benefit
 criterion because they would not improve the current poor ability to mitigate FEI's largest
 resiliency risk even at average winter temperature in the Lower Mainland. As shown in the figure:

- Supplemental Alternative 4 (Planning) and Supplemental Alternative 4A (Planning) would provide no additional risk mitigation against a winter T-South no-flow event when compared to FEI's current capabilities (represented by Supplemental Alternative 1). This is because, like Supplemental Alternative 1, Supplemental Alternatives 4 and 4A would not have a dedicated resiliency reserve, and therefore could not support the Lower Mainland for any duration during a winter T-South no-flow event.
- Supplemental Alternative 4's performance would not improve under a favourable "contingent" scenario. In particular, Supplemental Alternative 4 (Contingent), which assumes that 0.6 Bcf of LNG is available on the day of a no-flow event, would not mitigate additional risk (relative to Supplemental Alternative 1 (Contingent)) because the like-forlike replacement of the existing Base Plant does not address the governing constraint – limited regasification capacity of only 150 MMcf/d.
- 16 Supplemental Alternative 4A (Contingent), which assumes that 1.0 Bcf of LNG is available 17 despite the tank's use for peaking supply, would provide improved risk mitigation; 18 however, the improvement would not be material. As shown in Figure 4-2 above, the loss 19 reduction provided by Supplemental Alternative 4A (Contingent) would surpass that 20 provided by Supplemental Alternative 4 (Contingent), but is significantly less than what 21 would be provided by the larger supplemental alternatives (i.e., Supplemental Alternatives 22 8 and 9). As the basis for scoring this criterion is a relative comparison of the level of risk 23 mitigation each supplemental alternative provides (refer to Section 4.2.2.3.2), the 24 improvement in risk mitigation that would be provided by Supplemental Alternative 4A 25 (Contingent) is too small to warrant a positive score. As such, even when considering the 26 favourable contingent scenario for Supplemental Alternative 4A, the score for the 27 Resiliency Benefit criterion remains "No Impact".
- At average Lower Mainland winter conditions (+4°C), Supplemental Alternatives 8 and 9 would both provide significant risk mitigation. The figure above shows the risk mitigation as a reduction in annual expected GDP losses, totalling approximately \$160 million per year. Over a long period of time, those avoided expected GDP losses are very significant. The same is true when considering customer-outage-days and customer outages.

Section 4.7 below provides additional Exponent risk calculation results for Supplemental
 Alternative 9, including for various consequence measures and cumulative risk calculations over
 67-year and 23-year horizons. Please refer to Appendix C for a detailed review and discussion of
 the risk mitigation benefit, if any, provided by each supplemental alternative at average winter
 conditions.



1 4.5.1.3 Exponent's Cold Weather Analysis Results – Fixed Temperature (-1.4°C)

FEI requested that Exponent conduct additional analysis for the two options – Supplemental
 Alternatives 8 and 9 – that had demonstrated material risk reductions at average winter
 temperatures. The purpose of the analysis was to understand their relative effectiveness in

5 mitigating risk in colder conditions that are present for periods of the winter.¹⁶⁹ Under this analysis,

- 6 Supplemental Alternative 9 will provide superior risk mitigation under colder conditions when
- 7 compared to Supplemental Alternative 8.

8 As the intent was to understand the difference between the two Supplemental Alternatives under 9 cold weather, as opposed to the total expected risk under colder conditions, this analysis was

10 simplified to only focus on T-South, and only on customers impacted in the Lower Mainland.

11 Excluding customers outside of the Lower Mainland will result in the total expected risk at colder

12 temperatures being under-reported in this secondary analysis; however, the risk to the Lower

13 Mainland is FEI's largest outage risk and the simplification allows for a meaningful comparison of

14 expected Lower Mainland risk between the supplemental alternatives.

15 Figure 4-3 below shows the results from Exponent's expected annual risk calculations at -1.4°C,

16 with direct customer impact expressed in terms of customer-outage-days. The expected risk

17 under Supplemental Alternative 8 is significantly higher than Supplemental Alternative 9. This

18 shows that the latter provides superior risk mitigation at -1.4°C when compared to Supplemental

19 Alternative 8.

¹⁶⁹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Section 9 and Report Appendix U, Section U.6.1.



1Figure 4-3: Expected Annual Winter-only Customer Outage-days Loss for Different Supplemental2Alternatives for T-South (AV-1, -2, -3, and -54) at -1.4°C



T-South (AV-1, -2, -3, and -54)

3

4 The risk results shown above align with the expected load support duration for each of 5 Supplemental Alternatives 8 and 9. Supplemental Alternative 9 provides a 3-day support duration 6 at -1.4°C whereas the support duration provided by Supplemental Alternative 8 would be 7 approximately a full day shorter (see Figure 4-4 below). With a load support duration of less than 8 three days, Supplemental Alternative 8 would be less likely to bridge a no-flow event prompted 9 by, for example, a precautionary regulatory shutdown following an integrity incident on one of the 10 T-South pipelines. The residual risk for Supplemental Alternative 9 is due to possible failure 11 modes that can result in an outage duration exceeding the duration of support provided to FEI's 12 entire Lower Mainland load. In this situation, the Supplemental Alternative 9 tank allows FEI to 13 enact a controlled shutdown, which reduces the number of customer outage days.





Figure 4-4: Lower Mainland Support Duration Provided by Supplemental Alternatives at Various Temperatures



4 4.5.1.4 Exponent's Cold Weather Analysis Results – Variable Temperature

5 Exponent also performed a variable temperature analysis showing the relationship between below 6 average winter temperatures and risk mitigation. The result, described below, is that nearly a 7 quarter of winter days fall in the range in which Supplemental Alternative 9 can support the Lower 8 Mainland load for at least three days but Supplemental Alternative 8 could not. The material 9 difference in load support duration at a given temperature has significant consequences for the 10 GDP losses and customer-outage-days. A likely outcome of a T-South failure is that only one of 11 the two pipelines is physically damaged, and the adjacent undamaged line is shut-in as a 12 precaution by the regulator (i.e., a regulatory shutdown). The regulatory shutdown could 13 reasonably be expected to last three days.

- 14 FEI provided Exponent with the Lower Mainland load support duration for each of Supplemental
- 15 Alternatives 8 and 9 at -10°C, -1.4°C, and +4°C as follows.¹⁷⁰

¹⁷⁰ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Section 9 and Report Appendix U.6.2.



1 2

Table 4-8: Load Support Duration in Days Provided by Supplemental Alternatives 8 and 9 at
Different Temperatures

	Load support duration [days]			
Temperature	Alternative 9			
[degrees Celsius, °C]	Alternative 8	(Preferred)		
4	3.33	4.54		
-1.4	2.54	3.5		
-10	1.92	2.71		

4 Using this data, Exponent determined the load support duration for Supplemental Alternatives 8 5 and 9 as a function of temperature. The key finding from this analysis was that Supplemental 6 Alternative 9 is superior to Supplemental Alternative 8 throughout a temperature range of 7 approximately -6.8°C to +1.7°C. Above +1.7°C, Supplemental Alternatives 8 and 9 both provide 8 at least a 3-day support duration. This translates to similar amounts of risk mitigation because 9 they can both bridge a regulatory shutdown of an undamaged T-South pipe for up to 3 days. 10 However, between -6.8°C and +1.7°C, only Supplemental Alternative 9 will provide 3 days of load 11 support. Thus, in this temperature range Supplemental Alternative 9 will provide a superior 12 resiliency benefit.

13 Exponent then considered the load support duration as a function of temperature with respect to

14 historical daily average temperature data. Figure 74 from the Exponent Report,¹⁷¹ included below

15 as Figure 4-5, presents the results from this analysis in the context of exceeding an assumed 3-

16 day regulatory shutdown. In the figure, the temperature range at which the load support duration

17 exceeds the assumed regulatory shutdown period for Supplemental Alternative 9, but not

18 Supplemental Alternative 8, is highlighted in burgundy in the figure, being -6.8°C to +1.7°C. And

19 the temperature range at which neither Supplemental Alternative 8 nor 9 would meet the 3-day

20 regulatory shutdown period is the blue at the left hand side of the distribution when temperatues 21 are lower than -6.8°C and for which the probability density is very low.

22 Exponent explained the significance of the results as follows:¹⁷²

23 This review of the temperature data and supply duration for each temperature 24 reveals the substantial benefit of using Alternative 7 or 9 (Preferred) over 25 Alternative 8, as nearly a quarter of winter days fall in the range in which 26 Alternatives 7 and 9 (Preferred) can bridge the regulatory shutdown period but 27 Alternative 8 cannot. This has significant consequences for the GDP losses and 28 CODs for failures of a single pipeline on AV-1, -2, -3, and -54 [i.e., the four 29

segments comprising T-South].

¹⁷¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 228.

¹⁷² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 229.



1Figure 4-5: Distribution of Average Daily Winter (December through February) Temperature in the2Lower Mainland (Vancouver Airport) between January 2013 and December 2022



3

4 4.5.1.5 Supplemental Alternative 9 Best Mitigates Risk

5 Supplemental Alternative 9 will provide a significant reduction in customer outage risk that is 6 superior to either Supplemental Alternatives 4, 4A, or 8 because it will increase both the 7 regasification capacity and the storage at Tilbury to the point where it can support FEI's load for 8 much longer than can be supported today.

9 The benefit of more regasification capacity: Supplemental Alternatives 8 and 9 provide 10 800 MMcf/d of regasification capacity, thereby addressing the existing governing 11 regasification constraint at Tilbury. The present regasification constraint guarantees a Day 12 1 widespread customer outage even at average winter temperatures. The proposed 13 regasification capacity of Supplemental Alternative 9 (and Supplemental Alternative 8) is 14 sufficient to support the Lower Mainland daily load in all but the coldest of conditions that 15 occur relatively infrequently for only short durations. Supplemental Alternative 4 would not 16 address the constraint, and Supplemental Alternative 4A, while increasing the 17 regasification capacity and thus providing some improvement, would not fully address the 18 constraint. This is because at average winter conditions (+4°C) the 400 MMcf/d 19 regasification capacity would remain undersized for the Lower Mainland demand.



- The benefit of having a resiliency reserve over and above gas supply: Supplemental 1 • 2 Alternative 9 sets aside 2 Bcf in the 3 Bcf tank as a resiliency reserve, such that this will 3 be the minimum amount available for resiliency when a T-South no-flow event occurs. 4 Once the regasification constraint is addressed, the additional LNG allows FEI to serve 5 the daily load for multiple days, with the specific duration of that support depending on the 6 system load, which is primarily driven by temperature. The longer Tilbury can serve the 7 winter load, the greater the potential to bridge a no-flow event and avoid depressurization 8 and customer outages. Exponent observed that more than 1 Bcf is required to make a 9 material risk reduction:173
- 10 The on-system LNG of a Tilbury Alternative plays an important role in 11 determining the reduction in losses compared to the status quo 12 configuration (Alternative 1 (Planning)). When the Tilbury Facility has 13 limited on-system LNG (1 Bcf or less), relatively few hazards are mitigated 14 by its presence because there is not enough volume to bridge the 15 regulatory shutdown period or shorter repair durations for AVs with higher 16 loads, such as AV-1, AV-2, AV-3, AV-54, and AV-18.
- While Supplemental Alternative 8 would have a 1.4 Bcf resiliency reserve, which would be
 an improvement relative to the existing Base Plant, the additional 0.6 Bcf of LNG provided
 by Supplemental Alternative 9 makes a difference to the level of risk mitigation provided
 (particularly at temperatures below the average Lower Mainland winter temperature).
- 21 A precautionary shut down of the unaffected adjacent T-South pipeline is a likely 22 trigger of a no-flow event: Exponent's risk assessment evaluates a T-South failure 23 assuming a range of no-flow durations and failure modes (i.e., hazards). In Exponent's 24 analysis, in instances where there are parallel pipelines (as is the case for T-South), the 25 analysis considers the probability of each hazard causing simultaneous failure of both 26 pipelines. As many of the hazards considered have a low probability of simultaneous 27 failure, a likely outcome of a T-South failure is that only one line is physically damaged 28 and the adjacent undamaged line is shut-in as a precaution by the regulator (i.e., a 29 regulatory shutdown). As such, FEI can mitigate significant risk by being able to continue 30 supporting load until the federal regulator allows Westcoast to resume flowing gas on the unaffected pipeline. Exponent explained:¹⁷⁴ 31
- Larger on-system LNG volumes (such as 2 Bcf) provide enough backup supply to bridge the three-day regulatory shutdown period on AVs-1, 2, 3, 54, and 18, which are AVs with parallel pipeline segments. Bridging the regulatory shutdown period significantly reduces losses on these AVs, except for cases in which the two parallel pipeline segments fail simultaneously.

¹⁷³ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 192.

¹⁷⁴ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 192.



Supplemental Alternative 9, by virtue of the amount of regasification capacity and LNG
 resiliency reserve, is likely to achieve this outcome. Supplemental Alternative 8 would
 likely achieve this outcome at average winter temperatures, but not if the temperature
 dropped to +1.7°C or below.

5 Lower Mainland outages expected to occur in an uncontrolled manner today will be • 6 controlled: At minimum, even if a no-flow event exceeds the duration of load support and 7 an outage occurs, Supplemental Alternative 9 will provide an adequate response time in 8 most temperature conditions to ensure a controlled shutdown of the Lower Mainland 9 system. An uncontrolled shutdown extends the time to restore service to customers relative to a controlled shutdown. As such, a controlled shutdown reduces the number of 10 customer outage days and GDP losses (please also refer to Section 3.2.2 of this 11 12 Supplemental Evidence). Supplemental Alternative 9 is likely to provide the amount of 13 time necessary for FEI to arrange for a controlled shutdown. Supplemental Alternative 8 14 would not be able to achieve this outcome when the temperature is +1.7°C or below, which 15 represents a significant portion of the winter.

4.5.2 Availability of Dependable Gas Supply During Peak Demand Scoring Criterion

18 With respect to the criterion "Availability of Dependable Gas Supply During Peak Demand", FEI 19 has scored Supplemental Alternatives 4 and 8 as "Medium Positive Impact", and Supplemental 20 Alternatives 4A and 9 as "High Positive Impact". The difference is attributable to the incremental 21 gas supply that Supplemental Alternatives 4A and 9 provide relative to today.

Supplemental Alternative 4 ("Medium Positive"): With a 0.6 Bcf gas supply reserve and 150 MMcf/d of send-out, Supplemental Alternative 4 would retain FEI's existing firm peaking gas supply capabilities that have long been included in the ACP. Further, it would improve FEI's gas supply capabilities relative to Supplemental Alternative 1 by:

- Resolving the age-related challenges associated with the end-of-life Tilbury Base Plant by
 replacing it with a new facility, which increases the availability of dependable gas supply;
 and
- Providing a new tank that effectively provides the amount of LNG that FEI has long included in its ACP, allowing FEI to discontinue the stop-gap measure of relying on 0.25
 Bcf from Tilbury 1A (i.e., LNG that is intended for RS 46 sales) to compensate for the Base
- 32 Plant tank being operated at 0.35 Bcf for seismic reasons.
- Supplemental Alternative 4A ("High Positive"): Supplemental Alternative 4A, with a gas supply
 reserve of 1.0 Bcf paired with 200 MMcf/d¹⁷⁵ of send out, was designed to meet FEI's full peaking

¹⁷⁵ Supplemental Alternative 4A includes 400 MMcf/d of send out capacity, however for gas supply planning purposes only 200 MMcf/d is required.



- supply requirement in an optimal manner without adding resiliency.¹⁷⁶ It would provide the benefits
 outlined above for Supplemental Alternative 4. The additional 0.4 Bcf dedicated to gas supply.
- 3 paired with the additional 50 MMcf/d of send-out, would provide FEI considerable flexibility for gas
- 4 supply planning and winter operation, and potentially displace other less optimal gas portfolio
- 5 assets. Without those additional peaking supply capabilities, FEI would need to continue to incur
- 6 millions of dollars of costs each year to maintain its existing year-round pipeline capacity that FEI
- 7 relies on as a sub-optimal source of peaking supply. Section 4.5.4.1.2 below quantifies the
- 8 avoided gas supply costs associated with Supplemental Alternative 4A.

9 Supplemental Alternative 8 ("Medium Positive"): Supplemental Alternative 8 increases the 10 send-out capacity relative to Supplemental Alternative 4 (800 MMcf/d vs. only 150 MMcf/d). 11 However, this alternative would only allocate 0.6 Bcf of the 2.0 Bcf tank to gas supply, such that 12 it provides the same energy as Supplemental Alternative 4. The additional regasification would 13 provide some additional operational flexibility, but the extent of the benefit would be constrained 14 due to the limited gas supply reserve of only 0.6 Bcf. Like other storage assets, both daily send-15 out capacity and total storage supply are needed to make the asset useful in FEI's gas supply 16 portfolio.

17 LNG peaking is the last resource in the portfolio to provide up to 10 days¹⁷⁷ of peaking supply 18 during winter season. The LNG send-out is only called when additional supply is required on cold 19 winter days. Due to the unpredictability of cold weather events, having additional send-out 20 capacity provides optionality to increase peak day LNG supply in the planning model as well as 21 the operational flexibility to call on LNG send out. Unless there was enough LNG reserve to cover 22 the whole winter period, the increased daily send-out capacity provides limited benefits because 23 one-day send-out could deplete the whole LNG reserve and leave a shortage of peaking supply 24 for the remainder of winter. For instance, sending out at that maximum rate of 800 MMcf/d would 25 consume 0.6 Bcf (i.e., 600 MMcf) in less than a day. As such, FEI has not differentiated between 26 Supplemental Alternatives 4 and 8 for the "Availability of Dependable Gas Supply During Peak 27 Demand" criterion, and both are scored as "Medium Positive".

Supplemental Alternative 9 ("High Positive"): Supplemental Alternative 9 was sized for optimal gas supply and optimal outage risk mitigation. With a gas supply reserve of 1.0 Bcf paired with 800 MMcf/d of send out, Supplemental Alternative 9 will provide the same gas supply benefits as Supplemental Alternative 4A.¹⁷⁸ That is, Supplemental Alternative 9 offers incremental benefits beyond those offered by Supplemental Alternatives 4 and 8. The additional 0.4 Bcf that is dedicated to gas supply, paired with 200 MMcf/d of send-out, will provide FEI with considerable flexibility for gas supply planning and winter operation, and potentially displace or secure other

¹⁷⁶ Section 3.3.4 discusses FEI's peaking supply requirements and how resources are optimized to meet those requirements.

¹⁷⁷ FEI includes the maximum peak day send out and the total LNG reserve in it planning model but not all LNG send out is at the maximum send out capacity.

¹⁷⁸ Supplemental Alternative 9 includes 800 MMcf/d of send out capacity, however for gas supply planning purposes only 200 MMcf/d is required.



higher cost gas portfolio assets. Section 4.5.4.1.2 of this Supplemental Evidence quantifies the
 avoided gas supply costs associated with Supplemental Alternative 9.

3 4.5.3 Resolves Age-Related Base Plant Challenges Scoring Criterion

4 The key consideration for the "Resolves Age-Related Base Plant Challenges" criterion is whether 5 the alternative addresses the issues related to the end-of-life Base Plant (outlined in Section 3.3 6 of this Supplemental Evidence) that are preventing the Base Plant from reliably performing its 7 critical gas supply function. These challenges include unreliable and obsolete regasification 8 equipment that is difficult to repair, the seismic design challenges that have caused FEI to operate 9 the tank at only 59 percent of its design capabilities, the flooding risk, and the environmental 10 challenges associated with how the Tilbury Base Plant tank vents to atmosphere. As 11 Supplemental Alternatives 4, 4A, 8, and 9 all include the installation of a new tank and new 12 regasification equipment built to current standards, each alternative received the same score. A 13 "High Positive Impact" score was selected as the alternatives would be resolving all age-related 14 challenges associated with the Base Plant.

15 4.5.4 Levelized Total Rate Impact Scoring Criterion

The levelized total rate impact criterion compares the incremental levelized total rate impact over a 67-year period between each supplemental alternative. Alternatives with a higher levelized total rate impact over the 67-year analysis period will score lower (i.e., worse) than those alternatives with a lower levelized total rate impact. The levelized total rate impact includes:

- The impact to FEI's delivery rates due to the capital and operating costs of each alternative; and
- The impact to FEI's cost of gas rates (which include both commodity and midstream costs)
 due to the incremental gas supply costs/benefits to FEI's customers resulting from each
 alternative (which are discussed in Section 4.5.4.1.2 below).

The 67-year analysis period is used for the financial analysis to cover the expected useful life of the assets pertaining to all alternatives, which is 60 years for an LNG storage tank, plus seven prior years from 2024 to 2030 (assuming all alternatives are placed in-service by 2030).

Table 4-9 provides the financial results for the four feasible supplemental alternatives (Supplemental Alternatives 4, 4A, 8, and 9) based on the capital and operating cost estimates, as well as the estimate of gas supply costs/savings of each alternative over the 67-year analysis period. Although this section focuses on the results of Supplemental Alternatives 4, 4A, 8 and 9, FEI has also undertaken the financial analysis for the remaining Supplemental Alternatives (i.e., Supplemental Alternative 1, 2, 3, 5, 6, and 7)¹⁷⁹ in Appendix C to this Supplemental Evidence.

¹⁷⁹ No financial analysis was completed for Supplemental Alternatives 10, 11, and 12.



1 2 Table 4-9: Summary of Capital Costs, Cost of Service, Gas Supply Costs/Savings, and Levelized **Total Rate Impacts for Feasible Supplemental Alternatives**

	Alt 4 - 0.6 BCF 150 MMcf/d	Alt 4A - 1 BCF 400 MMcf/d	Alt 8 - 2 BCF 800 MMcf/d	Alt 9 - 3 BCF 800 MMcf/d
	(No resl)	(No resl)	(1.4 BCF resl)	(2 BCF resl)
Total Capital Costs during Construction, As-Spent \$ (\$000s)	826,921	893,199	1,030,287	1,140,962
PV of Cost of Service, excl. Gas Supply Costs/Savings (\$000s) over 67 years	790,047	892,612	1,133,983	1,240,821
PV of Gas Supply Cost/Savings (\$000s) over 67 years	(366,362)	(517,554)	(366,362)	(517,554)
Total PV of Cost of Service over 67 years (\$000s)	423,685	375,059	767,621	723,267
Levelized Total Rate Impact (Incl. Cost of Gas) 67 years (%)	1.44%	1.27%	2.60%	2.45%
Levelized Total Rate Impact (Incl. Cost of Gas) 67 years (\$/GJ)	0.134	0.118	0.242	0.228

3

4 Based on the results of the levelized total rate impact analysis, Supplemental Alternatives 4 and

5 4A received a score of "Low Negative Impact", while Supplemental Alternatives 8 and 9 received

6 a score of "Medium Negative Impact".

7 4.5.4.1 Key Observations from the Financial Analysis Results for the Four Feasible Supplemental Alternatives (4, 4A, 8 and 9) 8

9 Based on the financial results for Supplemental Alternatives 4, 4A, 8, and 9 from Table 4-9 above, the levelized total rate impact to FEI's customers between each alternative over the life of the

10 11 assets (i.e., 67 years) is significantly influenced by two factors:

- 12 The strong economies of scale in the construction capital costs of the facility; and •
- The extent to which an alternative is capable of providing the necessary peaking supply 13 •

14 to avoid curtailments during normal operations, being 200 MMcf/d and 1.0 Bcf (i.e., 5 days

- 15 x 200 MMcf/d) as discussed in Section 3.3.4.2, thereby avoiding the need for FEI to incur
- 16 annual gas supply costs for holding less efficient peaking supply on expanded off-system
- storage or regional pipeline infrastructure (assuming that is even available). 17
- These factors are further discussed below. 18

19 **BENEFITS FROM SIGNIFICANT ECONOMIES OF SCALE** 4.5.4.1.1

As previously discussed in Section 4.4.1.2 of the Application,¹⁸⁰ LNG storage infrastructure is 20 21 characterized by significant economies of scale, where the capital cost per unit of storage 22 decreases as the size of the LNG storage increases. Figure 4-6 below provides a graphical illustration of the strength of the economies of scale for new facilities at Tilbury ranging from a 23 24 like-for-like replacement (0.6 Bcf / 150 MMcf/d) up to a 3 Bcf / 800 MMcf/d replacement. The costs

25 shown are based on updated estimates.

¹⁸⁰ Exhibit B-1-4, Application, p. 107.



\$1,200 +38% \$1,000 -25% +8% Total Capital Costs (\$ million) \$800 \$600 0.6 BCF 3.0 BCF 1.0 BCF 2.0 BCF (\$M/BCF (\$M/BCF (\$M/BCF (\$M/BCF = \$1,378) = \$380) = \$893) = \$515) \$400 +400% \$200 +233% +67% **\$0** 0 0.5 1 1.5 2 2.5 3 LNG Tank Storage Size (BCF)

1 Figure 4-6: Graphical Illustration of Economies of Scale by Tank Capacity (from 0.6 Bcf to 3.0 Bcf)

2

3 The capital cost per Bcf comparison demonstrates that:

When comparing against a 0.6 Bcf and 150 MMcf/d facility that would offer no resiliency benefits (Supplemental Alternative 4), Supplemental Alternative 9 (3.0 Bcf and 800 MMcf/d) provides five times the storage (+400 percent) and over five times the regasification capacity (+433 percent) for approximately \$314 million more (or 38 percent more) in capital cost. The unit cost for Supplemental Alternative 9 with a 3.0 Bcf tank (including ground improvement, auxiliary systems, and regasification) is approximately \$998 million per Bcf lower than the unit cost for Supplemental Alternative 4.

- When comparing against a 1.0 Bcf and 400 MMcf/d facility that would offer no resiliency benefits (Supplemental Alternative 4A), Supplemental Alternative 9 (3.0 Bcf and 800 MMcf/d) provides three times the storage (+200 percent) and two times the regasification capacity (+100 percent) for approximately \$248 million more (or 28 percent more) in capital cost. The unit cost for Supplemental Alternative 9 with a 3.0 Bcf tank (including ground improvement, auxiliary systems, and regasification) is approximately \$513 million per Bcf lower than the unit cost for Supplemental Alternative 4A.
- When compared to a 2 Bcf tank with the equivalent regasification capacity (Supplemental Alternative 8), Supplemental Alternative 9 (3.0 Bcf and 800 MMcf/d) provides 50 percent more storage for an additional capital cost of only 11 percent (approximately \$111 million).
 The unit cost for Supplemental Alternative 9 with a 3 Bcf tank (including ground


- improvement, auxiliary systems, and regasification) is therefore approximately
 \$135 million less per Bcf than the unit cost of a 2 Bcf tank.
- 3 Thus, the economies of scale significantly favour Supplemental Alternative 9.

4 4.5.4.1.2 SIGNIFICANT GAS SUPPLY BENEFITS

5 As explained in Section 3.3.4.2 of the Supplemental Evidence, FEI's current peaking supply 6 requirements are 1.0 Bcf paired with 200 MMcf/d. A supplemental alternative must achieve this 7 level of peaking supply to avoid curtailments of firm customers under normal operations. This 8 could be achieved through: (a) on-system LNG; (b) the regional gas supply market, with reliance 9 on future large regional infrastructure expansions; or (c) a combination of both. Where an 10 alternative involves FEI relying on the market, the annual peaking supply costs would be paid by 11 FEI's customers through the cost of gas, and these annual costs are reflected in the levelized total rate impact calculation. Having on-system LNG for peaking gas supply would either partially 12 13 or fully avoid the annual market-based gas supply costs, which represents a benefit for customers.

This section explains how FEI calculated annual market-based gas supply costs / avoided costs
 in respect of the various supplemental alternatives. FEI considers there are three separate phases
 of gas supply costs/savings over the 67-year analysis period:

- From the present to when, absent capital upgrades, LNG from Tilbury would no longer be available either due to the existing Tilbury Base Plant no longer being usable or due to an inability to continue using LNG from Tilbury 1A to supplement the Base Plant's reduced capacity (assumed to be **present to 2030**);
- The period between when, absent capital upgrades, Tilbury LNG is no longer available
 and when a sufficiently large regional infrastructure expansion could replace the lost LNG
 (assumed to be between 2030 and 2035); and
- The period thereafter, when relying on expanded regional infrastructure would cause FEI to incur higher annual tolls / charges on all capacity held on that infrastructure (assumed to be **2035 and onwards**).

27 Annual Gas Supply Costs up to 2030

Since FEI requires 200 MMcf/d and 1.0 Bcf of peaking gas supply, the existing Tilbury Base Plant is already undersized. To make up this level of peaking gas supply, the current ACP includes 150 MMcf/d and 0.6 Bcf of LNG from Tilbury, plus 50 MMcf/d of year-round pipeline capacity on T-South, which FEI then must try to resell (mitigate) during most of the year.¹⁸¹ FEI estimates the annual costs it incurs at present for holding the 50 MMcf/d of pipeline capacity on T-South yearround are approximately \$7 million, net of mitigation of the unused capacity.

In the baseline scenario where there are no capital upgrades (i.e., Supplemental Alternative 1), FEI could, if required, continue to hold 50 MMcf/d of pipeline capacity on T-South indefinitely

¹⁸¹ FEI's 2024/25 ACP provides a detailed discussion of T-South holdings.



- because it has rollover rights. However, FEI's ability to continue relying on the 0.6 Bcf / 150
 MMcf/d of LNG peaking supply in the ACP is time limited due to the following:
 - The Base Plant has already reached end-of-life (as discussed in Section 3).
- As discussed in Section 4.4.1 above, FEI is unlikely to be able to continue relying on the
 Tilbury 1A tank to augment FEI's peaking energy given the recent regulatory approval of
 the Tilbury Jetty and the anticipated deliveries to service the Port of Vancouver.

FEI has assumed that the end date for the existing Base Plant is 2030, which FEI believes is reasonable based on its deteriorating condition. As such, for the purposes of the financial analysis, FEI assumes that it would continue to incur approximately \$7 million annually (net of mitigation) for holding the 50 MMcf/d of pipeline capacity on T-South until 2030 for all supplemental alternatives (including those alternatives that would have new storage and/or regasification equipment since any new assets pertaining to each alternative would not be inservice before 2030).

14 Annual Gas Supply Costs between 2030 and 2035

As explained in Section 3.3.4.3, the regional infrastructure is already fully contracted which 15 16 precludes replacing Tilbury's existing peaking capabilities with long-term commercial contracts. 17 Once the Base Plant ceases to operate in the baseline scenario (i.e., Supplemental Alternative 18 1), firm customers would face curtailments in normal operations absent the completion of a 19 regional infrastructure upgrade that is sufficiently large to replace the lost peaking supply from 20 Tilbury. FEI believes there is no possibility of this occurring by 2030. Currently, the only planned 21 regional infrastructure upgrade that FEI is aware of is the T-South Sunrise Expansion which will 22 only offset the capacity of Woodfibre LNG and does not provide any added capacity to the region.¹⁸² FEI is also not aware of any other potential for regional market area storage upgrades, 23 24 other than the North Mist Expansion Project which is anticipated to complete in 2029. Any 25 subsequent expansion at Mist, if possible, would take multiple years to develop post 2029. 26 Therefore, FEI considers 2035 is a reasonable assumption for regional infrastructure upgrades.

- As such, in the baseline scenario there would be a period where FEI is curtailing customers instead of incurring annual peaking supply costs.
- FEI's financial analysis therefore had to make assumptions as to if and when regional infrastructure might be expanded sufficiently to meet peaking supply requirements so that FEI could cease having to curtail customers in the normal course. FEI assumed this would occur in 2035, as previously described.
- As such, for the baseline scenario (i.e., Supplemental Alternative 1), FEI continue to incur millions of dollars annually (net of mitigation) to hold the existing 50 MMcf/d of pipeline capacity, but

¹⁸² See Section 3.3.4.3.1.3 of the Supplemental Evidence.



otherwise will likely to have to curtail firm loads in normal operations until another expansion in
 regional pipeline or storage facility becomes available for FEI to contract.

3 Annual cost of holding existing 50 MMcf/d of pipeline capacity increases with 2028 4 Sunrise Expansion: Since the currently planned T-South Sunrise Expansion is expected 5 to result in tolls of \$0.95/GJ, which is a significant increase (46 percent) over current 6 embedded tolls on T-South, FEI assumes it would incur those \$0.95/GJ tolls to continue 7 to hold 50 MMcf/d of pipeline capacity that it currently uses to supplement Tilbury LNG in 8 the ACP. Paying the higher toll on 50 MMcf/d represents approximately \$17 million per 9 year of annual peaking supply costs, before mitigation. Assuming FEI's current ability to mitigate that pipeline capacity, the post mitigation cost remains at approximately \$7 million 10 11 annually.

12 Curtailments of up to 150 MMcf/d get reflected as gas cost savings: Since there is no 13 possibility for FEI to replace the lost peaking capacity from the existing Tilbury Base Plant once it ceases operation, assumed to be 2030, the financial analysis assumes no costs to 14 15 replace the lost 0.6 Bcf and 150 MMcf/d of peaking gas supply for those supplemental 16 alternatives that do not provide new replacement for Tilbury LNG (i.e., those without a new 17 LNG storage tank and no allocation for gas supply). However, as discussed above, FEI 18 expects it would have to curtail customers to reduce demand by up to 150 MMcf/d until 19 another regional expansion becomes available.

In comparison to the baseline scenario (i.e., Supplemental Alternative 1), Supplemental Alternatives 4 and 8 provide at least 150 MMcf/d of regasification, thereby avoiding curtailments. However, both of these alternatives would only provide 0.6 Bcf of LNG storage, therefore, FEI would still have to incur the estimated \$7 million post mitigation gas supply costs to augment the remaining 0.4 Bcf of peaking supply in order to meet the requirement of having an optimized portfolio of 200 MMcf/d and 1.0 Bcf of peaking supply. This results in no avoided costs for Supplemental Alternatives 4 and 8 from 2030 to 2035.

Supplemental Alternatives 4A and 9, which would have 1 Bcf of LNG storage and at least 200
MMcf/d of regasification at Tilbury for peaking gas supply purposes, would avoid the estimated
\$7 million tolls from the regional market.

30 Annual Gas Supply Costs for 2035 and Beyond

31 FEI has assumed for the purposes of this analysis that the period of curtailment following the

32 failure of the Base Plant would end when construction of a regional infrastructure (pipeline or

33 storage) expansion is large enough to meet FEI's full requirements of 1.0 Bcf and 200 MMcf/d.

34 Table 4-10 below provides the alternative resource capacity required to provide the equivalent of

35 1 Bcf and 200 MMcf/d of on-system LNG.



3

Table 4-10: Alternative Resource Capacity Required to Replace 1 Bcf / 200 MMcf/d of On-System Peaking Supply

	Annual Capcity (Bcf)	Daily Deliverability (MMcfd)
TLSE Peaking Supply	200*5/1000 =1 Bcf	200 MMcfd
1) Off-system Storage	200*10/1000 = 2 Bcf	200 MMcfd
2) T-South Expansion (Sunrise 300 MMcfd)	200 *365/1000 = 73 Bcf	200 MMcfd
3) T-South Expansion (Potential)	200 *365/1000 = 73 Bcf	200 MMcfd

4 FEI assumes the earliest upgrade in regional infrastructure (either a regional storage or pipeline 5 expansion) could be in 2035 (given the T-South Sunrise Expansion is expected to complete in 6 2028 and FEI is not aware of any other potential for regional market area storage upgrades, other 7 than the North Mist Expansion Project which is anticipated to complete in 2029). As such, for the 8 purposes of the financial analysis, FEI would begin to incur higher tolls in 2035 (whether it is from 9 new regional storage or pipeline expansion) under the baseline scenario (i.e., Supplemental 10 Alternative 1) as well as any other Supplemental Alternatives (i.e., 2, 3, 5, 6, and 7) that do not have on-system LNG storage for peaking gas supply. In contrast, for those Supplemental 11 12 Alternatives that have on-system LNG storage either fully or partially allocated for gas supply (i.e., 13 4, 4A, 8, and 9), the higher tolls in 2035 would effectively be gas supply costs savings or avoided 14 costs reflected in the calculation of the levelized total rate impact and to the benefit of FEI's 15 customers.

16 The annual costs of holding capacity on regional infrastructure will increase significantly with 17 upgrades. These capital costs get reflected in higher tolls or demand charges, which are passed 18 along to FEI customers through Storage and Transportation charges. Specifically:

- For access to regional storage like JPS and Mist, FEI pays the storage demand charges, plus pipeline tolls on Northwest Pipeline for capacity to deliver the volumes to FEI's system. The cost of JPS or Mist expansions are borne by the party that requires the incremental capacity, in this case FEI. The same is true for expansions on Northwest Pipeline. The higher storage demand charge, and associated transportation charge, are passed on to FEI customers through Storage and Transportation charges.
- In the case of a T-South expansion, due to rolled-in tolling design, FEI would pay higher
 tolls than today for <u>all</u> of its capacity (existing holdings plus new incremental holdings),
 which means that FEI's existing capacity holdings on the regional infrastructure would also
 become more expensive for customers.

As the cost of holding capacity on regional infrastructure is a function of the expected tolls and charges, the costs will be higher than they are today by virtue of the capital costs of upgrades being reflected in the tolls and charges. Therefore, as discussed below, FEI used information from currently proposed pipeline and storage expansions to derive the cost estimates for peaking resources on expanded regional pipeline or storage infrastructure. The underlying market conditions and alternatives employed are similar as that used in the confidential section 71 application recently approved by Order G-241-24.



- 1 Assuming regional storage expansion or regional pipeline infrastructure would be upgraded
- 2 sufficiently by 2035, Figure 4-7 below shows the estimated annual costs for holding peaking gas
- 3 supply of 1.0 Bcf and 200 MMcf/d could range from \$63 million to \$79 million (net of mitigation),
- 4 depending on if the regional infrastructure upgrade is storage or pipeline. For the purposes of the 5 financial analysis and comparing between all supplemental alternatives, FEI conservatively used
- financial analysis and comparing between all supplemental alternatives, FEI conservatively used
 the lower annual cost of storage shown in Figure 4-7 below for gas supply costs/savings. Using
- 7 the higher pipeline costs would have the effect of improving the levelized total rate impact of those
- 8 supplemental alternatives that meet all of FEI's peaking gas supply requirements (Supplemental
- 9 Alternatives 4A and 9) relative to those options that do not (i.e., 1, 2, 3, 4, 5, 6, 7, and 8).

10Figure 4-7: Annual Cost (Post Mitigation) of Using Expanded Regional Infrastructure to Supply11Equivalent of 1 Bcf, 200 MMcf/d



- 13 The estimated annual costs of holding peaking gas supply of 1 Bcf and 200 MMcf/d from a 14 regional storage or pipeline expansion are based on the following assumptions:
- Expected Regional Storage Cost Increase (Basis for Financial Calculations): The cost of using regional storage involves: (1) a storage demand charge; and (2) the associated transportation charge:
- 18 The assumed storage demand charge is based on a Mist storage expansion 0 project which FEI is currently involved with. FEI's current ACP portfolio includes 19 20 approximately 115 MMcf/d of Mist storage, out of which up to 50 percent capacity 21 will be recalled by the storage owner NW Natural. FEI is currently participating in 22 an expansion project in order to maintain the same level of Mist capacity in future 23 ACP portfolios. This expansion project does not have additional capacity to replace 24 the required Tilbury send out. FEI is uncertain if another expansion would be 25 feasible at the current Mist facility. Recent study of this incremental project at Mist 26 indicates the transportations charge of moving the Mist supply will be significantly



- higher. FEI has used existing tolls for the transportation charges in these avoided
 cost calculations. In addition, FEI has used the storage demand charge from the
 North Mist Expansion Project.
- 4 The associated transportation charge is the cost of obtaining pipeline capacity on 0 5 Northwest Pipeline (NWP) to deliver gas (by displacement) to FEI's service 6 territory. The expected regional storage cost increase shown in Figure 4-7 above 7 includes transportation charge based on current NWP tolls. However, the regional 8 storage facilities (JPS and Mist) are in the US, and future expansion of either storage facility will also require an expansion on the NWP system because existing 9 10 transportation capacity required to access these locations is fully contracted and utilized during the winter period. Therefore, if an expansion was required, the costs 11 12 to upgrade the NWP system would have to be paid by the expansion shippers (i.e., 13 FEI) to NWP. As such, the expansion costs to FEI would be even higher than the 14 current NWP tolls used in FEI's calculation.
- Expected T-South Toll Increases (Not Used in Financial Analysis): The T-South toll for the Huntingdon delivery area is currently \$0.65/GJ, however:
- As noted above, the currently planned 2028 T-South Sunrise Expansion is
 expected to result in tolls of \$0.95/GJ, which is a significant increase (46 percent)
 over the current embedded tolls on T-South.
- 20 Given the 2028 T-South Sunrise Expansion is only offsetting the incremental 0 21 demand in the region when Woodfibre LNG comes into service, it would not 22 provide added capacity for replacing the required on-system peaking supply from 23 Tilbury. As such, FEI expects the next expansion after the 2028 T-South Sunrise 24 Expansion will be much more costly, resulting in a further increase from the 25 expected tolls of \$0.95/GJ in 2028. FEI's internal hydraulic modeling suggests that 26 a future post-Sunrise expansion of T-South that would provide sufficient capacity 27 and energy to replace FEI's peaking supply would require approximately 200 km 28 of looping. Based on the capital cost of the T-South Sunrise Expansion, FEI 29 estimates that the T-South Long-Haul toll could increase to approximately 30 \$1.50/GJ (i.e., a 58 percent increase over the expected toll of \$0.95/GJ by 2028) 31 which FEI would have to pay year-round.
- FEI would expect to continue to seek mitigation for its gas supply costs by selling
 underutilized pipeline capacity into the market in non-peak times. Based on the
 observed T-South values between January 2016 and March 2023, FEI assumes a
 mitigation value of \$1.62/GJ for the expected T-South toll increase shown in Figure
 4-7 above. FEI also assumed that it would be able to mitigate the T-South capacity
 at this value for 120 days based on mitigation during the winter period.

Using the same methodology as outlined above and assuming sufficient regional infrastructureupgrades at that time, FEI also estimated the annual gas supply costs to replace the existing 0.6



1 Bcf and 150 MMcf/d of peaking supply from the existing Tilbury Base Plant by 2035 for the

2 financial analysis of Supplemental Alternatives 4 and 8, both of which would maintain the existing

3 0.6 Bcf and 150 MMcf/d of regasification. These amounts are shown in Figure 4-8 below, which

4 range from \$46 million to \$59 million (post mitigation), depending on whether the regional

5 infrastructure upgrade is from a storage facility or pipeline. Similar to the discussion above, FEI

6 used the lower bound of \$46 million for the gas supply cost calculations for these two alternatives.

7 Figure 4-8: Annual Cost (Post Mitigation) of Using Expanded Regional Infrastructure to Supply 8 Equivalent of 0.6 Bcf, 150 MMcf/d



9

10 Avoided Costs of Holding Capacity on Expanded Regional Infrastructure

11 In general, for those supplemental alternatives that do not have on-system LNG storage for 12 peaking gas supply, such as the baseline scenario (i.e., Supplemental Alternative 1), FEI would 13 be relying on a combination of curtailment and regional market supply to meet the peaking gas 14 requirement of 1.0 Bcf and 200 MMcf/d of regasification, thus there would be annual gas supply costs reflected in customers' rates as cost of gas and included in the calculation of total levelized 15 rate impact over the 67-year analysis period. In contrast, for those supplemental alternatives that 16 17 have on-system LNG storage either fully or partially allocated for gas supply (i.e., 4, 4A, 8, and 9), the annual gas supply costs from off-system regional market would effectively become savings 18 19 or avoided costs reflected in the calculation of the levelized total rate impact and to the benefit of 20 FEI's customers.

Table 4-11 below summarizes the annual gas supply costs for the baseline scenario (Supplemental Alternative 1) as well as the feasible alternatives that passed the Step 1 and Step screens discussed in Sections 4.3 and 4.4 above (Supplemental Alternatives 4, 4A, 8, and 9). The incremental costs/savings from the baseline scenario (i.e., the avoided costs of holding capacity on expanded regional infrastructure), which are used for the financial analysis are also shown in the table below. Although this section focuses on the results of Supplemental



- 1 Alternatives 4, 4A, 8 and 9, FEI has also provided the annual gas supply costs and incremental
- costs/savings for the remaining supplemental alternatives (i.e., Supplemental Alternative 2, 3, 5,
 6. and 7)¹⁸³ in Appendix C to this Supplemental Evidence.
- As noted above, for the purposes of the financial analysis for all supplemental alternatives, FEI conservatively used the lower annual cost of storage shown in Figures 4-7 and 4-8 above for gas supply costs/savings. Using the higher pipeline costs would have the effect of improving the levelized total rate impact of those supplemental alternatives that meet all of FEI's peaking gas supply requirements (Supplemental Alternatives 4A and 9) relative to those options that do not (Supplemental Alternatives 4 and 8).

Table 4-11: Avoided Annual Gas Supply Costs for Supplemental Alternatives 4, 4A, 8, and 9 (\$ millions)

		Annual Gas Supply Costs (\$millions)			Incremental to Baseline / (Avoided Costs) (\$ millions)		
Supplemental Alternatives	Description	Present to 2030	2030 to 2035	2035 onwards	Present to 2030	2030 to 2035	2035 onwards
1	No Capital Upgrades (Continue to rely on existing Base Plant until it fails. No on-system peaking gas supply thereafter and no resiliency reserve)	7.0	7.0	63.0			
4	Like-for-Like Replacement for 0.6 Bcf and 150 MMcf/d (No Resiliency Reserve)	7.0	7.0	17.0	-	-	(46.0)
4A	New 1 Bcf Tank and 400 MMcf/d Regasification (No resiliency reserve)	7.0	-	-	-	(7.0)	(63.0)
8	New 2 Bcf Tank and 800 MMcf/d Regasification (1.4 Bcf resiliency reserve and 0.6 Bcf for peaking gas supply)	7.0	7.0	17.0	-	-	(46.0)
9	New 3 Bcf Tank and 800 MMcf/d Regasification (2 Bcf resiliency reserve and 1 Bcf for peaking gas supply)	7.0	-	-	-	(7.0)	(63.0)

12

13 Table 4-11 above demonstrates that having new on-system storage and added regasification at

14 Tilbury would provide significant benefit in terms of avoided gas supply costs from the regional

15 market (assuming expanded regional infrastructure is available in the future).

In summary, Supplemental Alternatives 4 and 8 would partially avoid annual peaking gas supplycosts:

 Both of these alternatives would avoid curtailment of firm customers compared to Supplemental Alternative 1 as they can provide more than 150 MMcf/d of regasification. However, both alternatives would still leave FEI with a shortage of energy for peaking supply (i.e., there is still only 0.6 Bcf allocated to gas supply, whereas FEI requires 1 Bcf). As such, from 2030 to 2035, FEI would still incur gas supply costs, estimated to be approximately \$7 million post mitigation as discussed above from the regional market to augment the remaining 0.4 Bcf of peaking supply in order to meet the requirement of

¹⁸³ The financial analysis was not completed for non-viable Alternatives 10, 11, and 12.



having an optimized portfolio of 200 MMcf/d and 1.0 Bcf of peaking supply. This results in
 no avoided costs for Supplemental Alternatives 4 and 8 from 2030 to 2035.

 For 2035 and onwards, given both Supplemental Alternatives 4 and 8 would include onsystem LNG storage with gas supply allocation of 0.6 Bcf and 150 MMcf/d, FEI's customers would avoid annual gas supply costs from the expanded regional infrastructure for the equivalent amount of supply that could range from \$46 million to \$59 million as shown in Figure 4-8 above (conservatively using the lower end of \$46 million for the purpose of the financial analysis).

- However, since both of these alternatives would only have 0.6 Bcf for gas supply purposes, they still require FEI to hold peaking gas supply capacity from regional infrastructure from 2035 onwards for the remaining 0.4 Bcf in order to meet the optimized portfolio requirement. As such, the avoided gas supply costs for Supplemental Alternatives 4 and 8 would be less than the avoided gas supply costs for Supplemental Alternative 4A and 9 as shown in Table 4-11 above.
- 15 In contrast, Supplemental Alternatives 4A and 9 would avoid all annual peaking gas supply costs:
- Since Supplemental Alternatives 4A and 9 would have 1 Bcf and at least 200 MMcf/d of regasification at Tilbury for peaking gas supply purposes, there is no need to incur additional gas supply costs, thereby avoiding the tolls from the regional market estimated to be approximately \$7 million from 2030 to 2035.
- As reflected in the financial analysis, the additional 50 MMcf/d and 0.4 Bcf provided by
 Alternatives 4A and 9 creates the flexibility to shed other supply resources in the ACP, all
 else equal.
- For 2035 and onwards, given Supplemental Alternatives 4A and 9 include on-system LNG storage with gas supply allocation that already meet the optimized portfolio of 1.0 Bcf and more than 200 MMcf/d of regasification, FEI's customers would avoid annual gas supply costs from the expanded regional infrastructure for the equivalent amount of supply that could range from \$63 million to \$79 million¹⁸⁴ as shown in Figure 4-7 above (conservatively using the lower end of \$63 million for the purposes of the financial analysis).

¹⁸⁴ In September 2021, in the response to BCUC IR1 46.2 (Exhibit B-15), FEI estimated that the financial value to customers of the "third Bcf" (i.e., the storage available in a 3 Bcf tank over and above the 2 Bcf retained as a resiliency reserve) combined with the proposed regasification capacity was approximately \$30 million per year. FEI estimated this value based on the T-South market value in 2021 net of winter mitigation revenue, which was close to the 2021 T-South toll for 365 days a year. That is, by having 1 Bcf of LNG available, FEI would be avoiding \$30 million in annual pipeline tolls. FEI explained that this calculation understated the benefits by virtue of being based on the assumption that FEI was only avoiding the T-South embedded toll, whereas in reality FEI would need to pay a premium. As discussed in Section 3.3.2.3. of this Supplemental Evidence, there have been several market changes since 2021 including: (1) Woodfibre making a final investment decision; (2) WEI's new T-South expansion project (i.e., the T-South Sunrise Expansion) was fully contracted; and (3) the winter of 2022/23 showed market participants that the costs for energy in western North America are high due to, for instance, gas-fired electricity demand and coal plant retirements. FEI's Supplemental Evidence financial analysis reflects these significant market developments.



14.5.4.2Summary of Levelized Total Rate Impact Results for the Four Feasible2Supplemental Alternatives (4, 4A, 8 and 9)

3 In summary, for the criterion of comparing the Levelized Total Rate Impact over the expected life 4 of the assets of each feasible Supplemental Alternative (4, 4A, 8, and 9), the results confirm that 5 it makes financial sense for customers to build for more gas supply capabilities than currently 6 provided by Tilbury in order to meet FEI's full peaking supply requirements of 200 MMcf/d and 7 1.0 Bcf. Although Supplemental Alternative 4A would have the lowest levelized total rate impact 8 out of the four feasible supplemental alternatives, it would only be sized for meeting FEI's peaking 9 supply requirements in an optimal manner, whereas Supplemental Alternative 9 will be able to 10 provide the required peaking supply in an optimal manner while also significantly mitigating 11 customer outage risk.

- 12 The results of the financial analysis indicate the following:
- Supplemental Alternative 4 would be the smallest facility (offering the same peaking gas supply capabilities installed in 1971 with no resiliency reserve), and yet would not have the lowest total rate impact. It would have a lower capital cost; however, it would necessitate FEI continuing to supplement its undersized LNG peaking supply with an additional 50 MMcf/d and 0.4 Bcf from the market at a cost of \$7 million to \$17 million annually.
- Supplemental Alternative 4A would have the lowest levelized total rate impact of the four alternatives because it is sized to provide the optimum amount of peaking gas supply, without mitigating FEI's largest customer outage risk. It would have a higher capital cost than Supplemental Alternative 4 because it is a larger facility; however, its larger gas supply reserve and expanded regasification capacity would avoid Supplemental Alternative 4's annual gas supply costs.
- Supplemental Alternative 8 would have the highest levelized total rate impact of any of the four alternatives, despite its 2 Bcf tank being one-third smaller than that of Supplemental Alternative 9. Its levelized total rate impact would be highest because: (1) the difference in capital cost between a 2 Bcf tank and a 3 Bcf tank is relatively small due to economies of scale; and (2) as with Supplemental Alternative 4, Supplemental Alternative 8 would require FEI to continue to acquire an additional 50 MMcf/d and 0.4 Bcf of peaking resources from the market at a cost of \$7 million to \$17 million annually.
- 32 Supplemental Alternative 9, the largest facility, will optimize the gas portfolio like 33 Supplemental Alternative 4A, while also optimizing the risk mitigation against a winter T-34 South no-flow event. The levelized total rate impact is higher than Supplemental Alternatives 4 and 4A (which do not provide resiliency), but lower than Supplemental 35 36 Alternative 8 (which would provide less resiliency and less gas supply). As explained in 37 Section 4.5.4.1.1, the economies of scale are very significant up to a 3 Bcf tank, and FEI 38 would avoid the need to incur annual peaking supply costs in the market to provide the 39 required supply.



- Table 4-12 below also provides the risk reduction per dollar of rate impact between Supplemental
 Alternatives 4, 4A, 8, and 9 based on the risk reduction values from Exponent's analysis at
- 3 average winter conditions.¹⁸⁵
- 4

	Supplemental	Supplemental	Supplemental	Supplemental
Parameter	Alternative 4	Alternative 4A	Alternative 8	Alternative 9
(1) 67-Year Expected GDP				
Loss Reduction (\$millions)	-	-	10,877	11,093
(2) Total PV of Cost of				
Service (\$millions)	424	375	768	723
Ratio (1)/(2)	-	-	14.2	15.3

6 As demonstrated in the table above, Supplemental Alternative 9 offers a lower cost than

7 Supplemental Alternative 8, and greater risk reduction per dollar spent than Supplemental

8 Alternatives 4, 4A, or 8.¹⁸⁶

9 **4.5.5 Future Use Scoring Criterion**

In the Adjournment Decision, the BCUC identified a need for FEI to further consider the potential for the transition towards a lower carbon future to affect the appropriate sizing of the TLSE Project.¹⁸⁷ FEI's "Future Use" criterion evaluates the degree to which an alternative will be useful for FEI's own resiliency and gas supply portfolio (i.e., to serve load or generate mitigation revenue), and its potential to be underutilized.

FEI acknowledges the uncertainty in forecasting future load over a long period of time. To that end, as discussed in this section, FEI has assessed the future use for the viable Supplemental Alternatives (4, 4A, 8 and 9) using two hypothetical adverse load sensitivities that modify FEI's DEP Scenario from the 2022 LTGRP. Specifically, these sensitivities reflect significant hypothetical customer losses in the Lower Mainland of 2 percent and 5 percent per year between 2030 and 2050 (the "modified DEP" sensitivities are abbreviated as mDEP 2% and 5%).

Even under the most adverse hypothetical sensitivity (mDEP 5%), FEI would still be serving
hundreds of thousands of customers in the Lower Mainland in 2050. Customers in the Lower
Mainland and FEI's other service areas would still need peaking supply. The Lower Mainland
would (absent mitigation) still be exposed to a significant customer outage – with all of the

¹⁸⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Section 8.3, Table 9.

¹⁸⁶ The resiliency performance of Supplemental Alternative 9 at temperatures below the average winter temperature is not captured in Table 4-12. At colder temperatures, specifically between to -6.8°C to +1.7°C, the difference in the value ratio for Supplemental Alternatives 8 and 9 is expected to be much larger.

¹⁸⁷ Adjournment Decision, p. 52. "Further, if the throughput of natural gas is reduced due to a decrease in demand, the size of a tank and the amount of regasification required would likely be reduced." And p. 22: "The larger tank provides flexibility to accommodate future load growth that may occur. However, given the current emphasis on electrification and decarbonization in BC, it is unclear whether FEI will experience significant, or even any, future natural gas load growth. The larger tank means greater risk of a stranded, or partially stranded, asset in the event that FEI's increased load does not emerge or decreases beyond the current load."



- associated social, human health and economic consequences following a no-flow event on T South.
- The hypothetical adverse load loss sensitivities illustrate how an on-system LNG facility is a unique asset when it comes to the flexibility afforded in response to changing load. There would be a spectrum of choices available to FEI in terms of how to reallocate the tank as between a
- 6 resiliency reserve and gas supply:
- 7 One end of the spectrum would be to maximize resiliency, for instance by maintaining the 8 same resiliency reserve to achieve progressively more customer outage risk reduction. 9 The resiliency value of on-system LNG increases if load declines (other things equal) 10 because the facility will support less load for a longer period of time following a no-flow 11 event. There is still residual risk with any of the Supplemental Alternatives at the current 12 load. FEI's system modelling and Exponent's risk calculations indicate that a longer load 13 support duration under hypothetical reduced load sensitivities would be valuable for the 14 remaining customers in 2050.
- 15 On the other end of the spectrum, for Supplemental Alternatives that provide a resiliency 16 reserve from the outset. FEI could progressively reallocate some of the resiliency reserve 17 to gas supply. FEI could maintain a consistent level of risk mitigation over time, while 18 realizing additional gas supply benefits for all FEI customers. FEI reoptimizes its ACP 19 portfolio annually based on the available supply options. The additional LNG available for 20 gas supply could provide optionality in future gas supply portfolio by either replacing other 21 gas supply resources or displacing other peaking supply¹⁸⁸ that is exposed to potential 22 commodity price spikes on cold days.
- In either case, customers receive value from the TLSE Project. As a result, all four of the viable
 Supplemental Alternatives (4, 4A, 8 and 9) received the same score, "No impact", reflecting that
 stranding risk does not increase relative to retaining the existing Base Plant with no capital
 upgrades (Supplemental Alternative 1), which is not viable.

4.5.5.1 Defining Usefulness vs. Underutilization: Whether the Alternative Provides Excessive Resiliency or Gas Supply

FEI considered two illustrative "book-end" approaches to reallocating the TLSE Project's capabilities in the face of hypothetical declining load – resiliency maximization and resiliency retention. The focus of the future use analysis was to determine whether, under either "bookend" approach, FEI's customers would be faced with underutilized resiliency capability or underutilized gas supply capability in 2050 based on the mDEP (2% and 5%) adverse sensitivities.

¹⁸⁸ FEI's gas supply portfolio includes commercial arrangements (i.e., peaking call options), which are priced based off a daily index. Gas prices generally increase when the region experiences a cold weather event. Having additional on-system LNG allows gas supply to displace the peaking supply received from the market and reduce the costs to customers.



- For the resiliency maximizing strategy, FEI determined the load support duration under average winter conditions using the 2050 Year mDEP (2% and 5%) load scenario and maintaining the current resiliency and gas supply allocations defined for each alternative; and
- For the resiliency retention strategy, FEI determined the resiliency reserve volume required such that, using the 2050 Year mDEP (2% and 5%) load scenario, the alternative would provide the same support duration as under current year load. As this would result in a lower resiliency reserve volume, and thus a larger gas supply reserve volume, this approach assessed if a Supplemental Alternative would be underutilized for gas supply.
- FEI and Exponent modelled Supplemental Alternatives 8 and 9, since they would be the largest facilities and any smaller facilities would necessarily be useful if the largest facilities are useful. Alternatives 4 and 4a were not included in this analysis because they are significantly smaller than 8 and 9. If the larger alternatives prove to be useful under a future scenario, then both the smaller alternatives would also be useful. The system results of each "bookend" strategy are set out in Sections 4.5.5.3 and 4.5.5.4 below.

16 4.5.5.2 Approach and Inputs for mDEP Hypothetical Adverse Sensitivities

FEI's hypothetical adverse sensitivities modified FEI's DEP Scenario from the 2022 LTGRP in the manner described below to estimate the customer numbers, and annual and peak load from 2024 to 2050. FEI then used the results from these hypothetical mDEP sensitivities in transient modelling, as described in Section 3, to determine how many days of load support a 2 Bcf and 1.4 Bcf resiliency reserve would provide in 2050. This also informed how much of the 2 Bcf and 1.4 Bcf resiliency reserves could be re-allocated to FEI's gas supply portfolio.

23 4.5.5.2.1 <u>2050 CUSTOMER COUNT IN HYPOTHETICAL ADVERSE SCENARIOS</u>

Starting with the DEP Scenario, which includes FEI's core customers in Rate Schedules (RS) 1
 to 5 and 7 in the Lower Mainland, FEI adjusted the customer count for these hypothetical adverse
 sensitivities as follows:

- FEI adjusted the customer count downward by 40,000 which is the approximate number of customers (categorized as Lower Mainland in the DEP Scenario) that FEI serves who are east of FEI's interconnection with T-South (Huntingdon). FEI made this adjustment because gas resiliency at Tilbury would not directly benefit these customers because FEI cannot flow gas eastward into T-South at Huntingdon.
- FEI assumed that new residential and commercial customers would continue to connect to FEI's gas distribution system until the year 2030. After 2030, commencing in 2031, FEI assumed it would stop adding new customers and residential and commercial customers would begin to decrease by either 2 percent or 5 percent per year. The 2 percent decline assumption (mDEP 2%) is a modification from the DEP Scenario, but is in line with the estimated demolition rate provided in FEI's 2022 LTGRP (Appendix C-1 to the 2022)



- LTGRP, page 34). FEI considers the 5 percent assumption (mDEP 5%) to be an extreme
 hypothetical sensitivity.
- FEI did not adjust the number of industrial customers, which account for approximately
 0.1 percent of FEI's customers and approximately 30 percent of FEI's load.
- FEI did not include volumes of gas associated with RS 23 or 25 in the Lower Mainland.
 The results are therefore understated given that RS 23 and 25 are firm delivery; however,
 they buy their gas from a third party where FEI takes possession at its interconnections
 with upstream pipelines.
- 9 Figure 4-9 below shows the 2050 customer results of the hypothetical sensitivity using a 2 percent
- annual decrease in customers (mDEP 2%). FEI would still be serving approximately 400,000
- 11 customers in the Lower Mainland in 2050.

12Figure 4-9: Lower Mainland 2050 Customers at 2 Percent Customer Decrease Per Year (mDEP 2%13Adverse Sensitivity)



14

15 Figure 4-10 below shows that, even hypothetically assuming an extreme 5 percent customer

- 16 decrease per year starting in 2031, FEI would still be serving approximately 220,000 customers
- 17 in the Lower Mainland in 2050.





4 4.5.5.2.2 2050 ANNUAL LOAD IN HYPOTHETICAL ADVERSE SENSITIVITIES

FEI's two hypothetical adverse mDEP (2% and 5%) sensitivities use the use per customer (UPC)
included in the DEP Scenario, which includes UPCs to 2042. FEI trended UPC to 2050 using the
prior 10 years of forecast UPC from the DEP Scenario (2033 – 2042) and then multiplied the
UPCs by the number of customers as discussed above, resulting in an estimated Lower Mainland
load to 2050.

10 As shown in Figure 4-11 below, FEI's load in the Lower Mainland would still be substantial at 80

11 PJ under the mDEP 2% sensitivity.





Figure 4-11: Lower Mainland 2050 Gas Load at 2 Percent Customer Decrease Per Year (mDEP 2% Adverse Sensitivity)



4 When FEI uses a 5 percent customer decrease per year starting in 2031 (mDEP 5%), the Lower

5 Mainland load would be approximately 60 PJ per year, or approximately half of the 2024 level.

Figure 4-12: Lower Mainland 2050 Gas Load at 5 Percent Customer Decrease Per Year (mDEP 5%
 Adverse Sensitivity)





1 4.5.5.2.3 2050 PEAK DEMAND IN HYPOTHETICAL ADVERSE SENSITIVITIES

- 2 FEI used the same transient modelling process described in Section 3 of this Supplemental
- 3 Evidence to determine how long a 2 Bcf resiliency reserve at Tilbury would last following a winter
- 4 T-South no-flow event in a hypothetical future where there are fewer customers and lower load in
- the Lower Mainland. As the risk being addressed is a winter no-flow event, the transient modelling
 requires a load duration curve for mDEP 2% and 5% that shows FEI's load profile throughout the
- 7 year. FEI prepared the load duration curve in the manner described below.
- 8 FEI started with the Lower Mainland's year 2041 load duration curve (LDC) developed for FEI's
- 9 2024/25 ACP so that FEI could consider what resources it may require in the future by matching
- 10 resources with the characteristics of FEI's demand. FEI further trended the LDC from 2042 to
- 11 2050. As with the UPCs described above, FEI trended each day's demand to 2050 using the prior
- 12 10 years of forecast (2032 to 2041). In order to apply the LDC (trended to 2050), which does not

13 account for a per year loss in residential and commercial customers from 2031 onward, FEI used

14 the trended LDC to proportion the annual load, as determined above, across 365 days.

15 Figure 4-13 shows the load duration curve over the gas year for 2030 and 2050 under the mDEP

16 2% sensitivity. The peak day demand for the mDEP 2% sensitivity would be 874.9 MMcf/d in 2030

17 and 581.3 MMcf/d in 2050.



18Figure 4-13: Lower Mainland Load Duration Curve 2030 & 2050 Showing Peak Day Demand19Assuming 2% Per Year Customer Decrease (mDEP 2% Adverse Sensitivity)

20

Assuming a 5 percent per year decrease in customers (mDEP 5%), which FEI considers to be an extremely adverse hypothetical sensitivity, the peak day demand would fall to 459.9 MMcf/d

23 in 2050.



Figure 4-14: Load Duration Curve 2030 & 2050 Showing Peak Day Demand Assuming 5% Per Year 2 Customer Decrease (mDEP 5% Adverse Sensitivity)

Million Cubic Feet

1

4.5.5.3 Option of Maximizing Resiliency in 2050 Would Still Be Providing 4 5 Significant Risk Mitigation Under Adverse Sensitivities

6 As noted previously, there is a spectrum of potential approaches that FEI could take over time in 7 response to changing load. The resiliency maximization approach is one potential strategy that 8 represents a "bookend" on the spectrum. Under this strategy, each Supplemental Alternative 9 would provide a longer load support duration in 2050 under the two adverse load loss sensitivities 10 when compared to the current load. However, none of the Supplemental Alternatives would be 11 underutilized for resiliency under a resiliency maximization strategy. As discussed below, the load 12 support duration for the largest facility under the most adverse load loss sensitivity would be akin 13 to what is provided by Mt. Hayes on Vancouver Island today (assuming it has LNG volumes 14 available for resiliency in the event of a supply outage). This translates into a meaningful and 15 valuable customer outage risk reduction in 2050.

16 4.5.5.3.1 FEI USED TRANSIENT MODELLING TO DETERMINE HOW LONG THE SAME RESILIENCY RESERVE 17 WOULD SUPPORT LOAD IN 2050

- 18 FEI undertook the same type of transient modelling described in Section 3 of this Supplemental
- 19 Evidence to determine the load support duration for the Lower Mainland at average winter
- temperatures (+4°C), assuming 2 Bcf and 1.4 Bcf resiliency reserves at Tilbury (i.e., Supplemental 20
- 21 Alternatives 9 and 8, respectively) with consideration for the lower number of customers and the

FORTIS BC^{*}



- estimated change in peak load. FEI focused on these two alternatives as they have the largest
 resiliency reserves and longest load support durations.
 - The table below sets out the assumed number of customers in 2050, the estimated annual demand, the estimated daily demand under average winter conditions (+4°C), and the duration of load support that a 2 Bcf and 1.4 Bcf resiliency reserve at Tilbury would provide to the Lower Mainland under the two hypothetical mDEP (2% and 5%) sensitivities.¹⁸⁹ As shown in the table, the load support duration increases as customers and load decrease. That is, with sufficient regasification capacity in place from the outset, the same volume of LNG would be expected (other things being equal) to serve reduced load for a longer period of time following a winter no-
- 10 flow event.

11 Table 4-13: Resiliency Reserve Support Under Two Hypothetical Customer Loss Sensitivities

Parameter	2050 mDEP (2%)	2050 mDEP (5%)
Lower Mainland Customers	409,831	220,395
Annual Throughput (TJ)	78,847	62,380
Lower Mainland Load +4°C (MMcf/d)	406	321
Supplemental Alternative 8 Lower Mainland Support Duration +4°C	7 days and 19 hours	9 days and 20 hours
Supplemental Alternative 9 Lower Mainland Support Duration +4°C	10 days and 20 hours	13 days and 14 hours

12

3

4

5 6

7

8

9

13 The transient modelling outputs for Supplemental Alternative 9, which is the largest facility among

14 all viable alternatives, are shown in Figures 4-15 and 4-16 below.

¹⁸⁹ To perform this analysis FEI had to reduce the number of customers within its system planning models. FEI assumed that the change in customers, related to these two hypothetical sensitivities, occurred proportionately across its system in the Lower Mainland.







3 4

5

Figure 4-16: Impact to Lower Mainland due to Loss of T-South Supply with 5% per Year Customer Loss (mDEP 2% Adverse Sensitivity)



6

As long as there is sufficient regasification capacity, under average winter conditions, 2 Bcf of
LNG earmarked for resiliency will initially (i.e., upon commissioning) allow FEI to maintain service
to 100 percent of firm Lower Mainland customers for approximately 4.5 days. If the peak load in
the Lower Mainland were to decline, as calculated with the 5 percent customer reductions



- 1 sensitivity (mDEP 5%), the same 2 Bcf resiliency reserve could support the load for approximately 2 13.5 days.
- 3 As discussed in the Exponent Report, there are various modes of failure that could result in a no-
- flow event longer than 4.5 days.¹⁹⁰ Some of those modes of failure (e.g., flooding) are events that 4 could increase in likelihood with climate change. Continuing to have access to the same resiliency
- 5 6 reserve decreases the potential for catastrophic social, health and economic harm to result from
- 7 those modes of failure.
- 8 By 2050, the support duration provided by Supplemental Alternatives 8 and 9 under the mDEP 9 (2% and 5%) adverse load sensitivities would increase to be more in line with the support currently 10 provided by Mt. Hayes LNG to Vancouver Island customers. The longest load support duration 11 under the most adverse hypothetical sensitivity (mDEP 5%) is 13 days and 14 hours, associated 12 with Supplemental Alternative 9. This duration is useful given the range of no-flow events 13 considered in the 2024 Resiliency Plan (i.e., the load support duration does not increase to the 14 level of surpassing the no-flow durations considered in the analysis). 15 While customers and load/peak load decrease, there would still be 220,000 – 410,000 customers
- 16 left in the Lower Mainland in 2050 under the hypothetical adverse sensitivities (respectively,
- 17 mDEP 5% and 2%). All of these customers would lose service if a T-South no-flow event occurred.
- 18 In the intervening years between when the TLSE Project is constructed and 2050, the Project
- 19 would be providing resiliency support for hundreds of thousands of customers in the Lower
- 20 Mainland. These customers would all be exposed to the significant social, health and economic 21 impacts.

22 4.5.5.3.2 EXPONENT'S 2050 RISK CALCULATIONS SHOW SIGNIFICANT RISK MITIGATION

23 Exponent calculated the risk mitigation provided in 2050, measured by customer outage-days, 24 using the parameters of the two hypothetical sensitivities (mDEP 2% and 5%). As shown in the 25 following figure, the risk mitigation benefit for customers is material relative to the baseline.¹⁹¹ For this purpose, the baseline represents the scenario where there is no available LNG volume at 26 27 Tilbury for resiliency use and no linepack. To simplify the analysis, FEI instructed Exponent to 28 consider only the impact to the Lower Mainland due to a T-South failure (i.e., customers that 29 would be impacted by a T-South failure in other regions such as Vancouver Island and the Interior

30 are not included in this specific analysis).

¹⁹⁰ Refer to Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Report Appendix U, Table U.3 for the noflow durations considered in the 2024 Resiliency Plan.

¹⁹¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Report Appendix X, Figure X.4.



1Figure 4-17: Expected Annual Customer Outage-Days With and Without Mitigation from TLSE2Project – Current and Hypothetical Future Adverse Load Sensitivities (mDEP 2% and 5%)



T-South (AV-1, -2, -3, and -54)

3

4 4.5.5.4 Resiliency Retention Strategy – Reallocating Part of Resiliency Reserve to 5 Gas Supply Would Avoid More Gas Supply Costs in 2050

6 As discussed above, if load declines, FEI could elect to retain the initial level of resiliency and 7 reallocate some of the resiliency reserve to gas supply. The allocation of more of the tank to the 8 gas supply portfolio would create opportunities for FEI to optimize its gas supply portfolio for the 9 benefit of customers. These benefits could come from substituting LNG for other resources or 10 generating mitigation revenue by making peaking supply available in the market.

FEI also used transient modelling to determine what size of Tilbury resiliency reserve would be required in 2050 under the adverse load loss scenarios to maintain the same support duration of 4.5 days (at average winter temperatures) for the approximately 600,000 customers currently in the Lower Mainland. Table 4-14 shows the results, along with how much LNG would then be available for gas supply if that approach was taken.



 Table 4-14: LNG Volume Required Under the 2050 mDEP (2% and 5%) Load – Resiliency

 Retention Approach (i.e., Maintain Equivalent Support Duration as Current Year)

Parameter	Alternative 8	Alternative 9
Support Duration Under Current Load +4°C	3 days and 8 hours	4 days and 13 hours
Current Load Resiliency Reserve Volume	1.4 Bcf	2.0 Bcf
Current Load Gas Supply Volume	0.6 Bcf	1.0 Bcf
Target Support Duration Under 2050 mDEP (2%) Load Sensitivity +4°C	3 days and 8 hours	4 days and 13 hours
2050 mDEP (2%) Load Sensitivity – Required Resiliency Reserve to Achieve Target Support	0.53 Bcf	0.76 Bcf
2050 mDEP (2%) Load Sensitivity – Resulting Gas Supply Volume	1.47 Bcf	2.24 Bcf
Target Support Duration Under 2050 mDEP (5%) Load Sensitivity +4°C	3 days and 8 hours	4 days and 13 hours
2050 mDEP (5%) Load Sensitivity– Required Resiliency Reserve to Achieve Target Support	0.39 Bcf	0.58 Bcf
2050 mDEP (5%) Load Sensitivity– Resulting Gas Supply Volume	1.61 Bcf	2.42 Bcf

3

4 The largest gas supply allocation under the most adverse hypothetical sensitivity (mDEP 5%) is

5 2.42 Bcf and is associated with Supplemental Alternative 9. As explained below, the gas supply

6 volume is considered useful because FEI has opportunities to optimize its portfolio by substituting

7 this volume for other contracted resources in FEI's supply portfolio, or at a minimum by selling

8 peaking supply in the market to generate mitigation revenue. The volume is less than the

9 established market demand volume of 3 to 4 Bcf.¹⁹² Given that the 2.42 Bcf available for gas

10 supply in Supplemental Alternative 9 will be useful in the future, the smaller Supplemental

11 Alternatives would also be useful.

12 4.5.5.4.1 ON-SYSTEM LNG CAN SUBSTITUTE FOR OTHER SUPPLY RESOURCES DURING ANNUAL SUPPLY 13 PORTFOLIO OPTIMIZATION

FEI conducts portfolio optimization annually to provide an outlook for future resource requirements. Maximizing the utilization of the TLSE Project would provide FEI with valuable supply portfolio flexibility, allowing it to adjust other elements of its supply portfolio to meet the shifting load profile, and is supported by the expert evidence of Raymond Mason.

The objective of the portfolio optimization process is to develop a cost-effective portfolio to meet FEI's design load, which helps FEI to contract supply, transportation and storage capacities with the resources available in the region. The outcomes of the portfolio optimization help FEI to adjust the resource mix by including new resources or de-contracting resources that are no longer required. One of the key objectives of the portfolio is to have flexibility in the resource mix and contract terms over time. For example, some of FEI's storage service agreements have short

¹⁹² Refer to Appendix F, Raymond Mason Report, pp. 6 and 36.



- terms, which require FEI to negotiate renewals after a few years. The flexibility in resource mix and contract terms facilitates adjusting the resources over time to match changing customer
- 3 demand and/or the resources available in the marketplace.
 - For context, 1 Bcf could compensate for approximately 20 percent of FEI's current off-system storage holdings. As noted in Table 4-14 above, the largest gas supply allocation under the most adverse hypothetical sensitivity (mDEP 5%) is 2.42 Bcf and is associated with the Supplemental Alternative 9. FEI expects it will be able to make use of the remainder under even the largest of the Supplemental Alternatives for gas supply portfolio purposes by adjusting other resources holdings as they come up for renewal given the different contract terms within the portfolio.
- Mr. Mason also noted the potential for resource substitution within the gas supply portfolio if the demand profiles of FEI's customers were to shift over time (i.e., lowering annual demand while maintaining the need for winter supply).¹⁹³ Resource substitution within FEI's portfolio can be attractive because of the favourable attributes of on-system LNG in normal operations, described
- 14 by Mr. Mason as follows:¹⁹⁴

5

6

7

8

9

- 15 On-system peaking resource(s) must be able to provide (at a minimum) continuous 16 availability to bridge any gaps in winter weather events. As such, in anticipation of 17 a forecasted weather event, a peaking resource can remain idle (i.e., operational 18 ready to deploy gas supply) while actual weather unfolds and can typically be 19 ramped up within hours. This operational flexibility enhances the assets value 20 compared to an alternate resource that requires 24-hours notice to be deployed 21 (e.g., off-system storage and mainline transportation) or if there is an unplanned 22 operational mainline transportation disruption. For example, when a utility elects 23 to nominate an off-system resource and the weather forecast doesn't materialize 24 as expected, it will be left with excess gas that will require mitigation. After 25 mitigation, the off-system resource will be depleted should future weather events 26 or unplanned outages occur. This contrasts with an on-system peaking resource 27 which can be deployed when access to gas supplies from mainline transportation or off-system storage faces an unplanned outage. 28
- Mr. Mason also observed that the TLSE Project "would provide FEI with operational backup for
 disruption...to existing off-system storage and/or mainline transmission."¹⁹⁵

31 4.5.5.4.2 GAS SUPPLY MITIGATION OPPORTUNITIES TO GENERATE VALUE FOR CUSTOMERS

- 32 At minimum, FEI could generate mitigation revenue for customers to offset gas supply costs.
- 33 FEI routinely realizes gas supply mitigation revenues for customers from gas supply portfolio
- 34 elements that it does not require on any given day, and its approach to the TLSE Project would
- 35 be no different. Peaking capacity is scarce and valuable in the Pacific Northwest market, where

¹⁹³ Appendix F, Raymond Mason Report, pp. 5 and 35.

¹⁹⁴ Appendix F, Raymond Mason Report, p. 17.

¹⁹⁵ Appendix F, Raymond Mason Report, pp. 5 and 35.



- energy demand has grown, and utilities are making significant long-term investments in natural
 gas-fired generation.
- 3 Mr. Mason summarized this opportunity as follows:¹⁹⁶

4 The proposed TLSE Project's proximity to a larger US export market could also be 5 a benefit to FEI's customers. In particular, peer utilities in the US, servicing a 6 growing gas-powered electricity and industrial demand are, and will be for the 7 foreseeable future, evaluating the cost/benefit of limiting pricing exposure to 8 Huntingdon/Sumas, in response to growing peaking demand, by either: (a) 9 sourcing peaking services like what the proposed TLSE Project could provide; or 10 (b) committing to long-term mainline transportation.

Mr. Mason has undertaken an analysis of the financial value of LNG storage in the Lower Mainland from the perspective of mitigation (i.e., FEI selling peaking capacity into the market to entities throughout the Pacific Northwest region). Mr. Mason's opinion is that LNG storage in the Lower Mainland will continue to have financial value regardless of how FEI's own customer demand evolves:¹⁹⁷

16 Assuming that the TLSE Project is constructed, and FEI were to have spare 17 capacity that is not required to meet customer demand or resiliency, FEI would be 18 able to generate revenues to offset the cost of service of the facility by selling its 19 excess supply into the market. Based on my assessment of the available supply and demand in the Huntingdon/Sumas natural gas market, and assuming current 20 21 market conditions persist, I expect the daily market can reasonably absorb 300-22 400 MMcf/d of natural gas across multiple days (e.g., 10 days) during winter 23 without influencing daily prices in a manner that could limit monetization values 24 (i.e., materially decreasing the revenues generated through mitigation into the 25 market).

Mr. Mason explains why, in his view, peaking resources will remain important throughout the Pacific Northwest region, such that there will be a market for any LNG that is surplus to FEI's requirements.¹⁹⁸ The volume is less than the established market demand volume of 3 to 4 Bcf.¹⁹⁹

Mr. Mason provides estimates of the potential gas supply mitigation value of various assumed LNG volumes at Tilbury using forward prices, while acknowledging that some assumptions must be made about market conditions in the future. The analysis supports FEI's belief that, at minimum, LNG surplus to FEI's own requirements could generate significant gas supply mitigation revenue to benefit customers.

¹⁹⁶ Appendix F, Raymond Mason Report, p. 5.

¹⁹⁷ Appendix F, Raymond Mason Report, pp. 6 and 36.

¹⁹⁸ Appendix F, Raymond Mason Report, pp. 36-43.

¹⁹⁹ Appendix F, Raymond Mason Report, pp. 6 and 36.



- 1 Mr. Mason's calculations assume 300 MMcf/d of regasification to reflect a reasonable proxy for
- 2 what the market could absorb on a given day, and various LNG volumes, e.g., 0.9 Bcf (3 days x
- 3 300 MMcf/d), 1.5 Bcf (5 days x 300 MMcf/d, and 3 Bcf (10 days x 300 MMcf/d). They are presented
- 4 before adding a standing demand charge, which would be a significant additional revenue stream
- 5 if FEI was able to commit to providing any of that capacity on a firm basis.

6 The figure below, using data from the Raymond Mason Report, shows annual and five-year 7 cumulative values for the various mitigation scenarios.²⁰⁰ The highest mitigation values (\$73.0 8 million to \$78.8 million) are predicated on selling 3 Bcf over 10 peak days in the winter, each year. 9 In practice, FEI expects to retain some of the LNG for resiliency and gas supply, so there is a low 10 likelihood of needing to mitigate 3 Bcf.²⁰¹ However, even the results for 1.5 Bcf (300 MMcf/d over 11 5 days) suggest a high mitigation value for customers: \$36.5 million to \$39.4 million over five 12 years, before accounting for any revenue from a standing demand charge.

13Figure 4-18: Gas Supply Cost Mitigation Scenarios – Selling Peaking Supply in the Market – 5-14Year Cumulative Value



- 16 Mr. Mason estimates that FEI could generate the following range of <u>additional</u> incremental value
- 17 through an annual standing demand charge: \$5.2 million to \$7.0 million for every 50 MMcf/d.²⁰²

 $^{^{\}rm 200}\,$ Based on forward markets dated February 29, 2024.

²⁰¹ Achieving the \$73.0 million to \$78.0 million would also require that FEI has full access to the existing liquefaction at Tilbury each year so that FEI can completely refill the 3 Bcf tank after emptying it in year one. The lower Scenario #7 results in Figure 4-18 are applicable if FEI is assumed to be limited to the 5 MMcf/d liquefaction currently allocated to the Base Plant (the rest being allocated to Tilbury 1A), which slows the pace of refilling the tank once sold. The smaller volume scenarios are not subject to the liquefaction constraint.

²⁰² Appendix F, Raymond Mason Report, pp. 7 and 58 and Report Appendix C, pp. 10-11.



1 4.6 SUMMARY OF ALTERNATIVES ANALYSIS RESULTS

- 2 Figure 4-19 and Table 4-15 below summarize the results from each step of the alternative
- 3 selection process that led to Supplemental Alternative 9 being confirmed as the Preferred
- 4 Alternative. The results from each step are discussed in the following sections.
- 5

Figure 4-19: Results from Structured Process to Identify the Preferred Alternative



Table 4-15: Results from Three-Step Process to Identify the Preferred Alternative

SUMMARY OF RESULTS						
Supp. Alt. #	Name Step 1 Results		Step 2 Results	Step 3 Overall Score		
	Alternatives Reliant on Existing Facilities ²⁰³					
Alt. 1	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)	×	N/A	N/A		
Alt. 2	New Regasification Only - 400 MMcf/d (No Resiliency Reserve)		×	N/A		
Alt. 3	New Regasification Only - 600 MMcf/d (No Resiliency Reserve)	1	×	N/A		
New Facility with Gas Supply But No Resiliency Reserve ²⁰⁴						
Alt. 4	Like-for-Like (No Resiliency Reserve)	✓	✓	1.4		
Alt. 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification	✓	✓	1.8		

²⁰³ These alternatives include prolonged reliance on the Base Plant tank with no dependable resiliency reserve, declining reliability, and a high likelihood of relying on the market for some replacement gas supply.

²⁰⁴ These alternatives do not include a dependable resiliency reserve but provide different amounts of peaking gas supply and improved reliability.



SUMMARY OF RESULTS					
Supp. Alt. #	Name	Step 1 Results	Step 2 Results	Step 3 Overall Score	
	New Facility with Resiliency Re	serve But No G	as Supply ²⁰⁵		
Alt. 5	Like-for-Like (Full Resiliency Reserve)	✓	×	N/A	
Alt. 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification			N/A	
Alt. 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	×	N/A		
	New Facility with Both Resilien	cy Reserve and	Gas Supply		
Alt. 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	✓	✓	1.9	
Alt. 9	9 New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification (the Preferred Alternative)		✓	2.9 (Preferred)	
Non-Tilbury Alternatives ²⁰⁶					
Alt. 10	Alt 1 plus VITS Reverse Flow	×	N/A	N/A	
Alt. 11	LNG from Woodfibre	×	N/A	N/A	
Alt. 12	Floating LNG	×	N/A	N/A	

Table 4-16 below summarizes the Step 3 evaluation scoring results for the alternatives that passed the Step 1 and Step 2 screening (Supplemental Alternatives 4, 4A, 8 and 9). The scoring is relative to the existing Base Plant with no capital upgrades (Supplemental Alternative 1) given that there are no zero-cost alternatives

5 that there are no zero-cost alternatives.

6 Supplemental Alternative 9 (New 3.0 Bcf Tank with 2.0 Bcf Resiliency Reserve and 800 MMcf/d

7 Regasification) scored the highest and was therefore confirmed as the Preferred Alternative.

²⁰⁵ These alternatives include a full resiliency reserve but still rely on the market for replacement of the gas supply functions.

²⁰⁶ FEI considers these alternatives to be non-viable approaches to providing winter resiliency and peaking gas supply.



4		

Table 4-16:	Step 3	Scoring	Results
-------------	--------	---------	---------

Evaluation Criterion	Criterion Weightin	Alternative 4	Alternative 4A	Alternative 8	Alternative 9
Resiliency Benefit	30%	No Impact	No Impact	Medium Positive Impact	High Positive Impact
Availability of Dependable Gas Supply During Peak Demand	20%	Medium Positive Impact	High Positive Impact	Medium Positive Impact	High Positive Impact
Resolves Age Related Base Plant Challenges	20%	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact
Levelized Total Rate Impact	20%	Low Negative Impact	Low Negative Impact	Medium Negative Impact	Medium Negative Impact
Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP) Between the In- Service Date and 2050	10%	No Impact	No Impact	No Impact	No Impact
	Total Weighted Score:	1.4	1.8	1.9	2.9

Figure 4-20 is an alternate presentation of the results of the structured alternatives analysisprocess.







Figure 4-20: Step 3 Scoring Results

2

FEI's strong view is that Supplemental Alternative 9 will provide the best value in meeting the Project objectives by providing the optimum peaking gas supply and providing material customer outage risk reduction. Comparing the 67-year expected GDP loss reduction and the levelized total rate impact of Supplemental Alternative 4A to Supplemental Alternatives 8 and 9 leads to the following conclusions:

For an additional 1.3 percent and 1.2 percent levelized total rate impact, Supplemental Alternatives 8 and 9 would both provide significant loss reduction against a T-South no-flow event when compared to Supplemental Alternative 4A, which would optimize peaking



- supply only. This is true for direct customer impacts (customer outage-days) and GDP
 impacts.
- The increase in levelized total rate impact between Supplemental Alternative 4A (which would optimize gas supply only) and Supplemental Alternatives 8 and 9 (which would have varying amounts of gas supply plus resiliency reserves) is small compared to the significant loss reduction that would be provided by Supplemental Alternatives 8 and 9.
- 7 Supplemental Alternatives 8 and 9 would provide similar levels of risk mitigation against a 8 T-South winter no-flow event at average winter temperature. However, as discussed in 9 Section 4.5.1.5 and summarized in Figure 4-3 above (see "Expected Annual COD - Cold Weather (-1.4°C)"), Supplemental Alternative 9 will provide superior risk mitigation at 10 temperatures below the average winter temperature (i.e., the expected annual customer 11 12 outage-day loss at -1.4°C is lower under Supplemental Alternative 9 than under 13 Supplemental Alternative 8) for a lower levelized total rate impact. As discussed in Section 14 4.5.1.5, based on historical temperature data, nearly one-guarter of Lower Mainland winter days fall in the range (-6.8°C to +1.7°C) in which Supplemental Alternative 9 can bridge 15 16 the regulatory shutdown period but Supplemental Alternative 8 could not. This shows the 17 benefit of the Supplemental Alternative 9's incremental resiliency reserve (i.e., an 18 additional 0.6 Bcf relative to Supplemental Alternative 8) and incremental gas supply 19 capabilities (i.e., an additional 0.4 Bcf), which come at a relatively small incremental cost.



4.7 ADDITIONAL INFORMATION ON THE PREFERRED ALTERNATIVE'S 2 RESILIENCY BENEFITS

The Resiliency Benefit criterion in the structured alternatives analysis focuses on Exponent's calculated annual risk reduction in respect of a winter T-South no-flow event only. While that evaluation provides a common basis for alternatives evaluation in respect of FEI's single largest customer outage risk exposure, it understates the resiliency benefits from Supplemental Alternative 9 (Preferred Alternative). This section provides additional information on those considerable resiliency benefits, in particular:

- The Preferred Alternative provides very significant cumulative risk mitigation in respect of
 a winter T-South no-flow event over 67-years (the expected life of the Project) and 23 year time horizons (the adverse sensitivity used to address BCUC commentary);²⁰⁷
- The Preferred Alternative introduces the potential for FEI to undertake a staged shutdown to maintain service to some portion of FEI's customers for longer. This makes it much more likely that many customers will remain uninterrupted, while having the knock-on benefit of shortening the outage duration for curtailed customers.
- In a worst-case scenario where a winter T-South no-flow event is longer than the Preferred
 Alternative's load support duration, the Preferred Alternative will still (1) avoid the
 uncontrolled depressurization and the attendant safety risks expected today; and (2)
 provide valuable time for customers, governments and social / health services to prepare.
- The Preferred Alternative will mitigate the calculated customer outage risk for 13 other customer outage vulnerabilities across FEI's system.
- In a circumstance where access to off-system storage is temporarily impaired for
 operational reasons during cold periods, the Preferred Alternative can be used to backstop
 off-system storage supply and limit the service impact on FEI's customers.

4.7.1 Cumulative Risk Reduction in Respect of a Winter T-South No-Flow Event

Exponent's risk calculations show that the Preferred Alternative provides very significant cumulative risk mitigation in respect of a winter T-South no-flow event over 67-years (the expected life of the Project) and 23-year time horizons (the shorter adverse sensitivity used to address BCUC commentary).

The following figures prepared by Exponent show the overall winter risk mitigation provided by the Preferred Alternative in respect of T-South <u>only</u>. Exponent's calculations are based on average Lower Mainland winter temperatures (+4°C) and the assumption that the existing Base Plant with no capital upgrades (mirroring the Supplemental Alternative 1 scenarios) and the Preferred Alternative will operate to the extent of their capabilities until the LNG is exhausted. The figures express risk mitigation in terms of reductions in customer-outage-days, customer outages

²⁰⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, pp. 119 - 122.



1 and expected GDP losses (\$), respectively. The same pattern is evident regardless of the 2 consequence measure used.

3 4.7.1.1 Annual Winter Risk Mitigation (T-South Only)

4 The following figures show the winter mitigation <u>in relation to T-South only</u> for a time horizon of 5 one year.²⁰⁸

6 Figure 4-21: Preferred Alternative Annual T-South-Only Expected Customer Outage Days 7 Reduction²⁰⁹



²⁰⁸ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 61, 67 and 55, respectively.

²⁰⁹ Figures 4-21 to 4-29 are at average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



1 Figure 4-22: Preferred Alternative Annual T-South-Only Expected Customer Outages Reduction



Total expected annual customer outages reduction at average winter temperature (AV-1, -2, -3, and -54)

Figure 4-23: Preferred Alternative Annual T-South-Only Expected GDP Loss Reduction



Total expected annual GDP loss reduction at average winter temperature (AV-1, -2, -3, and -54)



1 4.7.1.2 Cumulative Winter Risk Mitigation 67 Years (T-South Only)

2 The following figures show the cumulative winter mitigation in relation to T-South only over 67

3 years.²¹⁰ FEI considers that this is the appropriate horizon over which to assess risk, as it is the

4 expected life of the TLSE Project and FEI expects to continue using it throughout that period for

5 the reasons stated in Section 6 of this Supplemental Evidence.

6 7

Figure 4-24: Preferred Alternative 67-Year T-South-Only Expected Customer Outage Days Reduction



²¹⁰ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 65, 71 and 59, respectively.



1 Figure 4-25: Preferred Alternative 67-Year T-South-Only Expected Customer Outages Reduction



Total expected 67-year lifetime customer outages reduction at average winter temperature (AV-1, -2, -3, and -54)

Figure 4-26: Preferred Alternative 67-Year T-South-Only Expected GDP Loss Reduction

Total expected 67-year lifetime GDP loss reduction at average winter temperature (AV-1, -2, -3, and -54)



2


1 4.7.1.3 Cumulative Winter Risk Mitigation 23 Years (T-South Only)

- The following figures show the cumulative winter mitigation <u>in relation to T-South only</u> over 23 years.²¹¹ FEI asked Exponent to prepare a sensitivity based on 23 years, as the BCUC had raised the issue of the usefulness of the Preferred Alternative beyond 2050. FEI regards this period as far too short as it expects to continue using the facility throughout its expected service life for the
- 6 reasons stated in Sections 6.2.2 and 6.4 of this Supplemental Evidence.
- 7 8

Figure 4-27: Preferred Alternative 23-Year T-South-Only Expected Customer Outage Days Reduction



²¹¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Report Appendix, Figures 63, 69 and 57, respectively.



1 Figure 4-28: Preferred Alternative 23-Year T-South-Only Expected Customer Outages Reduction



Total expected 23-year lifetime customer outages reduction at

2 3

Figure 4-29: Preferred Alternative 23-Year T-South-Only Expected GDP Loss Reduction



Total expected 23-year lifetime GDP loss reduction at average winter temperature (AV-1, -2, -3, and -54)



14.7.2Preferred Alternative Offers Potential for Staged Shutdown to Maintain2Service to Some Customers

Currently, the system will depressurize before FEI has time to assess and respond to a winter T-South no-flow event. The Preferred Alternative will give FEI time to assess conditions and the flexibility to adjust its curtailment response based on its real-time expectations about when the no-flow event will be resolved. This time and flexibility makes it far more likely that at least some portion of FEI's customers would continue to receive uninterrupted service during a no-flow event. Having fewer depressurized sections of the system to restore has the knock-on benefit of shortening the overall outage duration for curtailed customers.

- 10 In this section, FEI presents transient modelling load support duration results for multiple no-flow
- 11 / FEI response scenarios to illustrate the extent of the flexible risk mitigation provided by the TLSE
- 12 Project.

13 4.7.2.1 Response Options Available to FEI With Preferred Alternative

- 14 The following table outlines, in broad terms, the response options that will be available once the
- 15 Preferred Alternative is in place.
- 16

Table 4-17: FEI's Response Options to No-Flow Event with Preferred Alternative

Potential Response Options to a No-flow Event	Circumstances When Option Used
Continue to serve all firm load for as long as possible (the basis for the Resiliency Benefit criterion in Section 4.2.2.3.2 above). In this option FEI's deployment of the asset would entail sending out at a flow rate that matches or exceeds system demand and thus maintains or increases the pressure in the system.	When FEI is confident that the no-flow event will be rectified before the available LNG is depleted.
Once the issue is understood, begin to curtail firm loads progressively according to FEI's System Preservation and Restoration (P&R) Plan to reduce demand and thus extend the duration of support for the remaining customers. FEI's deployment of the asset in this option would be the same as described above. However, in this option system demand has been decreased through curtailment and therefore the send out flow rate required to maintain or increase the system pressure is also decreased, and thus the duration of support for the non-curtailed customers is extended.	When FEI is not confident that the no-flow event will be rectified before the available LNG is depleted.

17

4.7.2.2 Option to Maintain Service to All Firm Customers When FEI Confident Supply Will Resume Before Tank is Empty

The Preferred Alternative will enable FEI to continue to support all firm load without forced curtailments if FEI is confident that a no-flow event would be resolved before the tank was emptied. This is the scenario that was the focus of FEI's evidence in the TLSE Project CPCN proceeding to date, and the analysis remains valid.



- 1 FEI used its transient modelling to determine the length of time the Preferred Alternative could
- 2 support the entire Lower Mainland load under a variety of winter temperature conditions in the
- 3 event of a loss of supply from T-South. Table 4-18 below summarizes this analysis.
- 4 Table 4-18: Preferred Alternative Support of Lower Mainland Under Various Winter Temperatures

Temperature Condition (°C)	Existing Tilbury Base Plant (assuming full to 0.35 Bcf) ²¹²	Preferred Alternative – 2 Bcf resiliency reserve & 800 MMcf/d (Minimum 2 Bcf available in 3 Bcf tank on first day of no-flow event)
-10.0 (very cold winter day) ²¹³	2 hours	2 days and 17 hours
-1.4 (warmest winter in last 10 years) ²¹⁴	5 hours	3 days and 12 hours
+4.0 (average Lower Mainland winter) ²¹⁵	7 hours	4 days and 13 hours

- 6 Figures 4-30 and 4-32 below are the outputs of FEI's transient system modelling for column three
- 7 in Table 4-18 above (i.e., the modelling results at various temperatures for a 2.0 Bcf resiliency
- 8 reserve and 800 MMcf/d of regasification). The figures show that in each temperature scenario
- 9 the current regasification constraint has been addressed, such that the overall load support
- 10 duration is defined by the amount of LNG present. Please refer to Section 3.2.2.1.2 of this
- 11 Supplemental Evidence for an explanation of the various components of these transient modelling
- 12 graphs.

²¹² Although there is no resiliency reserve at present that would provide dependable resiliency, FEI has assumed that at least 0.35 Bcf of LNG is available on the day of the no-flow event (i.e., FEI assumed that the Base Plant is full based on current operating levels, meaning that FEI has used no supply to serve peak load). In the alternatives analysis, this is referred to as Alternative 1 (Contingent)).

²¹³ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

²¹⁴ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

²¹⁵ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



Figure 4-30: Impact to Lower Mainland Due to T-South No-Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification at -10°C



3 ⊿

Figure 4-31: Impact to Lower Mainland Due to T-South No-Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification at -1.4°C





3

Figure 4-32: Impact to Lower Mainland Due to T-South No Flow Event with 2 Bcf Resiliency Reserve and 800 MMcf/d Regasification) at Average Winter Conditions (+4°C)



4 4.7.2.3 Option to Initiate Staged Shutdown When Supply Resumption Date 5 Unknown

6 The Preferred Alternative also provides FEI with the flexibility to perform a staged shutdown to 7 extend the duration of load support for remaining customers, which is unlikely to be an option 8 currently in winter because of how quickly the system will currently depressurize. This flexibility 9 will be a significant benefit when responding to a no-flow event. A staged shutdown can reduce 10 the number of customers who experience an outage. It can also shorten the time to restore service 11 to all customers because there are fewer customers to relight.

As described in the Application,²¹⁶ in circumstances where FEI is not confident that the no-flow 12 13 event will be resolved before stored LNG is exhausted and FEI has time to respond, it will begin a staged shutdown of the system in a manner consistent with the BCUC-approved Gas Supply 14 Shortfall System Preservation and Service Restoration Plan (P&R Plan).²¹⁷ By pre-emptively 15 shutting down customers while the Preferred Alternative's tank still has volume, FEI can reduce 16 17 the system demand. This reduces the rate at which the tank storage is depleted and increases 18 the tank supply duration for the remaining online customers (i.e., customers that were not 19 shutdown pre-emptively).

²¹⁶ Exhibit B-1-4, Application, pp. 23-24; Exhibit B-46-1, Rebuttal Evidence to RCIA.

²¹⁷ The P&R Plan is confidential and security sensitive. However, in broad terms, contemplates initially shedding larger customers and then successively isolating portions of the Lower Mainland to maintain system pressure. Decisions on whether to curtail are made in real time, with reference to the most up-to-date information on when the no flow event is expected to be resolved.



At present, even under average winter temperatures (+4°C), FEI expects that a T-South failure on certain segments of T-South will cause the system to depressurize too quickly for FEI to employ a staged shutdown. The Preferred Alternative, in contrast, will provide sufficient regasification capacity and a resiliency reserve (i.e., guaranteed supply) to provide FEI with enough time to make the decision as to whether to initiate a staged shutdown.

FEI used its transient system modelling to determine the length of time the Preferred Alternative
could support portions of the Lower Mainland under various illustrative staged shutdown
scenarios. The results of these illustrative scenarios are presented in Table 4-19 below.

9Table 4-19: Preferred Alternative (2 Bcf Resiliency Reserve & 800 MMcf/d) Lower Mainland10Support Durations Under Various Staged Shutdown Scenarios

	Lower Mainland Support Duration					
Temperature Condition (°C)	Without Staged Shutdown	Example 1: 20% Customer Shutdown After 24 hrs	Example 2: 20% Customer Shutdown After 24 hrs Additional 20% Customer Shutdown After 48 hrs	Example 3: 20% Customer Shutdown After 24 hrs Additional 20% Customer Shutdown After 48 hrs Additional 20% Customer Shutdown	Example 4: 40% Customer Shutdown After 24 hrs Additional 20% Customer Shutdown After 48 hrs	
-10.0C (very cold winter	2 days and 17	3 days and 5	3 days and 8	3 days and 11	4 days and 3	
10 years) ²¹⁹	a days and 12 hours	4 days and 5 hours	4 days and 9 hours	5 days and 3 hours	5 days and 17 hours	
+4.0°C (average Lower Mainland winter) ²²⁰	4 days and 13 hours	5 days and 6 hours	5 days and 17 hours	6 days and 20 hours	7 days and 11 hours	

11

12 Given these results, there are a variety of scenarios where FEI could significantly reduce the scale

13 and duration of a customer outage by performing a partial staged shutdown; for instance, where

14 a no-flow event occurring at -1.4°C ends up lasting 4 days. Without a staged shutdown, 100

²¹⁸ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3C, and all subsequent days are 4°C.

²¹⁹ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

²²⁰ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



percent of Lower Mainland customers would lose service after the support duration (an estimated 3 days and 12 hours). However, if FEI executed a staged shutdown, then approximately 80 percent of Lower Mainland customers would retain service throughout the entirety of the event. While 20 percent of Lower Mainland customers would still lose service, the staged shutdown approach significantly reduces the overall consequence of the no-flow event. Restoring service to 20 percent of the Lower Mainland would take far less time than a Lower Mainland-wide customer outage.

4.7.3 Preferred Alternative Has Resiliency Value Even Where T-South No-Flow Event Is Too Long to Bridge

In a worst-case scenario where a T-South no-flow event is longer than the Preferred Alternative's load support duration, the Preferred Alternative will still: (1) avoid the uncontrolled depressurization and the attendant safety risks expected today; and (2) provide valuable time for customers, governments and social / health services to prepare.

14 4.7.3.1 Preferred Alternative, at Minimum, Prevents Uncontrolled Shutdown

FEI described in its prior evidence²²¹ the difference between a controlled and uncontrolled shutdown, emphasizing the safety and service restoration issues associated with an uncontrolled shutdown. A controlled shutdown requires time to implement,²²² and FEI lacks the necessary time at present to respond to a T-South winter no-flow event. At minimum, even if a no-flow event exceeds the duration of load support, the Preferred Alternative will provide adequate response time in most temperature conditions to ensure a controlled shutdown.

To summarize FEI's prior evidence, an uncontrolled shutdown occurs when the gas distribution system is naturally lost due to a collapse of system pressure and gas supply. An uncontrolled shutdown is a serious scenario both in terms of service disruptions to customers as well as the potential for safety concerns.

- From a safety perspective, the uncontrolled drop in system gas pressure can introduce
 the possibility of air being drawn into the distribution system. This is a potentially
 hazardous situation as the gas-air mixture can result in fire or explosion risks. Entrained
 air can also blow out the flames in customer appliances or equipment, resulting in improper
 operation.
- From an outage duration perspective, any air within the gas distribution system must be
 purged and the segment re-pressurized with gas prior to relighting any customers. This
 purge and regasification/re-pressurization process extends the duration of the customer
 outage when compared to a controlled shutdown.

In a controlled shutdown, which will be possible with the Preferred Alternative, gas pressure in the system is maintained. This reduces the possibility of air being drawn into the system. As such,

²²¹ Exhibit B-46-1, Rebuttal Evidence to RCIA, p. 9.

²²² Exhibit B-1-4, Application, p. 24.



- 1 notwithstanding a third-party line hit, in a controlled shutdown the safety hazard due to entrained
- 2 air is less likely to occur. Additionally, the extended outage duration associated with the need for
- 3 purging and repressurizing due to entrained air is avoided.

4 4.7.3.2 Preferred Alternative, at Minimum, Provides Valuable Preparation Time for 5 Customers, Governments and Health / Social Services

In a worst case scenario where the Preferred Alternative only delays a widespread customer
outage, this still provides valuable preparation time for customers, governments and health /
social services that can reduce harm.

- 9 Exponent addressed this in Section 11 of its Report:²²³
- 10 There can be significant downtime following a pipeline rupture or failure at a compressor station, control station, or valve assembly, as demonstrated by a 11 12 review of outage durations in Appendix T. In certain situations, a backup source of 13 gas may not provide sufficient load support to an AV to avoid an outage: for 14 example, if the time required to repair a ruptured pipe exceeds the load support 15 duration provided by the backup source. However, even when an outage 16 eventually occurs, having a backup supply of gas for three, five, or more days may 17 allow several emergency preparedness strategies to be implemented for 18 protecting individuals, communicating critical information to the public, and 19 sustaining medical services in the event of a winter failure. This chapter 20 summarizes some of the key preparedness measures that a backup supply of gas 21 might enable.
- 22 The following preparedness measures assume a certain degree of planning prior 23 to the disruption of gas service, including: the identification of regions with high 24 population density; the identification of areas with high concentrations of 25 vulnerable populations such as children, individuals with disabilities, and residents 26 of long-term care facilities; and the creation of evacuation plans for those regions. 27 It is also assumed that a communications system is available for timely 28 dissemination of relevant information to the public. Examples of preparedness measures in the days ahead of the gas outage include: 29
 - Relocating vulnerable members of the population, such as residents of long-term care facilities and nursing homes, individuals with disabilities, and persons requiring home-health services.
 - Setting up warming shelters for evacuated individuals and those who cannot shelter at home. The warming shelters may be equipped with heaters that run on electricity, gasoline or propane, and stockpiled

30

31

32

33 34

²²³ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, paras. 247-250.

2



with blankets, cots, water, and ready-to-eat meals, or self-heating meals.

- 3 c. Supporting sheltering-in-place for capable individuals, if possible, by 4 distributing heaters that run on other fuel sources than natural gas, 5 and providing timely communication of the imminent disruption event. 6 and safe practices for sheltering-in-place. For example, individuals 7 should be advised on protecting themselves from cold-related 8 injuries, preparing heating alternatives, and avoiding carbon 9 monoxide poisoning. Timely dissemination of information should 10 include plans to alert individuals with sensory or cognitive disabilities.
- 11d. Supplying all essential facilities such as government buildings,12emergency centers, hospitals and long-term care facilities with13electricity-, gasoline- or propane-powered heaters, backup14generators, and a sufficient supply of fuel.
- e. Bolstering hospital staffing in preparation of the increased demands,
 and stockpiling medical equipment needed to treat cold weatherrelated conditions in hospitals and emergency centers.
- 18f.Setting up the necessary backup systems for hospitals to run the
chillers, which keep the hospital servers and equipment running, and
keep medications accessible by maintaining functioning computer
systems. Temperature management of hospital facilities may also be
critical for proper storage of medications. It may become necessary
to temporarily relocate critical patients to other hospitals.
- 24g. Developing a strategy for the management and possible relocation25of incarcerated persons.
- 26 h. Managing traffic to support evacuation efforts, and possibly 27 implementing contraflow strategies, wherein all highway lanes are 28 temporarily converted to outbound lanes for maximizing traffic flow. 29 There may be a need to set up shelters along the designated evacuation routes and running special-service public transit to 30 31 transport individuals to sheltering locations. Designating refuges of 32 last resort may also be necessary for individuals which were unable 33 to reach the warming shelters.
- i. Pre-positioning medical personnel and supplies at shelter locations
 and establishing out-of-hospital care centers to respond to cold
 weather-related conditions such as frostbites and hypothermia.
- 37j.Identifying and pre-staging big kitchen locations for creating meals38and feeding the sheltered population.



- While additional days of backup gas supply may not always prevent an outage
 following a failure along a pipeline or at a station, they may provide enough time
 to mitigate the impacts of an outage on affected customers, including in particular
 the most vulnerable community members.
- 5 These benefits are not reflected in Exponent's risk calculations, but they are significant and very 6 real. NERC noted recently that hundreds of people died as a result of an electric outage in Texas
- 7 that left people without heat for only four days, in part due to "hypothermia, carbon monoxide
- 8 poisoning, and medical conditions exacerbated by freezing conditions."²²⁴ Minimizing the risk of
- 9 these outcomes would require advance preparation and outreach by various agencies.

10 4.7.4 Preferred Alternative Mitigates Risk of 13 Other Assessed Vulnerabilities

The Preferred Alternative will mitigate the calculated winter-only customer outage risk for 13 other customer outage vulnerabilities (referred to in the 2024 Resiliency Plan as Assessed Vulnerabilities or AVs) across FEI's system. The subsections below provide Exponent's calculations for the 13 other AVs <u>only</u>, and on a combined basis with the T-South risk to give an overall total risk reduction.

16 4.7.4.1 Risk Reduction for 13 Non-T-South Assessed Vulnerabilities Only

Figure 4-33 below is from Exponent's report and shows the <u>annual winter-only</u> risk mitigation provided by the Preferred Alternative relative to the status quo for <u>non-T-South</u> AVs. Please refer to the Exponent Report for Exponent's equivalent figures for horizons of 23 and 67-years.²²⁵ A similar pattern exists, although the mitigation values are a lot higher, when the risk mitigation is

21 considered on a cumulative basis over 23 and 67 years.

²²⁴ <u>https://www.nerc.com/pa/rrm/ea/Documents/February 2021 Cold Weather Report.pdf.</u>

²²⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 62, 68 and 56, respectively.



Figure 4-33: Annual Winter Only Expected Customer Outage Days Reduction – All AVs Except T South²²⁶



²²⁶ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



Figure 4-34: Annual Winter Only Expected Customer Outages Reduction – All AVs Except T-South²²⁷



3

1

²²⁷ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



1 Figure 4-35: Annual Winter-Only Expected GDP Loss Reduction – All AVs Except T-South²²⁸



Total expected annual GDP loss reduction at average winter temperature (all AVs except AV-1, -2, -3, and -54)

2

The risk mitigation evident in Exponent's figures above for these non-T-South AVs is due to the load support duration provided by the Preferred Alternative. The following table summarizes the estimated length of time the Preferred Alternative would support these non-T-South AVs under average winter conditions for the areas impacted by the AVs.²²⁹

Table 4-20: Preferred Alternative Support Durations Under Average Winter Conditions for Non-T South AVs Only²³⁰

Assessed Vulnerability	Mitigation Type ¹	Load Support Duration from Preferred Alternative
AV-4	Direct & Displacement	13.3 days
AV-12	Direct	19.1 days
AV-13	Direct	58.8 days
AV-14	Direct	95.2 days
AV-16	Direct	52.6 days
AV-17	Direct	19.1 days

²²⁸ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.

²²⁹ The average winter temperature in the Lower Mainland and the relevant parts of Vancouver Island is +4°C, but the average winter temperature in various parts of the Interior is much colder.

²³⁰ FEI did not conduct transient modelling to determine the support duration for all AVs. For these AVs, the tank support durations are estimated by dividing the tank volume by the AVs' daily flow rate. This method does not account for linepack; therefore, some durations may be under reported.



Assessed Vulnerability	Mitigation Type ¹	Load Support Duration from Preferred Alternative
AV-18	Direct	4.9 days
AV-20	Direct	4.5 days
AV-23	Direct & Displacement	13.3 days
AV-30	Direct	19.1 days
AV-40	Direct & Displacement	16.9 days
AV-41	Direct & Displacement	13.5 days
AV-42	Direct & Displacement	16.9 days

1 Note to Table:

Direct mitigation and mitigation displacement provide the equivalent benefit for customers. The only
 difference is the source of the physical gas molecules.

Table 4-20 above identifies two categories of mitigation, direct and by displacement, although the
effect from a customer perspective is the same in both instances.

- Direct mitigation involves instances where the re-gasified LNG from Tilbury physically
 flows to the region of the system that is experiencing the supply shortfall. An example of
 direct mitigation is the support provided by the Preferred Alternative to the Lower Mainland
 following a T-South winter no-flow event.
- 10 **Mitigation by displacement** refers to instances where the LNG supports the region at risk indirectly. As FEI discussed in the TLSE Workshop and prior written evidence,²³¹ the 11 12 storage provided by the TLSE Project would also allow FEI to meet Interior customer 13 demand for the vast majority of the year even if one of the gas transmission lines in the 14 Interior was disrupted. For example, if there was reduced capacity or a no-flow event on 15 the TC Energy pipeline that provides supply for the ITS at Yahk, the TLSE Project could also help FEI manage such an event. FEI could divert supply from the T-South system 16 17 into the ITS to replace the lost capacity from TC Energy, and then use the TLSE storage and regasification to back-fill the reduced supply into the LML which would have previously 18 19 been supplied from the T-South system.

20 4.7.4.2 Total Combined Risk Reduction for All Assessed Vulnerabilities

The following Exponent figures show their overall risk mitigation results for the Preferred Alternative at average winter temperatures, accounting for <u>T-South</u>, plus all other AVs.²³² The figures express risk mitigation in terms of, respectively, customer-outage-days, customer outages and expected GDP losses (\$). The figures show the winter mitigation for a time horizon of only one year. A similar pattern exists, although the mitigation values are much higher, when the risk mitigation is considered on a cumulative basis over 23 and 67 years.

²³¹ Transcript Volume 1, Web-Based Workshop March 11, 2021, pp. 187-188; Exhibit B-17, BCOAPO IR1 5.1.

²³² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 47, 48 and 46, respectively.



1 Figure 4-36: Total (All AVs) Expected Annual Winter-Only Customer Outage Days Reduction²³³



²³³ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



Figure 4-37: Total (All AVs) Expected Annual Winter-Only Customer Outages Reduction²³⁴



2

²³⁴ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



Figure 4-38: Total (All AVs) Expected Annual Winter-Only GDP Loss Reduction²³⁵



Total expected annual GDP loss reduction at average winter temperature (all AVs)

2

1

4.7.5 Preferred Alternative Protects Customers from Off-System Storage Disruptions

5 FEI contracts storage capacity at JPS and Mist to provide seasonal supply during the winter 6 period. Both are AVs in FEI's 2024 Resiliency Plan. Under normal operating conditions, FEI 7 receives storage supply at Huntingdon through gas displacement which works essentially as an 8 exchange of third-party T-South supply with FEI's off-system storage supply. However, the 9 displacement service is not available during major disruptions which limit the withdraw from off-10 system storage. Operational issues can occur for a variety of reasons along the supply chain 11 which result in supply shortfall and restricted capacity. For example:

- During a recent cold snap in the Pacific Northwest region during January 2024, both
 TransCanada (i.e., Foothills) and an outage at the JPS facility required additional natural
 gas to be sent out from Tilbury and Mt. Hayes during the cold event.
- The January 2023 T-South "swamp gas" incident discussed in Section 3.2.3.2.1 of this
 Supplemental Evidence resulted in FEI experiencing an unexpected reduction of gas

²³⁵ At average winter temperature and compared to the three baseline (status quo) scenarios provided to Exponent.



supply to its downstream system over approximately a 24-hour period. FEI relied on the
 Mt. Hayes and Tilbury facilities to make up for the supply shortfall.

If operational issues occur on the cold winter days, it would be very challenging for FEI to replace
a significant amount of storage supply in the market within a few hours. Under these
circumstances, TLSE Project LNG can be used to backstop storage supply to meet FEI's
customer demand and limit the impact on customers.

7 4.8 CONCLUSION

8 FEI's expanded alternatives analysis provides a sound basis for selecting Supplemental 9 Alternative 9 as the Preferred Alternative. An alternative must, at a minimum, be commercially 10 and technical viable and allow FEI to continue meeting its firm load to avoid firm customer 11 curtailments in normal operations. The analysis confirms that, among the viable options, 12 Supplemental Alternative 9 delivers on the Project objectives in a way that provides superior 13 customer value. In particular, Supplemental Alternative 9 will:

- Maintain and expand FEI's on-system firm peaking supply;
- Significantly mitigate FEI's largest customer outage risks associated with T-South, plus other AVs and off -system storage disruptions;
- Fully address end-of-life issues with the existing Tilbury Base Plant which has reached
 end-of-life and is experiencing age-related challenges;
- Compare favourably to smaller replacement facilities in terms of levelized total rate impact;
 and
- Be fully utilized throughout its expected service life.

As FEI has demonstrated in this section, it is in customers' best interest to replace the Base Plant and size the new facility based on near and mid-term load requirements, rather than under-sizing the TLSE Project for resiliency and gas supply in anticipation of uncertain adverse long-term future load scenarios. FEI will retain the flexibility to determine, in the future, how much of the tank should be set aside as a "resiliency reserve", and how much can be made available to optimize the gas supply portfolio. FEI's adverse load loss sensitivities (mDEP 2% and 5%) show that customers can expect to continue benefitting from the Preferred Alternative over time.



1 5. PROJECT DESCRIPTION

2 **5.1** INTRODUCTION

In this section, FEI provides an update on the Project description, including the approach for
 calculating the updated Project cost estimate and the updated Project schedule.

5 The scope of the TLSE Project has not changed, and much of the information contained in Section 6 5 of the Application remains current and applicable. There have been no changes made to the 7 key Project components, how they advance the Project objectives,²³⁶ FEI's assessment of 8 resources to complete the Project,²³⁷ the risk management approach taken²³⁸ or FEI's 9 assessment of the key regulatory permits and approvals.²³⁹ FEI has undertaken the necessary 10 work to reflect new geotechnical standards, and to update the cost estimate and schedule, as 11 further described below.

12 **5.2 PROJECT COMPONENTS REMAIN THE SAME**

The Project components remain the same. As discussed in Section 4 of this Supplemental Evidence, the preferred alternative (Supplemental Alternative 9) continues to be to replace the existing Base Plant with 3 Bcf of storage and 800 MMcf/d of regasification capacity. The tank will be allocated between a 2 Bcf resiliency reserve and 1 Bcf allocated to gas supply as a peaking resource.

- 18 Table 5-1 below provides an overview of the Project components and how each component
- 19 serves the Project objectives described in Section 3 of the Supplemental Evidence.

²³⁶ Discussed in Section 3 of the Supplemental Evidence and Exhibit B-1-4, Application, Section 5.2.

²³⁷ Discussed in Exhibit B-1-4, Application, Section 5.6.

²³⁸ Discussed in Exhibit B-1-4, Application, Section 5.7.

²³⁹ Discussed in Exhibit B-1-4, Application, Section 5.8.



Table 5-1: Overview of Project Component
--

Key Project Component	How Component Serves Project Objective
Regasification capacity of 800 MMcf/day. ²⁴⁰	 800 MMcf/day of regasification capacity serves a gas supply and resiliency function: The current regasification equipment is end-of-life. It is obsolete, experiencing increasing rates of failure and reliability issues, and is difficult to maintain or repair.
	• FEI's peaking supply requirements (200 MMcf/d) exceed the current regasification capacity at Tilbury (150 MMcf/d). The increased regasification allows FEI to inject sufficient natural gas from Tilbury into the Lower Mainland system each day to ensure that it is able to meet peak load consistently as it has for decades.
	 The current regasification capacity is far too small to be able to prevent a Day 1 system depressurization following a T- South winter no-flow event.
	• The proposed equipment will provide quicker response time than the present configuration. The response time will be two hours (between notification from FEI Gas Control to gas delivered to the system). This is beneficial for both gas supply and resiliency.
LNG storage Tank of 3 Bcf (142,400 m ³).	The Base Plant was constructed to lower engineering and safety standards that were in place at the time of construction. There are seismic, environmental and flooding issues inherent in the original Base Plant design. External experts have advised against refurbishment. The new LNG tank will be designed according to current design standards to provide safe and reliable operations.
	A 3 Bcf tank provides a 2 Bcf resiliency reserve and 1 Bcf allocated to gas supply as a peaking resource:
	• The resiliency reserve provides sufficient LNG supply to serve FEI's Lower Mainland average winter load for approximately 4.5 days if a T-South no-flow event were to occur. This, in combination with the increased regasification capacity, provides significant customer outage risk mitigation.
	 The 1 Bcf gas supply allocation will ensure that FEI has sufficient peaking resources to serve firm demand during peak periods, and allow FEI to optimize its gas supply portfolio.
Addition or modification of any necessary auxiliary systems including power supply, utility pipe racks, in-tank pumps, piping, cable trays, instrument air compressors, boil-off gas compressors, connectivity to Tilbury 1A LNG storage tank, and connections to the sendout gas pipeline.	These systems are required to provide the necessary power, control, monitoring, and interconnection systems to safely and reliably operate the facility.

²⁴⁰ 4x200 MMcf/d. Each unit is capable of an output range of 50 to 200 MMcf/d (i.e., 50 MMcf/d is the lowest capacity at which a vapourizer can operate).



Key Project Component	How Component Serves Project Objective
Demolition of above-ground portion of the	As explained in Section 3 of this Supplemental Evidence, the Base
Tilbury Base Plant LNG storage tank and	Plant has reached end-of-life and needs to be replaced as part of the
liquefaction facilities (Base Plant).	Project.

5.3 GEOTECHNICAL REQUIREMENTS HAVE CHANGED DUE TO SEISMIC DESIGN STANDARD CHANGES

The geotechnical requirements for the Preferred Alternative have changed due to changes in the
seismic design standards since the Application was filed.

5 The 2020 geotechnical costs, linked to design based on an earlier version of the CSA Z276 code,

6 are no longer valid due to significant changes in seismic hazard and design criteria according to

7 the latest code CSA Z276: 2022 (April 2023 version). The geotechnical costs from 2020 have

8 been re-evaluated using new construction rates from 2023. To improve the accuracy of the cost

9 estimate, WSP reviewed the design assumptions and consulted with a specialty ground

10 improvement contractor to ensure the design requirements could be met.

11 Detailed geotechnical work will be carried out prior to commencing detailed design to ensure the

12 proposed ground improvements will meet the limits of the ground settlement specified by the tank

13 vendor.

14 5.4 APPROACH TO UPDATED TLSE PROJECT COST ESTIMATE

As discussed in Section 6.1 of this Supplemental Evidence, FEI has updated the AACE Class 3
Project cost estimate to account for industry-wide inflationary pressures, a larger contingency,
and P70-based escalation (vs. P50 previously). The updated total cost estimate of the TLSE

18 Project is \$1,143.899 million in as-spent dollars, including AFUDC.

195.4.1 Project Cost Estimate Updated Based on AACE with New Contingency20and Escalation Reports

Consistent with the original estimate completed in 2020, FEI, in conjunction with Linde, Horton CB&I (HCBI), WSP (previously Golder), and Solaris Management Consultants Inc. (SMCI), updated the Project capital cost estimate using AACE International Recommended Practices 18R-97 and 97R-18 as guides. Please refer to Section 6 and Confidential Appendix G to this Supplemental Evidence for the updated summary of the total base capital cost estimate for the Project, and Confidential Appendices I and J for the Validation Estimating Contingency Report and Escalation Report, respectively.

Please refer to the Application, Section 5.4.1 for an overview of the external experts, their credentials, experience and their scope of work for the estimate.



1 5.4.2 Basis of Estimate Remains the Same Except for Timing

- 2 Please refer to Section 5.4.2 of the Application for more information on the basis of estimate.3 Aside from a shift in the construction dates, the basis of estimate remains the same.

4 5.4.3 Risk Analysis Reviewed

The risk assessment process for the Project has not changed. As noted in the Validation
 Estimating Contingency Report:²⁴¹

The risk analysis workshops to obtain team inputs were held in 2020. No new workshops were held for this update; however, the systemic and project-specific
risks were reviewed with project management in October 2023. Systemic risk inputs were only changed slightly and project-specific risk quantification values were updated as appropriate (e.g., for changes between 2020 and 2023).

12 FEI has set contingency and escalation amounts in addition to the Project base cost estimate to

13 achieve a P50 confidence level (with a P70 level for escalation risk) to address foreseeable risks

14 and changes in market conditions over time. Contingency and escalation amounts included in the

15 Project cost are discussed further in Section 6 of this Supplemental Evidence.

16 **5.5 UPDATED TLSE PROJECT SCHEDULE**

17 FEI has updated the detailed Project schedule to reflect the passage of time since the Adjournment Decision, during which FEI was preparing a comprehensive resiliency plan and 18 addressing the BCUC's commentary from the Adjournment Decision through this Supplemental 19 20 Evidence. The overall schedule has been delayed approximately four years from what was 21 presented in the Application (filed in 2020), such that the in-service date is now at the end of 2030, 22 rather than the end of 2026. The current Project schedule was created by FEI, with input from its 23 consultants, to ensure that all activity durations are based on reasonable assumptions and agreed 24 upon by experts with experience scheduling projects of this nature.

25 As in the Application, the Project construction is divided into the following five main sub-projects:

- Ground Improvement and Early Works;
- Regasification Package;
- Auxiliary Systems (Utility Pipe Rack and Equipment);
- 3 Bcf LNG Storage Tank; and
- Base Plant Demolition.

³¹

²⁴¹ Supplemental Evidence, Confidential Appendix I, p. 3.



- 1 Each sub-project will have its own activities, including pre-construction, construction, and post-
- construction. Sub-projects will be planned and coordinated to optimize limited space on site and
 to ensure a streamlined process.
- 4 The Project schedule assumes a BCUC approval of the CPCN by Q2 2025, with the execution 5 phase beginning in Q4 2026.
- 6 Table 5-2 below is a summary of the Project schedule and key milestones, with a comparison to
- 7 the schedule and milestones from the Application (see Table 5-9 of the Application). The basis of
- 8 estimate and detailed Project schedule is included as Appendix H.
- 9

Table 5-2: TLSE Project Schedule and Milestones

Activity	Dates from Application	Revised Dates
Environmental Assessment (EA)		
Submit EA Final Application	Jan-22	Jun-25
Phase 2 EA Certificate - Provincial/Federal	Jul-22	Mar-26
Contractor Selection and Award		
Award Engineering Procure Construct (EPC) and Engineering Contract(s)	Jul-22	Jun-26
Permitting		
BC FLNRORD (HCA Inspection Permit)	Jun-23	May-27
BCOGC Permits	Jun-23	Aug-27
Boundary Bay Airport - Email Notification	Jul-23	May-27
Ministry of Transportation and Infrastructure (Highway Use Permit)	Aug-23	May-27
BC 1 Call registration	Jan-24	Aug-27
Port of Vancouver (Notice of Shipping)	Jan-24	May-27
NavCanada (Land Use Program - Tower Care)	Jan-24	Oct-27
Transport Canada (Aeronautical Clearance Permit)	Jan-24	Oct-27
City of Delta Permits	May-24	Jun-27
WorkSafeBC - Worker Compensation Act/OHS Regulation	Jul-24	Aug-27
Technical Safety BC - Safety Standards Act Permits	Dec-24	Jun-27
Metro Vancouver Permits	Jan-25	Jan-28
Construction		
Start of Ground Improvement Work in Regasification and Auxiliary Piping Area	Jan-23	Oct-26
Ground Improvement in Tank Area Start of Construction	Mar-23	Feb-27
First Regasification Units Construction Complete (Phase 1)	Jul-24	Aug-28
Balance of Regasification Units Construction Completion (Phase 2)	Apr-25	Jun-29
Auxiliary System Construction Completion	Sep-26	Oct-30
LNG Storage Tank Expansion Completion	Sep-26	Oct-30
Project Technical Close-out (new)		Dec-30



- 2 There have been several schedule adjustments, mostly in the timing of permits based on FEI's
- 3 current understanding of permitting timelines; however, the overall schedule has been delayed
- 4 approximately four years from the original schedule in the Application (in service at the end of
- 5 2030 rather than the end of 2026).

6 **5.6 CONCLUSION**

The scope of the TLSE Project has not changed, and much of the information in the Application
remains current and applicable. FEI has undertaken the necessary work to reflect new
geotechnical standards and to update the cost estimate and Project schedule.



6. FINANCIAL ANALYSIS FOR PREFERRED ALTERNATIVE 1

2 This section provides the updated financial analysis for the Preferred Alternative, a new LNG 3 facility with 3 Bcf of LNG storage capacity and 800 MMcf/d of regasification. FEI has made changes to the analysis to address specific commentary in the Adjournment Decision, and to 4 5 better reflect the overall customer bill impact.

- 6 The section is organized as follows:
- 7 Section 6.1 – Updated TLSE Project Cost Estimate: FEI updated the AACE Class 3 • 8 cost estimate, including to account for industry-wide inflationary pressures, a larger contingency, and P70-based escalation. The updated total cost estimate for the TLSE 9 10 Project is \$1,143.889 million in as-spent dollars, including AFUDC.
- 11 Section 6.2 - Updated Financial Analysis: FEI has undertaken its primary financial 12 evaluation of the Preferred Alternative in a similar manner as in the Application, with the 13 exception that FEI has now also recognized offsetting reductions in the cost of gas due to 14 the Preferred Alternative's incremental gas supply benefits (avoided costs) relative to 15 Supplemental Alternative 1.
- 16 Section 6.3 - Updated Rate Impact for the Preferred Alternative: The Preferred 17 Alternative results in a levelized total rate impact of approximately \$21 per year over the 18 67-year analysis period relative to Supplemental Alternative 1. FEI's bill impact analysis 19 reflects the incremental delivery rate impact associated with the Project, partially offset by 20 gas supply benefits (reflected in a lower cost of gas rate) when compared to current 2024 Approved rates. 21
- 22 • Section 6.4 – FEI Performed a Sensitivity with a Shorter (27-year) Amortization 23 Period to Address BCUC Commentary: In recognition of the BCUC's commentary in the Adjournment Decision, FEI has also provided sensitivities based on a shorter 24 amortization period (i.e., full depreciation by 2050). 25

UPDATED TLSE PROJECT COST ESTIMATE 6.1 26

27 As discussed below, FEI updated the AACE Class 3 cost estimate. The updated cost estimate 28 incorporates industry-wide inflationary pressures, a larger contingency, and P70-based escalation 29 (compared to the P50-based escalation used in the Application). The updated total cost estimate 30

is \$1,143.889 million in as-spent dollars, including AFUDC.

6.1.1 Approach to Updated Project Costs Estimate, Contingency and 31 Escalation 32

33 Consistent with the original estimate completed in 2020, FEI, in conjunction with Linde, HCBI, WSP (previously Golder), and SMCI, updated the Project capital cost estimate using AACE 34 International Recommended Practices 18R-97 and 97R-18 as guides. 35



1 The design and quantities remain largely unchanged. With the exception of the ground

- 2 improvement work, the updated AACE Class 3 cost estimate is based on quantities that were
- 3 developed using the designs and material take-offs completed as part of the original estimate in
- 4 2020. The changes in the ground improvement work are due to a change to the seismic standards.

5 The updated base capital cost estimate is \$731.067 million in 2023 dollars (before contingency. 6 deferred costs and financing costs), which is an approximate 38 percent increase from the original 7 base capital cost estimate of \$529.103 million. The difference between the original and the 8 updated base capital cost estimate is primarily due to inflationary increases in material and 9 equipment costs and, to a lesser extent, increased labour costs consistent with the increases 10 experienced across the industry since 2020 when the original estimate was completed. Please 11 refer to Confidential Appendix G to this Supplemental Evidence for the updated summary of the 12 total base capital cost estimate for the Project.

- FEI also re-engaged Validation Estimating to update the quantitative analysis on Project specific risks and systemic risks used to determine the appropriate level of contingency, as well as to update the escalation risk analysis used to determine the appropriate level of escalation funding.
- Updated contingency: For the updated contingency, based on the updated quantitative risk analysis, a total capital budget at a P50 confidence level was recommended by Validation Estimating, resulting in a contingency estimate of \$135.800 million in 2023 dollars, which is approximately 19 percent of the updated base capital cost estimate. No specific management reserve was recommended by Validation Estimating. Please refer to Confidential Appendix I to this Supplemental Evidence for the updated Validation Estimating Contingency Report.
- 23 Updated escalation, now based on P70: The original escalation was performed based • 24 on a P50 confidence level. For the updated escalation, based on the updated escalation 25 risk analysis completed by Validation Estimating, FEI is now applying a P70 escalation 26 value of \$154.460 million. Validation Estimating recommended a higher (P70) confidence 27 level for escalation in consideration that the scale of the TLSE Project could put significant 28 demands on local markets that would generate localized escalation. Further, the current 29 base price forecast by S&P Global (formerly IHS Markit) is showing a trend of low prices 30 through 2026. likely reflecting a combination of reduced steel prices and reduced industry 31 capital spending in BC. However, from a pricing risk perspective, a low base forecast 32 means the risk is high should there be a resurgence of inflation and/or competing capital 33 spending, thus emphasizing the need to consider a higher confidence level (e.g., P70) 34 when budgeting for escalation. Please refer to Confidential Appendix J to this 35 Supplemental Evidence for the Validation Estimating Escalation Report.

36 6.1.2 Summary of Updated Project Cost Estimate

Table 6-1 below summarizes the updated total Project capital cost estimate in both 2023 and asspent dollars. The updated capital cost estimate for the Preferred Alternative meets the criteria
for an AACE Class 3 cost estimate.



	2023 \$	As-Spent \$
LNG Tank (3 BCF)	359.749	423.480
Regasification Equipment	141.483	166.547
Ground Improvement	60.944	71.740
Auxiliary System	153.964	181.239
Base Plant Demolition	14.927	17.571
Subtotal Capital Cost	731.067	860.578
Contingency	135.800	160.749
Subtotal Project Capital Costs w/ Contingency	866.867	1,021.327
CPCN Application	4.945	4.945
CPCN Preliminary Stage Development	1.546	1.546
Subtotal w/ Deferral Costs	873.358	1,027.818
AFUDC	-	120.096
Tax Offset	-	(4.025)
TOTAL Project Cost	873.358	1,143.889

Table 6-1: Breakdown of the TLSE Project Cost Estimate (\$ millions)

2

1

3 The updated TLSE Project cost estimate, reflected in the table above, is based on the following:

- An updated capital cost estimate of \$731.067 million in 2023 dollars developed by FEI, in conjunction with Linde, HCBI, WSP, and SMCI. The capital cost estimate includes:
- 6 o \$688.321 million of base capital costs comprised of a new 3 Bcf LNG storage tank,
 7 800 MMcf/d of regasification equipment, ground improvement work including
 8 installation of stone columns necessary for the new LNG storage tank, new
 9 auxiliary system, and base plant demolition. Please refer to Section 5.3 of the
 10 Application for a detailed description of each component of the Project; and
- \$42.746 million of pre-construction capitalized development costs, which include
 actuals of \$28.347 million from 2020 to 2023, and a forecast of \$14.399 million
 from 2024 to 2025.
- A contingency estimate of \$135.800 million in 2023 dollars (approximately 18 percent of the base cost estimate of \$731.067 million in 2023 dollars), as discussed in Section 6.1.1 above, provides a total capital budget at a P50 confidence level.
- A P70 escalation value of \$154.460 million as discussed in Section 6.1.1 above for the Project over the period from 2024 to 2030 applied to both the base capital cost and contingency.²⁴² The escalation is used to convert the Project capital cost, including contingency, from 2023 dollars to as-spent dollars (i.e., from \$866.867 million in 2023 dollars to \$1,021.327 million in as-spent dollars).

²⁴² No escalation has been applied on actual costs incurred by FEI prior to December 2023.



- Estimated \$6.491 million of total deferred costs, which includes \$4.945 million of
 Application Costs and \$1.546 million of Preliminary Stage Development Costs. Please
 refer to Section 6.1.2.2 of this Supplemental Evidence for further discussion.
- Financing costs based on FEI's 2024 approved AFUDC rate of 6.24 percent, equivalent to FEI's after-tax weighted average cost of capital (WACC).²⁴³

6 6.1.2.1 Updated Project In-Service Years by Asset Components

The TLSE Project is expected to complete in multiple phases from 2025 to 2030. Table 6-2 below
provides the estimated amount of the Project costs (i.e., \$1,143.889 million) to be complete
(deferral) or in-service (capital costs) each year between 2025 and 2030. Consistent with FEI's
treatment of major project capital costs, including CPCNs, once the assets are placed into service,
the associated capital cost will be added to FEI's rate base on January 1 of the following year.

- 12 For Project deferral costs, please refer to Section 6.1.2.2 below for the proposed treatment.
- 13

Table 6-2: Breakdown of Project Capital Costs by In-Service Year (2025-2030)

	Project complete and in-service each year, incl. Project Deferral (\$ millions) (To be transferred to Rate Base January 1 of each following year)						
As-Spent \$, incl. AFUDC and Tax Offset	2025	2026	2027	2028	2029	2030	TOTAL
LNG Tank (3.0 BCF)	-	-	-	-	-	572.204	572.204
Regasification Equipment	-	-	-	111.627	102.920	-	214.547
Ground Improvement	-	-	95.484	-	-	-	95.484
Auxiliary System	-	-	-	129.432	48.509	58.063	236.005
Total Charged to Gas Plant in Service	-		95.484	241.058	151.429	630.267	1,118.238
Base Plant Demolition	-	-	-	-	22.724	-	22.724
Project Deferral Costs	2.927	-	-	-	-	-	2.927
Total Project Costs	2.927	-	95.484	241.058	174.153	630.267	1,143.889
Annual Project % In-Service	0.3%	0.0%	8.3%	21.1%	15.2%	55.1%	100.0%

14

15 6.1.2.2 Updated Application and Preliminary Stage Development Costs

16 Consistent with the request in the Application (as discussed in Section 6.4.4 of the Application), 17 FEI is seeking approval under sections 59 to 61 of the UCA for deferral treatment of the 18 Application and Preliminary Stage Development Costs. The Application Costs incurred by FEI 19 include BCUC costs and BCUC-approved intervener costs for the regulatory process, as well as 20 FEI's external legal, consultant, and studies costs for the preparation of the Application, including 21 this Supplemental Evidence, and review as required to complete the regulatory process. The 22 Preliminary Stage Development Costs are related to expenses incurred for engaging third-party 23 consultants for feasibility evaluation, preliminary development, and assessment of potential 24 design and alternatives as required to complete this Application.

²⁴³ As approved for 2024 by Order G-41-24. The actual AFUDC will be calculated based on the approved AFUDC rate at the time of construction.



Table 6-3 below provides an updated forecast of the Application and Preliminary Stage
 Development Costs. The total forecast pre-tax deferral costs are \$6.491 million and include:

3 \$4.945 million of Application costs, with \$3.245 million of actual Application costs incurred 4 from June 2020 to December 2023 and a forecast of \$1.700 million from January 2024 to 5 the end of the remaining regulatory process. The actual Application costs up to December 2023 include the costs to prepare the Application and the subsequent regulatory process. 6 7 The regulatory process thus far has spanned from December 2020 to March 2023, when 8 it was adjourned by Order G-62-23. The actual Application costs also include the 9 additional legal, consultant, and studies costs incurred by FEI for the preparation of the Supplemental Evidence (i.e., from April 2023 to December 2023). The forecast Application 10 costs from January 2024 to the end of the remaining regulatory process include the legal, 11 12 consultant, and studies costs in 2024 and 2025 for the Supplemental Evidence, and 13 assuming a written hearing process with an expert-led workshop when the regulatory 14 process is restarted.

\$1.546 million of actual Preliminary Stage Development Costs from 2019 to 2020 for the
 preparation of the TLSE Project and Application. There are no additional preliminary stage
 development costs incurred by FEI since the Application was filed in December 2020.

Table 6-3: Forecast Application and Preliminary Stage Development Costs Deferral Account (\$ millions)^{244,245}

	Application (Actual: 2020-2023; Forecast	Preliminary Stage Development	
Particular	2024 Onwards)	(Actual: 2019-2020)	TOTAL
Pre-Tax Costs	4.945	1.546	6.491
WACC Return	0.555	(0.094)	0.461
Total Before Tax Offset	5.500	1.452	6.952
Tax Offset - Costs held in Deferral Account	(1.335)	(0.417)	(1.753)
Tax Offset - Capitalized Costs	-	(2.272)	(2.272)
Total	4.165	(1.238)	2.927
Annual Amortization for 3 years	1.388	(0.413)	0.976

The actual Application and Preliminary Stage Development Costs will be recorded in the proposed non-rate base deferral account, i.e., the Application and Preliminary Stage Development Costs deferral account, attracting FEI's WACC until it enters rate base. Consistent with FEI's previous CPCN applications, FEI proposes to transfer the balance in the deferral account to rate base on January 1 of the year following BCUC approval of the Application. Consistent with FEI's request

²⁴⁴ Income tax offset on the deferred costs (i.e., \$1.753 million) equals to the sum of \$4.945 million for the Application costs and \$1.546 million for the development costs times the income tax rate of 27 percent.

²⁴⁵ Income tax offset on the capitalized costs is related to the pre-construction development costs that were capitalized but are eligible for deduction for tax purposes. The amount (i.e., \$2.272 million) is equal to the capitalized costs of \$8.416 million times the income tax rate of 27 percent.



- 1 in Section 6.4.4 of the Application, FEI is proposing an amortization period of three years for the
- 2 deferral account.

3 6.2 UPDATED FINANCIAL ANALYSIS – 67 YEAR ANALYSIS PERIOD

FEI has updated the financial evaluation of the Preferred Alternative based on the updated capital cost estimate shown above in Section 6.1.2 over a 67-year analysis period. This is a similar approach used in the Application, with the exception that FEI has now also recognized offsetting reductions in the cost of service due to the Preferred Alternative's incremental gas supply benefits (avoided costs) relative to Supplemental Alternative 1. Please refer to Section 6.2.3 below for further explanation on the gas supply benefits in the financial evaluation.

- 10 FEI has treated Supplemental Alternative 1 as the Base Case Scenario (i.e., all financial results are assessed as incremental to the Base Case Scenario) for the purpose of the financial 11 12 evaluation. As explained in Section 4 and Appendix C of this Supplemental Evidence, FEI has 13 ruled out Supplemental Alternative 1 because adequate peaking supply is not currently available, 14 and off-system resources are not a replacement for on-system LNG²⁴⁶; nevertheless, off-system peaking supply was considered as an alternative to capital infrastructure investment and can 15 16 serve as a reasonable financial comparator for the TLSE Project alternatives. FEI assumed for 17 this financial analysis that there would be another upgrade to the regional infrastructure by 2035 18 (with associated costs and an increase in tolls) to continue to serve FEI peak load under the Base 19 Case Scenario.

20 6.2.1 Summary of Results – 67-Year Analysis Period

Table 6-4 below provides the present value (PV) of the incremental revenue requirement and the levelized total rate impact to FEI's non-bypass customers over a 67-year analysis period. Details of the financial evaluation of the Preferred Alternative are provided in the Financial Schedules included in Confidential Appendix K to this Supplemental Evidence.

- As shown in Table 6-4 below, the PV of the incremental revenue requirement due to the Preferred Alternative is approximately \$723.267 million over the 67-year analysis period, which includes approximately \$1,240.821 million of PV delivery margin impact related to the capital cost of the Project that is partially offset by approximately \$517.554 million related to the avoided cost of gas provided by the additional 1 Bcf of LNG storage reserved for gas supply (compared to the Base Case Scenario, as discussed in Section 4.5.4.1.2 of this Supplemental Evidence).
- The levelized total rate impact of the Preferred Alternative over the 67-year analysis period, including both the delivery rate impact and the cost of gas savings, is 2.45 percent, which is equivalent to \$0.228 per GJ. For clarity, the new assets associated with the Project will be included in FEI's rate base when construction completes, with the capital costs recovered through FEI's delivery rates. The avoided gas supply costs, in contrast, will be a benefit to FEI's cost of gas which is not part of FEI's delivery rates. As such, in order to evaluate the Preferred Alternative,

²⁴⁶ See discussion in Section 3.3.4 of this Supplemental Evidence.



- 1 taking into account both the impacts due to the capital cost and the gas supply benefits, FEI
- 2 presents a levelized total rate impact (which is the levelized total bill impact) over the 67-year
- period in Table 6-4 below. If presented separately, the levelized delivery rate impact due to the
 Project is 6.90 percent and the levelized cost of gas impact due to the gas supply benefits are
- 5 savings of 4.49 percent.²⁴⁷
- 6

Table 6-4: Financial Analysis of the Project – 67-year Analysis Period

Line	Particular	TOTAL	(Confidential Appendix K, Financial Schedule)			
1	Total Charged to Gas Plant in Service (\$ millions)	1,118.238	Schedule 6; Line 65			
2	Base Plant Demolition Costs (\$ millions)	22.724	Schedule 6; Sum of Line 62 (2025 to 2029)			
3	Total Project Deferral Cost, Net of Tax (\$ millions)	2.927	Schedule 9; Line 6 + Line 15			
4	Total Project Cost (\$ millions)	1,143.889	Sum of Line 1 to Line 3			
5						
6	Incremental Delivery Margin in 2035 (\$ millions)	118.671	Schedule 1; Line 12 (2035)			
7	Incremental Cost of Gas Benefits in 2035 (\$ millions)	(63.000)	Schedule 1; Line 2 (2035)			
8	Net Incremental Revenue Requirement in 2035 (\$ millions)	55.671	Line 6 + Line 7			
9						
10	PV of Incremental Delivery Margin 67 years (\$ million)	1,240.821	Schedule 10; Line 22			
11	PV of Incremental Cost of Gas Benefits 67 years (\$ million)	(517.554)	Schedule 10; Line 32 - Line 22			
12	Net PV of Incremental Revenue Requirement 67 years (\$ million)	723.267	Line 10 + Line 11			
13						
14	Delivery Rate Impact in 2035 (%)	10.40%	Schedule 10; Line 25 (2035)			
15	Total Rate Impact (incl. Cost of Gas) in 2035 (%)	2.97%	Schedule 10; Line 35 (2035)			
16						
17	Levelized Total Rate Impact (Incl. Cost of Gas) 67 years (%)	2.45%	Schedule 10; Line 39			
18	Levelized Total Rate Impact (Incl. Cost of Gas) 67 years (\$/GJ)	0.228	Schedule 10; Line 46			

8 6.2.2 Basis for 67-Year Analysis Period

9 The 67-year analysis period is based on a 60-year post-Project analysis period plus seven prior

years from 2024 to 2030 (with all new assets to be placed in-service by 2030 as shown in Table

11 6-2 above).

As discussed in Section 6.4.1 of the Application, the 60-year post-Project analysis period was chosen based on the average service life for a new 3 Bcf LNG tank, as recommended by Concentric Advisors, ULC (Concentric), who completed FEI's most recent Depreciation Study (2022) which was filed as part of FortisBC's 2025-2027 Rate Setting Framework Application.²⁴⁸ As discussed in Section 4.5.5 of the Supplemental Evidence, FEI expects the Preferred Alternative will continue to be used and useful through the energy transition and, as such, the 60-

²⁴⁷ The levelized delivery rate impact of 6.90 percent is calculated based on a PV of incremental revenue requirement over a 67-year period of \$1,240.821 million related to the capital cost and divided by the PV of FEI's 2024 approved delivery margin over a 67-year period of \$17.984 billion (see Confidential Appendix K to this Supplemental Evidence, Schedule 10, Line 29). The levelized cost of gas impact of 4.49 percent is calculated based on a PV of incremental cost of gas benefits over a 67-year period of \$517.554 million divided by the PV of FEI's 2024 approved cost of gas over a 67-year period of \$11.527 billion (see Confidential Appendix K to this Supplemental Evidence, Schedule 10, Line 38 – Line 28). The total levelized rate impact (delivery and cost of gas) of 2.45 percent is calculated based on the total PV of revenue requirement of \$723.267 million divided by the total PV of FEI's 2024 approved revenue requirement of \$29.511 billion (i.e., sum of \$17.984 billion and \$11.527 billion).

²⁴⁸ Filed for BCUC review on April 8, 2024.



- 1 year post-Project analysis period (based on the expected average service life of the LNG storage
- 2 tank) remains appropriate. As part of this Supplemental Evidence, FEI continues to seek approval
- 3 of a depreciation rate of 1.67 percent and a net salvage rate of 0.67 percent (equivalent to 60
- 4 years) for the new 3 Bcf LNG tank.
- 5 Please refer to Section 6.4 of the Supplemental Evidence for a financial evaluation based on a
- 6 shorter post-Project analysis period in response to the BCUC commentary from the Adjournment
- 7 Decision.

8 6.2.3 Importance of Recognizing Avoided Gas Supply Costs

In preparing the updated financial evaluation of the Preferred Alternative (3 Bcf of LNG storage
 capacity and 800 MMcf/d of regasification), FEI has also included the gas supply benefits that

11 were discussed in Section 4.5.4.1.2 of the Supplemental Evidence as avoided costs.

12 These avoided costs are related to the "third Bcf" allocated for planning purposes to gas supply 13 (i.e., not the 2 Bcf resiliency reserve) and the associated increased regasification capacity. As 14 highlighted in Section 4.5.4.1.2 of this Supplemental Evidence, FEI has estimated the avoided 15 costs to be approximately \$7 million (post mitigation) between 2030 and 2035. The avoided costs 16 are based on the assumptions that peaking gas supply ceases to be available from Tilbury (when 17 the existing Base Plant ceases operation due to its age and Tilbury 1A is fully subscribed through 18 RS 46) and that third party regional pipeline or storage upgrades will not be completed until at 19 least 2035. From 2035 onwards, FEI estimated the avoided costs to be between approximately 20 \$63 million and \$79 million annually, as shown in Figure 4-7 of this Supplemental Evidence, 21 reflecting the gas supply costs that FEI would incur by relying on regional infrastructure under the 22 Base Case Scenario (i.e., Supplemental Alternative 1). For the financial analysis, FEI has used 23 the lower bound of gas supply costs of \$63 million.

FEI discussed the gas supply benefits in the Application, as well as throughout the regulatory process prior to the Adjournment Decision; however, the original financial evaluation in the Application did not include the benefits of avoided costs, as the focus at that time was on the delivery rate impact only. FEI considers it reasonable and appropriate to include the gas supply benefits in the updated financial evaluation as part of this Supplemental Evidence to provide a more fulsome evaluation of the Preferred Alternative's total impact on customer bills, including both delivery rates and cost of gas.

31 6.2.4 Assumptions Used

32 With the exception of the gas supply benefits discussed above, FEI applied the same assumptions

- to the financial evaluation in this Supplemental Evidence as was used in the Application; however,
- 34 FEI updated the assumptions from 2020 dollars to 2023 dollars.
- 35 The assumptions applied to the financial analysis are summarized below:

10



- Inflation: 2 percent annually for incremental O&M, property tax, and future capital replacement costs during the post-Project analysis period starting from 2031 (i.e., since the new LNG tank is scheduled to be in-service in 2030, the first full year of the new LNG tank in-service will be 2031). This is consistent with the Bank of Canada inflation target of 2 percent which Canada is expected to return to in 2025.²⁴⁹
- Incremental O&M: An estimate of incremental O&M costs resulting from the TLSE Project of approximately \$5.729 million in 2023 dollars (\$6.953 million in 2031 dollars²⁵⁰). These costs are comprised of:
 - approximately \$8.818 million in 2023 dollars (\$10.585 million in 2031 dollars) of new O&M costs, including electricity costs, associated with the new 3 Bcf LNG tank, the new 800 MMcf/d regasification equipment, and auxiliary systems;
- o offset by O&M savings, including electricity costs, of approximately \$3.089 million
 in 2023 dollars (\$3.631 million in 2031 dollars) due to the demolition of the Tilbury
 Base Plant.
- 15 The O&M estimates for the new tank, regasification equipment, and auxiliary systems are 16 based on the escalated estimates from 2019 developed by Partners in Performance (PiP), 17 as presented in Confidential Appendix N of the Application. As part of this Supplemental 18 Evidence, FEI updated the maintenance, utilities, chemicals and reagents, and insurance costs, as well as the cross-charge expenses, to 2023 dollars using actual CPI inflation 19 20 from 2019 to 2023. FEI also updated the labour costs to 2023 dollars based on FEI's 21 actual increase in costs for the relevant roles. For electricity costs, FEI updated the 22 calculation based on BC Hydro's RS 1830 for transmission service customers, which was 23 recently approved by the BCUC and will be transitioning from RS 1823 over a three-year period from 2024 to 2026.²⁵¹ FEI assumed a 2 percent general electricity rate increase for 24 2026 and beyond. 25
- The offsetting savings reflect the average of historical O&M costs for the Tilbury Base Plant over the most recent 10-year period from 2014 to 2023. These costs will no longer be incurred once the Tilbury Base Plant is decommissioned.
- Property Tax: Incremental property tax as a result of the new 3 Bcf tank based on the 2023 tax rate. The incremental property tax is assumed to occur in phases based on percentage completion of the LNG tank construction between 2025 and 2030.
- Incremental Sustainment Capital for Mechanical Equipment: FEI continued to use an average of 1 percent per year of the revised mechanical equipment capital expenditures (LNG tank, regasification equipment, auxiliary equipment) in this Supplemental Evidence as an estimate of incremental sustainment capital due to the Project. As explained in

²⁴⁹ Bank of Canada Monetary Policy Report – April 2024 (<u>https://www.bankofcanada.ca/2024/04/mpr-2024-04-10/#:~:text=The%20Bank%20projects%20that%20inflation,April%201%20to%20March%2031</u>).

²⁵⁰ Based on 2 percent annual inflation for all O&M costs, except electricity. Since the new LNG tank is scheduled to be in-service in 2030, the first full year of the new LNG tank in-service will be 2031.

²⁵¹ Order G-353-23; three-year transition from RS 1823 to 1830 from BC Hydro's Fiscal 2025 to Fiscal 2027 (i.e., April 2024 to April 2026).



- Section 6.3 of the Application, this assumption was developed by PiP based on the
 industry benchmark of similar operations and interviews with third party industry
 experts.²⁵² This benchmark applies to the capital cost of the mechanical equipment only,
 which does not include other indirect costs such as mobilization, engineering, contingency,
 etc.
- 6 Future capital replacement: The average service life for the regasification equipment 7 and auxiliary system is 41 years, which is shorter than the 60-year post-Project period 8 used for the financial analysis. As such, FEI's financial analysis includes future 9 replacement of the regasification and auxiliary systems at the end of their average service 10 life at 41 years. The future replacement costs are based on the 2023 estimate plus an annual escalation of 2 percent. Consistent with the assumption made in the financial 11 12 evaluation completed in 2020 as part of the Application, the future capital replacement 13 does not include the replacement of ground improvement work related to stone columns. 14 FEI does not expect the stone columns will need to be replaced within the 60-year post-15 Project period.

For clarity, FEI is not seeking approval of the incremental sustainment capital, incremental O&M, or the future capital replacements as part of this proceeding. These costs are included as a proxy to ensure a fulsome analysis of the financial impact of the TLSE Project over the expected life of the new assets. If these requirements materialize in the future, FEI will seek approval from the BCUC, as required, for these incremental costs in future applications such as FEI's revenue requirement applications.

22 6.3 UPDATED RATE IMPACT FOR THE PREFERRED ALTERNATIVE

The incremental delivery rate impact from 2026 to 2031 includes the amortization from the Application and Preliminary Stage Development Costs deferral account between 2026 and 2028, as discussed in Section 6.1.2.2 above, and the incremental impact of the capital costs which will enter FEI's rate base in phases between 2028 and 2031, as shown in Table 6-2 above.

- 27 The updated incremental delivery rate impact due to the Project is summarized as follows:
- Based on the updated capital cost estimate in Section 6.1.2 above, the Preferred
 Alternative will result in a cumulative incremental delivery rate impact of 11.03 percent by
 2031 when compared to FEI's 2024 approved delivery rates.
- The year-over-year increase in delivery rate is shown in Table 6-5 below, with an average incremental delivery rate impact per year of approximately 1.78 percent over the six-year period from 2026 to 2031 when compared to FEI's 2024 approved delivery rates.

²⁵² Exhibit B-1-4, Application, Confidential Appendix N.



Table 6-5: Summary of Delivery Rate Impact for the TLSE Project

		2026	2027	2028	2029	2030	2031
	Annual Delivery Margin, Incremental to 2026 Approved, Non-Bypass (\$ millions)	1.328	1.456	13.834	33.280	54.792	125.920
	% Increase to 2026 Approved Delivery Margin, Non-bypass	0.12%	0.13%	1.21%	2.92%	4.80%	11. 03 %
2	Incremental % Delivery Rate Impact (Year-over-Year)	0.12%	0.01%	1.08%	1.68%	1.83%	5.95%

However, when accounting for the cost of gas savings starting from 2031 (as discussed in Section 6.2.3 above) in addition to the incremental delivery rate impact from 2026 to 2031, the levelized total rate impact (levelized total bill impact) over the 67-year analysis period due to the Project is 2.45 percent or \$0.228 per GJ. For the typical residential customer with an average annual consumption of 90 GJ, this is equivalent to a bill impact of approximately \$20.55 per year over the 67-year analysis period.

9 6.4 FEI PERFORMED A SENSITIVITY WITH A SHORTER (27-YEAR) AMORTIZATION 10 PERIOD TO ADDRESS BCUC COMMENTARY

11 Regarding the rate impact analysis time horizon, the BCUC stated in the Adjournment Decision:²⁵³

However, the cost justification for the larger [3 Bcf vs 2 Bcf] tank is based on forecasts over 67 years at a time when the future of natural gas and the pipeline system is uncertain. The Panel reviews these uncertainties in Section 5.2.1 of this Decision." Page 40: "Because of these concerns we are unable to find a 60-year life to be appropriate for the purpose of amortization. Given the uncertainties around the useful life, a shorter amortization period may be more appropriate.

18 FEI continues to believe, based on the discussion in Section 4.5.5 of this Supplemental Evidence, 19 that the TLSE Project will provide both resiliency and gas supply benefits for customers for the 20 duration of its 60-year post-commissioning expected service life. However, to be responsive to 21 the BCUC's commentary, FEI has also provided sensitivities based on a shorter amortization 22 period by increasing the depreciation rates of the assets. FEI has performed an ancillary financial 23 evaluation of the TLSE Project based on the PV of the incremental revenue requirement and the 24 levelized total rate impact over a 27-year analysis period (including the construction years) assuming all new assets related to the Project will be fully depreciated (or amortized) by 2050²⁵⁴. 25 The 27-year analysis period, with all new assets being fully depreciated by 2050, aligns with BC's 26 target of net-zero by 2050²⁵⁵ and the timeframe suggested by the BCUC in the Adjournment 27 Decision.256 28

Table 6-6 below provides the financial analysis with a shorter analysis period of 27 years compared to the proposed 67-year analysis period. As a result of higher depreciation rates for all

²⁵³ Adjournment Decision, p. 21.

²⁵⁴ The 27-year period includes the Project period from 2024 to 2030 and the post-Project period by assuming all assets will be fully depreciated by 2050.

²⁵⁵ <u>cleanbc_roadmap_2030.pdf (gov.bc.ca)</u>.

²⁵⁶ Adjournment Decision, pp. 39-41.


- 1 new assets related to the Project (i.e., in order to be fully depreciated by 2050), the levelized total 2 rate impact would be 3.56 percent, which is approximately 1.11 percent higher than the levelized
- 3 total rate impact of 2.45 percent if the new assets are depreciated over a 60-year period (i.e., a
- 4 67-year analysis period when including the seven prior years starting from 2024). For a typical
- 5 residential customer with an average annual consumption of 90 GJ, this is equivalent to a
- 6 levelized total bill impact of approximately \$29.88 per year if the assets are to be fully depreciated
- 7 by 2050, which is approximately \$9.34 per year higher than the \$20.55 per year shown in Section
- 8 6.3 above.

9 FEI notes that using a shorter amortization period might entail additional costs for managing
 10 external financial reporting.²⁵⁷ These additional costs are unknown and therefore have been
 11 excluded from the financial analysis presented in Table 6-6 below.

12Table 6-6: Financial Analysis of the Project over 27 Years Compared to 67 Years (20 Years vs. 6013Years Expected Service Life)

		TOTAL	TOTAL	
Line	Particular	(20-years)	(60-years)	Difference
1	Total Charged to Gas Plant in Service (\$ millions)	1,118.238	1,118.238	-
2	Base Plant Demolition Costs (\$ millions)	22.724	22.724	-
3	Total Project Deferral Cost, Net of Tax (\$ millions)	2.927	2.927	-
4	Total Project Cost (\$ millions)	1,143.889	1,143.889	-
5				
6	Incremental Delivery Margin in 2035 (\$ millions)	163.014	118.671	44.344
7	Incremental Cost of Gas Benefits in 2035 (\$ millions)	(63.000)	(63.000)	-
8	Net Incremental Revenue Requirement in 2035 (\$ millions)	100.014	55.671	44.344
9				
10	PV of Incremental Delivery Margin over analysis period (\$ million)	1,199.226	1,240.821	(41.595)
11	PV of Incremental Cost of Gas Benefits over analysis period (\$ million)	(337.830)	(517.554)	179.724
12	Net PV of Incremental Revenue Requirement over analysis period (\$ million)	861.396	723.267	138.128
13				
14	Delivery Rate Impact in 2035 (%)	14.28%	10.40%	3.89%
15	Total Rate Impact (incl. Cost of Gas) in 2035 (%)	5.34%	2.97%	2.37%
16				
17	Levelized Total Rate Impact over analysis period (%)	3.56%	2.45%	1.11%
18	Levelized Total Rate Impact over analysis period (\$/GJ)	0.332	0.228	0.104

15 Considering the likelihood that the assets will remain used and useful for the full 60-year expected

16 service life, as discussed in Section 4.5.5 of this Supplemental Evidence, FEI considers it

17 reasonable to depreciate the assets over their 60-year useful life.

18 **6.5 CONCLUSION**

14

19 FEI's updated financial analysis and incremental rate impact for the Preferred Alternative account

20 for the BCUC's commentary in the Adjournment Decision. FEI updated the capital cost estimate,

21 the contingency, and used a P70 escalation. FEI has accounted for the incremental gas supply

22 benefits associated with avoiding the cost of holding capacity on expanded regional infrastructure,

²⁵⁷ See Exhibit B-39, BCUC Panel IR1 6.9.1.



- 1 which would be necessary under the Base Case Scenario once the Base Plant ceases operation.
- 2 Although FEI has provided the BCUC-suggested sensitivity based on the assets being fully
- 3 depreciated by 2050, this period is too short in light of the ongoing resiliency and gas supply
- 4 benefits discussed in Section 4.5.5 of this Supplemental Evidence.

5



1 7. ENVIRONMENT AND ARCHAEOLOGY

2 Although there has been no change to the TLSE Project scope or footprint, over the past three 3 years, FEI has been advancing the Environmental Assessment (EA) for the Tilbury Phase 2 LNG 4 Expansion Project (Phase 2), which includes components of the TLSE Project. This has involved 5 additional environmental and archaeological activities, including a Stage 2 Preliminary Site Investigation (PSI), Limited Detailed Site Investigation (DSI) and removal of soil stockpiles, 6 7 discarded rail ties, etc., and completion of an Archaeological Impact Assessment (AIA) for the 8 site. These additional steps have provided FEI the opportunity to further understand the 9 environmental and archaeological risks of the Project, as described below.

10 Based on the additional assessments undertaken, and prior to FEI's planned mitigation activities,

- 11 FEI's assessment of the potential impacts of the project have changed.
- 12 This section is organized around the following points:
- Section 7.1 Environmental Assessment: The potential environmental impacts of the
 TLSE Project, which were described as "moderate" in the Application, have been
 downgraded to "low to moderate".
- Section 7.2 Archaeological Assessment: The potential archaeological impacts of the
 TLSE Project, which were described as "moderate" in the Application, have been
 downgraded to "low".

FEI expects to mitigate potential impacts through additional assessments, permitting, and
 standard protection and mitigation measures so there would be negligible environmental and
 archaeological effects from the Project.

22 7.1 ENVIRONMENTAL ASSESSMENT

Technical Data Reports (TDRs) for biophysical receptors identified in the Environmental Overview
Assessment (EOA) for the Tilbury site were conducted in preparing the Phase 2 EA application.
These reports included a Terrestrial Biophysical TDR (including wildlife and wildlife habitat
surveys, vegetation and invasive species surveys and wetlands characterization) and an Aquatic
Biophysical TDR (including fish and fish habitat surveys). The findings from these reports support
the previous filing's "low" risk rating for the following biophysical receptors:

- Surface water quality and quantity;
- Fish and fish habitat;
- Vegetation and wetlands;
- Wildlife and wildlife habitat; and
- Land use.



Biophysical receptors identified as having additional risk in Section 7.2.1.3 of the Application (i.e.,
 the atmospheric and contaminated soils and ground water receptors) were also further
 investigated for the Phase 2 EA application. The results of this further investigation are as follows:

- The risk rating of the atmospheric biophysical receptor has not changed, as risks associated with Metro Vancouver permitting are still considered "medium to high".
- 6 For the contaminated soil and/or groundwater biophysical receptor, a Stage 1 and Stage 7 2 PSI were completed to further understand the potential contamination at the site. These 8 investigations concluded that seven of the eight areas of potential environmental concern 9 (APECs) outlined in the EOA did not show contamination. One APEC, APEC 1 - former 10 sawmill site, was identified to have contaminated soil. A Limited DSI was conducted for 11 APEC – 1 to further delineate the extent of soil contamination, and remediation works are 12 scheduled for the near future. Therefore, the risk rating for the contaminated soils and 13 groundwater biophysical receptor has been downgraded from "medium to high" to 14 "negligible to low".

As described in the Application, FEI will follow best management practices and mitigation
measures applicable to the Tilbury site during construction and will obtain all required
environmental permits and approvals as required.

18 7.2 ARCHAEOLOGICAL ASSESSMENT

An AIA was completed for the site in 2021. This work was conducted under *Heritage Conservation Act* Permit #2020-137 and the following Cultural Heritage Investigation permits were obtained:

- Katzie Development Limited Partnership Archaeological/Heritage Permit 2020-03;
- Seyem' Qwantlen Land Development Ltd. Heritage Investigation Permit SQ 2022-63;
- Stó:lō Research and Resource Management Centre Heritage Investigation Permit 2019-225;
- x^wməθk^wəýəm (Musqueam Indian Band) Heritage Investigation Permit MIB-2019-163 AIA;
- Skwxwú7mesh Úxwumixw (Squamish Nation) Archaeological Investigation Permit 20 0131; and
- səlilwətał (Tsleil-Waututh Nation) Cultural Heritage Investigation Permit 2019-149-1.
- 30 The objectives of the AIA were as follows:
- Identify the location and extent of archaeological sites that may be affected by the Project;
- Assess the heritage significance of any identified archaeological sites;
- Evaluate the nature and magnitude of direct and indirect impacts of the Project on
 archaeological sites; and



- 1 2
- Formulate management options for avoiding or mitigating Project impacts to archaeological sites.

3 During the AIA, 186 test pits were started with four being terminated early due to encountering 4 inactive utilities and two locations not being excavated due to overlaps with known buried utilities. 5 Work was conducted with in-field assistance from representatives from Sevem' Qwantlen, 6 Skwxwú7mesh Úxwumixw, and Katzie Development Limited Partnership. Due to concerns 7 regarding the COVID-19 pandemic, several Nations did not attend field work, but remote 8 monitoring was provided by representatives from salilwatał, Katzie Development Limited 9 Partnership, x^wmə0k^wəýəm. Daily post-fieldwork summaries were also provided to Cowichan 10 Tribes, Katzie Development Limited Partnership, Lyackson First Nation, xwmə@kwəýam, Skwxwú7mesh Úxwumixw, səlilwətał, and Tsawwassen First Nation. 11

No archaeological materials or features were identified as part of the AIA. Recommendationsresulting from the AIA included the following:

- No further archaeological work for the Project footprint, provided that it is not altered to
 include areas that were not assessed during this AIA;
- No further archaeological work for the FEI property (Project area); and
- Project-specific Archaeological Chance Find Management Procedure is to be available to contractors prior to undertaking ground-altering activities.

In 2022, a project-specific Archaeological Chance Find Management Procedure was developed
 for the Project site, with input from Nations who requested to participate.

21 With the recommendations from the AIA and development of the Archaeological Chance Find

Management Procedure, the risk of archaeological impact has been downgraded from "moderateto low" to "low".

24 **7.3 CONCLUSION**

Over the past three years, FEI has undertaken additional environmental and archaeological activities, including a Stage 2 Preliminary Site Investigation, Limited Detailed Site Investigation, and completion of an Archaeological Impact Assessment for the site. Based on the additional assessments undertaken, FEI has downgraded the potential environmental impacts of the TLSE Project from "moderate" to "low to moderate" and has downgraded the potential archaeological impacts from "moderate" to "low". FEI expects to mitigate potential impacts through additional assessments, permitting, and standard protection and mitigation measures.



1 8. CONSULTATION AND ENGAGEMENT

2 **8.1 INTRODUCTION**

This section provides an update on consultation and engagement with Indigenous groups, the public, government and other stakeholders. These activities are integral components of FEI's project development process and meet the requirements of the the BCUC's CPCN Guidelines.

- 6 This section is organized around the following points:
- Section 8.2 FEI Has Continued to Engage with Indigenous Groups: FEI has undertaken, and will continue to undertake, significant engagement with Indigenous groups regarding the TLSE Project, including through the ongoing Environmental Assessment process. FEI has also entered into capacity funding and other agreements which have enabled Indigenous groups to meaningfully participate and contribute to these concurrent regulatory processes. FEI's engagement is sufficient, reflecting the TLSE Project's current stage of development.
- 14 • Section 8.3 – FEI Has Continued to Engage with the Public, Governments and 15 Stakeholders: FEI is committed to meaningful, proactive and ongoing engagement with 16 the public, governments and other stakeholders regarding the TLSE Project through, in 17 particular, the ongoing Environmental Assessment process. FEI's engagement approach has focused on sharing project information through various activities and public 18 19 communications channels, which have provided stakeholders with the opportunity to provide feedback. FEI's engagement is sufficient, reflecting the TLSE Project's current 20 21 stage of development.

22 8.2 FEI HAS CONTINUED TO ENGAGE WITH INDIGENOUS GROUPS

The Application and evidence presented throughout the regulatory proceeding demonstrates that consultation and engagement with Indigenous groups has been meaningful, timely and sufficient to date, given the nature of the approvals sought. These consultation and engagement activities remain consistent with the BCUC's CPCN Guidelines, are in alignment with other FEI applications approved by the BCUC, and reflect the TLSE Project's current stage of development.

FEI has continued to engage with Indigenous groups regarding the TLSE Project, responding to all issues, concerns and opportunities as they are raised.

30 8.2.1 Summary of FEI's Approach to Engagement with Indigenous Groups

As discussed in the Application and other evidence in this proceeding,²⁵⁸ FEI's engagement with Indigenous groups with respect to the TLSE Project is taking place through this CPCN Application

33 and the Environmental Assessment (EA) process for the FortisBC Holdings Inc. and FortisBC

²⁵⁸ See e.g., Exhibit B-1-4, Application, Section 8; Exhibit B-44, Rebuttal Evidence to TWN.



- 1 Energy Inc. (collectively, FortisBC) Tilbury Phase 2 LNG Expansion Project (Tilbury Phase 2 Expansion Project), which includes components of the TLSE Project.²⁵⁹ As these regulatory 2 3 processes remain underway concurrently, and in recognition of the resource constraints facing 4 many Indigenous groups, FEI sought to limit consultation fatigue by synchronizing engagement 5 activities between these regulatory processes where possible. FEI applies comments received 6 from Indigenous groups through this synchronized process to all applicable aspects of the 7 developments at Tilbury, including the TLSE Project, to ensure they are appropriately captured 8 and addressed.
- 9 Engagement with Indigenous groups with respect to the TLSE Project has continued to be guided 10 by FEI's Statement of Indigenous Principles (Appendix R-1 to the Application) and FEI's 11 Engagement Plan (Appendix Q-2 to the Application). FEI's engagement through the EA process 12 has been robust, and consistent with the BC Environmental Assessment Office's (BC EAO) 13 framework for consensus-seeking with Indigenous groups, as outlined in the Assessment Plan 14 for the Tilbury Phase 2 Expansion Project.²⁶⁰

15 8.2.2 Indigenous Groups Identified for TLSE Project

- 16 As set out in the Application, FEI identified the following 21 potentially affected Indigenous groups
- 17 to engage with specifically in respect of the TLSE Project.
- 18

Table 8-1: Indigenous Groups Potentially Affected by the TLSE Project

Indigenous Groups				
Cowichan Tribes	Musqueam Indian Band	Soowahlie First Nation		
Halalt First Nation	Penelakut Tribe	Squamish First Nation		
Katzie First Nation	Seabird Island Band	Stó:lö Nation		
Kwantlen First Nation	Semiahmoo First Nation	Stó:lö Tribal Council		
Lake Cowichan First Nation	Shxw'ōwhámél First Nation	Stz'uminus First Nation		
Lyackson First Nation	Skawahlook First Nation	Tsawwassen First Nation		
Métis Nation British Columbia	Snuneymuxw First Nation	Tsleil-Waututh Nation		

19 FEI also continues to engage with the following 22 additional Indigenous groups that have been

- 20 identified as potentially affected Indigenous Nations by the BC EAO as part of the ongoing EA
- 21 process for the Tilbury Phase 2 Expansion Project.²⁶¹

²⁵⁹ Components of the TLSE Project scope (e.g., the TLSE tank) forming part of the Tilbury Phase 2 Expansion Project is a function of the law governing the scope of environmental assessments under the *Environmental Assessment Act*, S.B,C. 2018, c. 51 (BC EAA) and the *Impact Assessment Act*, S.C. 2019, c. 28, s. 1 (IAA).

²⁶⁰ Tilbury Phase 2 Expansion Project, Schedule B – Assessment Plan: <u>https://projects.eao.gov.bc.ca/api/public/document/62c75af8e04a3a00225b7d84/download/Tilbury%20LNG%20P</u> <u>hase%202%20-%20Assessment%20Plan%20-%20Rev1%20-</u> <u>%20June%2013%202022%20%28EPIC%20Posting%29.pdf</u>.

²⁶¹ Tilbury Phase 2 Expansion Project, Schedule B – Assessment Plan, Sections 2.0 – 2.3 Consultation with Indigenous Nations: https://projects.eao.gov.bc.ca/api/public/document/62c75af8e04a3a00225b7d84/download/Tilbury%20LNG%20P



Indigenous Groups			
Aitchelitz First Nation	Pauquachin First Nation	Tsartlip Indian Band	
Chawathil First Nation	Popkum First Nation	Tsawout First Nation	
Cheam First Nation	Semá:th (Sumas) First Nation	Tseycum First Nation	
Leq'á:mel First Nation	Shxwhá:y Village	Tzeachten First Nation	
Kwaw-Kwaw-Apilt First Nation	Skowkale First Nation	Yakweakwioose First Nation	
Kwikwetlem First Nation	Skwah First Nation	Yale First Nation	
Malahat First Nation	Sq'éwlets (Scowlitz) First Nation		
Matsqui First Nation	Squiala First Nation		

Table 8-2: Indigenous Groups Identified as Potentially Affected by the BC EAO

2

1

3 8.2.3 TLSE Project-Specific Engagement with Indigenous groups

FEI's engagement with potentially affected Indigenous groups has continued to be meaningful,
timely and sufficient to date, consistent with the BCUC CPCN Guidelines, and reflects the TLSE
Project's current stage of development. In particular, FEI has focused its engagement activities

7 on ensuring that Indigenous groups:

- Are kept informed about the TLSE Project and have access to Project-related information
 as it becomes available;
- Are able to provide feedback, including opportunities to identify issues and concerns
 regarding the TLSE Project; and
- Have opportunities to describe how the TLSE Project may interact with their Indigenous interests.

FEI has provided these Indigenous groups with updates regarding the TLSE Project as new
 information arises, and has also addressed issues, concerns and opportunities through the
 following engagement activities, in particular:

- Regular Meetings with Indigenous Groups: FEI has continued to discuss the TLSE
 Project through regular project meetings with Indigenous groups. A summary of the
 issues, concerns and opportunities discussed with Indigenous groups since FEI's last
 evidentiary update regarding Indigenous engagement is provided in Section 8.2.4 below.
- Tilbury Site Tours: Since November 2021, FEI has held 16 site tours with Indigenous groups regarding the Tilbury Phase 2 Expansion Project. These tours provide Indigenous groups the opportunity to ask questions about the TLSE Project and to identify areas of concern, interest or opportunity relating to the Project.

hase%202%20-%20Assessment%20Plan%20-%20Rev1%20-%20June%2013%202022%20%28EPIC%20Posting%29.pdf.



- 1 • Updates Regarding the CPCN Application: FEI sent a notification letter to the 21 2 potentially affected Indigenous groups (see Table 8-1 above) providing an update 3 regarding the TLSE Project, including the status of the Project and this Application before 4 the BCUC. Indigenous groups were offered an opportunity to ask questions and seek 5 additional information. A copy of this letter is included as Appendix L to the Supplemental 6 Evidence. FEI received two responses: (1) on December 20, 2023, TWN confirmed receipt 7 of the letter; and (2) on January 3, 2024, Lyackson First Nation sought clarification 8 regarding the full name of FEI, which FEI provided in response on January 10, 2024.
- **TWN Final Argument:** In November 2022, TWN filed its Final Argument in this proceeding advising the BCUC that they do not oppose the TLSE Project and are not seeking to delay the Project from moving forward. TWN has not indicated that this position has changed since this filing.²⁶²

8.2.4 FEI Has Responded to Issues, Concerns and Opportunities Raised by Indigenous Groups

FEI has continued to respond to and address concerns raised by Indigenous groups regarding
 the TLSE Project. Since November 2021, when FEI provided its last evidentiary update in this
 proceeding regarding its engagement activities, FEI received inquiries regarding the following:

- The potential environmental impacts of the TLSE Project; including cumulative effects,
 greenhouse gas (GHG) emissions and marine impacts; and
- Business opportunities related to the TLSE Project.

21 *8.2.4.1* Potential Environmental Impacts

FEI has continued to discuss concerns raised by some Indigenous groups regarding cumulative effects of increased development on and near Tilbury Island, especially as it relates to GHG emissions and increased shipping on the Fraser River related to waterborne delivery of Project materials. In response to the concern regarding increased shipping on the Fraser River, FEI has removed all waterborne delivery of Project materials. FEI has also engaged Indigenous groups regarding air quality, human health and GHG emissions through BC EAO-led workshops and regular project meetings.

- 29 Some Indigenous groups also remain concerned about up-stream emissions from the production
- 30 of LNG, and downstream emissions from LNG's use as a marine fuel. While these concerns are
- 31 largely related to issues beyond the scope of the TLSE Project, FEI continues to work closely with
- 32 these Indigenous groups to incorporate their concerns in a manner consistent with the CPCN
- 33 Guidelines and the EAO effects assessment methodology.

²⁶² See TWN Final Argument: <u>https://docs.bcuc.com/documents/arguments/2022/doc_68841_2022-11-21-twn-final-argument.pdf</u>.



1 8.2.4.2 Business Opportunities

2 FEI has continued to discuss business opportunities on the TLSE Project with Indigenous groups.

In particular, FEI has held multiple meetings with Indigenous groups and provided information via
 email to address concerns or further discuss opportunities. Specific areas of interest have
 included:

- 6 Potential contracting opportunities;
- Job training and mentoring opportunities, including Indigenous monitoring;
- Post-secondary opportunities, and future opportunities for youth in archaeological
 programming, environmental stewardship, and fish habitat programs; and

Requests for equitable access to business opportunities, particularly for Indigenous groups that are not in immediate proximity to the TLSE Project site.

In November 2022, FEI held an in-person business-to-business session with local Indigenous affiliated and member-owned businesses. In October 2023, FEI also facilitated a business-tobusiness session with Musqueam Indian Band to discuss procurement opportunities for Musqueam affiliate and member-owned businesses. These sessions included a presentation on the Tilbury development, including TLSE-specific project components, and an opportunity to network with attendees. FEI will continue to engage Indigenous groups regarding business opportunities as the Project progresses, through various methods and forums.

In addition, during an April 2022 Process Planning workshop led by the BC EAO, Seabird Island Band inquired about Project safety and response procedures at the Tilbury Project site, in the case of an emergency. FEI responded by providing additional information regarding Project safety

22 and associated response procedures. FEI has not received any further response on the matter

and, therefore, considers this concern to be addressed.

24 8.2.5 Agreements with Indigenous Groups

Given the scope of the Project, which requires an EA and significant participation from Indigenous groups, FEI has finalized funding agreements with 15 Indigenous groups that support their capacity to actively participate in and engage regarding the Tilbury Phase 2 Expansion Project, including the TLSE Project. FEI has offered a capacity funding agreement to one additional Indigenous group that is not yet finalized. FEI continues regular dialogue with Indigenous groups to discuss capacity constraints and advance agreement-specific deliverables.

- In addition to the funding agreements described above, FEI has executed project agreementswith the following two Indigenous groups:
- Musqueam Indian Band (August 2022);²⁶³ and

²⁶³ See Exhibit A2-1 and Exhibit B-54-1, BCUC IR4 114.1.



1 • Snuneymuxw First Nation (January 2023).²⁶⁴

Both agreements reflect FortisBC's collective efforts to build strong relationships with Indigenous groups regarding the TLSE Project and other Tilbury Projects, to meaningfully engage with potentially affected Indigenous groups and to seek their free, prior, and informed consent. These agreements demonstrate the value of meaningful engagement undertaken in close collaboration

6 and the spirit of reconciliation for customers of a regulated utility.

7 8.2.6 Engagement with Indigenous Groups Will Continue

FEI will continue engaging with potentially affected Indigenous groups as development of the TLSE Project progresses, including engagement through the ongoing EA process, and other future permitting processes. In particular, FEI will continue to work collaboratively with Indigenous groups to address existing concerns, address new questions that arise through future engagement, and identify opportunities for Indigenous groups and their members where possible. FEI will also continue to provide updates to Indigenous groups regarding the Application, including notifying these groups about the filing of this Supplemental Evidence.

15 8.3 FEI HAS CONTINUED TO ENGAGE WITH THE PUBLIC, GOVERNMENTS, AND 16 STAKEHOLDERS

FEI is committed to meaningful engagement with the communities it serves, particularly where its major projects are taking place. Since its last evidentiary update, FEI has continued to actively engage with the public, including its customers, residents, businesses and landowners located near the facility and surrounding area, governments and other stakeholders regarding the TLSE Project using a variety of engagement methods and, in particular, through the ongoing EA process.

As discussed further in the sections below, FEI's engagement with the public, governments and other stakeholders has been meaningful, timely and sufficient to date, given the nature of the approvals sought. These consultation and engagement activities remain consistent with the BCUC's CPCN Guidelines, are in alignment with other FEI applications approved by the BCUC,

and reflect the TLSE Project's current stage of development.

8.3.1 Summary of FEI's Approach to Engagement with the Public, Government and Other Stakeholders

30 FEI's engagement efforts are designed to ensure that the public, governments, and other 31 stakeholders are kept informed about the Project, have access to relevant information, and are 32 provided with multiple opportunities to provide feedback throughout project development. This

²⁶⁴ Snuneymuxw First Nation and FortisBC Holdings Inc. sign agreement for Tilbury LNG projects, strengthening longstanding relationship (Jan. 27, 2023): <u>https://www.fortisbc.com/news-events/media-centre-details/2023/01/27/snuneymuxw-first-nation-and-fortisbcholdings-inc.-sign-agreement-for-tilbury-Ing-projects-strengthening-long-standing-relationship.</u>



- 1 engagement approach supports FEI's goal of understanding public interests and incorporating
- 2 feedback into the Project design. Since FEI's last evidentiary update, it has continued to create
- 3 opportunities for stakeholders to learn more about the TLSE Project, to ask questions, and to
- 4 provide feedback, including identifying issues, concerns and opportunities, as applicable.
- 5 FEI has also, and will continue to, maintain the positive relationships developed during 6 engagement to date, particularly with those located closest to the TLSE Project, and with those 7 who have demonstrated a high level of interest. These engagement activities remain consistent 8 with the communication and consultation objectives identified in Section 8.2.2 of the Application
- 8 with the communication and consultation objectives identified in Section 8.3.2 of the Application.

9 Further, as described in Section 8.2.1 above, FEI sought to limit consultation fatigue of 10 stakeholders by synchronizing engagement activities where possible between the concurrent 11 regulatory processes. FEI applies comments received from stakeholders through this 12 synchronized process to all applicable aspects of the developments at Tilbury, including the TLSE 13 Project, to ensure they are appropriately captured and addressed.

14 8.3.2 Engagement with the Public and Other Stakeholders

- FEI has continued to engage with the public and other stakeholders through several engagementactivities:
- Site Tours & LNG Demonstrations: Since November 2021, FEI has held 76 site tours with stakeholders and governments. These tours have provided opportunities for interested stakeholders to ask questions regarding the TLSE Project, and for FEI to educate interested groups on the day-to-day operations of FEI's existing LNG storage facility. FEI has also provided multiple opportunities for stakeholders to view live demonstrations of LNG during FEI-led site tours, and upon request.
- Meetings & Presentations: Since November 2021, FEI has held 41 project meetings and presentations with interested members of the public and stakeholders. These meetings have taken place in both a virtual and in-person format to ensure project information is accessible for interested groups.
- 27 Broad Digital & Customer Communications: FEI has established and maintained 28 various communication channels. including project-specific а website on 29 TalkingEnergy.ca, where educational materials such as videos and articles have been 30 posted to help the public understand LNG and learn more about the TLSE Project. FEI 31 has also shared these materials on social media and on FortisBC websites to further 32 engage the public. For example, in August 2023, FEI posted an article on Talking Energy 33 focusing on how the TLSE Project will improve the resilience of FEI's gas system.²⁶⁵
- Community Events: FEI actively participates in local community events held throughout
 the year, providing many opportunities for the public to ask question and learn about the
 TLSE Project, in addition to general information about FortisBC and its operations.

²⁶⁵ https://talkingenergy.ca/stories/tilbury-expansion-ensuring-energy-there-when-you-need-it.



- EA Open Houses: In March 2022, FEI participated in two virtual open houses hosted by the BC EAO as part of the prescribed 45-day public comment period during the Process Planning phase of the EA. These open houses allowed FEI to continue to understand key issues, engage with the public and propose further mitigations, where necessary.
- Government representatives also participated in a number of the activities summarized above.
 FEI addresses government-specific engagement below.

7 8.3.3 Engagement with Government

8 FEI has engaged directly with nearby local, provincial, and federal government agencies to share 9 updates and seek feedback regarding the TLSE Project. In particular, FEI meets regularly with the City of Delta to provide updates related to the Tilbury LNG facility, including with respect to 10 the TLSE Project, and provides advance notice to government officials of FEI related activities 11 12 taking place in their communities. FEI has also provided ongoing updates to the nearby City of 13 Richmond. Further, FEI has engaged local government staff, first responders, and other 14 stakeholders in full-scale emergency exercises at the Tilbury LNG facility, fostering collaboration 15 and preparedness, in addition to providing specific LNG training for firefighters.

16 Finally, as part of the ongoing EA process for the Tilbury Phase 2 Expansion Project, which 17 includes components of the TLSE Project, FEI has maintained regular engagement with 18 government and government agencies through the Technical Advisory Committee (TAC). The 19 TAC's role as part of the EA process is to advise the BC EAO and participating Indigenous Nations 20 on technical matters related to the assessment; and to review FortisBC's application for an EA 21 certificate.²⁶⁶ The TAC includes participating Indigenous Nations, local governments, provincial, 22 and federal agencies. Since its last evidentiary update, FEI has engaged with government 23 representatives through more than 910 interactions, including 45 site tours, 22 project meetings, 24 and over 840 individual engagements by email or phone.

8.3.4 FEI Has Responded to Issues, Concerns and Opportunities Raised by the Public, Governments and Other Stakeholders

The following sub-sections describe the inquiries, including issues, concerns and opportunities, regarding the TLSE Project that FEI has received from the public, governments and other stakeholders since its last evidentiary update. In general, FEI has continued to reach a wide crosssection of stakeholders through diverse channels, consistent with the communication and consultation objectives identified in Section 8.3.2 of the Application.

32 *8.3.4.1* Inquiries from the Public and Other Stakeholders

33 Since the last evidentiary update regarding its public engagement activities, FEI has received 34 inquiries from interested members of the public and other stakeholders through various

²⁶⁶ Tilbury Phase 2 Expansion Project, Technical Advisory Committee Terms of Reference, June 2022: <u>https://projects.eao.gov.bc.ca/api/public/document/629a5bd4eedf690022c8657b/download/TAC%20Terms%20of</u> <u>%20Reference TilburyLNG Final.pdf</u>.



1 communication channels, including by email and through project meetings and site tours. These

2 inquiries have generally focused on aspects of project development, economic opportunities,

3 environmental impacts, as well as issues that are beyond the Project scope. The primary interests

- 4 raised are specifically related to:
- The purpose and need for the proposed Project;
- Rate impacts associated with the proposed Project;
- GHG emissions associated with LNG for export and marine fuelling; and
- Contracting and workforce opportunities associated with the TLSE Project.

9 FEI has responded directly to public and other stakeholder inquiries by providing further 10 information regarding the purpose of the TLSE Project, associated rate impacts, as well as the 11 projected economic benefits associated with Project construction. Where applicable, FEI has also 12 provided additional information regarding how potential environmental impacts, such as GHG 13 emissions, are assessed through the ongoing parallel EA process for the Tilbury Phase 2 14 Expansion.

FEI has committed to providing further information regarding the TLSE Project when it becomes
available, to address questions on topics such as the construction schedule, workforce impacts,

17 and job opportunities as development of the Project progresses. FEI has received positive

18 feedback regarding site tours and presentations and will continue to use these engagement

19 methods to respond to issues raised by stakeholders and members of the public.

FEI also recognizes that multiple stakeholder groups and members of the public have concerns regarding issues that are beyond the Project scope (e.g., marine shipping regulations and upstream natural gas extraction). FEI has, to the best of its ability, provided interested parties with supplemental information regarding these issues. One representative also inquired about visual

24 effects of LNG facilities once in operation.

25 *8.3.4.2* Inquiries from Governments

FEI has continued to engage with government representatives to address outstanding concerns, issues and inquiries relating to:

- Project construction and LNG facility operation;
- Traffic impacts;
- Environmental impacts;
- Safety; and
- Issues beyond the Project scope.
- 33 FEI addresses each of these inquiries, as well as FEI's response, further below.



1 8.3.4.2.1 PROJECT CONSTRUCTION AND LNG FACILITY OPERATION

During a process planning workshop as part of the ongoing EA process, FEI received an inquiry
regarding hydro-testing during construction of the TLSE Project. In particular, FEI was asked
about the use of surface water for the hydro testing of tanks and raised concerns regarding water
contamination. In response, FEI confirmed that water from the Fraser River would be used for
hydro testing, but that the water would be treated prior to discharge.

The City of Richmond inquired about the technical equipment at the Tilbury LNG facility, as well
as the associated energy usage and GHG emissions with the use of a hot oil heater at the LNG
facility.

10 8.3.4.2.2 ENVIRONMENTAL IMPACTS

Interest in and concerns regarding environmental impacts were raised by multiple government representatives during regular project engagement. Questions and discussion during regular project meetings generally focused on FEI's decarbonization goals, upstream GHG emissions and how GHG emissions are being examined in the EA process.

Environment and Climate Change Canada inquired about whether FEI is considering mitigations to reduce interactions with migratory birds. FEI stated that there has not been a completed assessment to identify a need for mitigations. Further, if a need for mitigations is identified, they will be implemented by FEI.

19 Finally, the City of Vancouver asked FEI whether the potential difference in GHG emissions

20 depending on the use of barges or trucks to deliver Project materials would be assessed. FEI

21 confirmed that GHG emissions will be included in the assessment during the EA process.

22 8.3.4.2.3 ISSUES OUTSIDE OF THE PROJECT SCOPE

One local government inquired about the use of renewable natural gas (RNG) as a fuel for customers. FEI re-iterated that its focus is exploring opportunities to use RNG in its operations at the Tilbury facility, including the TLSE Project, as a means to manage the facility's GHG emissions and meet the Project's net-zero planning requirements.

27 8.3.4.2.4 OTHER ISSUES

During a project update meeting, one local government asked questions about the emergency response procedures of the Tilbury LNG facility, and if alternatives to the project's location at the existing facility had been considered. FEI provided information about the safety protocols for the facility, including engagement undertaken with local first responders to participate in LNG training and emergency planning exercises. FEI also noted that the CPCN Application and EA draft application have more detailed information on alternatives that have been considered for the TLSE Project.

FEI received one question from a local government representative related to traffic impacts, and what plans are in place to assess truck traffic resulting from construction of the Project. FEI identified trucking routes and potential loads during those transfer windows. FEI confirmed that



traffic routes are assessed in incremental change and the cumulative effects assessment notes
 increased congestion if multiple projects will be occurring during the same time.

8.3.5 Engagement with the Public, Governments and Other Stakeholders Will Continue

5 FEI is dedicated to maintaining and strengthening positive relationships with the public, 6 governments and other stakeholders through an open and transparent engagement process 7 throughout the duration of the Project. Please refer to the Section 8.3.10 of the Application which 8 discusses FEI's approach to continued engagement. This approach, and the associated areas of 9 focus, remain relevant. In particular, FEI will continue to provide regular updates as this 10 proceeding advances, and through subsequent stages of project development, such as BCER 11 permitting.

FEI will also continue to actively engage with the public, governments and other stakeholders regarding the TLSE Project through the remainder of the EA process. FEI currently anticipates submitting the Draft Application as part of the EA process in Q3 2024. After the Draft Application has been submitted, a public comment period will open which will include further opportunities for engagement such as both in-person and virtual open houses.

Finally, FEI will explore further opportunities to host live LNG demonstrations to educate stakeholders and help the public better understand the properties of LNG, including continuing to seek participation from municipal staff and local stakeholders in future emergency preparedness exercises.

21 **8.4 CONCLUSION**

FEI has actively engaged with Indigenous groups, the public, governments and other stakeholders regarding the TLSE Project through open, transparent and timely engagement methods that has resulted in meaningful two-way dialogue. FEI has made best efforts to ensure Indigenous groups and stakeholders are informed and engaged about the Project holistically and to allow for synchronized engagement activities with the parallel Provincial EA and Federal IA processes, which provide significant engagement opportunities.

To date, FEI has identified and responded to concerns raised by Indigenous groups, the public, and stakeholders. FEI's consultation and engagement has been sufficient, reflecting the TLSE Project's stage of development and meeting the requirements of the the BCUC's CPCN Guidelines. FEI will continue to engage directly with Indigenous groups, stakeholders and government to address outstanding issues on the Project, to incorporate feedback into Project mitigations, and to share relevant Project updates through the engagement activities outlined in the Application and in this Supplemental Evidence.

35



BC ENERGY OBJECTIVES AND LONG TERM GAS RESOURCE PLAN

3 **9.1 INTRODUCTION**

7

8

4 This section discusses the factors that the BCUC must consider pursuant to section 46(3.1) of 5 the *Utilities Commission Act* (UCA) when determining whether to issue a CPCN:

- 6 a) the applicable of British Columbia's energy objectives,
 - b) the most recent long-term gas resource plan filed by the public utility under section 44.1, if any, and
- 9 c) the extent to which the application for the certificate is consistent with the applicable 10 requirements under sections 6 and 19 of the *Clean Energy Act*.

11 Sections 6 and 19 of the *Clean Energy Act* (CEA), as referred to in subsection (c) above, do not 12 apply to FEI. FEI addresses the other two requirements below.

13 9.2 BRITISH COLUMBIA'S ENERGY OBJECTIVES

The applicable of British Columbia's energy objectives support the TLSE Project. FEI anticipates positive socio-economic benefits to the area as a result of the Project. Otherwise, British Columbia's energy objectives are generally neutral in relation to the Project. Some objectives do not apply to FEI. Other objectives either do not apply to the TLSE Project, or are not in conflict with the Project, as the Project is designed to allow FEI to continue meeting winter peak demand and provide energy resiliency in the Lower Mainland. There is currently no feasible alternative peak resource available to serve this load or provide the necessary resiliency.

Section 3 of the Supplemental Evidence discusses the Project drivers. The Project is a critical asset within FEI's gas supply portfolio (i.e., FEI's ACP). Since 1971, FEI's gas supply portfolio has included on-system LNG storage and regasification capacity at Tilbury in its planning to meet customer demand. Losing access to some or all of the peaking capacity and energy currently provided by Tilbury would impair FEI's ability to provide uninterrupted service to customers in winter. Adding on-system LNG is also the only way to materially reduce the customer outage risk associated with a winter no-flow event on T-South.

Table 9-1 below sets out each of British Columbia's energy objectives and their applicability to the Project.



	4	
	Т	

Table 9-1: British Columbia's Energy Objectives²⁶⁷

ltem	Objective	Comments
(a)	to achieve electricity self-sufficiency;	The Project does not affect the generation or acquisition of electricity or otherwise impact the Province's achievement of electricity self-sufficiency.
(b)	to take demand-side measures and to conserve energy, including the objective of the authority reducing its expected increase in demand for electricity by the year 2020 by at least 66%;	FEI is implementing its accepted 2024-2027 DSM Expenditures Plan, which includes the Lower Mainland, to take demand-side measures and conserve energy. The peak load served by the Project is net of demand side measure savings (and the 66 percent reduction in demand applies to BC Hydro and is not applicable to FEI).
(c)	by 2030, to ensure that 100% of the electricity generated in British Columbia and supplied to the integrated grid is generated from clean or renewable resources, and to ensure that the infrastructure necessary to transmit that electricity is built;	The Project does not affect the generation or supply of electricity.
(d)	to use and foster the development in British Columbia of innovative technologies that support energy conservation and efficiency and the use of clean or renewable resources;	The Project does not impact FEI's ability to deliver renewable and low carbon gas supplies to its customers or utilize other innovative technologies that support energy conservation and efficiency and the use of clean or renewable resources. The Project replaces existing storage and regassification capacity while also mitigating exposure to significant resiliency risks for which there are no other more reasonable alternatives.
(e)	to ensure the authority's ratepayers receive the benefits of the heritage assets and to ensure the benefits of the heritage contract under the <i>BC Hydro Public Power Legacy and Heritage</i> <i>Contract Act</i> continue to accrue to the authority's ratepayers;	This objective applies to BC Hydro and is not applicable to FEI.
(f)	to ensure the authority's rates remain among the most competitive of rates charged by public utilities in North America;	This objective applies to BC Hydro and is not applicable to FEI.
(f.1)	 to ensure that changes to the authority's rates (i) are reasonably predictable, and (ii) are reasonably consistent from year to year; 	This objective applies to BC Hydro and is not applicable to FEI.
(f.2)	to ensure that increases to the authority's rates do not exceed cumulative inflation;	This objective applies to BC Hydro and is not applicable to FEI.

 $^{^{\}rm 267}$ As set out in section 2 of the CEA, as amended on February 15, 2024.



ltem	Objective	Comments
(g)	 to reduce BC greenhouse gas emissions: (i) by 2012 and for each subsequent calendar year to at least 6% less than the level of those emissions in 2007, (ii) by 2016 and for each subsequent calendar year to at least 18% less than the level of those emissions in 2007, (iii) by 2020 and for each subsequent calendar year to at least 33% less than the level of those emissions in 2007, (iv) by 2050 and for each subsequent calendar year to at least 80% less than the level of those emissions in 2007, (iv) by 2050 and for each subsequent calendar year to at least 80% less than the level of those emissions in 2007, and (v) by such other amounts as determined under the <i>Climate Change Accountability Act</i>; 	The Project does not conflict with the reduction of greenhouse gas emissions in BC. The Project is designed to support energy system resiliency and peak energy demand requirements in the Lower Mainland during cold winter conditions and there is currently no feasible alternative peak resource available to serve this load. The "resiliency reserve" is stored, unused, until it is required in a supply emergency. Further, the Project's allocation to the gas supply portfolio will facilitate customers' continued use of renewable natural gas even during peak demand conditions, as the renewable natural gas is blended on FEI's system and allocated to FEI's Sales customers, to reduce emissions in BC.
(g.1)	to ensure that the authority holds rights to a sufficient amount of clean or renewable electricity to enable British Columbia to meet the objective set out in paragraph (g);	This objective applies to BC Hydro and is not applicable to FEI.
(h)	to encourage the switching from one kind of energy source or use to another that decreases greenhouse gas emissions in British Columbia;	The Project is designed to meet energy system resiliency and peak demand and will not prevent the switch to other energy sources that can decrease greenhouse gas emissions, such as electricity or renewable natural gas. The Project does not affect customer use of renewable natural gas, which is blended on FEI's system and allocated to FEI's Sales customers, to reduce emissions in BC.
(i)	to encourage communities to reduce greenhouse gas emissions and use energy efficiently;	The Project is designed to meet energy system resiliency and peak demand and will not prevent communities from reducing greenhouse gas emissions or using energy efficiently.
(j)	to reduce waste by encouraging the use of waste heat, biogas, and biomass;	The Project does not affect customer use of renewable natural gas, which is blended on FEI's system and allocated to FEI's Sales customers, to reduce emissions in BC.



ltem	Objective	Comments
(k)	to encourage economic development and the creation and retention of jobs;	The Project will benefit the local economy during the construction phase by creating jobs in BC through FEI's contractors, and result in the procurement of goods and services from locally owned and operated vendors and subcontractors (i.e., the use of local hotels and restaurants for employees working on the construction sites). FEI is committed to working with Indigenous groups, community leaders and local organizations, developing the local workforce, supporting local businesses, and connecting them to Project opportunities. The Project will also ensure adequate capacity is available to support economic activity and growth in the region.
		The British Columbia energy objective related to retention of jobs is also served by reducing the potential for a T-South no-flow event to cause a widespread and lengthy outage in the Lower Mainland, British Columbia's most important economic region. A loss or disruption of gas supply would impact many hundreds of thousands of natural gas customers who use gas in their homes and businesses, plus those who indirectly rely on natural gas for access to goods or services. The PwC Report ²⁶⁸ provides additional analysis of the potential GDP implications of an outage.
(I)	to foster the development of first nation and rural communities through the use and development of clean or renewable resources;	The Project does not affect the development of clean or renewable resources.
(m)	to maximize the value, including the incremental value of the resources being clean or renewable resources, of British Columbia's generation and transmission assets for the benefit of British Columbia;	The Project does not affect BC's generation and transmission assets.
(n)	to be a net exporter of electricity from clean or renewable resources with the intention of benefiting all British Columbians and reducing greenhouse gas emissions in regions in which British Columbia trades electricity while protecting the interests of persons who receive or may receive service in British Columbia;	The Project does not affect the generation or export of electricity.
(o)	to achieve British Columbia's energy objectives without the use of nuclear power;	The Project does not affect the generation of electricity.

1

²⁶⁸ Appendix RP 3 to the 2024 Resiliency Plan, PwC Report.



- Section 4 of the CEA indicates that the objectives in section 2(f.2) and (g) of the CEA have priority,
 as follows:
- 3 4 The energy objectives set out in section 2 (f.2) and (g) of the Act have 4 priority over the other energy objectives set out in that section.

5 The objective in section 2(f.2) applies only to BC Hydro and is therefore not applicable to the 6 Project. As noted in Table 9-1 above, the Project does not conflict with the objective in section 7 2(g) to reduce GHG emissions, as it is designed to strengthen long term energy system resiliency 8 and serve peak demand through the winter for which there is no available alternative, and will 9 facilitate FEI's customers' use of renewable natural gas – even during these peak periods. Since 10 the Project is not in conflict with this objective, the priority to be given to it has no bearing on the 11 Draiest

- 11 Project.
- 12 In summary, a consideration of British Columbia's energy objectives supports the Project, and
- 13 FEI anticipates positive socio-economic benefits to the area as a result of the Project. A
- 14 consideration of the remaining objectives is neutral in relation to the Project.

15 9.3 Long Term Gas Resource Plan

16 This Application considers and is consistent with the outcome of the 2022 LTGRP proceeding, 17 which identified the need for FEI to revise and resubmit its resiliency plan. FEI has performed that 18 work, and the 2024 Resiliency Plan provides the basis for this Supplemental Evidence. Further, 19 Section 6.3.2 of the 2022 LTGRP explains the vital role of the existing Tilbury Base Plant in 20 providing gas supply throughout the year and, in particular, during peak demand events²⁶⁹ and 21 that this type of critical service is difficult to replace with market alternatives.

The need for the TLSE Project as a resiliency measure was considered at length in the 2022
 LTGRP proceeding,²⁷⁰ supported by the initial version of FEI's Resiliency Plan. This initial version
 of the plan was appended to the 2022 LTGRP and included in the LTGRP Action Plan.²⁷¹

The BCUC Panel for the 2022 LTGRP agreed with the findings in the Adjournment Decision²⁷² that additional analysis and information about FEI's overall risk exposure was needed for the BCUC to complete its review of the Resiliency Plan and make any decisions regarding the infrastructure needed to implement it.²⁷³ While the Panel in the 2022 LTGRP rejected the initial version of the Resiliency Plan as presented in the 2022 LTGRP, it noted that:

²⁶⁹ 2022 LTGRP, Exhibit B-1, Section 6.3.2, pp. 6-25.

²⁷⁰ 2022 LTGRP, Exhibit B-1: Section 7.5 discusses gas system resiliency; Section 10, Action Item 6 discusses inclusion of the TLSE Project as part of the Resiliency Plan in the actions FEI intends to pursue, numerous information requests in the LTGRP regulatory proceeding discussed the TLSE Project and the resiliency plan. 2022 LTGRP, Decision and Order G-78-24, Section 3.5, pp. 38-40, discusses the TLSE Project and FEI's initial resiliency plan.

²⁷¹ 2022 LTGRP, Exhibit B-1, Section 10, Action Plan Item 6, p. 10-5.

²⁷² The Adjournment Decision was made after the 2022 LTGRP was submitted, but prior to the 2022 LTGRP Decision and Order G-78-24.

²⁷³ 2022 LTGRP Decision and Order G-78-24, p. 40.



- ...FEI has committed to preparing a new resiliency plan that will include a more
 comprehensive and robust analysis, and intends to include the latest version of its
 resiliency plan in future long-term gas resource plans for BCUC review. The Panel
 considers this commitment to be reasonable and appropriate.
- 5 The 2024 Resiliency Plan filed concurrently with this Supplementary Evidence completes FEI's 6 commitment made during the 2022 LTGRP to revise its resiliency plan and is resubmitting that 7 part of the 2022 LTGRP that was rejected as set out in section 44.1(7) of the UCA.
- 8 As discussed in Section 3 of this Supplemental Evidence, the 2024 Resiliency Plan confirms that 9 a winter no-flow event on T-South is the largest customer outage risk by a wide margin. The risk
- 10 is significant and should be mitigated with the addition of on-system LNG.

11 **9.4 CONCLUSION**

- 12 The TLSE Project is consistent with the applicable of British Columbia's energy objectives set out
- 13 in Section 2 of the *Clean Energy Act.* FEI's 2024 Resiliency Plan, which is filed concurrently with
- 14 this Supplemental Evidence, completes FEI's commitment made during the 2022 LTGRP (and
- 15 confirmed by the BCUC Panel in the 2022 LTGRP Decision), and confirms the resiliency need for
- 16 the TLSE Project.

17



1 10. CONCLUSION

2 A total loss of supply to the Lower Mainland during winter is, and will remain, FEI's largest 3 customer outage risk by an order of magnitude. The consequences of a winter no-flow event, 4 lasting only a matter of hours on T-South, are known, and if left unmitigated, will be catastrophic. 5 It will, without question, result in at least 600,000 customers losing service within a day of the 6 event. Service would not be fully restored to those customers for many weeks. A lengthy loss of 7 gas supply during winter will expose vulnerable populations to serious health and mortality risks 8 and have widespread social and economic consequences for British Columbians. Exponent has 9 calculated a high cumulative probability of a winter no-flow event, even assuming a 23-year evaluation horizon sensitivity that is less than half of the expected service life of the TLSE Project. 10 11 Since the 2018 T-South Incident there have been two incidents on T-South that highlight FEI's 12 risk exposure to upstream supply disruptions given its heavy reliance on T-South.

13 It is also now even more clear that the 53-year-old Base Plant needs to be replaced soon 14 irrespective of resiliency. The Base Plant has reached end-of-life and is unable to reliably perform 15 its critical gas supply function. A strategy of prolonged reliance on the Base Plant for critical 16 peaking supply would leave firm customers at serious risk of losing service in normal operations, 17 given the Base Plant's age-related issues and the absence of any feasible contingency plan for 18 peaking supply. Allowing this to occur would contradict the gas supply planning approach that has 19 been in place for decades.

Replacing the Base Plant with a facility that will both meet FEI's gas supply requirements and mitigate a known catastrophic risk associated with the loss of gas supply to the Lower Mainland is appropriate and necessary. FEI's analysis, presented in this Supplemental Evidence, shows that FEI customers will obtain much greater value from the Preferred Alternative (Supplemental Alternative 9) than any other viable alternative.

FEI respectfully submits that the TLSE Project is in the public interest and should be approved as proposed, along with the associated deferral account and depreciation rate approvals requested in Section 6 of the Application.

28

Appendix A REFERENCES TO PREVIOUS FEI EVIDENCE BY TOPIC



No.	Application Topic	Reference to Evidence		
Pro	Project Need and Justification			
1.	Customers recognize the importance of resiliency.	• Exhibit B-15, BCUC IR1 7.1 and Attachment 7.1.		
2.	Limited gas distribution infrastructure in the Pacific Northwest region	• Exhibit B-1-4, Application, Section 3.4.2.1, pp. 37-39		
	South system).	 Exhibit B-1-4, Application, Appendix A (Guidehouse Report on Natural Gas System Resiliency), pp. 29 to 38. 		
		• Transcript Vol 1, p. 41, l. 7 to p. 42, l. 23.		
		• Transcript Vol 1, p. 122, l. 7 to p. 123, l. 9.		
3.	The 2018 T-South Incident resulted in a no-flow event lasting approximately two days, during which the Lower Mainland system was at material risk of hydraulic collapse.	• Exhibit B-1-4, Application, Section 3.4.2.2, pp. 39 to 50.		
4.	T-South and other natural gas systems in North America have experienced major service disruptions and near misses.	 Exhibit B-1-4, Application, Appendix B (PwC – The Case for Improved System Resiliency), p. 17. 		
		• Exhibit B-15, BCUC IR1 3.1.		
5.	FEI's Lower Mainland system only avoided a widespread and	• Exhibit B-1-4, Application, pp. 45-46 and 52.		
	weather conditions and a favourable pipeline rupture location.	• Transcript Vol 1, p. 73, l. 9 to p. 74, l. 8.		
6.	Contractual rights to additional gas supply and capacity on the T- South system are of no assistance to FEI during a no-flow event on	 Exhibit B-1-4, Application, p. 22 and Appendix A (Guidehouse Report on Natural Gas System Resiliency), pp. 18 to 23. 		
	T-South.	• Exhibit B-24, Sentinel IR1 55.		
		• Transcript Vol 1, p. 31, ll. 10 to 18.		



No.	Application Topic	Reference to Evidence
7.	Other potential supply sources, including market area storage (Mist and JPS), mutual aid, and the Mt. Hayes LNG facility cannot be relied on during a winter no-flow event on T-South.	Market Area Storage
		• Exhibit B-1-4, Application, pp. 69 to 73.
		 Exhibit B-1-4, Application, Appendix A (Guidehouse Report on Natural Gas System Resiliency), p. 14.
		• Exhibit B-15, BCUC IR1 16.14, 16.16 and 46.1.
		• Exhibit B-19, CEC IR1 25.1 and 25.2.
		• Transcript Vol 1, p. 43, II. 5 to 21.
		• Transcript Vol 1, p. 52, II. 2 to 20.
		• Transcript Vol 1, p. 61, l. 17 to p. 62, l. 13.
		• Transcript Vol 1, p. 133, II. 3 to 12.
		Mutual Aid
		• Exhibit B-1-4, Application, p. 52.
		• Exhibit B-15, BCUC IR1 4.2 and Attachment 4.2.
		• Exhibit B-26, BCUC IR2 74.1.
		• Transcript Vol 1, p. 155, l. 26 to p. 156, l. 11.
		<u>Mt Hayes LNG</u>
		• Exhibit B-15, BCUC IR1 11.8.
		• Transcript Vol 1, p. 149, l. 22 to p. 150, l. 1.
		• Transcript Vol 1, p. 174, ll. 16-26.
		• Transcript Vol 1, p. 175, ll. 7-11.
8.	FEI's ability for FEI's Lower Mainland system to withstand a no-flow	• Exhibit B-1-4, Application, p. 65 (Figure 3-14).
	event requires both dependable "energy" (regasification) and "capacity" (storage).	 Exhibit B-1-4, Application, Appendix C (2020/21 ACP Compliance Report), p. 15.
		• Exhibit B-26, BCUC IR2 76.1 and 78.1.



No.	Application Topic	Reference to Evidence
9.	Restoring service to hundreds of thousands of customers after a	Exhibit B-46, Rebuttal Evidence to RCIA.
	BCUC-approved P&R Plan, regulations, standards and the experience of other utilities.	• Transcript Vol 1, p. 137, l. 17 to p. 138, l. 23.
10.	Where the consequence of events are unacceptably severe, an	Exhibit B-18, BCSEA IR1 2.1.
	appropriate risk management approach is to mitigate the consequences to tolerable levels irrespective of the calculated	• Exhibit B-28, RCIA IR2 31.2.
	probabilities of the triggering event.	• Exhibit B-32, BCOAPO IR2 2.3.
		• Exhibit B-39, BCUC Panel IR1 4.1.
11.	The existing Base Plant is well-over its average service life and needs to be replaced.	• Exhibit B-15, BCUC IR1 16.21, 16.22 and 40.1.
		• Exhibit B-22, RCIA IR1 18.1.
12.	12. The BCUC directed FEI to describe the utility's plans to address	BCUC Letter L-31-20, dated June 5, 2020.
resiliency in the short, medium and long terms in re 2018 T-South Incident.	resiliency in the short, medium and long terms in response to the 2018 T-South Incident.	 Exhibit B-1-4, Application, Appendix C (2020/21 ACP Compliance Report).
Des	cription of Alternatives	
13.	The optimal resiliency portfolio should align with the optimal gas	Exhibit B-4, Workshop Presentation, slide 35.
	supply portfolio.	• Transcript Vol 1, p. 30, l. 7 to p. 31, l. 18.
		• Transcript Vol 1, p. 164, l. 26 to p. 166, l. 1.
14.	On-system storage is the only dependable, practical and effective	• Exhibit B-1-4, Application, pp. 77-92.
	source of supply to avoid or reduce the impact of a widespread and prolonged outage as a result of a T-South no-flow event occurring during the winter	 Exhibit B-1-4, Application, Appendix A (Guidehouse Report on Natural Gas System Resiliency), p. 46.
	-	• Transcript Vol 1, p. 114, l. 11 to p. 115, l. 5.
		• Transcript Vol 1, p. 140, II. 16-21.



No.	Application Topic	Reference to Evidence
15.	Regional pipeline alternatives would not prevent a widespread outage in the Lower Mainland.	• Exhibit B-1-4, Application, pp. 84 to 92.
		Exhibit B-4, Workshop Presentation, slide 33.
		• Exhibit B-15, BCUC IR1 16.3, 16.5, 16.6 and 16.9.
		• Transcript Vol 1, p. 116, l. 6 to p. 117, l. 14.
		• Transcript Vol 1, p. 132, l. 23 to p. 133, l.2.
16.	There are significant advantages to locating on-system storage at the	• Exhibit B-15, BCUC IR1 16.18 and 24.3.
	existing Tilbury facility, relative to other locations.	• Exhibit B-24, Sentinel IR1 77.
		• Exhibit B-4, Workshop Presentation, slide 46.
		• Transcript Vol 1, p. 48, ll. 2-9.
		• Transcript Vol 1, p. 171, l. 25 to p. 172, l. 9.
		• Transcript Vol 1, p. 186, l. 23 to p. 188, l. 12.
17.	17. It is impractical and insufficient to add regasification capacity without	• Exhibit B-1-4, Application, pp. 57 and 65.
also replacing the Base Plant tank	also replacing the Base Plant tank.	• Exhibit B-26, BCUC IR2 78.1.
		• Transcript Vol 1, p. 18, ll. 13-16.
18.	FEI has properly sized the TLSE Project to prepare for, withstand and recover from a high impact event.	 Exhibit B-1-4, Application, Appendix A (Guidehouse Report on Natural Gas System Resiliency), pp. 49 to 50.
		• Exhibit B-5, Guidehouse Workshop Presentation, slide 21.
19.	Proposed 800 MMcf/d of regasification capacity provides resiliency, reliability and optionality, as well as being cost-effective.	 Exhibit B-1-4, Application, Appendix A, (Guidehouse Report on Natural Gas System Resiliency), p. 48.
		• Exhibit B-1-4, Application, pp. 116 to 118.
		• Exhibit B-15, BCUC IR1 19.3 to 19.6.
		• Transcript Vol 1, p. 18, l. 20 to p. 19, l. 1.



No.	Application Topic	Reference to Evidence
20.	A 3 Bcf tank is the best way to avoid or mitigate a widespread outage	• Exhibit B-1-4, Application, pp. 103 to 116.
	following a T-South no-flow event, and provides a variety of ancillary benefits unavailable with a smaller tank	• Exhibit B-4, Workshop Presentation, slides 39, 41.
		• Exhibit B-18, BCSEA IR1 4.1.
		• Transcript Vol 1, p. 182, l. 1 to p. 183, l. 15.
21.	Inherent economies of scale, as well as other environmental,	• Exhibit B-1-4, Application, pp. 99 to 100 and 107 to 108.
	reliability and operational benefits of a 3 Bcf storage tank are preferrable to keeping the existing Base Plant tank in-service.	• Exhibit B-15, BCUC IR1 16.21, 16.22 and 16.27.
		• Exhibit B-28, RCIA IR2 37.3.
22.	New facility sizing should account for gas supply benefits, as a larger facility can avoid the cost of acquiring other supply resources.	• Exhibit B-1-4, Application, Appendix C (2020/21 ACP Compliance Report), pp. 14 to 15.
		• Exhibit B-4, Workshop Presentation, slide 42.
		• Exhibit B-15, BCUC IR1 22.7, 46.1 and 46.2.
		 Transcript Vol 1, p. 182, l. 1 to p. 183, l. 15 (preliminary \$30 million/year avoided cost estimate).
23.	TLSE Project enables a controlled shutdown if a no-flow event(s) exceeds the capacity of the storage tank.	• Exhibit B-26, BCUC IR2 88.1.2.
No.	FEI Evidence	Торіс
Pro	ject Description	
24.	The TLSE Project will meet or exceed all safety standards.	• Exhibit B-1-4, Application, pp. 121 to 133.
		• Exhibit B-15, BCUC IR1 25.1.
		• Exhibit B-18, BCSEA IR1 4.6.
Env	ironmental and Archaeological Impacts	
25.	The TLSE Project will be constructed on an existing brownfield site	• Exhibit B-1-4, Application, pp. 169 to 182, Appendices O and P.
	and the potential adverse environmental and archaeological impacts can be mitigated.	• Exhibit B-44, Rebuttal Evidence to TWN, p. 22.



No.	Application Topic	Reference to Evidence
26.	The components of the TLSE Project are subject to additional regulatory scrutiny through the environmental assessment process for the Tilbury Phase 2 LNG Expansion Project.	 Exhibit B-44, Rebuttal Evidence to TWN, pp. 5-13, 14 to 16 and 23 to 24.
Indi	genous and Stakeholder Engagement	
27.	FEI has engaged with stakeholders regarding developments at Tilbury since 2012 and continues to do so consistent with the requirements of the BCUC's CPCN Guidelines.	• Exhibit B-1-4, Application, pp. 183 to 196.
28.	Engagement activities regarding the TLSE Project have been synchronized with the environmental assessment process for the Tilbury Phase 2 LNG Expansion Project.	 Exhibit B-1-4, Application, pp. 184 to 185 and Appendix Q-2 (Engagement Plan).
		• Exhibit B-18, BCSEA IR1 3.4.
29.	Consultation and engagement with Indigenous groups has been meaningful, timely and sufficient consistent with the requirements of the BCUC's CPCN Guidelines, and will continue.	• Exhibit B-1-4, Application, pp. 196 to 205.
		• Exhibit B-15, BCUC IR1 58.1 and 58.2.
		• Exhibit B-26, BCUC IR1 102.1.
		 Exhibit B-44, Rebuttal Evidence to TWN, pp. 13 to 14, 18 and 19 to 20.
30.	Collaboration and partnership with the Musqueam Indian Band embodies the spirit of reconciliation and demonstrates FEI's commitment to robust engagement.	 Exhibit A2-1, Press Release - Musqueam Indian Band and FortisBC Holdings Inc. sign Tilbury LNG Projects Agreement.
		 Exhibit B-54, BCUC IR4 114 and 115 series, and Attachment 114.1.
		Exhibit C8-1, Musqueam Indian Band – Request to Intervene.
No.	FEI Evidence	Торіс
British Columbia Energy Objectives		
31.	The TLSE Project will encourage economic development and the creation and retention of jobs.	 Exhibit A2-1, Press Release - Musqueam Indian Band and FortisBC Holdings Inc. sign Tilbury LNG Projects Agreement.
		• Exhibit B-1-4, Application, pp. 206 to 207.



No.	Application Topic	Reference to Evidence
32.	The TLSE Project enables greater system resiliency and is not expected to contribute to greenhouse gas emissions.	• Exhibit B-15, BCUC IR1 63.1.
		• Exhibit B-30, BCSEA IR2 11.1.
		• Exhibit B-44, Rebuttal Evidence to TWN, p. 24.

Appendix B CONCORDANCE WITH BCUC ADJOURNMENT DECISION COMMENTARY



Appendix B Concordance with BCUC Adjournment Decision Commentary



- 1 This Appendix summarizes how FEI has addressed the BCUC's commentary in the TLSE Project CPCN Adjournment Decision.
- 2 Table B-1 addresses the BCUC's comments regarding further evidence needed to evaluate the TLSE Project. Table B-2 addresses
- 3 the BCUC's comments on the specific content of FEI's holistic resiliency plan.
- 4

BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Need for comprehensive Resiliency Plan	"The Panel considers resiliency objectives are best assessed on a holistic level by comparing various resiliency options and prioritizing and planning against various outage scenarios, and then developing a comprehensive resiliency plan. Ideally, this planning would be completed in the context of the development of a Long Term Gas Resource Plan (LTGRP). A robust resiliency plan should consider multiple credible threats to the FEI system, along with an assessment of the likelihood and consequence of each threat. Proposed solutions to mitigate those threats should consider the ability of a solution to mitigate one or more of the threats and a cost benefit analysis of that solution. The Panel considers that the assessment of resiliency through such a plan is needed before concurring with FEI that "storage is the only practical and effective way to bridge a winter no-flow event on the T-South system." (p. 12)	 The 2024 Resiliency Plan considers: 58 Assessed Vulnerabilities that have the potential to lead to material customer outages; and Quantitative consequence and probability-based risk analysis based on engineering analysis and system modelling. (See 2024 Resiliency Plan, Section 5) FEI has considered a number of Supplemental Alternatives to mitigate the risks of a winter no-flow event on T-South to varying degrees. FEI is not proposing to address any other Assessed Vulnerabilities at this time. (See Supplemental Evidence, Section 4)

Table B-1: How and Where FEI Addresses the Adjournment Decision



BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Consequence of T- South no-flow event	"However, we do find that the consequence of a no-flow event is proportional to the duration of the no-flow event. Further, the consequence is higher during colder months. A five day no-flow event coinciding with a particularly cold spell in January has significantly greater consequence than a one day no-flow event in July. Rupture location and availability of alternative supply options also affect the consequence of a no-flow event, but we have no evidence of the specific risk related to these factors." (p. 11)	 For each of the 58 Assessed Vulnerabilities: FEI has modelled system pressure and impacts at multiple temperatures; FEI's system modelling assumes that all available sources of alternative supply are being used and that non-firm interruptible load has already been curtailed; Exponent's quantitative risk analysis is based on a three-month winter period only, and uses average Jan-Feb temperatures for the relevant areas (e.g., +4°C for the Lower Mainland); and Exponent has considered the potential duration of a no-flow event given the vulnerability-specific causes of failure it identified (i.e., hazards). Exponent included those considerations in its Monte Carlo analysis.
Probability of T-South no flow event (time of year)	"Further, there is no evidence in the proceeding that a rupture is more likely to occur at any particular time of the year or in any season or within a specific location on the T-South System, The Panel also considers that it may well be the case that the probability of a no-flow event caused by weather risk is somewhat elevated in the winter, but there is no evidence on the record to support that conjecture." (p. 9)	Exponent has reviewed the hazards specific to each Assessed Vulnerability and determined which could occur during the winter period. Exponent has not assumed that integrity related rupture risk is elevated in winter, as most hazards considered in Exponent's analysis are not seasonal. (See 2024 Resiliency Plan, Section 5 and Appendix RP 2, Exponent Report, Section 4.6 (paras. 126-127))



APPENDIX B CONCORDANCE WITH BCUC ADJOURNMENT DECISION COMMENTARY

BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Probability of T-South no flow event (time horizon)	"An additional consideration is the time scale over which the risk of a no-flow event is considered. FEI states that over the service life of the TLSE Project, a multi-day no-flow event is likely. While we would not go so far as to characterize a no-flow event as likely, we do agree that the longer the service life the greater the probability of a no-flow event – it is a truism that as the period under examination increases, the probability of the event happening during the period increases towards 100 percent. Conversely, the shorter the service life, the less likely the occurrence of a no-flow event."	 FEI continues to believe that the 67-year expected life of the TLSE Project is an appropriate time scale for assessing risk and rate impacts. FEI has provided additional evidence in support of this view including: The long-term usefulness of the facility for providing gas supply and resiliency to FEI customers based on adverse load loss sensitivities; and The mitigation revenues that could
		potentially be achieved to the extent that FEI no longer requires the full facility for its own gas supply or resiliency purposes.
		(See Supplemental Evidence, Sections 4.5.5, 4.7; Appendix F, Raymond Mason Report, pp. 36-53)
		FEI asked Exponent to calculate cumulative probabilities over a 23-year time scale. The cumulative probability of a winter no-flow event on T-South is still significant over 23 years.
		(See Supplemental Evidence, Section 4.7.1; 2024 Resiliency Plan, Section 5)


BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Duration of T-South no flow event (Residual risk with TLSE Project)	"The Panel recognizes that the proposed TLSE Project would improve resiliency, but only in certain circumstances. It will not mitigate all resiliency risks, and indeed, FEI may never be able to mitigate all resiliency risks.	The TLSE Project significantly reduces the risk associated with a T-South no-flow event, both in terms of: (i) reducing probability of a customer outage; and (ii) the consequences
	There may be circumstances where FEI still needs to initiate a widespread controlled shutdown, even with the TLSE Project in place – for example, a no-flow event lasting longer than 3 days in winter." (p. 16)	of any customer outage (customers lost, customer-outage-days, economic, safety risks associated with uncontrolled depressurization).
	"nor is there conclusive evidence of the duration of the ensuing no-	(See Supplemental Evidence, Section 4.7)
	flow event. However, in FEI's view there would be a higher likelihood of inclement weather or snow making access to a rupture site more challenging, and, therefore, increasing the time to investigate, repair, and determine if and when service on one or both pipelines could be restored." (p. 9)	Exponent considered the potential duration of a no-flow event, having regard to the potential hazards it identified for each Assessed Vulnerability. Exponent also prepared a Monte Carlo analysis to account for the potential for different no flow event durations
	"JANA's evidence is that 27 out of 30 reported pipeline ruptures and 22 of 23 ignited ruptures resulted in an outage of more than three days. Based on this evidence we find an event of more than three days more likely than an event of 3 days or less. In this regard, we also note FEI and JANA's submissions that even a precautionary shutdown of one of the two pipelines due to a rupture in the other pipeline would likely result in the adjacent line being out for a period of two days or longer [Emphasis added]." (p. 9)	different no-flow event durations. (See 2024 Resiliency Plan, Section 3.4.1.2.1 and Appendix RP 2, Exponent Report, Section 6.5)
	"Further, as discussed in Section 2.1 of this Decision, we have no evidence concerning the specific probabilities associated with outage duration, time of year or rupture location." (p. 11)	
Other causes of Lower Mainland outage	"Additionally, uncontrolled shutdowns could still potentially occur, for instance if there was an earthquake that ruptured the pipeline in Delta running to/from Tilbury. This underlines the importance of developing different use cases representing catastrophic failures and approaches to mitigating those failures identified and the cost of those approaches considered in order to properly assess resiliency needs." (p. 16)	The 2024 Resiliency Plan evaluated 58 Assessed Vulnerabilities, including a number in the Lower Mainland. The TLSE Project addresses the Assessed Vulnerabilities determined by Exponent to give rise to the largest risks.
		(See Resiliency Plan, Section 5 and Appendix RP 2, Exponent Report, Section 7.3)



BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Future Developments (Impact on Probability and Consequence of T-South no flow	"The Panel agrees with FEI that there is a potential for a multi day no- flow event on the T-South System. The T-South Incident in 2018 demonstrates this potential and also illustrates some of the potential consequences of such an event.	FEI has considered hypothetical load loss sensitivities to demonstrate the continued usefulness of the TLSE Project in the event of significant future load loss.
event)	There is, however, uncertainty whether the risk of no-flow events on the T-South System may increase or decrease in the future. Factors that could contribute to increasing the probability of a no-flow event include the aging of the T-south pipeline, increased severity or extreme weather events and the potential for increased cyber and physical security incursions. With regards to extreme weather events, we note the significant physical exposure the pipeline faced during the floods of 2021 when water erosion left significant portions of the pipeline exposed – although that did not lead to a no-flow event. On the other hand, there are factors that reduce the probability of a no-flow event, including the development of enhanced integrity management practices and technology and improved cyber security practices and potential actions taken by the utility to replace aging sections of the pipeline." (pp. 7-8)	 (See Supplemental Evidence, Section 4.5.5) The 2024 Resiliency Plan presents FEI's assessment of future developments on probability (e.g., climate change increasing potential of no flow events). (See Resiliency Plan, Section 6). The Federal Government's assessment of increasing cyber risk for utilities is addressed in the 2024 Resiliency Plan, Section 5.2.
Ancillary (supply) benefits	"FEI's analysis of ancillary benefits, as discussed further in Section 3.3 of the Decision, does not assess the ability of the TLSE Project to replace the functions or capacity currently provided by the Base Plant with regards to peaking supply." (p. 14)	The TLSE Project replaces the current functions of the Base Plant, and provides additional flexibility to use the facility to avoid investments in other gas supply resources. FEI assessed the financial value for customers in terms of avoided supply costs of LNG located in the Lower Mainland. This assessment shows that the avoided supply cost benefit is much higher than originally estimated. (See Supplemental Evidence, Section 4.5.4.1.2)



APPENDIX B CONCORDANCE WITH BCUC ADJOURNMENT DECISION COMMENTARY

BCUC Commentary Topic	TLSE Adjournment Decision Commentary	Where Commentary is Addressed
Other Potential Alternatives to TLSE Project	The BCUC identified various potential options that could potentially provide differing levels of mitigation. (pp. 26-33)	FEI has evaluated various Supplemental Alternatives, including all of the alternatives identified by the BCUC.
		(See Supplemental Evidence, Section 4 and Appendix C)
Stranding risk of the TLSE Project	"We acknowledge the difficulty of navigating a path to clean gas given these new technologies and business practices that must be considered. However, we share the Commercial Energy Consumers Association of British Columbia's concerns that "a higher level of confidence in terms of the risk being assessed and the expected life for the assets to be used and useful" is necessary to assess whether further resiliency investments are in the public convenience and necessity. In light of the current uncertainty with respect to the continued role of the natural gas system in British Columbia, we find insufficient evidence to conclude that the risk of stranding of the Project is acceptable, especially considering its expected life. FEI is invited to file further evidence that addresses the Panel's concerns about the stranding risk of the TLSE Project." (p. ii)	FEI has considered hypothetical load loss sensitivities to demonstrate the continued usefulness of the TLSE Project in the event of significant future load loss. (See Supplemental Evidence, Section 4.5.5)

1

2

Table B-2: How and Where FEI Addresses BCUC Commentary on Resiliency Plan Content

BCUC Commentary	BCUC Decision Commentary	How and Where BCUC Commentary is
Topic	(Page references are to Adjournment Decision unless indicated.)	Addressed
Loss of resiliency from PGR Project	"The Panel is concerned with the loss of resiliency [from PGR Project] and FEI's lack of a firm plan to replace the lost resiliency. Given this apparent existing non-systemic approach to addressing resiliency of its system, the Panel considers it necessary for FEI to address resiliency in a more comprehensive and holistic manner." (PGR Decision, p. 9)	The 2024 Resiliency Plan provides a holistic risk assessment of FEI's 58 most material vulnerabilities. Based on that risk analysis, at this time FEI is not recommending a resiliency-driven project to address the loss of resiliency from the PGR Project. Resiliency would continue to be considered in the course of typical planning. (See 2024 Resiliency Plan, Section 4.3.2.1)



APPENDIX B CONCORDANCE WITH BCUC ADJOURNMENT DECISION COMMENTARY

BCUC Commentary Topic	BCUC Decision Commentary (Page references are to Adjournment Decision unless indicated.)	How and Where BCUC Commentary is Addressed
Identification of vulnerabilities	"What are the current and future threats to the resiliency of FEI's system in addition to the 3 day no-flow event identified in this Application?" (p. i)	FEI identified 87 system outage vulnerabilities through a holistic review. Exponent quantified the risk associated with the 58 Assessed Vulnerabilities.
		(See 2024 Resiliency Plan, Section 3)
		FEI also addresses future risk, considering factors that affect both probability and consequences. (See 2024 Resiliency Plan, Section 3.4)
Current risk assessment / resiliency gap analysis	"What assets provide resiliency in FEI's current system and what and where are the gaps in resiliency?" (p. i)	FEI identified 87 system outage vulnerabilities through a holistic review. Exponent's quantitative risk analysis of the 58 Assessed Vulnerabilities accounts for available supply assets, infrastructure (including AMI), and personnel. (See 2024 Resiliency Plan, Section 3.6)





BCUC Commentary Topic	BCUC Decision Commentary (Page references are to Adjournment Decision unless indicated.)	How and Where BCUC Commentary is Addressed
Current risk assessment / gap analysis (consequences/prob ability)	"Typically, a probability/consequence analysis involves multiplying probabilities and consequences. However, in this instance a quantification of probability multiplied by consequence would likely be impossible, since there are a number of factors affecting consequence (e.g. outage duration, time of year, rupture location, availability of alternate supply options), some of which could be estimated but others of which are uncertain. Further, as discussed in Section 2.1 of this Decision, we have no evidence concerning the specific probabilities associated with outage duration, time of year or rupture location. However, we do find that the consequence of a no-flow event is proportional to the duration of the no-flow event. Further, the consequence is higher during colder months. A five day no-flow event coinciding with a particularly cold spell in January has significantly greater consequence than a one day no-flow event in July. Rupture location and availability of alternative supply options also affect the consequence of a no-flow event, but we have no evidence of the specific risk related to these factors." (p. 11)	 Exponent has quantified risk for all 58 of the Assessed Vulnerabilities based on probability x consequence. The analysis addresses the variables that the BCUC identified, including: Exponent has accounted for a variety of modes of failure (hazards) as applicable; The risk analysis only accounts for a three-month winter period; The customer outage breadth and timing is based on hydraulic modelling that accounts for available alternative supply and uses average winter temperatures in the relevant area (e.g., +4°C for the Lower Mainland); Customer outage duration is based on the BCUC-approved System Preservation and Restoration Plan;¹ Exponent used a Monte Carlo analysis to address uncertainty in the duration of the no-flow event;
		(See 2024 Resiliency Plan and Appendix RP 2, Exponent Report, Sections 6 and 7)
Impact of potential future developments on risk	"The impact, if any, of the loss of contracted storage on resilience." (p. i)	The Plan provides a future risk assessment that includes discussion of the impact of losing market storage on system resilience.
		(See 2024 Resiliency Plan, Section 4.2)

¹ These assumptions were discussed extensively in FEI's Rebuttal Evidence to RCIA in this proceeding.



APPENDIX B CONCORDANCE WITH BCUC ADJOURNMENT DECISION COMMENTARY

1

BCUC Commentary Topic	BCUC Decision Commentary (Page references are to Adjournment Decision unless indicated.)	How and Where BCUC Commentary is Addressed
Impact of potential future developments on risk	"How do FEI's other planned projects address or mitigate these gaps – e.g. AMI, RGSD - and what is the relationship and extent of overlap between those planned projects and the TLSE Project?" (p. i)	The risk analysis contained in the 2024 Resiliency Plan assumes the AMI Project is in place.
		A Southern Crossing Pipeline expansion (e.g., RGSD) is considered as a TLSE Project alternative. FEI explains that RGSD on its own would not be able to prevent a system collapse in the Lower Mainland following a T- South no-flow event because FEI would not be able to access the supply quickly enough.
		Supplemental Evidence, Appendix C)
Options to address risks	"What steps can be taken to fill those gaps in the short, medium and long term and what are the costs associated with these options? This should include analysis of some of the alternatives discussed in the proceeding, including:	Based on the risk assessments of the 58 Assessed Vulnerabilities, at this time FEI is only recommending the TLSE Project. It addresses FEI's largest customer outage
	o Additional regasification and liquefaction at Tilbury;	(See 2024 Resiliency Plan, Section 5.1)
	o Assessment of the remaining life of the existing Base Plant" (p. i)	FEI evaluated additional alternatives to the TLSE Project, including all of those that the BCUC identified in the Adjournment Decision.
		(See Supplementary Evidence, Section 4 and Appendix C).

Appendix C
DETAILED ANALYSIS OF SUPPLEMENTAL ALTERNATIVES



Appendix C

Detailed Analysis of Supplemental Alternatives



Table of Contents

1.	Intro	oduction	6
2.	Des	cription of Alternatives and Summary of Results	8
	2.1	Alternative and Contingent Scenario Descriptions and Modelling Parameters	8
	2.2	Step 1 Results: Non-Tilbury Alternatives and "No Capital Upgrades" are Technically and Commercially Non-Viable	15
	2.3	Step 2 Results: Supplemental Alternatives 2, 3, 5, 6, and 7 Do Not Retain FEI's Existing On-System Firm Peaking Gas Supply	15
	2.4	Step 3 Results for Supplemental Alternatives 2 to 9	16
	2.5	Resiliency Benefit Assessment Summary	19
		2.5.1 Results of Exponent's Risk Assessment for Technically and Commercially Viable Alternatives	19
		2.5.2 Understanding the Results of Exponent's Risk Mitigation Calculations	27
3.	Fina	Incial Impacts For All Supplemental Alternatives	33
	3.1	Financial Analysis Results	33
	3.2	Financial Analysis Assumptions: Base Cost Estimates	34
	3.3	Financial Analysis Assumptions: Peaking Supply Costs / Benefits Reflected in Levelized Total Rate Impact	34
4.	Dev Add	elopment of Planning and Contingent Modelling Scenarios to ress Adjournment Decision	37
	4.1	- Description of Planning and Contingent Scenarios	38
	4.2	Contingent Modelling Scenarios Are Not Dependable	38
		4.2.1 FEI Peaking Gas Supply Requirements Exceed 0.6 Bcf in a Typical Winter	38
		4.2.2 Availability of Tilbury 1A Volumes on a Given Day Will Depend on Overall RS 46 Demand and Use Patterns During a Year	39
		4.2.3 Changing How FEI Uses Liquefaction Would Not De-Risk Relying on Non- Dependable Volumes for Resiliency	42
	4.3	A Full Tilbury 1A Tank Would Not Materially Reduce Risk Exposure to Winter No- Flow Event	43
5.	Deta Alte	ailed Information Supporting Assessment of the Supplemental rnatives	45
	5.1	Alternatives Relying on Existing Facilities	45
		5.1.1 Supplemental Alternative 1	45
		5.1.2 Supplemental Alternative 2	52



	5.1.3 Supplemental Alternative 3	. 64
5.2	Alternatives With New Facility that Maintains Gas Supply But No Resiliency Reserve	. 75
	5.2.1 Supplemental Alternative 4	. 75
	5.2.2 Supplemental Alternative 4A	. 80
5.3	Alternatives With New Facility that Provides Resiliency Reserve but No Gas Supply	. 86
	5.3.1 Supplemental Alternative 5	. 87
	5.3.2 Supplemental Alternative 6	. 92
	5.3.3 Supplemental Alternative 7	. 97
5.4	Alternatives with New Facility that Provides Both Resiliency Reserve and Gas Supply	103
	5.4.1 Supplemental Alternative 8	104
	5.4.2 Supplemental Alternative 9 (Preferred Alternative)	110
5.5	Other Non-Tilbury Alternatives Identified by the BCUC	112
	5.5.1 Supplemental Alternative 10	112
	5.5.2 Supplemental Alternative 11	115
	5.5.3 Supplemental Alternative 12	119
5.6	Southern Crossing Pipeline Extension / RGSD Project	122
	5.6.1 The Lower Mainland Would Lose Service Before FEI Could Get Gas from a Southern Crossing Pipeline Extension	123
	5.6.2 Tilbury LNG Provides Peak Capacity, Whereas Pipeline Provides Year-Round Energy	124
	5.6.3 FEI Is No Longer Pursuing RGSD on its Own	124



List of Appendices

Appendix C-1 Floating LNG Storage Concept Analysis

Index of Tables and Figures

Table C-1: Supplemental Alternatives and Descriptions	.9
Table C-2: Summary of Non-Viable Alternatives1	15
Table C-3: Summary of Step 2 Screen Results1	16
Table C-4: Supplemental Alternatives that Materially Improve Resiliency at Average	
Winter Temperatures (+4°C)2	20
Table C-5: Costs and Rate Impacts of the Supplemental Alternatives	33
Table C-6: Avoided Gas Supply Market Resources Included in Financial Analysis	36
Table C-7: Supplemental Alternative 1 Planning and Contingent Modelling Scenarios4	17
Table C-8: Supplemental Alternative 1 Load Support Durations Under Different Winter	
Conditions	19
Table C-9: Supplemental Alternative 2 Planning and Contingent Modelling Scenarios	55
Table C-10: Supplemental Alternative 2 Load Support Durations Under Different Winter	
Conditions	57
Table C-11: Supplemental Alternative 3 Planning and Contingent Modelling Scenarios	36
Table C-12: Supplemental Alternative 3 Load Support Durations Under Different Winter	
Conditions	38
Table C-13: Supplemental Alternative 4 Planning and Contingent Modelling Scenario	76
Table C-14: Supplemental Alternative 4 Load Support Durations Under Different Winter	
Conditions	78
Table C-15: Supplemental Alternative 4A Planning and Contingent Modelling Scenario	31
Table C-16: Supplemental Alternative 4A Load Support Durations Under Different Winter	
Conditions	34
Table C-17: Supplemental Alternative 5 Planning and Contingent Modelling Scenarios	38
Table C-18: Supplemental Alternative 5 Load Support Durations Under Different Winter	
Conditions) 0
Table C-19: Supplemental Alternative 6 Planning Modelling Scenarios) 3
Table C-20: Supplemental Alternative 6 Load Support Durations Under Different Winter	
Conditions) 4
I able C-21: Supplemental Alternative 7 Planning Modelling Scenarios	98





Table C-22:	Supplemental Alternative 7 Load Support Durations Under Different Winter	100
	Conditions	100
Table C-23:	Alternative 8 Planning Modelling Scenarios	104
Table C-24:	Supplemental Alternative 8 Load Support Durations Under Different Winter	
	Conditions	106
Table C-25:	Supplemental Alternative 9 Planning Modelling Scenarios	111
Table C-26:	Comparison of Existing VITS to Upgraded VITS	114
Table C-27:	Offshore LNG Storage Viability Matrix	121

Figure C-1: Structured Process to Identify the Preferred Alternative	6
Figure C-2: Scoring Results for Technically and Commercially Viable Supplemental	
Alternatives	18
Figure C-3: T-South at Avg. Winter – Expected Annual Loss Reduction	22
Figure C-4: T-South at Avg. Winter – Expected 23-Year Loss Reduction	23
Figure C-5: T-South at Avg. Winter – Expected 67-Year Loss Reduction	24
Figure C-6: All AVs at Avg. Winter – Expected Annual Loss Reduction	25
Figure C-7: All AVs at Avg. Winter – Expected 23-Year Loss Reduction	26
Figure C-8: All AVs at Avg. Winter – Expected 67-Year Loss Reduction	27
Figure C-9: Duration of Lower Mainland Load Support Provided by Supplemental	
Alternatives	31
Figure C-10: T1A Tank Levels	42
Figure C-11: T-South at Avg. Winter – Expected Annual Loss Reduction	44
Figure C-12: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 1	50
Figure C-13: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 1	50
Figure C-14: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 1	51
Figure C-15: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 2 – 0.35bcf & 400MMcf/d	59
Figure C-16: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 2 – 0.35bcf & 400MMcf/d	60
Figure C-17: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 2 – 0.35bcf & 400MMcf/d	60
Figure C-18: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 2 – 0.75bcf & 400MMcf/d	61
Figure C-19: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 2 – 0.75bcf & 400MMcf/d	62
Figure C-20: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 2 – 0.75bcf & 400MMcf/d	62



Figure C-21: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	70
	70
Figure C-22: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 3 – Contingent 0.35bcf & 600MMcf/d	71
Figure C-23: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 3 – Contingent 0.35bcf & 600MMcf/d	71
Figure C-24: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 3 – Contingent 0.75bcf & 600MMcf/d	73
Figure C-25: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 3 – Contingent 0.75bcf & 600MMcf/d	73
Figure C-26: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 3 – Contingent 0.75bcf & 600MMcf/d	74
Figure C-27: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 6 – 1bcf & 800MMcf/d	95
Figure C-28: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 6 – 1bcf & 800MMcf/d	96
Figure C-29: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp.	
Alt. 6 – 1bcf & 800MMcf/d	96
Figure C-30: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp.	
Alt. 7 – 2 Bcf & 800MMcf/d	101
Figure C-31: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt. 7 – 2 Bcf & 800MMcf/d	102
Figure C-32: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supply	
Alt. 7 – 2 Bcf & 800MMcf/d	102
Figure C-33 [°] Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp	
Alt $8 - 1.4$ Bcf & 800MMcf/d	107
Figure C-34: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with	
Supp. Alt $8 = 1.4$ Bcf & 800MMcf/d	108
Figure C-35: Impact to Lower Mainland due to Loss of T-South Supply at -10° C with Suppl	
Alt 9 1 4 Bof 8 900MMof/d	100



1 **1. INTRODUCTION**

- 2 This Appendix provides supporting detail for the alternatives analysis described in Section 4 of
- 3 the Supplemental Evidence. FEI used the information in this Appendix throughout the structured
- 4 three-step alternatives analysis process described in Section 4.2.2 of the Supplemental Evidence
- 5 and depicted in Figure C-1 below.

6 In order to ensure that the BCUC has complete information, this Appendix includes a full analysis

- 7 and Step 3 scoring for all Supplemental Alternatives that are technically and commercially viable
- 8 (i.e., passed Step 1), even if they fail the Step 2 screen by jeopardizing FEI's ability to maintain
- 9 service in normal operations. As such, Supplemental Alternatives 2, 3, 4, 4A, 5, 6, 7, 8 and 9 were
- 10 scored relative to the existing Tilbury Base Plant with no capital upgrades (Supplemental
- 11 Alternative 1), having regard to resiliency, gas supply, age-related Base Plant challenges,
- 12 levelized total rate impact, and future use. This analysis confirms that Supplemental Alternative 9
- 13 provides superior overall customer value and is the Preferred Alternative.



- 16 Appendix C is organized as follows:
- 17 Section 2 - Description of Alternatives and Summary of Results: This section 18 provides more detailed alternative and contingent scenario descriptions and modelling parameters. It also summarizes the Step 1 and Step 2 screening results, and the Step 3 19 20 scoring results for all Supplemental Alternatives that are technically and commercially 21 viable. FEI evaluated every alternative that passed the Step 1 viability screening, despite 22 Supplemental Alternatives 2, 3, 5, 6, and 7 risking FEI's existing ability to serve firm 23 customers in normal operations. Expanding the scoring to include other alternatives that 24 failed the Step 2 screen only reinforces that Supplemental Alternative 9 is the preferred 25 alternative.
- Section 3 Financial Impacts for All Supplemental Alternatives: This section provides
 the financial impacts for Supplemental Alternatives 1 to 9 and the contingent scenarios,
 as well as additional information regarding the associated inputs. The financial impacts



- are based on updated capital cost estimates and consistent financial assumptions
 regarding gas supply requirements, availability in the market and associated costs.
- Section 4 Development of Planning and Contingent Scenarios to Address BCUC
 Commentary: FEI developed both "planning" and "contingent" scenarios for certain
 Supplemental Alternatives.
- 6 o The planning scenarios assume dependable LNG supply for resiliency, predicated on setting aside a "resiliency reserve".
- The contingent scenarios are included to address commentary from the BCUC in the Adjournment Decision. These contingent scenarios are sensitivities that assume a specified amount of non-dependable LNG (i.e., LNG allocated for gas supply or LNG sales) is present on the day of a winter T-South no-flow event.
- 12 The planning scenarios, with their focus on dependable resiliency reserves, reflect typical 13 utility planning principles that FEI believes should be applied. Proceeding on the 14 expectation that unreserved LNG volumes would *necessarily* be present at Tilbury and 15 available for resiliency is risky as these volumes are used for other purposes in normal 16 operations. However, Exponent's analysis confirms that none of the contingent scenarios 17 would materially improve FEI's ability to withstand a winter T-South no-flow event.
- Section 5 Detailed Information Supporting Assessment of the Supplemental Alternatives: FEI provides an alternative-by-alternative detailed discussion for every Supplemental Alternative, including those located at Tilbury, non-Tilbury alternatives raised by the BCUC in the Adjournment Decision, as well as a Southern Crossing Pipeline (SCP) extension.



1 2. DESCRIPTION OF ALTERNATIVES AND SUMMARY OF RESULTS

This section provides more detailed alternative and contingent scenario descriptions and
modelling parameters. It also summarizes the Step 1 and Step 2 screening results, and the Step
3 scoring results for all Supplemental Alternatives that are technically and commercially viable.

5 2.1 ALTERNATIVE AND CONTINGENT SCENARIO DESCRIPTIONS AND MODELLING 6 PARAMETERS

The following table sets out the names and descriptions for all 13 Supplemental Alternatives, plus
the "contingent" scenarios for Supplemental Alternatives 1, 2, 3, 4, 4A, and 5.

9 The base (i.e., non-contingent) alternatives reflect planning principles, meaning they reflect 10 dependable resources that are allocated for specific purposes. In contrast, the contingent 11 scenarios for a given alternative consider the case where some volume of LNG that is not set 12 aside as a "resiliency reserve" is available on the day of a no-flow event. In other words, some 13 volume of LNG from Tilbury 1A or the Base Plant (as applicable) that was intended to be used for Rate Schedule (RS) 46 sales or gas supply remains in the tank on the day of the no-flow event 14 15 and is used for resiliency instead of its intended purpose. FEI does not endorse using a contingent 16 approach for planning purposes because: (a) FEI cannot rely on these LNG volumes being 17 present on the day of a winter T-South no-flow event; and (b) the use of peaking supply for 18 resiliency leaves FEI very exposed to firm load curtailments during a subsequent typical cold 19 weather event. FEI has included these contingent scenarios to address the BCUC's commentary 20 in the Adjournment Decision about the potential resiliency value of such volumes. The concept of 21 the "planning" and "contingent" scenarios is discussed further in Section 4 of this Appendix.



Supp. Alt. #	Name	Description				
		Alternatives Reliant on Existing Facilities ¹				
Alt 1	No Capital	Run the Base Plant until it is no longer usable with no resiliency reserve.				
	Upgrades with Optimized Liquefaction (No Resiliency Reserve)	For peaking supply, FEI would continue to rely on the existing, non-refurbished Base Plant (i.e., 150 MMcf/d regasification and reduced tank capabilities of 0.35 Bcf) which has reached its end of life, augmented by 0.25 Bcf from Tilbury 1A to restore the original Base Plant design capabilities reflected in the Annual Contracting Plan (ACP). ² FEI would need to continue relying on 50 MMcf/d of year-round pipeline capacity to achieve the required peaking supply (1 Bcf and 200 MMcf/d ³).				
		This option would result in FEI ultimately losing its LNG peaking supply (150 MMcf/d and 0.6 Bcf), and dependable peaking supply on regional infrastructure would be unavailable. Firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d. FEI considers 2030 to be a reasonable estimate for when this period of curtailment would begin. Although it is uncertain if and when this period of curtailment would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.				
		Contingent Scenarios: FEI also considered two contingent scenarios for Supplemental Alternative 1 wherein LNG volume from the Base Plant (0.35 Bcf) (Alternative 1 Contingent) and Tilbury 1A (0.4 Bcf) ⁴ (Alternative 1 Contingent with T1A) is present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose of gas supply and RS 46 sales, respectively. This would leave FEI without peaking supply for the rest of the winter, thus exposing FEI customers to curtailments in normal operations.				

Table C-1: Supplemental Alternatives and Descriptions

¹ These alternatives include prolonged reliance on the Base Plant tank with no dependable resiliency reserve, declining reliability, and a high likelihood of relying on the market for some replacement gas supply.

² As discussed in Section 3.3.2.1 of the Supplemental Evidence, the Base Plant has been operating at a reduced tank capacity of 0.35 Bcf for seismic reasons and has been relying on 0.25 Bcf of peaking energy from Tilbury 1A to restore the ACP requirement. However, FEI does not expect to be able to continue relying on the 0.25 Bcf from Tilbury 1A, as discussed in Section 4.4.1 of the Supplemental Evidence.

³ The optimized gas supply portfolio is discussed in Section 3.3.4.1 of the Supplemental Evidence.

⁴ Section 4.2.2 in this Appendix discusses the basis for using 0.4 Bcf as a contingent volume in Tilbury 1A.



Supp. Alt. #	Name	Description
Alt 2	New Regasification Only – 400 MMcf/d (No Resiliency Reserve)	Replace the Base Plant regasification with 400 MMcf/d of new capacity, but continue to rely on a non-refurbished Base Plant tank until it is no longer usable. There is no resiliency reserve.
		For peaking supply, FEI would continue to rely on the existing, non-refurbished Base Plant (i.e., reduced tank capabilities of 0.35 Bcf) which has reached its end of life, augmented by 0.25 Bcf from Tilbury 1A to restore the original Base Plant design capabilities reflected in the ACP. ⁵ Despite the new regasification of 400 MMcf/d, FEI would need to continue relying on 50 MMcf/d of year-round pipeline capacity to achieve the required peaking supply (1 Bcf and 200 MMcf/d) due to insufficient storage.
		This option would result in FEI ultimately having insufficient peaking supply, and firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d. FEI considers 2030 to be a reasonable estimate for when this period of curtailment would begin; the new regasification will be of limited use after that point given there would be insufficient available LNG in storage at Tilbury beyond 2030. Dependable peaking supply on regional infrastructure would be unavailable. Although it is uncertain if and when this period of curtailments would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.
		Contingent Scenarios: FEI also considered two contingent scenarios for Supplemental Alternative 2 wherein LNG volume from the Base Plant (0.35 Bcf) (Alternative 2 Contingent) and Tilbury 1A (0.4 Bcf) (Alternative 2 Contingent with T1A) is present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose of gas supply and RS 46 sales respectively. This would leave FEI without peaking supply for the rest of the winter, thus exposing FEI customers to curtailments in normal operations.

⁵ As discussed in Section 3.3.2.1 of the Supplemental Evidence, the Base Plant has been operating at a reduced tank capacity of 0.35 Bcf for seismic reasons and has been relying on 0.25 Bcf of peaking energy from Tilbury 1A to restore the ACP requirement. However, FEI does not expect to be able to continue relying on the 0.25 Bcf from Tilbury 1A, as discussed in Section 4.4.1 of the Supplemental Evidence.



Supp. Alt. #	Name	Description		
Alt 3	New Regasification Only - 600 MMcf/d (No Resiliency Reserve)	New Regasification Only - 600	New Regasification Only - 600	Replace Base Plant regasification with 600 MMcf/d of new capacity but continue to rely on a non-refurbished Base Plant tank until it is no longer usable. There is no resiliency reserve.
		For peaking supply, FEI would continue to rely on the existing, non-refurbished Base Plant (i.e., reduced tank capabilities of 0.35 Bcf) which has reached its end of life, augmented by 0.25 Bcf from Tilbury 1A to restore the original Base Plant design capabilities reflected in the ACP. ⁶ Despite the new regasification of 600 MMcf/d, FEI would need to continue relying on 50 MMcf/d of year-round pipeline capacity to achieve the required peaking supply (1 Bcf and 200 MMcf/d) due to insufficient storage.		
		This option would result in FEI ultimately having insufficient peaking supply, and firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d. FEI considers 2030 to be a reasonable estimate for when this period of curtailments would begin; the new regasification will be of limited use after that point given there would be insufficient available LNG in storage at Tilbury beyond 2030. FEI would be unable to procure dependable replacement peaking supply on regional infrastructure. Although it is uncertain if and when this period of curtailment would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.		
		Contingent Scenarios: FEI also considered two contingent scenarios for Supplemental Alternative 3 wherein LNG volume from the Base Plant (0.35 Bcf) (Alternative 3 Contingent) and Tilbury 1A (0.4 Bcf) (Alternative 3 Contingent with T1A) is present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose of gas supply and RS 46 sales respectively. This would leave FEI without peaking supply for the rest of the winter, thus exposing FEI customers to curtailments in normal operations.		

⁶ As discussed in Section 3.3.2.1 of the Supplemental Evidence, the Base Plant has been operating at a reduced tank capacity of 0.35 Bcf for seismic reasons and has been relying on 0.25 Bcf of peaking energy from Tilbury 1A to restore the ACP requirement. However, FEI does not expect to be able to continue relying on the 0.25 Bcf from Tilbury 1A, as discussed in Section 4.4.1 of the Supplemental Evidence.

SECTION 2: DESCRIPTION OF ALTERNATIVES AND SUMMARY OF RESULTS



Supp. Alt. #	Name	Description						
	New Facility with Gas Supply But No Resiliency Reserve ⁷							
Alt 4	Like-for-Like (No Resiliency Reserve)	Replace the Base Plant like-for-like to restore the 1971 design capacity (150 MMcf/d regasification and 0.6 Bcf tank) and continue using Tilbury as a supply peaking resource, without a resiliency reserve.						
		FEI would continue to rely on 50 MMcf/d of year-round pipeline capacity to achieve required peaking supply (1 Bcf and 200 MMcf/d), which is suboptimal from a portfolio design standpoint.						
		Contingent Scenario: FEI also considered a contingent scenario for Supplemental Alternative 4 wherein 0.6 Bcf of LNG volume is present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose for gas supply. This would leave FEI without peaking supply for the rest of the winter, thus exposing FEI customers to curtailments in normal operations.						
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification	Replace the Base Plant with the smallest new facility capable of providing FEI's optimum peaking gas supply, including a 1 Bcf tank and 200 MMcf/d regasification (with an additional 200 MMcf/d for redundancy). Continue using it as a supply peaking resource, without a resiliency reserve.						
		The facility provides sufficient capability to meet FEI's required peaking supply and therefore does not require FEI to augment with regional market resources. This is optimal from a gas supply portfolio design perspective.						
		Contingent Scenario: FEI also considered a contingent scenario for Supplemental Alternative 4A wherein 1.0 Bcf of LNG volume is present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose for gas supply. This would leave FEI without peaking supply for the rest of the winter, thus exposing FEI customers to curtailments in normal operations.						

⁷ These alternatives do not include a dependable resiliency reserve but provide different amounts of peaking gas supply and improved reliability.



Supp. Alt. #	Name	Description					
	N	ew Facility with Resiliency Reserve But No Gas Supply ⁸					
Alt 5	Like-for-Like (Full Resiliency Reserve)	Replace the Base Plant like-for-like to restore the 1971 design capacity (150 MMcf/d regasification and 0.6 Bcf tank) and allocate the entire tank as a resiliency reserve.					
		FEI would continue to hold the existing 50 MMcf/d of sub-optimal year-round pipeline capacity for peaking supply. However, this option would result in FEI losing its LNG peaking supply (150 MMcf/d and 0.6 Bcf) upon commissioning of the new facility, and firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d due to an inability to procure dependable replacement peaking supply on regional infrastructure. Although it is uncertain if and when this period of curtailment would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.					
		Contingent Scenario: FEI also considered a contingent scenario for Supplemental Alternative 5 wherein 0.4 Bcf of LNG volume from Tilbury 1A is also present on the day of a no-flow event and available for outage risk mitigation, despite its intended purpose of being set aside for RS 46 sales.					
Alt 6	New 1 Bcf Tank (Full	Replace the Base Plant with a 1.0 Bcf tank and 800 MMcf/d regasification. Allocate the entire tank as a resiliency reserve.					
Resiliency Reserve) and 800 MMcf/d Regasification		FEI would continue to hold the existing 50 MMcf/d of sub-optimal year-round pipeline capacity for peaking supply. However, this option would result in FEI losing its LNG peaking supply (150 MMcf/d and 0.6 Bcf) upon commissioning of the new facility, and firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d due to an inability to procure dependable replacement peaking supply on regional infrastructure. Although it is uncertain if and when this period of curtailment would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.					
Alt 7	New 2 Bcf Tank (Full	Replace the Base Plant with a 2.0 Bcf tank and 800 MMcf/d regasification. Allocate the entire tank as a resiliency reserve.					
	Reserve) and 800 MMcf/d Regasification	FEI would continue to hold the existing 50 MMcf/d of sub-optimal year-round pipeline capacity for peaking supply. However, this option would result in FEI losing its LNG peaking supply (150 MMcf/d and 0.6 Bcf) upon commissioning of the new facility, and firm customers would thereafter begin to be curtailed in normal operations by up to 150 MMcf/d due to an inability to procure dependable replacement peaking supply on regional infrastructure. Although it is uncertain if and when this period of curtailment would end, in order to provide a consistent basis for the financial analysis FEI assumed that the period of curtailment would end in 2035 with the construction of regional infrastructure upgrades.					

⁸ These alternatives include a full resiliency reserve but still rely on the market for replacement of the gas supply functions.



Supp. Alt. #	Name	Description
	New Facilit	ty with Both Resiliency Reserve and Replacement of Gas Supply
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct the smallest facility that allows FEI to both avoid curtailments of firm peak load in normal operations and provide a certain level of resiliency reserve. This includes replacing the Base Plant with a 2 Bcf tank and 800 MMcf/d regasification, of which 1.4 Bcf is allocated as a resiliency reserve, and 0.6 Bcf is allocated to replace the existing gas supply functions at Tilbury. FEI would continue to rely on 50 MMcf/d of year-round pipeline capacity to achieve required peaking supply (1 Bcf and 200 MMcf/d), which is suboptimal from a portfolio design standpoint.
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	Construct a facility that both significantly mitigates FEI's largest customer outage risks and meets FEI's required peaking gas supply in an optimal manner. Replace the Base Plant with a 3 Bcf tank and 800 MMcf/d regasification and allocate 2 Bcf as a resiliency reserve, and 1 Bcf to gas supply. The facility provides sufficient capability to meet FEI's required peaking supply and therefore does not require FEI to augment with regional market resources. This is optimal from a gas supply portfolio design perspective.
		Non-Tilbury Alternatives ⁹
Alt 10	Alt 1 plus VITS Reverse Flow	FEI would retain the existing Tilbury facilities with no capital upgrades (i.e., Supplemental Alternative 1). FEI would also construct the necessary facilities to allow significant reverse flows on the Vancouver Island Transmission System (VITS) at all times during the year, such that the combined daily delivery is at least 550 MMcf/d.
Alt 11	LNG from Woodfibre LNG	 Use the existing Tilbury facilities with no capital upgrades (i.e., Supplemental Alternative 1). FEI would also contract with Woodfibre LNG for a long-term firm supply of LNG and FEI would either: Custom build a vessel for transporting LNG up the Fraser River to Tilbury, construct a facility at Tilbury for offloading the LNG from the vessel, and add more regasification capacity at Tilbury to address the existing regasification constraint; or Acquire property rights from Woodfibre LNG on which FEI constructs a regasification facility at Woodfibre, plus FEI constructs facilities to permit reversing the flow of the VITS.
Alt 12	Floating LNG	Purchase a vessel to provide floating LNG storage. Acquire a water lot that would allow for permanent mooring. Add more regasification capacity, either as an integrated component of the LNG storage vessel or on the adjacent shoreline. Construct onshore facilities, including a jetty and interconnecting pipe.

⁹ FEI considers these alternatives to be non-viable approaches to providing winter resiliency and peaking gas supply.



1 2.2 STEP 1 RESULTS: NON-TILBURY ALTERNATIVES AND "NO CAPITAL 2 UPGRADES" ARE TECHNICALLY AND COMMERCIALLY NON-VIABLE

The BCUC stated in the Adjournment Decision that it required further information about the remaining life of the Base Plant, specific non-Tilbury alternatives (constructing facilities to permit reverse-flow on the VITS, LNG from Woodfibre LNG, and floating LNG storage), and a Southern Crossing Pipeline extension. The following table summarizes why each of these Supplemental Alternatives was found to be non-viable at Step 1. More detailed discussion of each alternative is provided in Section 5 of this Appendix.

9

Table C-2: Summary of Non-Viable Alternatives

No.	Description	Rationale
Alt 1	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)	Supplemental Alternative 1 is not technically viable because the Base Plant has reached end-of-life and can no longer reliably perform its intended function without capital upgrades. Please refer to Section 3.3 of the Supplemental Evidence for more details on the Base Plant.
Alt 10	Alt 1 plus VITS Reverse Flow	Supplemental Alternative 10 would have a significant scope of work compared to Tilbury-based alternatives that provide a similar resiliency benefit. For example, this alternative would involve looping significant portions of FEI's VITS and completing multiple compressor station upgrades. Further, a hydraulic constraint exists which limits the amount of reverse flow that is possible on the VITS during winter.
Alt 11	LNG from Woodfibre	Supplemental Alternative 11 would be inconsistent with Woodfibre LNG's business model and LNG markets. Any LNG storage that Woodfibre has on the site is inventoried to ensure the next customer vessel can be filled on schedule. Further, even if LNG from Woodfibre were available, the reverse flow constraint identified in Supplemental Alternative 10 applies to Supplemental Alternative 11 as well.
Alt 12	Floating LNG	There are no viable sites for floating LNG. All options also have issues with technical feasibility, either due to the tie-in point pressure rating and pipeline capacity or execution difficulties caused by the location of the tie-in point in the Fraser River.
n/a	Southern Crossing Extension (e.g., RGSD Project)	FEI has re-confirmed that a Southern Crossing Pipeline extension, regardless of size or end point, would not be able to provide supply fast enough to prevent a Lower Mainland outage on a similar scale to what would occur today following a winter T-South no-flow event. FEI notes that although a future Southern Crossing Pipeline extension remains possible, FEI is no longer pursuing RGSD as an FEI project.

10

2.3 STEP 2 RESULTS: SUPPLEMENTAL ALTERNATIVES 2, 3, 5, 6, AND 7 DO NOT *RETAIN FEI'S EXISTING ON-SYSTEM FIRM PEAKING GAS SUPPLY*

In Section 4.2.2.2 of the Supplemental Evidence, FEI outlines and explains the rationale for the
 Step 2 screen. Step 2 screens out supplemental alternatives that do not, at minimum, retain FEI's

15 existing on-system peaking resources of 0.6 Bcf and 150 MMcf/d. The intent is to ensure that FEI

16 remains able to serve firm load dependably in cold weather without curtailments. Based on this



- 1 Step 2 screen, Supplemental Alternatives 2, 3, 5, 6, and 7 were screened out. The results are
- 2 summarized in Table C-3 below. Please refer to Section 4.4 of the Supplemental Evidence and
- 3 Section 5 of this Appendix for more details.
- 4

Table C-3: Summary of Step 2 Screen Results

Supp Alt	Name	On-Systen Modelling	n Gas Supply Parameters ¹⁰	Retains FEI's Existing On-System Firm	
#		То 2030	After 2030	Peaking Gas Supply Capabilities?	
Alt 2	New Regasification Only - 400 MMcf/d (No Resiliency Reserve)	0.6 Bcf 400 MMcf/d	< 0.6 Bcf 400 MMcf/d	×	
Alt 3	New Regasification Only - 600 MMcf/d (No Resiliency Reserve)	0.6 Bcf 600 MMcf/d	< 0.6 Bcf 600 MMcf/d	×	
Alt 4	Like-for-Like (No Resiliency Reserve)	0.6 Bcf 150 MMcf/d	0.6 Bcf 150 MMcf/d	~	
Alt 4A	New 1 Bcf Tank (No Resiliency Reserve) and 400 MMcf/d Regasification	1 Bcf 400 MMcf/d	1 Bcf 400 MMcf/d	~	
Alt 5	Like-for-Like (Full Resiliency Reserve)	0 Bcf 150 MMcf/d	0 Bcf 150 MMcf/d	×	
Alt 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	0 Bcf 800 MMcf/d	0 Bcf 800 MMcf/d	×	
Alt 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	0 Bcf 800 MMcf/d	0 Bcf 800 MMcf/d	×	
Alt 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	0.6 Bcf 800 MMcf/d	0.6 Bcf 800 MMcf/d	~	
Alt 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	1 Bcf 800 MMcf/d	1 Bcf 800 MMcf/d	~	

5

6 2.4 STEP 3 RESULTS FOR SUPPLEMENTAL ALTERNATIVES 2 TO 9

In Section 4.2.2.3 of the Supplemental Evidence, FEI describes Step 3 of the structured alternative analysis process. As part of Step 3, FEI scored all alternatives relative to the existing Tilbury Base Plant with no capital upgrades (Supplemental Alternative 1) against five criteria. The discussion in Section 4 of the Supplemental Evidence focuses on how the criteria apply to Supplemental Alternatives 4, 4A, 8, and 9, since these are the only Supplemental Alternatives that passed both the Step 1 and Step 2 screens. However, as explained above, as part of this

¹⁰ The modelling parameters for each Supplemental Alternative are discussed in greater detail in Section 4 of this Appendix.



Appendix FEI also assessed and scored the five alternatives screened out at Step 2
 (Supplemental Alternatives 2, 3, 5, 6 and 7) to be fully responsive to the Adjournment Decision.

3 Figure C-2 below presents scoring results for all technically and commercially viable supplemental 4 alternatives (i.e., those that passed the Step 1 screening). In Figure C-2, Supplemental 5 Alternatives that did not pass the Step 2 screen are identified by a red filled "No Impact" score for 6 the "Availability of Dependable Gas Supply During Peak Demand" criterion. FEI has scored Step 7 3 on a relative basis; therefore, "No Impact" means that the option does not improve on the 8 adverse service implications of continuing to rely on the existing Tilbury Base Plant with no capital 9 upgrades (Supplemental Alternative 1). As Figure C-2 demonstrates, when the additional Supplemental Alternatives are considered, Supplemental Alternative 9 remains the Preferred 10 11 Alternative.

٠	1		

Figure C-2: Scoring Results for Technically and Commercially Viable Supplemental Alternatives										
Evaluation Criterion	Criterion Weighting	Alternative 2	Alternative 3	Alternative 4	Alternative 4A	Alternative 5	Alternative 6	Alternative 7	Alternative 8	Alternative 9
Resiliency Benefit	30%	No Impact	No Impact	No Impact	No Impact	No Impact	No Impact	High Positive Impact	Medium Positive Impact	High Positive Impact
Availability of Dependable Gas Supply During Peak Demand	20%	No Impact	No Impact	Medium Positive Impact	High Positive Impact	No Impact	No Impact	No Impact	Medium Positive Impact	High Positive Impact
Resolves Age Related Base Plant Challenges	20%	Medium Positive Impact	Medium Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact	High Positive Impact
Levelized Total Rate Impact	20%	Low Negative Impact	Low Negative Impact	Low Negative Impact	Low Negative Impact	Medium Negative Impact	High Negative Impact	High Negative Impact	Medium Negative Impact	Medium Negative Impact
Useful Under the Modified Diversified Energy (Planning) Scenario (mDEP) Between the In- Service Date and 2050	10%	Medium Negative Impact	Medium Negative Impact	No Impact	No Impact	No Impact	No Impact	No Impact	No Impact	No Impact
	Total Weighted Score:	0.1	0.1	1.4	1.8	0.4	0.0	1.5	1.9	2.9
										Preferred





1 2.5 RESILIENCY BENEFIT ASSESSMENT SUMMARY

This section presents the results from Exponent's analysis of how effective each technically and
commercially viable Supplemental Alternative (i.e., Supplemental Alternatives 2 to 9) is at
mitigating the risk from a winter T-South no-flow event.

5 Exponent used the same approach to determining risk and risk reduction across all alternatives. 6 FEI summarized the approach in Section 3.2 of the Supplemental Evidence, and Exponent also 7 describes its approach in its report.¹¹ At a high-level, Exponent: (1) determined the status quo risk 8 associated with a winter T-South no-flow event relying on the existing Tilbury Base Plant with no 9 capital upgrades (represented by Supplemental Alternative 1 (Planning)); and then (2) compared 10 this status quo risk to the risk of a winter T-South no-flow event assuming each Supplemental 11 Alternative is in place. The amount of risk mitigation provided by an alternative is then the 12 difference between the status quo risk and the risk with the alternative in place.

- 13 FEI summarizes the results as follows:
- Most of the Supplemental Alternatives would not materially improve FEI's current ability to withstand a winter T-South no-flow event. In other words, the Lower Mainland system would still depressurize, and all customers would still lose service, on the day of a T-South no-flow event at average winter temperatures;
- Even the contingent scenarios do not result in a significant improvement; and
- Supplemental Alternatives 7, 8 and 9 offer significantly greater risk mitigation relative to
 the other alternatives.

Exponent's risk reduction results are reflective of the importance of regasification capacity and
storage volume in determining the support duration, and that certain hazards are likely to cause
a failure of only a single T-South pipeline, leading to a regulatory shutdown of the intact pipeline.

24 2.5.1 Results of Exponent's Risk Assessment for Technically and 25 Commercially Viable Alternatives

26 Table C-4 below summarizes the resiliency modelling parameters for each Supplemental 27 Alternative and identifies the alternatives that, based on Exponent's risk calculations, materially 28 improve FEI's resiliency against a winter T-South no-flow event. The resiliency modelling 29 parameters are the send-out capacity and volume of LNG that is used for resiliency. As will be 30 discussed in Section 2.5.2 of this Appendix, the resiliency modelling parameters, combined with 31 the system demand, determine how long a Supplemental Alternative can support the entirety of 32 FEI's Lower Mainland load, which in turn dictates the level of risk mitigation. FEI used the amount 33 of risk reduction provided by an alternative against a T-South no-flow event at average winter

¹¹ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report.



- 1 conditions, as calculated by Exponent, to establish whether the alternative materially improves
- 2 resiliency.
- 3 4

Table C-4: Supplemental Alternatives that Materially Improve Resiliency at Average Winter Temperatures (+4°C)

Supp Alt #	Name	Resiliency Modelling Parameters	Material Resiliency Improvement?
Alternative 1		0 Bcf 150 MMcf/d	×
Alternative 1 Contingent	No Capital Upgrades with Optimized Liquefaction (No Resiliency Reserve)	0.35 Bcf 150 MMcf/d	×
Alternative 1 Contingent w/T1A		0.75 Bcf 150 MMcf/d	×
Alternative 2		0 Bcf 400 MMcf/d	×
Alternative 2 Contingent	New Regasification Only – 400 MMcf/d (No Resiliency Reserve)	0.35 Bcf 400 MMcf/d	×
Alternative 2 Contingent w/T1A		0.75 Bcf 400 MMcf/d	×
Alternative 3		0 Bcf 600 MMcf/d	×
Alternative 3 Contingent	New Regasification Only – 600 MMcf/d (No Resiliency Reserve)	0.35 Bcf 600 MMcf/d	×
Alternative 3 Contingent w/T1A		0.75 Bcf 600 MMcf/d	×
Alternative 4	Like-for-Like (No Resiliency	0 Bcf 150 MMcf/d	×
Alternative 4 Contingent	Reserve)	0.6 Bcf 150 MMcf/d	×
Alternative 4A	New 1 Bcf Tank (No Resiliency	0 Bcf 400 MMcf/d	×
Alternative 4A Contingent	Regasification	1 Bcf 400 MMcf/d	×
Alternative 5	Like-for-Like (Full Resiliency	0.6 Bcf 150 MMcf/d	×
Alternative 5 Contingent w/T1A	Reserve)	1 Bcf 150 MMcf/d	×
Alternative 6	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	1 Bcf 800 MMcf/d	×
Alternative 7	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification	2 Bcf 800 MMcf/d	✓



Supp Alt #	Name	Resiliency Modelling Parameters	Material Resiliency Improvement?
Alternative 8	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	1.4 Bcf 800 MMcf/d	✓
Alternative 9	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification	2 Bcf 800 MMcf/d	✓

2 The results in Table C-4 were determined based on Exponent's risk analysis. Figure C-3 below, 3 prepared by Exponent, summarizes the annual expected loss reduction associated with each 4 Supplemental Alternative relative to the existing Base Plant with no capital upgrades (the 5 Supplemental Alternative 1 (Planning) scenario).¹² Exponent also prepared figures based on 6 customer outages and customer-outage-days consequence metrics, and the pattern is the same. 7 Exponent's calculations show that, even at average winter temperatures, the calculated risk for 8 most Supplemental Alternatives remains similar to the existing Base Plant with no capital 9 upgrades. Even the contingent scenarios do not significantly improve risk mitigation. 10 Supplemental Alternatives 7, 8 and 9 offer significantly greater risk mitigation relative to the others. For example, Supplemental Alternative 9 provides over 4 times more risk mitigation than 11 12 Supplemental Alternative 6.

¹² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figure 41.





- 3 The choice of time horizon does not affect the relativities among the various Supplemental
- 4 Alternatives. This can be seen in the expected 23- and 67-year GDP loss results included below.13

¹³ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 42 and 43.





1





1

3 The above figures show risk mitigation for T-South related risks only (AVs 1, 2, 3 and 54), since 4 that is the measure used for the "Resiliency Benefits" criterion in the alternatives analysis. 5 However, Exponent also calculated the risk mitigation for each Supplemental Alternative on a 6 combined basis for all Assessed Vulnerabilities. The corresponding figures for all Assessed 7 Vulnerabilities are provided below as Figures C-6 to C-8.14 When these figures are compared to 8 the corresponding T-South-only figures above, they show that the larger Supplemental 9 Alternatives provide more risk mitigation in respect of non-T-South AVs than the smaller 10 Supplemental Alternatives.

¹⁴ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Figures 38-40.





Figure C-6: All AVs at Avg. Winter - Expected Annual Loss Reduction

2

1





1





1

3 2.5.2 Understanding the Results of Exponent's Risk Mitigation Calculations

Exponent's risk reduction results shown in the above figures are reflective of the importance of regasification capacity and storage volume in determining the support duration, and that certain hazards are likely to cause a failure of only a single T-South pipeline, leading to a regulatory shutdown of the intact pipeline. The implications of those considerations for the risk mitigation results are discussed below.

9 1. The Risk Mitigation Provided by a Supplemental Alternative is Dictated by its 10 Support Duration:

An alternative's load support duration (i.e., the duration the entire Lower Mainland system can be supported by the alternative until customer outages occur) influences the level of risk mitigation provided. For example, if the load support duration exceeds a given no-flow duration, then the Lower Mainland component of the risk will have been mitigated.



2. Regasification Capacity and Storage Volume Dictate Support Duration:

The support duration is, in turn, a function of the regasification capacity, the available volume of LNG, and the system demand. FEI explains in Section 3.2.2 of the Supplemental Evidence, for instance, how the constraint in regasification is presently limiting the Tilbury support duration to less than 1 day, even at average winter temperatures.

7 **3.** A 3-Day Support Duration is Necessary to Provide Significant Risk Mitigation:

8 Supplemental Alternatives that only marginally extend the load support duration are 9 unlikely to significantly mitigate risk, as many modes of failure on T-South are expected to 10 result in a no-flow event lasting approximately 3 days.

11 Exponent's risk assessment evaluates a T-South failure assuming a range of no-flow 12 durations and failure modes (i.e., hazards). In instances where there are parallel pipelines 13 (as is the case for T-South), Exponent's analysis considers the probability of each hazard 14 causing simultaneous failure of both pipelines. As many of the hazards considered have 15 a low probability of simultaneous failure, a likely outcome of a T-South failure is that only 16 one line is physically damaged and the adjacent undamaged line is shut-in as a precaution 17 by the regulator (i.e., a regulatory shutdown). For example, the duration of the no-flow 18 event during the 2018 T-South Incident was determined by the duration of the regulatory 19 shutdown.

- Exponent's calculations assume a regulatory shutdown will be resolved in 3 days; thus,
 alternatives that provide at least 3 days of support are able to bridge the regulatory
 shutdown and mitigate the risk for this specific failure scenario. This results in a significant
 risk reduction, given that a regulatory shut-down is a likely outcome of a T-South failure.
 Exponent explained:¹⁵
- 25 The on-system LNG of a Tilbury Alternative plays an important role in 26 determining the reduction in losses compared to the status quo 27 configuration (Alternative 1 (Planning)). When the Tilbury Facility has 28 limited on-system LNG (1 Bcf or less), relatively few hazards are mitigated 29 by its presence because there is not enough volume to bridge the 30 regulatory shutdown period or shorter repair durations for AVs with higher 31 loads, such as AV-1, AV-2, AV-3, AV-54, and AV-18. Larger on-system 32 LNG volumes (such as 2 Bcf) provide enough backup supply to bridge the 33 three-day regulatory shutdown period on AVs-1, 2, 3, 54, and 18, which are 34 AVs with parallel pipeline segments. Bridging the regulatory shutdown 35 period significantly reduces losses on these AVs, except for cases in which 36 the two parallel pipeline segments fail simultaneously. [Emphasis added]

¹⁵ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 192.


The assumption of a 3-day regulatory shutdown is considered reasonable given that the 1 2 regulatory shutdown during the 2018 T-South Incident was two days, and that was in 3 favourable conditions that facilitated crews reaching and assessing the condition of the 4 adjacent pipe. The 2018 T-South Incident timeline and the favourable conditions that 5 influenced it are discussed in the Application and several responses to information 6 requests.¹⁶ FEI considers it reasonable to assume that the regulator will perform its safety 7 assessment with due haste, as it will recognize that an unnecessarily prolonged 8 precautionary shut-down of the unaffected line will have significant consequences for all 9 downstream gas users.

Increased Regasification is Required to Increase the Load Support Duration and Improve Resiliency Against a Winter T-South No-Flow Event:

12 Regardless of the volume of LNG available at Tilbury, 150 MMcf/d of regasification (i.e., 13 the existing capacity at the Tilbury Base Plant) is insufficient to meet the daily winter 14 system demand (see further discussion in Section 3.3.4.2 of the Supplemental Evidence). 15 At average winter temperatures the system demand greatly exceeds 150 MMcf/d, resulting in a supply/demand imbalance that causes the system to fail on the day of a 16 17 winter T-South no-flow event. Thus, alternatives that do not increase the regasification 18 capacity above 150 MMcf/d would not increase the load support duration. In other words, 19 they would not provide risk mitigation relative to the existing Base Plant with no capital upgrades in the event of a T-South winter no-flow event at average winter temperatures. 20

21 Supplemental Alternatives 1, 4 and 5, all of which have 150 MMcf/d of regasification 22 capacity, fall in this category.

Increasing Regasification Alone Will Not Materially Increase the Load Support Duration or Improve Resiliency Against a Winter T-South No-Flow Event – More than 1 Bcf is Required:

- Supplemental Alternatives 2 and 3 (which were screened out at Step 2 due to their inability to preserve FEI's peaking supply) would only increase regasification capacity, without expanding the tank. However, an adequate volume of LNG must also be available to materially improve resiliency, since a higher rate of regasification will empty the tank faster. Exponent observed that more than 1 Bcf is required to make a material risk reduction:¹⁷
- 32When the Tilbury Facility has limited on-system LNG (1 Bcf or less),33relatively few hazards are mitigated by its presence because there is not34enough volume to bridge the regulatory shutdown period or shorter repair35durations for AVs with higher loads, such as AV-1, AV-2, AV-3, AV-54, and36AV-18.

¹⁶ Exhibit B-1-4, Application, p. 52 and Exhibit B-15, BCUC IR1 1.1 to 1.4.

¹⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, para. 192.



- Figure C-9 below summarizes the Lower Mainland support duration provided by each 1 2 Supplemental Alternative at various winter temperatures (the average Lower Mainland winter
- temperature of +4°C, -1.4°C, and -10°C) as determined by FEI's transient system modelling. The
- 3 4 relationship between an alternative's load support duration and the extent of the risk mitigation
- 5 calculated by Exponent is evident when reviewing this figure in conjunction with Exponent's
- 6 figures above.







Figure C-9: Duration of Lower Mainland Load Support Provided by Supplemental Alternatives

2

1



- 1 Based on the points above, FEI concludes the following:
- Supplemental Alternatives 1, 4, and 5 would not materially improve FEI's ability to withstand a winter T-South no-flow event relative to the existing Base Plant with no capital upgrades (Supplemental Alternative 1 (Planning)) because they would not increase regasification above the existing capacity of 150 MMcf/d. The load support duration would remain far too short. Additionally, Supplemental Alternatives 1 and 4 do not contemplate a dedicated resiliency reserve, thus from a planning perspective there would be no LNG volume available for resiliency.
- 9 2. Supplemental Alternatives 2, 3, 4A, and 6 would provide increased regasification 10 capacity and would, therefore, improve resiliency (for Supplemental Alternatives 2, 3 and 11 4A, this is only true when one considers the contingent scenarios). However, the 12 improvement would not be material. In the case of Supplemental Alternatives 3 and 6, 13 which would provide at least 600 MMcf/d of regasification, this is due to the limited LNG 14 volume available (i.e., the high rate of regasification guickly consumes the available LNG, 15 resulting in depressurization). In the case of Supplemental Alternative 2, which would provide 400 MMcf/d of regasification capacity, this is due to the limited LNG volume 16 17 available and insufficient regasification capacity (i.e., even if the available volume of LNG was increased, there would not be a material resiliency improvement). In the case of 18 19 Supplemental Alternative 4A, this is due to insufficient regasification capacity. Even if the 20 regasification constraint were fully resolved, the limited LNG volume available would 21 prevent Supplemental Alternative 4A (Contingent) from providing a material resiliency 22 improvement, As a result, Supplemental Alternatives 2, 3, 4A, and 6 would not materially 23 improve resiliency.
- 24 3. Supplemental Alternatives 7, 8, and 9 would add significant levels of both regasification 25 and storage volume, such that they are able to provide, for example, at least a 3-day 26 support duration under average winter conditions of +4°C. Thus, they would provide 27 significant and material risk mitigation at average winter conditions. As set out in Section 28 4.5.1 of the Supplemental Evidence, these alternatives provide differing levels of risk 29 mitigation at colder winter temperatures. In particular, Supplemental Alternative 8 would 30 be unable to provide 3 days of load support at colder temperatures that are present during 31 large portions of the winter.



FINANCIAL IMPACTS FOR ALL SUPPLEMENTAL ALTERNATIVES 3. 1

2 Section 4.5.4 of the Supplemental Evidence discusses the methodology for determining levelized 3 rate impacts, one of the five criteria used in the Step 3 scoring analysis, and evaluates the 4 levelized rate impacts of the four Supplemental Alternatives that passed both Step 1 and Step 2 5 screens (i.e., Supplemental Alternatives 4, 4A, 8 and 9). In order to provide the BCUC with full information, this section of Appendix C provides the financial impacts for Supplemental 6 7 Alternatives 1 to 9 and contingent scenarios, as well as additional information regarding the 8 associated inputs. The financial impacts are based on updated capital cost estimates and 9 consistent financial assumptions regarding gas supply requirements, availability in the market, 10 and associated costs.

3.1 FINANCIAL ANALYSIS RESULTS 11

12 Table C-5 below summarizes, for Supplemental Alternatives 1 to 9 and the associated contingent 13 scenarios, the following information: (1) construction capital cost; (2) total capital cost (including sustainment); (3) present value (PV) of the cost of service, which is borne by customers through 14 15 delivery charges; (4) PV of the cost of gas, which is borne by customers through Cost of Gas 16 charges; (5) PV of the revenue requirement, which is the sum of the aforementioned cost of 17 service and cost of gas; (6) the PV of FEI's current revenue requirement; (7) the levelized total rate impact; (8) the incremental levelized total rate impact when compared to Supplemental 18 19 Alternative 1; and (9) the payback period. The information is based on a 67-year analysis period, for the reasons described in Section 6.2.2 of the Supplemental Evidence. 20

21

Table C-5: Costs and Rate Impacts of the Supplemental Alternatives

		Total Capital			Total PV of	PV of 2024			
	Construction	Cost incl.			Revenue	Revenue	Levelized	Incremental	
	Capital Cost	Sustainment	PV Cost of	PV Cost of Gas	Requirement	Requirement	Total Rate	to Alt 1 - Base	Payback
Alternative	(\$MM)	(\$MM)	Service (\$MM)	(\$MM)	(\$MM)	(\$MM)	Impact (%)	Case (%)	Period (Years)
							(G) = (E) /	(H) = (G) -	
	(A)	(B)	(C)	(D)	(E) = (C) + (D)	(F)	(F)	Base Case(G)	(1)
Alt 1 (Base Case)	-	-	-	517.6	517.6	29,510.5	1.8%	0.00%	-
Alt 1 (Contingent)	-	-	-	517.6	517.6	29,510.5	1.8%	0.00%	-
Alt 1 (Contingent w/T1A)	-	-	-	517.6	517.6	29,510.5	1.8%	0.00%	-
Alt 2	391.5	1,290.7	464.2	517.6	981.7	29,510.5	3.3%	1.57%	-
Alt 2 (Contingent)	391.5	1,290.7	464.2	517.6	981.7	29,510.5	3.3%	1.57%	-
Alt 2 (Contingent w/T1A)	391.5	1,290.7	464.2	517.6	981.7	29,510.5	3.3%	1.57%	-
Alt 3	435.0	1,467.9	490.0	517.6	1,007.5	29,510.5	3.4%	1.66%	-
Alt 3 (Contingent)	435.0	1,467.9	490.0	517.6	1,007.5	29,510.5	3.4%	1.66%	-
Alt 3 (Contingent w/T1A)	435.0	1,467.9	490.0	517.6	1,007.5	29,510.5	3.4%	1.66%	-
Alt 4	826.9	1,887.9	790.0	151.2	941.2	29,510.5	3.2%	1.44%	22
Alt 4A	893.2	2,022.7	892.6	-	892.6	29,510.5	3.0%	1.27%	18
Alt 4 (Contingent)	826.9	1,887.9	790.0	151.2	941.2	29,510.5	3.2%	1.44%	-
Alt 5	826.9	1,887.9	790.0	517.6	1,307.6	29,510.5	4.4%	2.68%	-
Alt 5 (Contingent w/T1A)	826.9	1,887.9	790.0	517.6	1,307.6	29,510.5	4.4%	2.68%	-
Alt 6	933.5	2,224.0	943.1	517.6	1,460.7	29,510.5	4.9%	3.20%	-
Alt 7	1,030.3	2,276.9	1,134.0	517.6	1,651.5	29,510.5	5.6%	3.84%	-
Alt 8	1,030.3	2,276.9	1,134.0	151.2	1,285.2	29,510.5	4.4%	2.60%	27
Alt 9 (Preferred)	1,141.0	2,370.0	1,240.8	-	1,240.8	29,510.5	4.2%	2.45%	22



1 FEI highlights two points:

 The various Supplemental Alternatives can have an impact on the delivery rates through the capital costs, or an impact on the cost of gas related to the costs/savings for gas supply resources, or both. As such, focusing on only one of these components in isolation can distort the overall impacts for customers. As explained in Section 6.2.1 of the Supplemental Evidence, the levelized total rate impact used to compare among various alternatives includes both the impacts due to the capital costs and gas supply costs/savings over the 67-year period; and

9 The Contingent scenarios, which are sensitivities assuming a different amount of LNG is • 10 present on the day of a no-flow event, show the same financial results as compared to 11 their associated Planning Alternative (e.g., Supplemental Alternative 2 (Contingent) and 12 (Contingent w/T1A) would have the same costs and rate impacts as Supplemental 13 Alternative 2 (Planning)). This is because the Contingent scenarios do not change the 14 underlying planning for the utility; rather, they just take advantage of the resources that 15 are in place and may be available at the time of a no-flow event on T-South to determine 16 how non-dependable resources could affect resiliency. FEI discusses the Contingent 17 scenarios in the Section 4 of this Appendix.

18 3.2 FINANCIAL ANALYSIS ASSUMPTIONS: BASE COST ESTIMATES

As discussed in Section 6 of the Supplemental Evidence, FEI, in conjunction with Linde, Horton
CB&I (HCBI), Golder, and Solaris Management Consultants Inc. (SMCI), updated the AACE
Class 3 Project capital cost estimate for Supplemental Alternative 9 as well as Supplemental
Alternatives 7 and 8 using AACE International Recommended Practices 18R-97 and 97R-18 as
guides. These estimates were originally prepared in 2020 and updated in 2023.

In 2023, FEI retained these external experts to additionally prepare AACE Class 4 cost estimates
 for the new Supplemental Alternatives. Solaris prepared the Class 4 Ground Improvement cost
 with inputs from WSP and FEI. Linde and Solaris prepared the Class 4 Regasification Package
 cost estimate. HCBI prepared Class 4 LNG Storage Tank cost estimates and factored it for both
 0.6 Bcf and 1 Bcf Tank sizes. Solaris prepared Class 4 estimates for both Base Plant Demolition
 and Auxiliary Systems.

30 3.3 FINANCIAL ANALYSIS ASSUMPTIONS: PEAKING SUPPLY COSTS / BENEFITS 31 REFLECTED IN LEVELIZED TOTAL RATE IMPACT

FEI used the approach described in Section 4.5.4 of the Supplemental Evidence to determine the
 annual gas supply costs or benefits (avoided costs) for Supplemental Alternatives 1 to 9. In
 summary:

Supplemental Alternative 1 – No Capital Upgrade serves as the baseline in the analysis.
 It would entail several years when FEI would face curtailments of up to 150 MMcf/d for a



- period of time (2030-2035) due to an inability to obtain sufficient peaking supply to replace
 the loss of access to Tilbury LNG. It would also entail significant annual peaking gas supply
 costs throughout the entire assessment period. The annual costs increase over time as
 tolls / charges on regional infrastructure increase to reflect expansions;
- The same is true for Supplemental Alternatives 2, 3, 5, 6 and 7, all of which would depend on procuring capacity on regional infrastructure to provide sufficient peaking supply to avoid firm curtailments in normal operations. Supplemental Alternatives 2 and 3, which only replace the regasification equipment, rely on the market because they face the loss of access to sufficient stored energy at Tilbury by 2030. Supplemental Alternatives 5, 6 and 7 depend on the market because the new facility would be set aside in its entirety as a resiliency reserve. The annual gas costs are the same as for the baseline.
- Supplemental Alternatives 4, 4A, 8 and 9 avoid the above-described annual costs to varying degrees. Supplemental Alternatives 4 and 8 still require supplementing LNG with 50 MMcf/d on regional infrastructure. Supplemental Alternatives 4A and 9 avoid the need to incur any annual peaking gas costs for market resources, since they meet FEI's full requirements; and
- FEI conservatively used the (lower) annual cost of storage for gas supply costs/savings.
 Using the higher pipeline costs would have the effect of improving the levelized total rate
 impact of the Supplemental Alternatives that meet all of FEI's peaking gas supply
 requirements (i.e., Supplemental Alternatives 4A and 9).
- 21 Table C-6 provides the annual gas supply costs/benefits for each Supplemental Alternative.



Table C-6: Avoided Gas Supply Market Resources Included in Financial Analysis

		Annual Gas Supply Costs (\$millions)			Incremental to Baseline / (Avoided Costs) (\$ millions)		
Supplemental		Present to	2030 to	2035	Present to	2030 to	2035
Alternatives	Description	2030	2035	onwards	2030	2035	onwards
	No Capital Upgrades (Continue to rely on existing						
1	Base Plant until it fails. No on-system peaking	7.0	7.0	63.0			
	gas supply thereafter and no resiliency reserve)						
	New Regasification Only - 400 MMcf/d (Continue						
2	to rely on existing Base Plant until it fails. No on-	70	7.0	63.0		-	-
2	system peaking gas supply thereafter and no	7.0	7.0		-		
	resiliency reserve)						
	New Regasification Only - 600 MMcf/d (Continue						
2	to rely on existing Base Plant until it fails. No on-	7.0	7.0	63.0	-	-	-
5	system peaking gas supply thereafter and no	7.0					
	resiliency reserve)						
4	Like-for-Like Replacement for 0.6 Bcf and 150	70	70	17.0	_	_	(46.0)
	MMcf/d (No Resiliency Reserve)	7.0	7.0	17.0			(40.0)
40	New 1 Bcf Tank and 400 MMcf/d Regasification	70	_	_	_	(70)	(63.0)
	(No resiliency reserve)	7.0				(7.0)	(03.0)
	Like-for-Like Replacement for 0.6 Bcf and 150						
5	MMcf/d (Full resiliency reserve and no allocation	7.0	7.0	63.0	-	-	-
	for peaking gas supply)						
	New 1 Bcf Tank and 800 MMcf/d Regasification						
6	(Full resiliency reserve and no allocation for	7.0	7.0	63.0	-	-	-
	peaking gas supply)						
	New 2 Bcf Tank and 800 MMcf/d Regasification						
7	(Full resiliency reserve and no allocation for	7.0	7.0	63.0	-	-	-
	peaking gas supply)						
	New 2 Bcf Tank and 800 MMcf/d Regasification						
8	(1.4 Bcf resiliency reserve and 0.6 Bcf for peaking	7.0	7.0	17.0	-	-	(46.0)
	gas supply)						
_	New 3 Bcf Tank and 800 MMcf/d Regasification (2						· ·
9	Bct resiliency reserve and 1 Bcf for peaking gas	7.0	-	-	-	(7.0)	(63.0)
	supply)						

2

1

14.DEVELOPMENT OF PLANNING AND CONTINGENT MODELLING2SCENARIOS TO ADDRESS ADJOURNMENT DECISION

FORTIS BC^{**}

3 The alternatives analysis provided in the Application took a typical utility reliability planning 4 approach that is premised on sizing infrastructure and gas supply assets to be able to meet firm 5 customer requirements consistently. On a planning basis, it is not possible to treat a resource as 6 being dependable for two different purposes. That is, Tilbury 1A must be available on a 7 dependable basis for RS 46 LNG sales (the purpose for which it was constructed pursuant to 8 Direction No. 5 to the BCUC), making it non-dependable for resiliency. FEI's Annual Contracting 9 Plan treats a certain amount of LNG as dependable in the gas supply portfolio, making that same 10 volume non-dependable for resiliency. The focus on dependability requires basing the resiliency 11 analysis on a "resiliency reserve" that is dedicated to resiliency. In the absence of a resiliency 12 reserve, the dependable LNG for resiliency is zero.

13 The BCUC stated in the Adjournment Decision:¹⁸

We appreciate that the primary purpose of Tilbury Tank 1A is to serve FEI's existing and future LNG customers. Further FEI appears to operate the tank for that purpose. As a result, the average storage volumes are about two thirds of the tank capacity. At any given time, however, there have been volumes ranging from 200 MMcf/day [sic – 200 MMcf or 0.2 Bcf] to 1,000 MMcf/day [sic – 1000 MMcf or 1 Bcf] that appear to be available for resiliency purposes.

The evidence shows that FEI's existing liquification capacity would allow FEI to keep Tilbury Tank 1A more full than it currently chooses to do for operational reasons....

FEI remains of the view that resiliency planning should follow typical utility planning principles, and has thus assessed a variety of facility sizes with different allocations of the tank as between a "resiliency reserve" and gas supply functions. However, in response to the Adjournment Decision, FEI evaluated sensitivities for certain Supplemental Alternatives (referred to as "contingent" scenarios) to determine whether the availability of additional LNG volumes at Tilbury on the day of a no-flow event would materially mitigate customer outage risk.

FEI describes below why resiliency planning based on non-dependable LNG resources is risky, with reference to peaking gas supply utilization and forecasts for RS 46 LNG sales. In practice, it is a moot point when it comes to mitigating FEI's largest customer outage risk. Exponent's analysis confirms that none of the contingent scenarios would materially improve FEI's ability to withstand a winter T-South no-flow event even at average winter temperatures (refer to Section 2.5 of this Appendix). This would be true even under the assumption that the entirety of Tilbury 1A was present.

¹⁸ Adjournment Decision, p. 29.



1 4.1 DESCRIPTION OF PLANNING AND CONTINGENT SCENARIOS

- 2 The planning and contingent scenarios in FEI's alternatives analysis are as follows:
- Planning: Supplemental Alternatives 1 to 9 have planning scenarios. The planning scenarios reflect any LNG volumes being allocated to resiliency, while any LNG volumes at Tilbury planned for RS 46 LNG sales and gas supply have no dependable value for resiliency. On a planning basis, it is not possible to treat a resource as being dependable for two different purposes.
- 8 • **Contingent:** Supplemental Alternatives 1, 2, 3, 4, and 4A, which are the alternatives that contemplate the existing Base Plant (a new 0.6 Bcf tank or a new 1 Bcf tank) also have 9 10 contingent scenarios. These scenarios assume that some LNG is available from the 11 existing Base Plant or a replacement facility on the day of a no-flow event and is used for 12 resiliency instead of the intended purpose (i.e., gas supply). The amount of LNG assumed to be present is 0.35 Bcf for alternatives reliant on the existing derated Base Plant tank.¹⁹ 13 14 This volume could be attributable, for instance, to (a) partial consumption of the ACP gas 15 supply allocation for peaking supply, or (b) Tilbury 1A volume not being available due to 16 RS 46 sales, but the Base Plant tank being full to its reduced fill capacity of 0.35 Bcf. The 17 amount of LNG assumed to be present is 0.6 Bcf for an alternative that assumes a likefor-like 0.6 Bcf tank intended to maintain existing gas supply functions,²⁰ and 1 Bcf for an 18 alternative that contemplates a facility to optimize FEI's gas supply portfolio.²¹ 19
- Contingent with Tilbury 1A: These sensitivities are the most optimistic. They assume that both of the following are available: (a) LNG volumes from the existing Tilbury Base
 Plant or a new facility as per the "contingent" scenarios; and (b) 0.4 Bcf from Tilbury 1A is present to further support resiliency, despite the intended purpose of RS 46 sales.²²

24 4.2 CONTINGENT MODELLING SCENARIOS ARE NOT DEPENDABLE

Not having a dedicated resiliency reserve (as reflected in the contingent modelling scenarios) is a risky strategy, despite a modified approach to using liquefaction intended to maintain higher levels of LNG in the tanks. The three reasons for this are discussed in the following subsections.

4.2.1 FEI Peaking Gas Supply Requirements Exceed 0.6 Bcf in a Typical Winter

- 30 The first reason why the contingent scenarios are risky is that FEI's requirements for peaking gas
- 31 supply exceed 0.6 Bcf in a typical winter. This would leave the available LNG on the day of a
- 32 winter no-flow event below the amounts assumed in the contingent modelling scenarios.

Section 4: Development of Planning and Contingent Modelling Scenarios to Address Adjournment Decision PAGE 38

¹⁹ Supplemental Alternatives 1 (Contingent), 2 (Contingent), and 3 (Contingent).

²⁰ Supplemental Alternative 4 (Contingent).

²¹ Supplemental Alternative 4A (Contingent).

²² Supplemental Alternatives 1 (Contingent w/T1A), 2 (Contingent w/T1A), 3 (Contingent w/T1A), and 5 (Contingent w/T1A).



- FEI's use of LNG for peaking supply depends on weather, and significant volumes are commonly 1
- 2 used during the winter. For example, during winter 2022/23, the 0.6 Bcf allocation of Tilbury LNG
- 3 for supply purposes was reduced to 0.17 Bcf by Christmas 2022 and was down to 0.07 Bcf by the
- 4 beginning of March 2023 when FEI began filling the tank again. This was the product of high demand during cold weather periods in December and February, as well as operational reasons
- 5
- 6 (e.g., compressor outages and tool runs).
- 7 Because FEI now operates the Base Plant at 0.35 Bcf for seismic reasons, and peak loads exceed
- 8 the capabilities of the Tilbury Base Plant, FEI currently draws on LNG from Tilbury 1A to meet the
- 9 incremental demand.

10 Moreover, if peaking supply is used for a resiliency event, it would leave FEI dependent on 11 sourcing gas from the market to meet demand during typical cold weather events for the remainder of the winter. FEI discusses in Section 3.3.4.3 of the Supplemental Evidence how, 12 13 under current market conditions, it would not be realistic to expect that FEI could replace the 14 Tilbury peaking supply at the Sumas market. Significant curtailments of firm load in normal

15 operations are a likely outcome.

16 4.2.2 Availability of Tilbury 1A Volumes on a Given Day Will Depend on 17 Overall RS 46 Demand and Use Patterns During a Year

18 LNG levels in the Tilbury 1A tank are a function of both overall RS 46 demand and the patterns 19 of sales over the course of a year. FEI discusses below how the past average storage levels and 20 use patterns within the year are unlikely to be replicated in the future, as the market is growing 21 and evolving. While RS 46 demand from Tilbury 1A has been impacted by the COVID-19 22 pandemic and delays in Tilbury Jetty approvals, RS 46 demand awaiting the jetty's completion 23 exceeds the capacity of Tilbury 1A. While some of the LNG in the Tilbury 1A tank will potentially 24 be available on a given day, realistically there are likely to be times at unpredictable intervals 25 when the tank is very depleted. FEI's analysis suggests that its contingent scenario assumption 26 of 0.4 Bcf being present at any one time should be considered an upper limit.

27 4.2.2.1 Future RS 46 Sales Growth and Evolving Use Patterns

28 As noted in the response to BCUC IR1 11.9.2 (Exhibit B-15), the tank volumes in Tilbury 1A over 29 the past number of years have fluctuated depending on volume of sales and the need to refill the 30 Base Plant tank (as the Base Plant tank is filled through an interconnecting line from the Tilbury 31 1A tank). The delay in utilizing more LNG production for LNG sales is linked to both the impacts 32 to the supply chain from the COVID-19 pandemic, as well as the delays in regulatory approvals 33 for the marine jetty planned adjacent to the Tilbury site (Tilbury Jetty). As discussed below, the 34 overall annual sales under RS 46 are expected to increase significantly and, as such, the patterns 35 of use will change as the market evolves. In particular, FEI expects the types of sales to change.

36 The types of sales impact the pattern of LNG use. LNG is currently sold in ISO containers or on 37 road transport containers which are loaded through one of the two existing truck loading bays at 38 the site. Once the Tilbury Jetty has been constructed, sales of LNG will likely shift from being



- loaded onto ISO containers towards being loaded onto ships for transport, most likely for use in
 displacing higher burning carbon fuels for ships in the Port of Vancouver. This shift in sales type
- 3 is significant, as one ISO container is 40 m³ (1,412ft³) while an LNG bunker vessel (fuelling ship)
- 4 can range from $3,000-18,000 \text{ m}^3$ (105,944 635,664 ft³). When considering the graph of tank
- 5 levels shown in the response to BCUC IR1 11.9.2 (reproduced below), the LNG was being sold
- 6 through ISO or on-road transport containers which is why the trends showed a period of filling
- 7 and then a long slow drawdown of the tank as LNG was extracted to provide fuel for customers
- 8 such as BC Ferries and Seaspan, as well as ISO customers.

9 With the construction of the Tilbury Jetty, larger vessels will be loading LNG, resulting in a much 10 quicker drawdown. The purpose of the Tilbury 1A tank as it relates to LNG sales is to provide 11 storage as a buffer for plant maintenance downtimes or for when sales temporarily exceed 12 capacity. The Tilbury 1A tank holds 1 Bcf of LNG. The Tilbury 1A liquefaction plant produces 13 approximately 33 MMcf/d with approximately 5 MMcf/d dedicated to utility use. As noted above, 14 LNG bunker vessels can range in size; however, for illustrative purposes, FEI refers to a vessel 15 recently announced by Seaspan which they plan to operate in the Port of Vancouver. This vessel 16 is planned to be 0.163 Bcf in size. Considering this, the Tilbury 1A tank holds enough LNG to fill 17 the Seaspan vessel just over 6 times (without refilling the tank).

18 Further, standard practice in the LNG industry is to develop annual delivery plans when 19 considering LNG sales. These plans are built around customer delivery needs as well as to ensure 20 the LNG facility can manage its maintenance activities. For Tilbury 1A, given FEI sells LNG both 21 through its truck loading bays (current state) and intends to sell LNG through the Tilbury Jetty, 22 the annual delivery plan would also assist with production planning to ensure FEI can meet all of 23 its sales obligations. There is considerable uncertainty inherent in the development of yearly 24 delivery plans as customer needs and requirements can change over time. A bunker vessel 25 picking up fuel at Tilbury will be serving more than one end use customer and as a result, the 26 delivery schedule for LNG will very likely be erratic during the year, although matched to the 27 production capabilities of the facility.

28 The Tilbury 1A tank provides a mechanism to deal with the erratic nature of LNG scheduling. For 29 example, as noted above. FEI could load six of the Seaspan bunker vessels (6 times 0.163 Bcf = 30 0.98 Bcf) if the plant was undergoing maintenance. If those ships were scheduled every third day, 31 this would provide nearly 20 days of maintenance outage before the tank ran out of LNG. As 32 another example, if the plant were running and considering the current liquefaction capacity of 33 0.25 MTPA, FEI could load one Seaspan vessel approximately every six days. In the first case, 34 the tank level would be dropping with every load until it was empty. In the second case, the tank 35 levels would remain relatively constant as delivery and production are matched. This could be 36 considered an ideal situation.

The reality of LNG sales is that vessels are not loaded on a regular interval. More realistic is the requirement that loading is required every day for a period of time and then there is an interval of time where no loading occurs. In those cases, it is conceivable that the LNG level in the Tilbury 1A tank would drop with every ship loaded (as the capacity of the ship exceeds the production



- 1 levels) until a time where no vessels can be loaded until inventory is replenished. The exact nature
- 2 and times of year that this would occur cannot be known until the actual customers' requirements
- are known. However, the flexibility provided by the Tilbury 1A tank is very valuable to the LNG
 business as it enables FEI to accommodate (to a point) the timing of customers' loading
- 5 requirements.

4.2.2.2 "Contingent w/T1A" Sensitivities Assumed 0.75 Bcf is Available (0.4 Bcf in Tilbury 1A Plus Full Base Plant)

Given expected future use of the Tilbury T1A tank and evolving use patterns discussed above,
FEI undertook preliminary modelling of tank levels based on several scenarios (see Figure C-10
below) and determined that 0.4 Bcf would be the upper limit of what could or should be considered
present in the Tilbury 1A tank at any one time. FEI's "Contingent w/T1A" sensitivities thus included
0.4 Bcf from Tilbury 1A, plus 0.35 Bcf in the Base Plant.

- FEI's selection of 0.4 Bcf as an upper limit assumption for volume present in Tilbury 1A was based
 on an analysis of production and usage. Figure C-10 below shows different Tilbury 1A production
 and usage scenarios. FEI determined that 0.4 could be present in Tilbury 1A by considering likely
- 16 loading scenarios based on FEI's current understanding of customer requirements:
- Scenario #1 reflects an idealized balancing of loading with production (this example considers only Tilbury 1A production and reflects a vessel approximately every 6 days).
 This is an unrealistic scenario, since (as discussed above) the reality of LNG sales is that vessels are not loaded on a regular interval.
- Scenarios #2 and #3 reflect situations where customer requirements result in periodic loading requirements throughout the month. Scenarios #2 and #3 are illustrative of potential sales patterns and were selected to show that any type of loading scenario that is not equally balanced throughout the year will result in fluctuating tank volumes. The actual loading scenarios will depend on the type of customers, their schedules and their fuelling requirements.





Importantly, while FEI's analysis of certain contingent modelling scenarios has treated 0.4 Bcf as potentially available, in any given year the annual delivery plans developed for FEI LNG customers may require use of the LNG inventory below that level in order to meet sales commitments. As noted above, this means that unlike a resiliency reserve, these volumes are not dependable.

16 Moreover, as noted above and discussed in Section 3.3 of the Supplemental Evidence, FEI now

operates the Base Plant tank at 0.35 Bcf and FEI's peaking gas requirements exceed that amount.

18 FEI is temporarily relying on Tilbury 1A to maintain its Annual Contracting Plan requirements of

19 0.6 Bcf, pending the replacement of the Base Plant.

4.2.3 Changing How FEI Uses Liquefaction Would Not De-Risk Relying on Non-Dependable Volumes for Resiliency

Depending on the profile of the LNG sales, the LNG levels at Tilbury could remain depleted evenwith increased liquefaction.

24 Operationally, Tilbury works as follows:

Tilbury 1A houses the only functioning liquefaction equipment. The Tilbury 1A tank can be
 refilled in approximately 30 days from empty. Operationally, FEI strives to maintain tank
 levels in Tilbury 1A at a minimum of 60 percent through the winter months to strike a
 balance between maintaining sufficient inventory versus frequent starting and stopping of
 the liquefaction equipment. The liquefaction equipment was designed to run continuously
 so frequent starting and stopping accelerates wear on equipment due to metal-on-metal
 wear in machinery, accelerated piping/vessel corrosion from thermal cycling, and



- accumulation of moisture in electrical equipment including motor windings and junction
 boxes, all of which are undesirable outcomes; and
- 3 Since only the Base Plant has regasification equipment, and since the Base Plant no 4 longer has functioning liquefaction equipment, the LNG in Tilbury 1A must be transferred 5 to the Base Plant. The Base Plant tank is filled using Tilbury 1A production and the 6 interconnecting line between the Tilbury 1A tank and the Base Plant tank. The maximum 7 transfer/fill rate is approximately 5 MMcf/d. As such, it takes approximately 72 days to fill 8 0.35 Bcf of LNG into the Base Plant tank from empty, which means that refilling the Base 9 Plant tank with additional LNG volumes from Tilbury 1A if the Base Plant is depleted for gas supply purposes would take the majority of the winter. 10

114.3A FULL TILBURY1A TANK WOULD NOT MATERIALLY REDUCE RISK12Exposure to WINTER NO-FLOW EVENT

Exponent's results (see Figure C-11 below) confirm that none of the contingent scenarios would
 materially improve FEI's ability to withstand a winter T-South no-flow event even at average winter
 temperatures.

- In the case of the contingent scenarios for Supplemental Alternatives 1, 4, and 5, the 150
 MMcf/d regasification capacity, which is insufficient to support system demand at average
 winter temperatures or colder, is not addressed and remains the primary constraint; and
- In the case of the contingent scenarios for Supplemental Alternative 2, the primary constraint is either insufficient regasification capacity (Supplemental Alternative 2 (Contingent w/ T1A)) or an insufficient available volume of LNG (Supplemental Alternative 2 (Contingent)). In the case of the contingent scenarios for Supplemental Alternative 3, there is an insufficient volume of LNG. In the case of Supplemental Alternative 4A, there is insufficient regasification capacity.

25 FEI did not specifically model a contingent sensitivity based on the assumption that the Tilbury 26 1A tank is completely full, given that this was considered to be highly unrealistic. However, 27 Exponent's results in Figure C-11 for Supplemental Alternative 6 (1 Bcf and 800 MMcf/d) can be 28 used as a proxy for the hypothetical scenario where the Tilbury 1A tank is full and the 29 regasification constraint is removed. The level of risk mitigation provided by Supplemental 30 Alternative 6 in Figure C-11 shows that a full Tilbury 1A tank would not materially reduce the risk 31 associated with a winter T-South no-flow event, even if the existing limited regasification was 32 replaced with new regasification equipment with over five times more regasification capacity.





Figure C-11: T-South at Avg. Winter – Expected Annual Loss Reduction

2

1

15.DETAILED INFORMATION SUPPORTING ASSESSMENT OF THE2SUPPLEMENTAL ALTERNATIVES

3 This section provides the underlying information supporting FEI's assessment of each 4 Supplemental Alternative, including those located at Tilbury, non-Tilbury alternatives raised by 5 the BCUC in its Adjournment Decision, as well as a Southern Crossing Pipeline extension.

6 **5.1** ALTERNATIVES RELYING ON EXISTING FACILITIES

7 In response to the Adjournment Decision, this section addresses three Supplemental Alternatives

8 (1, 2, and 3) that contemplate continued reliance on the existing Tilbury Base Plant regasification

9 and/or tank. These alternatives assume a "run until it is no longer usable" approach for some or all

- 10 of the Base Plant assets.
- 11 For the reasons described below, this approach would be high-risk for customers in terms of:
- Exposure to widespread and prolonged customer outages following a no-flow event;
- Potential for an unplanned, multi-year loss of dependable peaking resources between the
 Base Plant's end-of-life and commissioning of a replacement facility; and
- A high likelihood of increasingly unreliable service in normal conditions.

FEI determined that Supplemental Alternative 1 is technically non-viable at Step 1 because the Tilbury Base Plant has reached end-of-life and can no longer reliably perform its intended function. Supplemental Alternatives 2 and 3 were found to be non-viable at Step 2 because they put FEI's ability to continue meeting peak load in normal conditions at risk.

20 **5.1.1 Supplemental Alternative 1**

- 21 Supplemental Alternative 1 includes the following:
- Continuing to rely on the existing regasification equipment (150 MMcf/d), which has reached end-of-life;
- Continuing to rely on a non-refurbished Base Plant tank, which has reached end-of-life and is operating at a lower maximum volume (0.35 Bcf), and leave unaddressed the seismic, environmental and flooding issues inherent in the original Base Plant design;
- Continuing to operate with no resiliency reserve; that is, from a planning perspective, all
 of the LNG at Tilbury would still be allocated to either gas supply or RS 46 LNG sales,
 such that those volumes may or may not be present upon a no-flow event;
- Changing how FEI operates the existing Tilbury 1A liquefaction to replenish consumed
 LNG faster, so as to increase the potential for LNG to be present on the day of a no-flow

FORTIS BC[®]



- event (i.e., 0.35 Bcf in the Base Plant tank and 0.4 Bcf in Tilbury 1A despite RS 46 LNG
 sales);²³
- From the gas supply perspective, meet the peaking supply requirement of 1 Bcf and 200
 MMcf/d by continuing to rely on the non-refurbished Base Plant until it is no longer usable,
 plus continuing to hold 50 MMcf/d of year-round pipeline capacity;
- When the existing Base Plant is no longer usable (assumed to be 2030 for the purpose of the financial analysis), FEI would be unable to replace the lost peaking capabilities of 150 MMcf/d and 0.6 Bcf storage with regional infrastructure since they are already fully contracted. As such, FEI would still hold the existing 50 MMcf/d of market pipeline capacity but would likely have to curtail firm loads in normal operations by up to 150 MMcf/d unless and until future regional infrastructure upgrades occur that could provide replacement peaking supply; and
- If and when regional infrastructure upgrades occur (assumed to be 2035 for the financial analysis), FEI could potentially end curtailments by holding gas supply contracts for 1.0
 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that includes the costs of the upgrades.

17 5.1.1.1 Modelling Parameters

Table C-7 below sets out the resiliency and gas supply modelling parameters for theSupplemental Alternative 1 scenarios.

The planning scenario uses the maximum regasification (150 MMcf/d), paired with the volume of dependable LNG for resiliency (0 Bcf – no resiliency reserve) and, for gas supply, the amount included in the Annual Contracting Plan (0.6 Bcf).

FEI also modelled two contingent (i.e., non-dependable) scenarios based on more optimistic assumptions about how much LNG is present at Tilbury on the day of a no-flow event. While these additional scenarios address comments in the Adjournment Decision, as explained in Section 4.2 of this Appendix, FEI cannot rely on the LNG volumes assumed in the contingent scenarios being present on the day of a no flow event

27 present on the day of a no-flow event.

²³ This assumption addresses the following BCUC commentary in the Adjournment Decision, p. 29: "With additional liquefaction the average storage volume could be increased thereby providing additional capacity available for resiliency purposes. While there is evidence that this would be more costly than the current operational strategy, the cost benefit analysis is incomplete. In order to fully evaluate a potential role of Tilbury Tank 1A in FEI's resiliency portfolio, a more fulsome analysis is required."



Supp. Alt. 1 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 1 (Planning)	No Resiliency Reserve ²	0 Bcf at 150 MMcf/d	0.6 Bcf at 150 MMcf/d
Supplemental Alternative 1 (Contingent)	Assume Partial LNG Availability on Day of No-Flow ³	0.35 Bcf at 150 MMcf/d	0.6 Bcf at 150 MMcf/d
Supplemental Alternative 1 (Contingent w/T1A)	Assume Base Plant Fully Available plus 0.4 Bcf in Tilbury 1A on Day of No-Flow	0.75 Bcf at 150 MMcf/d	0.6 Bcf at 150 MMcf/d

Table C-7: Supplemental Alternative 1 Planning and Contingent Modelling Scenarios

2

1

- 3 <u>Notes to Table:</u>
- Per the ACP, FEI allocates 0.6 Bcf from Tilbury for gas supply portfolio purposes, with a 0.25 Bcf portion temporarily
 obtained from Tilbury 1A. Actual volumes will decline with use for gas supply during winter, or if LNG volumes were
 used for resiliency.
- When there is no resiliency reserve, on a planning basis there is no dependable LNG available on occurrence of no-flow event. Hence the resiliency modelling parameter is 0 Bcf.
- Reduced volume available could be attributable to either: (a) Tilbury 1A volumes being unavailable due to RS 46 sales, leaving a maximum of 0.35 Bcf in the Base Plant tank at currently reduced fill levels; or (b) partial consumption for peaking gas supply of the 0.6 Bcf currently allocated to gas supply at Tilbury (which is currently comprised of 0.35 Bcf from the Base Plant plus 0.25 Bcf Tilbury 1A). The former is expected to occur in the future, and the latter is common over the course of a typical winter.

14 *5.1.1.2* Summary of Information and Analysis

Supplemental Alternative 1 did not pass the Step 1 viability screen because the Base Plant has reached end-of-life, such that continuing to rely on it without capital upgrades is not technically viable. However, for completeness, FEI assesses Supplemental Alternative 1 based on the five criteria utilized in Step 3 of the alternatives analysis. Further, as explained in Section 2.4 of this Appendix, FEI scored Alternatives 2 to 9 relative to Supplemental Alternative 1 based on the five criteria; thus, Supplemental Alternative 1 does not have an impact score for each criterion.

- The following bullets provide a high-level description of how this alternative performs against the criteria, followed by sections providing additional details.
- Resiliency: Supplemental Alternative 1 would provide no additional risk mitigation against a winter T-South no-flow event. In particular, the limited regasification capacity at Tilbury of only 150 MMcf/d means Supplemental Alternative 1 (including the most optimistic contingent scenario) would not materially improve resiliency against FEI's largest customer outage risk. Even increased use of liquefaction does not change the consequence associated with a winter T-South no-flow event.



- Gas Supply: Supplemental Alternative 1 would retain FEI's existing on-system peaking capabilities for only as long as the Tilbury Base Plant remains functional and FEI is able to continue relying on LNG from Tilbury 1A to supplement the Base Plant's reduced capabilities. After that, there would likely be at least several years where there are significant curtailments in normal operations before FEI would be able to acquire enough replacement peaking supply on upgraded regional infrastructure.
- Base Plant Challenges: Supplemental Alternative 1 would not resolve the existing Base
 Plant's age-related challenges as it assumes no capital upgrades at Tilbury.
- Levelized Total Rate Impact: Supplemental Alternative 1 would have a levelized total rate impact of 1.8 percent over a 67-year analysis period. This is lower than Supplemental Alternative 9 because there are no capital costs pertaining to Supplemental Alternative 1, but it assumes FEI would be curtailing firm customers in normal operations for several years instead of incurring peaking gas costs.
- Future Use: FEI does not expect the Base Plant to still be in-service by 2050; therefore,
 FEI does not expect the facility to underutilized.
- 16 FEI discusses each of the five criteria in further detail below.

17 5.1.1.2.1 <u>RESILIENCY</u>

Supplemental Alternative 1 would provide no additional protection against a winter T-South noflow event on T-South because it assumes no capital upgrades at Tilbury (i.e., maintains the status quo) and, therefore, does not address the regasification capacity constraint at Tilbury of

21 only 150 MMcf/d for the contingent scenarios.

Regardless of how much liquefaction capacity, storage volume or inventory exists behind these regasification units, the most gas that FEI can access from Tilbury on any day is 150 MMcf. This is only a fraction of the daily load in the Lower Mainland on winter days. Even assuming an optimistic contingent scenario, the system would still fail on the day of a winter T-South no-flow event.

These results are confirmed in Exponent's figure (reproduced in Section 2.5.1 above) which summarizes the annual expected loss reduction associated with each Supplemental Alternative relative to the existing Base Plant with no capital upgrades (the Supplemental Alternative 1 (Planning) scenario).

- The following table summarizes how long the existing Base Plant, in combination with FEI's other existing capabilities, will be able to sustain the Lower Mainland load following a winter no-flow event. FEI has consistently modelled three temperatures that are reflective of local conditions,
- 34 since FEI's load increases as temperatures decrease.



1 Table C-8: Supplemental Alternative 1 Load Support Durations Under Different Winter Conditions

Supp. Alt. 1 Modelling Scenario	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ²⁴				
	-10°C (very cold winter day) ²⁵	-1.4°C (warmest winter in 10 years, as defined in footnote) ²⁶	+4.0°C (average Lower Mainland winter) ²⁷		
Planning (0 Bcf, 150 MMcf/d)	0 days and 0 hours	0 days and 0 hours	0 days and 0 hours		
Contingent (0.35 Bcf, 150 MMcf/d)	0 days and 2 hours	0 days and 5 hours	0 days and 7 hours		
Contingent w/T1A (0.75 Bcf, 150 MMcf/d)	0 days and 2 hours	0 days and 5 hours	0 days and 7 hours		

2

3 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan, 4 under average winter conditions, the support duration provided by Supplemental Alternative 1

5 (regardless of the scenario) would not be enough to execute a controlled shutdown. Therefore, a

6 failure occurring on some parts of T-South could be expected to result in an uncontrolled

7 shutdown in the Lower Mainland system, with attendant safety risks and service restoration

8 challenges.

9 Figures C-12 to C-14 below show the transient modelling outputs used to determine the load support duration values in the table above. The figures show the results at various temperatures (+4°C, -1.4°C and -10°C) for the <u>most-optimistic</u> contingent scenario, which assumes both the Base Plant is full and 0.4 Bcf of LNG inventory is on hand in Tilbury 1A (totalling 0.75 of LNG at Tilbury on the day of a winter T-South no-flow event). FEI explains how to interpret these figures in Section 3.2.2.1.2 of the Supplemental Evidence.

²⁴ This represents the approximate duration of full firm load support for customers in the Lower Mainland. Except for the 0 Bcf case which does not account for linepack, the analysis considers support from on-system LNG and linepack from the CTS. Linepack was not considered in the 0 Bcf case such that an understanding of the absolute baseline risk could be had. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

²⁵ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

²⁶ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from the Vancouver International Airport (YVR).

²⁷ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.

Figure C-12: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 1²⁸



2 3

1

Figure C-13: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 1



4

FORTIS BC[®]

²⁸ FEI's transient modelling for Supplemental Alternative 1 assumed not only that the full derated capacity of the Base Plant is present (0.35 Bcf), but also 0.4 Bcf in Tilbury 1A, for a total of 0.75 Bcf. There would be less LNG available if FEI had used any of this for peaking supply.

Figure C-14: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 1



3 5.1.1.2.2 GAS SUPPLY

1

2

Supplemental Alternative 1 would retain FEI's existing on-system peaking capabilities; however,
as explained below, this is predicated on the continued availability of peaking capacity and energy
at Tilbury when required, which is not technically viable.

7 The value of on-system LNG to a supply portfolio is the capacity it provides (i.e., the capability for 8 rapid send-out when load spikes), backed by enough energy (LNG) to last the duration of the 9 peak demand event. Peak loads already exceed Tilbury's capabilities. Further, FEI is entirely 10 dependent on the existing Base Plant regasification equipment to access LNG for peaking 11 capacity. As a result, Supplemental Alternative 1 only maintains the current peaking supply 12 capabilities for so long as both: (1) the Base Plant remains operational; and (2) Tilbury 1A is not 13 fully subscribed under RS 46, which would otherwise preclude FEI from continuing its existing 14 temporary practice of using 0.25 Bcf from Tilbury 1A in the Annual Contracting Plan.

Further, as discussed in Section 3.3.4 of the Supplemental Evidence, FEI could not replace the 15 16 Tilbury peaking supply resources currently included in the Annual Contracting Plan (150 MMcf/d 17 and 0.6 Bcf) in the market and does not have control over regional infrastructure expansions that, 18 if those expansions were to occur in the future, would entail significant annual gas supply costs for FEI customers. FEI's assessment, which is supported by the report of Mr. Raymond Mason, 19 20 an expert in the regional gas supply market who FEI retained to opine on the role of on-system 21 LNG storage in a supply portfolio, is that fully-contracted regional infrastructure would preclude 22 replacing Tilbury's existing peaking capabilities with high-capacity long-term commercial contracts for peaking supply arrangements. In the absence of on-system LNG at Tilbury, the only 23 24 way to secure dependable replacement peaking supply would be to participate in regional 25 infrastructure upgrades.

FORTIS BC[®]



1 5.1.1.2.3 BASE PLANT CHALLENGES

- 2 Supplemental Alternative 1 would not resolve the existing Base Plant's age-related challenges as
- 3 it assumes no capital upgrades at Tilbury. For the reasons described in Sections 3.3 of the
- 4 Supplemental Evidence, prolonging FEI's reliance on the end-of-life Base Plant would entail 5 increasing risk that peaking capacity and energy will be unavailable when required in normal
- 5 increasing risk that peaking capacity and energy will be unavailable when required in normal
- 6 conditions:

7 5.1.1.2.1 LEVELIZED TOTAL RATE IMPACT

- 8 The levelized total rate impact for Supplemental Alternative 1 is estimated to be 1.8 percent over 9 a 67-year analysis period. The levelized total rate impact is lower than Supplemental Alternative 10 9 because this alternative would involve no capital costs at Tilbury. However, once the 150 MMcf/d
- 11 and 0.6 Bcf of on-system LNG is no longer available, FEI would be forced to curtail customers in
- 12 normal operations which is reflected in the analysis as a gas supply savings.
- There would still be rate impacts associated with maintaining the existing 50 MMcf/d of pipeline capacity that is currently used for peaking gas supply. If and when replacement peaking supply is available on regional infrastructure to replace the lost 150 MMcf/d and 0.6 Bcf (assumed to be 2035) FEI would pay higher tolls for its full holdings on expanded regional infrastructure. Please refer to Section 3.3 of this Appendix for discussion on the gas supply cost assumptions and the
- 18 costs/savings used in the financial analysis for Supplemental Alternative 1.

19 5.1.1.2.2 FUTURE USE

FEI does not expect the Base Plant to still be in-service by 2050 as the facility would be 79 years old.

22 **5.1.2 Supplemental Alternative 2**

- In the Adjournment Decision, the BCUC raised the potential of increasing the regasification
 capacity of the Base Plant while retaining the existing tank:²⁹
- 25 FEI submits that "most of the existing Base Plant infrastructure is not adequately 26 sized for the volume of regasification required" and that "to increase its 27 regasification would be technically challenging and costly to the point where FEI 28 would not consider it to be a prudent investment." However, in the absence of an 29 assessment of the remaining life of the Base Plant and the guantum of the costs 30 that FEI describes as "other significant engineering and capital costs to ensure the 31 existing system could operate reliably under very different operating parameters," 32 we are not able to definitively determine the prudency of investing in an upgrade 33 to the existing regasification capacity to 800 MMcf/day. In addition, there is no 34 evidence concerning what level of regasification capacity could be added while still 35 remaining, in FEI's view, a prudent investment; or the duration and nature of a no-

²⁹ Adjournment Decision, p. 32.



- 1 flow event that the existing infrastructure (Base Plant and Tilbury Tank 1A) with 2 increased gasification could withstand.
- FEI also provides no information on whether, should the regassification capacity
 of the existing facility be increased, it would be compatible with a new tank, should
 one subsequently be approved and built.
- 6 We recommend that FEI consider the potential costs and benefits associated with 7 supplementing the existing storage assets at Tilbury with increased gasification. 8 We note that such an alternative provides a potential bridging mechanism to 9 enhance resiliency while allowing more time to understand the future of natural 10 gas demand and supply in the LML.
- 11 FEI confirms that, leaving aside whether the Project objectives are met, it is technically feasible 12 to construct additional regasification and connect it to the existing Tilbury 1A and Base Plant tank 13 infrastructure. As discussed in the Application, FEI evaluated numerous regasification 14 technologies and sizes and selected submerged combustion vessel technology at a unit size of 15 200 MMcf/d (which can be operated between 50 and 200 MMcf/d). Therefore, FEI could choose 16 to install either one (200 MMcf/d), two (400 MMcf/d) or three (600 MMcf/d) units initially. A fourth 17 unit, which would bring the total regasification capacity to 800 MMcf/d, would only be constructed 18 at the same time as a new tank, as that level of regasification would very rapidly deplete the 19 existing available tank volume, even assuming 0.4 Bcf of additional LNG supply from Tilbury 1A.
- As explained in Section 4.4 of the Supplemental Evidence, FEI eliminated Supplemental Alternative 2 in Step 2 of the screening process. However, to be responsive to the BCUC's commentary in the Adjournment Decision, FEI has assessed this alternative against the five criteria utilized in Step 3 of the screening process as part of this Appendix.
- 24 Supplemental Alternative 2 includes the following:
- Replacing the existing Base Plant regasification units with 400 MMcf/d of regasification;
- Continuing to rely on a non-refurbished Base Plant tank that has reached end-of-life and is operating at a lower maximum capacity (0.35 Bcf), and leaving unaddressed the seismic, environmental and flooding issues inherent in the original Base Plant design;
- There continuing to be no resiliency reserve; that is, from a planning perspective, all of the
 LNG at Tilbury would still be allocated to either gas supply or RS 46 LNG sales, such that
 those volumes may or may not be present upon a no-flow event;
- Changing how FEI operates existing Tilbury 1A liquefaction to replenish consumed LNG faster, so as to increase the potential for LNG to be present on the day of a no-flow event (i.e., 0.35 Bcf in the Base Plant tank and 0.4 Bcf in Tilbury 1A despite RS 46 LNG sales).³⁰

³⁰ This assumption addresses the following BCUC commentary in the Adjournment Decision, p. 29: "With additional liquefaction the average storage volume could be increased thereby providing additional capacity available for resiliency purposes. While there is evidence that this would be more costly than the current operational strategy, the



- From the gas supply perspective, continue meeting the peaking supply requirement of 1
 Bcf and 200 MMcf/d by continuing to rely on the Base Plant, plus continuing to hold 50
 MMcf/d of year-round pipeline capacity. The latter is still necessary despite the new
 regasification of 400 MMcf/d because the amount of LNG is a limitation;
- 5 When the existing Base Plant tank must be operated at further reduced levels or is no longer usable, or Tilbury 1A LNG is unavailable (assumed to be 2030 for the purpose of 6 7 the financial analysis), the new regasification will be of limited use. FEI would therefore lose the peaking capabilities of 0.6 Bcf and 400 MMcf/d from Tilbury and would be unable 8 9 to replace it with capacity on regional infrastructure. As such, FEI would still hold the 10 existing 50 MMcf/d of market pipeline capacity but would likely have to curtail firm loads 11 in normal operations by up to 150 MMcf/d (the original regasification level currently 12 included in the ACP) until future regional infrastructure upgrades occur; and
- If and when regional infrastructure upgrades occur (assumed to be 2035 for the financial analysis), it would offer the potential for FEI to end curtailments by holding gas supply contracts for 1.0 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that includes the costs of the upgrades.

17 5.1.2.1 Modelling Parameters

Table C-9 below sets out the resiliency and gas supply modelling parameters for theSupplemental Alternative 2 scenarios.

The planning scenario uses the maximum regasification (400 MMcf/d), paired with the volume of dependable LNG for resiliency (0 Bcf – no resiliency reserve) and, for peaking gas supply, the amount included in the Annual Contracting Plan (0.6 Bcf).

FEI also modelled two "contingent" (i.e., non-dependable) scenarios based on more optimistic assumptions about how much LNG is present at Tilbury on the day of a no-flow event. These scenarios are described in Section 4 of this Appendix.

cost benefit analysis is incomplete. In order to fully evaluate a potential role of Tilbury Tank 1A in FEI's resiliency portfolio, a more fulsome analysis is required."



Supp. Alt. 2 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹	
Supplemental Alternative 2 (Planning)	New Regasification – 400 MMcf/d (No Resiliency Reserve) ²	0 Bcf at 400 MMcf/d	0.6 Bcf at 400 MMcf/d	
Supplemental Alternative 2 (Contingent)	Assume Partial LNG Availability on Day of No-Flow ³	0.35 Bcf at 400 MMcf/d	0.6 Bcf at 400 MMcf/d	
Supplemental Alternative 2 (Contingent w/T1A)	Assume Base Plant Fully Available plus 0.4 Bcf in Tilbury 1A on Day of No-Flow	0.75 Bcf at 400 MMCf/d	0.6 Bcf at 400 MMcf/d	

Table C-9: Supplemental Alternative 2 Planning and Contingent Modelling Scenarios

2
3

1

Notes to Table:

- Per the ACP, FEI allocates 0.6 Bcf from Tilbury for gas supply portfolio purposes, with a 0.25 Bcf portion temporarily
 obtained from Tilbury 1A. Actual volumes will decline with use during winter, including if LNG volumes were used
 for resiliency to respond to a no-flow event.
- When there is no resiliency reserve, on a planning basis there is no dependable LNG available on occurrence of no-flow event. Hence the resiliency modelling parameter is 0 Bcf.
- Reduced volume available could be attributable to either: (a) Tilbury 1A volumes being unavailable due to RS 46 sales, leaving a maximum of 0.35 Bcf in the Base Plant tank at currently reduced fill levels; or (b) partial consumption for peaking gas supply of the 0.6 Bcf currently allocated to gas supply at Tilbury (which is currently comprised of 0.35 Bcf from the Base Plant plus 0.25 Bcf Tilbury 1A). The former is expected to occur in the future, and the latter is common over the course of a typical winter.

14 *5.1.2.2* Summary of Information and Analysis

FEI investigated Supplemental Alternative 2 with respect to the criteria discussed in Section 4.2.2.3 of the Supplemental Evidence. Supplemental Alternative 2 did not pass the Step 2 screen since it would not retain FEI's existing on-system peaking supply capabilities³¹ due to the expected increase in RS 46 sales and the age-related seismic challenges associated with the Base Plant tank, which creates a risk that the operating level of the tank could be further reduced.³² However, for completeness, FEI assesses Supplemental Alternative 2 based on the five criteria utilized in Step 3 of the alternatives analysis.

- The following bullets provide a high-level description of how this alternative performs against the criteria, followed by sections providing additional details.
- **Resiliency (No Impact):** Supplemental Alternative 2 would not include a resiliency reserve and, even assuming some LNG inventory is present at Tilbury on the day of a noflow event, would only provide very limited incremental mitigation against a winter T-South

³¹ 0.6 Bcf and 150 MMcf/d of regasification capacity.

³² See Section 4.4.1 of the Supplemental Evidence for further discussion regarding why Supplemental Alternative 2 did not pass the Step 2 screen.



- no-flow event relative to the Supplemental Alternative 1 (Planning) scenario. A failure
 would result in similar consequences as the Supplemental Alternative 1 (Planning)
 scenario.
- Gas Supply (No Impact): Supplemental Alternative 2 would only maintain the current peaking supply capabilities for so long as: (a) Tilbury 1A is not fully subscribed under RS 46; and (b) the Base Plant tank can continue to operate at its current operating level without having to be further reduced. This is unlikely to occur beyond 2030. Thereafter, FEI would lose a significant portion of its dependable peaking resources unless and until a replacement LNG facility or costly regional infrastructure upgrades could be constructed.
- Base Plant Challenges (Medium Positive Impact): The replacement of the Base Plant's regasification equipment would address its reliability issues; however, the age-related challenges inherent in the Base Plant tank would remain unresolved.
- Levelized Total Rate Impact (Low Negative Impact): Supplemental Alternative 2 would have a levelized total rate impact of 3.3 percent over a 67-year analysis period. This is lower than Supplemental Alternative 9 since it only involves replacement of the regasification equipment, but it assumes FEI would be curtailing firm customers in normal operations for several years instead of incurring peaking gas costs.
- Future Use (Medium Negative Impact): FEI does not expect the Base Plant tank to still
 be in-service by 2050 as the facility would be 79 years old. Additionally, FEI expects
 Tilbury 1A to be sold out from RS 46 sales. This creates a stranded asset risk for the new
 regasification equipment, which would not be useful without the Base Plant tank or
 available LNG from Tilbury 1A.
- 23 FEI discusses the analysis for each of the five criteria in further detail below.

24 5.1.2.2.1 RESILIENCY (NO IMPACT)

Supplemental Alternative 2 would not include a resiliency reserve and, even assuming some LNG
 inventory is present at Tilbury on the day of a no-flow event, would only provide very limited
 incremental mitigation against a winter T-South no-flow event. FEI addresses the resiliency
 benefit of the Supplemental Alternative 2 scenarios below.

29 Supplemental Alternative 2 (Planning) would improve the regasification capacity relative • 30 to the Supplemental Alternative 1, but would not fully address the existing regasification constraint at Tilbury. This is because, at average Lower Mainland winter temperature 31 32 (+4°C), the send out capacity of 400 MMcf/d remains undersized for the system demand. 33 Further, without a resiliency reserve, all LNG would continue to be allocated for planning 34 purposes to either gas supply or RS 46 LNG sales, such that it may or may not be present 35 on the day of a winter T-South no-flow event. Therefore, the system would still fail on the day of a winter T-South no-flow event, resulting in similar consequences as the 36 Supplemental Alternative 1 (Planning) scenario. 37



1 Supplemental Alternative 2 would also only provide limited incremental mitigation against • 2 a winter T-South no-flow event under either of the contingent scenarios, which both 3 assume that some LNG inventory is present on the day of a winter T-South no-flow event. 4 Based on Exponent's analysis, the customer outage risk would nonetheless remain 5 significant under these optimistic contingent scenarios. In particular, the system would still 6 fail on Day 1 (cold weather of -1.4°C) or Day 2 (at an average winter temperature of +4°C) 7 of a winter no-flow event, resulting in similar consequences as the Supplemental 8 Alternative 1 (Planning) scenario.

9 These results are confirmed in Exponent's figure (reproduced in Section 2.5.1 above)
10 summarizing the annual expected loss reduction associated with each Supplemental Alternative
11 relative to the Supplemental Alternative 1 (Planning) scenario.

12 The following table summarizes how long Supplemental Alternative 2, in combination with FEI's

13 other existing capabilities, will be able to sustain the Lower Mainland load following a winter T-

14 South no-flow event. FEI has consistently modelled three temperatures that are reflective of local

15 conditions, since FEI's load increases as temperatures decrease.

16 17

Table C-10: Supplemental Alternative 2 Load Support Durations Under Different Winter Conditions

Supp. Alt. 2	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ³³				
Modelling Scenario	-10°C (very cold winter day) ³⁴	-1.4°C (warmest winter in 10 years, as defined in footnote) ³⁵	+4.0°C (average Lower Mainland winter) ³⁶		
Planning (0 Bcf, 400 MMcf/d)	0 days and 0 hours	0 days and 0 hours	0 days and 0 hours		
Contingent (0.35 Bcf, 400 MMcf/d)	0 days and 5 hours	0 days and 12 hours	1 day and 1 hour		

³³ This represents the approximate duration of full firm load support for customers in the Lower Mainland. Except for the 0 Bcf case which does not account for linepack, the analysis considers support from on-system LNG and linepack from the CTS. Linepack was not considered in the 0 Bcf case such that an understanding of the absolute baseline risk could be had. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

³⁴ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

³⁵ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

³⁶ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.

Supp. Alt. 2	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ³³				
Modelling Scenario	-10°C (very cold winter day) ³⁴	-1.4°C (warmest winter in 10 years, as defined in footnote) ³⁵	+4.0°C (average Lower Mainland winter) ³⁶		
Contingent w/T1A (0.75 Bcf, 400 MMcf/d)	0 days and 5 hours	0 days and 12 hours	1 day and 19 hours		

2 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

3 under average winter conditions, the support duration provided by the contingent scenarios for

4 Supplemental Alternative 2 may or may not be enough to execute a controlled shutdown.

5 Therefore, a failure occurring on some parts of T-South could result in a controlled or uncontrolled

6 shutdown in the Lower Mainland system. The Supplemental Alternative 2 (Planning) scenario

7 does not provide a support duration, and therefore, FEI expects a failure occurring on some parts

8 of T-South would result in an uncontrolled shutdown.

9 The resiliency support provided by each Supplemental Alternative 2 scenario is discussed further 10 below.

11 Supplemental Alternative 2 (Planning) – 0 Bcf, 400 MMcf/d

As shown in Table C-10 above, the Supplemental Alternative 2 (Planning) scenario assumes there is no LNG inventory present on the day of a winter T-South no-flow event, as there is no resiliency reserve. Therefore, this scenario would provide no resiliency support (i.e., the result is

15 the same as the Supplemental Alternative 1 (Planning) scenario).

16 Supplemental Alternative 2 (Contingent) – 0.35 Bcf, 400 MMcf/d

17 Supplemental Alternative 2 (Contingent) assumes 0.35 Bcf of LNG inventory is present on the

18 day of a winter T-South no-flow event, which amounts to approximately 58 percent of the planning

19 LNG inventory for gas supply purposes. The reduced LNG volume available could be attributable

- 20 to either:
- Tilbury 1A volumes being unavailable due to RS 46 sales, leaving a maximum of 0.35 Bcf
 in the Base Plant tank at currently reduced fill levels; or
- Partial consumption for peaking gas supply of the 0.6 Bcf currently allocated to gas supply at Tilbury (which is currently comprised of 0.35 Bcf from the Base Plant plus 0.25 Bcf Tilbury 1A).

26 Because of the higher regasification capacity, this contingent scenario provides an increased 27 resiliency benefit, in terms of duration of load support following a winter T-South no-flow event,

across all temperature conditions considered relative to both the Supplemental Alternative 1

29 (Planning) and (Contingent) scenarios. However, the benefit is mainly observed under average



- winter conditions (+4°C) as, similar to Supplemental Alternative 1 (Contingent), under the colder temperature conditions of -10°C and -1.4°C, customer outages would still occur on the day of the no-flow event. This is due to the 400 MMcf/d regasification capacity being insufficient at the colder temperature conditions. Even at +4°C the 400 MMcf/d regasification capacity is insufficient for the entire Lower Mainland load. However, the capacity shortfall is smaller than in Supplemental Alternative 1 (Contingent) and thus Supplemental Alternative 2 (Contingent) shows more improvement at this temperature condition when compared to the colder temperature conditions
- 8 of -10°C and -1.4°C.
- Figures C-15 to C-17 below show the transient modelling outputs FEI used to determine the load
 support duration values for this scenario. As shown in Figure C-15, under average winter
 conditions (+4°C), the primary constraint is insufficient storage capacity.³⁷ However, as seen in
 Figures C-16 and C-17, the constraint switches back to being insufficient regasification under
- 13 colder temperatures.³⁸

Figure C-15: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 2 – 0.35bcf & 400MMcf/d



16

³⁷ The Tilbury facility sendout flow rate (yellow line) drops to zero before the first customer outage occurs.

³⁸ The Tilbury facility sendout flow rate (yellow line) remains above zero at the time of the first customer outage, indicating that there is still volume left in the tank when customer outages occur.



Figure C-16: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 2 – 0.35bcf & 400MMcf/d



3 4

5

Figure C-17: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 2 – 0.35bcf & 400MMcf/d



6

7 Supplemental Alternative 2 (Contingent w/T1A) – 0.75 Bcf, 400 MMcf/d

8 Supplemental Alternative 2 (Contingent w/T1A) assumes 0.75 Bcf of LNG inventory is present on
9 the day of a winter T-South no-flow event as follows:

The Base Plant tank being full at its current reduced operating capacity (0.35 Bcf) and
 Tilbury 1A having 0.4 Bcf of LNG inventory on hand on the day of the winter T-South no flow event.



- 1 Because of the higher regasification capacity, this contingent scenario would provide an increased
- 2 resiliency benefit in terms of the duration of load support relative to Supplemental Alternative 1
- 3 (Contingent w/T1A).

5

6

7

8

9

10

- Compared to Supplemental Alternative 2 (Contingent), under average winter conditions (+4°C), the additional available LNG volume would improve the load support duration to 1 day and 19 hours (an increase of 18 hours compared to Supplemental Alternative 2 (Contingent)). However, the support durations under the -1.4°C and -10°C temperature conditions are the same as for the Supplemental Alternative 2 (Contingent) scenario. This result reflects the constraining factor at these colder temperatures, insufficient regasification capacity, which is not remedied by increasing the available storage volume from 0.35 Bcf to 0.75 Bcf.
- 11 Figures C-18 to C-20 below show the transient modelling outputs FEI used to determine the load
- 12 support duration values for this scenario.

13Figure C-18: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 2 –140.75bcf & 400MMcf/d



15



Figure C-19: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 2 – 0.75bcf & 400MMcf/d



3 4

5

Figure C-20: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 2 – 0.75bcf & 400MMcf/d



6

7 5.1.2.2.2 GAS SUPPLY (NO IMPACT)

8 Supplemental Alternative 2 would not retain FEI's existing on-system peaking capabilities and, 9 therefore, did not pass the Step 2 screen because of: (1) the expected increase in RS 46 sales 10 which would prevent FEI from supplementing the operating volume of the Base Plant tank from 11 Tilbury 1A; and (2) the increasing risk that even the reduced Base Plant tank volumes will be 12 unavailable. FEI addresses each reason below.



- First, Supplemental Alternative 2 would only maintain the current peaking supply capabilities for
 so long as Tilbury 1A is not fully subscribed under RS 46. Tilbury 1A currently supplements the
- 3 current operating volume of the Base Plant tank (0.35 Bcf) with an additional 0.25 Bcf of LNG,
- 4 such that the required LNG volume for peaking supply is available at Tilbury. Section 4.2.2.1
- 5 above discusses how FEI expects RS 46 demand to grow, particularly now that the Tilbury Jetty
- 6 has received the necessary regulatory approvals to proceed. Once Tilbury 1A is fully subscribed,
- 7 it would be difficult for FEI replace the lost gas supply support in the market as dependable
- 8 peaking resources are not available in the market.
- 9 Second, Supplemental Alternative 2 would only maintain the current peaking supply capabilities 10 if the Base Plant remains in operation at its current capacity. While the regasification challenges
- 11 are resolved in Supplemental Alternative 2, the original 1971 Base Plant tank would still be relied
- 12 on for gas supply. If the operating level of the tank were further reduced due to the seismic
- 13 challenges that are inherent in the Base Plant design, FEI would again need to replace the lost
- 14 supply by acquiring the equivalent resources on regional pipelines or storage.
- As discussed in Section 3.3.4 of the Supplemental Evidence, FEI could not replace the Tilbury
 peaking supply resources currently included in the Annual Contracting Plan (150 MMcf/d and 0.6
 Bcf) in the market and does not have control over regional infrastructure expansions that, if those
- 18 expansions were to occur in the future, would entail very significant annual gas supply costs for
- 19 FEI customers.

20 5.1.2.2.3 BASE PLANT CHALLENGES (MEDIUM POSITIVE IMPACT)

The replacement of the Base Plant's regasification equipment would address the reliability of the send-out equipment. However, this alternative would not address the Base Plant tank, which has reached end-of-life and, as such, the seismic, environmental and flooding issues inherent in the existing Base Plant tank design discussed in Section 3.3.2 of the Supplemental Evidence would remain unresolved.

26 5.1.2.2.1 LEVELIZED TOTAL RATE IMPACT (LOW NEGATIVE IMPACT)

- The levelized total rate impact for Supplemental Alternative 2 is estimated to be 3.3 percent over a 67-year analysis period.
- 29 This reflects the estimated capital cost of \$391.5 million to install the 400 MMcf/d of new 30 regasification, operating costs over the expected life of the new assets, and the costs for gas 31 supply from the regional market, if available, to meet FEI's full peaking gas requirement of 1.0 Bcf 32 and 200 MMcf/d once there is no longer sufficient LNG available for peaking gas supply. For years 33 when it is not possible to replace the peaking gas supply level currently provided by the Base 34 Plant, FEI would have to curtail firm customers under normal operations. Please refer to Section 35 3.3 of this Appendix for details on the gas supply cost assumptions and the costs/savings used 36 in the financial analysis for Supplemental Alternative 2.



1 5.1.2.2.2 FUTURE USE (MEDIUM NEGATIVE IMPACT)

- 2 Supplemental Alternative 2 creates a stranded asset risk for the new regasification equipment,
- 3 which would not be useful without the Base Plant tank or available LNG from Tilbury 1A.
- FEI does not expect the Base Plant tank to still be in-service by 2050 as the facility would be 79
 years old and has reached end-of-life. Further, as Tilbury 1A will no longer be available due to an
 expected increase in RS 46 sales, unless a new tank is built, the regasification equipment would
- 7 become a stranded asset.
- 8 The question of whether or not the facility contemplated under a Supplemental Alternative would 9 be used and useful must be distinguished from whether or not it would deliver on the Project 10 objectives. As discussed above, Supplemental Alternative 2 would not deliver improved resiliency 11 against a winter T-South no-flow event, nor would it ensure that FEI is able to continue serving 12 customers in normal operations (i.e., maintaining the critical gas supply function and other
- 13 operational benefits provided by the Base Plant).

14 **5.1.3 Supplemental Alternative 3**

As explained in Section 4.4 of the Supplemental Evidence, FEI eliminated Supplemental Alternative 3 in Step 2 of the screening process. However, to be responsive to the BCUC's commentary in the Adjournment Decision, FEI has assessed this alternative against the five criteria utilized in Step 3 of the screening process as part of this Appendix.

- 19 Supplemental Alternative 3 includes the following:
- Replacing the existing Base Plant regasification units with 600 MMcf/d of regasification;
- Continuing to rely on a non-refurbished Base Plant tank that is operating at a lower maximum capacity (0.35 Bcf), and leaving unaddressed the seismic, environmental and flooding issues inherent in the original Base Plant design;
- There continuing to be no resiliency reserve; that is, from a planning perspective, all of the
 LNG at Tilbury would still be allocated to either gas supply or RS 46 LNG sales, such that
 those volumes may or may not be present upon a no-flow event;
- Changing how FEI operates existing Tilbury 1A liquefaction to replenish consumed LNG faster, so as to increase the potential for LNG to be present on the day of a no-flow event (i.e., 0.35 Bcf in the Base Plant tank and 0.4 Bcf in Tilbury 1A despite RS 46 LNG sales);³⁹
- From the gas supply perspective, continue meeting the peaking supply requirement of 1
 Bcf and 200 MMcf/d by continuing to rely on the Base Plant, plus continuing to hold 50

³⁹ This assumption addresses the following BCUC commentary in the Adjournment Decision, p. 29: "With additional liquefaction the average storage volume could be increased thereby providing additional capacity available for resiliency purposes. While there is evidence that this would be more costly than the current operational strategy, the cost benefit analysis is incomplete. In order to fully evaluate a potential role of Tilbury Tank 1A in FEI's resiliency portfolio, a more fulsome analysis is required."


- 1 MMcf/d of year-round pipeline capacity. The latter is still necessary despite the new 2 regasification of 600 MMcf/d because the amount of LNG is a limitation;
- 3 When the existing Base Plant tank must be operated at further reduced levels or is no 4 longer usable, or Tilbury 1A LNG unavailable (assumed to be 2030 for the purpose of the 5 financial analysis), the new regasification will be of limited use. FEI would therefore lose the peaking capabilities of 0.6 Bcf and 600 MMcf/d from Tilbury and would be unable to 6 7 replace it with capacity on regional infrastructure. As such, FEI would still hold the existing 8 50 MMcf/d of market pipeline capacity but would likely have to curtail firm loads in normal 9 operations by up to 150 MMcf/d (the original regasification level currently included in the ACP) until future regional infrastructure upgrades occur; and 10
- If and when regional infrastructure upgrades occur (assumed to be 2035 for the financial analysis), it would offer the potential for FEI to end curtailments by holding gas supply contracts for 1.0 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that includes the costs of the upgrades.

15 This alternative is the same as Supplemental Alternative 2, except the regasification capacity has

been increased from 400 MMcf/d to 600 MMcf/d; therefore, many of the same considerations

17 outlined in Supplemental Alternative 2 apply here as well.

18 5.1.3.1 Modelling Parameters

Table C-11 below sets out the resiliency and gas supply evaluation parameters for theSupplemental Alternative 3 scenarios.

The planning scenario uses the maximum regasification (600 MMcf/d), paired with the volume of dependable LNG for resiliency (0 Bcf – no resiliency reserve) and, for peaking gas supply, the amount included in the ACP (0.6 Bcf).

24 FEI also modelled two "contingent" (i.e., not dependable) resiliency scenarios based on more

optimistic assumptions about how much LNG is present at Tilbury on the day of a no-flow event.
These scenarios are described in Section 4 of this Appendix.



Supp. Alt. 3 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 3 (Planning)	New Regasification – 600 MMcf/d (No Resiliency Reserve) ²	0 Bcf at 600 MMcf/d	0.6 Bcf at 600 MMcf/d
Supplemental Alternative 3 (Contingent)	Assume Partial LNG Availability on Day of No-Flow ³	0.35 Bcf at 600 MMcf/d	0.6 Bcf at 600 MMcf/d
Supplemental Alternative 3 (Contingent w/T1A)	Assume Base Plant Fully Available plus 0.4 Bcf in Tilbury 1A on Day of No-Flow	0.75 Bcf at 600 MMcf/d	0.6 Bcf at 600 MMcf/d

Table C-11: Supplemental Alternative 3 Planning and Contingent Modelling Scenarios

2	
3	

1

<u>Notes to Table:</u>

- Per the ACP, FEI allocates 0.6 Bcf from Tilbury for gas supply portfolio purposes, with a 0.25 Bcf portion temporarily
 obtained from Tilbury 1A. Actual volumes will decline with use during winter, including if LNG volumes were used
 for resiliency to respond to a no-flow event.
- When there is no resiliency reserve, on a planning basis there is no dependable LNG available on occurrence of no-flow event. Hence the resiliency modelling parameter is 0 Bcf.
- Reduced volume available could be attributable to either (a) Tilbury 1A volumes being unavailable due to RS 46 sales, leaving a maximum of 0.35 Bcf in the Base Plant tank at currently reduced fill levels; or (b) partial consumption for peaking gas supply of the 0.6 Bcf currently allocated to gas supply at Tilbury (which is currently comprised of 0.35 Bcf from the Base Plant plus 0.25 Bcf Tilbury 1A). The former is expected to occur in the future, and the latter is common over the course of a typical winter.

14 *5.1.3.2* Summary of Information and Analysis

FEI investigated Supplemental Alternative 3 with respect to the criterion discussed in Section 4.2 of the Supplemental Evidence. Supplemental Alternative 3 did not pass the Step 2 screen since it would not retain FEI's existing on-system peaking supply capabilities⁴⁰ due to the expected increase in RS 46 sales and the age-related seismic challenges associated with the Base Plant tank which create a risk that the operating level of the tank could be further reduced. However, for completeness, FEI assesses Supplemental Alternative 3 based on the five criteria utilized in Step 3 of the alternatives analysis.

- The following bullets provide a high-level description of how this alternative performs against the criteria, followed by sections providing additional details.
- Resiliency (No Impact): Supplemental Alternative 3 would not include a resiliency reserve. Even assuming some LNG inventory is present at Tilbury on the day of a no-flow event, it would only provide very limited incremental mitigation against a winter T-South no-flow event relative to the Supplemental Alternative 1 (Planning) scenario. It would

⁴⁰ 0.6 Bcf and 150 MMcf/d of regasification capacity.



- provide more mitigation than the Supplemental Alternative 2 scenarios at below-average
 winter temperatures. A failure would have similar consequences as the Supplemental
 Alternative 1 (Planning) scenario.
- Gas Supply (No Impact): Supplemental Alternative 3 would only maintain the current peaking supply capabilities for so long as: (a) Tilbury 1A is not fully subscribed under RS 46; and (b) the Base Plant tank does not require a further reduced fill elevation due to its seismic design challenges that are inherent in the Base Plant design. Thereafter, FEI would lose a significant portion of its dependable peaking resources unless and until a replacement LNG facility or costly regional infrastructure upgrades could be constructed.
- Base Plant Challenges (Medium Positive Impact): The replacement of the Base Plant regasification equipment would address its reliability issues; however, the age-related challenges inherent in the Base Plant tank would remain unresolved.
- Levelized Total Rate Impact (Low Negative Impact): Supplemental Alternative 3 would have a levelized total rate impact of 3.4 percent over a 67-year analysis period. This is lower than Supplemental Alternative 9 since it only involves replacement of the regasification equipment, but it assumes FEI would be curtailing firm customers in normal operations for several years instead of incurring peaking gas costs.
- Future Use (Medium Negative Impact): FEI does not expect the Base Plant tank to still
 be in-service by 2050 as the facility would be 79 years old. Additionally, FEI expects
 Tilbury 1A to be sold out from RS 46 sales. This creates a stranded asset risk for the new
 regasification equipment, which would not be useful without the Base Plant tank or
 available LNG from Tilbury 1A.
- 23 FEI discusses each of the five analysis criteria in further detail below.

24 5.1.3.2.1 RESILIENCY (NO IMPACT)

Supplemental Alternative 3 would not include a resiliency reserve and, even assuming some LNG
 inventory is present at Tilbury on the day of a no-flow event, would only provide very limited
 incremental mitigation against a winter T-South no-flow event, albeit more than the Supplemental
 Alternative 2 scenarios under below-average winter temperatures. FEI addresses the resiliency
 benefit of the Supplemental Alternative 3 scenarios below.

- 30 The Supplemental Alternative 3 (Planning) scenario goes further than Supplemental • Alternative 2 towards addressing the regasification constraint at Tilbury by increasing it to 31 32 600 MMcf/d. Further, without a resiliency reserve, all LNG would continue to be allocated 33 for planning purposes to either gas supply or RS 46 LNG sales, such that it may or may 34 not be present on the day of a winter T-South no-flow event. Therefore, from a planning 35 perspective, the system would still fail on the day of a winter T-South no-flow event, 36 resulting in the same consequences as the Supplemental Alternative 1 (Planning) 37 scenario.
- Supplemental Alternative 3 would also only provide limited incremental mitigation against
 a winter T-South no-flow event under either of the contingent scenarios, which both



assume that some LNG inventory is present on the day of a winter T-South no-flow event. 1 2 For example, while the most optimistic Supplemental Alternative 3 (Contingent w/T1A) 3 scenario, which assumes that 0.75 Bcf of LNG inventory is present on the day of the no-4 flow event, would provide limited incremental mitigation against a winter T-South no-flow 5 event relative to Supplemental Alternative 2 under average Lower Mainland winter 6 temperatures (+4°C), the customer outage risk would nonetheless remain significant. In 7 particular, the system would still fail on Day 1 (extreme cold weather of -10°C) or Day 2 8 (at -1.4°C and +4°C) of a winter no-flow event, resulting in similar consequences as the Supplemental Alternative 1 (Planning) scenario. 9

These results are confirmed in Exponent's figure (reproduced in Section 2.5.1 above)
summarizing the annual expected loss reduction associated with each Supplemental Alternative
relative to the Supplemental Alternative 1 (Planning) scenario.

The following table summarizes how long Supplemental Alternative 3, in combination with FEI's other existing capabilities, would be able to sustain the Lower Mainland load following a winter T-South no-flow event. FEI has consistently modelled three temperatures that are reflective of local conditions, since FEI's load increases as temperatures decrease.

17 18

Table C-12: Supplemental Alternative 3 Load Support Durations Under Different Winter Conditions

Supp. Alt. 3	Approximate Time	Until Customers in the Lower Mainland Begin Losing Service ⁴¹		
Modelling Scenario	-10°C (very cold winter day) ⁴²	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁴³	+4.0°C (average Lower Mainland winter) ⁴⁴	
Planning (0 Bcf, 600 MMcf/d)	0 days and 0 hours	0 days and 0 hours	0 days and 0 hours	
Contingent (0.35 Bcf, 600 MMcf/d)	0 days and 7 hours	0 days and 18 hours	1 day and 1 hour	

⁴¹ This represents the approximate duration of full firm load support for customers in the Lower Mainland. Except for the 0 Bcf case which does not account for linepack, the analysis considers support from on-system LNG and linepack from the CTS. Linepack was not considered in the 0 Bcf case such that an understanding of the absolute baseline risk could be had. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁴² Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁴³ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁴⁴ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



Supp. Alt. 3	Approximate Time	Until Customers in the Lower Mainland Begin Losing Service ⁴¹		
Modelling Scenario	-10°C (very cold winter day) ⁴²	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁴³	+4.0°C (average Lower Mainland winter) ⁴⁴	
Contingent w/T1A (0.75 Bcf, 600 MMcf/d)	0 days and 7 hours	1 day and 10 hours	1 day and 22 hours	

2 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

3 under average winter conditions, the support duration provided by the contingent scenarios for

4 Supplemental Alternative 3 may or may not be enough to execute a controlled shutdown.

5 Therefore, a failure occurring on some parts of T-South may result in a controlled or uncontrolled

6 shutdown in the Lower Mainland system. The Supplemental Alternative 3 (Planning) scenario

7 does not provide a support duration, and therefore, FEI expects a failure occurring on some parts

8 of T-South would result in an uncontrolled shutdown.

9 The resiliency support provided by each Supplemental Alternative 3 scenario is discussed further

10 below.

11 Supplemental Alternative 3 (Planning) - 0 Bcf, 600 MMcf/d

As shown in the table above, the Supplemental Alternative 3 (Planning) scenario assumes there

13 is no LNG inventory present on the day of a winter T-South no-flow event, as there is no resiliency

14 reserve. Therefore, this scenario would provide no more resiliency support than Supplementary

15 Alternative 1 (Planning).

16 Supplemental Alternative 3 (Contingent) - 0.35 Bcf, 600 MMcf/d

Supplemental Alternative 3 (Contingent) assumes 0.35 Bcf of LNG inventory is present on the
day of a winter T-South no-flow event, which amounts to approximately 58 percent of the planning
LNG inventory for gas supply purposes. The reduced LNG volume available could be attributable
to either:

- Tilbury 1A volumes being unavailable due to RS 46 sales, leaving a maximum of 0.35 Bcf in the Base Plant tank at currently reduced fill levels; or
- Partial consumption for peaking gas supply of the 0.6 Bcf currently allocated to gas supply at Tilbury (which is currently comprised of 0.35 Bcf from the Base Plant plus 0.25 Bcf Tilbury 1A).

This contingent scenario would provide an increased resiliency benefit in terms of duration of load support relative to Supplemental Alternative 1 (Contingent). However, compared to Supplemental Alternative 2 (Contingent), Supplemental Alternative 3 (Contingent) would provide the same support duration under average winter conditions (+4°C), as both would be constrained by the available volume of LNG under this condition. Under the colder temperature conditions of -10°C



and -1.4°C, Supplemental Alternative 3 (Contingent) would provide slightly longer support
 durations than Supplemental Alternative 2 (Contingent).

Figures C-21 to C-23 show the modelling outputs used to determine the load support duration values for this scenario. As shown in Figure C-21 and Figure C-22, at the +4°C and -1.4°C temperature conditions the primary constraint is insufficient storage capacity.⁴⁵ However, as seen in Figure C-23, the constraint switches back to being insufficient regasification under colder temperatures.

Figure C-21: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 3 – Contingent 0.35bcf & 600MMcf/d



10

⁴⁵ The Tilbury facility send-out flow rate (yellow line) drops to zero before the first customer outage occurs.



Figure C-22: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 3 – Contingent 0.35bcf & 600MMcf/d



3 4

5

Figure C-23: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 3 – Contingent 0.35bcf & 600MMcf/d



6

7 Supplemental Alternative 3 (Contingent w/T1A) – 0.75 Bcf, 600 MMcf/d

8 Supplemental Alternative 3 (Contingent w/T1A) assumes 0.75 Bcf of LNG inventory is present on

9 the day of a winter T-South no-flow event as follows:



• The Base Plant tank being full at its current operating capacity (0.35 Bcf) and Tilbury 1A having 0.4 Bcf of LNG inventory on hand on the day of the winter T-South no-flow event.

3 This contingent scenario would provide an increased resiliency benefit in terms of duration of load 4 support relative to Supplemental Alternative 1 (Contingent w/T1A). Further, compared to 5 Supplemental Alternative 2 (Contingent w/T1A) under average winter conditions (+4°C), the 6 additional regasification capacity would improve the load support duration to 1 day and 22 hours 7 (an increase of 3 hours compared to Supplemental Alternative 2 (Contingent w/T1A)). Increasing 8 the regasification from 400 MMcf/d to 600 MMcf/d also improves the support duration under the -9 1.4°C temperature condition. At this temperature, Supplemental Alternative 3 (Contingent w/T1A) 10 provides a support duration that is 22 hours longer than Supplemental Alternative 2 (Contingent 11 w/T1A). However, the support duration at the -10°C temperature condition is only 2 hours longer 12 than that provided by Supplemental Alternative 2 (Contingent w/T1A).

13 When compared to Supplemental Alternative 3 (Contingent) at the +4°C and -1.4°C temperature 14 conditions (i.e., the conditions where both modelling scenarios are constrained by the available 15 volume of LNG), Supplemental Alternative 3 (Contingent w/ T1A) provides a longer load support 16 duration because of the higher available LNG volumes. At the -10°C temperature condition the 17 support durations between the two modelling scenarios are the same. This is because for both 18 modelling scenarios the constraint at this temperature condition is due to the regasification 19 capacity, which is not remedied by increasing the available LNG volume from 0.35 Bcf to 0.75 20 Bcf. This result reflects that the constraining factor at -10°C remains insufficient regasification 21 capacity for Supplemental Alternative 3.

Figures C-24 to C-26 below show the modelling outputs used to determine the load support duration values in the table above for the second contingent scenario.



Figure C-24: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 3 – Contingent 0.75bcf & 600MMcf/d



3 4

5

Figure C-25: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 3 – Contingent 0.75bcf & 600MMcf/d



6







3

4 5.1.3.2.2 GAS SUPPLY (NO IMPACT)

5 The same assessment from Supplemental Alternative 2 applies to Supplemental Alternative 3.

6 Supplemental Alternative 3 would not retain FEI's existing on-system peaking capabilities and,

7 therefore, did not pass the Step 2 screen.

8 5.1.3.2.3 BASE PLANT CHALLENGES (MEDIUM POSITIVE IMPACT)

9 The same assessment from Supplemental Alternative 2 applies to Supplemental Alternative 3.

10 5.1.3.2.1 LEVELIZED TOTAL RATE IMPACT (LOW NEGATIVE IMPACT)

11 The levelized total rate impact for Supplemental Alternative 3 is estimated to be 3.4 percent over

12 a 67-year analysis period.

13 This reflects the estimated capital cost of \$435.0 million to install the 600 MMcf/d of new 14 regasification, the operating costs over the expected life of the new assets, and the costs for gas 15 supply from regional market, if available, to meet FEI's full peaking gas requirement of 1.0 Bcf 16 and 200 MMcf/d once there is no longer sufficient LNG available for peaking gas supply. For years 17 when it is not possible to replace the peaking gas supply level currently provided by the Base 18 Plant, FEI would have to curtail firm customers under normal operations. Please refer to Section 19 3.3 of this Appendix for details on the gas supply cost assumptions and the costs/savings used 20 in the financial analysis for Supplemental Alternative 3.

21 5.1.3.2.2 FUTURE USE (MEDIUM NEGATIVE IMPACT)

22 The same assessment from Supplemental Alternative 2 applies to Supplemental Alternative 3.

- 23 The new regasification equipment would have a stranded asset risk as it would not be useful
- 24 without the Base Plant tank or available LNG from Tilbury 1A.



1 5.2 ALTERNATIVES WITH NEW FACILITY THAT MAINTAINS GAS SUPPLY BUT NO 2 RESILIENCY RESERVE

3 This section discusses Supplemental Alternatives 4 and 4A, which would involve a replacement 4 facility intended to mitigate the age-related challenges associated with the Base Plant and either 5 avoid further deterioration of Tilbury's existing gas supply function (Supplemental Alternative 4) or optimize FEI's gas supply portfolio to provide sufficient peaking supply to meet FEI's entire 6 7 requirements (Supplemental Alternative 4A). These Supplemental Alternatives would not add a 8 resiliency reserve and, as such, customers would remain exposed to the current high risk of a 9 widespread and prolonged outage in winter. Both Supplemental Alternatives 4 and 4A are viable 10 alternatives, but would be inferior to Supplemental Alternative 9 from an overall customer value 11 standpoint.

12 **5.2.1 Supplemental Alternative 4**

- 13 Supplemental Alternative 4 includes the following:
- Replacing the Base Plant with a new 0.6 Bcf tank and 150 MMcf/d of regasification capacity (the original 1971 design capacity of the Base Plant);
- There continuing to be no resiliency reserve; that is, from a planning perspective, all LNG
 in the new tank would be allocated to gas supply and all of Tilbury 1A would be allocated
 to RS 46 LNG sales, such that those volumes may or may not be present in a no-flow
 event;
- Using the new facility as a gas supply peaking resource, akin to how the Base Plant has
 always been used; and
- Continuing to rely on 50 MMcf/d of year-round pipeline capacity to achieve required peaking supply (1 Bcf and 200 MMcf/d).

24 5.2.1.1 Modelling Parameters

Table C-13 below sets out the resiliency and gas supply modelling parameters for the Supplemental Alternative 4 scenarios.

The planning scenario uses the maximum regasification (150 MMcf/d), paired with the volume of dependable LNG for resiliency (0 Bcf - no resiliency reserve) and, for peaking gas supply, the amount included in the Annual Contracting Plan (0.6 Bcf).

- 30 FEI also modelled a contingent (i.e., non-dependable) scenario based on optimistic assumptions
- 31 about how much LNG is present at Tilbury on the day of a no-flow event. These scenarios are
- 32 described in Section 4 of this Appendix.



Supp. Alt. 4 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 4 (Planning)	Like-for-Like (No Resiliency Reserve) ²	0 Bcf at 150 MMcf/d	0.6 Bcf at 150 MMcf/d
Supplemental Alternative 4 (Contingent)	Like-for-Like (Assume New Tank Full on Day of No-Flow)	0.6 Bcf at 150 MMcf/d	0.6 Bcf at 150 MMcf/d

Table C-13: Supplemental Alternative 4 Planning and Contingent Modelling Scenario

2

1

3 <u>Notes to Table:</u>

4 ¹ FEI allocates 0.6 Bcf from Tilbury for gas supply portfolio purposes. Actual volumes will decline with use during 5 winter, including if LNG volumes were used for resiliency to respond to a no-flow event.

When there is no resiliency reserve, on a planning basis there is no dependable LNG available on occurrence of no-flow event. Hence the resiliency modelling parameter is 0 Bcf.

8 *5.2.1.2* Summary of Information and Analysis

9 FEI investigated Supplemental Alternative 4 with respect to the criteria discussed in Section 4.2 10 of the Supplemental Evidence. Supplemental Alternative 4 would be technically and commercially 11 viable, while retaining FEI's existing on-system firm peaking supply; therefore, it was evaluated 12 under Step 3 in the structured process to select a preferred alternative. Alternative 4 would be 13 inferior to Supplemental Alternative 9 from an overall customer value standpoint.

14 The following bullets provide a high-level description of how this alternative performs against the 15 criteria, followed by sections providing additional details.

- Resiliency (No Impact): Supplemental Alternative 4 would provide no additional risk mitigation against a winter T-South no-flow event relative to Supplemental Alternative 1. In particular, the limited regasification capacity of only 150 MMcf/d and the absence of a dedicated resiliency reserve mean that Supplemental Alternative 4 would provide the same inadequate level of resiliency as the existing Base Plant. A winter T-South no-flow event would have the same consequences as the Supplemental Alternative 1 (Planning) scenario.
- Gas Supply (Medium Positive Impact): As a new facility with a dedicated gas supply reserve of 0.6 Bcf, Supplemental Alternative 4 would improve the availability of dependable gas supply during peak demand when compared to Supplemental Alternative 1. However, FEI would need to continue acquiring an additional 50 MMcf/d and 0.4 Bcf of peaking resources from the market annually, which would be sub-optimal.
- Base Plant Challenges (High Positive Impact): The new facility would address the age related issues with the existing Base Plant.
- Levelized Total Rate Impact (Low Negative Impact): Supplemental Alternative 4 would
 have a levelized total rate impact of 3.2 percent over a 67-year analysis period.



- Future Use (No Impact): Supplemental Alternative 4 would remain useful and not underutilized for the period between the in-service date and 2050.
- 3 FEI discusses each of the five analysis criteria in further detail below.

4 5.2.1.2.1 RESILIENCY (NO IMPACT)

Supplemental Alternative 4 would not include a resiliency reserve and, even assuming some LNG
inventory is present at Tilbury on the day of a no-flow event, would provide no additional risk
mitigation against a winter T-South no-flow event relative to Supplemental Alternative 1. FEI
addresses the resiliency benefit of the Supplemental Alternative 4 scenarios below.

- Like Supplemental Alternative 1, the Supplemental Alternative 4 (Planning) scenario would not address the regasification constraint of only 150 MMcf/d at Tilbury. This is only a fraction of the daily load in the Lower Mainland on winter days. Further, without a resiliency reserve, all LNG would continue to be allocated for planning purposes to either gas supply or RS 46 LNG sales, such that it may or may not be present on the day of a winter T-South no-flow event.
- The Supplemental Alternative 4 (Contingent) scenario, which assumes that 0.6 Bcf of LNG
 is available on the day of a no-flow event, would similarly be unable to mitigate additional
 risk because, like the planning scenario, the like-for-like replacement of the existing Base
 Plant would not address the regasification constraint of only 150 MMcf/d.

Under both the planning and contingent scenarios, the system would still fail on the day of a winter
T-South no-flow event, resulting in similar consequences as the Supplemental Alternative 1
(Planning) scenario. These results are confirmed in Exponent's figure (reproduced in the Section
2.5.1 above) summarizing the annual expected loss reduction associated with each Supplemental
Alternative relative to the Supplemental Alternative 1 (Planning) scenario.

The following table summarizes how long the Supplemental Alternative 4 scenarios would be able to support the Lower Mainland load following a winter T-South no-flow event. FEI has consistently modelled three temperatures that are reflective of local conditions, since FEI's load increases as temperatures decrease.



Table C-14: Supplemental Alternative 4 Load Support Durations Under Different Winter Conditions

Supp. Alt. 4	Approximate Time I	te Time Until Customers in the Lower Mainland Begin Losing Service ⁴⁶		
Modelling Scenario	-10°C (very cold winter day) ⁴⁷	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁴⁸	+4.0°C (average Lower Mainland winter) ⁴⁹	
Planning (0 Bcf, 150 MMcf/d)	0 days and 0 hours	0 days and 0 hours	0 days and 0 hours	
Contingent (0.6 Bcf, 150 MMcf/d)	0 days and 2 hours	0 days and 5 hours	0 days and 7 hours	

3

4 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

5 under average winter conditions, the support duration provided by the Supplemental Alternative

6 4 scenarios would not be enough to execute a controlled shutdown. Therefore, FEI expects that

7 a failure occurring on some parts of T-South would result in an uncontrolled shutdown in the Lower

8 Mainland system, with attendant safety risks and service restoration challenges.

- 9 The transient modelling outputs for the Supplemental Alternative 4 (Contingent) scenario are
- 10 identical to those of the Supplemental Alternative 1 (Contingent w/T1A) scenario, as provided in
- 11 Section 2.5 above. This is due to Supplemental Alternative 4 having the same regasification
- 12 capacity as Supplemental Alternative 1, as well as the regasification constraint not being fully
- 13 addressed by either Supplemental Alternative.

14 5.2.1.2.2 GAS SUPPLY (MEDIUM POSITIVE IMPACT)

- Supplemental Alternative 4 would improve the availability of dependable gas supply during peak
 demand relative to Supplemental Alternative 1 and, therefore, passed the Step 2 screen.
- _____
- 17 Supplemental Alternative 4 would involve constructing a new LNG storage tank and regasification
- 18 that is dedicated to gas supply, thus retaining FEI's existing firm peaking gas supply capabilities

⁴⁶ This represents the approximate duration of full firm load support for customers in the Lower Mainland. Except for the 0 Bcf case which does not account for linepack, the analysis considers support from on-system LNG and linepack from the CTS. Linepack was not considered in the 0 Bcf case such that an understanding of the absolute baseline risk could be had. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁴⁷ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁴⁸ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁴⁹ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



- that have long been included in the Annual Contracting Plan (0.6 Bcf gas supply reserve and 150 1 2
- MMcf/d of send-out) while also resolving the gas supply concerns associated with the existing
- 3 Base Plant, which are driven by the age-related dependability challenges discussed in Section
- 4 3.3 of the Supplemental Evidence. This alternative would also address the challenges associated 5 with the stop-gap measure of relying on 0.25 Bcf from Tilbury 1A (i.e., LNG that is intended for
- 6 RS 46 sales) to compensate for the Base Plant tank being operated at 0.35 Bcf for seismic
- 7 reasons. This represents an improvement from Supplemental Alternative 1.
- 8 However, while Supplemental Alternative 4 would avoid the need to replace the existing on-9 system peaking gas supply provided by Tilbury with capacity on regional infrastructure at considerable cost, it would not provide FEI's full peaking supply requirements (200 MMcf/d and 1 10 11 Bcf). As such, FEI would need to continue acquiring an additional 50 MMcf/d and 0.4 Bcf of 12 peaking resources from the market annually - which would be sub-optimal and increase FEI's
- 13 gas supply costs.
- 14 Further, if the Supplemental Alternative 4 (Contingent) scenario were to be realized (i.e., a failure
- 15 event occurred, and FEI utilized the hypothetically available LNG volume from the new tank for
- 16 resiliency), FEI would be forced to try to make up the lost peaking gas supply from the market for
- 17 the remainder of the winter. To the extent that this was possible, it would increase FEI's gas
- 18 supply costs for the remainder of the year.

19 5.2.1.2.3 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

- 20 Like other alternatives that would involve a new tank and regasification, Supplemental Alternative
- 21 4 would address the age-related challenges associated with continuing to rely on the Base Plant
- 22 and, in particular, would restore the reliability of Tilbury's critical gas supply function.

23 5.2.1.2.4 LEVELIZED TOTAL RATE IMPACT (LOW NEGATIVE IMPACT)

- 24 The levelized total rate impact for Supplemental Alternative 4 is estimated to be 3.2 percent over 25 a 67-year analysis period.
- 26 This reflects the estimated capital cost of \$826.9 million to install a new 0.6 Bcf tank and 150 27 MMcf/d of new regasification as well as the operating costs over the expected life of the new 28 assets. Supplemental Alternative 4 will avoid the costs of securing 0.6 Bcf and 150 MMcf/d for 29 peaking gas supply from the regional market, resulting in estimated savings of \$46 million starting 30 in 2035, but FEI would continue to incur gas supply costs for the year-round 50 MMcf/d to make 31 up FEI's full peaking gas requirement of 1.0 Bcf and 200 MMcf/d. Please refer to Section 3.3 of 32 this Appendix for discussion on the gas supply cost assumptions and the costs/savings used in 33 the financial analysis for Supplemental Alternative 4.
- 34 FEI notes the following regarding the results of the financial analysis:
- Supplemental Alternative 4 would have a lower levelized total rate impact than 35 • 36 Supplemental Alternative 9, but would provide no additional protection against a winter T-37 South no-flow event relative to Supplemental Alternative 1; and



There are significant economies of scale with LNG facility construction, as Supplemental 1 • 2 Alternative 9 (which is five times larger) costs only 28 percent more than Supplemental 3 Alternative 4.

4 5.2.1.2.5 FUTURE USE (NO IMPACT)

5 Supplemental Alternative 4 would be useful and not underutilized between the in-service date and 6 2050.

7 In Section 4.5.5 of the Supplemental Evidence, FEI describes how an on-system LNG facility is a unique asset when it comes to the flexibility afforded in response to changing load. Supplemental 8 9 Alternative 4, as an LNG facility, would remain used and useful between the in-service date and 10 2050. Even under the mDEP (2% and 5%) hypothetical adverse load sensitivities, the 2050 load 11 would still be too large for the facility to be underutilized for peaking gas supply.

12 The guestion of whether or not the facility contemplated under a Supplemental Alternative would 13 be used and useful must be distinguished from whether or not it would deliver on the Project 14 objectives. Supplemental Alternative 4 will never be able to avoid the consequences of a winter 15 T-South no-flow event at any time between the in-service date and 2050. The current daily load 16 vastly exceeds the regasification capacity of Supplemental Alternative 4, as do the 2050 mDEP 17 (2% and 5%) daily loads. For example, under the most pessimistic future load sensitivity 18 considered by FEI (mDEP (5%)), in 2050 the Lower Mainland load at +4°C is estimated to be 19 approximately 321 MMcf/d, which is more than double the Supplemental Alternative 4 send-out

20 capacity of 150 MMcf/d.

21 Similarly, FEI's peaking supply requirements currently exceed what Supplemental Alternative 4

22 could provide, such that FEI is already augmenting its supply with year-round pipeline capacity.

23 This would remain the case for some time under the mDEP (2% and 5%) hypothetical adverse

24 load loss sensitivities discussed in Section 4.5.5 of the Supplemental Evidence.

25 5.2.2 Supplemental Alternative 4A

26 Supplemental Alternative 4A would involve replacing the existing Tilbury Base Plant with a new 27 facility for gas supply. The sizing of the new facility reflects the smallest possible facility that would 28 meet FEI's peaking supply requirements and thus optimize FEI's gas supply portfolio. Due to 29 significant increases in load, FEI's on-system peak gas supply needs have increased since the 30 construction of the Base Plant in 1971. As a result, a facility that is larger than the Base Plant is 31 necessary to meet those peaking supply requirements and optimize FEI's gas supply portfolio.

- 32 Supplementary Alternative 4A includes the following:
- 33 Replacing the Base Plant with a new 1 Bcf tank and 400 MMcf/d (2 x 200 MMcf/d • 34 vapourizers) of regasification capacity;
- 35 There continuing to be no resiliency reserve; that is, from a planning perspective, all LNG • 36 in the new tank would be allocated to gas supply and all of Tilbury 1A would be allocated



- to RS 46 LNG sales, such that those volumes may or may not be present upon a no-flow
 event; and
- Allocating all of the 1 Bcf tank volume to gas supply, therefore providing sufficient peaking
 supply to meet FEI's full requirements of 1.0 Bcf and 200 MMcf/d and avoiding the need
 to continue augmenting the peaking supply with regional market resources.

6 5.2.2.1 Modelling Parameters

7 Table C-15 below sets out the resiliency and gas supply modelling parameters for the8 Supplemental Alternative 4A scenarios.

- 9 The planning scenario uses the maximum regasification (400 MMcf/d), paired with the volume of
- 10 dependable LNG for resiliency (0 Bcf no resiliency reserve) and 1 Bcf of peaking gas supply.
- 11 FEI also modelled a contingent (i.e., non-dependable) scenario based on optimistic assumptions

12 about how much LNG is present at Tilbury on the day of a no-flow event. These scenarios are

13 described in Section 4 of this Appendix.

14 Table C-15: Supplemental Alternative 4A Planning and Contingent Modelling Scenario

Supp. Alt. 4A Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 4A (Planning)	New 1 Bcf Tank and 400 MMcf/d Regasification ²	0 Bcf at 400 MMcf/d	1 Bcf at 400 MMcf/d
Supplemental Alternative 4A (Contingent)	New 1 Bcf Tank and 400 MMcf/d Regasification (Assume New Tank Full on Day of No- Flow)	1 Bcf at 400 MMcf/d	1 Bcf at 400 MMcf/d

15

16 <u>Notes to Table:</u>

In Supplemental Alternative 4A FEI allocates 1 Bcf from Tilbury for gas supply portfolio purposes. Actual volumes
 will decline with use during winter, including if LNG volumes were used for resiliency to respond to a no-flow event.

When there is no resiliency reserve, on a planning basis there is no dependable LNG available on occurrence of no-flow event. Hence the resiliency modelling parameter is 0 Bcf.

21 *5.2.2.2* Summary of Information and Analysis

FEI investigated Supplemental Alternative 4A with respect to the criteria discussed in Section 4.2 of the Supplemental Evidence. Supplemental Alternative 4A would be technically and commercially viable, while retaining FEI's existing on-system firm peaking supply and, therefore, was evaluated under Step 3 in the structured process to select a preferred alternative. Supplemental Alternative 4A would be superior to some options but inferior to Supplemental Alternative 9 from an overall customer value standpoint.



- 1 The following bullets provide a high-level description of how this alternative performs against the 2 criteria, followed by sections providing additional details.
- Resiliency (No Impact): Supplemental Alternative 4A would not include a resiliency reserve and, even assuming some LNG inventory is present at Tilbury on the day of a no-flow event, would provide no material incremental risk mitigation against a winter T-South no-flow event relative to Supplemental Alternative 1 because the send-out capacity of 400 MMcf/d remains undersized for the system demand.
- Gas Supply (High Positive Impact): As a new facility with a dedicated gas supply reserve of 1 Bcf, Supplemental Alternative 4A would improve the availability of dependable gas supply during peak demand when compared to Supplemental Alternative 1. It would also provide the same gas supply benefits as Supplemental Alternative 9 by virtue of meeting FEI's entire peaking supply requirements. This would provide considerable flexibility for gas supply planning and winter operation, and potentially would allow FEI to displace other sub-optimal gas portfolio assets.
- Base Plant Challenges (High Positive Impact): The new facility would address the age related issues with the existing Base Plant.
- Levelized Total Rate Impact (Low Negative Impact): Supplemental Alternative 4A
 would have a levelized total rate impact of 3.0 percent over a 67-year analysis period,
 which is lower than Supplemental Alternative 4 due to the additional gas supply benefits.
- **Future Use (No Impact):** Supplemental Alternative 4A would remain useful and not underutilized for the period between the in-service date and 2050.
- 22 FEI discusses each of the five analysis criteria in further detail below.

23 5.2.2.2.1 RESILIENCY (NO IMPACT)

Supplemental Alternative 4A would not include a resiliency reserve and, even assuming some LNG inventory is present at Tilbury on the day of a no-flow event, would provide no material incremental risk mitigation against a winter T-South no-flow event relative to Supplemental Alternative 1. FEI addresses the resiliency benefit of the Supplemental Alternative 4A scenarios below.

- The Supplemental Alternative 4A (Planning) scenario would provide no additional protection against a winter T-South no-flow event relative to the Supplemental Alternative 1 (Planning) scenario. This is because Supplemental Alternative 4A (Planning) would not have a dedicated resiliency reserve, and therefore, could not support the entire Lower Mainland for any duration on the day of a winter T-South no-flow event, resulting in the same consequences as the Supplemental Alternative 1 (Planning) scenario.
- The Supplemental Alternative 4A (Contingent) scenario, which assumes that 1 Bcf of LNG is available on the day of a winter T-South no-flow event, would provide improved risk mitigation relative to both the Supplemental Alternative 1 (Contingent) and Supplemental Alternative 1 (Contingent w/ T1A) scenarios; however, the improvement would not be



- material. This is because, like Supplemental Alternative 2, Supplemental Alternative 4A
 would improve the regasification capacity relative to the capacity of the existing Base Plant
 (150 MMcf/d), but would not fully address the existing regasification constraint at Tilbury.
- 4 At average Lower Mainland winter conditions (+4°C), the send-out capacity of 400 MMcf/d 5 would remain undersized for the system demand. As a result, the Supplemental 6 Alternative 4A (Contingent) scenario would provide the same level of risk mitigation as the 7 Supplemental Alternative 2 (Contingent w/T1A) scenario. Although the Supplemental 8 Alternative 4A (Contingent) scenario assumes more LNG would be available than the 9 Supplemental Alternative 2 (Contingent w/ T1A) scenario (1.0 vs. 0.75 Bcf), the support duration and risk mitigation results are identical since the constraining factor, regasification 10 11 capacity, remains unresolved for both Supplemental Alternatives. As such, the system 12 would still fail on Day 1 (cold weather of -1.4°C) or Day 2 (at an average winter temperature of +4°C) of a winter no-flow event, resulting in similar consequences as the Supplemental 13 14 Alternative 1 (Planning) scenario.
- 15 These results are confirmed in Exponent's figure (reproduced in the Section 2.5.1 above) 16 summarizing the annual expected loss reduction associated with each Supplemental Alternative 17 relative to the existing Base Plant with no capital upgrades (the Supplemental Alternative 1
- 18 (Planning) scenario).
- 19 The following table summarizes how long the Supplemental Alternative 4A scenarios would be
- 20 able to support the Lower Mainland load following a winter T-South no-flow event. FEI has
- 21 consistently modelled three temperatures that are reflective of local conditions, since FEI's load
- 22 increases as temperatures decrease.



Table C-16: Supplemental Alternative 4A Load Support Durations Under Different Winter Conditions

Supp. Alt. 4A	Approximate Time I	Fime Until Customers in the Lower Mainland Begin Losing Service ⁵⁰		
Modelling Scenario	-10°C (very cold winter day) ⁵¹	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁵²	+4.0°C (average Lower Mainland winter) ⁵³	
Planning (0 Bcf, 400 MMcf/d)	0 days and 0 hours	0 days and 0 hours	0 days and 0 hours	
Contingent (1 Bcf, 400 MMcf/d)	0 days and 5 hours	0 days and 12 hours	1 day and 19 hours	

3

4 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

under average winter conditions, the support duration provided by the Supplemental Alternative
4A (Planning) scenario would not be enough to execute a controlled shutdown. Therefore, a
failure occurring on some parts of T-South could be expected to result in an uncontrolled

8 shutdown in the Lower Mainland system, with attendant safety risks and service restoration

9 challenges. Under average winter conditions, the support duration provided by the Supplemental

10 Alternative 4A (Contingent) scenario may or may not be enough to execute a controlled shutdown.

11 Therefore, a T-South failure in some segments could result in a controlled or uncontrolled

12 shutdown in the Lower Mainland system.

13 The transient modelling outputs for Supplemental Alternative 4A (Contingent) are identical to

14 those of Supplemental Alternative 2 (Contingent w/T1A). This is due to Supplemental Alternative

15 4A having the same regasification capacity as Supplemental Alternative 2 (Contingent w/ T1A),

16 as well as the regasification constraint not being fully addressed by either Supplemental

17 Alternative. As a result, while the Supplemental Alternative 4A (Contingent) scenario assumes a

18 greater volume of LNG being available than the Supplemental Alternative 2 (Contingent w/ T1A)

19 scenario, it does not result in a longer support duration or additional risk mitigation.

⁵⁰ This represents the approximate duration of full firm load support for customers in the Lower Mainland. Except for the 0 Bcf case which does not account for linepack, the analysis considers support from on-system LNG and linepack from the CTS. Linepack was not considered in the 0 Bcf case such that an understanding of the absolute baseline risk could be had. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁵¹ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁵² The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁵³ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



1 5.2.2.2.2 GAS SUPPLY (HIGH POSITIVE IMPACT)

- 2 Supplemental Alternative 4A would improve the availability of dependable gas supply during peak
- 3 demand relative to Supplemental Alternative 1 and, therefore, passed the Step 2 screen.

4 Supplemental Alternative 4A would involve constructing a new LNG storage tank and 5 regasification that is dedicated to gas supply (1 Bcf paired with 200 MMcf/d⁵⁴ of send out), thus 6 optimizing FEI's gas supply function while resolving the gas supply concerns associated with the 7 existing Base Plant, which are driven by the age-related dependability challenges discussed in 8 Section 3.3 of the Supplemental Evidence. The additional 0.4 Bcf that is dedicated to gas supply, 9 paired with the additional 50 MMcf/d of send-out, would provide FEI considerable flexibility for gas supply planning and winter operation, and potentially displace other higher cost gas portfolio 10 assets. Similar to Supplemental Alternative 4, this alternative would also address the challenges 11 12 associated with the stop-gap measure of relying on 0.25 Bcf from Tilbury 1A (i.e., LNG that is 13 intended for RS 46 sales) to compensate for the Base Plant tank being operated at 0.35 Bcf for 14 seismic reasons.

Further, by expanding both the available volume of LNG for gas supply and the regasification capacity beyond what FEI currently relies on for on-system peaking supply in the Annual

17 Contracting Plan (i.e., 0.6 Bcf and 150 MMcf/d), Supplemental Alternative 4A would also provide

- 18 incremental gas supply benefits, including allowing FEI to optimize its ACP by potentially
- 19 displacing other gas portfolio assets.
- 20 FEI notes that if the Supplemental Alternative 4A (Contingent) scenario were to be realized (i.e.,
- 21 a failure event occurred and FEI utilized the hypothetically available LNG volume from the new
- 22 tank for resiliency), FEI would be forced to try to make up the lost peaking gas supply from the
- 23 market for the remainder of the winter. To the extent that this was possible, it would increase FEI's
- 24 gas supply costs for the remainder of the year.

25 5.2.2.2.3 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

- 26 Like other alternatives that involve a new tank and regasification, Supplemental Alternative 4A
- would address the age-related challenges associated with continuing to rely on the Base Plant
- and, in particular, would restore the reliability of Tilbury's critical gas supply function.

29 5.2.2.2.4 LEVELIZED TOTAL RATE IMPACT (LOW NEGATIVE IMPACT)

- The levelized total rate impact for Supplemental Alternative 4A is estimated to be 3.0 percent over
 a 67-year analysis period.
- This reflects the estimated capital cost of \$893.2 million to install a new 1.0 Bcf tank and 400 MMcf/d of new regasification as well as the operating costs over the expected life of the new assets. Given Supplemental Alternative 4A would be able to meet FEI's full peaking gas requirement of 1.0 Bcf and 200 MMcf/d under the optimized portfolio, it would avoid the costs of securing the same amount of peaking gas from the regional market. The benefits begin

⁵⁴ Supplemental Alternative 4A includes 400 MMcf/d of send out capacity, however for gas supply planning purposes only 200 MMcf/d is required.



- 1 immediately following the in-service date, as FEI would initially avoid the need to continue holding
- 2 50 MMcf/d of year-round pipeline capacity. The estimated avoided costs reach \$63 million starting
- in 2035. Please refer to Section 3.3 of this Appendix for discussion on the gas supply cost
 assumptions and the costs/savings used in the financial analysis for Supplemental Alternative 4A.
- 5 FEI notes the following regarding the financial analysis:
- Supplemental Alternative 4A would have a lower levelized total rate impact than the
 smaller like-for-like facility (Supplemental Alternative 4) by virtue of the economies of scale
 in LNG facility construction and the additional gas supply benefits;
- Supplemental Alternative 4A would have a lower capital cost and levelized total rate
 impact than Supplemental Alternative 9 but would provide no additional protection against
 a winter T-South no-flow event relative to Supplemental Alternative 1; and
- There are significant economies of scale with LNG facility construction, as Supplemental
 Alternative 9 (which is three times larger) will cost only 22 percent more than Supplemental
 Alternative 4.

15 5.2.2.2.5 FUTURE USE (NO IMPACT)

- Supplemental Alternative 4A would be useful and not underutilized between the in-service dateand 2050.
- In Section 4.5.5 of the Supplemental Evidence, FEI describes how an on-system LNG facility is a unique asset when it comes to the flexibility afforded in response to changing load. Supplemental Alternative 4A, as an LNG facility, would remain used and useful between the in-service date and 2050. Even under the mDEP (2% and 5%) hypothetical adverse load sensitivities, the 2050 load would still be too large for the facility to be underutilized for peaking gas supply.
- The question of whether or not the facility contemplated under a Supplemental Alternative would be used and useful must be distinguished from whether or not it would deliver on the Project objectives. Without a dedicated resiliency reserve, Supplemental Alternative 4A would not be able to avoid the consequences of a winter T-South no-flow event at any time between the in-service date and 2050.

28 5.3 ALTERNATIVES WITH NEW FACILITY THAT PROVIDES RESILIENCY RESERVE 29 BUT NO GAS SUPPLY

Supplementary Alternatives 5, 6 and 7 would involve replacing the Base Plant with new 0.6, 1.0 or 2 Bcf facilities, where the entire volume would be set aside as a dependable resiliency reserve – necessitating that FEI try to secure replacement gas supply in the market. FEI's ability to withstand a winter T-South no-flow event would increase as the regasification capacity and resiliency reserve increase in tandem, and a 2 Bcf resiliency reserve would provide appropriate risk reduction. However, as set out in Table C-5 above, the capital cost of building a smaller facility approaches the cost of Supplemental Alternative 9 due to significant economies of scale in tank



1 construction. For example, a 1 Bcf facility with 800 MMcf/d regasification would only cost 18

2 percent less to construct than a 3 Bcf facility with the equivalent regasification. FEI customers

would also be subject to curtailments in normal operations and significant annual gas costs to
 meet FEI's peaking supply requirements with capacity on regional infrastructure. As a result,

- 5 these options would have lower overall customer value relative to Supplemental Alternative 9.
- 5 these options would have lower overall customer value relative to Supplemental Alternative 9

6 **5.3.1 Supplemental Alternative 5**

Supplemental Alternative 5 is the same as Supplemental Alternative 4 in terms of the physical
facilities constructed, but would differ in terms of how the capabilities are allocated for planning
purposes.

- 10 Supplemental Alternative 5 would involve:
- Replacing the Base Plant with a new 0.6 Bcf tank and 150 MMcf/d of regasification capacity (the original 1971 design capacity of the Base Plant);
- Allocating all of the 0.6 Bcf tank volume to a resiliency reserve and no allocation for peaking gas supply;
- FEI would, therefore, have to rely entirely on the regional market to meet its full peaking gas supply requirements. However, given that regional infrastructure is already fully contracted, FEI would be unable to replace the entire peaking gas supply requirement immediately. As such, FEI would still hold the existing 50 MMcf/d of market pipeline capacity but would likely have to curtail firm loads in normal operations by up to 150 MMcf/d until future regional infrastructure upgrades occurred; and
- If and when regional infrastructure upgrades occur (consistently assumed to be 2035 for the financial analysis), it would offer the potential for FEI to end curtailments by holding gas supply contracts for 1.0 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that includes the costs of the upgrades.

25 **5.3.1.1 Modelling Parameters**

Table C-17 sets out the resiliency and gas supply modelling parameters for the SupplementalAlternative 5 scenarios.

The planning scenario uses the maximum regasification (150 MMcf/d), paired with 0.6 Bcf of dependable LNG for resiliency (a full resiliency reserve) and 0 Bcf of peaking gas supply –

30 meaning FEI would need to attempt to replace lost peaking gas supply with capacity on regional

- 31 pipeline or storage infrastructure at additional cost.
- 32 FEI also modelled a contingent (i.e., non-dependable) scenario based on optimistic assumptions

33 about how much LNG is present at Tilbury on the day of a no-flow event. These scenarios are

34 described in Section 4 of this Appendix.



Supp. Alt. 5 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 5 (Planning)	Like-for-Like (Full Resiliency Reserve) ²	0.6 Bcf at 150 MMcf/d	0 Bcf at 150 MMcf/d
Supplemental Alternative 5 (Contingent w/ T1A)	Like-for-Like (Full Resiliency Reserve and T1A Available on Day of No-Flow)	1 Bcf at 150 MMcf/d	0 Bcf at 150 MMcf/d

Table C-17: Supplemental Alternative 5 Planning and Contingent Modelling Scenarios

2 3 Not

1

Notes to Table:

4 ¹ On a planning basis, since the tank is fully allocated to a resiliency reserve, FEI would not be able to meet the
5 requirements of the ACP. In order to have dependable peaking supply, FEI would have to obtain those resources
6 in the form of pipeline capacity or regional storage capacity.

When there is a resiliency reserve, on a planning basis there is dependable LNG available on occurrence of no flow event. Hence the resiliency modelling parameter is 0.6 Bcf.

9 *5.3.1.2* Summary of Information and Analysis

FEI investigated Supplemental Alternative 5 with respect to the criteria discussed in Section 4.2
 of the Supplemental Evidence. Supplemental Alternative 5 did not pass the Step 2 screen
 because, without a dedicated gas supply reserve, it would fail to retain FEI's existing on-system
 peaking gas supply capabilities.⁵⁵

14 The following bullets provide a high-level description of how this alternative performs against the 15 criteria, followed by sections providing additional details.

- Resiliency (No Impact): Although Supplemental Alternative 5 would allocate all of the 0.6
 Bcf tank to a resiliency reserve, it would still not provide material risk mitigation against a
 winter T-South no-flow event because it would not address the regasification constraint at
 Tilbury.
- Gas Supply (No Impact): Since Supplemental Alternative 5 would allocate the entire 0.6
 Bcf tank to a resiliency reserve, FEI would have to attempt to replace FEI's existing
 peaking gas supply from Tilbury in the market. Similar to Supplemental Alternative 1, there
 would likely be at least several years where there are significant curtailments in normal
 operations before FEI would be able to acquire enough replacement peaking supply on
 upgraded regional infrastructure.
- **Base Plant Challenges (High Positive Impact):** The new facility would address the agerelated challenges associated with the existing Base Plant.

 $^{^{\}rm 55}\,$ 0.6 Bcf and 150 MMcf/d of regasification capacity.



- Levelized Total Rate Impact (Medium Negative Impact): Supplemental Alternative 5 would have a levelized total rate impact of 4.4 percent over a 67-year analysis period.
- Future Use (No Impact): Supplemental Alternative 5 would remain useful and not underutilized for the period between the in-service date and 2050.
- 5 FEI discusses each of the five analysis criteria in further detail below.

6 5.3.1.2.1 RESILIENCY (NO IMPACT)

1

2

7 While Supplemental Alternative 5 would add a resiliency reserve, it would still not provide material
8 risk mitigation against a winter T-South no-flow event relative to Supplemental Alternative 1
9 because it would not address the regasification constraint at Tilbury. FEI addresses the resiliency
10 benefit of the Supplemental Alternative 5 scenarios below.

- 11 The Supplemental Alternative 5 (Planning) scenario would provide additional risk 12 mitigation against a winter T-South no-flow event on T-South when compared to the Supplemental Alternative 1 (Planning) scenario. This is due to Supplemental Alternative 13 14 5 having a dedicated resiliency reserve whereas Supplemental Alternative 1 does not. 15 However, the amount of additional mitigation is not material and could not support the entire Lower Mainland for any material duration. Supplemental Alternative 5 (Planning) 16 17 scenario is equivalent to the Supplemental Alternative 1 (Contingent) and Supplemental 18 Alternative 1 (Contingent w/T1A) scenarios. This is because the alternative would not address the regasification constraint at Tilbury (150 MMcf/d), which would not be sufficient 19 20 to meet the Lower Mainland daily demand in winter.
- The Supplemental Alternative 5 (Contingent w/T1A) scenario, which assumes that an additional 0.4 Bcf from Tilbury 1A is available to use for resiliency, would also not resolve the regasification constraint and, therefore, would provide the same amount of outage risk mitigation as Supplemental Alternative 5 (Planning).
- Under both planning and contingent scenarios, the system would still fail on the day of a winter
 T-South no-flow event, resulting in similar consequences as the Supplemental Alternative 1
 (Planning) scenario. These results are confirmed in Exponent's figure (reproduced in the Section
 2.5.1 above) summarizing the annual expected loss reduction associated with each Supplemental
 Alternative relative to the Supplemental Alternative 1 (Planning) scenario.
- The following table summarizes how long Supplemental Alternative 5 would be able to support the Lower Mainland load following a winter T-South no-flow event. FEI has consistently modelled three temperatures that are reflective of local conditions, since FEI's load increases as temperatures decrease.



Table C-18: Supplemental Alternative 5 Load Support Durations Under Different Winter Conditions

Supp. Alt. 5	Approximate Time	nate Time Until Customers in the Lower Mainland Begin Losing Service ⁵⁶		
Modelling Scenario	-10°C (very cold winter day) ⁵⁷	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁵⁸	+4.0°C (average Lower Mainland winter) ⁵⁹	
Planning (0.6 Bcf, 150 MMcf/d)	0 days and 2 hours	0 days and 5 hours	0 days and 7 hours	
Contingent (1 Bcf, 150 MMcf/d)	0 days and 2 hours	0 days and 5 hours	0 days and 7 hours	

3

4 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

5 under average winter conditions, the support duration provided by Supplemental Alternative 5

6 would not be enough to execute a controlled shutdown. Therefore, a failure occurring on some

7 portions of T-South could be expected to result in an uncontrolled shutdown in the Lower Mainland

8 system, with attendant safety risks and service restoration challenges.

9 The transient modelling outputs for Supplemental Alternative 5 are identical to those of the

10 Supplemental Alternative 1 (Contingent w/T1A) scenario, as provided in Section 5.1.1 above.

11 5.3.1.2.2 GAS SUPPLY (NO IMPACT)

12 Supplemental Alternative 5 would not retain FEI's existing on-system peaking capabilities and,

13 therefore, did not pass the Step 2 screen.

Supplemental Alternative 5 allocates the entire tank to a resiliency reserve (like Supplemental Alternatives 6 and 7) and, as a result, FEI would have to attempt to replace the existing peaking gas supply at considerable cost by acquiring the equivalent resources on regional pipelines or

17 storage. However, as discussed in Section 3.3.4 of the Supplemental Evidence, FEI could not

18 replace the Tilbury peaking supply resources currently included in the Annual Contracting Plan

19 (150 MMcf/d and 0.6 Bcf) in the market.

⁵⁶ This represents the approximate duration of full firm load support for customers in the Lower Mainland. The analysis considers support from on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁵⁷ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁵⁸ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁵⁹ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



1 5.3.1.2.3 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

- 2 Like other alternatives that involve a new tank and regasification, Supplemental Alternative 5
- would address the age-related challenges associated with continuing to rely on the aging BasePlant.

5 5.3.1.2.1 LEVELIZED TOTAL RATE IMPACT (MEDIUM NEGATIVE IMPACT)

- 6 The levelized total rate impact for Supplemental Alternative 5 is estimated to be 4.4 percent over
- 7 a 67-year analysis period.
- 8 This reflects the estimated capital cost of \$826.9 million to install a new 0.6 Bcf tank and 150 9 MMcf/d of new regasification as well as the operating costs over the expected life of the new 10 assets. However, since this alternative would leave no allocation of the on-system LNG for 11 peaking gas supply, FEI would have to secure its entire peaking gas requirement of 1.0 Bcf and 12 200 MMcf/d from the regional market. The annual peaking supply costs would initially consist of 13 the cost of continuing to hold the 50 MMcf/d of year-round pipeline capacity, with significant 14 curtailments in normal operations. The annual gas supply costs would reach an estimated \$63 15 million starting in 2035 when regional infrastructure upgrades are assumed to be possible. Please 16 refer to Section 3.3 of this Appendix for discussion on the gas supply cost assumptions and the 17 costs/savings used in the financial analysis for Supplemental Alternative 5.
- 18 FEI notes the following regarding the financial analysis:
- Supplemental Alternative 5 has the same capital cost as Supplemental Alternative 4 (\$826.9 million), as both of these alternatives would involve using the same asset (a new 0.6 Bcf tank and 150 MMcf/d of regasification). The only difference is in the way the facility is allocated for planning purposes (i.e., this alternative dedicates the entire volume to a resiliency reserve, while Supplemental Alternative 4 allocates the entire volume to gas supply); and
- Supplemental Alternative 5 is inferior to Supplemental Alternative 9 with a higher levelized total rate impact due to the increase in gas supply costs, which fully offset the lower capital cost for the smaller facility (given the significant economies of scale with LNG facility construction).

29 5.3.1.2.2 **FUTURE USE (NO IMPACT)**

- Supplemental Alternative 5 would be useful and not underutilized between the in-service date and2050.
- 32 In Section 4.5.5 of the Supplemental Evidence, FEI describes how an on-system LNG facility is a
- 33 unique asset when it comes to the flexibility afforded in response to changing load. Supplemental
- 34 Alternative 5, as an LNG facility, would remain used and useful between the in-service date and
- 35 2050. Even under the mDEP (2% and 5%) hypothetical adverse load sensitivities, the 2050 load
- 36 would still be too large for the facility to be underutilized for peaking gas supply.



- 1 The guestion of whether or not the facility contemplated under a Supplemental Alternative would 2 be used and useful must be distinguished from whether or not it would deliver on the Project
- objectives. Supplemental Alternative 5 would never be able to avoid the consequences of a winter
- 3 4 T-South no-flow event at any time between the in-service date and 2050. The current daily load
- 5 vastly exceeds the regasification capacity, as do the 2050 mDEP (2% and 5%) daily loads. For
- 6 example, under the most pessimistic future load sensitivity considered by FEI (mDEP (5%)), in
- 7 2050 the Lower Mainland load at +4°C is estimated to be approximately 321 MMcf/d, which is
- 8 more than double the Supplemental Alternative 5 sendout capacity of 150 MMcf/d.

9 5.3.2 Supplemental Alternative 6

- Supplemental Alternative 6 includes the following: 10
- 11 Replacing the Base Plant with a new 1 Bcf tank and 800 MMcf/d (4 x 200 MMcf/d 12 vapourizers) of regasification capacity;
- 13 Allocating all of the 1 Bcf tank volume to a resiliency reserve and no allocation for peaking 14 gas supply;
- 15 FEI would, therefore, have to rely entirely on the regional market to meet its full peaking gas supply requirements. However, given that regional infrastructure is already fully 16 17 contracted, FEI would be unable to replace the entire peaking gas supply requirement 18 immediately. As such, FEI would still hold the existing 50 MMcf/d of market pipeline 19 capacity but would likely have to curtail firm loads in normal operations by up to 150 MMcf/d until future regional infrastructure upgrades to occur; and 20
- 21 If and when regional infrastructure upgrades occur (consistently assumed to be 2035 for 22 the financial analysis), it would offer the potential for FEI to end curtailments by holding 23 gas supply contracts for 1.0 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that 24 includes the costs of the upgrades.

25 5.3.2.1 Modelling Parameters

26 Table C-19 sets out the resiliency and gas supply modelling parameters for Supplemental 27 Alternative 6.

- 28 The only modelling scenario is a planning scenario, focused on dependable resources. The 29 scenario treats the entire tank (1 Bcf) as a resiliency reserve that would be available on the day
- 30 of a no-flow event. As such, FEI would need to replace lost peaking gas supply at additional cost
- 31 with capacity on regional pipeline or storage infrastructure.



Supp. Alt. 6 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 6 (Planning)	New 1 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification ²	1 Bcf at 800 MMcf/d	0 Bcf at 800 MMcf/d

Table C-19: Supplemental Alternative 6 Planning Modelling Scenarios

3 <u>Notes to Table:</u>

1

2

- 4 ¹ On a planning basis, since the tank is fully allocated to a resiliency reserve, FEI would not be able to meet the
 5 requirements of the ACP. In order to have dependable peaking supply, FEI would have to obtain those resources
 6 in the market in the form of pipeline capacity or regional storage capacity.
- When there is a resiliency reserve, on a planning basis there is dependable LNG available on occurrence of no flow event. Hence the resiliency modelling parameter is 1 Bcf.

9 *5.3.2.2* Summary of Information and Analysis

- FEI investigated Supplemental Alternative 6 with respect to the criteria discussed in Section 4.2
 of the Supplemental Evidence. Supplemental Alternative 6 did not pass the Step 2 screen
 because, without a dedicated gas supply reserve, Supplemental Alternative 6 would fail to retain
 FEI's existing on-system peaking gas supply capabilities.⁶⁰
- 14 The following bullets provide a high-level description of how this alternative performs against the 15 criteria, followed by sections providing additional details.
- Resiliency (No Impact): Although Supplemental Alternative 6 would allocate all of the 1
 Bcf tank volume to a resiliency reserve and significantly increase the regasification
 capacity, it would provide little additional protection against a winter T-South no-flow event
 because the tank volume is insufficient to bridge a regulatory shutdown of at least 3 days.
- Gas Supply (No Impact): Since Supplemental Alternative 6 would allocate the entire 1
 Bcf tank to a resiliency reserve, FEI would have to attempt to replace the existing peaking
 gas supply from Tilbury in the market. Similar to Supplemental Alternative 1, there would
 likely be at least several years where there are significant curtailments in normal
 operations before FEI would be able to acquire enough replacement peaking supply on
 upgraded regional infrastructure.
- **Base Plant Challenges (High Positive Impact):** The new facility would address the agerelated challenges associated with the existing Base Plant.
- Levelized Total Rate Impact (High Negative Impact): Supplemental Alternative 6 would
 have a levelized total rate impact of 4.9 percent over a 67-year analysis period.

⁶⁰ 0.6 Bcf and 150 MMcf/d of regasification capacity.



- Future Use (No Impact): Supplemental Alternative 6 would remain useful and not underutilized for the period between the in-service date and 2050.
- 3 FEI discusses each of the five analysis criteria in further detail below.

4 5.3.2.2.1 RESILIENCY (NO IMPACT)

5 Supplemental Alternative 6 would improve resiliency relative to Supplemental Alternative 1 by 6 increasing regasification capacity at Tilbury to 800 MMcf/d (addressing the existing regasification 7 constraint at Tilbury) and including a 1 Bcf resiliency reserve, which would ensure LNG volumes 8 are available to support the system on the day of a winter T-South no-flow event. However, 9 Exponent's analysis confirms that 1 Bcf does not materially improve FEI's ability to withstand a 10 winter T-South no-flow event relative to the existing capabilities of the Tilbury Base Plant. This 11 result is confirmed in Exponent's figure (reproduced in Section 2.5.1 above) summarizing the 12 annual expected loss reduction associated with each Supplemental Alternative relative to the 13 Supplemental Alternative 1 (Planning) scenario.

The following table summarizes the load support duration modelling results at varioustemperatures.

16 17

Table C-20: Supplemental Alternative 6 Load Support Durations Under Different Winter Conditions

Supp. Alt. 6	Approximate Time	Until Customers in the Lower Mainland Begin Losing Service ⁶¹		
Modelling Scenario	-10°C (very cold winter day) ⁶²	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁶³	+4.0°C (average Lower Mainland winter) ⁶⁴	
Planning (1 Bcf, 800 MMcf/d)	1 day and 9 hours	1 day and 21 hours	2 days and 10 hours	

18

¹⁹ Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan,

²⁰ under average winter conditions, the support duration provided by the Supplemental Alternative

^{21 6} may or may not be enough to execute a controlled shutdown. Therefore, a T-South failure in

⁶¹ This represents the approximate duration of full firm load support for customers in the Lower Mainland. The analysis considers support from on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁶² Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁶³ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁶⁴ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



- some segments could result in a controlled or uncontrolled shutdown in the Lower Mainland
 system.
- 3 Figures C-27 to C-29 below show the modelling outputs used to determine the load support
- 4 duration values at various temperatures (+4°C, -1.4°C and -10.0°C) in the table above. As noted
- 5 above, Supplemental Alternative 6 assumes there is 1 Bcf of LNG inventory present on the day
- 6 of a winter T-South no-flow event, as there would be a resiliency reserve.

Figure C-27: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 6 – 1bcf & 800MMcf/d



9



Figure C-28: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 6 – 1bcf & 800MMcf/d



3 4

5

Figure C-29: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 6 – 1bcf & 800MMcf/d



6

7 5.3.2.2.2 GAS SUPPLY (NO IMPACT)

- 8 Supplemental Alternative 6 would not retain FEI's existing on-system peaking capabilities and,
- 9 therefore, did not pass the Step 2 screen.



- 1 Supplemental Alternative 6 would be identical to Supplemental Alternatives 5 and 7 from a gas
- 2 supply perspective insofar as the entire tank would be allocated to a resiliency reserve. As a
- 3 result, FEI would need to attempt to replace the existing peaking gas supply in the market, which
- 4 would not be possible given that regional infrastructure is fully contracted.

5 5.3.2.2.3 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

- 6 Like other alternatives that would involve a new tank and regasification, Supplemental Alternative
- 7 6 would address the age-related challenges associated with continuing to rely on the Base Plant.
- 8 5.3.2.2.1 LEVELIZED TOTAL RATE IMPACT (HIGH NEGATIVE IMPACT)
- 9 The levelized total rate impact for Supplemental Alternative 6 is estimated to be 4.9 percent over
- 10 a 67-year analysis period.

11 This reflects the estimated capital cost of \$933.5 million to install a new 1.0 Bcf tank and 800 12 MMcf/d of new regasification as well as the operating costs over the expected life of the new 13 assets. However, since this alternative would leave no allocation of the on-system LNG for 14 peaking gas supply, FEI would have to secure its entire peaking gas requirement of 1.0 Bcf and 15 200 MMcf/d from the regional market. The annual peaking supply costs would initially consist of 16 the cost of continuing to hold the 50 MMcf/d of year-round pipeline capacity, with significant 17 curtailments in normal operations. The annual gas supply costs would reach an estimated \$63 18 million starting in 2035 when regional infrastructure upgrades are assumed to be possible. Please 19 refer to Section 3.3 of this Appendix for discussion on the gas supply cost assumptions and the 20 costs/savings used in the financial analysis for Supplemental Alternative 6.

Based on the results of the levelized total rate impact, Supplemental Alternative 6 is inferior to Supplemental Alternative 9 with a higher levelized total rate impact due to the increase in gas supply costs outweighing the reduction in capital costs for a smaller facility (given the significant economies of scale with LNG facility construction). Supplemental Alternative 9 would provide both

25 resiliency and availability of dependable gas supply during peak demand.

26 5.3.2.2.2 FUTURE USE (NO IMPACT)

- Supplemental Alternative 6 would be useful and not underutilized between the in-service date and2050.
- 29 In Section 4.5.5 of the Supplemental Evidence, FEI describes how an on-system LNG facility is a
- 30 unique asset when it comes to the flexibility afforded in response to changing load. Supplemental
- 31 Alternative 6, as an LNG facility, would remain used and useful between the in-service date and
- 32 2050. Even under the mDEP (2% and 5%) hypothetical adverse load sensitivities, the 2050 load
- 33 would still be too large for the facility to be underutilized for peaking gas supply.

34 **5.3.3 Supplemental Alternative 7**

- 35 Supplemental Alternative 7 includes the following:
- Replacing the Base Plant tank with a new 2 Bcf tank;



- Replacing the existing Base Plant regasification capacity with 800 MMcf/d (4 x 200 MMcf/d vapourizers);
- Allocating all of the 2 Bcf tank volume to a resiliency reserve and no allocation for peaking gas supply;
- FEI would, therefore, have to rely entirely on the regional market to meet its full peaking gas supply requirements. However, given that regional infrastructure is already fully contracted, FEI would be unable to replace the entire peaking gas supply requirement immediately. As such, FEI would still hold the existing 50 MMcf/d of market pipeline capacity but would likely have to curtail firm loads in normal operations by up to 150 MMcf/d until future regional infrastructure upgrades to occur; and
- If and when regional infrastructure upgrades occur (consistently assumed to be 2035 for the financial analysis), it would offer the potential for FEI to end curtailments by holding gas supply contracts for 1.0 Bcf and 200 MMcf/d. FEI would pay a higher toll / charge that includes the costs of the upgrades.

Supplemental Alternatives 7 and 8 are similar in terms of the physical asset being constructed but contemplate different uses for the tank for planning purposes. Whereas Supplemental Alternative 8 contemplates splitting the tank between gas supply and resiliency, this alternative dedicates the entire tank to resiliency.

19 *5.3.3.1 Modelling Parameters*

Table C-21 sets out the resiliency and gas supply modelling parameters for SupplementalAlternative 7.

The modelling scenario is a planning scenario, focused on dependable resources. The scenario treats the entire tank (2 Bcf) as a dependable resiliency reserve that would be available on the day of a no-flow event. As such, FEI would need to replace lost peaking gas supply at additional cost with capacity on regional pipeline or storage infrastructure.

26

1

2

Table C-21: Supplemental Alternative 7 Planning Modelling Scenarios

Supp. Alt. 7 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 7 (Planning)	New 2 Bcf Tank (Full Resiliency Reserve) and 800 MMcf/d Regasification ²	2 Bcf at 800 MMcf/d	0 Bcf at 800 MMcf/d

27

28 <u>Notes to Table:</u>

On a planning basis, since the tank is fully allocated to resiliency reserve, FEI would not be able to meet the requirements of the ACP. In order to have dependable peaking supply, FEI would have to obtain those resources in the market in the form of pipeline capacity or regional storage capacity.



 When there is a resiliency reserve, on a planning basis there is dependable LNG available on occurrence of noflow event. Hence the resiliency modelling parameter is 2 Bcf.

3 *5.3.3.2* Summary of Information and Analysis

FEI investigated Supplemental Alternative 7 with respect to the criteria discussed in Section 4.2
of the Supplemental Evidence. Supplemental Alternative 7 did not pass the Step 2 screen
because, without a dedicated gas supply reserve, Supplemental Alternative 7 would fail to retain
FEI's existing on-system peaking gas supply capabilities.⁶⁵

8 The following bullets provide a high-level description of how this alternative performs against the 9 criteria, followed by sections providing additional details.

- Resiliency (High Positive Impact): At both average winter and colder temperatures,
 Supplemental Alternative 7 would provide superior risk mitigation against a winter T-South
 no-flow event relative to all other alternatives apart from Supplemental Alternative 9 which
 has the same 2 Bcf resiliency reserve.⁶⁶
- Gas Supply (No Impact): Since Supplemental Alternative 7 would allocate the entire 2 Bcf tank to a resiliency reserve, FEI would have to attempt to replace FEI's existing peaking gas supply from Tilbury in the market. As with a no capital upgrades option, there would likely be at least several years where there are significant curtailments in normal operations before FEI would be able to acquire enough replacement peaking supply on upgraded regional infrastructure.
- Base Plant Challenges (High Positive Impact): The new facility would address the age related challenges associated with the existing Base Plant.
- Levelized Total Rate Impact (High Negative Impact): Supplemental Alternative 7 would
 have a levelized total rate impact of 5.6 percent over a 67-year analysis period.
- **Future Use (No Impact):** Supplemental Alternative 7 would remain useful and not underutilized for the period between the in-service date and 2050.
- 26 FEI discusses each of the five analysis criteria in further detail below.

27 5.3.3.2.1 RESILIENCY (HIGH POSITIVE IMPACT)

28 Supplemental Alternative 7 would significantly improve resiliency relative to Supplemental

- Alternative 1 by: (1) increasing regasification capacity at Tilbury to 800 MMcf/d; and (2) creating
- 30 a 2 Bcf resiliency reserve, which would ensure LNG volumes are available to support the system
- 31 on the day of a winter T-South no-flow event.
- 32 While Supplemental Alternatives 7, 8 and the 9 would provide similar risk mitigation under 33 average winter temperatures of $+4^{\circ}$ C (as shown in Section 2.5.1 above), a 2 Bcf resiliency reserve

⁶⁵ 0.6 Bcf and 150 MMcf/d of regasification capacity.

⁶⁶ As explained further in Section 2.5.1 of this Appendix, Supplemental Alternatives 7, 8 and 9 provide similar risk mitigation at average winter temperatures.



1 would provide superior risk mitigation at colder temperatures (a temperature range of 2 approximately +1.7°C to -6.8°C). This is because, within this temperature range, a 2 Bcf resiliency

3 reserve can bridge a 3-day regulatory shutdown while smaller reserves cannot (e.g., the 1.4 Bcf

- 4 resiliency reserve of Supplemental Alternative 8 cannot). Exponent describes the performance of
- 5 a 2 Bcf resiliency reserve with 800 MMcf/d of regasification as a "substantial benefit" over other
- 6 alternatives.67

7 The following table summarizes the load support duration modelling results at various8 temperatures.

9 10

Table C-22: Supplemental Alternative 7 Load Support Durations Under Different Winter				
Conditions				

Supp. Alt. 7 Scenario	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ⁶⁸		
	-10°C (very cold winter day) ⁶⁹	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁷⁰	+4.0°C (average Lower Mainland winter) ⁷¹
Planning (2 Bcf, 800 MMcf/d)	2 days and 17 hours	3 days and 12 hours	4 days and 13 hours

11

12 Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan, 13 under the +4°C and -1.4°C temperature conditions, the support duration provided by 14 Supplemental Alternative 7 is long enough to execute a controlled shutdown. Therefore, at these 15 temperature conditions, FEI expects that an outage caused by a T-South failure (i.e., a T-South failure where the no-flow duration exceeds the support duration and thus an outage occurs) would 16 17 result in a controlled shutdown in the Lower Mainland system. Under the -10°C temperature 18 condition (i.e., a support duration just under 3 days) a controlled shutdown may or may not be 19 possible, and therefore, the shutdown may be controlled or uncontrolled. However, given that the 20 support duration is almost 3-days, it is more likely that the shutdown would be controlled than

21 uncontrolled.⁷²

⁶⁷ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, p. 150.

⁶⁸ This represents the approximate duration of full firm load support for customers in the Lower Mainland. The analysis considers support from on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁶⁹ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁷⁰ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁷¹ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.

⁷² Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Report Appendix U, Figure U.2.


- 1 Figures C-30 to C-32 below show the modelling outputs used to determine the load support
- 2 duration values at various temperatures (+4°C, -1.4°C and -10.0°C) in the table above. As noted
- above, Supplemental Alternative 7 assumes there is 2 Bcf of LNG inventory present on the day
- 4 of a winter T-South no-flow event, as there is a resiliency reserve. FEI has provided an explanation
- 5 of how to interpret these figures in Section 3.2.2.1.2 of the Supplemental Evidence.

Figure C-30: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 7 – 2 Bcf & 800MMcf/d





1 2

Figure C-31: Impact to Lower Mainland due to Loss of T-South Supply at -1.4°C with Supp. Alt. 7 – 2 Bcf & 800MMcf/d



3 4 5

Figure C-32: Impact to Lower Mainland due to Loss of T-South Supply at -10°C with Supp. Alt. 7 – 2 Bcf & 800MMcf/d



6

7 5.3.3.2.1 GAS SUPPLY (NO IMPACT)

8 Supplemental Alternative 7 would not retain FEI's existing on-system peaking capabilities and,

9 therefore, did not pass the Step 2 screen.



- 1 Supplemental Alternative 7 is identical to Supplemental Alternatives 5 and 6 from a gas supply
- 2 perspective insofar as the entire tank would be allocated to a resiliency reserve. As a result, FEI
- 3 would need to attempt to replace the existing peaking gas supply in the market, which would not
- 4 be possible given that regional infrastructure is fully contracted.

5 5.3.3.2.2 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

- 6 Like other alternatives that would involve a new tank and regasification, Supplemental Alternative
- 7 7 would address the age-related challenges associated with continuing to rely on the Base Plant.
- 8 5.3.3.2.3 LEVELIZED TOTAL RATE IMPACT (HIGH NEGATIVE IMPACT)
- 9 The levelized total rate impact for Supplemental Alternative 7 is estimated to be 5.6 percent over
- 10 a 67-year analysis period.
- 11 This reflects the estimated capital cost of \$1,030.3 million to install a new 2.0 Bcf tank and 800 12 MMcf/d of new regasification as well as the operating costs over the expected life of the new 13 assets. However, since this alternative would leave no allocation of the on-system LNG for 14 peaking gas supply, FEI would have to secure its entire peaking gas requirement of 1.0 Bcf and 15 200 MMcf/d from the regional market. The annual peaking supply costs would initially consist of 16 the cost of continuing to hold the 50 MMcf/d of year-round pipeline capacity, with significant 17 curtailments in normal operations. The annual gas supply costs would reach an estimated \$63 18 million starting in 2035 when regional infrastructure upgrades are assumed to be possible. Please 19 refer to Section 3.3 of this Appendix for discussion on the gas supply cost assumptions and the 20 costs/savings used in the financial analysis for Supplemental Alternative 7.
- Based on the results of the levelized total rate impact, Supplemental Alternative 7 is inferior to Supplemental Alternative 9 with a higher levelized total rate impact due to the increase in gas supply costs which outweigh the reduction in capital costs for a smaller facility.

24 5.3.3.2.4 FUTURE USE (NO IMPACT)

- Supplemental Alternative 7 would remain useful and not underutilized for the period between the in-service date and 2050.
- In Section 4.5.5 of the Supplemental Evidence, FEI describes how an on-system LNG facility is a unique asset when it comes to the flexibility afforded in response to changing load. Supplemental Alternative 7, as an LNG facility, would remain used and useful between the in-service date and 2050. Even under the mDEP (2% and 5%) hypothetical adverse load sensitivities, the 2050 load would still be tea large for the facility to be under utilized for peaking are supply.
- 31 would still be too large for the facility to be underutilized for peaking gas supply.

32 **5.4** ALTERNATIVES WITH NEW FACILITY THAT PROVIDES BOTH RESILIENCY 33 RESERVE AND GAS SUPPLY

Supplemental Alternatives 8 and 9 involve replacing the existing Base Plant with new 2 and 3 Bcf facilities, respectively. In both cases a portion of the tank volume would be set aside as a dependable resiliency reserve and the remainder of the tank volume would be used for peaking



- 1 gas supply. While Supplemental Alternatives 8 and 9 provide similar risk mitigation under average
- 2 winter temperatures (refer to Section 2.5.1 above), the larger resiliency reserve provided by
- 3 Supplemental Alternative 9 will make it superior at colder temperatures. As a result, and along
- 4 with the superior gas supply benefit, FEI determined that Supplemental Alternative 9 is the
- 5 preferred alternative.

6 **5.4.1 Supplemental Alternative 8**

- 7 Supplemental Alternative 8 includes the following:
- Replacing the Base Plant tank with a new 2 Bcf tank;
- Replacing the existing Base Plant regasification capacity with 800 MMcf/d (4 x 200 MMcf/d
 vapourizers);
- Allocating 1.4 Bcf of the tank volume to a resiliency reserve;
- Allocating 0.6 Bcf of the tank volume to gas supply to maintain existing peaking gas supply
 at Tilbury; and
- Continuing to hold the existing 50 MMcf/d of year-round pipeline capacity to meet FEI's
 full peaking supply requirements of 200 MMcf/d and 1.0 Bcf.
- Supplemental Alternatives 7 and 8 are the same in terms of the physical asset being constructed but contemplate different uses for the tank for planning purposes. Whereas Supplemental Alternative 7 dedicates the entire tank to resiliency, this alternative contemplates splitting the tank
- 19 between gas supply and resiliency.

20 5.4.1.1 Modelling Parameters

- Table C-23 sets out the resiliency and gas supply modelling parameters for Supplemental Alternative 8.
- 23 The only modelling scenario is a planning scenario, focused on dependable resources. The
- maximum regasification capacity is 800 MMcf/d with the 2 Bcf tank divided between a dependable
 1.4 Bcf resiliency reserve and 0.6 Bcf of dependable gas supply.
- 26

Table C-23: Alternative 8 Planning Modelling Scenarios

Supplemental Alt 8 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 8 (Planning)	New 2 Bcf Tank (1.4 Bcf Resiliency Reserve) and 800 MMcf/d Regasification ²	1.4 Bcf at 800 MMcf/d	0.6 Bcf at 800 MMcf/d



1 <u>Notes to Table:</u>

- On a planning basis, since the tank is only partially allocated to resiliency reserve, FEI would be able to meet the
 requirements of the ACP but would not be able to achieve incremental supply benefits relative to Supplementary
 Alternative 1 (Planning).
- 5 ² When there is a resiliency reserve, on a planning basis there is dependable LNG available on occurrence of a noflow event. Hence the resiliency modelling parameter is 1.4 Bcf.

7 *5.4.1.2* Summary of Information and Analysis

8 FEI investigated Supplemental Alternative 8 with respect to the criteria discussed in Section 4.2

- 9 of the Supplemental Evidence. Supplemental Alternative 8 would be inferior to Supplemental
 10 Alternative 9 from an overall customer value standpoint.
- 11 The following bullets provide a high-level description of how this alternative performs against the
- 12 criteria, followed by sections providing additional details.
- 13 Resiliency (Medium Positive Impact): Supplemental Alternative 8 would provide significantly improved risk mitigation compared to the Supplemental Alternative 1 14 scenarios. Compared to Supplemental Alternatives 7 and 9, which contemplate resiliency 15 reserves of 2 Bcf, Supplemental Alternative 8 would provide similar risk mitigation at 16 17 average Lower Mainland winter temperatures (i.e., +4°C). However, at colder 18 temperatures that occur in typical winters (i.e., between -6.8°C to +1.7°C), the risk 19 mitigation provided by Supplemental Alternative 8 would be significantly less than 20 Supplemental Alternatives 7 and 9. As a result, its resiliency benefit would be inferior to 21 Alternatives 7 and 9.
- Gas Supply (Medium Positive Impact): As a new facility with a dedicated gas supply reserve of 0.6 Bcf, Supplemental Alternative 8 would improve the availability of dependable gas supply during peak demand when compared to the Supplemental Alternative 1 scenarios. However, FEI would need to continue holding its additional 50 MMcf/d of year-round pipeline capacity to meet FEI's 1.0 Bcf peaking supply requirement, which would be sub-optimal from a portfolio design perspective and increase FEI's gas supply costs.
- Base Plant Challenges (High Positive Impact): The new facility would address the age related challenges associated with the existing Base Plant.
- 31 Levelized Total Rate Impact (Medium Negative Impact): Supplemental Alternative 8 32 would have a levelized total rate impact of 4.4 percent over a 67-year analysis period. 33 Although this alternative involves constructing a smaller facility than Supplemental Alternative 9 (i.e., 2 Bcf vs. 3 Bcf storage tank), the levelized total rate impact is higher 34 35 than Supplemental Alternative 9. The annual costs of supplementing FEI's peaking supply 36 with an additional 0.4 Bcf from the regional market to make up FEI's full peaking gas 37 requirement of 1.0 Bcf would outweigh the benefit of lower capital costs for the smaller 38 facility.



- Future Use (No Impact): Supplemental Alternative 8 would remain useful and not underutilized for the period between the in-service date and 2050.
- 3 FEI discusses each of the five analysis criteria in further detail below.
- 4 5.4.1.2.1 RESILIENCY (MEDIUM POSITIVE IMPACT)
- 5 Supplemental Alternative 8 would improve resiliency relative to Supplemental Alternative 1 by:
- 6 (1) increasing Tilbury regasification capacity to 800 MMcf/d; and (2) creating a 1.4 Bcf resiliency
- 7 reserve, which ensures LNG volumes would be available to support the system on the day of a
- 8 winter T-South no-flow event.
- 9 The load support durations provided by a 1.4 Bcf resiliency reserve are materially shorter than
- 10 the durations provided by a 2 Bcf reserve. A 1.4 Bcf resiliency reserve would provide similar risk
- 11 mitigation under average winter temperatures of +4°C as a 2 Bcf resiliency reserve (as shown in
- 12 Section 2.5.1 above). However, a 1.4 Bcf resiliency reserve could not bridge a 3-day regulatory
- 13 shutdown period at temperatures colder than +1.7°C, and therefore, would lack the significant
- 14 resiliency benefit provided by Alternatives 7 and 9.73 Please refer to Section 4.5.1 of the
- 15 Supplemental Evidence which provides a detailed comparison of the risk mitigation provided by
- 16 Supplemental Alternatives 8 and 9 in colder conditions that are present for periods of the winter.
- The following table summarizes the load support duration modelling results at varioustemperatures.

Table C-24: Supplemental Alternative 8 Load Support Durations Under Different Winter Conditions

Supp. Alt. 8 Modelling Scenario	Approximate Time Until Customers in the Lower Mainland Begin Losing Service ⁷⁴				
	-10°C (very cold winter day) ⁷⁵	-1.4°C (warmest winter in 10 years, as defined in footnote) ⁷⁶	+4.0C (average Lower Mainland winter) ⁷⁷		
Planning (1.4 Bcf, 800 MMcf/d)	1 day and 22 hours	2 days and 13 hours	3 days and 8 hours		

21

19

⁷³ Appendix RP 2 to the 2024 Resiliency Plan, Exponent Report, Section 9.

⁷⁴ This represents the approximate duration of full firm load support for customers in the Lower Mainland. The analysis considers support from on-system LNG and linepack from the CTS. The analysis also assumes that all interruptible customers are offline within 4 hours of the no-flow event, except for the -10°C analysis, wherein interruptible customers would already be offline due to the cold temperature condition.

⁷⁵ Due to the low probability of having multiple -10°C days in a row in the Lower Mainland, the -10°C temperature condition analysis assumes the following: the first day is -10°C, the second and third days are -7°C, the fourth day is -3°C, and all subsequent days are +4°C.

⁷⁶ The warmest winter in the last 10 years was found by determining the minimum daily average temperature for each year over a 10-year period from 2013-2022, then selecting the highest value. The analysis was based on data from YVR.

⁷⁷ Defined as the average of the daily average temperatures for December, January, and February over a 10-year period from 2013-2022. The average winter day is based on data from YVR.



Consistent with the framework established in Section 3.4.1.2.2 of FEI's 2024 Resiliency Plan, under the +4°C temperature condition, the support duration provided by the supplemental alternative would be long enough to execute a controlled shutdown. Therefore, at this temperature condition, FEI expects that an outage caused by a T-South failure (i.e., a T-South failure where the no-flow duration exceeds the support duration and thus an outage occurs) would result in a controlled shutdown in the Lower Mainland system. The support duration at the -1.4°C and -10°C temperature conditions creates uncertainty as to whether a controlled shutdown would be

temperature conditions creates uncertainty as to whether a controlled shu
possible, and therefore, the shutdown may be controlled or uncontrolled.

9 Figures C-33 to C-35 below show the modelling outputs used to determine the load support 10 duration values at various temperatures (+4°C, -1.4°C and -10.0°C) in the table above. As noted 11 above, Supplemental Alternative 8 assumes there is 1.4 Bcf of LNG inventory present on the day

12 of a winter T-South no-flow event, as there is a resiliency reserve.

13Figure C-33: Impact to Lower Mainland due to Loss of T-South Supply at +4°C with Supp. Alt. 8 –141.4 Bcf & 800MMcf/d



15

1 2

3

4

5



1 2



3

4

5







1 5.4.1.2.2 GAS SUPPLY (MEDIUM POSITIVE IMPACT)

- 2 Supplemental Alternative 8 would improve the availability of dependable gas supply during peak
- 3 demand relative to Supplemental Alternative 1 and, therefore, passed the Step 2 screen.

4 Supplemental Alternative 8 would involve constructing a new LNG storage tank and regasification 5 that is partially dedicated to gas supply, thus retaining FEI's existing firm peaking gas supply 6 capabilities that have long been included in the Annual Contracting Plan (0.6 Bcf gas supply 7 reserve and 150 MMcf/d of send-out) while also resolving the gas supply concerns associated 8 with the existing Base Plant, which are driven by the age-related dependability challenges 9 discussed in Section 3.3 of the Supplemental Evidence. This alternative would also address the 10 challenges associated with the stop-gap measure of relying on 0.25 Bcf from Tilbury 1A (i.e., LNG that is intended for RS 46 sales) to compensate for the Base Plant tank being operated at 0.35 11 12 Bcf for seismic reasons. This represents an improvement from Supplemental Alternative 1.

13 However, while Supplemental Alternative 8 would avoid the need to replace the existing on-14 system peaking gas supply provided by Tilbury with capacity on regional infrastructure at 15 considerable cost, it would not provide the full 200 MMcf/d and 1 Bcf of peaking supply that FEI 16 requires. Having the ability to meet FEI's entire peaking supply requirements with LNG would: (1) 17 provide FEI considerable flexibility for gas supply planning and winter operation; and (2) 18 potentially displace other higher cost gas portfolio assets. While the additional regasification 19 associated with Supplemental Alternative 8 would provide some additional operational flexibility, 20 the extent of the benefit would be constrained due to the limited gas supply reserve of only 0.6 21 Bcf. As such, FEI would need to continue to hold an additional 50 MMcf/d year-round pipeline 22 capacity to deliver the remaining 0.4 Bcf of peaking resources - which would be sub-optimal from 23 a portfolio design perspective and increase FEI's gas supply costs.

24 5.4.1.2.3 BASE PLANT CHALLENGES (HIGH POSITIVE IMPACT)

25 Like other alternatives that would involve a new tank and regasification, Supplemental Alternative

- 26 8 would address the age-related challenges associated with continuing to rely on the aging Base
- 27 Plant and, in particular, would restore the reliability of Tilbury's critical gas supply function.

28 5.4.1.2.4 LEVELIZED TOTAL RATE IMPACT (MEDIUM NEGATIVE IMPACT)

The levelized total rate impact for Supplemental Alternative 8 is estimated to be 4.4 percent over a 67-year analysis period.

- 31 This reflects the estimated capital cost of \$1,030.3 million to install a new 2.0 Bcf tank and 800
- 32 MMcf/d of new regasification as well as the operating costs over the expected life of the new
- 33 assets. Supplemental Alternative 8 will avoid the costs of securing 0.6 Bcf and 150 MMcf/d for
- peaking gas supply from the regional market, estimated as savings of \$46 million starting in 2035,
 but FEI would continue to incur gas supply costs for the year-round 50 MMcf/d to make up the full
- 36 peaking gas requirement of 1.0 Bcf and 200 MMcf/d. Please refer to Section 3.3 of this Appendix
- 37 for discussion on the gas supply cost assumptions and the costs/savings used in the financial
- 38 analysis for Supplemental Alternative 8.
 - Section 5: Detailed Information Supporting Assessment of the Supplemental Alternatives



1 FEI notes the following regarding the financial analysis:

2 Supplemental Alternative 8 has the same capital cost as Supplemental Alternative 7 3 (\$1,030.3 million), as both of these alternatives would involve using the same asset (a 4 new 2.0 Bcf tank and 800 MMcf/d of regasification). The only difference is in the way the 5 facility is allocated for planning purposes (i.e., this alternative allocates 1.4 Bcf for 6 resiliency and the remaining 0.6 Bcf for gas supply purposes while Supplemental 7 Alternative 7 allocates the entire volume to resiliency). The value of the avoided gas supply 8 costs pertaining to the 0.6 Bcf allocated to gas supply result in the levelized total rate 9 impact for Supplemental Alternative 8 being less than Supplemental Alternative 7 (which 10 has no allocation for gas supply purposes); and

 Although Supplemental Alternatives 8 and 9 both provide resiliency and availability of dependable gas supply during peak demand, Supplemental Alternative 8 is inferior to Supplemental Alternative 9 with a slightly higher levelized total rate impact. The higher gas supply costs associated with Supplemental Alternative 8 outweigh the benefit of lower capital costs for a smaller facility.

16 5.4.1.2.5 FUTURE USE (NO IMPACT)

Supplemental Alternative 8 would remain useful and not underutilized for the period between thein-service date and 2050.

Supplemental Alternative 8 would involve building the same physical facility as Supplemental Alternative 7 (i.e., the two supplemental alternatives only differ in how the tank volume is allocated) and, as such, the Future Use analysis for both Supplemental Alternatives is the same.

22 **5.4.2** Supplemental Alternative 9 (Preferred Alternative)

- 23 Supplemental Alternative 9 is the Preferred Alternative and includes the following:
- Replacing the Base Plant tank with a new 3 Bcf tank;
- Replacing the existing Base Plant regasification with 800 MMcf/d regasification capacity
 (4 x 200 MMcf/d vapourizers);
- Allocating 2 Bcf of tank volume to a resiliency reserve; and
- Allocating 1 Bcf of tank volume to gas supply; therefore, FEI is not required to augment
 the peaking supply with regional market resources.
- Please refer to Section 5 of the Supplemental Evidence which provides an updated ProjectDescription.

32 5.4.2.1 Modelling Parameters

Table C-25 sets out the resiliency and gas supply modelling parameters for SupplementalAlternative 9.



- 1 The modelling scenario is a planning scenario, focused on dependable resources. The maximum
- 2 regasification capacity is 800 MMcf/d with the 3 Bcf tank divided between a dependable 2 Bcf
- 3 resiliency reserve and 1 Bcf of dependable gas supply.

Supp. Alt. 9 Modelling Scenario	Description	Resiliency Modelling Parameters	Peaking Gas Supply Allocation (Normal Operations) ¹
Supplemental Alternative 9 (Preferred)	New 3 Bcf Tank (2 Bcf Resiliency Reserve) and 800 MMcf/d Regasification ²	2 Bcf at 800 MMcf/d	1 Bcf at 800 MMcf/d

Table C-25: Supplemental Alternative 9 Planning Modelling Scenarios

5

4

6 <u>Notes to Table:</u>

- On a planning basis, since the tank is only partially allocated to a resiliency reserve, FEI would be able to meet the
 requirements of the ACP and would be able to achieve incremental supply benefits relative to Supplementary
 Alternative 1 (Planning).
- When there is a resiliency reserve, on a planning basis there is dependable LNG available on occurrence of noflow event. Hence the resiliency modelling parameter is 2 Bcf.

12 *5.4.2.2* Summary of Information and Analysis

Supplemental Alternative 9 will provide superior overall customer value, having regard to risk mitigation, restored peaking gas supply, resolving the age-related challenges associated with the Base Plant, the levelized total rate impact, as well as utilization and usefulness in the future. Supplemental Alternative 9 scored the highest in Step 3 of the alternative selection process, and is therefore the Preferred Alternative. The following bullets provide a high-level description of how this alternative performs against the criteria. Section 4 of the Supplemental Evidence addresses each of the five alternatives analysis criterion in detail.

- Resiliency (High Positive Impact): At both average winter and colder temperatures, Supplemental Alternative 9 will provide superior risk mitigation against a winter T-South no-flow event relative to all other alternatives apart from Supplemental Alternative 7, which has the same 2 Bcf resiliency reserve. Please refer to Section 5.3.3.2.1 above regarding Supplemental Alternative 7 and Sections 4.5 and 4.7 of the Supplemental Evidence for an in-depth discussion of the resiliency performance of Supplemental Alterative 9.
- 26 Gas Supply (High Positive Impact): As a new facility with a dedicated gas supply reserve • 27 of 1 Bcf and an increased send-out capacity of 800 MMcf/d, Supplemental Alternative 9 28 will improve the availability of dependable gas supply during peak demand when 29 compared to the Supplemental Alternative 1 scenarios. Supplemental Alternative 9 will also provide FEI with considerable flexibility for gas supply planning and winter operation, 30 31 and potentially displace other higher cost gas portfolio assets. Please refer to Section 5.2.2.2.2 above regarding Supplemental Alternative 4A and Section 4 of the Supplemental 32 33 Evidence for further discussion of the gas supply criterion.

1

2



- **Base Plant Challenges (High Positive Impact):** The new facility addresses the agerelated issues with the existing Base Plant.
- 3 Levelized Total Rate Impact (Medium Negative Impact): Supplemental Alternative 9 4 has a levelized total rate impact of 4.2 percent over a 67-year analysis period. Although 5 the LNG facility under this alternative would be the largest with the highest capital cost, it has the lowest levelized total rate impact amongst the Supplemental Alternatives that 6 7 include a new LNG storage tank with allocation reserved for resiliency. This is primarily due to the significant economies of scale associated with LNG facility construction, 8 9 combined with the benefits of avoided gas supply costs that are achieved due to the 1.0 10 Bcf gas supply reserve which meets the full peaking gas supply requirement. These 11 benefits outweigh the increase in capital cost. Please refer to Section 3.3 of this Appendix 12 for discussion on the gas supply cost assumptions and the costs/savings used in the 13 financial analysis for Supplemental Alternative 9.
- Future Use (No Impact): Supplemental Alternative 9 will remain useful and not underutilized for the period between the in-service date and 2050. Please refer to Section 4.5.5 of the Supplemental Evidence where FEI provides a detailed analysis of how Supplemental Alternative 9 performs against the Future Use criterion.

18 **5.5** OTHER NON-TILBURY ALTERNATIVES IDENTIFIED BY THE BCUC

The BCUC indicated in the Adjournment Decision that it required further information about other non-Tilbury alternatives. FEI has considered those alternatives as Supplemental Alternatives 10, 11, and 12. After further review, FEI determined that constructing facilities to permit reverse-flow on the VITS (Supplemental Alternative 10), LNG from Woodfibre LNG (Supplemental Alternative 11), and floating storage (Supplemental Alternative 12) would not be viable for a variety of reasons. FEI addresses each non-viable alternative below.

25 **5.5.1 Supplemental Alternative 10**

26 FEI explained in its prior evidence that, during winter, the VITS system hydraulics are such that 27 FEI cannot rely on Mt. Hayes LNG to serve the Lower Mainland following a T-South winter no-28 flow event. Gas typically flows westward on the VITS (i.e., from Coquitlam to Vancouver Island). 29 In normal operating conditions, FEI can serve the Lower Mainland from Mt. Haves notionally, by 30 displacement (i.e., FEI uses Mt. Hayes LNG to serve more load on Vancouver Island, thereby permitting FEI to use gas in the Lower Mainland that was otherwise going to Vancouver Island). 31 32 In the summer, it is hydraulically possible for small amounts of gas to physically flow eastwards 33 through the VITS into the Lower Mainland. However, that is not possible in winter.78

In this section, FEI considers a potential alternative of constructing the necessary facilities to allow
 FEI to reverse the flow on VITS in winter to flow sufficient gas towards the Lower Mainland to
 provide a material resiliency improvement. FEI's analysis concluded that Supplemental

⁷⁸ Exhibit B-15, BCUC IR1 11.7.



- Alternative 10 is not viable; therefore, it was not considered beyond Step 1 of the alternatives
 analysis.
- 3 Supplemental Alternative 10 would involve:
- Using the existing Tilbury facilities with no capital upgrades (i.e., Supplemental Alternative
 1), with the primary constraint remaining that regasification capacity at Tilbury would be
 insufficient to prevent a widespread customer outage within hours; and
- Constructing the necessary facilities to reverse flow 400 MMcf/d on the VITS, which is the upper bound of what could be reverse-flowed on the VITS (as described below).

9 5.5.1.1 VITS Reverse Flow is Limited to Approximately 400 MMcf/d Due to the 10 Coquitlam Watershed

FEI's VITS includes an approximately 28 km segment of NPS 12 pipeline that passes through the Coquitlam Watershed. Due to environmental and permitting challenges, constructing additional pipeline infrastructure in the Coquitlam Watershed would be insurmountable in a reasonable amount of time.⁷⁹ As a result, the capacity of the VITS through the watershed is limited to the existing capabilities of the NPS 12 pipeline. FEI conducted a hydraulic analysis and found that, based on the length, diameter, and maximum operating pressure, the flow rate through the watershed is limited to approximately 400 MMcf/d.

The constraint through the watershed creates a bottleneck for the entire VITS. While it may be possible to upgrade other segments of the VITS to flow more than 400 MMcf/d, the overall VITS reverse flow capability would still be limited by the watershed segment. As such, approximately

21 400 MMcf/d is the upper bound for what the VITS could potentially reverse flow.

5.5.1.2 Substantial Upgrades are Required to Reverse Flow 400 MMcf/d on the VITS

While the VITS could potentially reverse flow up to 400 MMcf/d, FEI would need to complete
 significant infrastructure upgrades. FEI undertook a hydraulic analysis to identify the necessary
 upgrades:⁸⁰

- Looping the VITS with approximately 273 km of new NPS 20 pipeline;81
- Substantially upgrading FEI's existing V3 and V4 compressor stations (e.g., HP expansion and re-configuration to allow for bi-directional flow);
- Re-configuring FEI's future V2 compressor station to allow for bi-directional flow; and

⁷⁹ When FEI constructed the VITS in the early 1990s, construction through the watershed included stringent conditions. This included that the pipeline be overbuilt such that FEI would not have to disturb the watershed in the future for capacity upgrade projects.

⁸⁰ The listed upgrades are high-level and are not exhaustive.

⁸¹ The analysis assumes that FEI's future Eagle Mountain Pipeline 610 is in place. Without this pipeline, additional looping would be required.



• More than tripling the regasification capacity of FEI's Mt. Hayes LNG Facility.

The required upgrades represent a significant and complex scope of work. While FEI has not
 developed any of the potential upgrades, it expects that a project of this scale would have

4 considerable challenges associated with permitting, environment, land / Statutory Right of Way

- 5 (SRW) acquisition, and consultation. Table C-26 below compares the existing parameters of the
- 6 VITS to the parameters required to reverse flow 400 MMcf/d.
- 7

Table C-26: Comparison of Existing VITS to Upgraded VITS

Parameter	Existing VITS	Upgraded VITS
Vancouver Mainland 273/323 Flow Rate ⁸²	155 MMcf/d ⁸³	400 MMcf/d
VITS Pipeline Length	624 km ⁸⁴	Existing + 273 km of NPS 20
V3 Cmp Power	7,300 HP	11,000 HP
V4 Cmp Power	7,300 HP	16,000 HP
Mt. Hayes LNG Regasification	150 MMcf/d	550 MMcf/d

5.5.1.3 Reversing Flow to the Lower Mainland Would Degrade Resiliency in the VITS

FEI used its transient hydraulic system modelling to determine the support duration Supplemental Alternative 10 would provide to the Lower Mainland under average winter conditions (+4°C). FEI assumed that the Lower Mainland would be supported by the Tilbury Base Plant with additional LNG volumes from Tilbury 1A (totalling 0.75 Bcf and 150 MMcf/d), and by reverse flow from Mt. Hayes (1 Bcf and 400 MMcf/d). For the purposes of modelling, it was assumed that 1 Bcf from Mt. Hayes would be available for reverse flow, while the remaining 0.5 Bcf would be used to support customers within the VITS.

17 Based on the modelling results, Supplemental Alternative 10 would provide 3 days and 16 hours of support duration to the Lower Mainland; however, by drawing from the Mt. Hayes storage tank 18 19 to supply the Lower Mainland, this alternative would degrade the resiliency of the VITS. At 20 present, assuming the VITS has access to the full Mt. Hayes storage volume, under average 21 winter conditions, FEI estimates a VITS support duration of approximately 13 days. By reversing 22 flow to the Lower Mainland, where the VITS is limited to 0.5 Bcf from Mt. Hayes, the support 23 duration would be reduced to approximately 4 days and 10 hours. Therefore, the VITS would experience an outage if a no-flow duration exceeded 4 days and 10 hours (compared to 13 days 24 25 under existing conditions).

⁸² The Vancouver Mainland 273 and Vancouver Mainland 323 pipelines are the pipelines within the VITS that bring gas from the Lower Mainland to Vancouver Island.

⁸³ Based on the current model year VITS Design Degree Day load.

⁸⁴ Approximate sum of the length of all pipelines that make up the VITS.



5.5.1.4 Supplemental Alternative 10 is Inferior to the Equivalent Tilbury Facility Alternative

As noted above, the support duration provided by Supplemental Alternative 10 is approximately 3 days and 16 hours. Based on FEI's hydraulic system modelling, Supplemental Alternative 8 (new 2 Bcf tank, 1.4 Bcf resiliency reserve and 800 MMcf/d regasification) would provide a duration of 3 days and 8 hours at average winter temperatures (+4°C) without the following challenges associated with Supplemental Alternative 10:

- A larger and more complex scope of work;
- Greater challenges associated with permitting, environment, land/SRW acquisition, and consultation;
- Degradation of the existing VITS resiliency; and
- The expected project costs per kilometre of the Eagle Mountain Pipeline of approximately
 \$30 million per kilometre, multiplied by 273 km, suggests that Supplemental Alternative
 10 could cost as much as \$8 billion to implement.
- Therefore, FEI concluded that Supplemental Alternative 10 is not viable, and it was not considered
 beyond Step 1 of the alternatives analysis.

17 **5.5.2 Supplemental Alternative 11**

18 In the Adjournment Decision, the BCUC stated:⁸⁵

19 The proposed Woodfibre LNG storage facility could potentially be utilized in an 20 emergency to provide additional resilience to the system. While we appreciate this 21 is (or will be) a customer owned facility and may require some supplementary 22 assets and infrastructure such as a tanker and a jetty, FEI could explore 23 contractual agreements with Woodfibre that would make the gas in the tank 24 available to the FEI system in the case of a force majeure event. In addition, there 25 may be significant line-pack in the Eagle Mountain pipeline that could be utilized 26 in the event of a no-flow incident. In any event, as a result of the lack of a detailed 27 resiliency plan assessing such option, the Panel is unable to conclude whether this 28 is a viable mitigation option to address a no-flow incident should the Project not 29 proceed.

The Application did not address this option because it was never regarded as plausible. In order to address the BCUC's commentary, FEI elaborates on why Supplemental Alternative 11 is not viable. FEI did not consider Supplemental Alternative 11 beyond Step 1 of the alternatives analysis.

⁸⁵ Adjournment Decision, p. 29.



- 1 Supplemental Alternative 11 would involve:
- Using the existing Tilbury facilities with no capital upgrades (Supplemental Alternative 1),
 with the primary constraint being that the regasification capacity is insufficient to prevent
 a widespread customer outage within hours;
- FEI entering into a contract with Woodfibre LNG for a long-term firm supply of LNG; and
- 6 FEI either:
- Custom building a vessel for transporting LNG up the Fraser River to the Tilbury facility, constructing a facility at Tilbury for offloading the LNG from the vessel, and adding more regasification capacity at Tilbury to address the existing regasification constraint; or
- Acquiring property rights from Woodfibre LNG on which FEI constructs a
 regasification facility at Woodfibre, while also constructing facilities to permit
 reversing the flow of VITS and using VITS linepack.

5.5.2.1 Supplying Tilbury Would Be Inconsistent with Woodfibre LNG's Business *Model and LNG Markets*

The realities of Woodfibre LNG's business and LNG markets rule out this alternative as discussedin further detail below.

First, Woodfibre LNG has publicly announced that it has entered into firm contracts for 1.95 MTPA – meaning that over 90 percent of their planned plant capacity (2.1 MTPA) is allocated to customers. As such, there is not sufficient uncontracted capacity available from that facility to support FEI resiliency. Further, given the nature of their business, any LNG storage they have on the site is being inventoried to ensure the next customer vessel can be filled on schedule. If that vessel has just been loaded and left the LNG facility, it is very likely there would be no LNG storage available to support FEI needs.

25 Second, even if sufficient contracted capacity were available for FEI to have access to LNG stored 26 at Woodfibre in the event of an emergency, Woodfibre LNG would have to negotiate contracts 27 with customers that would allow for the interruption of supply to their firm customers. This would 28 be a very non-standard commercial arrangement in the LNG bulk export market and would make 29 Woodfibre's offering less competitive than their competitors internationally. In particular, the 30 nature of the bulk export LNG business is that the LNG liquefaction plant and loading terminal are 31 just one part of a global supply chain which includes the upstream gas production, transportation 32 to the LNG facility, liquefying the natural gas, loading it onto ocean going vessels, transporting it to customers and typically to a regasification terminal where the LNG vessel is received and 33 34 offloaded. This supply chain is tightly controlled to ensure that production at each stage 35 (production of the natural gas, production of the LNG, shipping, receiving) is optimized to ensure 36 minimal, if any, interruptions. For example, if production of LNG is interrupted for any reason, it would cause ripple effects and would affect both upstream production, activities at the receiving 37 38 terminal and delays and additional costs to the shipping companies. In the event the LNG



- 1 production facility cannot provide the contracted volumes of LNG, that facility is then responsible
- 2 to provide what are commonly referred to as "make-up quantities" and may also pay a penalty for
- 3 not being available for service.
- 4 Ultimately, FEI does not believe that there is any commercial arrangement that FEI could make
- 5 with Woodfibre LNG that would not be extremely disadvantageous to FEI's customers, as
- 6 Woodfibre would be injecting significant uncertainty into their business model.

5.5.2.2 Required Infrastructure to Rely on Woodfibre LNG Supply Would be Impractical

9 Even assuming FEI were able to contract for LNG supply from Woodfibre, FEI would then need 10 new infrastructure to make use of the LNG:

- Option #1: Custom building a vessel for transporting LNG up the Fraser River to the
 Tilbury facility, constructing a facility at Tilbury for offloading the LNG from the vessel, and
 adding more regasification capacity at Tilbury to address the existing regasification
 constraint; or
- Option #2: Acquiring property rights from Woodfibre LNG on which FEI constructs a regasification facility at Woodfibre, while also constructing facilities to permit reversing the flow of VITS and using VITS linepack.
- 18 FEI discusses each option below.

19 5.5.2.2.1 CUSTOM BUILD VESSEL, NEW OFFLOADING FACILITY, NEW REGASIFICATION CAPACITY

20 Option #1 would involve FEI: (1) custom building a vessel for transporting LNG up the Fraser 21 River to Tilbury; (2) constructing a facility for offloading the LNG from the vessel; and (3) adding 22 more regasification capacity to address the existing regasification constraint at Tilbury. This option

- 23 is not viable for the reasons below.
- First, since there are restrictions on the sizes of vessels that can travel up the Fraser River, FEI would need to either lease or rent a vessel that could meet these size restrictions or custom build an LNG vessel to navigate the Fraser River to the Tilbury facility. The size of vessel that can transit up the Fraser is not a standard size in the LNG business, so it is highly unlikely that this could be found and, even if a vessel could be found, it is even less likely that it would be available at the time it would be required. This means that in order to ensure that the vessel is available when needed it would need to be custom built.
- Second, even if FEI were able to either lease an LNG carrier on short notice or construct a custom vessel and get it to the Woodfibre LNG facility to be loaded, FEI would still need to build an LNG receiving terminal and pipeline infrastructure to connect to FEI's CTS, thereby enabling FEI to deliver the LNG to the Lower Mainland.
- Third, the Environmental Assessment Certificate (EAC) granted to Woodfibre LNG for the shipping of LNG in and out of the Woodfibre facility did not consider alternative vessel routing or



- 1 the possibility of utilizing Woodfibre LNG to backstop FEI's emergency system needs. Therefore,
- 2 Woodfibre would require an amendment to their EAC to consider potential impacts from any
- 3 alternative routing required to support FEI's needs.

4 5.5.2.2.2 Acquire Land from WoodFibre LNG, Construct New Regasification and Reverse-Flow Capabilities on VITS

- 6 Option #2 would involve FEI: (1) acquiring property rights from Woodfibre LNG on which FEI
- 7 constructs a regasification facility at Woodfibre; and (2) constructing facilities to permit reversing
- 8 the flow of VITS and using VITS linepack. This option is not viable for the reasons below.
- In Section 5.5.1 to this Appendix, FEI explained why reverse flow on the VITS with FEI's Mt. Hayes LNG Facility as the source of supply is not viable. As the Woodfibre LNG facility is located much closer to the CTS than the Mt. Hayes LNG facility, the pipeline looping required in Supplemental Alternative 10 would be avoided. Further, the successful implementation of this option would not diminish the resiliency on the VITS provided by FEI's Mt. Hayes LNG facility. However, the constraint through the Coquitlam watershed would still apply, which creates a bottleneck for the entire VITS that limits the flow rate to approximately 400 MMcf/d.

FEI reached out to Woodfibre LNG regarding adding additional infrastructure at their site. Woodfibre LNG responded that their existing facility configuration was subject to an extensive Nation-led environmental assessment resulting in the Squamish Nation Environmental Assessment Agreement. Any changes to the infrastructure on the site would require lengthy consultation with, and approval from, the Squamish Nation – potentially involving an amendment to the Agreement which also could require a member referendum.

22 5.5.2.3 In Winter the Lower Mainland Cannot Access EGP Linepack

The BCUC noted the potential for linepack to be available to support the Lower Mainland. As such, FEI evaluated the potential for linepack in the new Eagle Mountain pipeline (EGP) to provide resiliency to the Lower Mainland. FEI found that:

- EGP linepack cannot reverse flow to the Lower Mainland in winter, and thus can only be used to support the VITS; and
- The support provided by the EGP linepack to the VITS cannot replace the support provided by on-system LNG.

In its prior evidence, FEI discussed how, in winter, reverse flow from the VITS to the Lower Mainland is not possible.⁸⁶ The hydraulic constraints which prevent FEI from reverse flowing from the VITS to the CTS in winter are not resolved by the EGP. Therefore, in winter, FEI does not expect to be able to access the EGP linepack to support the Lower Mainland. However, the linepack from the EGP could be used to support the VITS.

⁸⁶ Exhibit B-17, BCUC IR1 11.7.



- At average winter conditions (+4°C), and assuming the EGP were in place. FEI estimates the 1 2 support duration provided by the VITS linepack (i.e., from the existing VITS transmission pipelines
- and the new EGP) would be approximately 13.4 hours. Without the EGP, the support duration
- 3 4 provided by the VITS linepack would be approximately 7 hours (i.e., the EGP increases the
- 5 linepack support duration by 6.4 hours at average winter conditions). While the additional support
- 6 provided by the EGP linepack would be valuable, as it would provide time for FEI to implement
- 7 other resiliency capabilities (e.g., begin sending out from Mt. Hayes), it would not be a substitute
- 8 for the support provided by on-system LNG.
- 9 The above analysis assumes that Woodfibre LNG is immediately curtailed following the assumed
- 10 no-flow event. Any consumption of the VITS linepack by Woodfibre LNG would reduce the
- 11 estimated support duration.

5.5.3 Supplemental Alternative 12 12

FEI initially considered, but rejected, the potential to use floating LNG storage as an alternative 13 to the TLSE Project. The BCUC commented in the Adjournment Decision:⁸⁷ 14

- 15 FEI's rejection of the floating LNG storage options appears to be based on its 16 assessment that these facilities are primarily intended to take advantage of 17 offshore natural gas fields which would otherwise be difficult to access. Since FEI 18 is able to liquefy natural gas from its own transmission system for storage on 19 system at Tilbury, it views the floating LNG storage options as being much more 20 expensive and complex than the TLSE Project. However, this begs the question 21 whether, absent access to an expanded facility at Tilbury, the floating LNG storage 22 options are a viable means of mitigating the impacts of a three day no-flow event. 23 In this regard, the analysis of these options would have benefited from a holistic 24 resiliency plan that assessed the relative merits and demerits of various 25 alternatives having regard to the prioritization of resiliency needs on the entire FEI 26 system.
- 27 In response to the BCUC's commentary, FEI considered Supplemental Alternative 12. FEI 28 provides additional discussion regarding floating LNG storage below.
- 29 FEI's additional analysis has not changed its initial determination. In particular, Supplemental 30 Alternative 12 is so complex as to likely be infeasible. Even if it were feasible, it would be very 31 costly – without providing commensurately greater resiliency benefits. Therefore, FEI concluded 32 that Supplemental Alternative 11 is not viable, and it was not considered beyond Step 1 of the 33 alternatives analysis.

⁸⁷ Adjournment Decision, p. 30.



1 *5.5.3.1* Requirements for Using Floating LNG Storage

- 2 Supplemental Alternative 12 would involve:
- Purchasing a vessel to provide floating LNG storage to increase the available energy
 (LNG) in the event of a supply emergency affecting the Lower Mainland;
- Acquiring a water lot that would allow for permanent mooring;
- Adding sufficient new regasification capacity, either as an integrated component of the
 LNG storage vessel or on the adjacent shoreline, to address the existing regasification
 constraint at Tilbury that will currently result in a widespread customer outage within hours
 of a winter T-South no-flow event; and
- Other onshore facilities, including a jetty and pipes to interconnect with the CTS; and
- New environmental assessment approvals.
- 12 An example of a floating storage and regasification unit⁸⁸ (263,000m3) is depicted below.



- 13
- 14 To assess this alternative, FEI engaged an independent consultant (Ogee Development Inc.) to
- 15 assess the viability of a floating storage unit (FSU) at a number of locations with access to the
- 16 CTS. The consultant's analysis indicated that the availability of a suitably sized floating storage
- 17 vessel could be a challenge:89

⁸⁸ <u>https://www.world-energy.org/article/33637.html</u>.

⁸⁹ Appendix C-1, p. 13.



Most of the new build FSU's in the current market are approximately 170,000 m3 in capacity. The older previous generation FSU's were in the size range of 130,000-140,000 m3 capacity. This presents the challenge of using an older existing FSU by way of lease/time charter or purchase. This option of an older FSU would require extensive inspections/verifications and possible modifications, repair, or re-certifications to satisfy todays codes and industry standards.

7 In the sections below, FEI describes why this Supplemental Alternative is not viable.

8 5.5.3.2 There Are No Appropriate Sites

9 The 2024 Resiliency Plan confirmed that the greatest customer outage risk facing FEI is 10 associated with a winter T-South no-flow event. In order to mitigate the risk associated with a 11 winter no-flow event, floating LNG storage would need to be sited close to FEI's largest load in 12 the Lower Mainland to facilitate access to gas within hours of the no-flow event. FEI identified a 13 number of on-water sites (i.e., offshore) with nearby access to the CTS and sought expert analysis 14 to assess the feasibility of tying into each. The most feasible of these options was then compared 15 against the option of onshore storage and regasification at Tilbury.

16 Of the sites identified, all of the offshore sites had issues with technical feasibility either due to 17 the tie-in point pressure rating, pipeline capacity or execution difficulties caused by the location of 18 the tie-in point in the Fraser River. The matrix below shows the viability of the identified offshore 19 storage options.

20

	Base Case TLSE	Location 1 Tilbury Site	Location 2 Fraser River Crossing	Location 3 Burrard Inlet	Location 4 Squamish
T echnical					
Economic					
Environmental					
C ommercial					
Organizational					
Political					

Table C-27: Offshore LNG Storage Viability Matrix

Not viable	Viable but multiple	Viable
(i.e. "fatal flaw")	challenges	

21

The most practical of the offshore options is the location at Tilbury due to the proximity to the CTS main artery and existing infrastructure. This option was analysed in further detail against onshore



LNG storage at Tilbury; however, as described below, even this offshore option was deemed not
 viable.

3 *5.5.3.3* On-system Storage and Regasification on an Existing Site Is Preferable

- 4 Offshore LNG storage at Tilbury presents many issues in terms of practicality, including:
- The water depth of the Fraser River, which would likely require ongoing dredging around
 the floating storage vessel;
- The cost of additional marine infrastructure such as required mooring and jetty systems;
 and
- 9 Concerns from other river users or stakeholders regarding restrictions around an LNG
 10 storage vessel that would require exclusion zones to operate safely.

Further, based on its prior experience, FEI's independent consultant concluded that offshore storage alone would likely cost 20-35 percent more than the onshore equivalent. This does not include the additional costs associated with marine infrastructure or mooring requirements.

Following the analysis of the possible sites that could provide access to the CTS, FEI's consultant concluded that "[f]loating LNG Storage options are considered 'not viable' due to one or more fatal flaws at each of the four potential offshore locations identified" and recommended "that the onshore TLSE Base Case option be considered as the viable option for this project."⁹⁰

18 5.6 SOUTHERN CROSSING PIPELINE EXTENSION / RGSD PROJECT

19 The Application included discussion of pipeline alternatives, including two configurations 20 extending the South Crossing Pipeline that FEI was exploring, which is referred to as the Regional 21 Gas Supply Diversity (RGSD) Project. One potential configuration of the RGSD Project (Oliver to 22 Huntingdon) would bypass T-South and avoid single-point-of-failure risk, whereas the other route 23 would reduce the length of T-South where single-point-of-failure risk exists.⁹¹ FEI concluded that, 24 regardless of the configuration, the RGSD pipeline capacity is best viewed as complementary to 25 the TLSE Project, rather than a substitute for the resiliency on-system LNG provides. In this 26 regard, the BCUC stated in the Adjournment Decision:92

The RGSD project is currently being planned by FEI. RCIA has identified a difference in the intended capacity of that project. Given this uncertainty we are unable to make any finding regarding how the RGSD project may or may not impact the need for the TLSE Project. This further supports the need for a more

⁹⁰ Appendix C-1, p. 17.

⁹¹ Exhibit B-1-4, Application, p. 87.

⁹² Adjournment Decision, p. 48.



- holistic resiliency plan to better understand the interaction of different projects that
 FEI may be contemplating in order to achieve greater resiliency."⁹³
- 3

4 The resiliency plan states that the RGSD project "would allow FEI to split the 5 optimal amount of pipeline capacity between T-South and RGSD, thereby reducing 6 FEI's current heavy dependence on the T-South system." However, IR responses 7 filed in both proceedings suggest that reduced dependence on the T-South System 8 has little to no impact on the TLSE Project. In any event, due to uncertainties in 9 the scope of the RGSD project, we noted in Section 4.1 of our Decision that we 10 are unable to make any finding regarding how the RGSD project may or may not impact the need for the TLSE Project." 11

FEI has recently determined not to pursue RGSD on its own, although FEI has not foreclosed participating with others in a similar pipeline project. Regardless, in this section, FEI provides additional explanation for why a Southern Crossing Pipeline extension would fill a different role than new on-system LNG from the perspective of resiliency and within FEI's gas supply portfolio. Regardless of size or end-point, an extension to the Southern Crossing Pipeline could not prevent Day 1 depressurization of the Lewer Mainland system following a T South pe flow event

17 a Day 1 depressurization of the Lower Mainland system following a T-South no-flow event.

5.6.1 The Lower Mainland Would Lose Service Before FEI Could Get Gas from a Southern Crossing Pipeline Extension

Following a winter T-South no-flow event, the Lower Mainland system will depressurize before
 FEI could access sufficient gas from a Southern Crossing Pipeline extension to restore pressure.

22 It would take approximately two days for FEI to be able to deliver supply through a Southern 23 Crossing Pipeline extension to the Lower Mainland. Pipeline operators plan daily operations and 24 shippers mitigate unutilized capacity in the open markets by scheduling pipeline supply the day 25 before current day delivery. At 5:30AM each morning, FEI's gas traders make gas delivery 26 arrangements based on the demand forecast for the next day. If there was excess capacity, 27 traders mitigate the capacity through buy and sell gas at different gas markets. The revenue from 28 daily trading activities reduces the fixed pipeline costs paid by FEI's customers. FEI cannot rely on the new pipeline within the first two days of a T-South failure because commercial deals have 29 30 been transacted and parties are bound by their contracts.

As such, send out from the TLSE Project would be the only dependable supply to make up for the loss of T-South supply on the first two days of a no-flow event on T-South – a period in which FEI would need to make commercial arrangements in the market to bring additional gas from the new pipeline. The amount of additional supply FEI could secure on the Southern Crossing Pipeline extension would depend on the market conditions when an incident occurs. FEI plans its

⁹³ Adjournment Decision, p. 25.



- required supply months before gas year starts; therefore, it would likely be challenging to secure
 a significant amount of supply from the markets within a short period of time.
- FEI has characterized RGSD as complementary to the TLSE Project from a resiliency standpoint because, once sufficient on-system LNG is in place to bridge the initial period before piped gas can arrive, a new pipeline could add significant resiliency. In particular, it would provide an alternative source of supply that will reduce the risk posed by long-term capacity shortfalls or duration issues (such as those experienced during Phases 2 and 3 of the 2018 T-South Incident).

5.6.2 Tilbury LNG Provides Peak Capacity, Whereas Pipeline Provides Year Round Energy

10 From a gas portfolio standpoint, on-system LNG is providing critical <u>capacity</u> to serve daily load

11 in peak periods. By contrast, a Southern Crossing Pipeline extension would provide year-round

12 <u>energy</u> that would be a like-for-like substitute for FEI's existing long-duration pipeline capacity on

- 13 T-South.
- 14 In order to replicate the capacity function that Tilbury provides today, FEI would have to hold 150
- 15 MMcf/d of pipeline capacity year-round over and above what it required for year-round base load.
- 16 As discussed in Section 3.3.4.5 of the Supplemental Evidence, the 150 MMcf/d of peaking
- 17 capacity would be underutilized for all but a handful of days in winter. This is not economic in
- 18 comparison to on-system LNG storage and exposes FEI customers to significant risk that the
- 19 unused capacity could not be resold to mitigate the cost. In any event, there is not enough supply
- 20 available in the market to replace 150 MMcf/d and 0.6 Bcf at Tilbury.

21 **5.6.3 FEI Is No Longer Pursuing RGSD on its Own**

In its sixth quarterly progress report filed on April 30, 2024, FEI summarized the conclusions of
 its screening analysis, including how FEI evaluated three RGSD Project delivery options,
 discussed the material market developments that have an impact on FEI and the Pacific
 Northwest operating marketplace and discussed how FEI concluded that the analysis supports
 options for a regional infrastructure solution with other market participants.

FEI's evaluation of potential alternative pipeline routes as part of its development work on the RGSD Project found that collaborating with other regional market participants on an integrated solution could be beneficial for the region and FEI's customers. This strategy would enhance the use of existing regional infrastructure, potentially lower costs, and balance risks for FEI and its customers. Therefore, FEI considers the current phase of the RGSD Project to have concluded and intends to explore commercial discussions with other market participants to continue to advance an optimal integrated solution for the region. Appendix C-1 FLOATING LNG STORAGE CONCEPT ANALYSIS





Proje	ect: TLSE - Floating LNG Storage Concept Analysis			Job N	o.:	FEI24-04493		
Location: Tilbury, BC			SMCI Doc. No.: 04493-31-3700-I		-REP-0001			
				FEI Do	oc. No.:			
Rev	Date (dd	l-mmm-yyyy)	Status		Prepared by	Checked by	Approved by	
0	20-	Mar-2024	Issued for Use		BGO	TIP/DKH 🌶	NRM	
0	20-1	Mar-2024	Issued for Use		BGO	TIP/DKH 🆗	NRM	

FORTISBC ENERGY INC.

PROJECT DEVELOPMENT

FLOATING LNG STORAGE CONCEPT ANALYSIS

Prepared by:

Ogee Development Inc.



DISCLAIMER

The Document has been prepared by OgeeDev with the degree of reasonable skill and care to be expected of consultants specializing in energy consultancy. Such advice, opinions and information are provided in good faith and may include advice and information obtained from third party sources. The advice and opinions given are based upon the information available to OgeeDev at the time the report was prepared. All advice, opinions and information are open to interpretation and typographical error.

AUTHENTICATION

Ogeedev		
Title	Name	Authentication
Director, LNG	Nadir Meratla	
Director, Energy and Renewables	Terence Portugal, P. Eng	
Director, Process Technologies	Brijesh Goel	
Strategic LNG Advisor	Don Hill	Þ
Fortis Energy Inc.		
Title	Name	Authentication





Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
Project:	ILSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Droject	TISE Floating INC Storage Concept Analysis	Joh No .	EE124 04402

REVISION HISTORY

DATE	PAGE / SECTION	DESCRIPTION





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

TABLE OF CONTENTS

1.0	GENER	AL INFORMATION	6
1.1 1.2 1.3 1.4 1.5 1.6	Backgr Purpos Scope Terms Approa Basis	ound e of Services of Reference ach and Methodology	
2.0	HIGH-I	LEVEL EVALUATION	9
2.1	Floatin	g LNG Storage – Potential Locations	9
2.2 2.3	2.1.1 2.1.2 2.1.3 2.1.4 Base C Floatin	Tilbury Island Fraser River Crossing near the Port Mann Bridge Burrard Inlet Squamish, at a location near the Woodfibre LNG plant ase TLSE g LNG Storage – Tilbury Island	9
	2.3.1 2.3.2 2.3.3 2.3.4 2.3.5 2.3.6	Technical Challenges: Economic Challenges. Environmental, permits, regulatory approval challenges. Commercial challenges. Organizational. Political.	
3.0	RECON	/MENDATIONS	17





Executive Summary

OgeeDev's recommendation to use proven onshore LNG storage rather than floating LNG storage is based on numerous considerations, including technical, environmental, economic, commercial, organizational, and political (see section 2.3). It is true that LNG liquefaction with associated storage is often the best choice for remote stranded offshore gas reserves, where installing a long and potentially difficult pipeline to bring the gas onshore would otherwise make the development uneconomic. In addition to avoiding a costly subsea pipeline, the floating solution allows the project developers to relocate the facilities, allowing for smaller reserves to be produced. The floating solution will likely be faster and cheaper when faced with those challenges.

However, in most cases where suitable onshore land is available, the cost of a subsea pipeline is not a consideration, and there are no other requirements to install additional infrastructure to support marine facilities and their operations, such as the export or import of LNG by sea, the best and preferred solution is to install onshore facilities. This is because the floating solution often requires higher upfront costs and complexity than the onshore solution.

The exception to this rule was when the development of the onshore facility was in a remote location without significant infrastructure, limited or no labor availability, environmental sensitivity, or use restrictions. The other factors that have resulted in the use of floating LNG import, or FSRU facilities were the availability of low cost vessels that could be converted quickly and cheaply and third parties were willing to provide the capital investment to build and own the facility and provide the facilities on a leased basis where the project developer could avoid the higher upfront capital expense by agreeing to pay an ongoing operational expense.

This alternative commercial arrangement was particularly attractive for developers who could not obtain the capital to build the more conventional facility. At the time when this option was popular, both used and newly built vessels could be acquired quickly and, in some cases, cheaper than the onshore infrastructure and facilities could be developed. However, the current high demand for LNG ships and the limited shipyards capable of building specialty cryogenic vessels have increased both, the cost and schedule to obtain these vessels.

In this specific case, the added complexity of adopting a floating storage solution with the high cost and long procurement times for the floating vessels, designing and implementing the associated marine infrastructure, and resolving competing river access and usage issues is expected to result in the onshore storage option to remain the low cost and shorter schedule project with less uncertainty and risk.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

1.0 GENERAL INFORMATION

1.1 Background

Fortis Energy Inc. (FEI) has been requested by the BC Utilities Commission (BCUC) to provide a supplemental submission for its Tilbury LNG Storage Expansion (TLSE) project application. As part of the submission, FEI has been asked to address additional alternatives.

One of the alternatives FEI is to address is floating LNG storage, based on the following BCUC comments in their Adjournment Decision:

"FEI's rejection of the floating LNG storage options appears to be based on its assessment that these facilities are primarily intended to take advantage of offshore natural gas fields which would otherwise be difficult to access. Since FEI can liquefy natural gas from its own transmission system for storage on system at Tilbury, it views the floating LNG storage options as being much more expensive and complex than the TLSE Project. However, this begs the question whether, absent access to an expanded facility at Tilbury, the floating LNG storage options are a viable means of mitigating the impacts of a three day no-flow event. In this regard, the analysis of these options would have benefited from a holistic resiliency plan that assessed the relative merits and demerits of various alternatives having regard to the prioritization of resiliency needs on the entire FEI system."

1.2 Purpose

To perform a high-level qualitative screening assessment of the viability of conventional onshore LNG storage vs. floating LNG storage options for meeting the resiliency needs of the FEI system in the Lower Mainland area of British Columbia.

1.3 Scope of Services

OgeeDev has conducted a high-level concept / screening assessment on the viability of utilizing a Floating Storage Unit (FSU) or Floating Storage and Regasification Unit (FSRU) to support the resiliency needs of the FEI system.

This analysis has been presented in this technical memo.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493				
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001				
		FEI Doc. No.:					

1.4 Terms of Reference

Following cases have been evaluated in this document:



- 1. Base Case TLSE comprising of LNG Storage (onshore) and Regasification unit (onshore) installed at the existing FEI Tilbury LNG facility.
- 2. Floating LNG Storage (FSU or FSRU), installed at any one the following potential locations:
 - 2.1 Tilbury Island
 - 2.2 Fraser River Crossing near the Port Mann Bridge
 - 2.3 Burrard Inlet
 - 2.4 Squamish, at a location near the Woodfibre LNG plant

1.5 Approach and Methodology

The following approach and methodology have been used in this analysis:

- 1. The OgeeDev project team, in association with FEI's project team, identified potential locations for the installation of Floating LNG Storage (FSU or FSRU) and characterized potential key features of each location to determine the best solution for site selection.
- 2. The key factors for the identification / shortlisting of potential locations for the Floating LNG Storage were:
 - (i) Close to the FortisBC's Coastal Transmission System (CTS) that distributes gas to the lower mainland and Vancouver Island,
 - (ii) Large enough pipeline diameter to handle the required flowrates of the CTS to minimize hydraulic losses,
 - (iii) Similar coverage of customer density.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493					
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001					
		FEI Doc. No.:						

- (iv) Proximity to a suitable water lot, ability to obtain a water lot, create adequate physical access to the water lot, interactions with the shipping and ports.
- 3. Next, the OgeeDev project team conducted a high-level comparison of a TLSE (onshore) design versus an FSU/FSRU (offshore) design at Tilbury Island. The team examined the benefits and challenges of onshore versus offshore floating LNG storage.
- 4. To develop list of challenges and risks associated with the floating storage, the TEECOP framework was utilized whereby Technical, Economic, Environmental, Commercial, Organizational, Political (TEECOP) issues and concerns were screened and identified. TEECOP is a risk analysis framework used to help identify and identify assess risks in projects. The primary domain categories represent the major domains that a project must mature as the project moves through the project development cycle.
- 5. The analysis of the options above helped to prepare a summary recommendations table (go / no-go screening type) for the study.

1.6 Basis

The following data was used in the as the basis for the resiliency study:

LNG Storage Capacity considered: 28,500 – 142,400 m³



Floating LNG Storage Concept Analysis



Project:TLSE - Floating LNG Storage Concept AnalysisLocation:Tilbury, BC

2.0 HIGH-LEVEL EVALUATION

2.1 Floating LNG Storage – Potential Locations

2.1.1 Tilbury Island



Note: Yellow line is existing FEI pipeline routing

Key features:

- 1. Connects to the main arteries of the CTS and thereby serving the entirety of the Coastal Transmission System.
- 2. Close to existing Tilbury Island LNG Facility, comprising of the Base Plant and T1A LNG Plants. Has existing liquefaction and LNG Storage facilities. Has been operating as a peak shaving facility and LNG production facility for over 50 years.
- 3. Potential to use the existing LNG liquefaction plant to fill the FSU with LNG
- 4. Close to water lot on the Fraser River.



Project:

Job No.:



							FEI Doc. No.:									
Location:	Tilbury, BC							SM		Doc	c. N	lo.:	()4493	3-31	-
•		5	5	•	-											

2.1.2 Fraser River Crossing near the Port Mann Bridge

TLSE - Floating LNG Storage Concept Analysis



Note: Yellow line is existing FEI pipeline routing

Key Features:

- 1. Connects to the main arteries of the CTS and thereby serving the entirety of the Coastal Transmission System.
- 2. The location is close to a water lot on the Fraser River. However, this section of the Fraser River is one of the narrowest portions of the river and could lead to potential ship traffic issues in the busy Fraser River should the FSU have loss of control or any other mechanical / operational issues.
- 3. The CTS pipeline is buried deep in this location due to the HDD river crossing. Substantial jetty work and tie-in infrastructure is required to connect the FSRU to the pipeline at this location. These issues make it difficult and expensive to work at this location.

The BC Hydro 500 kV transmission grid overhead powerlines are just downstream of this location and the busy Port Mann Bridge is just upstream. The existing infrastructure can cause further limitation to the ease of any work in this area.



Tilbury, BC

TLSE - Floating LNG Storage Concept Analysis



2.1.3 Burrard Inlet

Project:

Location:



Note: Yellow line is existing FEI pipeline routing

Key Features:

- Indirect connection to the main artery of the CTS; it is connected through a 20" line that is planned to be derated in 2024 due to the decommissioning of the Burrard Thermal Generating Station and integrity management requirements through the BCUC approved CTS TIMC project. Injection at this location would require reinstatement to a higher pressure, which would require pipeline integrity validation and an Engineering Assessment.
- 2. Close to water lot on the Burrard Bay.


Tilbury, BC

Project:

Location:



2.1.4 Squamish, at a location near the Woodfibre LNG plant

TLSE - Floating LNG Storage Concept Analysis



Note: Yellow line is existing FEI pipeline routing

Key Features:

- 1. Indirect connection to the main artery of the CTS; the existing and planned pipeline network between this location and the CTS is unable to transport sufficient gas to support the CTS during cold weather conditions.
- 2. Pipeline access rights from the water.
- 3. Proximity to northwestern shoreline of upper Howe Sound, Squamish.

Based on the analysis above for the four potential site locations to install an FSU, the Tilbury Island location is most suited but still has several challenges which are discussed in the sections below.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

2.2 Base Case TLSE

Key Features:

Existing infrastructure at Tilbury Island may be utilized to support the new facilities of TLSE (LNG Tank and Regasification facilities), such as:

- 1. Existing LNG liquefaction plant, T1A can be used to fill the new TLSE LNG Tank, if no additional liquefaction is installed.
- 2. Space for a new LNG storage tank and additional liquefaction currently available.
- 3. Space for a new regasification unit currently available.
- 4. Existing Control room may be shared with the TLSE facility.
- 5. Operations and Maintenance staff and associated costs may be shared.
- 6. Short run for connectivity to the existing CTS pipeline.
- 7. No new marine or jetty interface is required.
- 8. No major challenges have been identified with this TLSE option.

2.3 Floating LNG Storage – Tilbury Island

Based on the analysis of potential locations analyzed in section 2.1 above, the Tilbury Island location will be evaluated in this section to identify the challenges associated with the offshore storage option.

For simplicity of the analysis, the FSU and FSRU will both be considered the same in this section, and whichever is easier to procure can be considered for the installation, subject to the review of the challenges for this option. The generic term FSU will be used in this evaluation.

2.3.1 Technical Challenges:

- 1. Availability
 - (i) Most of the new build FSU's in the current market are approximately 170,000 m³ in capacity. The older previous generation FSU's were in the size range of 130,000-140,000 m³ capacity. This presents the challenge of using an older existing FSU by way of lease/time charter or purchase. This option of an older FSU would require extensive inspections/verifications and possible modifications, repair, or re-certifications to satisfy todays codes and industry standards.
 - (ii) Alternately, a new custom build FSU with capacity 140,000 m3 may have to be procured which would be relatively much more expensive.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

Both options above require extensive market analysis, compatibility studies, appointment of brokerage, warranty surveyors etc. These skill sets are typically available with shipping operators and not with natural gas operator companies such as FEI.

- 2. Mooring of the FSU at Tilbury Island
 - (i) Water depth in the Fraser River will need to be further studied to handle LNG filled vessel draft. Dredging requirements may have to be established for the entire operating life of the FSU.
 - (ii) Assessment of the Massey Tunnel access with respect to water depth required for the FSU.
 - (iii) New marine infrastructure will be required at Tilbury including a new jetty, mooring systems etc.
 - (iv) New ship to shore interfaces such as a send-out gas pipeline, communication systems, utilities, power, controls and shutdown systems, and safety system etc. would be required.
 - (v) Availability of LNG for filling the FSU from Tilbury or relying on external sources of LNG.
 - (vi) Potential impact to the existing BOG system used for filling (from existing liquefaction train at Tilbury). The existing BOG compressor may require upgrade / modifications to handle the additional BOG generated during the FSU filling operation or may have to be managed by some limitations on existing operations during the filling cycle.
 - (vii) Fraser river ship traffic impact study due to FSU moored at Tilbury. Exclusion zones, safety system, logs etc. May cause restrictions to other users on the Fraser River.
 - (viii) Limits use of Tilbury jetty for future developments. For example, potential export projects. Space constraints between the jetty and new LNG Carriers coming in. Associated risk assessments need to be carried out.
 - All the above issues require further detailed specialist studies, resources, and time.
- 2.3.2 Economic Challenges
 - (i) CAPEX and OPEX of an existing FSU charter / purchase or a new custom-built FSU may be higher than a full containment onshore LNG storage tank depending on the global LNG ship market conditions. The current shipping market is very tight, as the fabrication yards are facing a severe capacity crunch due to an increased amount of global ship orders.
 - (ii) Comparing between the new build options, a new conventional onshore TLSE tank versus a new FSU, the latter is materially more expensive. Based on experience, this can be 20-35% higher in the order of magnitude.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

- (iii) CAPEX for new marine infrastructure including a dedicated jetty at Tilbury is required.
- (iv) CAPEX for the onshore-offshore interfaces including an LNG loading system and send out system is required.
- (v) The cost of a new pipeline to connect the FSU to the gas transmission pipeline across the jetty and/or a longer run of LNG offloading line to the onshore regasification unit is required.
- (vi) Long term CAPEX/OPEX costs for the FSU moored in the Fraser River. These costs would include items such as frequent dredging, debris clearance, mooring system, vessel dry docking for hull and topsides, potential tugboats operation, new operations and maintenance staff and shipping skillsets needed, etc. Ship marine classification requirements may be impacted.

All the above issues require further detailed specialist studies, resources, and time.

- 2.3.3 Environmental, permits, regulatory approval challenges.
 - New Environmental Assessment (EA) or change to existing EA application will be required. Impact to fisheries and other marine life will need to be further studied for all operating and construction scenarios. Could be significant impact to project schedule.
 - (ii) Permits for pipeline, jetty, marine infrastructure/mooring system.
 - (iii) Pipeline ROW permit.
 - (iv) Construction permits for offshore and onshore work.
 - (v) Assessment and handling of the emissions, effluent discharge, noise from the FSU.
 - (vi) Risk assessments (i.e. Quantitative Risk Analyses).

All the above issues require further detailed specialist studies, resources, and time.

2.3.4 Commercial challenges

Contracts – multiple international and local contracts would be required.

- (i) The FSU option would involve multiple international contracts and associated challenges for design, procurement, installation, and operation, depending on the origin and condition of the vessel. Also, permits, regulatory approvals etc.
- (ii) If an existing older FSU or a new build is purchased as the option going forward, it will be an asset outside of FEI's core business. Associated liabilities, labor issues, management and operational contracts and commercial impacts.
- (iii) Contractor for pipeline, jetty, marine infrastructure.
- (iv) Contract for FSU charter, purchase, or new build.
- (v) LNG supply/fill contracts, if required.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

(vi) Contracts for local surveys - water ways, mooring system, river traffic impact assessment etc.

All the above issues require further detailed specialist studies, specialist contracts and legal resources and time.

2.3.5 Organizational

- (i) New operators or training current operations staff for the FSU and jetty operations.
- (ii) Marine crew for the FSU.
- (iii) Interface between operators of FSU control and safety shutdown system and the onshore facilities.
- (iv) Development of offshore, onshore and interface operating protocols.
- (v) Re-classification of the FSU for permanent moored operation, as applicable.
- (vi) Development of FSU emergency safety interface with the shore facilities.
- (vii) Training of local support services for the FSU such as firefighting, bunker, river traffic management, municipal-provincial-federal interfaces etc.
- (viii) FSU Operation/maintenance/management is not core to FEI's business and would be a new area of business, the organization would have to be expanded to include this in the portfolio, along-with the associated risks which would need to be analyzed.

All the above issues require detailed specialist studies, specialist operator training, manuals, and time.

2.3.6 Political

- (i) First Nations engagements and approvals.
- (ii) Public social and political impact due to permanently moored FSU in the Fraser River.
- (iii) Impact to surrounding facilities. Municipal issues. Provincial and Federal issues.

All the above issues require detailed engagement with stake holders and impact assessment.





Project:	TLSE - Floating LNG Storage Concept Analysis	Job No.:	FEI24-04493
Location:	Tilbury, BC	SMCI Doc. No.:	04493-31-3700-REP-0001
		FEI Doc. No.:	

3.0 **RECOMMENDATIONS**

The Onshore TLSE and Offshore Floating LNG Storage options have been analyzed. The Floating LNG Storage options are considered "not viable" due to one or more fatal flaws at each of the four potential offshore locations identified.

It is recommended that the onshore TLSE Base Case option be considered as the viable option for this project.

	Base Case TLSE	Location 1 Tilbury Site	Location 2 Fraser River Crossing	Location 3 Burrard Inlet	Location 4 Squamish
T echnical					
E conomic					
E nvironmental					
C ommercial					
O rganizational					
P olitical					

Not viable	Viable but multiple	Viable
(i.e. "fatal flaw")	challenges	

Appendix D CB&I REPORT

FILED CONFIDENTIALLY

Appendix E WSP REPORT

FILED CONFIDENTIALLY

Appendix F RAYMOND MASON REPORT

Dated: July 2, 2024

FortisBC Energy Inc. Tilbury Liquefied Natural Gas Expansion Project: Gas Supply Considerations

Provided to:

Fasken

Fasken, Martineau DuMoulin LLP 550 Burrard Street, Suite 2900 Vancouver, BC V6C 0A3

Prepared by:

Raymond Mason M4 – Energy & Capital Markets Consulting

Victoria, British Columbia

TABLE OF CONTENTS

INTRODUCTION AND SUMMARY OF CONCLUSIONS	3
SUPPORTING REVIEW AND EVALUATION	9
Response to Question #1	9
Response to Question #2	22
Response to Question #2A	29
Response to Question #2B	30
Response to Question #3	
Response to Question #4	50

INTRODUCTION AND SUMMARY OF CONCLUSIONS

My consulting services have been engaged by Fasken Martineau DuMoulin LLP ("Fasken") as it relates to FortisBC Energy Inc. (FEI)'s Certificate of Public Convenience and Necessity for the Tilbury Liquefied Natural Gas Storage Expansion ("TLSE") Project (the "Regulatory Proceeding"). My industry knowledge and experience (provided in Appendix B) was considered valuable for the purposes of providing an independent expert assessment of questions set out in the Letter of Instruction dated April 18, 2024 and attached to my report as Appendix A. I am aware that I have a duty to assist the regulator and not to be an advocate for any party ("Duty of Independence"). I prepared this report in accordance with the Duty of Independence, and if called upon to give oral or written testimony, I will give testimony in conformity with this duty.

At a high level, I was asked to consider what commercial attributes of the proposed TLSE Project may carry throughout its operational lifetime. Set out below is a summary of my answers to the specific questions put to me in the Letter of Instruction with supplementary information included in the body of my report.

1. What are the elements of an optimal resource portfolio for FEI and its customers?

The development of an efficient natural gas supply portfolio for a utility is predicated on the resource options that are accessible in the area it operates. Though natural gas is a homogeneous form of energy, how the molecules move from the supply to demand region(s) is unique. The characteristics of supply and demand are also not static and will continue to shift to achieve economic balance. In particular, a natural gas portfolio that has been optimized by the utility will respond to shifts in supply and demand to maintain security of supply, while balancing the economic benefits to customers. The elements of an optimal resource portfolio for FEI are therefore multi-faceted, leveraging transportation, storage, peak shaving facilities, and third-party arrangements to meet the demands of its customers throughout the year.

- 2. FEI has, since 1971, included on-system liquified natural gas ("LNG") at the Tilbury facility in its supply portfolio as a peaking capacity resource (i.e., rapid send-out capability, backed by sufficient LNG to meet short-duration peak load). FEI's current Annual Contracting Plan includes 150 MMcf/d of regasification and 0.6 Bcf of LNG from the Tilbury facility. We seek your opinion on the peaking capacity alternatives available to FEI in the event that FEI was no longer able to include some or all of these peaking resources in its supply portfolio. In particular:
 - A. Based on your assessment of the natural gas markets accessible to FEI, would FEI be able to contract for peaking resources (capacity/energy supply) as a potential alternative to on-system LNG in its gas supply portfolio, and if so on what terms?
 - B. If appropriate peaking resources (capacity/energy supply) are unavailable to FEI in the market, what other potential investments could FEI explore as a potential alternative to on-system LNG as peaking resources (capacity/energy supply) in its gas supply portfolio? For clarity, in this regard we are requesting that you identify potential peaking supply (capacity/energy supply) alternatives based on your understanding of the regional system, as opposed to undertaking a financial analysis of those options.

In my opinion, FEI would not be able to contract for peaking capacity resources that I would consider dependable, beyond those that are currently committed, as an alternative to on-system LNG storage. The deployment duration and dispatch capability of a resource remain the main criteria for security of supply from a peaking capacity resource. Due to the nature of the regional demands of FEI's customers, third-party off-system storage and companies offering peaking gas supply arrangements, are constrained by the ability to transport the underlying supply of natural gas, as well as the potential for unplanned outages. Furthermore, based on the results of my market research evaluating third-party arrangements, the costs for peaking resources are extremely expensive, do not support a consistent long term supply resource, and would require a portfolio of participants to be able to meet FEI winter demand. To put this in perspective, FEI would be competing/accessing the Huntingdon/Sumas gas supply market, on the coldest days of the winter, for significant volumes historically destined to the PNW (rapidly escalating pricing throughout daily trading hours). These factors, when taken together, are not conducive to contract for dependable peaking supply resources.

In the absence of readily available dependable peaking resources (including sufficient existing on-system storage), FEI would need to make significant investments (including capital investments and/or contractual capacity commitments) to meet its peaking capacity needs. The types of investments could include expanding mainline transportation, and the associated interconnected off-system storage facilities, and/or developing on-system peak shaving LNG. Building these types of resources takes time, especially capital projects that require consultation

with multiple parties throughout the planning and construction phases. For example, increasing WEI mainline capacity and associated interconnected off-system storage facility (i.e., Aitken Creek) would likely require longer development periods than increasing off-system storage displacement in the Pacific Northwest and/or increasing on-system peak shaving LNG.

In my opinion, the development of a proprietary on-system asset, such as the proposed TLSE Project, has advantages over the alternative infrastructure investments when it comes to designing a gas supply portfolio. Constructing a mainline transportation resource, for the purposes of meeting winter demand, would also result in inefficient utilization due to underutilized capacity in non-winter months. Any investments considered by FEI should consider the assets' long-term utilization parameters and the timing in which the resource could be deployed.

The TLSE Project would provide FEI with operational backup for disruption: (a) to existing offsystem storage and/or mainline transmission; and (b) that may not necessarily affect FEI customers, but those of peer utilities, that source gas supply from Huntingdon/Sumas. The proposed TLSE Project's proximity to a larger US export market could also be a benefit to FEI's customers. In particular, peer utilities in the US, servicing a growing gas-powered electricity and industrial demand are, and will be for the foreseeable future, evaluating the cost/benefit of limiting pricing exposure to Huntingdon/Sumas, in response to growing peaking demand, by either: (a) sourcing peaking services like what the proposed TLSE Project could provide; or (b) committing to long-term mainline transportation.

Furthermore, while meeting the needs of FEI's customers for safe and reliable service, a proprietary LNG peaking facility will continue to carry long-term value for the utility and its customers. If the demand profiles of FEI's customers were to shift over time (i.e., lowering annual demand while maintaining the need for winter supply), the ease of de-contracting mainland transportation is more appropriate than shedding reliable on-system capacity as on-system storage is designed to be deployed only when it is required.

3. Assuming a hypothetical scenario where the TLSE Project is constructed with 3 Bcf of storage volume and at 800 MMcf/d of regasification capacity, and FEI no longer requires the full storage and regasification capacity to meet its own gas supply and resiliency requirements, do you expect that FEI would be able to monetize the surplus by selling peaking resources in the gas markets? If so, how much daily send-out capacity would the market reasonably absorb.

Assuming that the TLSE Project is constructed, and FEI were to have spare capacity that is not required to meet customer demand or resiliency, FEI would be able to generate revenues to offset the cost of service of the facility by selling its excess supply into the market. Based on my assessment of the available supply and demand in the Huntingdon/Sumas natural gas market and assuming current market conditions persist, I expect the daily market can reasonably absorb 300-400 MMcf/d of natural gas across multiple days (e.g., 10 days) during winter without influencing daily prices in a manner that could limit monetization values (i.e., materially decreasing the revenues generated through mitigation into the market).

4. Assuming FEI is able to monetize its hypothetical surplus from the TLSE Project as contemplated in the Question 3 scenario, what annual financial value could FEI recover for its customers in the following scenarios where, in one year, FEI can deliver and the market can absorb a maximum of

300 MMcf/d of send-out sold over 1.3 peak days (300 MMcf/d x 1.3 = Scenario #1: 0.4 Bcf of LNG): Scenario #2: 300 MMcf/d of send-out sold over 2 peak days (300 MMcf/d x 2 = 0.6 Bcf of LNG); Scenario #3: 300 MMcf/d of send-out sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG); Scenario #4: 300 MMcf/d of send-out sold over 5 peak days (300 MMcf/d x 5 = 1.5 Bcf of LNG); Scenario #5: 300 MMcf/d of send-out sold over 7 peak days (300 MMcf/d x 7 = 2.1 Bcf of LNG); and 300 MMcf/d of send-out sold over 10 peak days (300 MMcf/d x 10 = 3.0 Scenario #6: Bcf of LNG).

For each scenario, please calculate the financial value FEI could recover over a 5-year period, taking into account FEI's estimated variable operating costs for the Tilbury facility, which has been provided to you, and assuming FEI's commodity cost to produce LNG reflects gas prices in the summer when FEI typically fills its LNG tank. Please also add another five-year scenario:

Scenario #7: in year 1, 300 MMcf/d sold over 10 peak days (300 MMcf/d x 10 = 3.0 Bcf of LNG), and in each of years 2-5, 300 MMcf/d is sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG);

The results of my financial analysis are presented in the figure below. The figure identifies the range of financial values (i.e., Min to Max) for Scenarios #1 through #7, and the respective 5-year cumulative values assuming 300 MMcf/d of send-out capacity from the TLSE Project, before including any value that FEI would receive for a standing demand charge. The results of my analysis show that, over a 5-year period, FEI could recover between \$73.0 MM CDN to \$78.8 MM CDN of financial value for its customers over 10 peak days in the winter. Assuming Scenario #7, a combination of 10 peak days of send-out capacity in year one and 3 peak days of send-out capacity in years 2 through 5, FEI could recover between \$36.9 MM CDN to \$39.4 MM CDN of financial value for its customers. ¹



The financial results presented in figure above do not include additional incremental value that FEI could capture by implementing a standing demand charge. Please refer to Appendix C for a detailed analysis of this calculation. I estimate that FEI could generate the following range of additional incremental value through an annual standing demand charge:

• \$5.2 MM CDN - \$7.0 MM CDN for every 50 MMcf/d

¹ Based on forward markets dated February 29, 2024

Assuming Scenario #6 (i.e., 10 days of send-out, cumulative over 5-years) and Scenario #7 (i.e., 10 days of send-out year 1 and 3 days of send-out over years 2-5) and that FEI's gas supply department is successful in contracting a standing demand charge with a third-party for 50 MMcf/d of send-out capacity from the TLSE Project, FEI could increase the average cumulative financial value as follows:

Scenario #6: From \$75.9 MM CDN to \$106.4 MM CDN; and Scenario #7: From \$38.2 MM CDN to \$68.7 MM CDN

SUPPORTING REVIEW AND EVALUATION

Response to Question #1

What are the elements of an optimal resource portfolio for FortisBC Energy Inc. ("FEI") and its customers?

The development of an efficient natural gas supply portfolio for a utility is predicated on the resource options that are accessible in the area it operates. Though natural gas is a homogeneous form of energy, how the molecules move from the supply to demand region is unique. The characteristics of supply and demand are also not static and will continue to shift to achieve economic balance. In particular, a natural gas portfolio that has been optimized by the utility will respond to shifts in supply and demand to maintain security of supply, while balancing the economic benefits to customers. The elements of an optimal resource portfolio for FEI are therefore multi-faceted, leveraging transportation, storage, peak shaving facilities, and third-party arrangements to meet the demands of its customers throughout the year.

Figure 1 below shows a general stack of resources a utility might utilize for its natural gas portfolio. Assuming a 365-day distribution, the foundation of the stack represents the necessary transportation delivery to facilitate 100% load factor demand and the deliverability of off-system storage injection and withdrawal seasons. While the stacking of resources will evolve over time as demand profiles shift and marketed products and services are introduced (e.g., storage, transportation, industrial customers, etc.), the contractual commitments of certain resources can extend over long timeframes due to their underlying capital investment structure. As a result, acquiring and eliminating a resource can take a long time, as well as creating overlaps in a portfolio as one resource is implemented and another is retired.





In addition to being in a state of constant change over time, the region FEI operates faces unique challenges among utilities in North America when sourcing its gas supply portfolio. First, the consumer demand is located a significant distance from its source of natural gas. As a result, market participants must undertake additional measures to secure pipeline transportation to move supply to the consumer. Second, because of the characteristics of the supply region and FEI's customer use profiles (i.e., coincidental peaking for power and natural gas), FEI is required to invest in and/or contract for additional resources including: alternative pipeline transportation, off-system storage, on-system peaking alternatives (e.g., interruptible tariffs and Liquified Natural Gas ("LNG"), and contractual gas supply peaking arrangements. As no one resource will provide security of supply, FEI must combine resources and evaluate their dispatch abilities. By understanding a given resource's economic viability, market availability, and associated contractual commitment necessary to secure it, FEI is able to shape the resource development of its broader gas supply portfolio.

In the sections below, I discuss the types of resources that are available in the region, each resource's dependency on another for deployment and how these resources can be combined to create an economic portfolio for FEI's customers. Figure 2 below shows the resources currently available to FEI, as well as potential resources that could be developed in the future (in one form or another).

Figure 2



I. Transportation (Pipeline Capacity) Resources

Transportation resource diversification helps shape an efficient gas supply portfolio. The contract parameters and operation criteria of transportation resources require continuous evaluation to ensure they remain economical and continues to benefit FEI's customers. As part of this evaluation, a utility such as FEI will consider tenure commitments, renewal rights, mitigation capabilities (i.e., during periods of time when transportation is not required), upstream and downstream interconnection reliability, and corresponding capacity availability.

The regions served by FEI are supplied by two upstream transportation pipelines, the Westcoast Energy Inc. ("WEI") T-South pipeline and FEI's Southern Crossing Natural Gas Pipeline ("SCP"). While WEI is interconnected with SCP and the Williams Northwest Pipeline ("NWP"), the SCP is interconnected with the Foothills System ("Foothills"), which connects to the NOVA Gas Transmission System in Alberta.

WEI operates segments of transport in British Columbia that FEI has contracted, Transportation North ("T-North") and Transportation South ("T-South"). The T-North section of pipeline provides access to the Aitken Creek North Gas Storage Facility ("Aitken Creek") and is the critical linkage for many processing plants that bring natural gas production onto the WEI system. T-South is the mainline transportation pipeline from Northern BC (T-North interconnect) to the United States border at Huntingdon/Sumas. WEI's T-South pipeline transverses over some of the most difficult geological terrain in North America and was put into service in 1956.

SCP consists of a 24-inch pipe extending 300 kilometers from Yahk, BC to Oliver, BC in the Okanagan Valley. The SCP interconnects, in the East, to the Foothills pipeline that delivers supply from Alberta to the United States while in the West, SCP connects to the WEI's T-South pipeline. SCP was put into service in the early 2000's.

Foothills is 1,237-kilometer network of pipelines that sources Alberta natural gas and transports through the southern portion of British Columbia from Sparwood (BC/Alberta border) to Kingsgate (US border of Idaho). The original pipeline commenced service in 1981. This pipeline interconnects with Gas Transmission Northwest ("GTN") that continues through to Washington, Oregon, Nevada, and the California border.

The NWP system interconnects with WEI T-South at the Canadian/US border and GTN at Stanfield, Oregon. The NWP begins at Sumas, Washington and extends southeast through Oregon, Idaho, northern Utah, Wyoming, and southward into the San Juan Basin in southern Colorado. NWP typically flows north to south and, therefore, is heavily dependent on WEI's deliverability on a daily basis. There are also two storage facilities interconnected to NWP, Mist Underground Natural Gas Storage Facility ("Mist") located near the community of Mist in Columbia County, Oregon, and the Jackson Prairie Underground Natural Gas Storage Facility ("JPS") located in Southwest Washington.

The foundation of FEI's supply portfolio is mainline transportation that can service customers throughout the year. As such, diversifying access to alternate supply regions would strengthen FEI's supply portfolio and, if an opportunity of this kind arose, should be carefully considered (i.e., assuming the costs are reasonable for the consumers).

Producers consider the advantages and disadvantages of carrying the fixed costs of transporting natural gas, as these costs are generally significant. For example, a producer may need to balance accessing a premium priced market to offset the fixed costs of transportation and what shareholders and/or equity markets felt was necessary to diversify the producer's operating costs. Utilities consider the advantages and disadvantages of carrying the fixed costs of transporting natural gas when sourcing gas from suppliers, as these costs will be passed onto customers. Utilities will generally seek to balance security of supply with the economic benefits of sourcing gas supplies, while avoiding higher priced markets.

FEI has continued to maintain a strategic transportation resource portfolio. Responding to customer demand by contracting for incremental capacity and constructing new capacity when economically viable. The foundation of firm transportation, operating 365 days a year, has afforded consumers significant cost of gas savings and diversification of supply (i.e., Spectra Stn2 and/or AECO AB-NIT²). These cost savings will persist for those gas supplies that are currently serviced by the existing portfolio. However, given current market conditions, any incremental demand from consumers within FEI's and its peers' regions will require the construction of incremental transportation capacity to meet the markets call for supply – which will come with an associated cost.

II. Off-System Storage Resources

Off-system storage resource options exist upstream and downstream of FEI's core customer demand regions. Upstream, Aitken Creek is located northeast of Fort St. John, British Columbia. Downstream, the Mist storage facility is in Columbia County, Oregon and the JPS facility is located in Southwest Washington. A storage resource, and connectivity to mainline transportation, is critical to its usefulness. Upstream storage and its associated supply characteristic differ from those downstream of FEI's core customer demand regions. As such, I evaluated each resource in turn below.

i. Upstream Off-system Storage

Upstream, the gas supply regions of Northern British Columbia and Alberta differ to others in North America. Extreme low winters temperatures can create gas well performance issues, and

² ICE NGX - <u>https://www.ice.com/products/69723159/AB-NIT-Fixed-Price-Future</u>

from season to season, access to some remote locations can hamper the continuous supply of natural gas from producers. This is commonly known, to industry players, as winter freeze-off and spring break up. Natural gas production can be at risk following a gas well freeze-off as many operators must inspect pipe, valves, compressors, reservoir integrity, etc. prior to restarting production. Further, spring break up delays will affect those wells that have been drilled but not yet completed (i.e., uncompleted natural gas wells that have been drilled but not yet undergone well completion activities to start the production of gas). Northern upstream storage therefore minimizes market production variability while the supply region is continuously adjusting to meet its own economic balance and fulfill commitments to the market.

Aitken Creek, for example, has a working capacity of 77 billion cubic feet. This resource, combined with other supply commitments, can provide 168,000 GJs/day (approx. 12%) of the supply portfolio for FEI downstream customers during the winter (November - March). Supply sourcing in the region also typically carries a lower price in the summer months, resulting in a cost of gas benefit for customers during the winter withdrawal season. Beyond the significant savings to FEI's customers from avoiding the Huntingdon/Sumas market, Aitken Creek capacity provides supplemental gas pricing benefits even when incorporating the incremental costs of transporting this supply source to FEI customers. Finally, the facility offers operational flexibility for supply disruptions during busy maintenance periods for the entire energy industry.

ii. Downstream Off-system Storage

FEI and peer utilities located in Washington and Oregon share similar customer load profiles. In particular, seasonal demand is higher in the winter months and lower in the summer months. These peer utilities also rely on T-South transportation to ship gas supplies from northern British Columbia to the US border. Approximately 70%³ of the gas supply transported through T-South is exported to the US. Southern storage injects gas supply during the summer, commingled and accumulated, for future winter withdraw. The north to south hydraulic nature of T-South mainline transportation means FEI's contracted downstream storage is utilized by displacement (i.e., gas supplies destined to travel south of British Columbia, during the storage withdraw period, will be diverted to the Lower Mainland and the associated diverted gas supply will be replenished by

³ Canadian Energy Regulator, <u>https://www.cer-rec.gc.ca/en/data-analysis/facilities-we-regulate/pipeline-profiles/index.html</u>

storage capacity at Mist and/or JPS). This leaves the southern US market whole while the utilization of mainline transportation remains high.

Southern off-system storage facilities in the United States do not rely solely on the T-South system to utilize natural gas capacity. These facilities are interconnected to the NWP network, which gains access to alternative gas supply regions in Alberta (i.e., through interconnection with GTN), Wyoming, and Utah. FEI is sourcing 100,000 GJs/day and 110,000 GJs/day respectively from JPS and Mist storage facilities (i.e., 15% of FEI's peak day requirements), which demonstrates the importance of these off-system resources as part of FEI's gas supply portfolio. The NWP interconnecting pipeline network is fully contracted, limiting all incremental deliverability, from the JPS and/or Mist storage facilities. Any expansions in these regions will come at a cost and its recovery will be reflected in the contracting of service(s). The market is responding to demand in the region with the proposed Northwest Pipelines Gorge and GTN's XPress expansions. If these expansions proceed, incremental deliverability from interconnected storage facilities could be explored. For FEI, its ability to make use of any incremental resources secured in this region will be subject to displacement availability on the T-South transmission system.

As demonstrated above, off-system storage remains a necessary tool within FEI's resource portfolio. The resource helps meet median customer demand profiles during winter seasons (151 days from November through to March), deployable on a monitored basis as winter weather volatility persists. The deliverability of off-system storage relies on the pipeline infrastructure that it is interconnected with, which must be considered when developing a gas supply portfolio. Further, as it stands, all interconnecting services are fully utilized and the only way to increase capacity is through infrastructure expansion(s). As a potentially interested counterparty that manages an evolving resource portfolio, researching the cost/benefit of supporting (i.e., through contractual arrangements) such expansion(s) is important. In addition, determining the timing of any expansion is important as these types of projects can take considerable time to complete.

Given the heavy use of off-system storage, storage customers (including FEI) must remain diligent in monitoring the reservoirs deliverability (i.e., during maintenance and high demand periods).

III. Peaking Resources

Peaking resources are those resources that are deployed during a "super peak" when customer demand reaches its peak. This typically coincides with colder-than-normal temperatures and the visual response, by consumers, to winter conditions (e.g., snowfall).⁴

Ideally, peaking resources would be accessible by consumers within a 24-hour period and deployable for multiple days. Sizing the resource, from an energy supply perspective, will also depend on the velocity of incremental demand during regional weather changes. As depicted in Figure 3 below, the onset of colder than normal weather can create exponential increases in customer demand. The charts in Figure 3 show multiple winter weather scenarios. The left chart shows a colder-than-normal winter weather event followed by an average winter weather event and the right chart shows two consecutive colder-than-normal winter weather events (i.e., as heating degree days rise (temperature declines), customer demand increases). Utilities must model and forecast for such scenarios and, as discussed below, develop a portfolio of peak resources to manage these potential weather events.





The reliance on consistent weather forecasts is also critical to the operation of the distribution system that is highly sensitive to consumer demand. Velocity will also change even within a smaller geographical service area. Examples include the variability of consumer demand from coastal communities such as the City of Victoria versus the Lower Mainland. Though

⁴ Assuming temperatures are the same with and without snow, a customer will react differently to the sight of snow.

geographically close in proximity, their coastal weather differences can drive unique demand responses. While weather forecasting technology continues to improve, radical changes and its severity cannot always be predicted.

On-system peaking resource(s) must be able to provide (at a minimum) continuous availability to bridge any gaps in winter weather events. As such, in anticipation of a forecasted weather event, a peaking resource can remain idle (i.e., operational ready to deploy gas supply) while actual weather unfolds and can typically be ramped up within hours. This operational flexibility enhances the assets value compared to an alternate resource that requires 24-hours notice to be deployed (e.g., off-system storage and mainline transportation) or if there is an unplanned operational mainline transportation disruption. For example, when a utility elects to nominate an off-system resource and the weather forecast doesn't materialize as expected, it will be left with excess gas that will require mitigation. After mitigation, the off-system resource will be depleted should future weather events or unplanned outages occur. This contrasts with an on-system peaking resource which can be deployed when access to gas supplies from mainline transportation or off-system storage faces an unplanned outage.

Figure 4 below provides additional context regarding the velocity of load (i.e., peaking volume throughput) based on the demand of FEI's RS 1, 2, 3, 4, 5 and 7 customers. The dashed lines show YVR weather⁵ (°C) consecutive winter terms (November to March) from 2010 to 2022. The green line shows average winter temperatures during this term, while the red line shows the winter low temperature as recorded in a given month during the same period.

⁵ <u>https://climate.weather.gc.ca/climate_data/daily_data_e.html?StationID=51442</u>





I identified three results in my review of the data provided by FEI regarding RS 1, 2, 3, 4, 5 and 7 customers. First, in each year reviewed there are considerably more days within the throughput range of 400,000-500,000 GJs, when compared to the other higher throughput ranges. Second, FEI experiences a significant increase in system throughput when temperatures drop below historical lows. Third, there has been an underlying shift in customer demand since 2019. In particular, from 2010-2019, system throughput remained between 400,000 and 500,000 GJs during average gas winter temperatures for an average of 60 days, while between 2019-2022, this average increased to 85 days (i.e., 15 additional days of higher than historical average system throughput). Further, the 2019-2022 period also saw a significant increase in throughput when temperatures dropped below -5°C, reaching the 600,000-700,000 GJs and 800,000+ GJs ranges.

In conclusion, higher-than-historical-average system throughput during average winter temperatures, an increase in the number of days average winter demand persists, and the increase in peak day demand, operationally managing winter weather anomalies becomes increasingly more difficult. As such, on-system peaking resources are, and will likely continue to be, an important part of FEI's resource portfolio.

In the sections below, I provide a detailed review of the peaking resources accessible to FEI.

i. Interruptible Customer Peaking Resource

Access to "on-system" natural gas supplies, through the displacement of one customer (i.e., interruptible) for the benefit of others (i.e., core customers), is a method of managing capacity which should be maximized wherever possible.⁶ Utilities and pipelines offer interruptible tariffs with the understanding that the customer can be interrupted and receive compensation for their displacement. Typically, these customers have an ability to fuel switch, leaving their supply and capacity available for other customers. The compensation provided to these customers could be a lower tariff, or compensation mechanism(s) to cover the underlying fuel cost and any operational disruption nuance(s) (i.e., increase labor expenses). Regardless, the customer acknowledges that their service can be interrupted.

Based on information provided by FEI, its Lower Mainland region has seen a 10% decline in the number of interruptible customers between 2018 and 2023. As such, there is less of this contractual peaking resource available to FEI.

ii. LNG Resource (On-system storage)

LNG peak-shaving plants ensure that adequate supplies of natural gas are readily available when demand is at its peak. Throughout North America, natural gas transmission pipeline operators and/or utilities, use these facilities to liquefy natural gas for storage. When demand is lower (i.e., typically in the summer), the operator can liquefy natural gas for above ground storage and then regasify the LNG when demand is high.⁷

⁶ For the purposes of my analysis, I assume interruptible customer peaking resources may include capacity displacement and/or sequestering delivered natural gas supply for FEI's own use.

⁷ <u>https://www.phmsa.dot.gov/pipeline/liquified-natural-gas/lng-facility-siting</u>

These facilities typically provide reliable supply in areas where pipeline/distribution capacity limitations and/or weather conditions tend to cause supply and demand discrepancies. The development of LNG-peak shaving resources is generally strategic. In particular, utilities will undertake a deployment analysis to determine the cost/benefit of a given facility when compared to alternate resources such as compression addition(s) and infrastructure looping. Underlying gas supply cost avoidance is another benefit that utilities can gain as part of this evaluation process.



FEI's Mt. Hayes LNG facility continues to provide capacity, commercial peaking resource, and system operational benefits to customers on Vancouver Island and beyond. For example, the facility allows FEI to manage operational and balancing parameters where distribution assets interconnect with mainline transmission systems. This is especially important during periods of peak winter demand (i.e., 6 am and 6 pm as consumers go about their daily routines). While there are typically flow parameters in place to ensure system off-takes do not adversely affect mainline transmission operations during high draws from distribution assets, deploying a peak shaving facility during bursts of demand can alleviate some of the peak draws on mainline transportation interconnections. Within the regions FEI serves, assets for peak shaving can offset alternative options such as securing long-term third-party transportation and off-system storage, as well as avoiding one of the most volatile pricing indexes (i.e., constrained supply region with high demand) in North America, Huntingdon/Sumas.

iii. Peaking Supply Resource (Seasonal Supply)

Peaking supply resources, governed by contractual agreements from third parties, could be a viable resource. This service requires an evaluation of the third-party credit worthiness, their proprietary physical assets to meet the contractual commitments offered under an agreement, and historical operating experience in the regions FEI operates. The third-parties offering this service would need to maintain firm service transportation, storage capacity, and/or their own proprietary

interruptible customer base to emulate FEI's current displacement capabilities. I review third-party peaking resources, and their deliverables, further in my response to Question 2 below.

IV. Conclusions to Question #1

The physical location of FEI's core customer demand is in a region that is challenging to provide safe and reliable service at a reasonable price. As discussed above, a gas supply portfolio has to be constructed from the resources that are available in the market, while also considering the resources contractual commitments including, tenure, carry costs, and any ongoing benefits when secured.

FEI relies on third-party resources to service 100% of its consumer portfolio (i.e., to fulfill core customer demand from FEI's proprietary distribution, transportation, and/or on-system peaking assets, FEI requires interconnections with third-party upstream and downstream off-system storage, third-party mainline transportation, and third-party gas supplies to gain access to markets). Although FEI can leverage existing supplemental resources such as on-system LNG and interruptible arrangements, these resources must be supplied with gas from third-parties. Any interruption to these resources hampers the deliverability of supply and ongoing consumer satisfaction. With that said, price volatility follows an inconsistency in market dynamics driven by weather and infrastructure constraints servicing this demand. An ability to insulate oneself from price volatility generally correlates to the increased reliability of a resource. Alternatively, having a resource that can be deployed during price volatility brings immense commercial/financial value, as well as increased flexibility.

It is important to note, resource portfolio development continues to be a challenging task especially when resource availability becomes finite and peer regions are vying for similar assets to meet their needs. Regardless, an inflection point will transpire when the need for incremental infrastructure will be necessary to meet the current trajectory of consumer demand for energy. The necessary lead time for major projects is critical, and from what I have observed, FEI has continued to recommend key elements of an optimal resource portfolio to service its customers.

Response to Question #2

FEI has, since 1971, included on-system LNG in its supply portfolio as a peaking capacity resource (i.e., rapid send-out capability, backed by sufficient LNG to meet short-duration peak load). We seek your opinion on the peaking capacity alternatives available to FEI so as to inform our understanding of the financial value of on-system LNG from a gas supply portfolio perspective by reference to avoided supply portfolio costs. In particular:

- A. Based on your assessment of the natural gas markets accessible to FEI, would FEI be able to contract for peaking resources (capacity/energy supply) as a potential alternative to on-system LNG in its gas supply portfolio, and if so on what terms?
- B. If appropriate peaking resources (capacity/energy supply) are unavailable to FEI in the market, what other potential investments could FEI explore as a potential alternative to on-system LNG as peaking resources (capacity/energy supply) in its gas supply portfolio? For clarity, in this regard we are requesting that you identify potential peaking supply (capacity/energy supply) alternatives based on your understanding of the regional system, as opposed to an undertaking a financial analysis of those options.

In my opinion, FEI would not be able to contract for peaking capacity resources that I would consider dependable, beyond those that are currently committed, as an alternative to on-system LNG storage. The deployment duration and dispatch capability remain the main criteria for security of supply from a peaking capacity resource. Due to the nature of the regional demands, third-party off-system storage and companies offering peaking gas supply arrangements, are constrained by the ability to transport the underlying supply of natural gas as well as the potential for unplanned outages. Furthermore, based on the results of my market research evaluating third-party arrangements, the costs for peaking resources are extremely expensive, do not support a consistent long term supply resource, and would require a portfolio of participants to be able to meet FEI winter demand. To put this in perspective, FEI would be competing/accessing the Huntingdon/Sumas gas supply market, on the coldest days of the winter, for significant volumes historically destined to the PNW (rapidly escalating pricing throughout daily trading hours). These factors, when taken together, are not conducive to contract for dependable peaking supply resources.

In the absence of readily available dependable peaking resources (including sufficient on-system storage), FEI would need to make significant investments (including capital investments and/or contractual capacity commitments) to meet its peaking capacity needs. The types of investments could include expanding mainline transportation, and the associated interconnected storage facilities, and/or developing on-system peak shaving LNG. Building these types of resources takes time, especially capital projects that require consultation with multiple parties throughout

the planning and construction phases. Constructing a mainline transportation resource, for the purposes of meeting winter demand, would also result in inefficient utilization because it would remain underutilized in the non-winter months. Any investments considered by FEI should consider the assets long-term utilization parameters and the timing in which the resource could be deployed.

In the sections below, I first consider what alternative resources are operable in the region, determine if these alternatives are constrained, and discuss whether any of these investments could be an alternative to FEI's proposed on-system LNG project at its existing Tilbury facility. I begin by discussing on-system storage, before addressing off-system storage, and contractual third-party peaking gas supply.

I. "On-System" Gas Supply

On-system resources, are located where customer demand exists, and are the most dependable resources when compared to off-system storage and contractual third-party peaking supply agreements. As previously discussed, interruptible customers and LNG peak shaving gas supply can be deployed more quickly than other resources (i.e., in advance of a 24-hour notice period) and instantaneously support capacity.⁸ This assumes FEI has: (1) sequestered Interior gas supply that has already been moved through the T-South pipeline; and/or (2) initiated capacity displacement from interruptible customers in the Lower Mainland.

While displacement from interruptible customers is contractual and predicated on the customer reducing or ceasing their use of natural gas, this type of resource is more deployable than third-party contractual agreements for peaking capacity. This is because an interrupted customer has already moved gas supplies onto the FEI system, while a third-party must rely on transportation and/or storage resources to fulfil their commitments.

FEI's Mt. Hayes LNG and Tilbury LNG facilities continue to provide operational and commercial value for FEI customers. From an operational standpoint, the integration of these facilities provides sound flexibility to meet significant velocity swings in consumer demand. Furthermore, from an avoided cost perspective, with high demand correlating to higher energy prices, the

⁸ Here support capacity refers to gas supplies that have already been moved into the utilities distribution or pipelines mainline network and does not require additional compression to move gas supplies to a desired location.

facilities are serving their design purpose while minimizing price exposures to Huntingdon/Sumas. In particular, the Mt. Hayes LNG facility continues to meet core demand "pull" from consumers throughout the Sunshine Coast and Vancouver Island. The combination of compression, and the strategic location of Mt. Hayes on-system gas supply, ensures reliability for Vancouver Island customers. A resource in this location also helps minimize the need to move incremental gas supplies through the Lower Mainland system that could be experiencing coincidental demand constraints.

In conclusion, FEI's existing on-system storage resources are utilized to their designed capabilities. Persistent weather variability, coupled with high velocity demand from consumers requires FEI to continue to rely on on-system resources. Furthermore, interruptible customers remain a key element to peak day management, despite their declining numbers. These assets continue to be strategic to FEI in meeting on-system capacity constraints during peak demand and insulate consumers from the financial exposures in spot prices at Huntingdon/Sumas.

II. Off-system Storage

While, increasing existing upstream (i.e., Aitken Creek) and/or downstream (i.e., Mist and JPS) off-system storage is theoretically an alternative capacity resource, in practice, these facilities are currently fully contracted and will require expansions to meet any incremental capacity requests from FEI. In addition, increases to off-system storage resources will remain contingent on the deliverability of interconnected mainline pipeline networks (i.e., which are already facing capacity constraints).

For Aitken Creek, the interconnected deliverability of the T-South system is limited, as shown in Figure 5 below.⁹ ¹⁰ The upper chart provides T-South rated capacity (gold line) and actual system throughput (export in green and intra-Canada in blue) for the terms from 2006-2023. The lower chart shows the daily characteristics of T-South system from 2021-2023, which includes rated capacity (red line), actual throughput (grey line), and authorized capacity (blue line).

⁹ <u>https://www.cer-rec.gc.ca/en/data-analysis/facilities-we-regulate/pipeline-profiles/natural-gas/pipeline-profiles-westcoast-bc-pipeline.html</u>

¹⁰ <u>https://noms.wei-pipeline.com/customer-activities/pipeline.php</u>

Figure 5



The data shows winter throughput on the T-South system has surpassed rated pipeline capacity while summer throughput-to-rated-capacity utilization has averaged as high as 80%. Also, please note the authorized capacity adjustments during summer periods while the pipeline conducts maintenance (i.e., blue line limits grey line). Depending on the type of maintenance, capacity can be limited, impeding shippers' deliveries. The high summer demand identified in the lower chart corresponds to the higher summer export utilization, as identified in the shaded area of the upper chart.¹¹

¹¹ Higher summer utilization is an important observation during this analysis. All pipelines, storage facilities, and large industrial operations (i.e., Commercial LNG, Refining, etc.) require ongoing maintenance, a period that will affect the operations of resources in the region. A facility that can offer an alternative supply resource during these disruptions carries commercial/financial value (i.e., peak shaving LNG, southern off-system storage displacement, and/or line pack utilization). For example, commercial LNG economics are driven by world prices, and refining facilities are influenced by the liquids market they serve. An asset that can backstop natural gas supply, while mainline transportation maintenance is conducted, minimizes disruptions to an industry's feedstocks and revenue streams.

Considering the current contracted capacity of off-system storage and the associated winter utilization of T-South, incremental storage from Aitken Creek is therefore not a viable resource option without significant investments in mainline transmission and the storage reservoir. Furthermore, southern off-system storage facilities (i.e., Mist and JPS) remain fully contracted, eliminating an opportunity of securing incremental capacity without triggering an expansion. While a financial commitment for incremental southern off-system storage is likely to be less than committing to incremental WEI transportation and Aitken Creek storage,¹² the deliverability issues discussed above remain.

III. Third-party Peaking Gas Supply Contracts

Third-party peaking gas supply contracts are governed by agreements that provide firm service gas supplies delivered at a pre-determined delivery point. The contract to deliver, when called upon, is only as good as the delivery options the third-party utilizes. The entities offering this service should possess transportation, storage and/or their own portfolio of interruptible customers to physically backstop the contractual commitments for peaking gas supply.

These commercial arrangements typically maintain 24-hour notice periods prior to the deployment of the resource. A third-party offering this service will want to recover the underlying costs of maintaining their asset portfolio while earning a fair return for its use during the period it is contracted. In particular, a third-party's fee would consider the "stand by" nature of the arrangement (i.e., the resources remain at the ready to deploy whether it is utilized or not). The arrangement typically carries a contract fee that includes a demand charge (i.e., monthly fee over the tenure of the agreement), and a commodity charge (i.e., a price charged on the day any volumes are used).

The demand charge would incorporate underlying carry costs associated with pipeline(s) tolls, storage tariff(s), and any other proprietary services that are required to support a peaking supply arrangement. Though a peaking supply arrangement may lay dormant during summer months (April-October), its underlying costs do not. Cost recovery of un-utilized assets will be incorporated into winter periods when the arrangement is active (November-March). The dormant period costs will be market driven, meaning they will fluctuate in price, driven by the success of

¹² The contractual tenure commitments, to drive investments, is likely lower, and this resource alternative does not incur summer capacity mitigation exposure a pipeline alternative does.

Figure 6

cost mitigation attempts during the summer. Higher success of mitigating costs during the summer may translate to lower demand charges for winter peaking arrangements.

A third-party gas supply peaking arrangement also includes a commodity charge portion, which I discuss in further detail below. Since these agreements are required for daily deployment, their underling commodity charge(s) are based on daily spot pricing. Daily spot prices trade in a range throughout a trading day, and in some instances, can trade at a significantly wide range (as depicted in Figure 6 below). The figure shows a sample **range** of price, from the highest to the lowest, on any given day prior to the day's settled price.



For the purpose of selling peaking gas services, the third-party will want to ensure daily volatility is minimized by constructing a contract that explicitly details that the buyer of the resource is paying the higher range of prices on any given day. The higher price range ensures that the third-party selling the resource, is not economically disadvantaged by offering the service at average prices.

Figure 7 below shows the range of price fluctuation on a given day. The daily transaction data, as recorded by Platts from 2016-2022, captures all trades consummated (i.e., when a buyer and seller trade) throughout the day prior to the end-of-day settlement. During the winter (i.e., November-March) prices ranged from \$0.8689/GJ to \$47.30/GJ, while in the summer (i.e., April-October), prices ranged from \$0.1261/GJ to \$4.23/GJ. The values presented in the chart are derived by subtracting the high price from the lower price.




Figure 8 below shows actual price settles (i.e., not the range) for the term January 2017 to July 2023.¹³ The daily prices (blue filled line) have been stacked from the lowest price of \$0.73 CDN/GJ to the highest price at \$201.03 CDN/GJ. The average daily settled price over the entire term was \$5.40 CDN/GJ. 50% of the higher data range averaged \$8.21 CDN/GJ while the lower averaged \$2.58 CDN/GJ. The table embedded within the figure shows the daily high/low price ranges (i.e., not price settlement). For example, the actual settled price of \$201.03 had a Daily High Price "Max" of \$224.70 CDN/GJ, or \$23.67 CDN/GJ higher than the actual end-of-day price. The table embedded in the figure also shows the volatility in pricing within a given trading day.

¹³ The data represented has not been manipulated by removing time segments that could include price influences from mainline pipeline maintenance and/or outage as these events will likely occur in some form or another in the future. Further, the time period selected for this analysis was not selected because it included a pipeline outage, but rather, because it is included within the period of time analyzed.





The data shown in Figure 8 demonstrates the financial value potential a peaking resource can have operating in the Huntingdon/Sumas index region. For example, avoiding or participating in trading activities during the highest 60 days of pricing (which averaged \$48.18 CDN/GJ) would have a significant impact on a utility's rates.

In the sections below, I provide my conclusions in response to the specific sub-questions to Question 2.

Response to Question #2A

A. Based on your assessment of the natural gas markets accessible to FEI, would FEI be able to contract for peaking resources (capacity/energy supply) as a potential alternative to on-system LNG in its gas supply portfolio, and if so on what terms?

My assessment of the natural gas market has determined that FEI would not be able to contract for peaking capacity resources as a potential alternative to on-system LNG. First, off-system

storage alternatives remain constrained. In particular, as discussed above, the Aitken Creek storage option would require incremental WEI transportation capacity, and the Mist and/or JPS storage facilities remain fully contracted and would require their own expansions to meet any FEI incremental calls for capacity. Second, third-party peaking gas supply contracts are not a dependable resource and also remain cost prohibitive. To confirm my analysis of third-party peaking supply, I provide the following market research assessment which confirms my conclusion that FEI would not be able to contract for peaking resources (i.e., capacity/energy supply) as a potential alternative to on-system LNG in its gas supply portfolio.

FEI would require a portfolio of third-party peaking contracts to meet the needs of its gas supply portfolio as no one counterparty can provide sufficient supply. The pool of available counterparties is also limited and there is considerable turnover in participants in the marketplace. In particular, I only identified 9 entities that maintain T-South capacity with sufficient proprietary supply to be considered as a portfolio participant for FEI. Furthermore, due to the nature of the arrangement(s), these third-parties would face similar mainline transportation disruptions that FEI does. In an attempt to alleviate this risk, FEI would need to contract "more than face value" supply arrangements to offset the potential of being cut during a mainline disruption.

The availability of these resources will also remain market-based, potentially resulting in even higher premiums than previously witnessed. Furthermore, marketing companies tend to keep their proprietary trading assets free of any contractual commitments to extract maximum value from the market. With winter demand high, and an increasing summer demand persisting in the Pacific Northwest, these companies will be less interested in committing resources that could inhibit profits. All of these factors ultimately limit FEI's ability to contract peaking resources.

Response to Question #2B

B. If appropriate peaking resources (capacity/energy supply) are unavailable to FEI in the market, what other potential investments could FEI explore as a potential alternative to on-system LNG as peaking resources (capacity/energy supply) in its gas supply portfolio? For clarity, in this regard we are requesting that you identify potential peaking supply (capacity/energy supply) alternatives based on your understanding of the regional system, as opposed to an undertaking a financial analysis of those options.

With no capacity resource alternatives to on-system LNG currently available, the other alternatives contemplated would require significant financial investment(s) (whether contractual

or otherwise) to support expansions to mainline transportation and/or off-system storage. Keep in mind, expansion projects of this nature require the necessary time to implement, create asset utilization variability, and should include an analysis of incremental long-term benefits for the call on expanded capacity. Investments in alternative resources are also costly and shift market characteristics for long periods of time. In the sections below, I provide an overview of peaking supply resource alternatives based on my understanding of the regional system FEI operates within. My analysis does not suggest or contemplate estimates to the capital costs for these resource alternatives. FEI would need to consider the timing necessary to deploy each alternative, including engaging with interested parties, organizing investment capital (if necessary), gaining regulatory approval, and final construction deployment. All of these steps generally take considerable time to complete. During this process, FEI would have to manage customer expectations for secure, reliable, and cost-effective resources.

The following peaking supply resource alternative analysis is based on what resources <u>could</u> be implemented with the understanding that capital investments would be required to increase the call on capacity. I have broken my analysis into two parts. First, I review mainline transportation capacity increases through T-South and/or SCP, and any interconnected off-system storage. Second, I review southern alternatives such as increasing southern off-system storage displacement and/or increasing on-system LNG peak shaving.

I. Analysis of Mainline Transportation

In the sections below I assess the following two mainline transportation solutions: (i) increasing capacity on the T-South system; and/or (ii) increasing capacity on the SCP. There is an asset utilization consideration with pipelines expansions for peaking supply, which is avoided with storage. In particular sizing an expansion to provide winter peaking capacity for FEI leaves significant underutilization in the other times of the year that require mitigation efforts to offset those costs.

i. Increasing T-South Capacity

In the absence of peaking resources, FEI could consider sourcing additional T-South firm service from WEI. As discussed below, WEI has responded to market demand in the past by offering an open season for market interest in expanding T-South capacity.

First, in the Spring of 2017, WEI was successful in securing firm interest to expand T-South. This open season saw significant market interest for a T-South expansion. In particular, WEI received a competitive response with participants bidding contractual firm service tenures out as far as long as 60 years, which is the longest that I am aware of in my 30-year career. In addition, the demand for incremental capacity surpassed the 190 MMcf/d offered by WEI in its open season. This was the highest amount of incremental capacity it could develop without significant increases to existing tolls. Beyond this level, the incremental gains in capacity, versus the capital deployed, would not be as efficient.

Second, in November 2022, WEI announced another open season to gauge shipper interest in a further annual capacity increase for T-South of 300 MMcf/d (i.e., The Sunrise Expansion Program).¹⁴ Following a successful open season, that resulted in requests for additional transportation capacity, WEI is proposing this expansion to have an in-service date by late 2028.

Assuming WEI responded to FEI's call for incremental capacity above what had already been committed, FEI would need to evaluate the cost of this capacity and the duration of the associated commitment. In particular, while this type of annual commitment would meet FEI's winter demand requirement, it would also require FEI to mitigate capacity into the market during summer when demand is lower. Furthermore, mainline transportation expansion would trigger additional capital investment for any associated off-system Aitken Creek storage. This assumes a one for one matching of transportation capacity to meet send out capabilities for storage.

ii. SCP Expansion

As previously discussed, in the East, the SCP interconnects to the Foothills pipeline that delivers supply from Alberta to the United States while, in the west, SCP connects to the T-South mainline 240 km north of Vancouver at Kingsvale. Figure 9 below provides the historical southern throughput on the Foothills pipeline where SCP interconnects.¹⁵

¹⁴ https://www.enbridge.com/projects-and-infrastructure/projects/sunrise-expansion-program

¹⁵ <u>http://www.cer-rec.gc.ca/en/data-analysis/facilities-we-regulate/pipeline-profiles/natural-gas/pipeline-profiles-foothills.html</u>



Figure 9

Although the most recent data shows monthly throughput reaching 90-95% utilization, the NGTL West Path pipeline expansion ("NGTL Expansion") has been approved, underpinned by 250 MMcf/d of firm service contracts. The NGTL Expansion will increase North to South flows from Canada to the US and confirms that the Southern US demand continues to seek Alberta sourced supply. This expansion could represent incremental SCP take away capacity if the SCP were to expand through compression/looping. Increasing SCP throughput would have to include a parallel expansion on T-South from Kingsvale to Huntingdon/Sumas, or an extension of SCP from Oliver to Huntingdon/Sumas. All SCP expansion initiatives would provide the benefit of newer infrastructure than the T-South alternative, and would diversify FEI's supply and access to supply.

As part of my review of an SCP expansion I identified a number of supplemental benefits that should be considered when reviewing the embedded costs of the mainline transportation alternatives. First, the interconnectivity of SCP to a larger supply basin provides the financial and physical price risk management tools to insulate FEI customers from price volatility. Such insulation is more difficult to achieve at Spectra-Stn2. In particular, the Alberta supply basin increases the number of credit worthy counterparties FEI could engage, increasing the diversity of its supply portfolio including RNG. Second, although FEI would have the option to access more off-system storage, unlike a T-South expansion, this would not necessarily be required as the

physical liquidity is robust enough to meet FEI needs on any given day. Third, an SCP expansion would enable FEI to have greater access to a market that is aggressively, with the support of Federal and Provincial funding, exploring Carbon Capture Utilization and Storage ("CCUS"). This could represent environmental benefits for FEI customers and peer utilities throughout the Pacific Northwest; assuming that a SCP expansion was supported by other shippers during an open season process.

II. Analysis of Southern Resource Alternatives

In the sections below I assess increasing off-system storage in the US. Prior to delving into this alternative, I note the following.

First, as with the option of committing to more long-term mainline transportation, the intended use of the southern resource alternatives may shift as FEI's resource stack evolves over time. Even so, these resources will generally maintain underlying commercial deployment value over their useful life. For example, annual demand shifts do not represent a one for one change to peak demand. Annual demand could decline while peak demand escalates, requiring a different stack of resources to manage a shift in consumer demand. Both of these resource alternatives are intended to be deployed during the winter, thereby avoiding the summer mitigation costs associated with a mainline transportation alternative.

Second, the regional market has become accustomed to accessing the capacity on T-South that is contracted for the Woodfibre LNG facility but being released into the market until the facility enters service. Both the summer and winter periods will be shorter in supply (i.e., estimated at 300 MMcf/day) when the Woodfibre LNG facility becomes operational. In particular, while T-South will remain fully utilized, markets south of Huntingdon/Sumas will not be able to access this supply all year once the Woodfibre capacity is no longer available in the market in 2027. This shift will create a tighter price environment for buyers. The Sunrise Expansion Program will only offset the lost Woodfibre capacity, returning the market to its current constrained state.

A southern off-system storage resource from Mist and/or JPS is contractual in nature and has a number of benefits. It does not carry investment capital risk for FEI and its contract commitment would cease if FEI determined the resource was no longer required (i.e., it is believed the tenure of the contract commitment would be less than a T-South alternative). Furthermore, since this is

a storage option, its utilization would be seasonally dependent and not carry summer mitigation costs that a mainline transportation option requires.

There are nonetheless a couple of drawbacks to consider with a southern off-system storage option. First, there is a likelihood of a call back provision(s) from the counterparty offering the service. For example, FEI does not currently hold renewal rights for capacity with the Mist storage facility. This option provides an opportunity, by the peer utility, to terminate arrangement(s), sequestering the capacity to meet their own peak customer demand. Second, the southern off-system storage option lacks the resiliency value that would be provided from an on-system LNG peak shaving facility.

Conclusions to Question #2

In my opinion, the development of a proprietary on-system asset, such as the proposed TLSE Project, has advantages over other types of infrastructure investments from a gas supply perspective. The TLSE Project would provide FEI with operational backup for disruption: (a) to existing off-system storage and/or mainline transmission; and (b) that may not necessarily affect FEI customers, but those of peer utilities, that source gas supply from Huntingdon/Sumas. The proposed TLSE Project's proximity to a larger US export market could also be a benefit to FEI's customers. In particular, peer utilities in the US, servicing a growing gas-powered electricity and industrial demand are, and will be for the foreseeable future, evaluating the cost/benefit of limiting pricing exposure to Huntingdon/Sumas, in response to growing peaking demand, by either: (a) sourcing peaking services like what could be provided by the proposed TLSE Project; or (b) committing to long-term mainline transportation. Furthermore, while meeting the needs of FEI's customers for safe and reliable service, a proprietary LNG peaking facility will continue to carry long-term value for the utility and its customers. If the demand profiles of FEI's customers were to shift over time (i.e., lowering annual demand while maintaining the need for winter supply), the ease of de-contracting mainland transportation is more appropriate than shedding reliable onsystem capacity as on-system storage is designed to be deployed only when it is required.

Response to Question #3

Assuming a hypothetical scenario where the TLSE Project is constructed at 800 MMcf/d and 3.0 Bcf and FEI does not require the full regasification capacity and stored LNG for its own resiliency and gas supply requirements, do you expect that FEI would be able to monetize its uncommitted peaking capacity by selling it in the gas markets? If so, how much daily send-out capacity could the market reasonably absorb.

Assuming that the TLSE Project is constructed, and FEI were to have spare capacity that is not required to meet customer demand or resiliency, FEI would be able to generate revenues to offset the cost of service of the facility by selling its excess supply into the market. Based on my assessment of the available supply and demand in the Huntingdon/Sumas natural gas market, and assuming current market conditions persist, I expect the daily market can reasonably absorb 300-400 MMcf/d of natural gas across multiple days (e.g., 10 days) during winter without influencing daily prices in a manner that could limit monetization values (i.e., materially decreasing the revenues generated through mitigation into the market). In the sections below, I discuss the drivers of increased natural gas demand and provide a detailed discussion of my market absorption analysis.

I. Demand for Natural Gas is Expected to Continue to Increase Due to Several Drivers

Managing the financial value and any associated benefits of an asset cannot be reviewed in isolation. By virtue of their infrastructure connectivity, British Columbia, Alberta, the States of Washington, Oregon, Idaho, Utah, and California share similar energy challenges. In the discussion below, I summarize the drivers of market demand, such as: (i) seasonal weather; (ii) social and economic pressures to reduce GHG emissions; (iii) advancement of new technologies; and (iv) large industrial development, which inform my assessment of the future demand (post-2027) for natural gas resources in the Huntingdon/Sumas region.

i. The Impact of Seasonal Weather on Demand

Weather from Northern British Columbia to the Southern State of California varies from season to season due to its topography and geographical location. Typically, the integration of a region's electric and natural gas systems will operate harmoniously, moving resources from one region to another through a multitude of pipelines and transmission lines. However, during winter weather conditions, one region's system will draw on another to fulfil its customer demands. This may include a scenario where warmer than normal temperatures, in Oregon and California, create significant electrical demand for air-conditioning that is serviced by an assortment of hydro, wind,

solar and natural gas generation throughout their own region but also those regions they are interconnected with. Alternatively, colder than normal temperatures, in that same region, will draw on the same resources but the energy mix deployed will be different. Weather remains unpredictable, and there is the likelihood that regions will experience coincidental peak demand pressure from consumers regardless of their region. Taxing infrastructure to its limits should ensure that each region has a diversified portfolio of resources and, in particular, resources that provide security of supply, while remaining economical to the consumers they serve.

An example of natural gas usage in a region that has extensive renewable resources is California. Understanding that California is a significant distance from British Columbia, the State's energy landscape has evolved over a long period of time with a significant emphasis on the integration of gas and power systems and the advancement of wind, solar and battery storage technologies. Though tremendous progress has been made to reduce the use of fossil fuels in California (i.e., as illustrated in Figure 10 below which shows the time required for new energy technologies to be into integrated into the California market), the State's dependency on natural gas remains because it is a reliable, economical, and dispatchable resource when compared to the current state of renewable resources.



Figure 10

Despite the continued importance of natural gas in the California energy market, the share of renewable energy in the California market is gradually increasing. For example, on April 30, 2022,

the California Independent System Operator ("CISO") set a record of using 100% renewable energy for approximately 1 hour. However, CISO continues to forecast that natural gas will remain in the mix of assets well-beyond 2045. Again, the cost of transiting the resources in an energy market is expensive and can only progress at a pace in which the consumer can absorb.

Another region forecast to go through significant resource growth challenges is the Pacific Northwest. Within a period of 24 to 48 months, electricity in the region is forecasted to meet capacity for winter and summer weather peaks. In addition, there remains a tremendous effort to minimize existing coal fired generation. Though retirements had been scheduled, they have been postponed while utilities re-evaluate resource portfolios. As shown in Figure 11 below, the energy market in the Pacific Northwest is constrained in both the summer and winter periods.¹⁶



Figure 11

Therefore, while no one resource can be deployed to meet growing demand driven by seasonal weather, the expansiveness of interconnecting regions in the United States will help support the continued need for peaking resources (such as FEI's proposed TLSE Project) well into the future.

¹⁶ The figure also shows the region's move away from coal and expectation that natural gas resources play a steady role in servicing peak in the region.

This need is especially acute during extreme weather conditions present in the Pacific Northwest. As a result, I expect the need for additional natural gas power generation will be required to serve seasonal demand.

ii. Social & Economic Pressures to Reduce GHG Emissions

Beyond the seasonal weather conditions discussed above, all regions (including the Pacific Northwest) are moving towards minimizing GHG emissions. The speed of implementation from region-to-region will vary based on political will, a consumer's mindset, and the economics in which the market participants can bear. Governments can and will provide incentives to create change while social desire/pressure will influence a consumer's behavior. Moving electrification in any one direction too quickly will create challenges, especially when it is imposed on a market that must absorb the costs in a prudent manner. In addition, the influence of one region on another must be monitored. For example, all levels of government (i.e., Federal, Provincial/State, City, and Municipal) could impose their own transitional processes to meet a specific agenda. The path may include tax options (i.e., incentive rebates, financial penalties, etc.), new market implementations (i.e., carbon trading), or parameters for corporate reporting (i.e., Environmental, Social, and Corporate Governance ("ESG") reporting). These initiatives, depending on how they are deployed and digested by the market, will influence consumer and industry energy consumption.

As shown by the examples below, consumer and industry hunger for energy has already created pressures on existing infrastructure, with many regions witnessing record demand.

Northwest Pipeline (May 2022):17

Northwest Pipeline finished 2021 with the highest annual throughput in the pipeline's history, reaching over 869 billion cubic feet, and is already setting records in 2022. "It was the highest peak electric generation demand year in the history of Northwest Pipeline," said Gary Venz, director of commercial services "The Pacific Northwest is relying on Northwest Pipeline more than ever as coal-fired plants continue to be retired."

Northwest Regional Forecast of Power Loads and Resources (May 2023):¹⁸

Several utilities are steadily phasing out coal-fired generation. By 2026, these utilities will have reduced their reliance on coal by over 4,800 megawatts since 2019. As utilities keep some

¹⁷ <u>https://www.williams.com/2022/05/27/northwest-pipeline-at-peak-performance-to-meet-natural-gas-</u>

demand/#:~:text=Northwest%20Pipeline%27s%20bi%2Ddirectional%20design,already%20setting%20records%20in %202022.

¹⁸ <u>https://www.pnucc.org/wp-content/uploads/2023-PNUCC-Northwest-Regional-Forecast-final.pdf</u>

options on the table to meet near-term increases in demand during a period with high resource development risk, plans are shifting. Compared to last year's Forecast, 222 megawatts of planned coal exits were pushed back from no later than the end of 2025 to be no later than the end of 2029. Also, natural gas resources are forecast to have an increased role for reliability until energy storage and other emerging technologies are proven. Plans are progressing to convert Jim Bridger coal units 1 and 2 to natural gas in 2024. These resources provide a bridge to meet peak demand and fill in during potential low water years until sufficient new capacity resource technologies and transmission can be added.

Western Assessment of Resource Adequacy (November 2023):19

"Based on balancing authorities 'current resource plans, the West is not prepared to meet future demand over the next 10 years. An increasingly variable resource stack and uncertain demand growth make it harder for the region to close the gap, according to the Western Electricity Coordinating Council's 2023 resource-adequacy assessment...".

While demand for energy continues to escalate with population growth, and society's desire to experience new products and services (i.e., electric vehicles, personal devices and their reliance on server storage, smart homes, etc.), increasingly frequent warmer than normal temperatures have prompted change within the cooling industry to ensure access to air-conditioning. Further, although many consumers are attempting to lower their GHG emissions, the speed in which they can bear the associated costs will differ and are also driven, at least in part, by wider governmental policies such as changes to building codes and Demand Side Management ("DSM") initiatives.

According to Statistics Canada, and as shown in Figure 12 below, FEI's interconnecting regions are experiencing significant population growth similar to British Columbia. Assuming population in these regions grows as trending shows,²⁰ a peak shaving LNG facility, that can be deployed during higher velocity demand, will maintain significant commercial value. Higher peak demand, during the winter or summer, coincides with higher daily prices which provides commercial/financial asset value.

¹⁹ Read the full paper at the following link:

https://www.wecc.org/Administrative/2023%20Western%20Assessment%20of%20Resource%20Adequacy.pdf ²⁰ Statistics Canada,

https://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1710000901&cubeTimeFrame.startMonth=01&cubeTimeFrame.estartYear=2016&cubeTimeFrame.endMonth=10&cubeTimeFrame.endYear=2023&referencePeriods=20160101%2 C20231001.





iii. Introduction of New Technologies

The energy industry is on the cusp of advancing significantly into new technologies that are intended to assist in mitigating GHG emissions growth. The advancement of these mitigation efforts through already significant investment in new technologies will, in turn, ensure that natural gas remains a valuable component of the energy market. In particular, commercializing the measurement and quantification of GHG emissions has gained momentum since 2018 with an increase of participants from investors, bankers, and financial intermediaries. Measurement and quantification of emissions is a necessary precursor to: (a) enabling broader cap and trade activities worldwide; (b) the large-scale commercial deployment of CCUS; and (c) continuing investments in Low Carbon Fuels. I address each in turn below.

a) Cap and Trade Activities: Carbon trading provides the platform to quantify and value GHG emissions. Underpinning the financing for a resource's expansion will be easier if the market can quantify the displacement of these emissions. Broader cap and trade activities worldwide, that allow more countries to participate in the carbon market, will ensure natural gas continues to be a foundational peaking resource in the energy market as governmental GHG reduction policies mature.

- b) Carbon Capture Utilization and Storage: Investments in CCUS technology has the potential to align "large scale carbon capture with large scale power generation" in order to mitigate GHG emissions from conventional natural gas as a fuel source for electrical generation. For example, in Alberta, 25 CCUS projects have been selected for further evaluation.²¹ Of these, seven new projects, with known capacity and commissioning dates plus expansions of already existing projects, have the potential to increase the provincial CCUS capacity to about 56 million tonnes of (carbon dioxide) CO2 per year by 2030. This is equivalent to 22% of the 256.5 million tonnes of CO2 emissions in Alberta in 2020. The remaining 18 projects would further increase future provincial carbon capture capacity if completed. CCUS projects, if commercialized, could support incremental pipeline expansions and their access to regions that have carbon capture technology.
- c) Low Carbon Fuels: The blending of low carbon fuels, such as Renewable Natural Gas ("RNG")²² and hydrogen into the existing natural gas system continues to be an effective and efficient means of mitigating GHG emissions by increasing the amount of low carbon energies in the market. As reported by the Canadian Energy Regulator,²³ the number of RNG projects operating in Canada are expected to more than double between 2021 and 2025, increasing capacity from 7.2 petajoules (PJ) in 2021 to 17.1 PJ in 2025. Further, the 2019 British Columbia Hydrogen Study²⁴ demonstrated that concentrations of hydrogen between 5 and 15 percent by volume, blended into the existing natural gas system, could be a viable opportunity of delivering renewable energy to markets without significantly increasing risks/cost to the energy value chain (i.e., production, midstream, transportation, distribution, and consumption). Aligning a peak shaving (on-system) resource, with the potential of

²³ Canadian Energy Regulator, https://www.cer-rec.gc.ca/en/data-analysis/energy-markets/market-

snapshots/2023/market-snapshot-two-decades-growth-renewable-natural-gas-canada.html

²¹ Canadian Energy Regulator, <u>https://www.cer-rec.gc.ca/en/data-analysis/energy-markets/market-</u>

snapshots/2022/market-snapshot-new-projects-alberta-could-add-significant-carbon-storage-capacity-2030.html. ²² RNG to include but not limited to landfill waste, water treatment waste, and livestock digester gas

²⁴ BC Hydrogen Study, <u>https://www2.gov.bc.ca/assets/gov/government/ministries-organizations/zen-bcbn-hydrogen-study-final-v6.pdf</u>.

increased renewable benefits, will continue to bring social and commercial market value.

iv. Large Industrial Development

Unlike population growth demand, which typically grows progressively, large industrial demand grows in large increments, taxing infrastructure that was designed to assume a moderating growth pattern. Large industrial infrastructure, such as LNG turbine compression and power; fuel refining;²⁵ and data centres,²⁶ are increasingly being developed in the Pacific Northwest region, driving increased demand for energy and, as a result, demand for infrastructure investments to address the already constrained energy market. For example, the viability of the Woodfibre LNG project was driven by the proponent's ability to secure long-term transportation capacity on the T-South system to source natural gas in northern British Columbia. When the Woodfibre LNG facility becomes operable, the resulting increase in demand will further tax the infrastructure in the region which, in turn, could lead to higher prices. As such, in my opinion, peak shaving (on-system) resources will play an increasingly important role to support balancing energy peaks in the Pacific Northwest region and beyond.

II. Market Absorption Analysis of the Huntingdon/Sumas Region

As discussed above, there are several drivers of market demand which inform my assessment of the future demand (post-2027) for natural gas resources in the Huntingdon/Sumas region. While the Huntingdon/Sumas market is made up of annual, monthly, and daily trading activities,²⁷ for the purposes of this analysis, I use daily market data to review the pricing characteristics of the Huntingdon/Sumas region as it relates to the monetization of uncommitted peaking capacity sold by FEI into the gas market. This review was conducted to determine the level of send-out capacity that the market could reasonably absorb.

Before addressing the Huntingdon/Sumas market specifically, I note that my initial analysis relies on Electronic Exchange Trading ("EET") data, which is a public form of transaction data between buyers and sellers for physical natural gas. Unlike transactions which are undertaken directly

²⁵ Refining Crude Oil, https://www.eia.gov/energyexplained/oil-and-petroleum-products/refining-crude-oil-the-refining-process.php

²⁶ Energy, <u>https://www.iea.org/energy-system/buildings/data-centres-and-data-transmission-networks</u>.

²⁷ Daily pricing variability represents real time scenarios the FortisBC gas supply department would experience during peak day demand.

between counterparties (i.e., bilateral transactions), EET data is executed through the use of an electronic exchange - meaning data (i.e., such as the number of trades, volume, and pricing) is recorded and accessible by the market. While records of bilateral transactions are not publicly available, they typically use information that the electronic exchange is producing (i.e., EET data) during trading hours to help bilateral counterparties generate their own trade parameters (i.e., price and associated volume). In my professional experience, and as confirmed by the experience of FEI's gas supply department, the ratio between EET and bilateral trading is approximately 50/50. As such, my analysis of EET data represents approximately 50 percent of possible market absorption. I address the incorporation of the bilateral market into my analysis after first summarizing the findings from EET dataset.

i. EET Dataset Demonstrate Ample Market Absorption

As discussed below, the EET market is able to easily absorb 150 MMcf/d across the time period surveyed, including during winter, before prices tend to moderate when volumes surpass 200 MMcf/d. In Figure 13 below, I have compiled daily EET data from 2016 to January 2024 to visualize the seasonality characteristics of the market.





The green data (right Y axis), represents the number of trades on a given day (Deal Count), the blue data (left Y axis) represents the total volumes (MMcf) traded on the day, the black dotted lines represent the 60-day rolling average across the time period visualized, and the transparent blue boxes identify the winter season (November-March) for each year presented. The **red** lines across the chart correlate the volume to the number of trades and show that:

- 1. an average of 25 trades were required to sell 150 MMcf/d in the EET market;
- 2. an average of 45 trades were required to sell 300 MMcf/d in the EET market;
- **3.** the EET market could generally absorb 150 MMcf/d year-round; and
- 4. during the winter periods shown, the EET market could easily absorb 300 MMcf/d.

Figure 14 below uses the same EET dataset, but re-arranges the number of trades from lowest to highest. This chart confirms that an increase in the number of trades corresponds to an increase in volumes sold into the market on a given day.





Finally, Figure 15 below shows the daily pricing over the same period. The red line (right Y axis) shows the volume data arranged from lowest to highest, with the day's corresponding daily (settled) price at Huntingdon/Sumas represented by the blue data (left Y axis).





The figures above show that the EET market can easily absorb volumes up to 150 MMcf/d, however, prices for natural gas at Huntingdon/Sumas tend to moderate when volumes surpass the 200 MMcf/d threshold (blue transparent circle). Even so, there are examples in the dataset where unforeseen events (e.g., weather and/or supply constraints) create market conditions that require volumes above 200 MMcf/d and prices do not moderate. These events are identified in the blue transparent box and the associated blue bars within Figure 15, and are described below:

As reported by the U.S. Energy Information Administration (EIA) on December 21, 2022,²⁸ daily natural gas spot prices at three major trading hubs in the western United States (Pacific Gas & Electric [PG&E] Citygate, Northwest Sumas on the Canada-Washington border, and Malin, Oregon) were higher than \$50.00 per million British thermal units (MMBtu). These hub prices were higher than in any other market and averaged \$48.12/MMBtu above the Henry Hub benchmark, which was \$6.14/MMBtu on December 21. PG&E Citygate in Northern California and Malin, Oregon, the northern delivery point into the PG&E service territory, reported the highest natural gas spot prices since

²⁸ EIA,

https://www.eia.gov/todayinenergy/detail.php?id=55279#:~:text=On%20December%2021%2C%202022%2C%20dail y.British%20thermal%20units%20(MMBtu).

December 2000—in both real and nominal terms—according to pricing data from Natural Gas Intelligence. The price at Southern California (SoCal) Citygate was highest on December 13 at \$49.67/MMBtu.

Several events occurring simultaneously in the West contributed to prices rising to these levels:

- 1. Widespread, below-normal temperatures
- 2. High natural gas consumption
- 3. Lower natural gas imports from Canada
- 4. Pipeline constraints, including maintenance in West Texas
- 5. Low natural gas storage levels in the Pacific region

Please note the price occurrence above started its advancement on December 8, 2021.

- 2. As reported by the EIA on January 6, 2022,²⁹ In mid-February, an intense winter storm in the central United States both increased energy consumption and disrupted energy supply. Daily dry natural gas production in Texas fell by almost half on Wednesday, February 17, according to estimates from IHS Markit, mostly because of well freeze-offs, which occur when water in the raw natural gas stream freezes. The Henry Hub spot price of natural gas increased to nearly \$24/MMBtu on February 17, the highest daily price (in real, inflation-adjusted terms) since February 2003. According to Natural Gas Intelligence data, many natural gas pricing hubs throughout the country also saw record-high prices around that time [including Sumas].
- *ii.* Incorporation of Bilateral Trades Doubles Market Absorption

As noted above, the EET market represents approximately 50 percent of the volume traded at the Huntingdon/Sumas market and, therefore, only half of what the volumes the market was able to absorb. In order to account for the remaining 50 percent of the market (i.e., the bilateral), I have assumed a 100 percent escalation of the EET volume data. This approach ensures that the natural gas volumes traded through the bilateral market data are represented in my analysis, reflecting that the characteristics of bilateral trades tend to mirror EET trades (i.e., volume, price, and number of trades). While volumes traded in the bilateral market are not represented in the three previous figures, based on the results of my analysis, the price point at which natural gas prices at Huntingdon/Sumas moderated would be similar to the EET market at 150-200 MMcf/d. Therefore, 200 MMcf/d could be traded <u>in their respective portion of the Huntingdon/Sumas market</u> (i.e., 200 MMcf/d EET + 200 MMcf/d bilateral = 400 MMcf/d total) before prices tend to moderate. As noted above, such moderation generally does not occur where unforeseen events: (1) drive an increase in demand; or (2) cause a supply constraint in the market.

²⁹ EIA, <u>https://www.eia.gov/todayinenergy/detail.php?id=50778</u>.

III. Conclusions to Question #3: Analysis Results Confirm Market Absorption Will Likely Increase in the Future as the Market Becomes Further Constrained

When both the Huntingdon/Sumas EET and bilateral markets are considered, I expect the daily market could absorb 300-400 MMcf/d across multiple days (e.g., 10 days) during the winter without influencing daily prices in a manner that could limit monetization values. Assuming no other resources are implemented, the minimum amount the Huntingdon/Sumas market can absorb will likely increase as the market becomes further constrained – thereby increasing the volume threshold at which prices tend to moderate under normal market conditions.

For example, as previously discussed in Question 2B, the future operations of the Woodfibre LNG facility, which is currently scheduled to enter service in 2027, will have a significant impact on the available supplies currently being consumed by the Huntingdon/Sumas market. In particular, as noted above, the 300 MMcf/d currently contracted for the Woodfibre LNG facility no longer being sold into the market when the facility enters service will result in a corresponding decrease in available supply at Huntingdon/Sumas. This large decrease in available supply, coupled with a continued increase in demand for natural gas resources in the Pacific Northwest due to, for example, increasing electrification across the region, would only increase what the market can reasonably absorb on a given day. Therefore, even assuming the proposed WEI Sunrise Expansion Program is successfully implemented and constructed, this project would only *replace* Woodfibre LNG supplies, leaving the conditions at the Huntingdon/Sumas market in a similar supply/demand and pricing position as today (i.e., 300-400 MMcf/d of market absorption potential). This continues to present a significant market opportunity for increased calls on daily natural gas peaking resource(s) such as the proposed TLSE Project.

Ultimately, without new resources being added, such as the proposed TLSE Project, the Huntingdon/Sumas market will remain constrained after 2027 (when the Sunshine Expansion Program is expected to enter service) and, therefore, will continue to be able to absorb 300-400 MMcf/d across multiple days (e.g., 10 days) during the winter without influencing daily prices in a manner that could limit monetization values.

Response to Question #4

Assuming FEI is able to monetize its hypothetical surplus from the TLSE Project as contemplated in the Question 3 scenario, what annual financial value could FEI recover for its customers in the following scenarios where, in one year, FEI can deliver and the market can absorb a maximum of

Scenario #1: 300 MMcf/d of send-out sold over 1.3 peak days (300 MMcf/d x 1.3 = 0.4 Bcf of LNG); Scenario #2: 300 MMcf/d of send-out sold over 2 peak days (300 MMcf/d x 2 = 0.6 Bcf of LNG); Scenario #3: 300 MMcf/d of send-out sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG); Scenario #4: 300 MMcf/d of send-out sold over 5 peak days (300 MMcf/d x 5 = 1.5 Bcf of LNG); Scenario #5: 300 MMcf/d of send-out sold over 7 peak days (300 MMcf/d x 7 = 2.1 Bcf of LNG); Scenario #6: 300 MMcf/d of send-out sold over 10 peak days (300 MMcf/d x 10 = 3.0 Bcf of LNG).

For each scenario, please calculate the financial value FEI could recover over a 5-year period, taking into account FEI's estimated variable operating costs for the Tilbury facility, which has been provided to you, and assuming FEI's commodity cost to produce LNG reflects gas prices in the summer when FEI typically fills its LNG tank. Please also add another five-year scenario:

Scenario #7: in year 1, 300 MMcf/d sold over 10 peak days (300 MMcf/d x 10 = 3.0 Bcf of LNG), and in each of years 2-5, 300 MMcf/d is sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG);

Please note that the financial variables used to generate the financial value results below are based on the 5-year forward market prices (2024-2029)³⁰ at Huntingdon/Sumas and Spectra-Stn2. As previously discussed in Question 3, the ongoing supply/demand imbalance (lack of energy infrastructure to meet existing and growing customer demand), is likely to persists throughout the proposed TLSE Project's projected term of operations. While I realize that the proposed TLSE Project is not yet in service, in my opinion, using current forward market prices is appropriate for the purposes of the requested analysis given volatility in the market is expected to increase, or at minimum persist, beyond a 5-year time horizon.

To replicate the filling of the TLSE Project tank during the summer (April-October), my analysis uses summer forward prices for Spectra-Stn2, which I have adjusted to reflect their respective tolls/motor fuel for delivery to Huntingdon/Sumas. To replicate regasification during the winter

³⁰ Based on forward markets dated February 29, 2024.

(December-February), my analysis uses winter forward prices at Huntingdon/Sumas to which I then applied a "daily escalator" to replicate future daily volatility that would be experienced by FEI's gas supply department. Finally, I have generated scenarios for a standing demand charge, monetary value incremental to market price spreads (i.e., Spectra-Stn2 to Huntingdon/Sumas prices), that could be accessed by FEI's gas supply department. For clarity, a standing demand charge is a fee, collected by FEI from a third-party, who, in these scenarios, has contractually committed to LNG capacity from the TLSE Project. Please refer to Appendix C, Supplemental Energy Market Information, for a detailed breakdown of the calculations underpinning the financial value results for each scenario.

While the results of my analysis demonstrate that volatility in the Huntingdon/Sumas market can generate significant monetary value, accessing this value is predicated on access to a <u>dependable</u> peaking resource, which currently does not exist in the market. Put simply, if the TLSE Project were not constructed, FEI cannot currently rely on purchasing similar volumes from the market.

The results of my financial analysis are presented in Figure 16 below. The figure identifies a Spectra-Stn2 delivery point,³¹ the range of financial values (i.e., Min to Max) for Scenarios #1 through #7, and the respective 5-year cumulative values assuming 300 MMcf/d of send-out capacity. The results of my analysis show that, over a 5-year period, FEI could recover between \$73.0 MM CDN to \$78.8 MM CDN of financial value for its customers over 10 peak days in the winter. Furthermore, assuming Scenario #7, a combination of 10-days of send-out capacity in year one and 3 peak days of send-out capacity in years two through five, FEI could recover between \$36.9 MM CDN to \$39.4 MM CDN of financial value for its customers. I provide a summary of the 5-year cumulative results for Scenarios 1# through #7 in Figure 16 below.

³¹ ICE NGX - <u>https://www.ice.com/products/73447570/ICE-NGX-Spectra-Station-2-Fixed-Price-Daily.</u>





The financial results presented in Figure 16 above do not include additional incremental value that FEI could capture by implementing a standing demand charge. Please refer to Appendix C, Generating A Standing Demand Charge, for a detailed analysis of its structure. I estimate that FEI could generate the following range of additional incremental value through an annual standing demand charge:

• \$5.2 MM CDN - \$7.0 MM CDN for every 50 MMcf/d

Assuming Scenario #6 (i.e., 10 days of send-out, cumulative over 5-years) and Scenario #7 (i.e., 10 days of send-out year 1 and 3 days of send-out over years 2-5) and that FEI's gas supply department is successful in contracting a standing demand charge with a third-party for 50 MMcf/d

of send-out capacity from the TLSE Project, ³² FEI could increase the average cumulative financial value as follows:

Scenario #6: From \$75.9 MM CDN³³ to \$106.4 MM CDN;³⁴ and Scenario #7: From \$38.2 MM CDN³⁵ to \$68.7 MM CDN³⁶

³² Calculation of average standing demand charge over five years: ((\$5.2 MM + \$7.0 MM)/2) x 5 years = \$30.5 MM

³³ Calculation of 10-day cumulative average financial value for Scenario #6: (\$73.0 MM + \$78.8 MM)/2 = \$75.9 MM ³⁴ Calculation of average standing demand charge over five years and 10-day cumulative average financial value for

Scenario #6: \$30.5 MM + \$75.9 MM = \$106.4 MM

³⁵ Calculation of 10-day year 1 and 3-day years 2-5 financial value for Scenario #7: (\$36.9 MM + \$39.4 MM)/2 =

^{\$38.2} MM ³⁶ Calculation of average standing demand charge over 5 years and 10-day year 1 and 3-day years2-5 financial value for Scenario #7: \$30.5 MM + \$38.2 MM = \$68.7 MM



FASKEN

Fasken Martineau DuMoulin LLP Barristers and Solicitors Patent and Trade-mark Agents 550 Burrard Street, Suite 2900 Vancouver, British Columbia V6C 0A3 Canada T +1 604 631 3131 +1 866 635 3131 F +1 604 631 3232

fasken.com

Matthew T. Ghikas Direct +1 604 631 3191 Facsimile +1 604 632 3191 mghikas@fasken.com

April 15, 2024 File No.: 240148.00966

Via Email Privileged and Confidential

1459488 B.C. Ltd 1201 Fort Street, Suite 506 Victoria, British Columbia V8V 0G1

Attention: Raymond Mason

Dear Sirs/Mesdames

Re: FortisBC Energy Inc. ("FEI") – Tilbury Liquefied Natural Gas Storage Expansion ("TLSE") Project CPCN Application (the "Regulatory Proceeding")

We, as counsel for FEI in the above referenced Regulatory Proceeding, request that you prepare an independent expert report to be introduced into evidence in that Regulatory Proceeding. This letter outlines the matters to be addressed and provides some general guidance as to the format of your report.

Apart from our instructions below as to the matters to be addressed and the format of your report, the contents of your report are entirely for you in the exercise of your independent professional judgment. You have been retained to provide independent expert evidence for the above referenced Regulatory Proceeding, not as an advocate for our client. The integrity of your conclusions is dependent upon your objectivity.

Matters on Which Your Opinion is Requested

We request that your report set out your independent objective opinion with respect to the following:

- 1. What are the elements of an optimal resource portfolio for FEI and its customers?
- 2. FEI has, since 1971, included on-system liquified natural gas ("LNG") at the Tilbury facility in its supply portfolio as a peaking capacity resource (i.e., rapid send-out capability, backed by sufficient LNG to meet short-duration peak load). FEI's current Annual Contracting Plan includes 150 MMcf/d of regasification and 0.6 Bcf of LNG from the

FASKEN

Tilbury facility. We seek your opinion on the peaking capacity alternatives available to FEI in the event that FEI was no longer able to include some or all of these peaking resources in its supply portfolio. In particular:

- Based on your assessment of the natural gas markets accessible to FEI, would FEI be able to contract for peaking resources (capacity/energy supply) as a potential alternative to on-system LNG in its gas supply portfolio, and if so on what terms?
- If appropriate peaking resources (capacity/energy supply) are unavailable to FEI in the market, what other potential investments could FEI explore as a potential alternative to on-system LNG as peaking resources (capacity/energy supply) in its gas supply portfolio? For clarity, in this regard we are requesting that you identify potential peaking supply (capacity/energy supply) alternatives based on your understanding of the regional system, as opposed to a undertaking a financial analysis of those options.
- 3. Assuming a hypothetical scenario where the TLSE Project is constructed with 3 Bcf of storage volume and 800 MMcf/d of regasification capacity, and FEI no longer requires the full storage and regasification capacity to meet its own gas supply and resiliency requirements, do you expect that FEI would be able to monetize the surplus by selling peaking resources in the gas market? If so, how much daily send-out capacity would the market reasonably absorb?
- 4. Assuming FEI is able to monetize its hypothetical surplus from the TLSE Project as contemplated in the Question 3 scenario, what annual financial value could FEI recover for its customers in the following scenarios where, in one year, FEI can deliver and the market can absorb a maximum of
 - Scenario #1: 300 MMcf/d of send-out sold over 1.3 peak days (300 MMcf/d x 1.3 = 0.4 Bcf of LNG);
 - Scenario #2: 300 MMcf/d of send-out sold over 2 peak days (300 MMcf/d x 2 = 0.6 Bcf of LNG);
 - Scenario #3: 300 MMcf/d of send-out sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG);
 - Scenario #4: 300 MMcf/d of send-out sold over 5 peak days (300 MMcf/d x 5 = 1.5 Bcf of LNG);
 - Scenario #5: 300 MMcf/d of send-out sold over 7 peak days (300 MMcf/d x 7 = 2.1 Bcf of LNG); and
 - Scenario #6: 300 MMcf/d of send-out sold over 10 peak days (300 MMcf/d x 10 = 3.0 Bcf of LNG).

FASKEN

For each scenario, please calculate the financial value FEI could recover over a 5-year period, taking into account FEI's estimated variable operating costs for the Tilbury facility, which has been provided to you, and assuming FEI's commodity cost to produce LNG reflects gas prices in the summer when FEI typically fills its LNG tank. Please also add another five-year scenario:

Scenario #7: in year 1, 300 MMcf/d sold over 10 peak days (300 MMcf/d x 10 = 3.0 Bcf of LNG), and in each of years 2-5, 300 MMcf/d is sold over 3 peak days (300 MMcf/d x 3 = 0.9 Bcf of LNG);

In order to facilitate your analysis and the preparation of your report, FEI will make available information that your request. You can assume, for the purposes of your analysis, that any information provided by FEI is accurate.

Overview of the Structure of Your Report

We request that your independent expert report be set out generally consistent with the following structure.

A. Introduction and Summary of Opinion

Your introduction should

- reference the nature of your engagement as an independent expert as per this letter,
- identify the questions posed to you, and
- set forth, in a summary fashion, your independent objective opinions on each matter upon which your opinion is requested, as set out above in this letter.

Please state, in a summary fashion, the professional qualifications, technical education, training and experience of those individuals who are responsible for the content. Explain how the authors' expertise relates to the subject matter of your opinions. Detailed *curricula vitae* should be attached as an appendix.

We confirm that you have a duty to assist the regulator and are not to be an advocate for any party ("Duty of Independence"). In this section of your report, please certify the following:

- You are aware of your Duty of Independence,
- You have prepared your report in accordance with the Duty of Independence, and
- If called to give oral or written testimony, you will give that testimony in conformity with the Duty of Independence.

B. Supporting Review and Evaluation

FASKEN

Under this heading, you should set out in full your independent objective opinions on each of the matters in the same order that the matters are presented above. You should provide the reasons for your opinions including reference to pertinent facts or assumptions and any research you conducted that led you to form the opinion.

C. Conclusion

You may provide a conclusion if you wish.

Appendices

Please include this letter, and your *curricula vitae* as appendices to your report. If additional instructions are required, then supplementary letters of instruction from us should also be attached to your report. You may attach other documents or schedules that elaborate on, or are integral to your analysis.

In conclusion, if you have any questions with respect to the nature and scope of your engagement, please contact the writer at your soonest convenience.

Yours truly,

FASKEN MARTINEAU DUMOULIN LLP

Matthew T. Ghikas Personal Law Corporation

MTG/NR

cc Sarah Walsh Director, Regulatory Affairs FortisBC Energy Inc.

APPENDIX B Work Experience

Raymond Mason

M4 - Energy & Capital Markets Consulting

Victoria, British Columbia

Professional Summary:

I am a commodity & derivatives professional with over 28 years of experience focusing primarily on natural resources and capital markets. I have provided market risk, sales & origination knowledge to the entire value chain within the Oil & Gas, Renewable Natural Gas, Electricity, Agriculture, and Metals industries. Expertise with price risk management portfolios, underwriting derivative structures, commodity linked financing, regulatory oversight, and business development. My career has engaged with Bay Street and Wall Street capital markets, the Federal Energy Regulatory Commission, US Securities Exchange Commission, Municipal Bond markets, Western Canadian Utilities Commissions, and the BC Vegetable Marketing Commission.

I have maintained leadership, guidance, and sensibility roles in the ever-changing derivatives and commodities landscape. This has continued to demonstrate an ability to advise and partner with a wide range of clients (C-suite public and private) across global businesses while ensuring risk weighted profits are maintained. My experience provides key strategic understanding of the development of commodity franchises in new and evolving market dynamics. This is instrumental in assisting internal partners with market risk, regulatory, tax, legal, and front and back-office functions for franchise trading activities.

Skills / Experience:

- Business Development
- Sales & Marketing
- · Derivatives, Asset Backed Securities, and Commodity linked financing
- Financial and Physical Commodity Price Risk Management
- · Operations and Maintenance of Pipelines, Distribution, Storage, and Industrial
- · Trucking, Vessel, and Rail Logistics & Scheduling
- Market Risk Evaluation & Monitoring
- Font/Back Office Functions & Reporting (Billing, Accounting, Balance sheet analysis)
- · Capital Cost Analyses and Implementation

Professional Background:

Financial Institution – Capital Markets & Commodity Trading

For the past 16 years, I have been active with the implementation, development, and success of building an energy trading franchise for The Toronto Dominion Bank ("TD"). Was active under the TD franchises: TD Securities, TD Securities USA LLC., and TD Energy Trading Inc. These franchises encompassed my activities in the financial and physical commodities business, capital markets, municipal bonds, interest rates and currency exchange platforms.

The initial launch of TD's physical energy franchise in 2003, I was responsible for establishing client relations and successfully negotiating ISDA / GasEDI / NAESB / Energy Management Service contracts to service financial and physical derivative risk management and operations solutions for new clients. The initiative was a collaborative effort with industry, and peers within the Firm, to align the "go to market" products while ensuring a proper business foundation was being built. Throughout the early years of the franchise development, it was critical to ensure business operations were sound with the appropriate risk controls, deal entry, operations, and accounting functions. Industry communications via conferences and speaking engagements helped me articulate how TD was building its business within a capital intense industry. TD Energy soon became the leading marketing and trading franchise in Canada and the Western United States.

I continued to develop new initiatives within the Firm that consisted of balance sheet reconciliation, capital risk modelling, underwriting derivative portfolios, commodity-based financing, and the US expansion of physical trading.

To increase franchise efficiencies and profitability, I performed internal business balance sheet audits. This ensured new regulations were being properly measured and implemented. In addition, the audits helped the development of a new directional risk counter-party exposure model to lower Basel III CVA capital which improved business profitability.

I managed the derivative portfolios of the Firm's US private equity energy producer franchise. The franchise initiatives included the underwriting of derivative portfolios which required knowledge of capital exposures and price risks. This process was undertaken while coordinating with Investment

Banking initiatives during client mergers and acquisitions efforts and peer banks during underwriting processes.

Commodity-based financing, also known as Municipal Pre-Pays, was a newly formed US franchise that provided a new source of funding for TD. I managed all energy related aspects of the franchise which included 30 year financial and physical commodity transactions. Deal size funding typically ranged from \$750 MM to \$1,000 MM per transaction. Collaborated with investment banking, municipal bonds, asset back securities, corporate lending, and equities teams to create a detailed business plan and policy parameters while tracking performance on a regular basis. The creation of the franchise required significant educational process for internal partners in risk, legal, regulatory compliance, finance, and trading. This franchise furthered TD's clients reach into the entire energy value chain which included producers, municipal power & gas utilities, refineries, petrol-chemical, pulp & paper, transportation/aviation, auto manufacturing, peer financial institutions, and energy marketing companies.

Developed the initial business plan for an expanded physical natural gas franchise into the US; while providing guidance to outside legal and tax council for the regulatory opinions regarding the Firm's operations under Canadian and US Bank regulatory frameworks. This continued to the coordination with internal partners for the approval of risk, compliance, legal, Federal and State tax, and staffing policies for the new initiative. Supervised and maintained business development liaison with internal partners for the ongoing implementation of the Firm's United States physical natural gas initiative.

Physical Commodities – Distribution, Midstream, Transmission and Trading

Throughout the early part of the 1990's, I was active in developing and maintaining financial regulatory modelling for Westcoast Energy's natural gas assets on Vancouver Island, Sunshine Coast, and Whistler. The assets included the high-pressure natural gas transmission line, utility distribution assets, and piped propane networks in Whistler and Port Alice. Provided support for applications to the British Columbia Utilities Commission that included Integrated Resource Plans, Certificate of Public Convenience and Necessity, and Cost of Gas Passthrough applications.

The latter part of 1990's, I assumed the responsibilities of gas operations for Centra Gas Inc. As Gas Operations Coordinator, I was responsible for the movement of natural gas throughout the British Columbia resource portfolio (storage and midstream) and the interconnections with FEI and

Northwest Pipeline. The role required an understanding of natural gas pricing, supply and demand fundamentals, core customer demand needs, hydraulic compression & meter station measurement operations (Scada monitoring), line pack utilization, and any operation & maintenance with peer pipelines, power generation, and large industrial industries. The propane systems required knowledge of liquids pricing, and rail scheduling/pricing. In addition, I provided negotiation support for gas supply portfolio resources for Centra Gas's core customers. This included long and short-term transportation, peaking gas supply, storage, and seasonal supply contracts.

In early 2000, I expanded my experience to include physical commodities trading in electricity. This new role included managing a small regional trading franchise in Vancouver, British Columbia. The regional coverage of clients included British Columbia, Washington, Oregon, Idaho, and California, with the client portfolio consisting of pipelines, midstream, storage, utilities, power generation, and industrials.

Agriculture – Deregulation, Franchise Development & Commodity Trading

As head of sales & logistics, I ran the day-to-day operations of a continental greenhouse sales and logistics business, as well as the regional management of Canada. The launch of this franchise was made possible with the deregulation of the hothouse industry in British Columbia. Introduced marketing contracts that included take or pay arrangements, peak supply off-take, and backstopped production arrangements not typically seen in the produce industry.

The new franchise responsibilities included the implementation of deal entry, risk management systems and employee oversight. Operations included the sale of proprietary and third-party marketed produce and the logistics of trucking/rail throughout Canada, United States, and Mexico. Established new relations with wholesale, brokerage houses and retailers for the off take of produce, as well as corporate sales initiatives with associated marketing entities such as Dole Food Group, Safeway, Costco, Kroger, Winn Dixie, etc.

APPENDIX C Supplemental Energy Market Information

I have reviewed the following supplemental energy market information to determine market fundamentals derived by historical and futures pricing trends. I have then used the fundamental trends to generate the financial variables necessary to calculate financial values requested in Question 4.

Historical Pricing Analysis

In the discussion below, I provide a pricing analysis to support the estimation of financial value(s) for the TLSE Project. Similar to the high/low range analysis from Question 2, III "Third-party Peaking Gas Supply", I want to investigate the value opportunities of sourcing gas supply from Northern British Columbia (Spectra-Stn2), delivering it to Huntingdon/Sumas, and then selling it into the Sumas market. This process will replicate the financial value that could have been hypothetically generated by the TLSE Project had it been in service as early as 2020 (i.e., cost of gas supply and transportation prior to liquefaction and regasification). The analysis shows there would have been opportunities to generate significant financial benefit from selling gas supplies, from the TLSE facility, into the market.

Figure 1 below identifies the incremental transportation and fuel gas costs to source gas supplies from Northern British Columbia. Spectra-Stn2 pricing incurs the T-South toll and associated fuel gas (i.e., average of \$0.64 CDN/GJ). Adding the toll and fuel costs to the Spectra-Stn2 prices generates the total embedded costs to deliver gas supply to the Huntingdon/Sumas region prior to its resale.

Figure 1



Figure 2 below, uses the above methodology, as well as, incorporating daily Sumas prices (i.e., March 2020 to August 2023). The Sumas daily index is used to replicate a price that would have
been captured when selling the previously delivered gas supply from Northern British Columbia. For example, the Sumas daily prices were subtracted from the embedded delivered Spectra-Stn2 (light blue line) daily prices (i.e., which includes respective toll and fuel costs). The red line assumes the "break-even" of the pricing location in relation to the Sumas daily index. Any data point **below** the red line, indicates that Spectra-Stn2 pricing is <u>more expensive</u> than buying the Sumas index. Any data points **above** the red line indicates that pricing from Spectra-Stn2 is a <u>less expensive</u> delivered priced supply than buying the Sumas index. The chart indicates there are significant periods of time that sourcing gas supply from Northern British Columbia, delivering it to the Huntingdon/Sumas region, and selling into the Sumas market, generates financial value.



Figure 2

Using the same historical pricing data as above (March 2020 to August 2023), the following pie chart (Figure 3) was created to show the percentage break down of the delivered costs for Spectra-Stn2, and resulting profits that could have been generated by selling gas supplies at Sumas (e.g., Sumas price minus Spectra-Stn2 price plus T-South toll and fuel). The pie chart

data supports the pricing advantages of sourcing gas supply from either Spectra-Stn2 and delivering it to Huntingdon/Sumas (Toll and Fuel) for resale (Profit/(Loss)). For example, the Spectra-Stn2 pie chart represents a percentage breakdown of costs as it relates to gross financial value generate when selling gas supply at Sumas. Therefore, assuming 100% of the gross value of a Sumas sale, roughly 60% of the value is offset by Spectra-Stn2 commodity cost, transportation toll, and fuel gas.





To further refine my financial value analysis of sourcing gas supply from Northern British Columbia, I reviewed the seasonal benefits of sourcing gas supplies in the summer, storing it, and then selling it in the winter. For example, Figure 4 below "Historic Daily Delivered Prices" shows summer (April-October) Spectra-Stn2 (blue line) prices and subsequent winter (November-March) Sumas prices (grey line). As in previous figures, Spectra-Stn2 daily prices have been adjusted to reflect their respective transportation toll and fuel gas costs (Index price + toll + fuel gas).

Figure 4	ļ
----------	---



The result of this financial analysis is located in the embedded table in Figure 4 above. Summer Spectra-Stn2 numbers represent the average daily prices over the months of April to October for their respective terms (e.g., summer Spectra-Stn2, April 2020 to October 2020 = 2.69 CDN/GJ). The Winter Sumas numbers represent the average daily prices over the months of November to March for their respective terms (e.g., winter Sumas, November 2020 to March 2021 = 4.06 CDN/GJ). Finally, the resulting Sumas Spread for Spectra-Stn2 is generated by subtracting the summer average prices from the winter Sumas prices to generate the Sumas Spread (e.g., winter Sumas minus summer Spectra-Stn2 = Sumas Spread ~ 4.06 - 2.69 = 1.37 CDN/GJ). Therefore, Sumas Spread prices have been seasonally adjusted to generate "profit" from sourcing Spectra-Stn2 summer gas, transporting it to Huntingdon/Sumas, and then selling it in the winter market at Sumas prices (i.e., cost of gas supply and transportation prior to liquefaction and regasification costs).

Please note that this exercise is only identifying the pricing differentials between the points discussed and not representative of what someone, or an entity, could extract from the market in

its entirety. That would assume a person had executed every trade in the exact right moment, an action that is not plausible. Therefore, one must assume future financial values would be a subset of the spreads if FEI, as the owner of the TLSE facility, was acting commercially.

Futures Pricing Analysis

The following futures pricing analysis is similar to the above analysis, however, I have used the 5-year futures market monthly index prices (March 2024 to February 2029) for Spectra-Stn2 and Sumas. In addition, any associated tolls and fuel gas costs, to replicate delivery cost to move supply from Northern British Columbia, were based on historical averages. This analysis shows there are opportunities to generate significant financial benefit selling gas supplies, from the TLSE facility, into the market.

Using the futures pricing data, while applying the same methodology as before, the following pie chart was created to show the percentage break down of the delivered costs for Spectra-Stn2, and resulting profits that could be generated by selling gas supplies at Sumas. The futures market data continues to support pricing advantages of sourcing gas supply from Spectra-Stn2 and delivering it to Huntingdon/Sumas (Toll and Fuel) for resale (Profit/(Loss)). For example, the Spectra-Stn2 pie chart (Figure 5) below represents a percentage breakdown of costs as it relates to gross financial value generate when selling gas supply at Sumas. Therefore, assuming 100% of the gross value of a future Sumas sale, roughly 60% of the value is offset by Spectra-Stn2 commodity cost, transportation toll, and fuel gas.

Figure 5



Using the same futures data as above (March 2024 – February 2029) and adjusting Spectra-Stn2 to reflect the respective delivery cost to Huntingdon/Sumas, the following Figure 6 was developed. As identified in the chart, Sumas prices (red line) rise significantly higher than Spectra-Stn2 (light blue line) prices in the winter. The financial values presented in the embedded table, continues to show that over the next five-years, Sumas Spreads remain profitable. For example, the Sumas spread for Spectra-Stn2, ranges from \$7.38 CDN/GJ to \$5.29 CDN/GJ.



Figure 6

The above data analysis confirms that the Huntingdon/Sumas region is predicting that resource will remain constrained over the next 5-years and beyond.

Financial Variables Analysis

Using the data results from the historical and futures pricing analysis above, I generated the financial *variables* that would be used to forecast the financial *values*, in the hypothetical scenario where the TLSE Project is constructed and where FEI no longer requires the full regasification capacity and stored LNG for its own resiliency. As part of this stage of my analysis, I apply the financial variables to the futures market data in order to align the hypothetical daily operations of the TLSE facility.

My analysis of the financial variables is broken into three parts:

- I. Refining the forward monthly prices;
- II. Developing a daily price escalator to replicate daily trading activities; and
- III. Generating a standing demand charge.

Note: though historical natural gas pricing data has daily and monthly settles, the forward/futures pricing market only generates monthly prices. Therefore, I developed a methodology to replicate future daily prices (as discussed further below).

I. Refining Forward Monthly Prices

The natural gas industry breaks the calendar year into winter and summer time periods. The summer consists of the months of April to October and the winter November to March (i.e., carries over two calendar years; for example, November 2023 to March 2024). In the Huntingdon/Sumas pricing region there can be significant monthly variations in customer demand throughout the winter period. Specifically, the months of December, January, and February are typically the months that experience higher demand influenced by weather and/or unforeseen infrastructure outages. I have shortened the winter Sumas sales period to only include forward prices for the months December, January, and February, assuming that regasification from the TLSE Project would take place during periods of peak demand at Huntingdon/Sumas. I believe this is appropriate because it closely aligns peak demand with when gas supplies have historically been sold into the market.

II. Daily Price Escalator Development

Based on my experience in the Huntingdon/Sumas market, FEI's gas supply department would be active in the daily market when mitigating excess gas supply. As previously discussed in Question 2, III Third-party Peaking Gas Supply, there is an incremental daily range value that could be accessed by FEI during LNG peaking operations (i.e., Daily Escalator). Therefore, by utilizing the monthly forward prices, for December to February, and incorporating a daily range estimate, I was able to formalize the financial estimates requested in Question 4.

The development of the daily price escalator was based on historical Huntingdon/Sumas daily price ranges from 2017 to 2023. This data is presented in the chart (Sumas Daily High/Low

Ranges) on page 28-29 of the report. In addition to the historical Sumas daily price range data, I have incorporated peaking supply pricing mechanisms that I became familiar with as part of my role as gas supply marketer. I have created the following Daily Escalator Scenarios based the analysis of this historical data and the application of pricing mechanisms used in the market.

Daily Escalator Scenario A

Daily Escalator Scenario A was derived from historically recorded "high trades"¹ at Huntingdon/Sumas for the months December to February over the periods of 2017-2023. These high trades were subtracted from their respective daily settled price to generate the range value. These range values were then averaged generating the mean variable of \$0.47 CDN/GJ.

For example: Assuming the monthly forward index for the Month of December was \$7.00 CDN/GJ (i.e., \$7.00 x 31 days), the Daily Escalator Scenario A of \$0.47 CDN /GJ would be added to the \$7.00 (i.e., \$7.00 + \$0.47 = \$7.47) which represents the incremental daily variable estimate that FEI gas supply department could extract in the future daily market. For the purpose of applying this daily escalator, I assumed that FEI's gas supply department had constructed a peaking gas arrangement with a third-party who has agreed to pay the Huntingdon/Sumas high price, as recorded by the market, on the day gas supply was required. Therefore, in this scenario, the Daily Escalator Scenario A would be applied to the forward monthly prices for the winter terms of December, January, and February.

Daily Escalator Scenario B

For the Daily Escalator Scenario B, I conducted a descriptive statistics analysis² using the same term (2017-2023), Huntingdon/Sumas daily recorded high price, and the respective Sumas daily settled price as Daily Escalator Scenario A above. The descriptive statistics analysis of the historic data generates a lower range (mean – confidence level) and a

¹ "High trades" refers to those recorded by the EET market that set the highest point in the range on any given day prior to end-of-day settlement.

² Descriptive Statistics as described by Microsoft Excel Analysis Tools. The Descriptive Statistics analysis tool generates a report of univariate statistics for data in the input range, providing information about the central tendency and variability of the data.

higher range (mean + confidence level) of prices. The resulting high prices are subtracted from the settled prices to generate the Daily Escalator Scenario B.

For example: In the following Figure 7, the high recorded prices in column 1 and the settles prices in column 2, represent the average (mean), lower, and high ranges generated by the descriptive statistics analysis. The Daily Escalator Scenario B (i) and B (ii) (i.e., Escalator Column 1-2) provides the lower (\$0.3778 CND/GJ) and higher (\$0.5697 CND/GJ) range for this scenario.

Escalator A & B Sumas (Cad/GJ)	High Recorded Prices (Column 1)	Settled Prices (Column 2)	E (Co	scalator lumn 1-2)
Mean	\$8.5941	\$8.1203	Α	\$0.4738
Confidence Level @ 95.0%	0.972856166	0.87689191		
Lower (Mean - CL)	\$7.6212	\$7.2434	B (i)	\$0.3778
Higher (Mean + CL)	\$9.5670	\$8.9972	B (ii)	\$0.5697

Figure 7

Please note that the mean for the Daily Escalator Scenario A remains the same in this analysis.

Daily Escalator Scenario C

The Daily Escalator Scenario C was generated multiplying the forward winter (December, January, and February) Huntingdon/Sumas monthly price by 102% and deducting forward summer (April to October) monthly prices for Spectra-Stn2 (i.e., adjusted to reflect their respective toll and fuel costs). This scenario assumes that the buyer has agreed to pay 102% of the Huntingdon/Sumas daily settled price, prior to the settlement actually transpiring. This arrangement is similar to products that I have structured during my tenure as an energy marketer. The % charged by the seller fluctuates based on what the market will bear. For the purposes of this analysis, I have assumed that the FEI gas supply department had replicated this structure and has used 102%.

Resulting Financial Variables Analysis

Considering FEI's estimate of the TLSE LNG Variable Operating Costs (i.e., as referenced in the table below, Financial Variables), which have been provided to me, and assuming FEI's

commodity cost to produce LNG reflects gas prices in the summer when FEI typically fills its LNG tank, the following Financial Variables table (Figure 8) has been generated. Furthermore, the creation of the daily escalators is an attempt to replicate actual daily trading activities FEI's gas supply department could execute, assuming similar arrangements were constructed in agreements, for marketed peaking gas supply that would be sold into the market. These variables have been used while estimating the financial value scenarios requested in Question 4.

Figure 8

Financial Variables	Avera	ge ('24/'25)	A	verage ('25/'26)	1	Average ('26/'27)	A	verage ('27/'28)	1	Average ('28/'29)
Cad/GJ										
Forward Monthly Prices										
*Summer Spectra-Stn2 (Apr-Oct)	\$	1.97	\$	3.26	\$	3.72	\$	3.62	\$	3.79
Winter Sumas (Dec-Jan-Feb)	\$	11.41	\$	10.75	\$	10.27	\$	11.08	\$	9.51
* adjusted to reflect delivery costs to S	Sumas/Hunt	ingdon (i.e., T	oll 8	& Fuel)						
TLSE LNG Variable Operating Cost	\$	(3.00)	\$	(3.00)	\$	(3.00)	\$	(3.00)	\$	(3.00)
Daily Escalator A	\$	0.47	\$	0.47	\$	0.47	\$	0.47	\$	0.47
Daily Escalator B (i)	\$	0.38	\$	0.38	\$	0.38	\$	0.38	\$	0.38
(ii)	\$	0.57	\$	0.57	\$	0.57	\$	0.57	\$	0.57
Daily Escalator C		102%		102%		102%		102%		102%

Example: assuming the forward average prices for the term 2023/2024, Winter Sumas plus Daily Escalator A, less the Summer Spectra-Stn2, less TLSE LNG Variable Cost*, multiplied by send-out volume and # of days, the financial value generated would be (((\$11.41 + \$0.47 - \$1.97 - \$3.00) x 321,238 GJs*) x 2 Days) = **\$4.44 MM CDN**. {*300 MMcf/d = 321,238 GJs}

III. Generating a Standing Demand Charge

As previously discussed in Question 2 III above, I have assumed FEI's gas supply department would replicate peaking gas supply arrangements offered by companies in the market at Huntingdon/Sumas. Therefore, the service offered by FEI should attempt to recover the underlying costs of maintaining their asset portfolio while earning a fair return for its use during the period it is contracted. FEI's gas supply department could charge a "stand-by" fee (i.e., a monthly demand charge) if the capacity at the proposed TLSE facility is reserved by a third-party for a pre-determined period.

The demand charge could incorporate underlying carry costs associated with pipeline(s) tolls, storage tariff(s), and any other proprietary services that are required to support a peaking supply arrangement. Although a peaking supply arrangement may not be used (i.e., contracted) during summer months (April-October), there is still an underlying cost to the utility associated with having the asset year-round. As such, cost recovery from the period where an asset is underutilized during the summer could be incorporated into winter periods when an arrangement

with a customer for peaking gas supply is active (November-March). The underutilized period costs will be market-driven, meaning they will fluctuate in price, driven by the success of cost mitigation attempts during the summer. Higher success of mitigating costs during the summer may result in lower demand charges for winter peaking arrangements.

For the purposes of generating a standing demand charge, I have assumed the fee charged would recovery T-South tolls and associated fuel. Therefore, for estimating purposes, I used \$0.64 CDN/GJ (i.e., WEI T-South historical average).

Annual Standing Demand Charge Scenarios

Scenario #I: Recouping T-South tolls and fuel for 100% of the winter period (i.e., (\$0.64 CDN/GJ x contracted daily volume x 151 days [November – March])); and

Scenario #II: Recouping T-South tolls and fuel for 100% of the winter period and 25% for the summer period (i.e., (\$0.64 CDN/GJ x contracted daily volume x 151 days) + (\$0.64 CDN/GJ x contracted daily volume x 214 days [April - October]) x 25%)).

The results of the above scenarios generate incremental annual financial value to the commodity pricing spreads derived by FEI's gas supply department, buying summer gas from Spectra-Stn2 and selling winter gas at Huntingdon/Sumas. FEI's gas supply department could charge a "stand-by" fee (i.e., a monthly demand charge), assuming the capacity, at the proposed TLSE facility, is reserved by a third-party for a pre-determined period. For example, a third-party might contract winter (i.e., November-March) peaking capacity, with the right to call up to 10 days of 50 MMcf/d capacity over the term. Therefore, the third-party pays FEI the standing demand charge fee and the associated Huntingdon/Sumas day price (i.e., FEI crystalizing a price spread from buying summer gas and selling winter gas). The results of the scenarios for an annual standing demand charge are as follows:

Results of Annual Demand Charge Scenarios

Scenario #I:\$5.2 MM CDN annually for every 50 MMcf/d. (50 MMcf/d = 53,540 GJs/d)Scenario #II:\$7.0 MM CDN annually for every 50 MMcf/d (50 MMcf/d = 53,540 GJs/d)

Detailed Financial Values Tables

The results of my financial values analysis are presented in the following table. These results are derived using the above financial variables and futures prices for Spectra-Stn2 and Huntingdon/Sumas as of February 29, 2024. These variables were used when calculating Question 4 scenarios #1 through #7 (i.e., variables added to market prices and then multiplied by their respective scenario volumes and days). Furthermore, these financial values do not incorporate value from an Annual Demand Charge.

Spectra-Stn2	A	verage ('24/'25)	Α	verage ('25/'26)	,	Average ('26/'27)	1	Average ('27/'28)	Α	werage ('28/'29)
Cad Dollars										
Scenario #1 (a) Annual (300 MMcf/d x 1.3 I	Days)									
Daily Escalator A	\$	2,885,102	\$	2,075,025	\$	1,680,236	\$	2,057,370	\$	1,332,143
Daily Escalator B (i)	\$	2,847,518	\$	2,037,440	\$	1,642,652	\$	2,019,785	\$	1,294,558
(ii)	\$	2,926,863	\$	2,116,786	\$	1,721,997	\$	2,099,131	\$	1,373,903
Daily Escalator C	\$	2,784,143	\$	1,968,571	\$	1,569,769	\$	1,953,630	\$	1,215,306
Scenario #1 (b) 5-Year Cumulative (300 MM	//cf/d	x 1.3 Days)								
Daily Escalator A	ş	2,885,102	ş	4,960,127	ş	6,640,363	ş	8,697,733	ş	10,029,876
Daily Escalator B (I)	ş	2,847,518	Ş	4,884,957	Ş	6,527,609	Ş	8,547,394	ş	9,841,951
(II) Deily Feedlates C	ş	2,926,863	Ş	5,043,649	Ş	6,765,646	ş	8,864,777	ş	10,238,680
Sconario #2 (a) Appual (200 MMcf/d x 2 Da	Ş	2,784,143	Ş	4,752,714	Ş	0,322,484	Ş	8,276,114	Ş	9,491,420
Daily Escalator A	γ>) ¢	4 438 619	¢	3 192 346	¢	2 584 979	¢	3 165 184	ć	2 049 450
Daily Escalator A	ŝ	4,380,796	ŝ	3,134,523	ŝ	2,534,575	ŝ	3,107,361	ŝ	1,991,627
(ii)	ŝ	4,502,867	ś	3,256,593	ŝ	2,649,227	ŝ	3,229,432	ŝ	2.113.698
Daily Escalator C	ś	4,283,297	ś	3.028.571	ŝ	2,415,030	š	3.005.585	ś	1.869.701
Scenario #2 (b) 5-Year Cumulative (300 MM	/cf/d	x 2 Days)		-,,	Ť	_, ,	Ť	-,,	Ť	_,,
Daily Escalator A	\$	4,438,619	\$	7,630,965	\$	10,215,944	\$	13,381,128	\$	15,430,578
Daily Escalator B (i)	\$	4,380,796	\$	7,515,319	\$	10,042,475	\$	13,149,836	\$	15,141,464
(ii)	\$	4,502,867	\$	7,759,460	\$	10,408,687	\$	13,638,118	\$	15,751,816
Daily Escalator C	\$	4,283,297	\$	7,311,868	\$	9,726,898	\$	12,732,483	\$	14,602,184
Scenario #3 (a) Annual (300 MMcf/d x 3 Da	ys)									
Daily Escalator A	\$	6,657,929	\$	4,788,518	\$	3,877,469	\$	4,747,776	\$	3,074,175
Daily Escalator B (i)	\$	6,571,194	\$	4,701,784	\$	3,790,734	\$	4,661,042	\$	2,987,441
(ii)	\$	6,754,300	\$	4,884,890	\$	3,973,840	\$	4,844,148	\$	3,170,546
Daily Escalator C	\$	6,424,946	\$	4,542,856	\$	3,622,545	\$	4,508,377	\$	2,804,552
Scenario #3 (b) 5-Year Cumulative (300 MM	//cf/d	x 3 Days)								
Daily Escalator A	\$	6,657,929	\$	11,446,447	\$	15,323,916	\$	20,071,692	\$	23,145,867
Daily Escalator B (i)	ş	6,571,194	Ş	11,272,979	Ş	15,063,713	Ş	19,724,755	ş	22,712,196
(ii)	ş	6,754,300	ş	11,639,190	ş	15,613,030	ş	20,457,177	ş	23,627,724
Daily Escalator C	Ş	6,424,946	Ş	10,967,802	Ş	14,590,347	Ş	19,098,724	Ş	21,903,276
Scenario #4 (a) Annual (300 MMcf/d x 5 Da	ys)									
Daily Escalator A	ş	11,096,548	ş	7,980,864	Ş	6,462,448	ş	7,912,960	ş	5,123,625
Daily Escalator B (I)	ş	10,951,991	Ş	7,836,307	\$	6,317,891	Ş	7,768,403	ş	4,979,068
(11)	ş	11,257,167	Ş	8,141,483	Ş	6,623,067	Ş	8,073,579	Ş	5,284,244
Daily Escalator C	\$ •-664	10,708,244	Ş	7,571,427	Ş	6,037,575	Ş	7,513,962	Ş	4,674,253
Scenario #4 (b) 5-Year Cumulative (300 MM	/ict/d	x 5 Days)	~	40.077.442	~	25 520 050	~	22.452.020		20 576 445
Daily Escalator A	Ş	11,096,548	Ş	19,077,412	Ş	25,539,859	Ş	33,452,820	ş	38,576,445
(ii)	ç	11 257 167	ç	10,700,290	ç	25,100,188	ç	34,095,296	ç	39 379 540
Daily Escalator C	ç	10 708 244	ç	19,398,630	ç	20,021,710	ç	34,095,290	ç	36 505 460
Scenario #5 (a) Annual (300 MMcf/d x 7 Da	vel	10,700,244	ý	10,275,070	2	24,517,245	ý	51,651,207	Ş	50,505,400
Daily Escalator A	Ś	15.535.167	Ś	11.173.209	s	9.047.427	s	11.078.144	Ś	7,173,075
Daily Escalator B (i)	ŝ	15,332,787	ś	10,970,829	ś	8,845,047	š	10,875,764	ś	6,970,695
(ii)	ŝ	15,760,034	ŝ	11,398,076	\$	9,272,293	ŝ	11,303,011	ŝ	7,397,942
Daily Escalator C	\$	14,991,541	\$	10,599,997	\$	8,452,605	\$	10,519,546	\$	6,543,955
Scenario #5 (b) 5-Year Cumulative (300 MM	//cf/d	x 7 Days)								
Daily Escalator A	\$	15,535,167	\$	26,708,376	\$	35,755,803	\$	46,833,947	\$	54,007,023
Daily Escalator B (i)	\$	15,332,787	\$	26,303,617	\$	35,148,663	\$	46,024,428	\$	52,995,123
(ii)	\$	15,760,034	\$	27,158,110	\$	36,430,403	\$	47,733,414	\$	55,131,356
Daily Escalator C	\$	14,991,541	\$	25,591,538	\$	34,044,143	\$	44,563,689	\$	51,107,644
Scenario #6 (a) Annual (300 MMcf/d x 10 D	ays)									
Daily Escalator A	\$	22,193,096	\$	15,961,728	\$	12,924,895	\$	15,825,920	\$	10,247,250
Daily Escalator B (i)	ş	21,903,982	Ş	15,672,613	Ş	12,635,781	Ş	15,536,806	Ş	9,958,136
(ii)	ş	22,514,334	ş	16,282,966	ş	13,246,133	ş	16,147,158	ş	10,568,488
Daily Escalator C	Ş	21,416,487	Ş	15,142,853	Ş	12,075,150	Ş	15,027,923	Ş	9,348,507
Scenario #6 (b) 5-Year Cumulative (300 MM	/ict/d	x 10 Days)	~	20 454 024	~	51 070 710	~	66 005 630	<i>.</i>	77 452 880
Daily Escalator A	ç	22,193,090	ç	27 576 505	ç	51,079,719	ç	65 749 192	ç	75 707 219
(ii)	¢	22,503,582	ç	38 797 300	ç	52 043 433	ç	68 190 591	ç	78 759 079
Daily Escalator C	¢	22,514,554	ç	36 559 341	ç	48 634 490	ç	63 662 413	ç	73 010 920
Scenario #7 (a) Annual (300 MMcf/d x 10 D	ave V	ear 1 & 3 Days Ves	are 2	-51	Ý	40,054,450	Ý	05,002,415	Ý	75,010,510
Daily Escalator A	Ś	22.193.096	s	4.788.518	Ś	3.877.469	Ś	4.747.776	Ś	3.074.175
Daily Escalator B (i)	š	21,903,982	ś	4,701,784	ŝ	3,790,734	š	4.661.042	ś	2,987,441
(ii)	ŝ	22,514,334	ŝ	4,884,890	ŝ	3,973,840	ś	4,844,148	ś	3,170,546
Daily Escalator C	ŝ	21,416,487	ŝ	4,542,856	ŝ	3,622,545	ŝ	4,508,377	ŝ	2,804,552
Scenario #7 (b) 5-Year Cumulative (300 MM	//cf/d	x 10 Days Year 1 &	& 3 C	Days Years 2-5)	į.		Ċ.			
Daily Escalator A	\$	22,193,096	\$	26,981,614	\$	30,859,083	\$	35,606,859	\$	38,681,034
Daily Escalator B (i)	\$	21,903,982	\$	26,605,766	\$	30,396,500	\$	35,057,542	\$	38,044,983
(ii)	\$	22,514,334	\$	27,399,224	\$	31,373,064	\$	36,217,211	\$	39,387,758
Daily Escalator C	\$	21,416,487	\$	25,959,343	\$	29,581,888	\$	34,090,265	\$	36,894,817
Formulas:										
Daily Escalator A ((Sumas Winter + Daily E	scala	ntor A) - Summer S	pect	tra-Stn2) x 321,23	80	ājs				

Daily Escalator B ((Sumas Winter + Daily Escalator B) - Summer Spectra-Stra2) × 321,238 Gjs Daily Escalator C ((Sumas Winter × Daily Escalator C) - Summer Spectra-Stra2) × 321,238 Gjs

Appendix G
TLSE DETAILED COST ESTIMATE SUMMARY 3 BCF

FILED CONFIDENTIALLY

Appendix H DETAILED PROJECT SCHEDULE



Activity ID	Activity Name	Rem. Duration	Start	Finish		124	2025	+
Tilbury Island	I I NG Facility Expansion (3 BCE)	1570d	03-Jun-19A	16-Dec-30				2
				40.0				
Tilbury LNG S	Storage Expansion (TLSE)	1570d	03-Jun-19A	16-Dec-30				
Milestones		1537d	03-Jun-19A	16-Dec-30				Ť
Tilbury LNG Stora	ge Expansion (TLSE) Project Milestones	1537d	06-Jul-20 A	16-Dec-30	Y			-
MST102	Regas. Package Risk Evaluation	0d		06-Jul-20 A	♦ Regas. Package Risk Evaluation			
MST104	Regas. Package Basic Engineering Completion	0d		21-Jul-20 A	♦ Regas Package Basic Engineering Completion			
MST106	LNG Storage Tank/Auxiliary System FEED Completion	0d		31-Jul-20 A	◆ LNG Storage Tank/Auxiliary System FEED Com	pletion		
MST110	Submit CPCN Application	0d		29-Dec-20 A	 Submit CPCN Application 			
MST182	Re-Submit CPCN Application	0d		13-Sep-24		♦ Re-S	3ubmit C	PC
MST192	Submit Phase 2 Draft EA Application to EAO	0d		13-Sep-24		♦ Subr	nit Phas	e2
MST202	BC EAO Provides Feedback on Draft Application	0d		09-May-25			♦ BC F	EAC
MST122	Submit Phase 2 EA Application	0d		06-Jun-25			♦ Sut	۲mi
MST120	Receive CPCN Approval	0d		12-Sep-25			•	Re
MST121	Phase Gate Health Check (TBC)	0d		29-Jan-26				
MST134	Phase 2 EA Certificate (Provincial/Federal)	0d		16-Mar-26				
MST114	TLSE EPC Contract/SO Award	0d		29-Jun-26				
MST128	Auxiliary System EPC Kick-Off Meeting	0d		30-Jun-26				
MST131	Early Ground Work Mobilization/Construction Start - Regas Area	Od		02-Nov-26				
MST130	Base Plant Demolition Kick-Off Meeting	Od		17-Feb-27				
MST132	Early Ground Work Completion - Regas Area (TBC)	Od		23-Mar-27				
MST138	LNG Storage Tank Detailed Design Completion	0d		23-Jul-27				
MST142	Regas. Packages Auxiliary System Construction Completion	0d		07-Dec-27				
MST140	Regas. Packages/Auxiliary Systems Ph1 Commissioning	Od		22-Aug-28				
MST144	Base Plant Demolition Completion	0d		23-May-29				
MST148	Regas. Packages/Auxiliary Systems Ph2 Commissioning	0d		21-Jun-29				
MST150	LNG Tank Auxiliary System Completion	Od		18-Oct-30		·		
MST152	LNG Storage Tank Expansion Completion and Commissioning	0d		18-Oct-30				
MST162	TLSE Project In Service	0d		18-Oct-30				
MST172	TLSE Project Technical/Commercial Close-out	0d		16-Dec-30				
Tilbury LNG Stora	ge Expansion (TLSE) Assurance Milestones	866d	24-Apr-20 A	18-Oct-30				+
Tilbury LNG Stora	ge Expansion (TLSE) Executive Level Milestones	1296d	03-Jun-19A	16-Dec-30				
Execution		1530d	15-May-20 A	18-Oct-30				÷
CPCN Application	Review & Approval Process	274d	16-May-20 A	12-Sep-25				
Tank and Re-gasi	ification Equipment Inputs	Od	18-May-20 A	10-Jun-20 A			1 1 1 1 1 1 1 1 1	
CPT020102.1	Feasible Alternative Class 4 Estimate	0d	18-May-20 A	10-Jun-20 A			1 1 1 1 1 1 1 1 1	
Draft CPCN Appli	cation	Od	16-May-20 A	28-Oct-20 A				
CPT020202.1	Draft CPCN Application (All Sections)	Od	16-May-20 A	28-Oct-20 A			1 1 1 1 1 1 1 1 1	
CPCN Application	n Reviews/Submission	0d	18-May-20 A	29-Dec-20 A				
CPT020302	Revie, Finalize and Submit Application	Od	18-May-20 A	29-Dec-20 A				
CPT020304	Submit CPCN Application	Od		29-Dec-20 A	Submit CPCN Application			
		1	1			<u>i</u>	i	
Remaining Leve	el of Effort 🛛 💻 Actual Work 🛛 📕 🛏 Critical Remaining Work			Data Date:2	6-Jul-24	Date		
Actual Level of I	Effort Effort Remaining Work Critical Secondary			Page: 1	of 73	J-24		

SC MAN	DLARIS	5
2026 2027 2028	2029 2	2030 2031
+		
CNApplication		
2 Draft EA Application to EAO		
O Provides Feedback on Draft Applicat	on	
nit Phase 2 EA Application		
eceive CPCN Approval		
◆ Phase Gate Health Check (TBC)		
♦ Phase 2 EA Certificate (Provincial/F	ederal)	
♦ TLSE EPC Contract/SO Award		
♦ Auxiliary System EPC Kick-Off	<i>l</i> leeting	
♦ Early Ground Work Mobiliz	ation/Constructio	n Start - Regas Ar
♦ Base Plant Demolition I	Kick-Off Meeting	
🔶 Early Ground Work C	ompletion - Rega	sArea (TBC)
♦ LNG Storage Tan	k Detailed Desigi	n Completion
♦ Regas. Pack	ages Auxiliary Sy	stem Construction
♦ Reg	jas. Packages/Ai	uxiliary Systems Pl
	♦ Base Plant	Demolition Comp
	🔶 Regas. Pa	ackages/Auxiliary
		♦ LNG Tank
		♦ LNG Stora
		♦ TLSE Proje
		♦ TLSE Pre
		
· · · · · · · · · · · · · · · · · · ·		
Revision	Checked	Approved



	FORTIS BC	Tilbu TLSE	ry Island I <mark>Sub-Proj</mark> e	LNG Facili cts Integra	ty Expansion (3 BCF) ted Schedule-Detailed	l			*	1111		OLAR N A G E M I NSULTANTS	ENT INC.	
Activity ID	Activity Name	Rem.	Start	Finish	20 2021 2022	2023	2024	2025	2026	2027	2028	2029	2030	2031
			11 Mar 21 A	10 Con 05										
		2740	17-War-21A	12-Sep-25	BCLIC Information	Request 1 to EF								
CP1020412.01	ECUC Information Request 1 to FEI	DU Dd	17-Jun-21A	12 Cam 01 A			-							
CP1020412.03		04	22-JUII-21 A	13-Sep-21A	▲ Intervener Inform	ation Request 1	o FFI							
CP1020412.02		04	14 Son 21 A	12 Oct 21 A										
CP1020412.04		04	14-Sep-21A	12-00-21 A										
CP1020412.05	FEI Responses to BCOC and Intervenen IRZ	00	13-0ct-21 A	10-NOV-21 A										
CP1020412.00	Procedural Contenence Time & Formal TBD by BCOC	00	29-1100-21 A	29-IN0V-21 A	4									
CPT020412.12	Feller Information Request (IR) No. 1 to FEl	00	24-Jan 22 A	20-Jan-22 A										
CPT020412.14	TAIN Oral Intervener Evidence	00	05 Apr 22 A	04-IVIAI-22 A										
CP1020412.10	FEI Bespenses to IDe on Bebuttel Evidence /Depty Submission on Eurther Process	04	05-Api-22 A	00-Api-22 A										
CP1020412.21	FEI Written Denk Argument	00	24-Jun-22 A	28-JUN-22A										
CP1020412.23	PEI Whiten Reply Argument	DU	24-INOV-22 A	12-Dec-22 A										
CPT020412	CPCN Workshop	0d	11-Mar-21 A 11-Mar-21 A	23-Mar-23 A 11-Mar-21 A										
CPT020412 11	FEL& Intervener Submissions on TWN Request to Exclude Interveners from Heari	b0 D0	10-Jan-22 A	21-Jan-22 A										
CPT02041213	TWN Renty to Submissions on TWN Request to Exclude Interveners from Hearing	D0	26- Jan-22 A	28-Jan-22A										
CPT020412.15		D0	07-Mar-22 A	28-Mar-22 A										
CPT020412.15	1 TWN Written Evidence	D0	29-Mar-22 A	04-Apr-22 A										
CPT020412.10.	BCLIC FELand Intervener IRs on Intervener Written Evidence (except TWN Writte	D0	07-Apr-22 A	22-Apr-22 A										
CPT020412.17	BCUC FEI and Intervener IRS on TWN Written and Oral Evidence	Dd	25-Apr-22 A	22-Api-22 A										
CPT020412.17.	Responses to IRs on Intervener Written Evidence and TWN Oral Evidence	Dd	23-Api-22 A	20-May-22 A										
CPT020412.10	Rebuttal Evidence (if required)	Dd	31-May-22 A	02- lun-22 A										
CPT020412.10	IRs on Rebuttal Evidence	Dd	03- lun-22 A	23- Jun-22 A										
CPT020412.20	Intervener Submissions on Further Process	DQ DQ	29- lun-22 A											
CPT020412.21.	2 FEI responses to IRs on Rebuttal Evidence	D0	07- Jul-22 A	14- Jul-22 A										
CPT020412.21.2	3 FEI Renty submission on further process	Dd	14- μL22 Δ	14-Jul-22 Α										
CPT020412.21.0	1 EEI Written Einal Argument	Dd	15- Jul-22 A	24_Oct_22 A										
CPT020412.21.4		Dd	25-0ct-22 A	23-Nov-22 A										
CPT020412.22		Dd	13-Dec-22 A	23-Mor-22 A										
CPCN Application I		334	24 Mar 23 A	13 Son 24										
CPT020412.25	Review Questions by Management Team	0d	24-Mar-23 A	31-Aug-23 A					i					
CPT020412.26	Prepare Answers/TLSE Estimate Update & Resiliency	0d	02-Mav-23 A	26-Jul-24 A										
CPT020412.27	TLSE CPCN Application Final Review/Submission	33d	29-Jul-24	13-Sep-24			TLS	E CPCN Ap	plication Fir	nal Review/Su	Ibmission			
BCUC Final Review	v & Approval	241d	16-Sep-24	12-Sep-25										
CPT020404	BCUC Review Application Updates/Receive CPCN Approval	241d	16-Sep-24	12-Sep-25				BC	UC Review	Application L	Ipdates/Rec	eive CPCN A	oproval	
Environmental As	sessment - Phase 2 (Execution Phase)	392d	15-May-20 A	16-Mar-26					◀ ! !		4		JLJ 	
Detailed Project	Description Development	0d	15-May-20 A	03-Sep-21 A										
EAT020313	Pre-FEED information required for Field Programs	0d	15-May-20 A		Pre-FEED information required f	or Field Program	IS		1 5 1 1 5 5 1 5 7					
EAT020314	Develop DPD and AIR + FEI Reviews	0d	27-Jul-20 A	31-May-21 A										
EAT020315	Submit DPD Draft	0d		01-Jun-21 A	Submit DPD Draft									
EAT020315.1	DPD Part 2 Workshop	0d	16-Jun-21 A	16-Jun-21 A					! & I I I I I I I I			4		
EAT020315.2	DPD Technical Advisor Review	0d	17-Jun-21 A	30-Jun-21 A										
]					i		<u> </u>	i		ii	i
Remaining Leve	el of Effort Actual Work Eritical Remaining Work			Data Date:2	6-Jul-24	40.1	Date		F	Revision		Checked	d Ap	pproved
Actual Level of I	Effort Effort Remaining Work E			Page: 2	of 73	19-A	ug-24							
								1						



Activity	ID	Activity Name	Rem.	Start	Finish	020		20	21		2022		202	23		2024		2	2025	
			Duration			Q	Q	QQ	QC	QQ	QQ	QC	QQ	QQ	Q	QQ	Q	Q	ຊຸດ	≀ Q
	EAT20845	Submit DPD Final	0d	02-Jul-21 A	03-Sep-21 A															
	Readiness Decisi	on (Detailed Project Description Review)	0d	09-Sep-21 A	20-Jan-22 A					-				1				1		
	EAT020417	BC EAO Readiness Decision and Seeking Consensus	0d	09-Sep-21 A	12-Jan-22 A															
	EAT020529	IAAC IA Determination and Substitution Decision	0d	13-Jan-22 A	20-Jan-22 A					1										
	BC EAO Process	Planning	0d	21-Jan-22 A	13-Jun-22 A						-									
	EAT020519	Meetings with TAC/CAC, Finalize AIR, and Develop Permit Plan	0d	21-Jan-22 A	22-Apr-22 A															
	EAT020520	Public Comment Period on Draft Process Order	0d	24-Feb-22 A	11-Apr-22 A													1		
	EAT020521	Issue BC EAO Process Order	0d	12-Apr-22 A	13-Jun-22 A															
	Draft Application	Development	33d	04-Jan-22 A	13-Sep-24							:						1	1	1
	EAT020627	Draft Application Development + FEI Review	0d	04-Jan-22 A	26-Aug-22 A									1				1		1
	EAT020624	Field work planning	0d	31-Mar-22 A	23-Jun-22 A									1				1		1
	EAT020625	Biophysical Baseline Data Collection and Analysis	0d	31-Mar-22 A	23-Jun-22 A	1								1				1		1
	EAT020626	TEK/ TUS Studies	0d	31-Mar-22 A	23-Jun-22 A	1								1				1		
	EAT020628	Stakeholder Draft Application Review	0d	18-Jul-22 A	15-Dec-22 A							-								
	EAT020629	Finalize Draft Application *Pending executive decisions*	33d	16-Dec-22 A	13-Sep-24							🗖				-	F	naliz	e Dr	aftA
	EAT020629.1	EAO Revised Process Order	29d	04-Jul-23 A	09-Sep-24						1						I E/	NO R	levis	sed F
	EAT020630	Submit Draft Application to EAO	0d		13-Sep-24												s S	ıbmi	t Dra	aft Ap
	BC EAO Draft App	lication Review	155d	16-Sep-24	09-May-25														7	
	EAT020732	Meetings with TAC/CAC	60d	16-Sep-24	11-Dec-24								,				—	Me	eting	js wi
	EAT020733	Draft Application Review/Public Comment Period	30d	12-Dec-24	05-Feb-25						1			-			1	D	raft	Appl
	EAT020734	Meetings with First Nations	30d	12-Dec-24	05-Feb-25									-			1	N	leeti	ings
	EAT020734.1	Review Period (6 Months from Draft Submission)	65d	06-Feb-25	09-May-25									-				-	Re	eviev
	EAT020735	BC EAO Provides Feedback on Draft Application	0d		09-May-25									-				•	BC	CEA
	Final Application	Development	65d	13-Mar-25	13-Jun-25													V	•	
	EAT020737	Final Application Development (including FEI Review)	60d	13-Mar-25	06-Jun-25									-				Ļ	F	inal/
	EAT020738	Submit Final Application to EAO	0d		06-Jun-25						1			-					♦ S	ubr
	EAT020739	BC EAO Acceptance of Final Application	5d	09-Jun-25	13-Jun-25						1			-				1	I E	3C E
	BC EAO Effects A	ssessment and Recommendation	120d	16-Jun-25	08-Dec-25						1			-				1	-	
	EAT020841	Legislated EAO Review and Recommendation	90d	16-Jun-25	24-Oct-25															i i
	EAT020843	First Nations Expression of Consent	50d	27-Aug-25	07-Nov-25									-				1		ė '
	EAT020842	Final Application Public Comment Period	20d	10-Nov-25	08-Dec-25															
	Application Subm	ission and Approval Period	60d	08-Dec-25	16-Mar-26															▼
	EAT020844	BC EAO Submits Referral Package to Provincial Minister and IAA	0d		08-Dec-25									-						•
	EAT020844.1	IAAC Issues Report to Federal Ministers	5d	09-Dec-25	15-Dec-25	1														
	EAT020845	Decision by Provincial and Federal Minister (Assuming Substituted Process)	55d	16-Dec-25	16-Mar-26															
	MST134.1	Phase 2 EA Certificate (Provincial/Federal)	0d		16-Mar-26															
	Management Plan	for Ph2 EAC	240d	27-Aug-25	21-Aug-26													1	T	
	GIT500010	Management Plan Development and Approval	240d	27-Aug-25	21-Aug-26				-					1				1		÷
	TLSE Contracting		<u>191d</u>	15- <u>Sep-25</u>	29-Jun-26															
	Geotech/Ground	Improvement	80d	24-Sep-25	29-Jan-26									1				1		-
	GR0.0	Geotech/Detailed Design Contracting Process	80d	24-Sep-25	29-Jan-26				-	1	1			1				1 1 1	1	
	GR0.0	Geotech/Detailed Design Contracting Process	80d	24-Sep-25	29-Jan-26	1	1				1			1	1	1	1	-	_	

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data:26 Jul 24	Date	
			Data Date:20-Jul-24	19-Aug-24	IFI
			Fage. 5 01 75		

SOLARIS

2026		20	27			20	28			20	29			20	30		2	203	1
	QQ	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q
										_	_								-
		· · ·											1						
		, 																	
		· ·																	
		· ·																	
														;			;		
		· ·		1											1		1		
		· ·		1															
		· ·		1 1 1			-								1		-		
		· ·													-		-		1
+				+										;					¦ ; ;
plication	*Pen	ding	g ex	xec	utiv	e d	eci	sior	ıs*										1
ocess Or	der	· ·													-		-		1
olication to	⊳ĖAC	>																	
i TAC/CA	c													}					
ation Rev	view/F	ub	lic (Cor	nme	ent	Pei	iod							1		1		1
/ith First N	lation	s					-								-		-		1 1 1
Period (6	6 Mor	ths	fro	ml	Dra	ft S	ubr	niss	sion)			1		1		1		
) Provides	sFee	db	ack	on	Dra	aft A	App	lica	tior	י ז					-		-		
polication	Deve		me	nt	ínc	ludi	na	FF	I Re	vie	w/)								
Final Anr	licatio	on f	οE			uui		• -			,								
		of	U L	лс ы А	nnl	icat	ion						1		1		1		
	ance		1110	ai A	hhi	uai													
		Da	lici		- d 1	5	~~~~	m	nd	otic	n			;			1		
irst Motion		nte) pre		v al	iu i f C	160		1111E +	nu.	auO	11				-		-		1 1 1
list ivalior	is ⊏X	pie 2 P	5510	י ו ופ היי			en,	l Der							-		-		1
rinai Appl	ICatio	nΡ	upl	ic U	om	ime	nti	-er	οa						-		-		1
					_1 ~														1
BC EAO S	Subm	nits I	Ret	err	al P	ack	ag	e to) Pr	ovii	ncia	al M	INIS	ter	and	AI L	A		
IAAC Issu	ies R	ерс	ort t	οF	ede	eral	Mir	niste	ers						_	.			
Decisi	on by	/ Pr	ovir	ncia	ıl ar	nd F	ed	era	I M	inis	ter	(As	sur	ning	g S	ubs	stitu	ted	Pro
 Phase 	2 E/	١Ce	ertif	icat	e (F	۲o	ving	;ial/	Fee	der	al)				1		1		1
																			1
N	Mana	ger	ner	nt P	lan	De	vel	opr	ner	it ai	nd /	٩рр	rον	'al					1
		,, , , , ,								1									
7																			
Geotecl	∩/Þet	aile	d D)esi	gn	Cor	ntra	ctir	ng F	Pro	ces	s							
						. 1													'
	Revis	sion								(Che	cke	d			App	prov	ed	



Activity	D	Activity Name	Rem.	Start	Finish	020	2021	2022		2023		2024	20)25	2026	2027	2028	2	029	2030	203	1
			Duration			Q			QQ	QQ	QQ	QQ		QC			QQQ	QQC			QQQ	Q
	EPC		191d	15-Sep-25	29-Jun-26										7							:
	TRP296202	Prepare Bidding Documents	40d	15-Sep-25	12-Nov-25	5									Prepare Biddi	ng Docume	nts					
	TRP600201	Cash Flow Planned Start Date - Regas. Packages (See Act. Note)	0d	15-Sep-25								· · · ·		• (Cash Flow Plan	ned Start D	ate - Rega	s. Packa	ages (Se	e Act. No	ote)	
	TAX296100	Cash Flow Planned Start Date - Auxiliary (See Act. Note)	0d	15-Sep-25										• (Cash Flow Plan	ned Start D	ate - Auxilia	ary (See	Act. Note	e)		
	TBP296102	Cash Flow Planned Start Date - Base Plant Demo (See Act. Note)	0d	15-Sep-25										• 0	Cash Flow Plan	ned Start D	ate - Base	Plant D	emo (See	Act. No	ote)	
	TTK296102	Cash Flow Planned Start Date - LNG Tank (See Act. Note)	0d	15-Sep-25										• 0	Cash Flow Plan	ned Start D	ate - LNG	Tank (Se	ee Act. No	ote)		
	TRP296204	Contract Request Form/Email	5d	07-Nov-25	14-Nov-25	5								1	Contract Req	uest Form/E	mail					1
	TRP296206	Review/Draft Requirements and Discussion with Stakeholders	10d	17-Nov-25	28-Nov-25	5									Review/Draft	Requireme	nts and Dis	scussior	n with Sta	keholde	ers 🛛	:
	TRP296208	Review Competetive Bid Documents	10d	24-Nov-25	05-Dec-25	5									Review Com	petetive Bid	Documen	ts				
	TRP296210	Circulate RFQ Internally for Review, Comments and Amendments	10d	01-Dec-25	12-Dec-25	5									Circulate RF	Q Internally	for Review	ı, ¢omm	ents and	Amend	Iments	:
	TRP296212	Legal Review	10d	08-Dec-25	19-Dec-25	5									Legal Review	N						1
	TRP296214	Finalize RFQ	10d	12-Dec-25	07-Jan-26	5									📕 Finalize RF	Q						
	TRP296216	Issue TLSE EPC RFQ	0d		07-Jan-26	5									Issue TLSE	EPC RFQ						
	TRP296218	Receive Vendor/Contractors Proposals & Clarification	60d	08-Jan-26	02-Apr-26	5									E Receive	Vendor/Cor	tractors Pi	roposals	s & Clarific	ation		
	TRP296220	Finalize SOW/Update Cost Estimate/Bid Evaluation//Negotiation/PR Creation and /	60d	03-Apr-26	29-Jun-26	5									📕 Finali	ze SOW/Up	date Cost	Estimate	e/Bid Eva	uation//	Negotiatio	n/F
	TRP296222	Award EPC Contract (Excluding LNG Tank Proc./Const.)	0d	-	29-Jun-26	5									♦ Aware	d EPC Cont	act (Exclu	ding LN	G Tank P	roc./Coi	nst.)	
	Phase Gate Health	n Check (Required Deliverables/Documents and Review)	87d	15-Sep-25	29-Jan-26									-	▼							i
	Phase Gate Healt	th Check Required Deliverables/Documents	60d	15-Sep-25	10-Dec-25	5											$ \frac{1}{1} \frac{1}{1} \frac{1}{1}$					
	TLT240200	Phase Gate Health Check Scope/Deliverables (TBD)	60d	15-Sep-25	10-Dec-25	5									Phase Gate	Health Che	ck Scope/I	Delivera	bles (TBD))		
	Gate (Health Che	ck) Review	27d	10-Dec-25	29-Jan-26																	
	TLT240302	Issue Documents to Peer Review Team	0d		10-Dec-25	5									Issue Docun	nents to Pee	r Review 7	F eam				
	TLT240304	Peer Review	5d	11-Dec-25	17-Dec-25	5									Peer Review	/						
	TLT240306	Incorporate Comments and Provide Feedback to Peer Review Team	10d	18-Dec-25	13-Jan-26	;									Incorporate	Comments	and Provi	de Feec	lback to F	Peer Re	view Tean	ji -
	TLT240308	Generate Recommendation Completion	0d		15-Jan-26	;									♦ Generate F	Recommend	ation Com	pletion				
	TLT240310	Phase Gate Health Check Review Period	10d	16-Jan-26	29-Jan-26								1		Phase Gat	e Health Ch	eck Reviev	w Period	b			
	Mid Phase 4 Revie	ew - Required Deliverables/Documents and Review	55d	30-Mar-27	16-Jun-27								1			T						
	Mid Phase 4 Revie	ew - Required Deliverables/Documents	35d	30-Mar-27	18-May-27	7								1 1 1 1 1 1		₩						
	Lessons Learned - F	Phase 3	35d	30-Mar-27	18-May-27	7										₩						
	A9992	Prepare and Issue Previous Phase Lesson Learned - IFIR	20d	30-Mar-27	28-Apr-27											Prepar	e and Issu	ie Previo	ous Phas	e Lesso	on Learned	- L
	A9993	Internal/Peer Review - Previous Phase Lesson Learned	5d	28-Apr-27	04-May-27	7										I Interna	l/Peer Rev	view - Pı	revious P	hase Le	esson Lea	rne
	A9994	Incorporate Comments and Issue Previous Phase Lesson Learned - IFGR	10d	05-May-27	18-May-27	7										Incorp	orate Con	nments a	and Issue	Previo	us Phase	Les
	PEP (Project Execut	ition Plan) Final	35d	30-Mar-27	18-May-27	7																
	A9986	Prepare and Issue PEP - IFIR	20d	30-Mar-27	28-Apr-27	·										Prepar	e and Issu	ie PEP -	IFIR			
	A9987	Internal/Peer Review - PEP	5d	28-Apr-27	04-May-27	7										I Interna	l/Peer Rev	view - Pl	EP			
	A9988	Incorporate Comments and Issue PEP - IFGR	10d	05-May-27	18-May-27	7										Incorp	orate Con	nments a	and Issue	PEP - I	IFGR	
	Project Organization	n Plan: Final	35d	30-Mar-27	18-May-27	7								, 1 1 1 1 1		₩_						
	A10004	Prepare and Issue Project Organization Plan - IFIR	20d	30-Mar-27	28-Apr-27							1 1 1 1 1 1		· · · · · · · · · · · · · · · · · · ·		Prepar	e and Issu	ie Proje	ct Organi	zation P	ian - IFIR	
	A10005	Internal/Peer Review - Project Organization Plan	5d	28-Apr-27	04-May-27	7										I Interna	VPeer Rev	view - Pi	roject Org	anizatio	n Plan	
	A10006	Incorporate Comments and Issue Project Organization Plan - IFGR	10d	05-May-27	18-May-27	7							1	1 1 1 1 1 1		Incorp	orate Con	nments a	and Issue	Project	t Organiza	tior
	Cost Estimate: AAC	E Class 2	35d	30-Mar-27	18-May-27	7																
	A10007	Prepare and Issue Class 2 Cost Estimate - IFIR	20d	30-Mar-27	28-Apr-27											Prepar	e and Issu	le Class	∠ Cost E	stimate	- IFIK	
	A10008	Internal/Peer Review - Class 2 Cost Estimate	5d	28-Apr-27	04-May-27	7							1			I interna	vreer Ke	view - C	iass 2 Co	st⊨stim	ate	

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data:26 Jul 24	Date	
	Remaining Work		Data Date:20-Jul-24	19-Aug-24	IFI
			Fage. 4 01 75		

SOLARIS

Revision	Checked	Approved



Activ	ity ID	Activity Name	Rem.	Start	Finish)20	2021	2022	2023	2024	2025	2026		2027	2028	2029	20	030	2031
			Duration													QQQ			QQQ
	A10009	Incorporate Comments and Issue Class 2 Cost Estimate - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	nents and Is	sue Cla	iss 2 Co	st Estimat
	Master Schedules	e: AACE Class 2	35d	30-Mar-27	18-May-27									Proporo	nd loouo		bodulo		
	A10010	Prepare and Issue Class 2 Schedule - IFIR	20d	30-Mar-27	28-Apr-27												Rebodul		
	A10011	Internal/Peer Review Class 2 Schedule	50	28-Apr-27	04-May-27														la a du da - 1
	A10012	Incorporate Comments and Issue Class 2 Schedule - IFGR	10d	05-May-27	18-May-27				· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · ·		 · · · · · · · · · · · · · · · · · · ·	· · · · · · · · ·	Incorpor	ate Cornn	ients and is	sue cia	ISS Z SCI	1eaule - I
Ш	Schedule: AACE	Class 1 (as required)	35d	30-Mar-27	18-May-27									Propara	ndiesuo	Close 1 Se	bodulo		
	A12000	Prepare and Issue Class 1 Schedule - IFIR	20d	30-Mar-27	28-Apr-27	_											Sobodul		
ш	A12010	Internal/Peer Review Class 1 Schedule	5d	28-Apr-27	04-May-27														ا ماريام
Ш	A12020	Incorporate Comments and Issue Class 1 Schedule - IFGR	10d	05-May-27	18-May-27									Incorpor	ale Cornin	ients and is	sue Cia	ISS I SCI	iedule - I
	Risk Managemen	nt Plan: Audit	35d	30-Mar-27	18-May-27							 · 4		Prenare	ndleeuo	Rick Mana	nement	Plan - IE	- IP
	A10013	Prepare and Issue Risk Management Plan - IFIR	200	30-Mar-27	28-Apr-27	-									oor Povic		appaom	ont Dion	
	A10014	Internal/Peer Review - Risk Management Plan	50	28-Apr-27	04-May-27	_													a operate T
	A10015	Incorporate Comments and Issue Risk Management Plan - IFGR	10d	05-May-27	18-May-27									Incorpor	ale Comm	ients and is	sue Ris	k ivianaę	Jemente
	Construction Rea	adiness Assessment	35d	30-Mar-27	18-May-27									Prenare	nd lesue	Constructio	n Road	inose Ae	seesmer
	A9989	Prepare and Issue Construction Readiness Assessment - IFIR	200	30-Mar-27	28-Apr-27				· · · · · · · · · · · · · · · · · · ·			 · 1 1	••••••				In Incau		
	A9990	Internal/Peer Review - Construction Readiness Assessment	5d	28-Apr-27	04-May-27	_							I			w - Consur		eaumes	
Ш	A9991	Incorporate Comments and Issue Construction Readiness Assessment - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	ients and is	sue Co	nstructio	in Readin
	BCER Monitoring	g & Compliance Plan: Final	35d	30-Mar-27	18-May-27									Propara	ndiesuo		itoring 8	Compli	ianco Dla
	A9995	Prepare and Issue BCER Monitoring & Compliance Plan - IFIR	20d	30-Mar-27	28-Apr-27	-													moliopoo
	A9996	Internal/Peer Review - BCER Monitoring & Compliance Plan	5d	28-Apr-27	04-May-27							 		internal/F					mpilance
	A9997	Incorporate Comments and Issue BCER Monitoring & Compliance Plan - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	nents and is	sue BC	ER Mon	itoring &
Ш	Permit Monitoring	g & Compliance Plan: Final	35d	30-Mar-27	18-May-27									Proporo	nd loouo	Dormit Mar	anama	nt Dlon	
ш	A10034	Prepare and Issue Permit Management Plan - IFIR	20d	30-Mar-27	28-Apr-27												agemei	ni Pian -	
Ш	A10035	Internal/Peer Review - Permit Management Plan	5d	28-Apr-27	04-May-27									internal/F	eer Revie	w - Permit	vianage	ment Pi	an
	A10036	Incorporate Comments and Issue Permit Management Plan - IFGR	10d	05-May-27	18-May-27							 · · · · · · · · · · · · · · · · · · ·		Incorpor	ate Comn	nents and is	sue Pei	rmit ivian	agement
	Project Controls	Plan: Final Project Performance Baselines	35d	30-Mar-27	18-May-27									Dronoro		Drainat Ca	atrol Dia		
Ш	A10001	Prepare and Issue Project Control Plan - IFIR	20d	30-Mar-27	28-Apr-27														
Ш	A10002	Internal/Peer Review - Project Control Plan	5d	28-Apr-27	04-May-27								I	internal/F	eerkevie	w - Project	Control	Plan	
	A10003	Incorporate Comments and Issue Project Control Plan - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	nents and Is	sue Pro	oject Cor	itrol Plan
	Utility & Crossing	g Requirements	35d	30-Mar-27	18-May-27							 · 4		Droporo					oonto IEI
Ш	A9998	Prepare and Issue Utility & Crossing Requirements - IFIR	20d	30-Mar-27	28-Apr-27									Prepare	inu issue			equirei	
	A9999	Internal/Peer Review - Utility & Crossing Requirements	5d	28-Apr-27	04-May-27								I	Internal/F	eer Revie	ew - Utility &	Crossin	ig Requi	rements
	A10000	Incorporate Comments and Issue Utility & Crossing Requirements - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	nents and Is	sue Util	ity & Cro	ssing Re
Ш	Construction read	diness review	35d	30-Mar-27	18-May-27											Construction			1514 1515
Ш	A10022	Prepare and Issue Construction readiness review - IFIR	20d	30-Mar-27	28-Apr-27							 · ! !		Prepare a	ina issue	Constructio	n readir	iess rev	
	A10023	Internal/Peer Review - Construction readiness review	5d	28-Apr-27	04-May-27									internal/F	eer Revie	ew - Constru	uction re	adiness	review
	A10024	Incorporate Comments and Issue Construction readiness review - IFGR	10d	05-May-27	18-May-27								I	Incorpor	ate Comn	nents and Is	sue Co	nstructio	n readine
Ш	Engineering and	Design Plan	35d	30-Mar-27	18-May-27												~ ~ ~ ~ ~ ~		
Ш	A10016	Prepare and Issue Engineering and Design Plan - IFIR	20d	30-Mar-27	28-Apr-27	_								Prepare a	ina issue	Engineerin	gand D	esign Pi	
Ш	A10017	Internal/Peer Review - Engineering and Design Plan	5d	28-Apr-27	04-May-27				· · · · · · · · · · · · · · · · · · ·			 		internal/F	eer Revie	w - Engine	ering an	ia Desig	n Plan
	A10018	Incorporate Comments and Issue Engineering and Design Plan - IFGR	10d	05-May-27	18-May-27									Incorpor	ate Comn	nents and Is	sue Eng	gineerin	g and De
	BCUC Regulator	ry Requirements	35d	30-Mar-27	18-May-27										ad le			Dearth	mont-
	A10019	Prepare and Issue BCUC Regulatory Requirements - IFIR	20d	30-Mar-27	28-Apr-27									Prepare a			Julaiory	Require	ments - 1
	A10020	Internal/Peer Review - BCUC Regulatory Requirements	5d	28-Apr-27	04-May-27									internal/F	eer Revie	W - BCUC	Regulat	ory Req	urements
	Remaining Le	evel of Effort Actual Work Critical Remaining Work			Data Data 2	6. 1.1.1	24			Date			Revisio	on		Chec	ked	Аррі	roved
	Actual Level o	of Effort Effort Remaining Work Critical Secondary			Page: 5 c	o-Jui-4				19-Aug-24	IFI	 							
		• — — •								1									

SOLARIS

M A N A G E M E N T CONSULTANTS INC.



Activ	rity ID	Activity Name	Rem.	Start	Finish	020	2021	20)22	2023	2024		2025	2026	3	2027	202	28	2029	20	30	2031
	440004			05.14 07	40.14 07			QQQ		QQQ		QQ										
	A10021	Incorporate Comments and Issue BCUC Regulatory Requirements - IFGR	10d	05-May-27	18-May-27							+					porate t	Johnnen	is anu			ulaiory r
	Environmental Impa	act Assessment and Pre-Construction Surveys (field work)	35d	30-Mar-27	18-May-27			-								Prens	are and	lssue Fn	vironm	ental Imna	ct Asse	ssment a
	A10037	Prepare and Issue Environmental Impact Assessment and Pre-Construction Suive	200	30-IVIAI-27	20-Api-27	_		-									al/Peer	Review.	Enviro	nmental l	nnact Δ	Jeepeeme
	A10038	Internal/Peer Review - Environmental Impact Assessment and Pre-Construction St	50	28-Apr-27	04-IVIAy-27	_		1									norato (Common	te opd	lecuo Env	ironmar	ntal Impa
	A10039	Incorporate Comments and Issue Environmental Impact Assessment and Pre-Con	10d	05-May-27	18-May-27												porate	Johnnen	is anu			nainnpa
	Environmental Man	agement Plan (includes archaeology)	35d	30-Mar-27	18-May-27											Prena	are and	Issue Fn	vironm	ental Man	agemer	nt Plan - I
	A12030		200	30-IVIAI-27	20-Api-27	_)al/Peer	Review.	Enviro	nmental N	lanade	ment Pla
	A12040		50	28-Apr-27	04-Iviay-27	_		-									norato (Common	te opd	lecuo Env	ironmar	ntal Man
	A12050	Incorporate Comments and Issue Environmental Management Plan - IF GR	100	05-May-27	18-May-27			-									porate c	Joinmen	is and			
	A10025	Propage and Issue Habitat Assocsments IFIP	35d	30-Mar-27	18-May-27			-								Prepa	are and	Issue Ha	bitat As	sessmen	s - IFIR	2
	A10025		200 Ed	20-101al-27	20-Api-27											Intern	al/Peer	Review	Habita	at Assessn	nents	
	A10020		50	20-Api-27	04-IVIAy-27			-									morate (Commen	te and		itat Acc	ecomont
	A10027	Incorporate Comments and Issue Habitat Assessments - IFGR	100	05-May-27	18-May-27			1									porate C	Johnmen	is and	ISSUE I IAL		COSTICUE
	A12060	mits Propare and leave Environmental Permite IEIP	35d	30-Mar-27	18-May-27			-								Prena	are and	Issue Fn	vironm	ental Pern	nits - IFI	IR
	A12000		200	30-IVIAI-27	20-Api-27			-									nal/Peer	Review.	Fnvire	nmental F	Permits	· ·
	A12070	Internal/Peer Review - Environmental Permits	50	28-Apr-27	04-Iviay-27												norate (Commen	te and		ironmor	ntal Dorm
	A12080	Incorporate Comments and Issue Environmental Permits - IFGR	100	05-May-27	18-May-27			-									porate c	Johnnen	is and			
	Archaeological Imp	Direct Assessment - Final	35d	30-Mar-27	18-May-27											Prepa	are and	Issue Arc	haeok	dical Imp	ict Asse	essment -
	A10040		200	30-IVIAI-27	20-Api-27	_		-								Intern	al/Peer	Review.	Archa	eological l	mnact 4	Assessm
	A10041		50	28-Apr-27	04-IVIAy-27												morato (Common	fe and			
	A10042	Incorporate Comments and Issue Archaeological Impact Assessment - IF GR	10d	05-May-27	18-May-27							+							is anu		acolog	icai in ipa
	Archaeological Mor	nitoring Plan & implementation during construction	35d	30-Mar-27	18-May-27			-					· · ·			Prena	are and	Issue Arc	haeok	ngical Mon	itorina F	Plan - IFIF
	A12090		200	30-IVIAI-27	20-Api-27	_		-									al/Peer	Review.	Δrcha	eological I	<i>I</i> onitorii	ing Plan
	A12100		50	28-Apr-27	04-IVIAy-27	_		-									norato (Common	te and			right Moni
	A12110	Incorporate Comments and Issue Archaeological Monitoring Plan - IF GR	10d	05-May-27	18-May-27			1									porate	Johnnen	is anu		acolog	
	Heritage Conservat	Drepare and leave Haritage Concentration Act Dermite JEID	35d	30-Mar-27	18-May-27											Prena	are and	lssue He	ritade (Conservat	ion Act I	Permits -
	A10031		200	30-IVIAI-27	28-Apt-27	_		1									al/Peer	Review.	Horita		rvation	
	A10032	Internal/Peer Review - Heritage Conservation Act Permits	50	28-Apr-27	04-Iviay-27	_		-									norato (Common	te and		togo ()	opeonyoti
	A10033	Incorporate Comments and Issue Heritage Conservation Act Permits - IFGR	10d	05-May-27	18-May-27												porate	JUITINET	is anu		laye G	Juseivau
	HSS Compliance M	Ionitoring: Audit	35d	30-Mar-27	18-May-27											Prena	are and	lssue HS	S Plan	- IFIR		
	A10043		200	30-IVIAI-27	28-Apt-27							4					al/Door	Roview		Plan		
	A10044	Internal/Peer Review - HSS Plan	50	28-Apr-27	04-Iviay-27	_		1									an cei	Common	to opd		Dian	
	A10045	Incorporate Comments and Issue HSS Plan - IF GR	10d	05-May-27	18-May-27			-									porate	Johnmen	is anu		FIAIT-	IFGR
	Construction Safety	Property and leave Construction Sofety Plan IEIP	35d	30-Mar-27	18-May-27											Prens	are and	lssue Co	nstruct	ion Safetv	Plan - I	IFIR
	A12120		200	30-IVIAI-27	20-Api-27												ne ana i val/Peer	Review.	Const	ruction Sa	foty Pla	n in t
	A12130	Internal/Peer Review - Construction Safety Plan	50	28-Apr-27	04-Iviay-27												an cor	Common	te and		etructio	n Safaty
	A12140	Incorporate Comments and Issue Construction Safety Plan - IF GR	10d	05-May-27	18-May-27			1									porate	JOININEI	is anu		Silucio	IT Salety
	External Relations	Plan - Update: Final Prepare and leave External Depletion Plan IEID	35d	30-Mar-27	18-May-27			-								Prena	are and	lssue Fx	ernal F	Replation I	Plan - IF	-IR
	A10040		200	30-IVIAI-27	20-Api-27			-								Intern	al/Peer	Review.	Exteri	al Renlati	n Plan	, , , , , , , , , , , , , , , , , , ,
	A10047		DC	28-Apr-27	04-iviay-27			1									norato (Common	te and			enlation F
	A10048	Incorporate Comments and Issue External Replation Plan - IFGR	10d	05-May-27	18-May-27												Porate	Joninen	is ai lù		andi Re	
	Land Acquisition M	Property and Issue Land Acquisition Management Plan JEIP	35d	30-Mar-27	18-May-27											Prens	are and	Issuela	nd Aca	uisition Ma	nadem	ıent Plan
	A10049		200	30-IVIAI-21	20-Api-27			-									al/Poor	Review	l and		Manac	dement C
	00001A		DC	20-Apr-21	04-iviay-27			1											Latiu		manag	
	Remaining Leve	el of Effort Element Actual Work Element Critical Remaining Work			Data Date:2	6-Jul-	-24			-	Date	e			Revis	sion			Che	cked	Appr	roved
	Actual Level of	Effort Effort Remaining Work Critical Secondary			Page: 6	of 73				-	19-Aug-24		IFI					—				
																		1		1		

SOLARIS



Activity ID	Activity Name	Rem. Duration	Start	Finish Di	20 2021	2022	2023	2024	2025	2026 20	27 2028	2029	2030	2031
A10051	Incorporate Comments and Issue L and Acquisition Management Plan - IEGR	10d	05-May-27	18-May-27							Incorporate Con	ments and is	sue Land A	Acquisition M
Procurement & Co	ontracting Plan: Final	35d	30-Mar-27	18-May-27										
A12150	Prepare and Issue Procurement & Contracting Plan - IFIR	20d	30-Mar-27	28-Apr-27							Prepare and Issu	e Procureme	nt & Contra	acting Plan - I
A12160	Internal/Peer Review - Procurement & Contracting Plan	5d	28-Apr-27	04-May-27							Internal/Peer Rev	/iew - Procure	ment & Co	ontracting Pla
A12170	Incorporate Comments and Issue Procurement & Contracting Plan - IEGR	10d	05-May-27	18-May-27							Incorporate Con	ments and Is	sue Procur	rement & Cor
Fabrication Plan:	Final	35d	30-Mar-27	18-May-27										
A12180	Prepare and Issue Fabrication Plan - IFIR	20d	30-Mar-27	28-Apr-27							Prepare and Issu	e Fabrication	Plan - IFIR	4
A12190	Internal/Peer Review - Fabrication Plan	5d	28-Apr-27	04-Mav-27							Internal/Peer Rev	/iew - Fabrica	tion Plan	
A12200	Incorporate Comments and Issue Fabrication Plan - IFGR	10d	05-May-27	18-May-27							Incorporate Con	nments and Is	sue Fabrica	ation Plan - I
Materials Manage	ement Plan: Final	35d	30-Mar-27	18-May-27						•				
A12210	Prepare and Issue Materials Management Plan - IFIR	20d	30-Mar-27	28-Apr-27							Prepare and Issu	e Materials M	anagemen	ıt Plan - IFIR
A12220	Internal/Peer Review - Materials Management Plan	5d	28-Apr-27	04-May-27						I I I I	Internal/Peer Rev	/iew - Materia	ls Manager	nent Plan
A12230	Incorporate Comments and Issue Materials Management Plan - IFGR	10d	05-Mav-27	18-May-27					L J 		Incorporate Con	nments and Is	sue Materia	als Manager
Mid Phase 4 Rev	view	20d	19-May-27	16-Jun-27						•				
A10070	Mid Phase 4 Review Period *Timing/Successor TBD*	20d	19-May-27	16-Jun-27							Mid Phase 4 Re	view Period	*Timing/S	uccessor TE
Regasification P		737d	30- Jun-26	21- lun-29						_				
EPC (High Lovel	n	737d	30 Jun 26	21-Jun 20										
Detailed Engineer	ring	274d	20 Jun 26	21-Jun-29	L				L J I					
Phase 1	ing	2740 117d	30-Jun-26	16-Dec-26										
TRP300010	Kick-Off Meeting	0d	30-Jun-26	10 200 20						♦ Kick-Off Me	eting			
TRP300100	Detailed Engineering - Phase 1	116d	30-Jun-26	16-Dec-26						Deta	led Engineering	Phase 1		
Phase 2		153d	22-Dec-26	09-Aug-27							▼			
TRP300200	Detailed Engineering - Phase 2	153d	22-Dec-26	09-Aug-27							Detailed Engli	neering - Pha	se 2	
Procurement / Fab	brication	577d	07-Jul-26	02-Nov-28								7		
Phase 1		290d	07-Jul-26	08-Sep-27										
TRP400100	Procurement/Fabrication - Phase 1	290d	07-Jul-26	08-Sep-27							Procuremen	t/Fabrication	Phase 1	
Phase 2		577d	07-Jul-26	02-Nov-28										Burner
TRP400200	Procurement/Fabrication - Phase 2	577d	07-Jul-26	02-Nov-28								Procureme	nvFabricati	on - Phase A
Construction		428d	09-Aug-27	02-May-29										
TRD500100	Construction Phase 1 (High Lovel)	210d	09-Aug-27	14-Jun-28							Co	nstruction - P	hase 1 (Hio	uh Level)
Phase 2	Construction - Phase T (Fight Level)	2100	26 Jup 28	02 May 20										
TRP500200	Construction - Phase 2 (Install Remaining Vaporizers-High Level)	210d	26-Jun-28	02-May-29								Con	struction - P	hase 2 (Inst
Commissioning		210d	20 0an 20 20 May-28	21_ lun_29										
Phase 1		60d	29-May-28	22-Aug-28										
TRP600100	Commissioning - Phase 1	60d	29-May-28	22-Aug-28								Commissionir	g - Phase 1	1
Phase 2		50d	11-Apr-29	21-Jun-29										
TRP600200	Commissioning - Phase 2	50d	11-Apr-29	21-Jun-29								🗖 Co	mmissionin	g - Phase 2
EPC (Detailed - 0	Contractor)	737d	30-Jun-26	21-Jun-29										
Milestones		737d	30-Jun-26	21-Jun-29										
PM1050	Start EPC Phase	0d	30-Jun-26							♦ Start EPC I	hase			
A1A01070	60% Design Review	5d	09-Feb-27	16-Feb-27						I 60	% Design Reviev	V		
EN1100	Main Module Piping Isometrics IFC	40d	17-Feb-27	14-Apr-27							/lain Module Pipir	ng Isometrics	FC	
A1A01080	90% Design Review (full plant)	5d	15-Apr-27	21-Apr-27						19	90% Design Revi	ew (full plant)		
A1A01090	Start Civil works	0d	07-Sep-27								 Start Civil wo 	rks		
		1			<u></u>	. : : i	<u> </u>	Date	: : 	Revision	: : :	Chec	ked	Approved
	ever of Effort Actual Work Critical Remaining Work			Data Date:26-	Jul-24		19	-Aug-24	IFI					
Actual Level of				Page: 7 of	13			-						

SOLARIS



ty ID	Activity Name	Rem.	Start	Finish	020		2021		2022	2	2023		202	24	20)25
		Duration			Q	QQ	QQ	QQ	Q	QQ	QQC	2Q1	QQ	QQ	QQ	QQ
A1A01140	Module Set 3 at site	Od		06-Jul-28												
A1A01150	Module Set 2 at site	Od		08-Sep-28												
A1A01130	Module Set 1 at site	Od		30-Oct-28										1		
A1A01110	MC - Mechanical Completion	0d		22-Mar-29												
A1A01120	Performance Test Run Completed	Od		15-Jun-29						1		1				
PM1070	Plant in Operation	Od		21-Jun-29												
Engineering		271d	30-Jun-26	04-Aug-27			· · · · · · · · · · · · · · · · · · ·									· · · · · · · · · · · · · · · · · · ·
Systems Engineer		184d	30-Jun-26	30-Mar-27												
A1A01010		10	30-Jun-26	30-Jun-26	-											
A1EF 1100	MSB Analyser, L, I, P	15d	07-Jul-26	27-Jul-26	-											
A1EF1070	MSB other Inline Instruments	20d	07-Jul-26	04-Aug-26												
A1EF1120	P&ID Issue G	80d	07-Jul-26	29-Oct-26			*									
A1EF1130	P&ID Issue H	60d	30-Oct-26	01-Feb-27	-							1				
A1EF1170	Operating Manual and Training, Comm. Planning	80d	30-Nov-26	30-Mar-27												
Equipment Engine	eering	150d	21-Jul-26	02-Mar-27												
A1070	Equipment Engineering	150d	21-Jul-26	02-Mar-27												
Civil & Steel Struc	cture Engineering	225d	16-Jul-26	14-Jun-27												
A1EN1020	Steel Structures detail eng. for Modules (IFC)	130d	16-Jul-26	27-Jan-27	-							1				
A1EN1030	Steel Structures detail eng. Field (IFC)	120d	19-Aug-26	16-Feb-27												
A1EC1010	Civil detail engineering foundations (IFC)	90d	04-Feb-27	14-Jun-27												
Piping Design		193d	14-Aug-26	27-May-27												
A1EL1100	Detailed Piping Design	160d	14-Aug-26	09-Apr-27												
A1EC1040	Equipment & Structural Foundation Loads	45d	16-Dec-26	25-Feb-27								1				
A1EL1030	Final Dimensions of Control Valves	10d	21-Dec-26	11-Jan-27												
A1EL1070	Design Review 60%	5d	09-Feb-27	16-Feb-27												
A1EL1080	On Modules isometrics	40d	17-Feb-27	14-Apr-27												
A1EL1090	Design Review 90%	5d	15-Apr-27	21-Apr-27							· · ·					
A1EL1110	Off Modules isometrics	20d	29-Apr-27	27-May-27												· · ·
Piping Material		163d	13-Oct-26	10-Jun-27												
A1ER1000	Piping Material 1st MTO for Inquiry	10d	13-Oct-26	26-Oct-26												
A1ER1010	Piping Material 2nd MTO	5d	09-Dec-26	15-Dec-26												
A1ER1020	Piping Material 3rd MTO	5d	24-Mar-27	30-Mar-27							· · ·	1				
A1ER1030	Piping Material 4th MTO	5d	20-May-27	27-May-27												
A1ER1040	Piping Material Last MTO	5d	04-Jun-27	10-Jun-27												
Electrical & Instru	umentation Engineering	138d	14-Sep-26	07-Apr-27												
A1EE1100	Electrical Detail Engineering	125d	14-Sep-26	18-Mar-27												
A1EE1010	Inquiry Specification MV Motor	30d	14-Sep-26	26-Oct-26												
A1EJ1030	Inquiry Specification DCS & PESS	35d	14-Sep-26	02-Nov-26							· · · · · · · · · · · · · · · · · · ·					
A1EJ1100	Instrumentation Detail Engineering	125d	01-Oct-26	07-Apr-27								1				
A1EJ1020	Data sheets Critical CV's & Inline devices	20d	01-Oct-26	29-Oct-26							· · ·					
A1EE1020	Inquiry Specification MV Switchgear	20d	27-Oct-26	24-Nov-26												
A1EJ1050	Data sheets other Inline Instruments	20d	30-Oct-26	27-Nov-26												
Construction Engi	ineering	240d	14-Aug-26	04-Aug-27												
		2100	117.09.20											<u> </u>		
Remaining Lev	vel of Effort Actual Work Critical Remaining W	ork		Data Date:2	61	1-24						L	D	ate	$-\!$	
				Dago: 9	s 50 sf 72							19	9-Aug-	-24	IFI	

SOLARIS
CONSULTANTS INC.
2026 2027 2028 2029 2030 2031
2
♦ Module Set 2 at site
♦ Module Set 1 at site
MC - Mechanical Completion
♦ Performance Test Run Com
♦ Plant in Operation
Engineering Kick Off Meeting
MSBAnalyser, L, T, P
MSB other Inline Instruments
P&ID Issue G
P&ID Issue H
Operating Manual and Training, Comm. Planning
Steel Structures detail eng. for Modules (IFC)
Steel Structures detail eng. Field (IFC)
Civil detail engineering foundations (IFC)
v
Equipment & Structural Foundation Loads
Off Modules isometrics
Piping Material 1st MTO for Inquiry
Piping Material 2nd MTO
Piping Material 3rd MTO
Piping Material 4th MTO
I Piping Material Last MTO
Inquiry Specification DCS& PESS
Data sheets Critical CV's & Inline devices
□ Inquiry Specification MV \$witchgear
Data sheets other Inline Instruments
· · · · · · · · · · · · · · · · · · ·

Revision	Checked	Approved



tivit	' ID	Activity Name	Rem. Duration	Start	Finish				2025					2030	
	A1EX1100	Construction Engineering TBC	240d	14-Aug-26	04-Aug-27						Const	ruction Engi	neering TBC		
	Procurement / Fabr	ication	578d	02-Jul-26	30-Oct-28										
	Stationary Equipm	ent	438d	02-Jul-26	11-Apr-28							7			
	Inquiry & Order		138d	02-Jul-26	25-Jan-27				· · · ·		V	· · · · · · · · · · · · · · · · · · ·			
	A1PB1010	Order Vessels	25d	02-Jul-26	06-Aug-26										
	A1PB1020	Order Heat Exchangers	25d	02-Jul-26	06-Aug-26						ler Heat Exchai	igers			
	A1PB1000	Order Air Cooler	25d	02-Jul-26	06-Aug-26						ler Air Cooler				
	A1PB1030	Order LL Columns	25d	02-Jul-26	06-Aug-26						ler LL Columns				
	A1PB1050	Order Package Units	25d	02-Jul-26	06-Aug-26	· · · · · · · · · · · · · · · · · · ·			· · · ·		ler Package Ur	its			
	A1PB1100	Order ColdBox (internal)	3d	09-Jul-26	13-Jul-26					I Ord	er ColdBox (inte	rnal)			
	A1PB1110	Inquiry & Order Columns	58d	17-Sep-26	09-Dec-26						Inquiry & Orde	r Columns			
	A1PB1040	Inquiry & Order other Equipment	70d	08-Oct-26	25-Jan-27						Inquiry & Ord	ler other Eq	uipment		
I	Fabrication & Del	livery	400d	14-Jul-26	28-Feb-28										
	PC1040	Fabr. & Del. Cold Box	400d	14-Jul-26	28-Feb-28							⊢apr. & De	I. Cola Box		
	A1FB1020	Fabr. & Del. Vessels	185d	23-Jul-26	23-Apr-27						■ Fabr. & D	el. Vessels			
	PC1060	Fabr. & Del. of LL Columns	305d	07-Aug-26	29-Oct-27						Fa	or. & Del. of	LL Columns		
	A1FB1000	Fabr. & Del. Air Coolers	308d	07-Aug-26	03-Nov-27							br. & Del. Al	Coolers		
	A1FB1030	Fabr. & Del. Heat Exchangers	180d	07-Aug-26	30-Apr-27						Fabr. & D	el. Heat Exc	hangers		
	A1FB1050	Fabr. & Del. Package Units	285d	07-Aug-26	30-Sep-27						Fab	r. & Del. Pac	kage Units		
	A1FB1010	Fabr. & Del. Columns	190d	10-Dec-26	17-Sep-27						Fab	. & Del. Coli	imns		
	A1FB1040	Fabr. & Del. other Equipment	200d	21-Dec-26	13-Oct-27						Fat	r. & Del. oth	er Equipment		
	Transport		238d	26-Apr-27	11-Apr-28						Transa	▼	Vord		
	A1120	Transport Vessels to Yard	30d	26-Apr-27	07-Jun-27							rt vessels t			
	A1130	Transport Heat Exchangers to Yard	30d	03-May-27	14-Jun-27							IT HEATEX	nangers to ra	ra	
	A1110	Transport Columns to Yard	30d	20-Sep-27	01-Nov-27						Ira	Insport Coll	imns to Yard		
	A1200	Transport Package Units to site	30d	01-Oct-27	15-Nov-27							ansport Pac	kage Units to s	site	
	A1140	Transport other Equipment to Yard	30d	14-Oct-27	25-Nov-27							ansport oth	erEquipment	to Yard	
	A1100	Transport LL Columns to site	30d	01-Nov-27	13-Dec-27							ransport LL	Columns to si	e	
	A1090	Transport Air Coolers to Yard	30d	04-Nov-27	16-Dec-27						T	ransport Air	Coolers to Ya	rd	
	A1080	Transport ColdBox to Site	30d	29-Feb-28	11-Apr-28						I	Transpor	t ColdBox to S	Site	
	Rotating Equipmen	nt	372d	02-Jul-26	06-Jan-28										
	A1DM1000	Order Compressore	15d	02-Jul-26	22-Jul-26						er Compressor				
	A11M1000	Order Pumps	15d	02-Jul-20	22-Jul-20						er Pumps				
	Eabrication & Del		3274	02-Jul-20	17 Nov 27										
	PC1050	Fabr. & Del. Feed Gas Compressor	327d	23-Jul-26	17-Nov-27						Fa	br. & Del. F	eed Gas Com	oressor	
	A1FM1000	Fabr. & Del. Tank Return Gas Compressor	327d	23-Jul-26	17-Nov-27						Fa	br. & Del. Ta	ink Return Ga	s Compre	essor
	A1FM1010	Fabr & Del Pumps	305d	23-Jul-26	15-Oct-27						Fal	r. & Del. Pu	mps		
	PC1070	Fabr. & Del. Regeneration Gas Compressor	262d	23-Jul-26	13-Aug-27						Fabr.	& Del. Rege	neration Gas	Compres	sor
	PC1080	Fabr & Del Cycle Compressor	2624	23-Jul-26	13-Aug-27		1 1 1 1 1 - 1		- L		Fabr.	& Del. Cvcle	Compressor		
	Transport		95d	16-Aug-27	06-Jan-28										
	A1180	Transport Cycle Compressor to site	30d	16-Aug-27	27-Sep-27						🔲 Trai	sport Cycle	Compressor	to site	
	A1190	Transport Regeneration Gas Compressor to yard	20d	16-Aug-27	13-Sep-27						🛛 Tran	sport Rege	neration Gas (Compress	sor to yard
					·							· · · · ·			<u> </u>
	Remaining Leve	el of Effort Actual Work Critical Remaining Wo	rk		Data Date:26	-Jul-24		Date		R	evision		Checked	Ар	proved
	-				Jula Bulo.20			119-Aug-24	IIFI						

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Fage. 9 01 75	1	

SOLARIS



Activity ID)	Activity Name	Rem.	Start	Finish D2	20	2021	2022	2023	2024	2025		2026	2027	202	8	2029	20	30	2031
			Duration				QQQ					QQ	QQQ	Q Q Q					QQC	Q Q Q
	A1170	Transport Pumps to Yard	30d	18-Oct-27	29-Nov-27							¦ 				ort Pu	mps to ra	ra		
	A1150	Transport Feed Gas Compressor to site	30d	18-Nov-27	06-Jan-28	-										sport F	eed Gas	Compre	SOT TO S	SITE
	A1160	Transport Tank Return Gas Compressor to site	30d	18-Nov-27	06-Jan-28	-									Iran	sport la	ink Retur	n Gas Co	mpress	SOF TO SIT
	Piping Material		240d	27-Oct-26	15-Oct-27										ining Mat	vrial 1 et	MTO			
	A1PR1000	Inquiry Piping Material 1st MTO	40d	27-Oct-26	22-Dec-26										iping Mat					
	A1PR1010	Order Piping Material 2nd MTO	10d	23-Dec-26	13-Jan-27											anai zhi				
	A1FR1010	Fabr. & Del. UG Piping Material	60d	14-Jan-27	09-Apr-27											3 Pipin	g iviateria		~	
	A1FR1020	Fabr. & Del. AG Piping Material Off Module	190d	14-Jan-27	15-Oct-27										Habr. &	Del. AC	Piping N	laterial O	rModu	le
	A1FR1000	Fabr. & Del. Main Piping Material On Module	145d	14-Jan-27	11-Aug-27										-abr. & De	el. Main	Piping M	aterial O	1 Modu	ıle
	A1PR1020	Order Piping Material 3rd MTO	5d	31-Mar-27	07-Apr-27									l Orde	r Piping M	laterial	3rd MTO			
	A1PR1030	Order Piping Material 4th MTO	5d	28-May-27	03-Jun-27									I OI	der Piping	Materi	al 4th MT	C		
	A1PR1040	Order Piping Material last MTO	5d	11-Jun-27	17-Jun-27				1 1 1		1 1			ΙO	der Piping) Mater	al last MT	0		
	Structural Steel		100d	17-Feb-27	09-Jul-27															
	A1FN1010	Fabr. & Del. Structural Steel Field	100d	17-Feb-27	09-Jul-27										abr. & Del	. Struct	ural Steel	Field	1	
	Modules / Skids		385d	15-Apr-27	30-Oct-28										D		0			
	A1FN1000	Purchase & Del. Structural Steel for Modules	100d	15-Apr-27	07-Sep-27										Purchase	e & Del.	Structura	I Steel to	Modul	les
	Module Set 1		360d	20-May-27	30-Oct-28										Variahan		an Madul			
	A1FY1150	Workshop Drawings Modules Set 1	45d	20-May-27	23-Jul-27										Vorksnop	Drawin	gs wodu	es Set I		
	A1FY1000	Yard Steel Fabrication of Modules Set 1	110d	19-Jul-27	23-Dec-27										Yaro	Steel	aprication	oriviodu	es Set	
	A1FY1120	Yard Piping Prefabrication of Modules Set 1	120d	31-Aug-27	28-Feb-28										Yar	a Pipin	g Pretabr	cation of	viodule	es Set 1
	A1FY1020	Yard Fabrication of Modules Set 1	225d	29-Sep-27	24-Aug-28											Yaro	l ⊢abricat	ion of M	dules :	Set 1
	A1FY1060	Piping installation Modules Set 1	120d	24-Dec-27	21-Jun-28											Piping	installatio	n Module	s Set 1	1
	A1FY1040	Transport Modules Set 1 to Site	45d	25-Aug-28	30-Oct-28											📫 🎚	ansport N	/lodules	set 1 to	Site
	Module Set 2 - me	edium complexity	310d	11-Jun-27	08-Sep-28									V		-				
	A1FY1180	Workshop Drawings Modules Set 2	45d	11-Jun-27	16-Aug-27										Workshop	Drawi	ngs Modi	iles Set 2		-
	A1FY1010	Yard Steel Prefabrication of Modules Set 2	110d	03-Aug-27	14-Jan-28										 Yard	Steel	retabrica	tion of Ma	dules &	Set 2
	A1FY1140	Yard Piping Fabrication of Modules Set 2	110d	15-Sep-27	28-Feb-28										e Pa r	d Pipin	g Fabrica	tion of Mo	dules S	Set 2
	A1FY1030	Yard Fabrication of Modules Set 2	180d	14-Oct-27	06-Jul-28											Yard	-abricatio	n of Mod	ules Se	±2
	A1FY1070	Piping Installation Modules Set 2	120d	03-Dec-27	31-May-28											Piping	Installatio	n Module	s Set 2	
	A1FY1050	Transport Modules Set 2 to Site	45d	07-Jul-28	08-Sep-28										1	📕 Tra	nsport Mo	odules Se	t 2 to S	Site
	Module Set 3 - Pi	pe Racks	255d	25-Jun-27	06-Jul-28											, ;				
	A1FY1160	Workshop Drawings Modules Set 3	45d	25-Jun-27	30-Aug-27										Worksho	o Draw	ings Modi	ules Set :)	
	A1FY1080	Yard Steel Fabrication of Modules Set 3	110d	17-Aug-27	28-Jan-28										📕 Yaro	Steel	abricatio	n of Mod	ules Se	et 3
	A1FY1130	Yard Piping Prefabrication of Modules Set 3	121d	29-Sep-27	28-Mar-28										Ya Ya	ard Pipi	ng Prefab	rication o	f Modul	les Set 3
	A1FY1100	Yard Fabrication of Modules Set 3	130d	21-Oct-27	02-May-28										\	′ard Fa	brication (of Module	s Set 3	3
	A1FY1170	Piping Installation Modules Set 3	80d	17-Dec-27	18-Apr-28										P	iping In	stallation	Modules	Set 3	
	A1FY1110	Transport Modules Set 3 to Site	45d	03-May-28	06-Jul-28		· · · · ·									Trans	port Mod	ule s Set :	to Site)
	Electrical & Instru	mentation	245d	27-Oct-26	22-Oct-27								-		7					
	A1PE1000	Inquiry & Order MV Motors	40d	27-Oct-26	22-Dec-26									Inquiry 8	Order M	/ Motoi	S			
	A1PJ1000	Inquiry & Order Critical Control Valves & Inline Devices	35d	30-Oct-26	18-Dec-26									Inquiry 8	Order Cr	itical Co	ontrol Valv	es & Inlin	e Devic	ces
	A1PJ1010	Inquiry & Order DCS & ESD	85d	03-Nov-26	11-Mar-27	-								📕 Inquii	y & Order	DCS 8	ESD			
	A1PE1010	Inquiry & Order MV Switchgears	40d	25-Nov-26	27-Jan-27		*						++	Inquiry	& Order N	1V Swit	chgears			+
	A1FJ1000	Fabr. & Del. Critical Control valves & Inline Devices	190d	21-Dec-26	28-Sep-27										Fabr. & I	Del. Cri	ical Contr	olvalves	& Inline	• Device:
	 Remaining Level Actual Level of 	el of Effort Actual Work Critical Remaining Work			Data Date:26- Page: 10 of	Jul-24 73				Date 19-Aug-24	IFI		Re	vision			Chec	ked	Appro	oved

SOLARIS



rity ID	Activity Name	Rem.	Start	Finish	020		2	2021		202	22	2	023		20	24		2025		Ī
		Duration			Q	Q	Q	ג ג	QC	2 Q	QQ	QC	2 Q	Q	QQ	QC	ג Q	QQ	. Q	ົດ
A1FE1000	Fabr. & Del.MV Motors	190d	23-Dec-26	30-Sep-27														, , , , , , , , , , , , , , , , , , ,		
A1FE1010	Fabr. & Del. MV/LV Switchgear Container Bldgs.	150d	28-Jan-27	01-Sep-27										1						
A1FJ1010	Fabr. & Del. DCS & PESS FER Container Bldg.	155d	12-Mar-27	22-Oct-27																
Construction Subc	contracts	63d	28-Sep-26	04-Jan-27				1	1			1	 					1 1 1 1		
A1PX1000	Inquiry & Order Civil Construction Contractor	50d	28-Sep-26	08-Dec-26	-											1 1 1 1 1 1		1 1 1 1		
A1PX1020	Inquiry & Order General Site construction Contractor	50d	28-Sep-26	08-Dec-26	-													· · ·		
A1PX1010	Inquiry & Order Yard fabrication Contractor	50d	28-Sep-26	08-Dec-26							-							1		
A1PN1000	Inquiry & Order Structural Steel (Site)	50d	16-Oct-26	04-Jan-27																_
Regulatory Permits	(Contractor Scope)	228d	23-Nov-26	25-Oct-27										1						
City Of Delta MD		160d	23-Nov-26	16-Jul-27														1 1 1 1		
PGT106382	Propara Building Parmit Application	300 10d	08 Apr 27	19-May-27				1										1 1 1 1 1 1		
DCT106294		24	10 Apr 27	21-Api-27				1								1 1 1 1 1 1 1 1		1 1 1 1		
RG1100304			19-Apt-27	21-Apt-27																
RG1100380		200	22-Apr-27	19-Iviay-27	-		÷													
RG1106388	Receive Approval/Permit - Deita City Building	Ud		19-May-27														, 1 1 1		
Temporary Buildin	ngs Permit - Delta Building/Plumbing Bylaw (No. 6060)	30d	08-Apr-27	19-May-27	-											1 1 1 1 1 1 1 1		1 1 1 1 1 1		
RG1010759		100	10 Apr 27	21-Apt-27			-	1								1 1 1 1 1 1		1 1 1 1		
RG1010702		30	19-Apr-27	21-Apr-27																
RG1616760		200	22-Apr-27	19-May-27																
RG1616761	Receive Approval/Permit - Delta City Building	Ud		19-May-27			į					-								
Plumbing Permit	- Delta Building/Plumbing Bylaw (No. 6060)	40d	20-May-27	16-Jul-27			-											1 1 1 1		
RG1010747		200	20-IVIAy-27	17-Jun-27	_											1 1 1 1 1 1 1 1		1 1 1 1		
RG1616748		30	15-Jun-27	17-Jun-27																
RG1616749	Review Period	20d	18-Jun-27	16-Jul-27																
RGT616750	Receive Approval/Permit - Plumbing	0d		16-Jul-27			÷											, 1 1 1		
Sprinkler Permit	- Delta Building/Sprinkler Bylaw (No. 6060)	40d	20-May-27	16-Jul-27														1 1 1 1		
RG1616751		200	20-May-27	17-Jun-27	_													1 1 1 1 1 1		
RG1616752	Submit Application	30	15-Jun-27	17-Jun-27				!		· · · · · · · · ·			· · · ·			1 1 1 1 1 +		, , , , , , , , , , , , , , , , , , ,		. ,
RG1616753	Review Period	20d	18-Jun-27	16-Jul-27			ł				-							1 1 1 1 1		
RGT616754	Receive Approval/Permit - Sprinkler Permit	0d		16-Jul-27														1		
Occupancy Perm	it - Delta Building/Occupancy Bylaw (No. 6060)	40d	20-May-27	16-Jul-27																
RG1010755		200	20-IVIAy-27	17-Jun-27	_													1 1 1 1		
RG1616756		3d	15-Jun-27	17-Jun-27									- L J							
RGT616757	Review Period	20d	18-Jun-27	16-Jul-27	-		ł				-							1 1 1 1		
RGT616758	Receive Approval/Permit - Occupancy Permit	0d		16-Jul-27			÷													
Highway Use Per	rmit - Delta Bylaw No. 6922	90d	09-Feb-27	17-Jun-27														1 1 1 1		
Necessary Deliv	verables	100 10d	09-Feb-27	23-Feb-27				1								1 1 1 1 1 1 1 1		1 1 1 1 1 1		
Application	Deliverable(s)		09-Feb-27	23-Feb-27										. .		¦ +				
RGT106182	Prepare Highway Lise and Inspection Permits Application	900 10d	09-Feb-27	23-Eeb-27			ł											1		
RGT106184	Submit Application	34	10-Feb-27	23-Feb-27																
RGT106186	Review Period	PU8	21-Feh-27	17_ lun_27				-			-	1						1 1		
PGT106100	Receive Approval/Permit - Dolta City Highway Loo	ג <u>ט</u>	24-1 60-21	17 Jun 27														. I I I		
RG1100100	Receive Approval/Permit - Delta City Highway Ose		02 Nov 06	17-Jun-27	÷					·										
	mit (Commencial/mutstrial) - Della Official Community Plan Bylaws	1400	23-1107-20	17-JUN-27	:	:		1	:			<u> </u>		<u> </u>		<u>: :</u>		<u>:</u>	<u> </u>	_
Remaining Leve	el of Effort 🛛 🗖 Actual Work 🛛 🗖 Critical Remaining Work			Data Date:2	6-Jı	ul-24	4							F)ate				_
Actual Level of	Effort Effort Remaining Work E			Page: 11	of 7	′3									9-Aug	j-24				_
				· · · · · · · ·	••••	-												i		

MANNE S	OLARIS	
	A N A G E M E N T NSULTANTS INC.	
2026 2027 2028	2029 2030 203	31
		Q
Fabr. & Del.M	1V Motors	
Fabr. & Del. N	W/LV Switchgear Container Bldgs	s.
Fabr. & Del.	DCS & PESS FER Container Bld	g.
Inquiry & Order Civil Co	nstruction Contractor	
Inquiry & Order General	Il Site construction Contractor	
Inquiry & Order Yard fal	prication Contractor	
Inquiry & Order Structu	ural Steel (Site)	
▼ →		
₩		
Prepare Building P	ermitApplication	
I Submit Application		
Review Period		
♦ Receive Approval	/Permit - Delta City Building	
Prepare Building P	ermit Application	
I Submit Application		
Review Period		
	/Permit - Delta City Building	
Prepare Plumbin	d Permit Application	
u Submit Applicatio	n	
	al/Permit - Plumbing	
	dir pinia i dinang	
Prepare Sprinkle	r Permit Application	
Submit Applicatio	n	
Review Period		-1
Receive Approv	al/Permit - Sprinkler Permit	
	nov Permit Application	· · · · · · · · · · · · · · · · · · ·
	ncy reimicappication	
	n	1 1 1 1
	al/Permit - Occupancy Permit	
■ ■ Deliverable(s)		
Prepare Highway Us	e and Inspection Permits Applica	tion
I Submit Application		
Review Period		
	Permit - Delta City Highway Use	
Revision	Checked Approved	



y ID	Activity Name	Rem.	Start	Finish	20 2021 20;	22 2023 2024	2025	2026	2027	2028	2029	2030	2031
		Dulation										QQQ	QQ
Necessary Deli	Verables	60d	23-Nov-26	23-Feb-27					Deliverable	3			
RGT106412	Deliverable(S)	600	23-INOV-26	23-FeD-27						"			
Application	Pronara Davalanment Dermit (Commercial/Industrial) Application	<u>100d</u>	26-Jan-27	17-Jun-27					Prenare Dev	elonment Pe	ermit (Comme	ercial/Indus	strial) A
RG1100402		200	20-Jan-27	23-Feb-27						ation			
RG1106484	Submit Application	30	19-Feb-27	23-Feb-27						Daniad			
RGT106486	Review Period	80d	24-Feb-27	17-Jun-27					Review i	Peniod			
RGT106488	Receive Approval/Permit - Delta City Development (Commercial/Industrial)	0d		17-Jun-27					♦ Receive	Approval/Pe	mit - Delta Ci	ty Develop	ment
Development Per	rmit (Streamside Protection & Enhancement) - Bylaw No. 6349	140d	23-Nov-26	17-Jun-27				V					
Necessary Deli	verables	60d	23-Nov-26	23-Feb-27					Deliverable(<u>.</u>			
RG1106512	Deliverable(s)	60d	23-Nov-26	23-Feb-27						2			
Application	Duran are Daviden ment Demait (Stream Dratestian) Application	100d	26-Jan-27	17-Jun-27					Prenare Dev	elonment Pe	ermit (Stream	Protection	a) Annli
RG1106582	Prepare Development Permit (Stream Protection) Application	200	26-Jan-27	23-FeD-27) Abbi
RG1106584	SubmitApplication	3d	19-Feb-27	23-Feb-27						auon			
RGT106586	Review Period	80d	24-Feb-27	17-Jun-27			· · · · · · · · · · · · · · · · · · ·			riod			
RGT106588	Receive Approval/Permit - Delta City Development (Stream Protection)	0d		17-Jun-27					♦ Receive	Approval/Pe	mt - Delta Ci	ty Develop	ment
Demolition Perm	nit - Delta Official Community Plan Bylaws *BasePlant*	75d	02-Feb-27	19-May-27									
Necessary Deli	verables	15d	02-Feb-27	23-Feb-27						A			
RGT106612	Deliverable(s)	15d	02-Feb-27	23-Feb-27						5)			
Application		75d	02-Feb-27	19-May-27				· · · · · · · · · · · · · · · · · · ·		nolition Dorm	vit Application		
RG1106682	Prepare Demolition Permit Application	150	02-Feb-27	23-Feb-27							пдрисацоп		
RGT106684	SubmitApplication	3d	19-Feb-27	23-Feb-27						ation			
RGT106686	Review Period	60d	24-Feb-27	19-May-27					Review P	eriod			
RGT106688	Receive Approval/Permit - Delta City Demolition	0d		19-May-27					♦ Receive A	pproval/Per	nit - Delta Cit	y Demolitio	'n
MOTI		85d	26-Jan-27	27-May-27									
Highway Use Pe	rmit - Transportation Act	85d	26-Jan-27	27-May-27									
Necessary Deli	verables	20d	26-Jan-27	23-Feb-27									
RG1110112	Deliverable(s)	20d	26-Jan-27	23-Feb-27						2)			
Application		85d	26-Jan-27	27-May-27					Bropara High		ormit Applicati	on	
RG1110182	Prepare Highway Use Permit Application	20d	26-Jan-27	23-Feb-27						Iway Use Fe	micApplication		
RGT110184	SubmitApplication	3d	24-Feb-27	26-Feb-27						cauon			
RGT110186	Review Period	62d	01-Mar-27	27-May-27					Review H	eriod			
RGT110188	Receive Approval/Permit - MOTI Highway Use	0d		27-May-27					♦ Receive	Approval/Per	mit - MOTI Hi	ghway Use	e
Nav Canada		138d	08-Apr-27	25-Oct-27					Y				
Boundary Bay Ai	irport Email Notification	15d	08-Apr-27	28-Apr-27									
Email Notification	on	15d	08-Apr-27	28-Apr-27					<u>.</u>				
Application	Dranara and Sand Natification Email	150 Ed	08-Apr-27	28-Apr-27					Prepare ar	nd Send Not	fication Emai		
DOT404400		UC L	15 Arr 07	10-mpi-27					Receive C	onfirmation F	Resnonse - F	oundary F	łav ∆irr
RG1104188	Receive Confirmation Response - Boundary Bay Airport	TUD	15-Apr-27	28-Apr-27							Caponae - D	Suridary D	սյուլ
Land Use Progra	am - Grane	128d	22-Apr-27	25-Oct-27			· · · · · · · · · · · · · · · · · · ·						
RGT111112		50	22- <u>Αρι-27</u> 22-Δpr-27	<u>26-Αρι-27</u> 28-Δρr-27					Deliverable	e(s)			· · ·
Application		100d	22-741-21	20-Api-27									
RGT111182	Prenare Land Lise Crane Permit Application	50	22-Apr-27	28-Apr-27					Prepare L	and Use Cra	ine Permit Ap	plication	
DCT111102		24	22-Apr-27	03 May 27					I Submit An	plication	······································		
			23-Apr-27							iow Period			
RG 1111186	Keview Period	120d	04-May-27	25-Oct-27					Rev				
	Г								Dovision		Charlerd	A	
 Remaining Lev 	/el of Effort Actual Work Critical Remaining Work			Data Date:26	-Jul-24			ŀ	(EVISION		Checked	Appr	oved
				D	6 70	19-Aug-24	+ jir"i						

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data:26 Jul 24	Dale
			Data Date:26-Jui-24	10_Aug_2/
Δctual Level of Effort	Remaining Work	Critical Secondary	Page: 12 of 72	13-Aug-24
			Fage. 12 01 / 5	

SOLARIS

Revision	Checked	Approved



ID	Activity Name	Rem.	Start	Finish D2	0 2021	2022	2023	2024		2025	2026	2027	2028		2029	203	C	2031
-		Duration					Q Q Q Q			Q Q Q Q	QQQ	Q Q Q Q			QQO			
RGT111188	Receive Approval/Permit - Nav Canada (Crane)	0d		25-Oct-27									Receive	Approva	l/Permit	t - Nav Ca	nada (Crane)
Port of Vancouver		25d	22-Apr-27	27-May-27														
Notice to Shipping	g - Canada Marine Act	25d	22-Apr-27	27-May-27														
RGT113112	Peliverable(s)	5d	22-Apr-27	28-Apr-27		· · · · · · · · · · · · · · · · · · ·		J I J I				∎ ∎ Deli	/erable(s)					
Application	Deliverable(3)		22-Apr-27	20-Api-27														
RGT113184	Notice to Shipping Registration	<u></u>	22-Apr-27	27-May-27 28-Apr-27								Not	ce to Shipp	ing Regi	stration			
RGT113188	Receive Info/Confirmation - Notice of Shinning (Van Port)	20d	20_Apr_27	27-May-27								Re	ceive Info/	Confirmat	tion - N	otice of Sl	nippina	(Van, Po
Transport Canada		1200	22 Apr 27	27-Way-27									7					
Aeronautical Clea	arance - Canadian Aviation Regulations (CARs)	1280 128d	22-Apr-27	25-Oct-27								····						
Necessary Delive	verables	5d	22-Apr-27	28-Apr-27								▼						
RGT114112	Deliverable(s)	5d	22-Apr-27	28-Apr-27								I Deli	/erable(s)					
Application		128d	22-Apr-27	25-Oct-27									7					
RGT114182	Prepare Aeronautical Clearance Permit Application	5d	22-Apr-27	28-Apr-27								I Pre	bare Aeron	autical Cl	earanc	e Permit /	Applica	tion
RGT114184	Submit Application	3d	29-Apr-27	03-May-27						· · · · · · · · · · · · · · · · · · ·		I Sub	mit Applica	ion				
RGT114186	Review Period	120d	04-May-27	25-Oct-27									Review	Period				
RGT114188	Receive Approval/Permit - Transport Canada (Aeronautical)	0d		25-Oct-27									Receive	Approva	l/Permit	t - Transpo	ort Can	ada (Ae
TSBC - Safety Stan	ndards Act	53d	31-Mar-27	15-Jun-27														
Boiler and/or Pres	ssure Vessel Registration / Approval	33d	31-Mar-27	17-May-27								₩						
Necessary Delive	verables	20d	31-Mar-27	28-Apr-27								W						
RGT115112	Deliverable(s)	20d	31-Mar-27	28-Apr-27								Deli	verable(s)					
Application		33d	31-Mar-27	17-May-27									_ .					
RGT115182	Prepare Boiler and/or Pressure Vessel Registration Application	20d	31-Mar-27	28-Apr-27								Pre	bare Boller	and/or P	ressure	e vessel h	egistra	ition App
RGT115184	Submit Application	3d	29-Apr-27	03-May-27								I Sub	mit Applica	ion				
RGT115186	Review Period	10d	04-May-27	17-May-27								I Re	view Perioo				1	
RGT115188	Receive Approval/Permit - Boiler/Pressure Vessel Reg.	Od		17-May-27								♦ Re	ceive Appro	val/Pern	nit - Boil	er/Pressu	re Ves	sel Reg.
Pressure Piping R	Registration / Approval	33d	31-Mar-27	17-May-27								₩						
Necessary Delive	rerables	20d	31-Mar-27	28-Apr-27								₩ .						
RGT115212	Deliverable(s)	20d	31-Mar-27	28-Apr-27								Deli	/erable(s)					
Application		33d	31-Mar-27	17-May-27												4		
RGT115282	Prepare Pressure Piping Registration Application	20d	31-Mar-27	28-Apr-27									bare Press	ILE HIDIN	g Regis	stration Ap	plicatio	'n
RGT115284	Submit Application	3d	29-Apr-27	03-May-27								I Sub	mit Applica	ion				
RGT115286	Review Period	10d	04-May-27	17-May-27								🛯 Re	view Perioo					
RGT115288	Receive Approval/Permit - Pressure Piping Reg.	0d		17-May-27								♦ Re	ceive Appro	val/Pern	nit - Pre	ssure Pip	ng Reg	g.
Installation permit	ts - Electrical System	33d	31-Mar-27	17-May-27														
Necessary Delive	/erables	20d	31-Mar-27	28-Apr-27														
RGT115312	Deliverable(s)	20d	31-Mar-27	28-Apr-27									/erable(s)					
Application		33d	31-Mar-27	17-May-27														
RGT115382	Prepare Electrical System Installation Permit Application	20d	31-Mar-27	28-Apr-27									Jaie ⊟iectr	aı Syste	in Insta	uiauon Pe	ппсАр	plication
RGT115384	Submit Application	3d	29-Apr-27	03-May-27								I Sub	mit Applica	ion				
RGT115386	Review Period	10d	04-May-27	17-May-27								I Re	view Period					
RGT115388	Receive Approval/Permit - Installation Permit (EL System)	0d		17-May-27								♦ Re	ceive Appro	val/Pern	nit - Inst	allation Pe	ermit (E	L Syster
Installation permit	ts - Refrigeration System	33d	31-Mar-27	17-May-27								T						
Necessary Delive	rerables	20d	31-Mar-27	28-Apr-27								T						
RGT115412	Deliverable(s)	20d	31-Mar-27	28-Apr-27								Deli	verable(s)					
Application		33d	31-Mar-27	17-May-27														
Remaining Leve	el of Effort - Actual Work - Critical Remaining Work			Data Dato:26	lul-24			Dat	te			Revision			Check	ked	Appro	oved
Actual Level of E	Effort Effort Remaining Work Critical Secondary			Page: 13 of	73			19-Aug-2	.4	IFI								
				. ago: 10 01														

SOLARIS



Activit	y ID	Activity Name	Rem.	Start	Finish)2	.0 2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031
			Duration					Q Q Q Q								
	RGT115482	Prepare Refrigeration Installation Permit Application	20d	31-Mar-27	28-Apr-27							Prepare	e Refrigera	ation Installatio	on Permit App	olication
	RGT115484	Submit Application	3d	29-Apr-27	03-May-27							I Submit	Application	۱		
	RGT115486	Review Period	10d	04-May-27	17-May-27							Review	v Period			
	RGT115488	Receive Approval/Permit - Installation Permit (Refrigeration System)	Od		17-May-27						• • • • • • • • • • • • • • • • • • • •	♦ Receiv	e Approva	I/Permit - Inst	allation Perm	iit (Refrigera
	Installation permit	ts - Gas	53d	31-Mar-27	15-Jun-27											
	Necessary Deliv	erables	20d	29-Apr-27	27-May-27							\mathbf{T}				
	RGT115512	Deliverable(s)	20d	29-Apr-27	27-May-27							Delive	rable(s)			
	Application		53d	31-Mar-27	15-Jun-27							•••				
	RGT115582	Prepare Gas Installation Permit Application	20d	31-Mar-27	28-Apr-27							Prepare	e Gas Inst	allation Permi	Application	
	RGT115584	Submit Application	3d	28-May-27	01-Jun-27							Submi	t Applicatic	'n		
	RGT115586	Review Period	10d	02-Jun-27	15-Jun-27							🛿 Revie	w Period			
	RGT115588	Receive Approval/Permit - Installation Permit (Gas)	0d		15-Jun-27							♦ Recei	ve Approv	al/Permit - Ins	tallation Pern	nit (Gas)
	Operation permits	- Electrical System	33d	31-Mar-27	17-May-27											
	Necessary Delive	erables	20d	31-Mar-27	28-Apr-27							₩				
	RGT115612	Deliverable(s)	20d	31-Mar-27	28-Apr-27							Delivera	able(s)			
	Application		33d	31-Mar-27	17-May-27							••				
	RGT115682	Prepare Electrical System Operation Permit Application	20d	31-Mar-27	28-Apr-27							Prepare	e Electrical	System Ope	ation Permit	Application
	RGT115684	Submit Application	3d	29-Apr-27	03-May-27							I Submit	Application	ו		
	RGT115686	Review Period	10d	04-May-27	17-May-27							Review	/ Period			
	RGT115688	Receive Approval/Permit - Operation Permit (EL System)	0d		17-May-27							♦ Receiv	e Approva	I/Permit - Ope	eration Permi	it (EL Syster
	Operation permits	- Refrigeration System	33d	31-Mar-27	17-May-27											
	Necessary Delive	erables	20d	31-Mar-27	28-Apr-27							₩				
	RGT115712	Deliverable(s)	20d	31-Mar-27	28-Apr-27							Delivera	able(s)			
	Application		33d	31-Mar-27	17-May-27							₩.	_			
	RGT115782	Prepare Electrical System Operation Permit Application	20d	31-Mar-27	28-Apr-27							Prepare	Electrical	System Ope	ation Permit	Application
	RGT115784	Submit Application	3d	29-Apr-27	03-May-27							I Submit	Application	ו א		
	RGT115786	Review Period	10d	04-May-27	17-May-27							Reviev	/ Period			
	RGT115788	Receive Approval/Permit - Operation Permit (Refrigeratioin System)	Od		17-May-27							♦ Receiv	e Approva	ıl/Permit - Ope	eration Permi	it (Refrigerat
	Operation permits	(Boiler, Pressure Vessels)	33d	31-Mar-27	17-May-27							••				
	Necessary Delive	erables	20d	31-Mar-27	28-Apr-27							•				
	RGT115812	Deliverable(s)	20d	31-Mar-27	28-Apr-27							Delivera	able(s)			
	Application		33d	31-Mar-27	17-May-27								. D.a ila 0. I			
	RGT115882	Prepare Boiler & Pressure Vessels Operation Permit Application	20d	31-Mar-27	28-Apr-27								e Boller & I	ressure ves	seis Operatio	in Permit Ap
	RGT115884	Submit Application	3d	29-Apr-27	03-May-27							I Submit	Application	וו		
	RGT115886	Review Period	10d	04-May-27	17-May-27							Review	/ Period			
	RGT115888	Receive Approval/Permit - Operation Permit (Boiler/Pressure Vessel)	0d		17-May-27							♦ Receiv	e Approva	I/Permit - Ope	eration Permi	it (Boiler/Pre
	BC One Call		8d	26-Jul-27	06-Aug-27							▼				
	BC One Call Regi	stration	8d	26-Jul-27	06-Aug-27											
	Necessary Delive		5d	26-Jul-27	03-Aug-27								(orable(s)			
	RG1101112	Deliverable(s)	50	26-Jul-27	03-Aug-27							I Dei				
	Application	PC One Call Degistration	40	30-Jul-27	06-Aug-27								One Call F	edistration		
			10	30-JU-Z/	03-Aug-27							I Bon		onfirmation		
	RG1101188	Receive Into/Contirmation - BC one Call	3d	03-Aug-27	06-Aug-27											
	Work Safe BC * T	BC Might Be Part of Construction Contractor Scope:*	85d	22-Apr-27	23-Aug-27											
	Guidennes 20.3-2	Quanneu Coorumators	810	28-Apr-27	23-Aug-27											
	Remaining Leve	el of Effort Actual Work Critical Remaining Work			Data Data:36				Date			Revision		Check	ed A	pproved
	Actual Level of F	Effort Remaining Work Critical Secondary			Page 14 of	73			19-Aug-24	IFI						
		 •			. ugo. 14 01	. •										

SOLARIS



Units Units <th< th=""><th>ctivity</th><th>ID</th><th>Activity Name</th><th>Rem.</th><th>Start</th><th>Finish</th><th>20 202</th><th>1 202</th><th>2 2023</th><th>2024</th><th>2025</th><th>2026</th><th>2027</th><th>2028</th><th>2029</th><th>203</th><th>30 2031</th></th<>	ctivity	ID	Activity Name	Rem.	Start	Finish	20 202	1 202	2 2023	2024	2025	2026	2027	2028	2029	203	30 2031
No. 2007 (Production Control (Production) Control (Duration											QQQ	QQQ	
Bit (19912) Disk of Contract (2000) Disk of Contract (2000) Disk of Contract (2000) ROT (19912) Disk of Contract (2000) Disk of Contract (2		Necessary Delive	erables	5d	09-Aug-27	16-Aug-27							V	alli (arabla (a)			
Control Proper Approx App		RGT116112	Deliverable(s)	5d	09-Aug-27	16-Aug-27								eliverable(s)			
Implicit register product of udder # 2.0 and a mining of the construction of the c		Application	Designed Ann list time of Outidality of OO 2, O Out life of an and in store	81d	28-Apr-27	23-Aug-27							Prena	re Application	of Guide	lines 20 3	2 Qualified cool
No.1119/35 Scient/Appendix 23 10 Product 2 Product 2 </td <td></td> <td>RG1116182</td> <td>Prepare Application of Guidelines 20.3-2 Qualified coordinators</td> <td>10</td> <td>28-Apr-27</td> <td>28-Apr-27</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>III ICS 20.0-</td> <td></td>		RG1116182	Prepare Application of Guidelines 20.3-2 Qualified coordinators	10	28-Apr-27	28-Apr-27										III ICS 20.0-	
BC1116105 Peaker Montal 23 19.4/2/27 2.4/2/27 CC1116105 Peaker Montal Permit VSEC Quality Constructors 01 2.4/2/27 CC1116105 Peaker Montal Permit VSEC Quality Constructors 01 2.4/2/27 CC1116105 Decker Approximation 01 2.4/2/27 R01116105 Figure Value Approximation 01 2.4/2/27 R01116105 Figure Approximation 01 2.4/2/27 R01116105 Figure Value Approximation 01 2.4/2/27 R01116105		RG1116184	Submit Application	3d	16-Aug-27	19-Aug-27								unii Applicat			
Bit 11 (168) Recent Approximation Control 2.2.4/a,0.27 Bit 11 (168) Recent Approximation Control 2.2.4/a,0.27 Bit 11 (168) Recent Approximation Control Control Bit 11 (168) Recent Approximation Control Control Devection Bit 11 (168) Recent Approximation Control Control Devec		RGT116186	Review Period	2d	19-Aug-27	23-Aug-27							IR	eview Period			re i r
Col: Col: <t< td=""><td></td><td>RGT116188</td><td>Receive Approval/Permit - WSBC Qualified coordinators</td><td>0d</td><td></td><td>23-Aug-27</td><td></td><td></td><td></td><td></td><td></td><td></td><td>♦R</td><td>eceive Appro</td><td>val/Permit</td><td>- WSBC C</td><td>ualified coordin</td></t<>		RGT116188	Receive Approval/Permit - WSBC Qualified coordinators	0d		23-Aug-27							♦R	eceive Appro	val/Permit	- WSBC C	ualified coordin
Process District Disk Disk <thdisk< th=""> Disk Disk<td></td><td>G20.2(1)/20.2.1(1)</td><td>Notice of Project, Section 20.2(1) of the OHS Regulation</td><td>10d</td><td>22-Apr-27</td><td>05-May-27</td><td>· · · · · · · · · · · · · · · · · · ·</td><td></td><td></td><td>· · · · · · · · · · · · · · · · · · ·</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></thdisk<>		G20.2(1)/20.2.1(1)	Notice of Project, Section 20.2(1) of the OHS Regulation	10d	22-Apr-27	05-May-27	· · · · · · · · · · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·							
Notice Labelenities 100 200/200 200/200 200/200 RGT11028 Depreser Work Sele DC Appricaton - Notice of Project 10 200/200 100 <td></td> <td>Necessary Delive</td> <td></td> <td><u>50</u></td> <td>22-Apr-27</td> <td>28-Apr-27</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>∎ Delive</td> <td>rable(s)</td> <td></td> <td></td> <td></td>		Necessary Delive		<u>50</u>	22-Apr-27	28-Apr-27							∎ Delive	rable(s)			
Bit Integra Program Work Sale BC Application - Notes of Project 9 State Project Project Work Sale BC Application - Notes of Project Rest Int Sale Project Notes and Project 9 29/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale Project Notes and Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale Project Notes and Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale Project Notes Sale BC Application - Notes of Project 9 00/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 00/4/07/2 05/4/07/2 05/4/07/2 Rest Int Sale DE Application - Notes of Project 9 06/4/07/2 05/4/07/2 06/4/07/2		Application		5d	22-Api-27	26-Apt-27							v				
Bit III 1028 Source Argeneodies Stat Psychop 7 Oblight 20 RGT III 2028 Persone Argeneodie Persone 0.00		RGT116282	Prepare Work Safe BC Application - Notice of Project	1d	28-Apr-27	28-Apr-27							Prepa	re Work Safe	BC Applic	ation - No	tice of Project
Communic Col C		RGT116284	Submit Application	3d	20, pr 27	03-May-27							I Subm	it Application			
Product		PCT116286	Boviow Poriod	24	04 May 27	05 May 27							I Revie	w Period			
No.1002 <		DCT116200	Receive Approval/Permit W/SPC Natio of Project	2u	04-1viay-21	05-May-27							▲ Rece	ve Approval/F	ermit - W	SBC Notic	of Project
Normality Normality <t< td=""><td></td><td>20M22 Dormit O</td><td>Receive Approval/Fermin - WSBC Notic of Froject</td><td>00</td><td>00 Amr 07</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-</td><td></td><td></td><td>02011010</td><td>01110j001</td></t<>		20M22 Dormit O	Receive Approval/Fermin - WSBC Notic of Froject	00	00 Amr 07								-			02011010	01110j001
Bit111532 Delevable(s) 54 22-Apr-27 28-Apr-27 Additional Bit111532 Prepare Work Safe BC Application - 30M33 Permit 14 28-Apr-27 28-Apr-27 15-Apr-27 15-A		Necessary Delive	arables	5d	22-Apr-27	28-Apr-27							V				
Adjustion Both		RGT116312	Deliverable(s)	5d	22-Apr-27	28-Apr-27							Delive	rable(s)			
In CT116382 Prepare Work Safe BC Application - 30M33 Permit 1d 28A-m;27 03A-m;27 03A-m;27 </td <td></td> <td>Application</td> <td></td> <td>6d</td> <td>28-Apr-27</td> <td>05-Mav-27</td> <td></td>		Application		6d	28-Apr-27	05-Mav-27											
RCT116384 Submit Application 3d 28-Apr.27 03-May.27 05-May.27 RCT116386 Receive Approval/Permt - WSBC 30M33 0d 0 - App.27 05-May.27 05-May.27 05-May.27 Receive Approval/Permt - WSBC 30M33 Receive Approval/Permt - WSBC 30M34		RGT116382	Prepare Work Safe BC Application - 30M33 Permit	1d	28-Apr-27	28-Apr-27							Prepa	re Work Safe	BC Applic	ation - 301	VI33 Permit
RCT116388 Review Pend 2d 04-May-27 05-May-27 RGT116388 Receive ApprovalPermit - WSBC 30M33 0d 05-May-27 05-May-27 Centration 200 05-May-27 16-Jun-28 16-Jun-28 16-Jun-28 A1X2 C200 Site Mobilisation (Regas. Linde Detailed) 201 07-App.27 16-Jun-28 A1X2 C200 Site Mobilisation (Regas. Linde Detailed) 201 07-App.27 16-Jun-28 A1X2 C200 Site Mobilisation (Regas. Linde Detailed) 204 07-App.27 16-Jun-28 A1X2 C200 General Foundation Works, Tranchus & Roads 192 07-App.27 16-Jun-28 A1XY 1080 Foundations Set 1 66d 92-Bw-27 20-May-28 A1XY 1080 Foundations Set 2 66d 92-Bw-27 20-May-28 Endoment 60d 14-Abv-27 16-Jun-28 16-Jun-28 A1XM1060 Erection Package Units 60d 14-Abv-27 16-Jun-28 A1XM1040 Erection Fack Burg Max 20 16-Jun-28 20-Jun-28 10-Jun-28 A1XM1040		RGT116384	SubmitApplication	3d	29-Apr-27	03-May-27							I Subm	it Application			
RGT116388 Receive ApprovalPermt - WSBC 30M33 Od 0 5 Map-27 Centituction 400d 09-Aug-27 22 Max-29 Chill Works 2120 09-Aug-27 07-Sep-27 Chill Works Technologies 1200 09-Aug-27 07-Sep-27 AlXC2000 Site Mobilization (Rogas: Lindo Datalido) 200 09-Aug-27 07-Sep-27 AlXC2000 Foundations Set 3 130-00 03-Sep-27 16-Jun-28 Site Mobilization (Rogas: Lindo Datalido) AlXV1000 Foundations Set 1 65d 09-Dec-27 15-Jun-28 Foundations Set 1 AlXV1000 Foundations Set 2 65d 09-Dec-27 18-Jan-28 Foundations Set 1 AlXM1040 Erection Package Units 66d 14-Jan-28 02-Jul-28 Foundations Set 2 Foundations Set 3 AlXM1040 Erection Fackage Units 66d 14-Jan-28 02-Jul-28 Foundation Set 3 Foundation Set 3 AlXM1020 Erection Fackage Units 130-dt 14-Jan-28 03-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map-28 04-Map		RGT116386	Review Period	2d	04-May-27	05-May-27							Revie	w Period			
Optimizion Optimiz		RGT116388	Receive Approval/Permit - WSBC 30M33	0d		05-May-27							♦ Rece	ve Approval/F	Permit - W	SB¢ 30M3	3
Ori Works 212d 09-Aug.27 16-Jun-28 A1XC2000 Site Mobilisation (Regas: Linde Detailed) 20d 09-Aug.27 16-Jun-28 A1XC2020 General Foundation Works, Trenches & Roads 192d 07-Sep.27 16-Jun-28 A1XY1060 Foundations Set 1 06d 09-Doc.27 15-Jun-28 A1XY1060 Foundations Set 1 06d 09-Doc.27 20-Mup28 A1XY1070 Foundations Set 1 06d 20-Jun-28 Poundations Set 1 A1XY1070 Foundations Set 1 06d 16-Nov.27 20-Jun-28 A1XY1070 Foundations Set 1 06d 16-Nov.27 20-Jun-28 A1XM1060 Erection Package Units 04d 16-Nov.27 16-Jun-28 A1XM1060 Erection Fack Gage Units 04d 14-Dec.27 15-Mar-28 A1XM1020 Erection Fack Gage Units 04d 14-Dec.28 09-Map-28 A1XM1020 Erection Fack Gage Compressor 65d 04-Fe-28 09-Map-28 A1XM1020 Erection Fack Gage Compressor Erection Fack Gage		Construction		400d	09-Aug-27	22-Mar-29											
A1XC2000 Site Mobilisation (Regas Linde Detailed) 20d 09-Aug-27 07-Sep-27 image: state in the integrate Linde Detailed) A1XC2020 General Foundation Works, Trenches & Roads 192d 07-Sep-27 image: state integrate Linde Detailed) image: state integrate Linde Detailed) A1XY1000 Foundations Set 1 100d 03-Dec-27 20-Mar-28 image: state integrate Linde Detailed) image: state integrate Linde Detailed) image: state integrate Linde Detailed) A1XY1000 Foundations Set 2 65d 28-Jan-28 20-Mar-28 image: state integrate Linde Detailed) image: state integ		Civil Works		212d	09-Aug-27	16-Jun-28											
A 1XC2202 General Foundation Works, Tenches & Roads 192d 07.5ep.27 16.Jun-28 A 1XY1080 Foundations Set 1 130d 03.0ep.27 15.Jun-28 Foundations Set 3 Foundations Set 1 A 1XY1080 Foundations Set 1 66d 02.8un-28 02.4wn-28 Foundations Set 1 Foundations Set 1 A 1XY1070 Foundations Set 2 66d 28.Jan-28 02.4wn-28 02.4wn-28 Foundations Set 1 A 1XY1080 Enction Package Units 40d 16.Nov-27 18.Jan-28 02.4wn-28 Foundation Field Equipment A 1XM1040 Election of LL Columns 60d 14-Dec-27 16.4wn-28 0.94Ma-28 A 1XM1020 Encetion Field Equipment 100d 14-Dec-27 16.4wn-28 0.94Ma-28 A 1XM1020 Election Tank Return Gas Compressor 66d 04-Feb-28 0.94Ma-28 0.94Ma-28 A 1XM1030 Election Tank Return Gas Compressor 100d 04-Feb-28 0.94Ma-28 0.94Ma-28 0.94Ma-28 A 1XM1030 Election Tank Return Gas Compressor 100d 0.94Ma-28 0.94Ma-28 0.94Ma-28 0.94Ma-28 0.94Ma-28 0.94Ma-28		A1XC2000	Site Mobilisation (Regas. Linde Detailed)	20d	09-Aug-27	07-Sep-27							∎ S	ite Mobilisatio	h (Regas.	Linde Det	ailed)
A1XY1080 Foundations Set 3 130d 03-De-27 15-Jun-28 1<		A1XC2020	General Foundation Works, Trenches & Roads	192d	07-Sep-27	16-Jun-28								Gen	eral Foun	dation Wo	rks, Trenches &
A1XY1060 Foundations Set 1 65d 09-Dec-27 20-Mar-28 A1XY1070 Foundations Set 1 65d 28-Jan-28 02-Mar-28 A1XM1060 Erection Package Units 40d 16-Nov-27 18-Jan-28 A1XM1060 Erection of LL Columns 60d 14-Dec-27 15-Mar-28 A1XM1020 Erection ank Return Gas Compressor 65d 04-Feb-28 09-May-28 A1XM1020 Erection Feed Gas Compressor 65d 04-Feb-28 09-May-28 A1XM1020 Erection Compressor 65d 04-Feb-28 09-May-28 A1XM1020 Erection Feed Gas Compressor 65d 04-Feb-28 04-Apr-28 A1XM1020 Erection Feed Gas Compressor 100d 04-Feb-28 03-Jan-29 A1XM1020 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XM1020 Installation Modules Set 3 10d 07-Jul-28 10-Jan-29 A1XM1020 Installation Modules Set 3 10d 07-Jul-28 10-Jan-29 A1XM1020 Installation Modules Set 3 10d 07-Jul-28 10-Jan-29 A1XY1040 <		A1XY1080	Foundations Set 3	130d	03-Dec-27	15-Jun-28								Four	ndations S	et 3	
A1XY1070 Foundations Set 2 65d 28-Jan-28 02-May-28 Fujiment 168d 16-Nov-27 20-Jul-28 A1XM1060 Erection Package Units 40d 16-Nov-27 18-Jan-28 A1XM1040 Erection of LL Columns 60d 14-Dec-27 15-Mar-28 A1XM1020 Erection Tank Return Gas Compressor 60d 14-Jan-28 02-Jul-28 A1XM1020 Erection Tank Return Gas Compressor 60d 04-Feb-28 09-May-28 A1XM1020 Erection Tank Return Gas Compressor 40d 04-Feb-28 09-Jul-28 A1XM1030 Erection Compressors 100d 04-Feb-28 09-Jul-28 A1XM1030 Installation Steel Structures Field 20-Mar-28 10-Jan-29 A1XM1030 Installation Modules Set 3 100d 07-Jul-28 03-Jan-29 A1XM1040 Installation Modules Set 3 100d 07-Jul-28 03-Jan-29 A1XY1100 Modules Set 3 100d 07-Jul-28 03-Jan-29 A1XY1100 Installation Modules Set 3 100d 07-Jul-28 03-Jan-29 A1XY1100 Installation Modules Set 3 10		A1XY1060	Foundations Set 1	65d	09-Dec-27	20-Mar-28								🛑 Founda	tions Set	1	
Equipment 168d 16-Nov-27 20-Jul-28 A XM1060 Erection Package Units 40d 16-Nov-27 18-Jan-28 A XM1040 Erection of LL Columns 60d 14-Dec-27 15-Mar-28 A XM1020 Erection of LL Columns 10d 14-Jan-28 20-Jul-28 A XM1020 Erection Tank Return Gas Compressor 65d 04-Feb-28 09-May-28 A XM1050 Erection Compressors 40d 04-Feb-28 04-Apr-28 A XM1050 Erection Compressors 40d 04-Feb-28 04-Apr-28 A XM1050 Erection Compressors 10d 04-Feb-28 10-Jan-29 A XM1050 Erection Compressors 10d 04-Feb-28 10-Jan-29 A XM1050 Installation Steel Structures Field 200d 20-Jul-28 10-Jan-29 A XM1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A XM1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A XM1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A XM1100 Modules Set 3 10d <t< td=""><td></td><td>A1XY1070</td><td>Foundations Set 2</td><td>65d</td><td>28-Jan-28</td><td>02-May-28</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>📕 Found</td><td>lations Se</td><td>t 2</td><td></td></t<>		A1XY1070	Foundations Set 2	65d	28-Jan-28	02-May-28								📕 Found	lations Se	t 2	
A1XM1060 Erection Package Units 40d 16-Nov-27 18-Jan-28 A1XM1040 Erection Package Units 60d 14-Dec-27 15-Mar-28 A1XM1020 Erection fack acge Units Erection of LL Columms Installation Field Equipment A1XM1020 Erection Tank Return Gas Compressor 65d 04-Feb-28 09-May-28 A1XM1020 Erection Feed Gas Compressor 40d 04-Feb-28 09-May-28 A1XM1020 Erection Compressors 100d 04-Feb-28 09-May-28 A1XM1020 Erection Compressors 100d 04-Feb-28 09-May-28 Structural Steel 200d 20-Mar-28 10-Jan-29 A1XM1020 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XY1040 Installation Modules Set 3 10d 07-Jul-28 31-Jau-28 A1XY1010 Modules Set 3 hook-up 30d 21-Jul-28 31-Jau-28 A1XY1101 Modules Set 2 hook-up 30d 22-Sep-28 06-Nov-28 A1XY1101 Installation Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 A1XY1101 Installati		Equipment		168d	16-Nov-27	20-Jul-28							▼				
A1XM1040 Erection of LL Columns 60d 14-Dec-27 15-Mar-28 1 <		A1XM1060	Erection Package Units	40d	16-Nov-27	18-Jan-28								Erection F	Package U	nits	
A1XB1000 Installation Field Equipment 130d 14-Jan-28 20-Jul-28 09-May-28 A1XM1020 Erection Tank Return Gas Compressor 65d 04-Feb-28 09-May-28 09-May-28 04-Apr-28 09-May-28 A1XM1030 Erection Compressor 40d 04-Feb-28 09-May-28		A1XM1040	Erection of LL Columns	60d	14-Dec-27	15-Mar-28								Erection	of LL Col	umns	
A1XM1020 Erection Tank Return Gas Compressor 65d 04-Feb-28 09-May-28 A1XM1030 Erection Feed Gas Compressor 40d 04-Feb-28 04-Apr-28 A1XM1050 Erection Compressors 100d 04-Feb-28 28-Jun-28 A1XM1050 Irection Compressors 100d 04-Feb-28 28-Jun-28 A1XM1050 Irection Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XN1000 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XY1000 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY1100 Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY1100 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1101 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1101 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1101 Installation Modules Set 2 Modules Set 2 Modules Set 2 A1XY1101 Installation Modules Set 2 06-Nov-28 06-Nov-28 A1XY1101 Installating Wo		A1XB1000	Installation Field Equipment	130d	14-Jan-28	20-Jul-28	r			q - - - - - - - - - -				Inst	allation Fi	eld Equipr	nent
A1XM1030 Erection Feed Gas Compressors A1XM1050 Erection Compressors A1XM1050 Erection Compressors A1XM1050 Erection Compressors A1XM1050 Installation Steel Structures Field A1XM1050 Installation Modules Set 3 A1XM1050 Installation Modules Set 3 A1XY100 Modules Set 3 hook-up A1XY1010 Installation Modules Set 2 A1XY1010 Installation Modules Set 2 Modules Set 3 Modules		A1XM1020	Erection Tank Return Gas Compressor	65d	04-Feb-28	09-May-28								Erecti	on Tank R	eturn Gas	Compressor
A1XM1050 Erection Compressors 100d 04-Feb-28 28-Jun-28 Structural Steel 200d 20-Mar-28 10-Jan-29 A1XN1000 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XN1000 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 A1XN1040 Installation Modules Set 3 10d 07-Jul-28 03-Jan-29 A1XY1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY100 Installation Modules Set 3 30d 21-Jul-28 31-Aug-28 A1XY1010 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1010 Installation Modules Set 2 hook-up 10d 11-Sep-28 22-Sep-28 A1XY1101 Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 A1XY1101 Modules Set 2 hook-up 10d 11-Sep-28 22-Sep-28 A1XY1101 Modules Set 2 hook-up 10d 11-Sep-28 26-Sep-28 06-Nov-28 Imstallation Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 104 19-Aug-24 19-Au		A1XM1030	Erection Feed Gas Compressor	40d	04-Feb-28	04-Apr-28								Erection	Feed Ga	as Compre	essor
Structural Steel 200d 20-Mar-28 10-Jan-29 A1XN1000 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 Modules / Skids 120d 07-Jul-28 03-Jan-29 A1XY1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY1000 Modules Set 3 hook-up 30d 21-Jul-28 31-Aug-28 A1XY1010 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1100 Modules Set 2 hook-up 30d 25-Sep-28 60-Nov-28 Modules Set 2 hook-up 30d 25-Sep-28 60-Nov-28 Modules Set 2 hook-up Critical Remaining Work Critical Remaining Work Critical Secondary Page: 15 of 73 19-Aug-24 IFI IFI IFI		A1XM1050	Erection Compressors	100d	04-Feb-28	28-Jun-28								Erec	tion Com	oressors	
A1XN1000 Installation Steel Structures Field 200d 20-Mar-28 10-Jan-29 Modules / Skids 120d 07-Jul-28 03-Jan-29 A1XY1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY1100 Modules Set 3 hook-up 30d 21-Jul-28 31-Aug-28 A1XY1100 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1100 Modules Set 2 hook-up 10d 11-Sep-28 22-Sep-28 A1XY1100 Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 A1XY110 Modules Set 2 hook-up Modules Set 2 hook-up Modules Set 2 hook-up K Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 Modules Set 2 hook-up Modules Set 2 hook-up Critical Remaining Work Critical Remaining Work Modules Set 2 hook-up Actual Level of Effort Remaining Work Critical Secondary Critical Secondary		Structural Steel	· ·	200d	20-Mar-28	10-Jan-29									v		
Modules / Skids 120d 07-Jul-28 03-Jan-29 A1XY1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 A1XY100 Modules Set 3 hook-up 1 Installation Modules Set 3 Installation Modules Set 3 A1XY1010 Installation Modules Set 2 30d 21-Jul-28 31-Aug-28 A1XY1010 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 A1XY1100 Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 A1XY1100 Modules Set 2 hook-up 30d 25-Sep-28 06-Nov-28 Remaining Level of Effort Actual Work Critical Remaining Work Critical Remaining Work Critical Remaining Work Actual Level of Effort Remaining Work Critical Remaining Work Critical Secondary Critical Secondary		A1XN1000	Installation Steel Structures Field	200d	20-Mar-28	10-Jan-29									Installa	tion Steel S	Structures Field
A1XY1040 Installation Modules Set 3 10d 07-Jul-28 20-Jul-28 10d 11-Aug-28 11-Aug-28 </td <td></td> <td>Modules / Skids</td> <td></td> <td>120d</td> <td>07-Jul-28</td> <td>03-Jan-29</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>V</td> <td></td> <td></td>		Modules / Skids		120d	07-Jul-28	03-Jan-29									V		
A1XY1100 Modules Set 3 hook-up Modules Set 3 hook-up Modules Set 3 hook-up Modules Set 3 hook-up Installation Modules Set 2 A1XY100 Installation Modules Set 2 10d 11-Sep-28 22-Sep-28 Installation Modules Set 2 Installati		A1XY1040	Installation Modules Set 3	10d	07-Jul-28	20-Jul-28								Inst	allation Mo	odules Set	3
A1XY1010 Installation Modules Set 2 Installation Modules Set 2 A1XY110 Modules Set 2 hook-up Installation Modules Set 2 Installation Modules Set 2 A1XY110 Modules Set 2 hook-up Installation Modules Set 2 Installation Modules Set 2 Remaining Level of Effort Actual Work Critical Remaining Work Data Date:26-Jul-24 Page: 15 of 73 Date Revision Checked Approved		A1XY1100	Modules Set 3 hook-up	30d	21-Jul-28	31-Aug-28								■ M	odules Se	t 3 hook-u	р
A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up A1XY110 Modules Set 2 hook-up Image: Actual Work Image: Critical Remaining Work Image: Actual Level of Effort Image: Actual Work Image: Actual Level of Effort Image: Critical Secondary		A1XY1010	Installation Modules Set 2	10d	11-Sep-28	22-Sep-28								l lr	stallation	Modules S	Set 2
Date Revision Checked Approved Actual Level of Effort Critical Remaining Work Critical Remaining Work Date Revision Checked Approved Actual Level of Effort Remaining Work Critical Secondary Date Revision Checked Approved		A1XY1110	Modules Set 2 hook-up	30d	25-Sep-28	06-Nov-28					L J I 1 1 1 1 1 1 1				Modules	Set 2 hook	-up
Remaining Level of Effort Actual Work Critical Remaining Work Data Date:26-Jul-24 Actual Level of Effort Remaining Work Critical Remaining Work Page: 15 of 73				1		1						ļ i i i		ļ I I I	+ · · ·	· · · ·	
Actual Level of Effort Critical Secondary Page: 15 of 73		 Remaining Leve 	l of Effort			Data Date:26	-Jul-24			Date			Revision		Che	cked	Approved
	—	 Actual Level of E 	ffort Eritical Secondary			Page: 15 c	of 73			19-Aug-24							

SOLARIS



TODTIC			Tilbury Island LNG Facility Expansion (3 BCF)							SOLADIS					
		FORTISBC	TLSE	Sub-Proie	ects Integra	ted Schedule	-Detailed				1	200	JAKI	3	
		I OITITO DO		J -							-	CONSUL	GEMEN TANTS IN	T C.	
Activ	ity ID	Activity Name	Rem.	Start	Finish	20 2021	2022 2	2023 2024	2025	2026	2027	2028 2	2029	2030 2031	
			Duration												
	A1XY1000	Installation Modules Set 1	10d	31-Oct-28	14-Nov-28							Insta	Ilation Modu	les Set 1	
	A1XY1090	Modules Set 1 hook-up	30d	15-Nov-28	03-Jan-29							📕 Mo	dules Set 1	hook-up	
	Piping		160d	08-Jun-28	01-Feb-29										
	A1XR1010	Piping Field Installation & Testing	160d	08-Jun-28	01-Feb-29							P	ping Field In	stallation & lesting	
	Electrical & Instru		180d	16-Jun-28	12-Mar-29									uitebacar (Drofob	
	A1XE1010	Installation MV / LV Switchgear (Pretab Distr. Center)	20d	16-Jun-28	17-Jul-28										
	A1XJ1010	Installation DCS & ESD (Prefab Distr. Center)	20d	16-Jun-28	17-Jul-28									D (Preiad Distr. C	
	A1XE1000	Electrical Field Installation incl. Testing	160d	22-Jun-28	15-Feb-29									installation incl. le	
	A1XJ1000	Instrumentation Field Installation incl. Testing	130d	14-Aug-28	26-Feb-29				· · · · · · · · · · · · · · · · · · ·				istrumentatio	on Field Installation	
	A1XE1020	Electr. Substation energized	0d	30-Oct-28								♦ Elect	. Substation	energized	
	A1XJ1020	Loop Testing	70d	23-Nov-28	12-Mar-29								_oop Testing		
	Painting & Insulati	ion	136d	25-Aug-28	19-Mar-29							•			
	A1XW1000	Surface Protection Field	130d	25-Aug-28	09-Mar-29								Surface Prot	ection Field	
	A1XV1000	Insulation Field	130d	05-Sep-28	19-Mar-29								Insulation Fig	eld	
	Precommissioning		100d	23-Oct-28	22-Mar-29										
	A1YF1000	Precommissioning	100d	23-Oct-28	22-Mar-29								Precommiss	ioning	
	Commissioning & S	Start Up	100d	29-Jan-29	21-Jun-29								V Commiss	ioning 8 Start Lin	
	A1YF1010	Commissioning & Start Up	100d	29-Jan-29	21-Jun-29								Toot Dup	ioning a Start op	
	A1YF1030	Test Run	2d	13-Jun-29	15-Jun-29										
	_Ground Improven	nent & Early Works - Regasification Area	283d	30-Jan-26	23-Mar-27										
	Geotechnical Inv	estigation	110d	30-Jan-26	08-Jul-26										
	GR1.1	Geotechical Field Investigation Including Coordination	60d	30-Jan-26	27-Apr-26					Geoted	chical Field Invest	igation Includ	ng Coordina	ition	
	GR1.2	Geotechical Analyses and Reporting Including Laboratory Testing	80d	16-Mar-26	08-Jul-26					Geo Geo	techical Analyses	and Reportir	ig Including I	Laboratory lesting	
	Detailed Enginee	ring	80d	28-Apr-26	20-Aug-26										
	GIT500020	Outside Tank Area Area Ground Improvement Detailed Engineering	80d	28-Apr-26	20-Aug-26					Οι	itside Tank Area A	rea Ground	mprovemen	t Detailed Enginee	
	Ground Improv. (Construction Contract (Concrete Fnd. Excluded)	137d	16-Mar-26	29-Sep-26										
	GR01	Prepare Bidding Documents	20d	16-Mar-26	13-Apr-26					Prepare	e Bidding Docum	ents			
	GR02	Contract Request Form/Email	1d	14-Apr-26	14-Apr-26					Contra	ct Request Form/	Email			
	GR03	Review/Draft Requirements and Discussion with Stakeholders	10d	15-Apr-26	28-Apr-26					Review	v/Draft Requireme	ents and Disc	ussion with S	Stakeholders	
	GR04	Review Competetive Bid Documents	10d	29-Apr-26	12-May-26					Reviev	v Competetive Bi	d Documents			
	GR05	Circulate RFQ Internally for Review, Comments and Amendments	10d	13-May-26	27-May-26					Circul	ate RFQ Internall	y for Review,	Comments a	and Amendments	
	GR06	Legal Review	10d	28-May-26	10-Jun-26					Lega	l Review				
	GR07	Finalize RFQ	10d	, 11-Jun-26	24-Jun-26					Final	ize RFQ				
	GR08	Receive Vendor/Contractors Proposals/Clarification	40d	25-Jun-26	21-Aug-26					📕 Re	ceive Vendor/Co	ntractors Prop	oosals/Clarifi	cation	
	GR09	Finalize SOW/Lindate Cost Estimate/Bid Evaluation//Negotiation/PR Creation and	1/ 20d	24-Aug-26	21-Sep-26					F	nalize SOW/Upd	ate Cost Estir	nate/Bid Eva	luation//Negotiation	
	GR12	Prenare and Award Contract - Only Construction	10d	16-Sep-26	29-Sep-26						repare and Awar	d Contract - 0	July Constru	ction	
	Construction Wo		123d	21-Sep-26	23-Mar-27						_				
	Regulatory Permits		0d	21-Sep-20	01 Oct 26					•					
	BC One Call		9d	21-Sep-26	01-Oct-26					—					
	Necessary Delive	erables	5d	21-Sep-26	28-Sep-26										
	RGT301112	Prepare Ticket	5d	21-Sep-26	28-Sep-26					I P	repare Ticket				
	Application		4d	28-Sep-26	01-Oct-26					▼					
	RGT301184	BC One Call Registration	1d	28-Sep-26	28-Sep-26					(B	C One Call Regis	stration			
	Remaining Leve	el of Effort ———— Actual Work ———— Critical Remaining Work			Data Date:26	Jul-24		Date		R	evision		Checked	Approved	
						(3 0		19-Aug-24	IFI					1	

Remaining Level of Effort	Actual Work	Critical Remaining Work	Deta Deta 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 16 01 75		



Activi	ty ID	Activity Name	Rem. Duration	Start	Finish		2021 Q Q Q		2024 QQQ	2025			202 Q Q Q		2029 Q Q Q Q	20 2 0 0	30 QQQ	2031 0 0 0
	RGT301188	Receive Info/Confirmation - BC One Call	3d	29-Sep-26	01-Oct-26							Receive Inf	o/Confirm	ation - E	BC One C	al		
	Regasification Area	Civil Work	113d	05-Oct-26	23-Mar-27			 						••••				
	GR2.0	Site Mobilization *impact of through HW instad of MOF logistic ?*	20d	05-Oct-26	02-Nov-26							Site Mobiliz	zation *	impact	of through	n HW ins	tad of N	10F logi
	GR2.1	Remove and Dispose Asphalt Surface	15d	03-Nov-26	24-Nov-26							Remove a	and Dispo	se Asp	halt Surfa	æ		
	GR2.1.1	Warehouse Demolition (TBD)	12d	06-Nov-26	24-Nov-26							Warehou	se Demol	ition (TE	3D)			
	GR2.2	Excavate 1.0 m depth soil and replace *impact of through HW instad of MOF log	دِ 15d	25-Nov-26	15-Dec-26							Excavate	e 1 0 m de	epth soil	l and repla	ce *ir	npact of	through
	GR3.1	Install 1.0 m diameter stone column 18.5 m length [LNG Expansion Area Outisde t	r 45d	14-Dec-26	23-Feb-27							💼 Install	1.0 m diar	neter sl	tone colur	nn 18.5 i	n length	ILNG E
	GR2.3	Supply and install structural sand 1 m	20d	24-Feb-27	23-Mar-27							Supp	ly and insi	tall struc	ctural sand	l1m		
	Auxiliary Systems	(Utility Rack & Equipment) Tie-in to Vaporizer Package	737d	29-Jun-26	21-Jun-29													
	EPC (Detailed - Co	ontractor)	737d	29-Jun-26	21-Jun-29						· · · ·							
	General		737d	29-Jun-26	21-Jun-29													
	TAX296120	Award EPC Contract	0d		29-Jun-26						• A		ontract	~~~~~~				
	G000002	Detail Engineering Kick Off (118D Lag from EPC award to match CPCN App Sch.)	0d	30-Jun-26							♦L	Jetail Enginee	ring Kick (Jff (118	D Lag froi	n EPC a	ward to	match C
	G000005	HAZOP/SIL/LOPA	0d		30-Oct-26								IL/LOPA					
	G000007	60% Model Review	0d		07-Dec-26							♦ 60% Mod		V				
	E0001801	Construction Permit Approval	0d	06-Jan-27									ction Pern	nit Appr	oval			
	G000006	90% Model Review	0d		09-Feb-27							♦ 90% M	lodel Revi	ew	~			
	E0001811	Ground Improvement Completed (Aux. Sys.)	0d		23-Mar-27							♦ Grou	nd Improv	rement	Complete	d (Aux. S	Sys.)	
	G0000076	Construction Mobilization to Site	0d	24-Mar-27								♦ Cons	truction M	obilizati	ion to Site			
	G0000016	Issue for Construction (IFC)	0d		22-Apr-27							♦ Issu	e for Cons	struction	n (IFC)			
	G0000046	Final Document Class 2 Control Estimate Completed	0d		22-Apr-27							l	I Docume	nt Clas	s 2 Contro	ol Estima	te Comp	bleted
	G000086	Last Piece of Mechanical Equipment Delivered to Site	0d		25-Jun-27							♦La	ast Piece o	of Mech	ianical Equ	upment	Delivere	d to Site
	E0005891	Pipe Fabrication Complete	0d		21-Oct-27								Pipe ⊢a	ibricatio	on Comple	te		
	G0000066	Project Mechanical Completion - Vaporizers	0d		07-Dec-27									ct Mech	anical Col	npletion	- Vapor	zers
	G0000096	Commissioning & Startup Completed - Vaporizers	0d		21-Jun-29										◆ Co	nmissior	ning & S	tartup C
	G0000126	Turnover to Operations - Vaporizers	0d		21-Jun-29				1 1 1 1 1 1						♦ Tur	nover to	Operat	ions - Va
	Engineering & Desig	gn	200d	30-Jun-26	22-Apr-27													
	E0000071	ΗΔΖΟΡ	1200 5d	26-Oct-26	22-Apr-27 30-Oct-26							HAZOP						
	E0000081		20d	02-Nov-26	30-Nov-26								Closeout					
	G0000001	60% Model Review	200 5d	02-N00-20	07-Dec-26							- I 60% Moo	lel Review	v				
	G0000101	90% Model Review	5d	03-Eeb-27	09-Eeb-27			 				I 90% M	lodel Revi	ew				· - + + + + + +
	E0000121	Prenare Class 2 Control Estimate	30d	11-Mar-27	22-Apr-27							Prep	oare Class	2 Con	trol Estima	ite		
	Milestones		180d	28-Jul-26	22-Apr-27						-							
	Process		140d	28-Jul-26	24-Feb-27						-							
	E0003871	Simulation - IFC	0d		28-Jul-26						•	Simulation - IF	⁼C					
	E0003921	Heat & Material Balance - IFC	0d		28-Jul-26					1 1 1 1 1 1 1 1 1	•	Heat & Materi	ial Balance	ə - IFC				
	E0004201	Hydraulic Calculations/Line Sizing - IFC	0d		26-Aug-26						•	Hydraulic Ca	lculations	/Line \$i	zing - IFC			
	E0004251	Process Flow Diagram - IFH	0d		26-Aug-26						•	Process Flow	v Diagram	ו- IFH				
	E0004091	Utilities Requirement - IFH	0d		10-Sep-26						•	Utilities Requ	uirement -	IFH				
	E0000031	P&ID - IFH	0d		24-Sep-26							P&ID - IFH						
	E0004611	Loss Management Philosophy - IFH	0d		24-Sep-26							Loss Manag	gement Pl	hilosopl	hy - IFH			
	E0004781	Control Narrative - IFH	0d		24-Sep-26							Control Nar	rative - IF	H				
	Remaining Leve	el of Effort Actual Work Critical Remaining Work			Data Date:2 Page: 17	6-Jul-24 of 73		19-4	Date Aug-24	IFI		Revision			Check	ked	Appro	oved

SOLARIS



Activit	y ID	Activity Name	Rem.	Start	Finish	2020 2021 2022	2023 2024	2025 2026 2027 2028 2029 2030 2031
			Duration					
	E0006451	Vendor P&ID - IFH	0d		24-Sep-26			◆ Vendor P&ID - IFH
	E0003981	Material Selection Diagram - IFH	0d		08-Oct-26			♦ Material Selection Diagram - IFH
	E0004361	Utility Flow Diagram - IFH	0d		08-Oct-26			◆ Utility Flow Diagram - IFH
	E0004681	Operating Philosophy - IFH	0d		23-Oct-26			Operating Philosophy - IFH
	E0004741	Pressure Relief Summary - IFH	0d		23-Oct-26			♦ Pressure Relief Summary - IFH
	E0004311	Process Flow Diagram - IFC	0d		14-Dec-26			Process Flow Diagram - IFC
	E0000061	P&ID - IFC	0d		12-Jan-27			♦ P&ID - IFC
	E0004701	Loss Management Philosophy - IFC	0d		12-Jan-27	I I		Loss Management Philosophy - IFC
	E0006481	Control Narrative - IFC	0d		12-Jan-27			Control Narrative - IFC
	E0001831	C&SU Procedure - IFC	0d		19-Jan-27			♦ C&SU Procedure - IFC
	E0004141	Utilities Requirement - IFC	0d		19-Jan-27	I I		♦ Utilities Requirement - IFC
	E0004031	Material Selection Diagram - IFC	0d		02-Feb-27			♦ Material Selection Diagram - IFC
	E0004601	Utility Flow Diagram - IFC	0d		02-Feb-27			♦ Utility Flow Diagram - IFC
	E0006471	Vendor P&ID - IFC	0d		02-Feb-27			♦ Vendor P&ID - IFC
	E0004721	Operating Philosophy - IFC	0d		24-Feb-27			Operating Philosophy - IFC
	E0004761	Pressure Relief Summary - IFC	0d		24-Feb-27			♦ Pressure Relief Summary - IFC
	Civiil/Structural/A	rchitectural	150d	12-Aug-26	24-Mar-27			
	E0000341	Civil Engineering Design Criteria - IFC	0d		12-Aug-26			♦ Civil Engineering Design Criteria - IFC
	E0000611	Structural Engineering Design Criteria - IFC	0d		12-Aug-26			 Structural Engineering Design Criteria - IFC
	E0000781	Piperack Structural Capacity Report - IFC	0d		12-Aug-26			Piperack Structural Capacity Report - IFC
	E0000821	C/S/A Standard Drawings & Details - IFC	0d		12-Aug-26			♦ C/\$/A Standard Drawings & Details - IFC
	E0001131	C/S/A Specifications - IFC	0d		12-Aug-26			♦ C/\$/A Specifications - IFC
	E0000201	Structural Steel Location Plans - IFC	0d		10-Mar-27	I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I		 Structural Steel Location Plans - IFC
	E0000231	Building Drawings - IFC	0d		24-Mar-27			Building Drawings - IFC
	E0000261	Foundation Details & Drawings - IFC	0d		24-Mar-27			Foundation Details & Drawings - IFC
	E0000291	Earthworks/Civil Drawings - IFC	0d		24-Mar-27			Earthworks/Civil Drawings - IFC
	E0000791	MIscellaneous Steel - Support Details - IFC	0d		24-Mar-27			Miscellaneous Steel - Support Details - IFC
	E0000801	Structural Steel MTO for Estimating	0d		24-Mar-27			 Structural Steel MTO for Estimating
	Mechanical		100d	12-Aug-26	12-Jan-27			······
	E0006491	Equipment Specifications - IFC	0d		12-Aug-26			 Equipment Specifications - IFC
	E0001221	Spare Parts Listing	0d		02-Sep-26			♦ Spare Parts Listing
	E0000311	Equipment List - IFH	0d		24-Sep-26			♦ Equipment List - IFH
	E0001141	Fire Fighting Equipment List - IFH	0d		24-Sep-26			♦ Fire Fighting Equipment List - IFH
	E0000331	Equipment List - IFC	0d		12-Jan-27			Equipment List - IFC
	E0001171	Fire Fighting Equipment List - IFC	0d		12-Jan-27			Fire Fighting Equipment List - IFC
	Piping Engineerin	g	130d	12-Aug-26	24-Feb-27			
	E0004851	Piping Specifications - IFC	0d		12-Aug-26			♦ Piping Specifications - IFC
	E0000741	LDT - IFH	0d		24-Sep-26			◆ LDT:- IFH
	E0001851	LDT - IFC	0d		12-Jan-27			♦ LDT-IFC
	E0000871	Stress Analysis Reports - IFC	0d		24-Feb-27			 Stress Analysis Reports - IFC
	E0004791	SP Item List - IFC	0d		24-Feb-27			♦ SP Item List - IFC
	Piping Design		155d	12-Aug-26	31-Mar-27			
			1					Dovision Charled American
Remaining Level of Effort Actual Work Critical Remaining Work					Data Date:26	-Jul-24		IFI Cnecked Approved
	Actual Level of I	Effort Effort Remaining Work E			Page: 18 c	of 73	10 / lug 24	

SOLARIS



/ity ID	Activity Name	Rem. Start	Finish 020	0 2021 2022 2023	2024 2025 2026 2027	2028 2029 2030 2031
		Duration				
E0005021	Pipe Support Details - IFC	Od	12-Aug-26		♦ Pipe Support I	ietails - IFC
E0005041	Piping Standard Drawings & Details - IFC	Od	12-Aug-26		♦ Piping Standar	d Drawings & Details - IFC
E0006461	Plot Plan - IFH	Od	24-Sep-26		♦ Plot Plan - IF	4
E0001231	Tie-in List - IFH	Od	23-Oct-26		◆ Tie-in List - I	-H
E0005071	Fire Protection & Safety Equipment Layout - IFH	0d	23-Oct-26		♦ Fire Protecti	n & Safety Equipment Layout - IFH
E0005141	Plot Plan - IFC	Od	12-Jan-27		Plot Plan	-IFC
E0000901	Equipment Location Plan - IFC	Od	02-Feb-27		♦ Equipme	nt Location Plan - IFC
E0004971	Utility Location Plan - IFC	Od	02-Feb-27		♦ Utility Lo	ation Plan - IFC
E0001661	Piping Tie-in List - IFC	Od	24-Feb-27		♦ Piping T	i∉-in List - IFC
E0004961	Equipment General Arrangement Drawings - IFC	0d	24-Feb-27		◆ Equipm	ent General Arrangement Drawings - IFC
E0000851	Piping (PVF, Insulation Piping SP Items) MTO for Estimating	Od	24-Mar-27		♦ Piping	(PVF, Insulation Piping SP Items) MTO for Estimati
E0001711	General Arrangement Key Plan - IFC	Od	24-Mar-27		♦ Gener	al Arrangement Key Plan - IFC
E0004881	Model Key Plan - IFC	Od	24-Mar-27		♦ Model	Key Plan - IFC
E0004901	Construction Isos - IFC	Od	24-Mar-27		♦ Constr	uction Isos - IFC
E0005101	Fire Protection & Safety Equipment Layout - IFC	Od	24-Mar-27		♦ Fire Pr	otection & Safety Equipment Layout - IFC
E0004921	Piping General Arrangement Drawings - IFC	Od	31-Mar-27		♦ Piping	General Arrangement Drawings - IFC
Electrical		170d 12-Aug-26	22-Apr-27			
E0005341	Electrical Standard Drawings & Details - IFC	Od	12-Aug-26		◆ Electrical Stand	lard Drawings & Details - IFC
E0005361	Electrical Specifications - IFC	Od	12-Aug-26		♦ Electrical Spec	fications - IFC
E0005521	Grounding/Cathodic Protection Details - IFC	Od	12-Aug-26		♦ Grounding/Ca	hodic Protection Details - IFC
E0005221	Area Classification Diagram - IFH	Od	23-Oct-26		🔶 Area Classif	cation Diagram - IFH
E0005181	Single Line Diagram - IFC	Od	09-Feb-27		♦ Single L	ne Diagram - IFC
E0005201	Cable Schedule - IFC	Od	09-Feb-27		♦ Cable S	shedule - IFC
E0005161	Electrical Load List - IFC	Od	24-Feb-27		♦ Electrica	I Load List - IFC
E0005241	Area Classification Diagram - IFC	Od	24-Feb-27		♦ Area Cl	assification Diagram - IFC
F0005481	Tie-In List - IFC	Od	24-Feb-27		♦ Tie-In:Li	st-IFC
E0005961	FTAP/Arc Flash - IFC	Od	24-Feb-27		♦ ETAP/A	rc Flash - IFC
E0005261	24V Distribution Wiring Schematics - IEC		10-Mar-27		◆ 24V Di	stribution Wiring Schematics - IFC
E0005281	120V Schematics - IEC	b0	10-Mar-27		▲ 120V S	chematics - IFC
E0005201			10 Mar 27		▲ 600V/4	80V Schematics - IFC
E0005301			10-Iviai-27		▲ 4160V	Schematics - IEC
E0003321	4 100 V SUIEIIIdus - IFC		10-ividi-∠7			-lectrical (Cable/Trav/Conduit/Junction Box) MTO f
E0001211			24-ividi - 21		↓ linetic	m Box I avout - IFC
E0005401			24-ividi - 21		◆ Julique ▲ Contai	for Panel Lavout - IEC
E0005421		DU الم				Panel avout - IEC
E0005461			24-iviar-27		↓ Conuc	
	Distribution Panel Drawings - IFC	UQ	24-IVIAr-27			
E0005571	Electrical Equipment Layout - IFC		24-iviar-27			a Equipment Layout - IFC
E0005611		Ud	24-Mar-27		◆ New S	
E0005381	Input/Output Drawings - IFC	Ud	22-Apr-27			Popular Lighting & Craunding Lauguit
E0005551	Cable, Raceway, Lighting & Grounding Layout - IFC	0d	22-Apr-27			Traveway, Lighting & Grounding Layout - IFC
E0005631	Cable Tray Layout - IFC	Od	22-Apr-27			
E0006511	Heat Tracing Layout - IFC	0d	22-Apr-27		→ Heat	racing Layout - IFC
Remaining L	evel of Effort		Data Date:26-J	ul-24	Date Revision	Checked Approved
Actual Level	of Effort Remaining Work Critical Secondary		Page: 19 of	73	19-Aug-24 F	
						I I

SOLARIS



Activity	/ ID	Activity Name	Rem. Duration	Start	Finish		2024	2025		2029	2030 2031
	Instrumentation		170d	12 Aug 26	22 Apr 27						
	F0005931	Standard Drawings & Details - IFC	0d	12-Aug-20	12-Aug-26				Standard Drawings & Det	ails - IFC	
	E0005941	I&C Specifications - IFC	0d 0d		12-Aug-26				♦ I&C Specifications - IFC		
	E0001541	Shut Down Key - IFH	0d 0d		24-Sep-26				♦ Shut Down Key - IFH		
	E0001561	Instrument Index - IFH	Dd Dd		23-Oct-26				Instrument Index - IFH		
	E0001581	Alarm Set Point Table - IFH	D0		23-Oct-26				♦ Alarm Set Point Table -	FH	
	E0005701	Metering Schematics - IFH	D0		23-Oct-26				 Metering Schematics - I 	FH	
	E0005771	Fire & Gas Detection Drawings - IEH	Dd Dd		23-Oct-26				♦ Fire & Gas Detection D	awings - IFH	
	E0005911	PCS/PLC/SIS/SCADA Architecture Diagram - IEH	Dd 0d		23-Oct-26				♦ PCS/PLC/SIS/SCADA /	rchitecture Diagr	am - IFH
	E0000311	Shut Down Key - IEC	b0 Dd		12- Jan-27				Shut Down Key - IFC		
	E0001001 E0001911	Alarm Set Point Table - IFC	b0 Dd		24-Feb-27				♦ Alarm Set Point Tab	le - IFC	
	E0004551		b0 Dd		24-Feb-27				♦ Instrument Index -	FC	
	E0005641	System Block Diagram - IEC	Dd		24-Feb-27				Svstem Block Diag	am - IFC	
	E0005681		Dd		24-Feb-27				♦ I&C Tie-In Drawing	- IFC	
	E0005731	Metering Schematics - IEC	Dd		24-Feb-27				♦ Metering Schematic	s+IFC	
	E0005781	Fire & Cas Detection Drawings - IEC	Dd Od		24-1 CD-27				♦ Fire & Gas Detection	n Drawings - IFC	
	E0005701	PCS/PLC/SLS/SCADA Architecture Diagram - IEC	Dd		24-Feb-27				♦ PCS/PLC/SIS/SCA	DA Architecture)iagram - IFC
	E0003921		Dd Od		24-1 eb-27				▲ Instrumentation &	Controls MTO	
	E0005661		0d		24-iviai-27				Instrument Lavor	t - IFC	
	Process		1604	20 Jun 26	22-Api-27						
	Simulation		20d	30-Jun-26	24-Feb-27 28-Jul-26				•		
	E0000281	Simulation - IFC	20d	30-Jun-26	28-Jul-26				Simulation - IFC		
	Heat & Material Ba	alance	20d	30-Jun-26	28-Jul-26				•		
	E0000761	Heat & Material Balance - IFC	20d	30-Jun-26	28-Jul-26				📋 Heat & Material Balance - I	FC	
	Material Selection	Diagram	105d	27-Aug-26	02-Feb-27				V		
	E0000751	Material Selection Diagram - IFH	30d	27-Aug-26	08-Oct-26				Material Selection Diagram	am-IFH	
	E0000771	Material Selection Diagram - IFC	30d	15-Dec-26	02-Feb-27				Material Selection D	agram - IFC	
	Utilities		115d	29-Jul-26	19-Jan-27					,	
	E0000111	Utilities Requirement - IFH	30d	29-Jul-26	10-Sep-26						
	E0000221	Utilities Requirement - IFC	30d	01-Dec-26	19-Jan-27					- 1FC	
	Hydraulic Calculat	ions	20d	29-Jul-26	26-Aug-26				Hydraulic Calculations/Lin	e Sizing - IFC	
	E0004771		200	29-Jul-26	20-Aug-20						
	F0000091	Process Flow Diagram Development - IFH	20d	29-Jul-26	26-Aug-26				Process Flow Diagram D	evelopment - IFH	
	E0004641	Utility Flow Diagram Development - IFH	30d	27-Aug-26	08-Oct-26			- L J L 	🔲 Utility Flow Diagram Dev	elopment - IFH	
	E0000101	Process Flow Diagram Development - IFC	10d	01-Dec-26	14-Dec-26				Process Flow Diagram	n Development -	IFC
	E0004651	Litility Flow Diagram Development - IEC	30d	15-Dec-26	02-Feb-27				Utility Flow Diagram	Development - IF	c
	Piping & Instrumer	t Diagram (P&ID)	105d	27-Aug-26	02-Feb-27						
	E0000021	P&ID Development IFH	20d	27-Aug-26	24-Sep-26				P&ID Development IFH		
	E0004661	Vendor P&ID - IFH	20d	27-Aug-26	24-Sep-26				Vendor P&ID - IFH		
	E0000041	P&ID Development IFC	15d	15-Dec-26	12-Jan-27				P&ID Development I	=c	
	E0004671	Vendor P&ID - IFC	30d	15-Dec-26	02-Feb-27				📮 Vendor P&ID - IFC		
	Instrument Data SI	neet	10d	29-Jul-26	12-Aug-26						
							· · · · · · ·		<u>, , , , , , , , , , , , , , , , , , , </u>		
-	Remaining Level	of Effort Actual Work Critical Remaining Work			Data Data:2	a lul 24	Date		Revision	Checked	Approved

Actual Level of Effort

Remaining Work Critical Secondary

Data Date:26-Jul-24 Page: 20 of 73

19-Aug-24 IFI

SOLARIS

Revision	Checked	Approved



Tilbury Island LNG Facility Expansion (3 BCF) TLSE Sub-Projects Integrated Schedule-Detailed

ctivity ID	Activity Name	Rem.	Start	Finish 02	.0 2021 202	22 2023	2024 202	25 2026 2027 2028	2029	2030 2031
		Duration								
E0004691	Instruments Sizing Calculations - IFP	10d	29-Jul-26	12-Aug-26				Instruments Sizing Calculation	ıs - IFP	
Loss Managem	ent Philosophy	100d	13-Aug-26	12-Jan-27						
E0004561	Loss Management Philosophy - IFH	30d	13-Aug-26	24-Sep-26					ny-IFH	
E0004571	Loss Management Philosophy - IFC	30d	24-Nov-26	12-Jan-27				Loss Management Phil	sophy - I⊢C	
Operating Philo	osophy	100d	25-Sep-26	24-Feb-27				▼		
E0004581	Operating Philosophy - IFH	20d	25-Sep-26	23-Oct-26				Operating Philosophy - IF	1	
E0004591	Operating Philosophy - IFC	30d	13-Jan-27	24-Feb-27				Operating Philosophy	- IFC	
Pressure Relie	f Summary	100d	25-Sep-26	24-Feb-27				· · · · · · · · · · · · · · · · · · ·		
E0004621	Pressure Relief Summary - IFH	20d	25-Sep-26	23-Oct-26				Pressure Relief Summary	·IFH	
E0004631	Pressure Relief Summary - IFC	30d	13-Jan-27	24-Feb-27				Pressure Relief Summ	ary - IFC	
Control Narrati	ve	95d	27-Aug-26	19-Jan-27				vv		
E0000251	Prepare Control Narrative - IFH	20d	27-Aug-26	24-Sep-26				Prepare Control Narrative -	IFH	
E0001621	Prepare Control Narrative - IFC	30d	24-Nov-26	12-Jan-27				📋 Prepare Control Narrati	/e - IFC	
E0000141	Prepare C&SU Procedure - IFC	30d	01-Dec-26	19-Jan-27				🛑 Prepare C&SU Proced	ure - IFC	
Civil/Structural/	Architectural	180d	30-Jun-26	24-Mar-27						
Design Criteria		30d	30-Jun-26	12-Aug-26						
E0004801	Civil Engineering Design Criteria - IFC	30d	30-Jun-26	12-Aug-26				Civil Engineering Design Crite	eria - IFC	
E0004811	Structural Engineering Design Criteria - IEC	30d	30-Jun-26	12-Aug-26				🔲 Structural Engineering Desig	n Criteria - IFC	
Modeling		145d	30- Jun-26	02-Feb-27						
G000001	C/S/A Modeling to 60%	104d	30-Jun-26	27-Nov-26				C/S/A Modeling to 60%		
E0000171	C/S/A Modeling to 90%	35d	08-Dec-26	02-Feb-27				C/S/A Modeling to 90%		
Sitowork		204	10 Ech 27	24 Mar 27						
E0000271	Earthworks/Civil Drawings - IEC	30d	10-Feb-27	24-Ivial-27				Earthworks/Civil Drav	vings - IFC	
Econdation		204	10-1 CD-27	24-Mar 27					,	
E0000241	Foundation Details & Drawings - IFC	30d	10-Feb-27	24-Mar-27				Foundation Details &	Drawings - IFC	
Structural Stool		1904	20 Jun 26	24 Mar 27					U	
E0004861	Piperack Structural Capacity Report - IEC	30d	30-Jun-26	12-Aug-26				Piperack Structural Capacity	Report - IFC	
E0004001	Miscollanoous Stool Support Dataila JEC	304	10 Ech 27	24 Mar 27				Miscellaneous Steel	Support Details	- IFC
E0003001		300	10-Feb-27	24-IVIdI-27						
E0000181	Structural Steel Drawings - IFC	100	25-Feb-27	10-Mar-27					iys-ii C	
Buildings		30d	10-Feb-27	24-Mar-27				Building Drawings - I	c	
E0004831	Building Drawings - IFC	300	10-Feb-27	24-Mar-27					`	
Material Take C		30d	10-Feb-27	24-Mar-27					for Estimating	
E0004871	Structural Steel MIO for Estimating	30d	10-Feb-27	24-Mar-27						
Standard Draw		30d	30-Jun-26	12-Aug-26					Dotaile JEC	
E0004821	C/S/A Standard Drawings & Details - IFC	300	30-Jun-26	12-Aug-26						
Specification		30d	30-Jun-26	12-Aug-26						
E0004841	C/S/A Specifications - IFC	30d	30-Jun-26	12-Aug-26						
Requisition		20d	30-Jun-26	28-Jul-26				<u> </u>		
<u>20014-C-5302</u>	20014 C 5202: Columnized Steel Branava IED Data Shaata & Shaata	20d	30-Jun-26	28-Jul-26					teel - Prenare IEI	P Data Sheets & S
E0002901	20014-C-3302: Galvanized Steel - Prepare IFP Data Sheets & Specifications	200	30-Jun-26	28-JUF-26						
20014-C-5303	Concrete	20d	30-Jun-26	28-Jul-26				■ 20014-C-5303 - Concrete - P	renare IFP Nata	Sheets & Specifics
E0002961	20014-C-5303 - Concrete - Prepare IFP Data Sneets & Specifications	20d	30-Jun-26	28-Jul-26						
20014-C-53XX	NEIL MATERIALS	20d	30-Jun-26	28-Jul-26					- Prenare IFP D	ata Sheets & Shee
E0003021	20014-C-33XX - Fill Materials - Prepare IFP Data Sheets & Specifications	200	30-Jun-26	28-JUE26						
Mechanical		130d	30-Jun-26	12-Jan-27						
							Date	Revision	Checked	Annroved
Remaining Le	evel of Effort Actual Work Critical Remaining Work			Data Date:26-	Jul-24	19-4		I CEVISION		
Actual Level of	of Effort Effort Remaining Work E			Page: 21 of	73	13-				+

SOLARIS

Revision	Checked	Approved



ctivity	ID	Activity Name	Rem.	Start	Finish	020	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031
			Duration			QQ	QQQ										
	Equipment List	Equipment List (E1)	100d	13-Aug-26	12-Jan-27								Equipment List	- IFH		, , , , , , , , , , , , , , , , , , ,	
	E0000301		300	13-Aug-26	24-Sep-26	_							Eire Fighting E	uinmentList.	IEH		
	E0005501		300	13-Aug-26	24-Sep-26	_								liet IEC			
	E0000321		300	24-NOV-26	12-Jan-27									LISI - II C	int IEC		
	E0005511	Fire Fighting Equipment List - IFC	30d	24-Nov-26	12-Jan-27									J⊏quipmenu			
	Spare Parts		45d	30-Jun-26	02-Sep-26								auinment Sner	fications - IEC			
	E0006321		300	30-Jun-26	12-Aug-26	_							Spare Parts List	ng			
	E0005531	Spare Parts Listing	10d	20-Aug-26	02-Sep-26									ng			
	20014-M-5103: In	nstrument Air Package	85d	30-Jun-26	30-Oct-26												
	E0000381	20014-M-5103 - Instrument Air Package - Prepare RFP Data Sheets & Specification	20d	30-Jun-26	28-Jul-26							2	0014-M-5103 -	nstrumentAir	Package - Pro	epare RF	P Data Sh
	E0000391	20014-M-5103 - Instrument Air Package - Receive Vendor Info	D0	20-Aug-26								♦ 2	20014-M-5103-	Instrument Ai	r Package - R	eceive Ve	endor Info
	E0000401	20014-M-5103 - Instrument Air Package - Review Vendor Data	10d	20-Aug-26	02-Sen-26	-						1 2	20014-M-5103	InstrumentA	ir Packade - R	eview Ve	ndor Data
	E0000461	20014-M-5103 - Instrument Air Package - Code 1 Vendor Data	Dd	207/03/20	02-Sep-26							•	20014-M-5103	InstrumentA	ir Package - C	ode 1 Ve	ndor Data
	E0000401	20014 M 5103 Instrument Air Package - Code - Vendor Data	0d	10 Oct 26	02-069-20	_							20014-M-510	3 - Instrument	Air Package -	Receive	Vendor Da
	E0000471	20014 M 5103 - Instrument Air Package - Neceive Vendor Data Books	104	19-00-20	20 Oct 26	-							20014-M-510	3 - Instrument	Air Package -	Review \	Vendor Da
	E0000481	20014-M-5103 - Instrument Air Package - Review Vendor Data Books	100	19-OCI-20	30-Oct-26								20014-M-510	3 - Instrument	Air Packade -		nroval For
	E0000491	20014-IVI-5103 - Instrument Air Package - Issue Approval For Payment		00 1	30-Oct-26								20014-10-010		All Tackage -	issue Ap	
	20014-M-5104: N	100010-M-5104 - Nitrogen Generation (PSA) Package - Prenare REP Data Sheets	20d	30-Jun-26	30-Oct-26							1 20	0014-M-5104 -	Nitrogen Gen	eration (PSA)	Package	Prepare
	E0000651	20014 M 5104 - Nitrogen Concration (PSA) Package - Prepare Nr F Data Sheets	200 0d	20 Aug 26	20-30-20	-						A 2	20014-M-5104 -	Nitrogen Ger	neration (PSA)	Package	- Receive
	E0000051	20014 M 5104 - Nillogen Generation (FSA) Package - Receive Vendor Into	104	20-Aug-20	02.000.00	_							20014-M-5104	Nitrogen Gei	neration (PSA	Package	e - Review
	E0000601	20014-M-5104 - Nitrogen Generation (PSA) Package - Review Vendor Data	100	20-Aug-26	02-Sep-26								20014-M-5104	Nitrogen Gei	neration (PSA) Packade	
	E0000671	20014-M-5104 - Nitrogen Generation (PSA) Package - Code 1 Vendor Data	Ud	40.0.1.00	02-Sep-26	_						•	20014-0-010-	Nitrogen C	eneration (PS	A) Dacka	
	E0000911	20014-M-5104 - Nitrogen Generation (PSA) Package - Receive Vendor Data Bool	0d	19-Oct-26									20014 M 510		cheration (PS	A) Pooko	
	E0000921	20014-M-5104 - Nitrogen Generation (PSA) Package - Review Vendor Data Book	10d	19-Oct-26	30-Oct-26	_							20014-10-510			A) Pauka	
	E0000961	20014-M-5104 - Nitrogen Generation (PSA) Package - Issue Approval For Payme	0d		30-Oct-26								20014-10-510	4 - Nitrogen C	eneration (PS	А) Раска	ige - issue
	20014-M-51XX: L	NG Drain Vessel (With Emerson Heater)	85d	30-Jun-26	<u>30-Oct-26</u>								011.M51XX	I NIC Drain \/	esol (Mith Err	ereon He	aptor) - Dre
	E14801	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Prepare RFP Data S	200	30-Jun-26	28-JUF-26	_							0014 M 51XX				(alei) - i le
	E14811	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Receive Vendor Info	0d	20-Aug-26		_						◆ 2					
	E14821	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Review Vendor Data	10d	20-Aug-26	02-Sep-26												leater) - N
	E14831	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Code 1 Vendor Data	0d		02-Sep-26							• 4	20014-10-5122				
	E14841	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Receive Vendor Data	0d	19-Oct-26								•	20014-10-517	K - LING Drain	vessei (vvitn	⊑merson ⊢	Heater) - I
	E14851	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Review Vendor Data	10d	19-Oct-26	30-Oct-26							0	20014-M-51X	X - LNG Drair	NVessel (VVith	Emerson	Heater) -
	E14861	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Issue Approval For P	0d		30-Oct-26								▶ 20014-M-51X	X - LNG Drair	NVessel (With	Emerson	Heater) -
	Piping Engineering		165d	30-Jun-26	03-Mar-27												
	Stress	Stross Analysis Banarta JEC	10d	10-Feb-27	24-Feb-27								Stress An	alvsis Reports	- IFC		
	E0005541		100	10-Feb-27	24-Feb-27											; 	
	Ene Designation	IDT Development - IEH	30d	13-Aug-26	12-Jan-27 24-Sen-26								LDT Developm	ent - IFH			
	E0000731	I DT Development - IFC	30d	24-Nov-26	12-lan-27	-								opment - IFC			
	Specialty Itoms		204	24-IN0V-20	12-Jall-27												
	F0005561	SP Items List - IEC	30d	13-Jan-27	24-Feb-27								SP Items	_ist - IFC			
	Specifications		304	30- lun-26	12-Aug-26		· -								·····		
	E0006501	Piping Specifications - IFC	30d	30-Jun-26	12-Aug-26							F	Piping Specificati	ons - IFC			
	Requisition		60d	01-Dec-26	03-Mar-27												
						- L		- l · · · ·		Data		i	Revision	I	Checked	<u>۸</u> ۳	
-	Remaining Leve	el of Effort Actual Work Critical Remaining Work			Data Date:2	6-Jul-2	4			19-Aug-24	IFI					Ар	PIOVED
Actual Level of Effort Effort Remaining Work Critical Secondary				Page: 22 of 73													

SOLARIS


ctivity ID	Activity Name	Rem. Duration	Start	Finish)2) 2021	2022	2023	2024	2025	2026		2027	2028	2029	2030	2031
200141-5201	- Stainlass Staal Dina	60d	01 Dec 26	03 Mar 27												
E0002441	20014-I -5201 - Stainless Steel Pipe - Prepare REP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27							20	014-L-5201	- Stainless	Steel Pipe -	Prepare F	RFP Data S
E0002451	20014-L-5201 - Stainless Steel Pipe - Receive Vendor Info	0d	18-Feb-27			· · · · · · · · · · · · · · · · · · ·			·		♦ 2	0014-L-520	01 - Stainles	s Steel Pipe	- Receive	e Vendor In
E0002461	20014-L-5201 - Stainless Steel Pipe - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	01 - Stainles	s Steel Pipe	- Review	v Vendor Da
20014-L-5202	- Carbon Steel Pipe	60d	01-Dec-26	03-Mar-27												
E0002361	20014-L-5202 - Carbon Steel Pipe - Prepare RFP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27							20	014-L-5202	2 - Carbon S	iteel Pipe - P	repare R	.FP Data Sł
E0002371	20014-L-5202 - Carbon Steel Pipe - Receive Vendor Info	0d	18-Feb-27								♦ 2	0014-L-520	02 - Carbon	Steel Pipe -	Receive	Vendor Info
E0002381	20014-L-5202 - Carbon Steel Pipe - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	02 - Carbor	Steel Pipe -	Review \	Vendor Dat
20014-L-5203	- Stainless Steel Fittings	60d	01-Dec-26	03-Mar-27												
E0002681	20014-L-5203 - Stainless Steel Fittings - Prepare RFP Data Sheets & Specification	20d	01-Dec-26	05-Jan-27							20	014-L-5203	3 - Stainless	Steel Fittings	s - Prepar	re RFP Dat
E0002691	20014-L-5203 - Stainless Steel Fittings - Receive Vendor Info	0d	18-Feb-27								♦ 2	20014-L-520	03 - Stainles	s Steel Fitting	gs - Rece	vive Vendor
E0002701	20014-L-5203 - Stainless Steel Fittings - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	03 - Stainles	s Steel Fittin	gs - Revi	ew Vendor
20014-L-5204	- Carbon Steel Fittings	60d	01-Dec-26	03-Mar-27												
E14481	20014-L-5204 - Carbon Steel Fittings - Prepare RFP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27							20	014-L-5204	I - Carbon S	iteel Fittings	- Prepare	RFP Data
E14491	20014-L-5204 - Carbon Steel Fittings - Receive Vendor Info	0d	18-Feb-27								◆ 2	20014-L-520	04 - Carbon	Steel Fittings	s - Receiv	/e Vendor li
E14501	20014-L-5204 - Carbon Steel Fittings - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	04 - Carbor	Steel Fitting	s - Revie	w Vendor D
20014-L-5205	- Stainless Steel Valves	60d	01-Dec-26	03-Mar-27		; ;							0		Баста	
E0000581	20014-L-5205 - Stainless Steel Valves - Prepare RFP Data Sheets & Specification:	20d	01-Dec-26	05-Jan-27							20	014-L-5205	o - Stainless	Steel Valves	- Prepare	e RFP Data
E0000591	20014-L-5205 - Stainless Steel Valves - Receive Vendor Info	0d	18-Feb-27								◆ 2	20014-L-520)5 - Stainles	s Steel Valve	s - Recei	ve Vendor
E0000601	20014-L-5205 - Stainless Steel Valves - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	05 - Stainles	s Steel Valve	es - Revie	ew Vendor I
20014-L-5206 ·	- Carbon Steel Valves	60d	01-Dec-26	03-Mar-27								0141 5006	Carbon		Dranara	
E14631	20014-L-5206 - Carbon Steel Valves - Prepare RFP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27		· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · ·			20		- Carbon S	oteel valves -	Prepare	RFPData
E14641	20014-L-5206 - Carbon Steel Valves - Receive Vendor Info	0d	18-Feb-27								◆ 2	0014-L-520	06 - Carbon	Steel Valves	- Receiv	e Vendor In
E14651	20014-L-5206 - Carbon Steel Valves - Review Vendor Data	10d	18-Feb-27	03-Mar-27							0 2	20014-L-52	06 - Carbor	Steel Valves	s - Reviev	w Vendor D
20014-5207 - 0	Cryogenic Pipe Supports	60d	01-Dec-26	03-Mar-27								014 5207	Crucconia		o Drono	
E0002601	20014-5207 - Cryogenic Pipe Supports - Prepare RFP Data Sheets & Specification	20d	01-Dec-26	05-Jan-27								0 4-5207	Chyogonic	Dina Suppor	s-rieµa	
E0002611	20014-5207 - Cryogenic Pipe Supports - Receive Vendor Info	Ud	18-Feb-27		,	· · · · · · · · · · · · · · · · · · ·			· · · · · ·		• 4	0014-5207		, Fipe Suppu		
E0002621	20014-5207 - Cryogenic Pipe Supports - Review Vendor Data	10d	18-Feb-27	03-Mar-27								20014-5207	- Cryogeni	c Pipe Suppo	ons - Rev	new vendor
20014-L-5208	- Insulation	60d	01-Dec-26	03-Mar-27							20	014-1-5208	8 - Insulation	- Prenare R	EP Data	Sheets & S
E0002701	20014 L-5206 - Insulation - Prepare RFP Data Sheets & Specifications	200	01-Dec-20	05-Jan-27							▲ 2	0014-1-520	08 - Insulatio	n - Receive	Vendor Ir	nfo
E0002771	20014-L-5208 - Insulation - Receive Vendor Inio	00	18-Feb-27	02 Mar 07								200141-52	08 - Inculatio		Vendor D	Jata
E0002781	20014-L-5208 - Insulation - Review Vendor Data	100	18-Feb-27	03-IVIAF-27		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·			20014-02				
Piping Design Modeling		185d	30-Jun-26	31-Mar-27												
G0000003	Piping Modeling to 60%	104d	09-Jul-26	07-Dec-26							Pipi	ng Modeling	g to 60%			
E0000831	Piping Modeling to 90%	35d	08-Dec-26	02-Feb-27							P	iping Model	ing to 90%			
Tie-In		100d	25-Sep-26	24-Feb-27												
E0005591	Tie-In List - IFH	20d	25-Sep-26	23-Oct-26							Tie-In	1 List - IFH				
E0005601	Piping Tie-In List - IFC	30d	13-Jan-27	24-Feb-27							F	Piping Tie-In	List - IFC			
Key Plan		30d	10-Feb-27	24-Mar-27							\blacksquare					
E0005651	General Arrangement Key Plan - IFC	30d	10-Feb-27	24-Mar-27								GeneralAr	rangement l	Key Plan - IF	С	
E0005691	Model Key Plan - IFC	30d	10-Feb-27	24-Mar-27								Model Key	Plan - IFC			
Piping General	Arrangements/Isos	35d	10-Feb-27	31-Mar-27					·							
E0000891	Construction Isos - IFC	30d	10-Feb-27	24-Mar-27								Constructio	n Isos - IFC			
								Dete			Devia		·	Charlin		\nnm\/ad
Remaining Le	evel of Effort Actual Work Critical Remaining Work			Data Date:26-	lul-24						Revisio	ווע		Checked	A	hhionea
Actual Level of	of Effort Effort Remaining Work E			Page: 23 of	73			10-71ug-24								

SOLARIS



	ACIVITY Name	Rem.	Start	Finish	p20	20	J21		2022		202	3	20)24		202
		Duration			QQ	QQ	Q	QQ	QQ	QQ	Q	QQ	QC		20	Q
E0005671	Equipment General Arrangement Drawings - IFC	10d	10-Feb-27	24-Feb-27												
E0000881	Piping General Arrangement Drawings - IFC	15d	11-Mar-27	31-Mar-27					1	1						
Location Plan		105d	27-Aug-26	02-Feb-27										+ -		,
E0005491	Plot Plan - IFH	20d	27-Aug-26	24-Sep-26					1							
E0005121	Plot Plan - IFC	15d	15-Dec-26	12-Jan-27												
E0001871	Equipment Location Plan - IFC	15d	13-Jan-27	02-Feb-27					1 1 1				1			
E0005621	Utility Station Location Plan - IFC	15d	13-Jan-27	02-Feb-27					1	-						
Pipe Support		30d	30-Jun-26	12-Aug-26												
E0005721	Pipe Support Details - IFC	30d	30-Jun-26	12-Aug-26					1	1			1			
Standard Drawings	& Details	30d	30-Jun-26	12-Aug-26					1				1			1
E0005741	Piping Standard Drawings & Details - IFC	30d	30-Jun-26	12-Aug-26					1				-			
Material Take Off		30d	10-Feb-27	24-Mar-27					1							
E0000841	Piping (PVF, Insulation Piping SP Items) MTO for Estimating	30d	10-Feb-27	24-Mar-27			j j				j. j.					
Fire Protection		120d	25-Sep-26	24-Mar-27					1	-			-			, , , , , , , , , , , , , , , , , , ,
E0005751	Fire Protection & Safety Equipment Layout - IFH	20d	25-Sep-26	23-Oct-26									-			
E0005761	Fire Protection & Safety Equipment Layout - IFC	30d	10-Feb-27	24-Mar-27					1							
Electrical Engineeri	ng & Design	200d	30-Jun-26	22-Apr-27					1	-						1
Load List		30d	13-Jan-27	24-Feb-27												
E0005061	Electrical Load List - IFC	30d	13-Jan-27	24-Feb-27					1							
Single Line Diagra	m	20d	13-Jan-27	09-Feb-27					1	4 1 1			-			1
E0005451	Single Line Diagram - IFC	20d	13-Jan-27	09-Feb-27					1							
ETAP/Arc Flash		30d	13-Jan-27	24-Feb-27												1
E0005951	E IAP/Arc Flash - IFC	30d	13-Jan-27	24-Feb-27												÷
		20d	13-Jan-27	09-Feb-27					1							
E0005091	Cable Schedule - IFC	200	13-Jan-27	09-Feb-27					1							
Area Classification	Area Classification Diagram	100d	25-Sep-26	24-Feb-27					1							
E0001101		200	20-Sep-20	23-00-20					1	-						1
E0005441	Area Classification Diagram - IFC	300	13-Jan-27	24-FeD-27												
Schematics	241/ Distribution Wiring Schematics	20d	10-Feb-27	10-Mar-27												
E0005291		200	10-Feb-27	10-Iviar-27					1	-			-			1
E0005311	120V Schematics - IFC	20d	10-Feb-27	10-Mar-27												
E0005331	600V/480V Schematics - IFC	20d	10-Feb-27	10-Mar-27												1
E0005351	4160V Schematics - IFC	20d	10-Feb-27	10-Mar-27							-1					
Standard Drawing	s & Details	30d	30-Jun-26	12-Aug-26					1	1			-			
E0005411	Electrical Standard Drawings & Details - IFC	30d	30-Jun-26	12-Aug-26					1							1
Specifications		30d	30-Jun-26	12-Aug-26					1				1			
E0005431	Electrical Specifications - IFC	30d	30-Jun-26	12-Aug-26					1	1						
Input/Output Drawi		30d	11-Mar-27	22-Apr-27												;
E0005391	Input/Output Drawings - IFC	30d	11-Mar-27	22-Apr-27					1							
Junction Box & Pa	nel Drawings/Layout	30d	10-Feb-27	24-Mar-27					1	1			-			
E0005231	Junction Box Layout - IFC	300	10-Feb-27	24-Mar-27					1							
E0005251	Contactor Panel Layout - IFC	30d	10-Feb-27	24-Mar-27						1						
E0005271	Control Panel Layout - IFC	30d	10-Feb-27	24-Mar-27						, , ,						;
E0005371	Distribution Panel Drawings - IFC	30d	10-Feb-27	24-Mar-27					1							
Tie-In		30d	13-Jan-27	24-Feb-27												

SOLARIS

M A N A G E M E N T CONSULTANTS INC.

20	26			2027		20	28		202	9		20	30		2	2031	1
2 Q	Q	Q	Q	QQ	Q	QQ	QC	2 Q	Q	QQ	Q	Q	Q	Q	Q	Q	Q
1			0 1	Equipi	me	nt Ger	ieral/	۱rrar	igen	nent E	Dra	wing	js -	IF	С		
1	· · ·		Ľ	Pipin	g¢	Genera	ll Arra	nge	men	t Drav	vin	gs -	IFC	2	1		
			▼ .				 										
1		Ρ	lot P	'lan - I	FH					1							
		ļ	I Pl	ot Pla	n -	IFC											
				quipn	neņ	it Loca	tion F	Plan	- IFC				1		1	1	
1	. 1 1 1 1 1			Jtility S	tati	on Loo	cation	Plai	n - IF	C			: : :		1		-
		D :							+ -				4		+		
		чp	e Si	lpoor	υe	etalls -	iFC			1			1		1		
		Din	ina	Stand	ard	Draw	inde 8	2 00	taile				1		1		
1		ι-ih	<u> </u>	Janu	aru	Jaw	u iyə c	x De	allo	- 11-0			1		1		
1	· · ·			Pipin	a (F	>VF In	sulati	ion F	Pipin	a SP I	ten	າຣ) ໄ	ИТ) fr	or F	stin	nati
					9 (···-							
1]	⁻ire l	Protec	ctio	n & Sa	fety E	quip	bmer	nt Lay	ou	: : - IF	ΞH				
				Fire F	Pro	tection	& Sa	fety	Equ	ipmer	ht L	ayc	out -	IF	С		
				,						-			1				
			-			1				1			1		1		-
			– E	Electri	cal	Load	List - I	FC	+-	l 					+	1	
	· · · · · · · · · · · · · · · · · · ·				.												
1				ongle	Lin	e Diag	ram -	IFC		1 1 1					1		
: : :	· · ·			- 14	A	, El '				1 1 1							
!]	┛╎╹ ╼╼╶┼╴	_ IAP/	ч¢	, riash	ı - ı⊢C	1		·!							
: : :			• 7)ahle	Sch	Jedule	- IFC			1							
: : :										1							
: : :		,	∖rea	Class	sific	ation [Diagra	1m -	IFH	1 1 1			1				
1 1 1				Area (Clar	ssificat	ion Di	iagra	am -	IFC			1				
	;;- ;;-									· · · · · · · · · · · · · · · · · ·			·		+	;-	
1	· · ·			24V C	Dist	ributio	n Wiri	ng S	Sche	matic	s - İ	FC			1		
	· · ·			120V	Sd	hemat	ics - Il	FC		- - - -					1		
1				600V	48	0V Scl	hema	tics -	IFC	;			: : :		1		
1	/ 1 1 1 1 1			4160	٧s	chema	atics -	IFC		1 1 1			1			1	-
													· 4				
		Ele	ctric	al Sta	nd	ard Dra	awing	s & I	Deta	ils - IF	С						
	-					1	· · ·			1							
		Ele	ctric	al Spe	cifi	cation	s-IFC						1		1	1	
]								,				
	· · ·	-		Inpu	ıt/Ö	utput	Drawi	ngs	- IF (3				-	1		-
1					ы.	.			~	1			1		1		
1	1 1 1 1 1 1			Junc	uon	I ROX L	ayou	r IF	U.	~							
1				Cont	act	or Pan	iel La	yput	- IF(
	· · ·			Cont	rol	Panel	∟ayoι	ut - II	⊦C								
		-		Distri	but	ion Pa	nel D	rawi	ngs	- IFC				-	1	1	
1																	
		R	evisio	on					С	hecke	d	Т		App	rov	ed	
								+									

Activi	y ID	Activity Name	Rem.	Start	Finish	20 2021 2022 2023 2024 2025 2026	2027 2028 2029 2030 2031
			Duration				
	E0005191	Tie-In List - IFC	30d	13-Jan-27	24-Feb-27		j iię-in¦list-if-c
	Grounding/Cathod	lic Protection	30d	30-Jun-26	12-Aug-26		unding/Cathodic Protection Details - IEC
	E0005111	Grounding/Cathodic Protection Details - IFC	300	30-Jun-26	12-Aug-26		
		Heat Tracing Layout JEC	15d	01-Apr-27	22-Apr-27		■ Heat Tracing Layout - IFC
	Modeling		1454	20 Jup 26	02 Ech 27		
	F0001192	Electrical Modeling to 60%	1450 104d	30-Jun-26	27-Nov-26		Electrical Modeling to 60%
	E0001192	Electrical Modeling to 90%	35d	08-Dec-26	02-Eeb-27		Electrical Modeling to 90%
	Material Take Off		30d	10-Eeb-27	24_Mar_27	<u></u>	
	E0001201	Electrical (Cable/Tray/Conduit/Junction Box) MTO for Estimating	30d	10-Feb-27	24-Mar-27		Electrical (Cable/Tray/Conduit/Junction Box) MTO for Est
	Electrical Lavouts		50d	10-Feb-27	22-Apr-27		
	E0005151	Electrical Equipment Layout - IFC	30d	10-Feb-27	24-Mar-27		🗖 Electrical Equipment Layout - IFC
	E0005171	New Substation Layout - IFC	30d	10-Feb-27	24-Mar-27		New Substation Layout - IFC
	F0005081	Cable Tray Layout - IFC	15d	01-Apr-27	22-Apr-27		Cable Tray Layout - IFC
	E0005131	Cable Raceway Lighting & Grounding Layout - IFC	15d	01-Apr-27	22-Apr-27		Cable, Raceway, Lighting & Grounding Layout - IFC
	Requisition		80d	01-Dec-26	31-Mar-27		
	20014-E-5500: El	ectrical Building	80d	01-Dec-26	31-Mar-27		🚽 🗌 🔄 🔄 🔤 🔤 🔤 🔤
	E14731	20014-E-5500 - Electrical Building - Prepare RFP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27	╡╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴	20014-E-5500 - Electrical Building - Prepare RFP Data She
	E14741	20014-E-5500 - Electrical Building - Receive Vendor Info	0d	20-Jan-27			20014-E-5500 - Electrical Building - Receive Vendor Info
	E14751	20014-E-5500 - Electrical Building - Review Vendor Data	10d	20-Jan-27	02-Feb-27		20014-E-5500 - Electrical Building - Review Vendor Data
	E14761	20014-E-5500 - Electrical Building - Code 1 Vendor Data	0d		02-Feb-27		▶ 20014-E-5500 - Electrical Building - Code 1 Vendor Data
	E14771	20014-E-5500 - Electrical Building - Receive Vendor Data Books	0d	18-Mar-27			◆ 20014-E-5500 - Electrical Building - Receive Vendor Data
	F14781	20014-E-5500 - Electrical Building - Review Vendor Data Books	10d	18-Mar-27	31-Mar-27		20014-E-5500 - Electrical Building - Review Vendor Data
	F14791	20014-E-5500 - Electrical Building - Issue Approval for Payment	D0		31-Mar-27		◆ 20014-E-5500 - Electrical Building - Issue Approval for P
	20014-E-5500: 5k		80d	01-Dec-26	31-Mar-27		🛶
	E0001421	20014-E-5500 - 5kV Variable Speed Drive - Prepare RFP Data Sheets & Specifica	20d	01-Dec-26	05-Jan-27		20014-E-5500 - 5kV Variable Speed Drive - Prepare RFP [
	E0001431	20014-E-5500 - 5kV Variable Speed Drive - Receive Vendor Info	0d	20-Jan-27			20014-E-5500 - 5kV Variable Speed Drive - Receive Vend
	E0001441	20014-E-5500 - 5kV Variable Speed Drive - Review Vendor Data	10d	20-Jan-27	02-Feb-27		20014-E-5500 - 5kV Variable Speed Drive - Review Vend
	E0001451	20014-E-5500 - 5kV Variable Speed Drive - Code 1 Vendor Data	DO		02-Feb-27		20014-E-5500 - 5kV Variable Speed Drive - Code 1 Vend
	E0001461	20014-E-5500 - 5kV Variable Speed Drive - Receive Vendor Data Books	DQ	18-Mar-27	02.100.21		◆ 20014-E-5500 - 5kV Variable Speed Drive - Receive Ver
	E0001101	20014-E-5500 - 5kV Variable Speed Drive - Review Vendor Data Books	10d	18-Mar-27	31_Mar_27		1 20014-E-5500 - 5kV Variable Speed Drive - Review Ver
	E0001001	20014-E-5500 - 5kV/Variable Speed Drive - Issue Approval for Payment	Dd		31_Mar_27		◆ 20014-F-5500 - 5kV Variable Speed Drive - Issue Appro
	20014-E-5501: U	ninterruntable Rower Supply (IIRS)	204	01 Dec 26	21 Mar 27		
	E0002121	20014-E-5501 - Uninterruptable Power Supply - Prepare RFP Data Sheets & Spe	20d	01-Dec-26	05-Jan-27		20014-E-5501 - Uninterruptable Power Supply - Prepare R
	E0002131	20014-E-5501 - Uninterruptable Power Supply - Receive Vendor Info	0d	20-Jan-27		-	20014-E-5501 - Uninterruptable Power Supply - Receive V
	E0002141	20014-E-5501 - Uninterruntable Power Supply - Review Vendor Data	10d	20- Jan-27	02-Eeb-27		20014-E-5501 - Uninterruptable Power Supply - Review V
	E0002151	20014-E-5501 - Uninterruntable Power Supply - Code 1 Vendor Data	D0	20 0011 21	02-Feb-27		▶ 20014-E-5501 - Uninterruptable Power Supply - Code 1 V
	E0002101	20014-E-5501 - Uninterruntable Power Supply - Beceive Vendor Data	04	18_Mar_27	02-1 00-27		♦ 20014-E-5501 - Uninterruptable Power Supply - Receive
	E0002101	20014-E-5501 - Uninterruntable Power Supply - Receive Vendor Data Books	104	18_Mor 27	31_Mor 27		1 20014-F-5501 - Uninterruntable Power Supply - Review
	E0002171	20014 E 5501 Uninterruptable Power Supply - Review Verluor Data BOOKS	10u		21 Mar 07		▲ 20014-F-5501 - Uninterruntable Power Supply Review
				04 D00	31-IVIAF-27		
	E0002241	20014-E-5502 - Transformers - Prenara RED Data Shoots & Specifications	204	01-Dec-26	31-Mar-27	┩╴╴╷╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴	20014-E-5502 - Transformers - Prepare RFP Data Sheets
	E0002241	20014 E 5502 Transformers - Propaie INF Data Silvets & Specifications	200	20 100 27	03-Jan-27		20014-F-5502 - Transformers - Receive Vendor Info
	E0002251		Uu	20-Jan-27			

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 25 01 75		

SOLARIS

Revision	Checked	Approved



Activ	ity ID	Activity Name	Rem.	Start	Finish	020	2021	2022	2	023	202	4	2025		2026		2027	20)28	202	29	2030	2031
			Duration			QQ					QQ	QQ		QQ	QQO	QQ		QQ	QQ	QQ			
	E0002261	20014-E-5502 - Transformers - Review Vendor Data	10d	20-Jan-27	02-Feb-27				-								0014-E-	5502 -	Trans	formers	s - Review	Vendo	Data
	E0002271	20014-E-5502 - Transformers - Code 1 Vendor Data	0d		02-Feb-27											◆ 2	0014-E-t	502 -	Irans	formers	s - Code 1	Vendo	Data
	E0002281	20014-E-5502 - Transformers - Receive Vendor Data Books	0d	18-Mar-27												•	20014-	-5502	2 - Iran	storme	rs - Rece	ve Venc	lor Data Bo
	E0002291	20014-E-5502 - Transformers - Review Vendor Data Books	10d	18-Mar-27	31-Mar-27											0	20014-	E-5502	2 - Trar	nstorme	ers - Revie	w Vend	or Data Bo
	E0002301	20014-E-5502 - Transformers - Issue Approval for Payment	0d		31-Mar-27											•	20014-F	E-5502	2 - Trar	nsforme	ers - Issue	Approv	al for Paym
	20014-E-5503: S	Switchgear & LV MCC	80d	01-Dec-26	31-Mar-27													500	OB -F-			Dana	
	E0002001	20014-E-5503 - Switchgear & LV MCC - Prepare RFP Data Sheets & Specificatior	20d	01-Dec-26	05-Jan-27	_			-								JU14-E-5	503 - :	Switch	gear &		Prepar	BREP Data
	E0002011	20014-E-5503 - Switchgear & LV MCC - Receive Vendor Info	0d	20-Jan-27												● 2	0014-E-5	-503 -	Switch	igear &		- Receiv	e Vendor Ir
	E0002021	20014-E-5503 - Switchgear & LV MCC - Review Vendor Data	10d	20-Jan-27	02-Feb-27							-					0014-E-t	503 -	Switch	ngear ð		- Revie	w Vendor D
	E0002031	20014-E-5503 - Switchgear & LV MCC - Code 1 Vendor Data	0d		02-Feb-27			 							· · · ·	◆ 2	.0014-E-F	5503 -	- Switch	ngear &	LV MCC	- Code	1 Vendor D
	E0002041	20014-E-5503 - Switchgear & LV MCC - Receive Vendor Data Books	0d	18-Mar-27					-							•	20014-E	-5503	3 - Swite	chgear	& LV MC	C - Rece	vive Vendor
	E0002051	20014-E-5503 - Switchgear & LV MCC- Review Vendor Data Books	10d	18-Mar-27	31-Mar-27											0	20014-	E-5503	3 - Swit	tchgea	r & LV MC	C-Revi	ew Vendor
	E0002061	20014-E-5503 - Switchgear & LV MCC - Issue Approval for Payment	0d		31-Mar-27											•	20014-	E-5503	3 - Swi	tchgea	r & LV MC	C - Issu	e Approval
	20014-E-5504: E	Electrical Bulks	80d	01-Dec-26	31-Mar-27				1										<u> </u>				
	E0000931	20014-E-5504 - Electrical Bulks - Prepare RFP Data Sheets & Specifications	20d	01-Dec-26	05-Jan-27		+	 			1 1		1 1 1 1		1 1 1 1	20)014-E-5	504 - I	Electric	al Bulk	s - Prepa	eR⊦PI	Jata Sheets
	E0000941	20014-E-5504 - Electrical Bulks - Receive Vendor Info	0d	20-Jan-27												◆ 2	0014-E-5	-504 -	Electri	cal Bull	(s - Recei	/e Vend	or Info
	E0000951	20014-E-5504 - Electrical Bulks - Review Vendor Data	10d	20-Jan-27	02-Feb-27				-							0 2	.0014-E-8	5504 -	Electri	ical Bul	ks - Revie	w Vendo	or Data
	E0000991	20014-E-5504 - Electrical Bulks - Code 1 Vendor Data	0d		02-Feb-27											♦ 2	.0014-E-5	5504 -	Electri	ical Bul	ks - Code	1 Vendo	or Data
	E0001001	20014-E-5504 - Electrical Bulks - Receive Vendor Data Books	0d	18-Mar-27												•	20014-E	-5504	l - Elec	trical B	ulks - Rec	eive Ver	ıdor Data B
	E0001011	20014-E-5504 - Electrical Bulks - Review Vendor Data Books	10d	18-Mar-27	31-Mar-27											0	20014-	E-5504	4 - Elec	ctrical B	ulks - Rev	view Ven	dor Data B
	E0001021	20014-E-5504 - Electrical Bulks - Issue Approval for Payment	0d		31-Mar-27											•	20014-	<u>-550</u> 2	4 - Elec	ctrical B	ulks - Issu	ie Appro	val for Payr
	Instrumentation E	ngineering & Design	200d	30-Jun-26	22-Apr-27											-	1						
	Instrument Index		100d	25-Sep-26	24-Feb-27				-								umontin		EU				
	E0001551	Instrument Index - IFH	20d	25-Sep-26	23-Oct-26	_										insu		lex - II					
	E0004541	Instrument Index - IFC	30d	13-Jan-27	24-Feb-27			 									nstrumer	ninae	- 1⊢C	,			
	System Block Di	agram	30d	13-Jan-27	24-Feb-27				-								Svistem F	lock F)iadran	n-IFC			
	E0004891		300	13-Jan-27	24-Feb-27														lagian				
	E0004911	Instrument Lavout - IEC	15d	01-Apr-27	22-Apr-27	_											ı Instrun	ient La	avout -	IFC			
	Tie-in		30d	13- Jan-27	22-Api-27																		
	E0004931	I&C Tie-in Drawing - IFC	30d	13-Jan-27	24-Feb-27			 									I&C Tie-ir	Draw	/ing - IF	C			
	Metering Schema	atics	100d	25-Sep-26	24-Feb-27				-														
	E0004941	Metering Schematics - IFH	20d	25-Sep-26	23-Oct-26										į 📋	Mete	ring Sch	ematic	≍s-IF⊢	•			
	E0004951	Metering Schematics - IFC	30d	13-Jan-27	24-Feb-27				-								Metering	Scher	matics	IFC			
	Fire & Gas Detec	ction Drawings	100d	25-Sep-26	24-Feb-27										-								
	E0004981	Fire & Gas Detection Drawings - IFH	20d	25-Sep-26	23-Oct-26								1 1			Fire	& Gas Dr	etectio	n Drav	vings -	IFH		
	E0004991	Fire & Gas Detection Drawings - IFC	30d	13-Jan-27	24-Feb-27												Fire & Ga	is Detr	ection	Drawin	gs - IFC		
	Architectural Dia	igram	100d	25-Sep-26	24-Feb-27										▼								
	E0005001	PCS/PLC/SIS/SCADA Architecture Diagram - IFH	20d	25-Sep-26	23-Oct-26											PCS	/PLC/SIS	,/SCAI	DAArc	hitectui	re Diagrar	n-IFH	
	E0005011	PCS/PLC/SIS/SCADA Architecture Diagram - IFC	30d	13-Jan-27	24-Feb-27												PCS/PLC	;/SIS/S	SCADA	Archit	ecture Dia	gram - I	FC
	Standard Drawin	gs & Details	30d	30-Jun-26	12-Aug-26											~			D .4 "				
	E0005031	Standard Drawings & Details - IFC	30d	30-Jun-26	12-Aug-26				1							stande	rd Drawi	ngs &	Details	s-⊪C			
	Specifications		30d	30-Jun-26	12-Aug-26											200-	ocificatio	ne IF	C				
	E0005051	I&C Specifications - IFC	30d	30-Jun-26	12-Aug-26					1 1						sy sp	cuicatio	ידו - כו	<u> </u>				
	Remaining Lev	rel of Effort Actual Work Critical Remaining Work			Data Data 1		1				Da	ate				Revisi	on				Checked	Ap	proved
	Actual Level of	Effort Remaining Work E			Data Date:2 Pana: 26	of 73	4				19-Aug-	24	IFI										
					1 490. 20	0.75																	

SOLARIS



Actual Level of Effort

Remaining Work Critical Secondary

Tilbury Island LNG Facility Expansion (3 BCF) TLSE Sub-Projects Integrated Schedule-Detailed

ty ID	Activity Name	Rem.	Start	Finish	020 20	21	2022	202	23	2024		2025	2026		2027	202	28	202	9	2030	20
		Duration							QQ		Q			<u>≀ Q Q (</u>	2 Q Q	QQ	QQ	QQ		≀ Q Q	
Alarm/Set Poin	nt Table	100d	25-Sep-26	24-Feb-27						·				Alarn	n Set Po	int Tabl	e - IFH				
E0001571		200	25-Sep-26	23-Oct-26											Marm Sc	t Doint		IFC			
E0001901	Alarm Set Point Table - IFC	30d	13-Jan-27	24-Feb-27				1 1						F	Nami Se	L FOILIC		IFC			
Shutdown Key		90d	27-Aug-26	12-Jan-27										Shut [4				
E0001531	Shut Down Key - IFH	200	27-Aug-26	24-Sep-26												y - 11 1 1					
E0001881	Shut Down Key - IFC	30d	24-Nov-26	12-Jan-27											iul Dowi	i key -	IFC				
Material Take (30d	10-Feb-27	24-Mar-27											Inotrum	Intotio		atrala 1			
E0001601	Instrumentation & Controls MIO	30d	10-Feb-27	24-Mar-27											Insuume	FILAUOI					
Programming	De marche De de m	10d	25-Feb-27	10-Mar-27											Program	mina E	Packadu	_			
E0001521	Programming Package	10d	25-Feb-27	10-Mar-27											logian		acrage				
Requisition		135d	15-Jul-26	02-Feb-27							·		 								
E0001271	20014- L5600 - ESDV - Prenare Instrument Data Sheet - REP	450 20d	15-Jul-26	17-Sep-26										20014-	J-5600 -	ESDV	- Prepa	are Ins	trument	t Data {	Sheet - R
E0001271	20014 J E600 ESDV Propare instrument Data Sheet - Ni h	200	02 Son 26	12-Aug-20										20014	1-5600	- ESD\	/-Rec	eive Vr	endorlin	fo	
E0004341		00	03-Sep-20	47.0										2001/					ndor Da	ata	
E0004531		TUC	03-Sep-26	17-Sep-26										a 20044	1-5000					nici oto	
E0004711	20014-J-5600 - ESDV - Code 1 Vendor Data	0d		17-Sep-26	· · · · · · · · · · · · · · · · · · ·					· · · · · · · · · · · · · · · · · · ·		· · · · ·		■ 20014	-J-2000	- E9D1	7 - Coa	eive	ndor Da	na	
20014-J-5601	I: Orifice Plate Meter	40d	01-Dec-26	02-Feb-27											014 156	201· Or	ifico Dic	ata Ma	tor Dro	noro Ir	setrumon
E0003381	20014-J-5601: Ornice Plate Meter - Prepare Instrument Data Sheet - RFP	150	01-Dec-26	21-Dec-26											3044 16						Suumeni Voodor k
E0004131	20014-J-5601: Orifice Plate Meter - Receive Vendor Info	0d	20-Jan-27											• 20	JU14-J-5					ceive v	
E0006391	20014-J-5601: Orifice Plate Meter - Review Vendor Data	10d	20-Jan-27	02-Feb-27										0 20	J014-J-t	5601: C		iate ivi	eter - R	eview v	/endor D
E0006401	20014-J-5601: Orifice Plate Meter - Code 1 Vendor Data	0d		02-Feb-27										♦ 20	J014-J-5	5601: C)rifice P	late M	eter - C	ode 1 V	/endor D
20014-J-5602	2 - Flow Control Valve	45d	15-Jul-26	17-Sep-26											1 5000	___	S	Valua	Deserves		
E0003201	20014-J-5602 - Flow Control Valve - Prepare Instrument Data Sheet - RFP	20d	15-Jul-26	12-Aug-26										20014-	J-5602 -	FIOWC	ontrol	vaive -	- Prepar	einstru	Iment Da
E0003911	20014-J-5602 - Flow Control Valve - Receive Vendor Info	0d	03-Sep-26											▶ 20014	-J-5602	- Flow (Control	Valve	- Recei	ve Vend	dor Info
E0005581	20014-J-5602 - Flow Control Valve - Review Vendor Data	10d	03-Sep-26	17-Sep-26										20014	-J-5602	- Flow	Contro	I Valve	- Revie	w Venc	lor Data
E0005711	20014-J-5602 - Flow Control Valve - Code 1 Vendor Data	0d		17-Sep-26										♦ 20014	-J-5602	- Flow	Contro	I Valve	- Code	1 Venc	lor Data
20014-J-5602	2 - PSV's	40d	01-Dec-26	02-Feb-27														1 1	1		
E0003261	20014-J-5602 - PSV's - Prepare Instrument Data Sheet - RFP	15d	01-Dec-26	21-Dec-26										200)14-J-56	\$02 - P\$	SV's - P	repar	e Instrur	nent Da	ata Shee
E0004241	20014-J-5602 - PSV's - Receive Vendor Info	0d	20-Jan-27											♦ 20)014-J-5)602 - F	²SV's -	Recei	ve Vend	or Info	
E0006331	20014-J-5602 - PSV's - Review Vendor Data	10d	20-Jan-27	02-Feb-27										l 2	0014-J-\$	5602 - I	PSV's -	Revie	w Vend	or Data	a l
E0006341	20014-J-5602 - PSV's - Code 1 Vendor Data	0d		02-Feb-27										♦ 2′	0014-J-	5602 - I	PSV's -	Code	1 Vend	or Data	a l
20014-J-56XX	X: Fire/Gas Detedtion System	20d	01-Dec-26	05-Jan-27										₩						1	
E0003501	20014-J-56XX - Fire/Gas Detection System - Prepare Instrument Data Sheet - RF	20d	01-Dec-26	05-Jan-27										a 20	014-J-5	6XX - F	ire/Ga	s Dete	ction Sy	stem -	Prepare
Procurement		340d	30-Jun-26	15-Nov-27									-	÷ ; ; ;							
Civil/Structural/	/Architectural	40d	30-Jun-26	27-Aug-26									-	1							
20014-C-5302:	: Galvanized Steel	20d	30-Jun-26	28-Jul-26																	
E0002911	20014-C-5302 - Galvanized Steel - Refresh/Renegotiate	20d	30-Jun-26	28-Jul-26										20014-(-5302 -	Galvar	nzed S	teel - F	ketresh/	Reneg	otiate
E0002921	20014-C-5302 - Galvanized Steel - Develop MRP	20d	30-Jun-26	28-Jul-26										20014-0	-5302 -	Galvar	nized S	teel - [Jevelop	MRP	
E0002931	20014-C-5302 - Galvanized Steel - Issue PO to Vendor	0d		28-Jul-26									•	20014-0	2-5302 -	Galvar	nized S	teel - I	ssue PC) to Ver	ndor
20014-C-5303:	: Concrete	40d	30-Jun-26	27-Aug-26									-	/							
E0002971	20014-C-5303 - Concrete - Refresh/Renegotiate	20d	30-Jun-26	28-Jul-26										20014-0	2-5303 -	Concre	ete - Re	efresh/	Renego	otiate	
E0002981	20014-C-5303 - Concrete - Develop MRP	20d	30-Jun-26	28-Jul-26										20014-0	2-5303 -	Concr	ete - De	evelop	MRP		
E0002991	20014-C-5303 - Concrete - Issue PO to Vendor	0d		28-Jul-26									•	20014-0	2-5303 -	Concr	ete - I\$:	sue PC) to Ven	dor	
E0006301	20014-C-5303 - Concrete - Manufacture	15d	29-Jul-26	19-Aug-26										20014-	C-5303	- Conc	rete - IV	lanufa	cture		
•			1		<u> </u>					1 1		1 1 1		1 1	<u> </u>	<u> </u>					
	evel of Effort Actual Work Critical Remaining Work			Data Data - 2	6 101 24					Dat	e			Revisio	n			С	hecked		Approved
A ()					o-Jui-24				1	9-Aua-24	4	IFI									

Page: 27 of 73

SOLARIS

Revision	Checked	Approved



Activi	ty ID	Activity Name	Rem.	Start	Finish	020	2021	2022	2023	2024	2025	2026	2027	2028	2029	20	030	2031
	F0002004		Duiation		10 Aug 00										QQQQ - Readvite S			
	E0003001	20014-C-5303 - Concrete - Ready to Snip	Ud Od		19-Aug-26	_						↓ 2	014-0-5303	- Concrete	- FTA to Site	n ilp		
	E0003011	20014-C-5303 - Concrete - ETAto Site	0d	07 Aug 06	26-Aug-26						·	▼ 2 ▲ W	/BS-001 Rec	ive at Site	Concrete & F	oundat	ion Matei	rials
	20014 C 52XX 5	WBS-001 Receive at Site Concrete & Foundation Materials	Ud	27-Aug-26	27 4117 26											oundat		IIdio
	E0003031	20014-C-53XX - Fill Materials - Refresh/Renegotiate	20d	30-Jun-26	27-Aug-26							20	014-C-53XX	- Fill Materia	als - Refresh/	Renego	otiate	
	E0003041	20014-C-53XX - Fill Materials - Develop MRP	20d	30-Jun-26	28-Jul-26							 20	014-C-53XX	- Fill Materia	als - Develop	MRP		
	E0003051	20014-C-53XX - Fill Materials - Issue PO to Vendor	0d	00 0011 20	28-Jul-26							♦ 20	014-C-53XX	- Fill Materia	als - Issue PC) to Ven	dor	
	E0006311	20014-C-53XX - Fill Materials - Manufacture	15d	29-Jul-26	19-Aug-26		· + + + + + + + + + + +				· · · · · · · · · · · · · · · · · · ·	∎ 20	0014-C-53XX	- Fill Mater	ials - Manufa	cture		· - +
	E0003061	20014-C-53XX - Fill Materials - Ready to Ship	0d		19-Aug-26							♦ 20	0014-C-53XX	- Fill Mater	ials - Ready t	o Ship		
	E0003071	20014-C-53XX - Fill Materials - ETA to Site	0d		26-Aug-26							♦ 2	0014-C-53XX	- Fill Mater	ials - ETA to S	Site		
	E0008981	WBS-001 Receive at Site Fill Materials	0d	27-Aug-26								♦ W	/BS-001 Rece	eive at Site	Fill Materials			
	Mechanical		245d	30-Jun-26	28-Jun-27													
	20014-M-5103: I	nstrument Air Package	235d	30-Jun-26	14-Jun-27													
	E0000501	20014-M-5103 - Instrument Air Package - Refresh/Renegotiation	20d	30-Jun-26	28-Jul-26							□ 20	014-M-5103	Instrumen	tAir Package	e - Refre	sh/Rene	gotiation
	E0000551	20014-M-5103 - Instrument Air Package - Develop MRP	20d	30-Jun-26	28-Jul-26							1 20	014-M-5103	Instrumen	t Air Package	e - Deve	lop MRP	
	E0000561	20014-M-5103 - Instrument Air Package - Issue PO to Vendor	Od		28-Jul-26							♦ 20	014-M-5103	Instrumen	t Air Package	e - Issue	PO to Ve	endor
	E0006041	20014-M-5103 - Instrument Air Package - Manufacture	200d	29-Jul-26	20-May-27								20014	4-M-5103 -	Instrument A	ir Packa	ge - Mar	lufacture
	E0005821	20014-M-5103 - Instrument Air Package - Testing & Inspection	2d	21-May-27	25-May-27								I 2001	4-M-5103 -	Instrument A	ir Packa	ige - Tesl	ting & Ins
	E0000571	20014-M-5103 - Instrument Air Package - Ready to Ship	0d		28-May-27								♦ 2001	4-M-5103 -	Instrument A	vir Packa	age - Rea	ady to Sh
	E0000631	20014-M-5103 - Instrument Air Package - ETA to Site	0d		11-Jun-27								♦ 2001	4-M-5103	- Instrument /	Air Packa	age - ETA	A to Site
	E15091	WBS-001 Receive at Site - Instrument Air Package	Od	14-Jun-27									♦ WBS	-001 Rece	ive at Site - Ir	nstrumei	nt Air Pac	kage
	20014-M-5104: N	litrogen Generation (PSA) Package	245d	30-Jun-26	28-Jun-27													
	E0000971	20014-M-5104 - Nitrogen Generation (PSA) Package - Refresh/Renegotiation	20d	30-Jun-26	28-Jul-26							1 20	014-M-5104	Nitrogen (Generation (F	PSA) Pa	ckage - F	Refresh/F
	E0000981	20014-M-5104 - Nitrogen Generation (PSA) Package - Develop MRP	20d	30-Jun-26	28-Jul-26							1 20	014-M-5104	Nitrogen (Generation (F	PSA) Pa	ckage - D	Jevelop I
	E0001041	20014-M-5104 - Nitrogen Generation (PSA) Package - Issue PO to Vendor	0d		28-Jul-26							♦ 20	014-M-5104	Nitrogen C	Generation (F	PSA) Pa	ckage - I	ssue PO
	E0006051	20014-M-5104 - Nitrogen Generation (PSA) Package - Manufacture	200d	29-Jul-26	20-May-27								20014	4-M-5104 -	Nitrogen Ge	neration	(PSA) P	'ackage
	E0005831	20014-M-5104 - Nitrogen Generation (PSA) Package - Testing & Inspection	2d	21-May-27	25-May-27								I 2001	4-M-5104 -	Nitrogen Ge	neration	ı (PSA) P	'ackage
	E0001051	20014-M-5104 - Nitrogen Generation (PSA) Package - Ready to Ship	0d		28-May-27								♦ 2001	4-M-5104 -	Nitrogen Ge	neratior	n (PSA) F	[•] ackage
	E0001061	20014-M-5104 - Nitrogen Generation (PSA) Package - ETA to Site	0d		25-Jun-27								♦ 2001	14-M-5104	- Nitrogen G	eneratio	n (PSA)	Package
	E15071	WBS-001 Receive at Site - Nitrogen Generation (PSA) Package	Od	28-Jun-27									♦ WBS	5-001 Rece	eive at Site - N	litrogen	Generat	tion (PSA
	20014-M-51XX: I	NG Drain Vessel (With Emerson Heater)	245d	30-Jun-26	28-Jun-27													
	E14871	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Refresh/Renegot	iatic 20d	30-Jun-26	28-Jul-26							□ 20	014-M-51XX	- LNG Drai	n Vessel (Wit	h Emers	son Heat	er) - Refi
	E14881	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Develop MRP	20d	30-Jun-26	28-Jul-26							□ 20	014-M-51XX	- LNG Drai	n Vessel (Wit	h Emers	son Heat	er) - Dev
	E14891	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Issue PO to Vend	lor Od		28-Jul-26							♦ 20	014-M-51XX	- LNG Drai	n Vessel (Wit	h Emers	son Heat	er) - Issu
	E14931	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Manufacture	200d	29-Jul-26	20-May-27								20014	4-M-51XX -	LNG Drain \	/essel (\	With Eme	rson He
	E14921	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Testing & Inspecti	on 2d	21-May-27	25-May-27								I 2001	4-M-51XX -	LNG Drain	Vessel (\	Nith Eme	erson He
	E14901	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - Ready to Ship	Od		28-May-27								♦ 2001	4-M-51XX ·	- LNG Drain '	Vessel (With Eme	erson He
	E14911	20014-M-51XX - LNG Drain Vessel (With Emerson Heater) - ETA to Site	0d		25-Jun-27								♦ 200 ²	14-M-51XX	- LNG Drain	Vessel	(With Em	ierson H
	E15121	WBS-001 Receive at Site - LNG Drain Vessel (With Emerson Heater)	0d	28-Jun-27									♦ WBS	\$-001 Rece	eive at Site - L	NG Dra	in Vesse	l (With E
	Piping		235d	01-Dec-26	15-Nov-27													
	20014-L-5201: S		20d	01-Dec-26	05-Jan-27								20014	201 Stain	loss Stool Dir		och/Doc	onotioto
	E0002471	20014-L-5201 - Stainless Steel Pipe - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27	_							20014-L-5	201 - Siain				egoliale
	E0002481	20014-L-5201 - Stainless Steel Pipe - Develop MRP	20d	01-Dec-26	05-Jan-27								20014-L-5	∠ui - Stalh	iess Steel Pl	e - Dev		г
		vel of Effort Actual Work Critical Remaining Work			Data Data 2		4			Date		I	Revision		Chec	ked	Appro	oved
	Actual Level of	Effort Effort Remaining Work				of 73	+		1	9-Aug-24	IFI							
		; ,			. 490. 20	5.70												

SOLARIS



Activity	<i>i</i> ID	Activity Name	Rem. Duration	Start	Finish				2025	2026	2027	2028	2029	2030	2031
	F0002491	20014-L-5201 - Stainless Steel Pine - Issue PO to Vendor	b0		05-lan-27						20014-L-5	201 - Stainle	ss Steel Pipe	- Issue P	J to Vendor
	20014-L-5202: Car		204	01 Dec 26	05-Jan 27						, , , , , , , , , , , , , , , , , , , ,				
	E0002391	20014-L-5202 - Carbon Steel Pipe - Refresh/Renegotiate	20d	01-Dec-20	05-Jan-27						20014-L-5	202 - Carbo	n Steel Pipe -	Refresh/F	Renegotiate
	F0002401	20014-I -5202 - Carbon Steel Pipe - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	202 - Carbo	n Steel Pipe -	Develop I	MRP
	F0002411	20014-I -5202 - Carbon Steel Pine - Issue PO to Vendor	 b0		05-Jan-27						20014-L-5	202 - Carbo	n Steel Pipe -	Issue PO	to Vendor
	20014-I -5203: Sta	inless Steel Fittings	125d	01-Dec-26	07- lun-27										
	E0002711	20014-L-5203 - Stainless Steel Fittings - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27						20014-L-5	203 - Stainle	ss Steel Fittin	gs - Refre	sh/Renegotia
	E0002721	20014-L-5203 - Stainless Steel Fittings - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	203 - Stainle	ss Steel Fittin	gs - Deve	lop MRP
	E0002731	20014-L-5203 - Stainless Steel Fittings - Issue PO to Vendor	0d		05-Jan-27						20014-L-5	203 - Stainle	ss Steel Fittin	gs - Issue	PO to Vendo
	E0006111	20014-L-5203 - Stainless Steel Fittings - Manufacture	100d	06-Jan-27	28-Mav-27						2001	I-L-5203 - S	tainless Steel	Fittings - I	Manufacture
	E0002741	20014-L-5203 - Stainless Steel Fittings - Ready to Ship	0d		28-May-27						♦ 2001	1-L-5203 - S	tainless Steel	Fittings - I	Ready to Shir
	E0002751	20014-I -5203 - Stainless Steel Fittings - FTA to Site	0d		04-Jun-27						♦ 2001	4-L-5203 - S	tainless Steel	Fittings -	ETA to Site
	F14951	WBS-001 Receive at Site - Stainless Steel Fittings	0d	07-Jun-27							♦ WBS	-001 Receiv	e at Site - Stai	nless Stee	el Fittings
	20014-L-5204: Ca	rbon Steel Fittings	125d	01-Dec-26	07-Jun-27										Ŭ
	E14511	20014-L-5204 - Carbon Steel Fittings - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27		· · · · · · · · · · · · · · · · · · ·				20014-L-5	204 - Carbo	n Steel Fitting	s - Refres	h/Renegotiat
	E14521	20014-L-5204 - Carbon Steel Fittings - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	204 - Carbo	n Steel Fitting	s - Develo	p MRP
	E14531	20014-L-5204 - Carbon Steel Fittings - Issue PO to Vendor	0d		05-Jan-27						20014-L-5	204 - Carbo	n Steel Fitting	s - Issue F	20 to Vendor
	E14561	20014-L-5204 - Carbon Steel Fittings - Manufacture	100d	06-Jan-27	28-May-27						2001	1-L-5204 - C	arbon Steel F	ittings - M	lanufacture
	E14541	20014-I -5204 - Carbon Steel Fittings - Ready to Ship	0d		28-May-27						♦ 2001	1-L-5204 - C	arbon Steel F	ittings - R	eady to Ship
	E14551	20014-I -5204 - Carbon Steel Fittings - FTA to Site	0d		04-Jun-27				· · · · ·		♦ 2001	4-L-5204 - C	arbon Steel F	ittings - E	TA to Site
	F14961	WBS-001 Receive at Site - Carbon Steel Fittings	0d	07lun-27							♦ WBS	-001 Receiv	e at Site - Car	bon Steel	Fittings
	20014-L-5205: Sta	inless Steel Valves	235d	01-Dec-26	15-Nov-27										0
	E0000621	20014-L-5205 - Stainless Steel Valves - Refresh/Renegotiate	200d	01-Dec-26	05-Jan-27						20014-L-5	205 - Stainle	ss Steel Valve	s - Refre	sh/Renegotia
	E0000681	20014-L-5205 - Stainless Steel Valves - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	205 - Stainle	ss Steel Valve	s - Devel	op MRP
	E0000691	20014-L-5205 - Stainless Steel Valves - Issue PO to Vendor	0d		05-Jan-27					•	20014-L-5	205 - Stainle	ss Steel Valve	s - Issue	PO to Vendor
	E0006091	20014-L-5205 - Stainless Steel Valves - Manufacture	200d	06-Jan-27	21-Oct-27							20014-L-520)5 - Stainless	Steel Valv	es - Manufac
	E0000701	20014-L-5205 - Stainless Steel Valves - Ready to Ship	0d		21-Oct-27						• 2	20014-L-520)5 - Stainless	Steel Valv	es - Ready to
	E0000711	20014-I -5205 - Stainless Steel Valves - FTA to Site	0d		12-Nov-27						•	20014-L-52	05 - Stainless	Steel Val	/es-ETA to S
	F0009021	WBS-001 Receive at Site - Stainless Steel Valves	0d	15-Nov-27							•	WBS-001 F	eceive at Site	- Stainles	s Steel Valve
	20014-L-5206: Ca	rbon Steel Valves	235d	01-Dec-26	15-Nov-27										
	E14571	20014-L-5206 - Carbon Steel Valves - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27						20014-L-5	206 - Carbo	n Steel Valves	- Refrest	1/Renegotiate
	E14581	20014-L-5206 - Carbon Steel Valves - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	206 - Carbo	n Steel Valves	- Develo	p MRP
	E14591	20014-L-5206 - Carbon Steel Valves - Issue PO to Vendor	0d		05-Jan-27						20014-L-5	206 - Carbo	n Steel Valves	s - Issue P	O to Vendor
	E14621	20014-L-5206 - Carbon Steel Valves - Manufacture	200d	06-Jan-27	21-Oct-27							20014-L-520)6 - Carbon S	teel Valve	s - Manufactu
	E14601	20014-L-5206 - Carbon Steel Valves - Ready to Ship	0d		21-Oct-27						• 2	20014-L-520)6 - Carbon S	teel Valve	s - Ready to S
	E14611	20014-L-5206 - Carbon Steel Valves - ETA to Site	Od		12-Nov-27						•	20014-L-52	06 - Carbon S	teel Valve	es - ETA to Sit
	E14971	WBS-001 Receive at Site - Carbon Steel Valves	0d	15-Nov-27							•	WBS-001 F	eceive at Site	- Carbon	Steel Valves
	20014-L-5207: Cr	vogenic Pipe Supports	100d	01-Dec-26	30-Apr-27										
	E0002631	20014-L-5207 - Cryogenic Pipe Supports - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27						20014-L-5	207 - Cryog	enic Pipe Sup	ports - Re	fresh/Reneg
	E0002641	20014-L-5207 - Cryogenic Pipe Supports - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-L-5	207 - Cryog	enic Pipe Sup	ports - De	velop MRP
	E0002651	20014-L-5207 - Cryogenic Pipe Supports - Issue PO to Vendor	0d		05-Jan-27						20014-L-5	207 - Cryog	enic Pipe Sup	ports - Iss	ue PO to Ver
	E0006101	20014-L-5207 - Cryogenic Pipe Supports - Manufacture	75d	06-Jan-27	22-Apr-27						20014	L-5207 - Cr	ogenic Pipe	Supports	- Manufacture
	E0002661	20014-L-5207 - Cryogenic Pipe Supports - Ready to Ship	0d		22-Apr-27						20014	L-5207 - Cr	ogenic Pipe	Supports	- Ready to Sh
					·			Date	: : :	Re	vision		Checke	d	Approved
_	Remaining Leve	el or Eπort Actual Work Critical Remaining Work			Data Date:2	6-Jul-24	 19-Au	g-24	IFI					-	
-	Actual Level of E	_ποπ			Page: 29	of 73		•							

SOLARIS

M A N A G E M E N T CONSULTANTS INC.



Activity	ID	Activity Name	Rem. Duration	Start	Finish)/	.0	2021 2022 2	023 202	4 2025	2026	2027	2028	2029	2030	2031
	E0002671	20014 L 5207 Crucagonia Dina Supporta ETA ta Sita	0d		20 Apr 27							Q Q Q Q C 1-1 -5207 - Ci	vogenic Pine	Supports	- ETA to Site
	E0002071	WPS 001 Receive at Site Cryagonia Dine Supports	b0	20 Apr 27	29-Api-27					+ - + + +	◆ 2001	001 Receive	at Site - Crvo	denic Pine	= Supports
	E 1494 I	wBS-001 Receive at Site - Cryogenic Pipe Supports		30-Apr-27	OF Mar 07								ar one oryo	gernerip	Cuppons
	E0002791	20014-L-5208 - Insulation - Refresh/Renegotiate	20d	01-Dec-26	25-Mar-27						20014-L-	5208 - Insula	tion - Refresh	Reneaoti	ate
	E0002801	20014 - 5208 - Insulation - Develop MRP	20d	01-Dec-26	05- Jan-27						20014-L-	5208 - Insula	tion - Develop	MRP	
	E0002801	20014 L 5208 Insulation - Develop Millin	200 0d	01-Dec-20	05-Jan 27						20014-L-	5208 - Insula	tion - Issue P() to Vend	or
	E0002011	20014-L-5208 - Insulation - Issue FO to vendor	50d	06 lon 27	17 Mor 27						20014-	-5208 - Insi	ulation - Manu	facture	
	E0000121	20014-L-5208 - Insulation - Manufacture	500	00-Jan-27	17-IVIdI-27						▲ 20014-	- 5208 - Insi	ulation - Read	v to Shin	
	E0002821	20014-L-5208 - Insulation - Ready to Ship	bU								▲ 20014	L -5208 - Insi	ulation - FTA to	n Site	
	E0002831	20014-L-5208 - Insulation - ETAto Site	bU	05 Mar 07	24-Mar-27						✓ 2001-	1 Receive	at Site Insulati	n	
	E0009361	WBS-001 Receive at Site Insulation	Ud	25-Mar-27	05.11.07						▼ ₩D3-0				
	20014-E-5500 - E	Sectrical Building	230d	01-Dec-26	05-Nov-27										
	E14661	20014-E-5500 - Electrical Building - Refresh/Renegotiate	20d	01-Dec-20	05-Jan-27						20014-E-	500 - Electr	ical Building - I	Refresh/R	enegotiate
	E14671	20014-F-5500 - Electrical Building - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-E-	5500 - Electr	ical Building - I	Develop N	/RP
	E14681	20014-F-5500 - Electrical Building - Issue PO to Vendor	 Dd		05-Jan-27						◆ 20014-E-	5500 - Electri	ical Building - I	ssue PO	to Vendor
	E14721	20014-E-5500 - Electrical Building - Manufacture	150d	06- Jan-27	10-Aug-27						20	014-E-5500	- Electrical Bu	ulding - Ma	anufacture
	E14721	20014-E-5500 - Electrical Building - Testing & Inspection	2d	11-Δug-27	12-Aug-27					$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	1 20	014-E-5500	- Electrical Bu	ildina - Te	stina & Inspec
	E14691	20014-E-5500 - Electrical Building - Ready to Shin	2d 0d	117/09/27	17-Aug-27						♦ 20	014-E-5500	- Electrical Bu	uildina - Re	eadv to Ship
	E14031	20014-E-5500 - Electrical Building - FTA to Site	b0 b0		15-Sep-27						• 2	0014-E-550	0 - Electrical B	uildina - E	TA to Site
	E15021	W/BS 001 Receive at Site Electrical Building	DO	16 Son 27	10-0ep-27							VBS-001 Re	ceive at Site -	Flectrical	Building
	20014 E 5500 5	WDS-001 Neceive at Site - Liectlical Building	1504	01 Dec 26	12 101 27						· .				Dallallig
	E0001951	20014-F-5500 - 5kV Variable Speed Drive - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27						20014-E-	5500 - 5kV V	ariable Speed	Drive - R	efresh/Renec
	E0001961	20014-E-5500 - 5kV Variable Speed Drive - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-E-	5500 - 5kV V	ariable Speed	Drive - D	evelop MRP
	E0001971	20014-E-5500 - 5kV Variable Speed Drive - Issue PO to Vendor	0d	01 200 20	05-lan-27						● 20014-E-	5500 - 5kV V	ariable Speed	Drive - Is	sue PO to Ve
	E0006131	20014-E-5500 - 5kV Variable Speed Drive - Manufacture	105d	06- Jan-27	04- lun-27						200	4-E-5500 - 5	5kV Variable S	beed Driv	/e - Manufactı
	E0005841	20014-E-5500 - 5kV Variable Speed Drive - Testing & Inspection	2d	07- lun-27	08- lun-27						1 200	14-E-5500 -	5kV Variable S	Speed Driv	ve - Testina &
	E0001081	20014-E-5500 - 5kV Variable Speed Drive - Ready to Ship	2d 0d	07 001 27	11_ lun_27						♦ 200	14-E-5500 - :	5kV Variable S	speed Driv	ve - Readv to
	E0001901	20014 E 5500 - 5kV Variable Speed Drive - Ready to Ship	DO		12 101 27						▲ 200)14-E-5500 -	5kV Variable	Speed Dr	rive - FTA to S
	E14091	W/PS 001 Passive at Site 54/ (Variable Speed Drive - LTAto Site		12 101 07							▲ WF	S-001 Rece	ive at Site - 5k	V Variable	Speed Drive
	20014 E 5501		1754		19 Aug 27										
	E0002191	20014-E-5501 - Uninterruptable Power Supply - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27						20014-E-	5501 - Uninte	rruptable Pov	ver Suppl	y - Refresh/R
	E0002201	20014-E-5501 - Uninterruptable Power Supply - Develop MRP	20d	01-Dec-26	05-Jan-27						20014-E-	5501 - Uninte	erruptable Pov	ver Suppl	y - Develop M
	E0002211	20014-E-5501 - Uninterruptable Power Supply - Issue PO to Vendor	0d	01 200 20	05-Jan-27						• 20014-E-	5501 - Uninte	erruptable Pov	ver Suppl	y - Issue PO t
	E0006141	20014-E-5501 - Uninterruntable Power Supply - Manufacture	120d	06- Jan-27	25- lun-27						200	14-E-5501 -	Uninterruptab	le Power	, Supply - Man
	E0005851	20014-E-5501 - Uninterruptable Power Supply - Testing & Inspection	2d	28- lun-27	20 Jun-27						200	14-E-5501 -	Uninterruptak	le Power	Supply - Testi
	E0003031	20014-E-5501 - Uninterruptable Power Supply - Ready to Ship	2d 0d	20-001-27	05- Jul-27						♦ 200)14-E-5501 -	Uninterruptal	le Power	Supply - Rea
	E0002221	20014 E 5501 - Uninterruptable Power Supply - Ready to Ship									▲ 20	014-F-5501	- Uninterrupt	able Powe	er Supply - FT
	E0002231	WPS 001 Receive at Site Uninterruptable Power Supply - ETAto Site	D0	19 Aug 07	17-Aug-27						◆ _ \	/BS-001 Rec	eive at Site - I	Ininterrun	table Power S
	E 1500 I	WBS-001 Receive at Site - Onlinentuptable Power Supply	DU	10-Aug-27	OF Nev 07										
	E0002311	20014-F-5502 - Transformers - Refresh/Renegotiate	2300 20d	01-Dec-26	05-Jan-27						20014-E-	5502 - Trans	formers - Refr	esh/Rene	gotiate
	E0002321	20014-E-5502 - Transformers - Develop MRP	20d	01-Dec-26	05-lan-27						20014-E-	5502 - Trans	formers - Dev	elop MRP	
	E0002321	20014 E 0002 - Italisionneis - Develop With 20014-E-5502 - Transformers - Issue PO to Vendor	200 0d	01-060-20	05- Jan-27						■ 20014-F-	5502 - Trans	formers - Issu	e PO to V	endor
	E0002331	20014 E 5502 Transformers Manufacture	2004	06 Jan 27	21 Oct 27							20014-E-55	02 - Transforn	ners - Mar	nufacture
	L0000131		2000	00-Jai 1-21	21-06-21										
-	Remaining Lev	el of Effort Actual Work Fitical Remaining Work			Data Date:26-	Jul-24			ate	R	levision		Checke	d	Approved
-	Actual Level of	Effort Effort Remaining Work E			Page: 30 o	73		19-Aug-							

SOLARIS



Activ	ity ID	Activity Name	Rem. Duration	Start	Finish		2024	2025			2028	2029	2030 2031
	F0005861	20014-F-5502 - Transformers - Testing & Inspection	2d	22-Oct-27	25-Oct-27)14-E-550;	2 - Transformer	s - Testing & Inspec
	E0002341	20014-F-5502 - Transformers - Ready to Ship	0d	22 00021	28-Oct-27					♦ 200)14-E-550:	2 - Transformei	s - Ready to Ship
	F0002351	20014-F-5502 - Transformers - FTA to Site	0d		04-Nov-27					♦ 20)14-E-550	2 - Transforme	rs - ETA to Site
	E15011	WBS-001 Receive at Site - Transformers	0d	05-Nov-27						♦ WE	3S-001 Re	ceive at Site - 1	ransformers
	20014-E-5503 - S	Switchgear & LV MCC	165d	01-Dec-26	04-Aug-27								
	E0002071	20014-E-5503 - Switchgear & LV MCC - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27					20014-E-550	3 - Switchg	jear & LV M¢C	- Refresh/Renegoti
	E0002081	20014-E-5503 - Switchgear & LV MCC - Develop MRP	20d	01-Dec-26	05-Jan-27					20014-E-550	3 - Switchg	jear & LV M¢C	- Develop MRP
	E0002091	20014-E-5503 - Switchgear & LV MCC - Issue PO to Vendor	Od		05-Jan-27					◆ 20014-E-550	3 - Switchg	jear & LV M¢C	- Issue PO to Vend
	E0006161	20014-E-5503 - Switchgear & LV MCC - Manufacture	130d	06-Jan-27	12-Jul-27					20014	E-5503 - S	Switchgear & L\	/ MCC - Manufactur
	E0005871	20014-E-5503 - Switchgear & LV MCC - Testing & Inspection	2d	13-Jul-27	14-Jul-27					I 20014	E-5503 - S	Switchgear & L	√ MCC - Testing & Ir
	E0002101	20014-E-5503 - Switchgear & LV MCC - Ready to Ship	Od		19-Jul-27					♦ 20014	E-5503 - E	Switchgear & L	/ MCC - Ready to S
	E0002111	20014-E-5503 - Switchgear & LV MCC - ETA to Site	0d		03-Aug-27					♦ 20014	-E-5503 - '	Switchgear & L	V MCC - ETA to Site
	E14991	WBS-001 Receive at Site - Switchgear & LV MCC	0d	04-Aug-27						♦ WBS-	01 Receiv	/e at Site - Swit	chgear & LV MCC
	20014-E-5504 - E	Electrical Bulks	110d	01-Dec-26	14-May-27								
	E0001031	20014-E-5504 - Electrical Bulks - Refresh/Renegotiate	20d	01-Dec-26	05-Jan-27					20014-E-550	4 - Electrica	al Bulks - Refre	sh/Renegotiate
	E0001091	20014-E-5504 - Electrical Bulks - Develop MRP	20d	01-Dec-26	05-Jan-27					20014-E-550	4 - Electrica	al Bulks - Deve	lop MRP
	E0001101	20014-E-5504 - Electrical Bulks - Issue PO to Vendor	0d		05-Jan-27					♦ 20014-E-550	4 - Electrica	al Bulks - Issue	PO to Vendor
	E0006171	20014-E-5504 - Electrical Bulks - Manufacture	80d	06-Jan-27	29-Apr-27					20014-E-	5504 - Elec	ctrical Bulks - N	lanufacture
	E0001111	20014-E-5504 - Electrical Bulks - Ready to Ship	Od		29-Apr-27					◆ 20014-E-	5504 - Elec	ctrical Bulks - R	eady to Ship
	E0001121	20014-E-5504 - Electrical Bulks - ETA to Site	Od		13-May-27					♦ 20014-E	5504 - Ele	ctrical Bulks - E	TA to Site
	E0009751	WBS-001 Receive at Site - Electrical Bulks	Od	14-May-27						♦ WBS-00 ²	Receive a	at Site - Electric	al Bulks
	Instrumentation		325d	15-Jul-26	05-Nov-27								
	20014-J-5600: E	SDV	60d	15-Jul-26	09-Oct-26								· · · · · · · · · · · · · · · · · · ·
	E0001281	20014-J-5600 - ESDV - Refresh/Renegotiate	20d	15-Jul-26	12-Aug-26					0014-J-5600 - E	DV - Reire		e
	E0001341	20014-J-5600 - ESDV - Develop MRP	20d	15-Jul-26	12-Aug-26					0014-J-5600 - E	DV - Deve		
	E0001351	20014-J-5600 - ESDV - Issue PO to Vendor	0d		12-Aug-26				♦ 2	0014-J-5600 - E	DV - Issue	PO to vendor	
	E0006191	20014-J-5600 - ESDV - Manufacture	35d	13-Aug-26	01-Oct-26					20014-J-5600 - I	±SDV - Mar	nutacture	
	E0001361	20014-J-5600 - ESDV - Ready to Ship	0d		01-Oct-26				•	20014-J-5600 - I	SDV - Rea	ady to Ship	
	E0001371	20014-J-5600 - ESDV - ETA to Site	Od		08-Oct-26					20014-J-5600 - I	ESDV - ETA	A to Site	
	E0009321	WBS-001 Receive at Site - ESDV	Od	09-Oct-26						WBS-001 Receiv	/e at Site - I	ESDV	
	20014-J-5601 Or	ifice Plate Meter	65d	24-Nov-26	04-Mar-27								
	E0003391	20014-J-5601: Orifice Plate Meter - Refresh/Renegotiate	20d	24-Nov-26	21-Dec-26					20014-J-5601		ate Meter - Rei	resn/Renegotiate
	E0003401	20014-J-5601: Orifice Plate Meter - Develop MRP	20d	24-Nov-26	21-Dec-26					20014-J-5601			
	E0003411	20014-J-5601: Orifice Plate Meter - Issue PO to Vendor	Od		21-Dec-26							ate ivieter - Issu	le PO to vendor
	E0006241	20014-J-5601: Orifice Plate Meter - Manufacture	40d	22-Dec-26	24-Feb-27					20014-J-56		Plate Meter + N	lanutacture
	E0003421	20014-J-5601: Orifice Plate Meter - Ready to Ship	0d		24-Feb-27					◆ 20014-J-56	J1: Orifice I	Plate Meter + R	eady to Ship
	E0003431	20014-J-5601: Orifice Plate Meter - ETA to Site	0d		03-Mar-27					◆ 20014-J-56	01: Orifice	Plate Meter - E	IA to Site
	E15051	WBS-001 Receive at Site - Orifice Plate Meter	0d	04-Mar-27						◆ WBS-001 F	eceive at S	Site - Orifice Pla	ate Meter
	20014-J-5602: FI	ow Control Valve	165d	15-Jul-26	18-Mar-27						ou Control		/Deperation
	E0003211	20014-J-5602 - Flow Control Valve - Refresh/Renegotiate	20d	15-Jul-26	12-Aug-26				<u> </u>	0014 JE602 FK			
	E0003221	20014-J-5602 - Flow Control Valve - Develop MRP	20d	15-Jul-26	12-Aug-26					0014-J-2002-FK			
	E0003231	20014-J-5602 - Flow Control Valve - Issue PO to Vendor	Od		12-Aug-26					0014-J-5602 - Fk		vaive - issue F	
	E0006201	20014-J-5602 - Flow Control Valve - Manufacturing	140d	13-Aug-26	10-Mar-27					20014-J-56	02-Flow(Control Valve -	wanutacturing
	Remaining Lev	el of Effort - Actual Work - Critical Remaining Work			Data Date:26	- lul-24	Date			Revision		Checked	Approved
	Actual Level of	Effort Effort Remaining Work E			Page: 31 d	of 73	19-Aug-24	IFI					
		-	1									1	

SOLARIS



Activity ID	Activity Name	Rem. Duration	Start	Finish	20	2021	2022	2023	2024	2025	2026	2027	2028	2029	20	30 2031								
E0003241	20014 L5602 Elow Control Valva Roady to Ship	0d		10 Mar 27								QQQ ▲ 20014	2 2 2 2 - 1-5602 - E	QQQQ ow Control	Valve - Rea	adv to Ship								
E0003241	20014-J-5002 - Flow Control Valve - Ready to Ship	b0		17 Mar 27								▲ 20014	- 1-5602 - F	low Control	Valve - FTA	to Site								
E0003231	20014-J-3002 - Flow Control Valve - ETAto Site	b0	19 Mar 07	17-11/121-27	_							♦ WBS-	001 Receiv	e at Site - Ek	w Control	Valve								
E 1503 I	WBS-001 Receive at Site - Flow Control valve	DU	18-Mar-27	05 Nov 07							_	• WEG				Valve								
E0003271	20014- L5602 - PSV/s - Refresh/Renegatiate	2350 20d	24-Nov-26	21-Dec-26								20014-J-	5602 - PSV	's - Refresh/	Renegotia	te								
E0003281	20014-1-5602 - PSV/s - Nevelon MRP	20d	24-Nov-26	21-Dec-26								20014-J-	5602 - PSV	's - Develop	MRP									
E0003201	20014-1-5-5002 - 1 3V 3 - Develop Wilki	200	24-1107-20	21-Dec-20	_					1 1 1 1 1 1 1 1 1		20014-J-	5602 - PSV	's - Issue PC) to Vendo	r i i i								
E0005291	20014 L 5602 - FSVS - Issue FO to Vendor	2004	22 Dec 26	21-Dec-20	_					1 1 1 1 1 1 1 1 1			20014-1-5	602 - PSV/s	- Manufac	ture								
E0000211	20014-J-3002 - PSVS - Manufacture	2000	22-Dec-20	14-0ct-27	_								20014-1-5	602 - PSV/s	- Ready to	Shin								
E0003301	20014-J-5002 - PSVS - Ready to Ship	bU		14-0cl-27	_								20014-0	5602 - PS\/	s - FTA to S	Site								
E0003311	20014-J-5602 - PSV S - E IA to Site	Ud	05 No. 07	04-NOV-27							· · · · · · · · · · · · · · · · · · ·			Pacoivo at										
E15041	WBS-001 Receive at Site - PSVs	Ud	05-Nov-27										VUD3-001		Sile - FSV	5								
20014-J-56XX: F	Fire/Gas Detection System	110d	01-Dec-26	14-May-27								20014-1	-56XX Fire	Gas Detect	ion System	- Refresh/Re								
E0003511	20014 J 56XY: Fire/Cas Detection System - Develop MPP	200	01-Dec-20	05-Jan-27	_							20014-1	-56XX Fire	Gas Detect	ion System	- Develop MF								
E0003521	20014-J-30XX. File/Gas Detection System - Develop MRP	200	01-Dec-26	05-Jan-27	_							20014-1	-56XX: Fire	Gas Detect	ion System									
E0003531	20014-J-56XX: Fire/Gas Detection System - Issue PO to Vendor		00 1 07	05-Jan-27						· · · · · · · · · · · · · · · · · · ·			1/L_L56XX	Fire/Cas De	tection Sv	tom - Manufa								
E0006181	20014-J-56XX: Fire/Gas Detection System - Manufacture	DC8	06-Jan-27	06-May-27	_							▲ 200	14 15677	Fire/Cas Do	toction Sv	stom Poody								
E0003541	20014-J-56XX: Fire/Gas Detection System - Ready to Ship	Ud		06-May-27								◆ 200	14-J-JUAA.	Fire/Cas De	tection Sy	storm ETA to								
E0003551	20014-J-56XX: Fire/Gas Detection System - ETA to Site	0d		13-May-27									14-J-JUAA.	File/Gas De	Fire/Cos D									
E15061	WBS-001 Receive at Site - Fire/Gas Detection System	0d	14-May-27									• VVD	5-001 Rece		riie/Gas D									
Fabrication		140d	01-Apr-27	21-Oct-27										· · · · · · · · · · · · · · · · · · ·										
F0008211	WBS-001 - Structural Steel Fabrication	95d	01-Apr-27	17-Aug-27									VBS-001 - 9	Structural St	eel Fabrica	ition								
E0009001	WBS-001 Receive at Site Structural Steel Materials	0d	31_May_27	11 / lag 21	_					1 1 1 1 1 1 1 1 1 1 1 1		♦ WB	S-001 Rec	eive at Site S	Structural S	teel Materials								
Pining		130d	16-Apr-27	21_Oct_27																				
Carbon Steel		130d	16-Apr-27	21-Oct-27																				
E0006801	WBS-001 - CS Pipe Fabrication	130d	16-Apr-27	21-Oct-27							· · · · · · · · · · · · · · · · · · ·		WBS-001	- CS Pipe F	abrication									
Stainless Steel		130d	16-Apr-27	21-Oct-27									,											
E0006891	WBS-001 - SS Pipe Fabrication	130d	16-Apr-27	21-Oct-27									WBS-001	- SS Pipe F	abrication									
Construction		172d	01-Apr-27	07-Dec-27								V	•											
Civil		60d	01-Apr-27	25-Jun-27						; ;			20 001 64-											
E0008871	WBS-001 Siteworks	60d	01-Apr-27	25-Jun-27										WUINS										
Concrete & Found	dations	75d	16-Apr-27	03-Aug-27									/BS-001 Cc	ncrete & Fo	undations									
E0000901	WBS-001 COncrete & Foundations	750	21 May 27	03-Aug-27											, and a done									
F0009031	WBS-001 Structural Steel	70d	31-May-27	08-Sep-27									WBS-001 S	Structural Ste	el									
Buildings		30d	16-Sen-27	28-Oct-27									,											
E0010471	WBS-001 Building Erection	30d	16-Sep-27	28-Oct-27									WBS-001	Building Ere	ection									
Equipment		14d	14-Jun-27	02-Jul-27						· · · ·		▼												
E15151	WBS-001 Instrument Air Package Installation	2d	14-Jun-27	15-Jun-27						1 1 1 1 1 1 1 1 1		I WE	3S-001 Insti	ument Air P	ackage Ins	tallation								
E15161	WBS-001 Nitrogen Generation (PSA) Package Installation	2d	28-Jun-27	29-Jun-27								I W	B\$-001 Nitr	ogen Genei	ration (PSA) Package Ins								
E15171	WBS-001 LNG Drain Vessel (With Emerson Heater) Installation	2d	30-Jun-27	02-Jul-27	L	4				L		I W	B\$-001 LN	G Drain Ves	sel (With E	merson Heate								
Piping		105d	16-Jun-27	16-Nov-27									▼											
E0009081	WBS-001 Piping Installation	105d	16-Jun-27	16-Nov-27									WBS-00	1 Piping Inst	allation									
Electrical		100d	30-Jun-27	23-Nov-27								-	▼											
					-																			
Remaining Lev	vel of Effort Actual Work Critical Remaining Work			Data Data:2					Date		R	levision		Remaining Level of Effort Actual Work Critical Remaining Work Date Critical Remaining Work Approved										

Actual Level of Effort

Remaining Work Critical Secondary

Data Date:26-Jul-24 Page: 32 of 73

SOLARIS

M A N A G E M E N T CONSULTANTS INC.

Revision	Checked	Approved



ID	Activity Name	Rem.	Start	Finish	020		2021		20	022		2023		202	:4		2025	
		Duration			QQ	Q	QQ	Q	QQ		Q Q	QQ	QQ	≀ Q	QC	≀Q	QC	Į
E0009181	WBS-001 Electrical Installation	100d	30-Jun-27	23-Nov-27														_
Instrumentatiion &	Controls	95d	15-Jul-27	30-Nov-27														
E0009391	WBS-001 Install Instrumentation	95d	15-Jul-27	30-Nov-27														
Insulation		90d	29-Jul-27	07-Dec-27			-											
E0009401	WBS-001 Install Insulation	90d	29-Jul-27	07-Dec-27														
Phase 1 Commission	ioning & Start up	50d	12-Jun-28	22-Aug-28														
E0001761.1	WBS-001 Pre-Commissioning	45d	12-Jun-28	15-Aug-28	_													
E0006521.1	WBS-001 Commissioning	45d	19-Jun-28	22-Aug-28														
E0001771.1	WBS-001 Start Up	0d		22-Aug-28														
Phase 2 Commission	oning & Start up	50d	11-Apr-29	21-Jun-29														
E0001761.2	WBS-001 Pre-Commissioning (Phase 2)	45d	11-Apr-29	14-Jun-29														-
E0006521.2	WBS-001 Commissioning (Pahse 2)	45d	18-Apr-29	21-Jun-29													1	
E0001771.2	WBS-001 Start Up (Phase 2)	0d		21-Jun-29														
Base Plant Demol	lition	562d	17-Feb-27	23-May-29														
EPC (Detailed - S		_562d	17-Feb-27	23-May-29			1										1	
Engineering		90d	17-Feb-27	25-Jun-27														
TBP296122	Kick off Meeting	0d	17-Feb-27	20 0011 21					; ;		••							
TBP300000	Detailed Design	90d	17-Feb-27	25-Jun-27														
Regulatory Permits	(Contractor Scope)	1/8d	08-Mar-28	06-Oct-28														
City Of Delta MD		140d	08-Mar-28	05-Oct-28														
Highway Use Per	rmit - Delta Bylaw No. 6922	93d	25-May-28	05-Oct-28				-										
RGT606182	Prepare Highway Use and Inspection Permits Application	10d	25-May-28	08-Jun-28														
RGT606184	Submit Application	3d	08-Jun-28	13-Jun-28														
RGT606186	Review Period	80d	13-Jun-28	05-Oct-28														
RGT606188	Receive Approval/Permit	0d	10 0000 20	05-Oct-28														
Complex Building	a Permit - Delta Building/Blumbing Bulaw (No. 6060)	204	09 Jun 29	21 Jul 29														
RGT606382	Prepare Complex Building Permit Application	10d	08-Jun-28	22-Jun-28														-
RGT606384	Submit Application	34	10- lup-28	22- Jun-28														
DOT606396		Ju Du	19-Jun-20	22-Jul-20														
RG1000380		200	22-JUN-28	21-JUI-28														
RG1606388	Receive Approval/Permit - Complex Building Permit	Ud		21-Jul-28													-	
Temporary Buildi	ings Permit - Delta Building/Plumbing Bylaw (No. 6060)	30d	30-May-28	12-Jul-28		+												-
RG1616763	Prepare Complex Building Permit Application	10d	30-May-28	13-Jun-28														
RGT616766	SubmitApplication	3d	08-Jun-28	13-Jun-28														
RGT616764	Review Period	20d	13-Jun-28	12-Jul-28				-									1	
RGT616765	Receive Approval/Permit - Complex Building Permit	0d		12-Jul-28														
Development Per	mit (Commercial/Industrial) - Delta Official Community Plan Bylaws	143d	08-Mar-28	29-Sep-28											1			_
RGT606482	Prepare Development Permit (Commercial/Industrial) Application	60d	08-Mar-28	02-Jun-28														
RGT606484	SubmitApplication	3d	02-Jun-28	07-Jun-28														
RGT606486	Review Period	80d	07-Jun-28	29-Sep-28				-										
RGT606488	Receive Approval/Permit	0d		29-Sep-28														
Development Per	mit (Streamside Protection & Enhancement) - Bylaw No. 6349	143d	08-Mar-28	29-Sen-28														
RGT606582	Prepare Development Permit (Commercial/Industrial) Application	60d	08-Mar-28	02-Jun-28							• -							
RGT606584	Submit Application	34	02_lun_28	07_lun_28				-										
RCTENESOE	Review Period	۵0 k	07_ lup 20	20 Con 20														
000000		ouu	01-JUII-20	23-3ep-20								· · · ·						_
Remaining Leve	el of Effort Actual Work Critical Remaining Work			Data Data 2	6 1	24								D	ate			
Actual Level of	Effort Remaining Work - Critical Secondary			Data Date:2	o-Jul-2	+							19	-Aug-	24	I	IFI	_
,				raye. Jo	0173													





		FODTIC	Tilbu	ry Island	LNG Facili	ty Expansion (3 BCF)	ALALAS	SOLARIS
		FORTIS BC ^{**}	TLSE	Sub-Proje	ects Integra	ted Schedule-Detaile	d		MANAGEMENT CONSULTANTS INC.
Activi	ty ID	Activity Name	Rem. Duration	Start	Finish	20 2021 2022	2023 2024 2025 Q Q Q Q Q Q Q Q Q Q	2026 2027 202 2	8 2029 2030 2031 Q Q Q Q Q Q Q Q Q Q Q Q Q
	RGT606588	Receive Approval/Permit	Od		29-Sep-28			***********	♦ Receive Approval/Permit
	Demolition Permi	it - Delta Official Community Plan Bylaws	83d	08-Jun-28	05-Oct-28				➡
	RGT606682	Prepare Demolition Permit Application	20d	08-Jun-28	07-Jul-28				Prepare Demolition Permit Application
	RGT606684	Submit Application	3d	07-Jul-28	12-Jul-28			1	Submit Application
	RGT606686	Review Period	60d	12-Jul-28	05-Oct-28			· · · · · · · · · · · · · · · · · · ·	📕 Review Period
	RGT606688	Receive Approval/Permit	Od		05-Oct-28				♦ Receive Approval/Permit
	MOTI		85d	06-Jun-28	05-Oct-28			•	 ▼
	Highway Use Per	mit - Transportation Act	85d	06-Jun-28	05-Oct-28				
	RGT610182	Prepare Highway Use Permit Application	20d	06-Jun-28	05-Jul-28				Prepare Highway Use Permit Application
	RGT610184	Submit Application	3d	05-Jul-28	10-Jul-28				SubmitApplication
	RGT610186	Review Period	62d	10-Jul-28	05-Oct-28				📕 Review Period
	RGT610188	Receive Approval/Permit	Od		05-Oct-28				 Receive Approval/Permit
	Nav Canada		135d	27-Mar-28	06-Oct-28	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·		━━━
	Boundary Bay Air	rport Email Notification	15d	27-Mar-28	18-Apr-28			▼	
	Email Notificatio		15d	27-Mar-28	18-Apr-28			₩.	
	RGT604184	Prepare and Send Notification Email	5d	27-Mar-28	04-Apr-28			I Pr	epare and Send Notification Email
	RGT604188	Receive Confirmation Response	10d	04-Apr-28	18-Apr-28				eceive Confirmation Response
	Land Use Program	m - Crane	128d	06-Apr-28	06-Oct-28				▼
	RGT611182	Prepare Land Use Crane Permit Application	5d	06-Apr-28	13-Apr-28			1 P	repare Land Use Crane Permit Application
	RGT611184	Submit Application	3d	13-Apr-28	18-Apr-28			I S	ubmitApplication
	RGT611186	Review Period	120d	18-Apr-28	06-Oct-28				📕 Review Period
	RGT611188	Receive Approval/Permit	Od		06-Oct-28				♦ Receive Approval/Permit
	Port of Vancouver		25d	11-Aua-28	18-Sep-28				▼
	Notice to Shippin	g - Canada Marine Act	25d	11-Aug-28	18-Sep-28				₩
	RGT613184	Notice to Shipping Registration	5d	11-Aug-28	18-Aug-28				Notice to Shipping Registration
	RGT613188	Receive Info/Confirmation	20d	18-Aug-28	18-Sep-28				Receive Info/Confirmation
	Transport Canada		128d	29-Mar-28	29-Sep-28				━◀
	Aeronautical Clea	arance - Canadian Aviation Regulations (CARs)	128d	29-Mar-28	29-Sep-28				➡
	RGT614182	Prepare Aeronautical Clearance Permit Application	5d	29-Mar-28	06-Apr-28			I Pr	repare Aeronautical Clearance Permit App
	RGT614184	Submit Application	3d	06-Apr-28	11-Apr-28			1 Si	ubmit Application
	RGT614186	Review Period	120d	11-Apr-28	29-Sep-28				Review Period
	RGT614188	Receive Approval/Permit	Od		29-Sep-28				♦ Receive Approval/Permit
	Work Safe BC - W	orker Compenstion Act/OHS Regulation	15d	10-Aug-28	31-Aug-28				▼
	Guidelines 20.3-2	2 Qualified coordinators	6d	23-Aug-28	31-Aug-28				
	RGT616182	Prepare Application of Guidelines 20.3-2 Qualified coordinates 20.3-2 Reprint Prepare Application of Guidelines 20.3-2 Reprint Prepare	ators 1d	23-Aug-28	24-Aug-28				Prepare Application of Guidelines 20.3
	RGT616184	Submit Application	3d	24-Aug-28	29-Aug-28				I Submit Application
	RGT616186	Review Period	2d	29-Aug-28	31-Aug-28				Review Period
	RGT616188	Receive Approval/Permit	Od		31-Aug-28				 Receive Approval/Permit
	G20.2(1)/20.2.1(1)	Notice of Project, Section 20.2(1) of the OHS Regulation	6d	23-Aug-28	31-Aug-28				▼
	RGT616282	Prepare Work Safe BC Application - Notice of Project	1d	23-Aug-28	24-Aug-28				Prepare Work \$afe BC Application - No
	RGT616284	SubmitApplication	3d	24-Aug-28	29-Aug-28				I Submit Application
	RGT616286	Review Period	2d	29-Aug-28	31-Aug-28				I Review Period
	RGT616288	Receive Approval/Permit	Od		31-Aug-28				◆ Receive Approval/Permit
	G20.2.1(1) and (2)) Notice of project for asbestos - Ongoing work, OH&S 20.112	ba	10-Aug-28	18-Aug-28				\mathbf{v}
	······································	, , , , , , , , , , , , , , , , , , ,						<u></u>	
			aining Work				Date	Revision	Checked Approved
					Data Date:2	6-JUI-24			

Actual Level of Effort

Remaining Work
 Critical Secondary

Page: 34 of 73

Date 19-Aug-24 IFI

Revision	Checked	Approved



Activity	ID	Activity Name	Rem.	Start	Finish	20 2021 2022 2023 20	2025	2026 2027	2028	2029	2030	2031
			Duration									
	RGT616382	Prepare Work Safe BC Application - Notice of Project for asbestos	1d	10-Aug-28	11-Aug-28				Pre	pare Work S	ate BC Apr	plication - N
	RGT616384	Submit Application	3d	11-Aug-28	16-Aug-28				I Su	bmit Applicati	on	
	RGT616386	Review Period	2d	16-Aug-28	18-Aug-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		I Re	view Period		
	RGT616388	Receive Approval/Permit	0d		18-Aug-28				♦ Re	ceive Approv	al/Permit	
	G20.2.1(2) (c) Noti	ce of project -Significant distrubance of lead-containing material-OH&S 20.112	6d	23-Aug-28	31-Aug-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		▼			
	RGT616482	Prepare Work Safe BC Application - Notice of Project (Significant distrubance of lea	1d	23-Aug-28	24-Aug-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		Pr	epare Work S	3afe BC Ap	oplication - N
	RGT616484	Submit Application	3d	24-Aug-28	29-Aug-28				Si	ibmit Applicat	on	
	RGT616486	Review Period	2d	29-Aug-28	31-Aug-28				R	view Period		
	RGT616488	Receive Approval/Permit	0d		31-Aug-28				♦ R	ceive Approv	/al/Permit	
	G220.2.1(2) (d) No	tice of project - Other similar exposure work activities-OH&S 20.113	6d	23-Aua-28	31-Aug-28				▼			
	RGT616582	Prepare Work Safe BC Application - Notice of Project (Other similar exposure work	1d	23-Aug-28	24-Aug-28		1 1 1 1 1 1 3 3 4 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		Pr	epare Work \$	Safe BC Ap	oplication - N
	RGT616584	Submit Application	3d	24-Aug-28	29-Aug-28				Si	bmit Applicat	on	
	RGT616586	Review Period	2d	29-Aua-28	31-Aug-28				R	view Period		
	RGT616588	Receive Approval/Permit	b0	gc	31-Aug-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		● R	ceive Appro	val/Permit	
	30M33 Permit - O	HS Regulation Sections 19	64	22 Aug 28	31 Aug 28				V			
	RGT616682	Prepare 30M33 Permit Application	1d	23-Aug-28	24-Aug-28				ı Pr	epare 30M33	Permit Ap	plication
	RGT616684	Submit Application	3d	24-Aug-28	29-Aug-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		I Su	bmit Applicat	ion	
	RGT616686	Review Period	24	29-Aug-28	31_Aug_28		1 1 1 1 1 1 1 1 3 4 4 4 4 4 4 1 4 4 4 4 4 4 4 1 4 4 4 4 4 4 4		ı R	view Period		
	PGT616688	Revelve Approval/Permit	20 0d	20-Aug-20	31 Aug 28				A R	ceive Appro	val/Permit	
	RG1010000	Receive Approva/Fermic	0u 4d	00 Aug 00	31-Aug-28				• •••	concr ppro		
	BC One Call Regi	stration	40 4d	23-Aug-28	29-Aug-28				V			
	RGT601184	BC One Call Registration	1d	23-Aug-28	24-Aug-28				i BC	One Call Re	gistration	
	RGT601188	Receive Info/Confirmation	3d	24-Aug-28	29-Aug-28				I R	ceive Info/C	onfirmation	1
	LNG Product Remov	val and Tank De Inventory	31d	22-Aug-28	04-Oct-28						[
	Preparation		21d	22-Aug-28	20-Sep-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		$\mathbf{\overline{w}}$			
	C0102	Stop LNG Flow into Tank	1d	22-Aug-28	23-Aug-28				I St	p LNG Flow	into Tank	
	C0104	Main LNG Removal via Bottom Sump	20d	23-Aua-28	20-Sep-28				N	ain LNG Rer	noval via B	3ottom Sum
	C0106	Start Warming Tank	Dd	20-Sep-28			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		♦ S	tart Warming	Tank	
	De-Inventory		5d	27-Sen-28	04-Oct-28				▼			
	A0108	De-Inventory Product	5d	27-Sep-28	04-Oct-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		1	e-Inventory	Product	
	Demolition		165d	04-Oct-28	23-May-29		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					
	Base Plant Mobiliz	ation	22d	04-Oct-28	02-Nov-28				₩			
	Premobilization A	ssessments	20d	04-Oct-28	01-Nov-28							
	D0202	Hazmat (Oils, Batteries, etc.) Assessment	5d	04-Oct-28	11-Oct-28					lazmat (Oils,	Batteries, e	etc.) Assess
	D0204	NORMs Assessment	5d	04-Oct-28	11-Oct-28					IORMs Asse	ssment	
	D0206	Asbestos/Lead Assessments	5d	04-Oct-28	11-Oct-28				• •	sbestos/Lea	d Assessm	nents
	D0208	Prepare and Submit Asbestos Abatement Documentation	15d	11-Oct-28	01-Nov-28		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			Prepare and	Submit Asl	bestos Abat
	Mobilization		2d	01-Nov-28	02-Nov-28				▼			
	D0210	Contractor(s) Mobilization & Site Setup	2d	01-Nov-28	02-Nov-28					Contractor(s	Mobilizatic	on & Site Se
	LNG Tank Demoliti	on Work	145d	04-Oct-28	25-Apr-29		I I		—			
	Tank (Tag 200)		145d	04-Oct-28	25-Apr-29		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					
	D0302	Remaining Tank Warming	10d	04-Oct-28	18-Oct-28					kemaining la	nk vvarmin	ıg
	D0304	Nitrogen Purge	5d	18-Oct-28	25-Oct-28				I	Nitrogen Pur	je	
	D0306	Isolate Tank	2d	25-Oct-28	26-Oct-28				Î	lsolate Tank		
		;	,		1							

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
	Remaining Work		Data Date:20-Jul-24	19-Aug-24	IFI
			rage. 35 0175		

SOLARIS

Revision	Checked	Approved



Activ	ity ID	Activity Name	Rem.	Start	Finish	020	2021	2022	2023	2024	2025	2026	2027	202	3	2029	2030	2031
	D 0000			00.0.4.00	00.01.00													
	D0308	Air Purge	/d	26-Oct-28	06-Nov-28												(Domo (oontoinmor
	D0310	LNG lank Demo (containment excluded)	100d	06-Dec-28	25-Apr-29													Jonainnen
	Demo Other Items	s (Tags 202, 202A, 202B)	35d	04-Oct-28	22-Nov-28													
	D0322	Nitrogen Purge	100	04-Oct-28	18-Oct-28					<u>.</u>				+				
	D0324	Flush & Clean	10d	18-Oct-28	01-Nov-28													
	D0326	Demo 202, 202A/B	15d	01-Nov-28	22-Nov-28											emo 202, 20	ZAVB	
	Base Plant Demoli	ition Work	144d	02-Nov-28	23-May-29													
	Hazardous Materi	ial Abatement	33d	02-Nov-28	19-Dec-28												Diant Din	oina & Eauir
	D0402	N2 Purge Base Plant Piping & Equipment Interconnect at 11A	30	02-Nov-28	07-Nov-28				· · · · · · · · · · · · · · · · · · ·					4		Fulge Dase	riant rip	ning & Equip
	D0404	Asbestos Abatement (Allowance)	25d	07-Nov-28	12-Dec-28											spesios Apa	ement (A	llowance)
	D0406	HazmatAbatement	30d	07-Nov-28	19-Dec-28										H	azmat Abate	ement	
	D0408	NORMAbatement (Allowance)	15d	07-Nov-28	28-Nov-28										∎ NO	DRMAbater	nent (Allov	wance)
	Demo Crew A		54d	13-Nov-28	29-Jan-29													
	Foam System Bu	uilding/Water Tank	1d	13-Nov-28	14-Nov-28	· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·							▼			
	D0410	Demo Foam System-Building, Water Tanks	1d	13-Nov-28	14-Nov-28										De	mo Foam S	ystem-Bu	ilding, Wate
	LNG Tank Bund	Wall	7d	14-Nov-28	23-Nov-28										▼			
	D0420	LNG Tank Containment Wall Demo	7d	14-Nov-28	23-Nov-28											IG lank Cor	tainment	Wall Demo
	LNG Access Site	9	14d	23-Nov-28	13-Dec-28										W			
	D0432	Demo LNG Access Stile	1d	23-Nov-28	24-Nov-28				: : : : : : : : : :					· · · · · · · · · · · · · · · · · · ·	De	emo LNG Ac	cess Stile	
	D0434	Prepare for Tank Demo Crew Access	5d	23-Nov-28	30-Nov-28										I Pr	epare for Ta	nk Demo	Crew Acce
	D0436	Interconnect Piping Demo	5d	06-Dec-28	13-Dec-28										🕴 🛯 In	terconnect l	Piping Der	mo
	Foam Generating	g Building	2d	19-Dec-28	22-Dec-28										•			
	D0442	Inlet Area Piping Demo	2d	19-Dec-28	21-Dec-28										i In	llet Area Pipi	ng Demo	
	D0444	Demo Foam Generation Building	1d	21-Dec-28	22-Dec-28										i D	emo Foam	Generatio	on Building
	Maintenance Bu	ilding	2d	22-Dec-28	26-Dec-28			L	· · · · · · · · · · · · · · · · · · ·					4 4/- -	•		J	
	D0450	Demo Maintenance Building	2d	22-Dec-28	26-Dec-28										D	emo Mainte	nance Bu	uilding
	Office/Control R	oom & MCC	7d	26-Dec-28	04-Jan-29										•			
	D0460	Demo Office/Control Room/MCC	7d	26-Dec-28	04-Jan-29										j 🕴 🗆)emo Office/	Control R	:oom/MCC
	BOG & Cycle Co	omp Building	7d	04-Jan-29	15-Jan-29										- H			
	D0470	Demo BOG & Cycle Comp Building	7d	04-Jan-29	15-Jan-29										1	Demo BOG	& Cycle C	omp Buildir
	Station & Transfo	ormer	1d	15-Jan-29	16-Jan-29										•			
	D0480	Demo Station & Transformer	1d	15-Jan-29	16-Jan-29										[Demo Statio	n & Transf	former
	Emergency Gene	erator Building	1d	16-Jan-29	17-Jan-29													
	D0490	Demo Emergency Generator Building	1d	16-Jan-29	17-Jan-29											Demo Emer	ency Ge	nerator Bui
	Diesel Storage T	Tank	1d	17-Jan-29	18-Jan-29										V			
	D0510	Demo Diesel Storage Tank	1d	17-Jan-29	18-Jan-29										[Demo Diese	Storage	Tank
	Utility Building		1d	18-Jan-29	19-Jan-29													
	A0520	Demo Utility Building	1d	18-Jan-29	19-Jan-29										I I	Demo Utility	Building	
	Storage Shed Bu	uilding	6d	19-Jan-29	29-Jan-29				· · · · · · · · · · · · · · · · · · ·									
	D0532	Demo Storage Shed Building	1d	19-Jan-29	22-Jan-29											Demo Stora	ge Shed E	3uilding
	D0534	NORMs Abatement Contingency	5d	22-Jan-29	29-Jan-29											NORMs Aba	tement C	ontingency
	Demo Crew B		36d	19-Dec-28	07-Feb-29										•			
	Piperacks		5d	19-Dec-28	26-Dec-28										♥_			
	D0610	Piperacks Demo	5d	19-Dec-28	26-Dec-28				1 1 1 1	1 1 1 1 1 1 1 1 1 1 1 1					I P	iperacks De	mo	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	Truck Loading B	uilding	1d	26-Dec-28	27-Dec-28										V			
			1										<u> </u>					<u> </u>
	Remaining Leve	el of Effort			Data Date:2	6-Jul-24				Date			Revision			Checked	Ар	proved
	Actual Level of I	Effort Effort Remaining Work E			Page: 36	of 73			19	-Aug-24								
			1		-													

1			1	1			CL NO		C	L	A	R		S					-
		20	27			20	28			20	29			20	30		1	2()
Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q	Q		C



Name Rem Start Preim 202 2023 2024 2025 2026 D0620 Demo Truck Loading Building 1d 26-Dec-28 22-Dec-28	227 2028 2029 2030 2031 Q
D0620 Demo Truck Loading Building 1d 26-De-28 27-De-28 P0 Heater 1d 26-De-28 27-De-28 D0630 FG Heater Demo 1d 26-De-28 27-De-28 D0640 Demo N2 Building, Surge Tank/Sea Can 1d 26-De-28 27-De-28 D0640 Demo N2 Building, Surge Tank/Sea Can 1d 26-De-28 27-De-28 D0650 Vaporizars 2d 22-De-28 27-De-28 29-De-28 D0660 Vaporizars 2d 22-De-28 29-De-28 29-De-28 D0670 Demo N2 Building Ethane, Propane, Butane) 4d 29-De-28 04-Jan-29 D0670 Demo Mole Steve Unit 1d 11-Jan-29 11-Jan-29 D0680 Oemo Mole Steve Unit 1d 11-Jan-29 12-Jan-29 D070 Demo Faed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Faed Gas Comp & Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo Mole Steve Unit 1d 15-Jan-29 15-Jan-29	Cold Box Demo Cold Bo
FG Heater Demo 1d 28.9ec.88 27.0ec.28 D0630 FG Heater Demo 1d 28.0ec.88 27.0ec.28 D0640 Demo N2 Building, Surge Tank/ Sea Can 1d 26.0ec.78 27.0ec.28 D0640 Demo N2 Building, Surge Tank/ Sea Can 1d 26.0ec.78 27.0ec.28 Vaportzers 2d 27.0ec.28 27.0ec.28 29.0ec.28 D0660 Vaportzers Demo 2d 27.0ec.28 0.4an.29 D0670 Demo Bullets (Ethane, Propane, Bulane) 4d 29.0ec.28 0.4an.29 D0680 Cold Box Demo 5d 04.an.29 11.1an.29 D0680 Demo Mole Sieve Unit 1d 11.an.29 12.3an.29 D0690 Demo Mel Sieve Unit 1d 11.an.29 15.3an.29 D0710 Demo Mel Sieve Unit 1d 15.3an.29 10.5an.29 D0710 Demo Mel Sieve Unit 1d 15.3an.29 10.5an.29 D0720 Demo Mel Comp & Coolers 1d 15.3an.29 10.5an.29 D0730 Demo Mel Comp & Stander Tim Coolers 1d 15.4an.29 10.5an.29 <tr< td=""><td>FG Heater Demo Demo N2 Building, Surge Tank / S Vaporizers Demo Demo Bullets (Etharie, Propane, Cold Box Demo Demo Mole Sieve Unit</td></tr<>	FG Heater Demo Demo N2 Building, Surge Tank / S Vaporizers Demo Demo Bullets (Etharie, Propane, Cold Box Demo Demo Mole Sieve Unit
D0630 FG Heater Demo 1d 26-Dec-28 27-Dec-28 N2 60 fulling Surge Tax/SER Can 1d 26-Dec-28 27-Dec-28 D0640 Demo N2 Buiking, Surge Tank / Sea Can 1d 26-Dec-28 27-Dec-28 D0660 Vaportains 2d 27-Dec-28 29-Dec-28 29-Dec-28 D0660 Vaportains 2d 27-Dec-28 29-Dec-28 24-Dec-28 D0670 Demo N2 Buiking, Surge Tank / Sea Can 4d 29-Dec-28 04-Jan-29 D0670 Demo Buikits (Ehnne, Propane, Butane) 4d 29-Dec-28 04-Jan-29 C6d Box Editors 5d 04-Jan-29 11-Jan-29 D0680 Cold Box Demo 5d 04-Jan-29 11-Jan-29 D0690 Demo Mole Sizev Unit 1d 11-Jan-29 12-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Tim Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Tim Coolers 1d 15-Jan-29	FG Heater Demo Demo N2 Building, Surge Tank / S Vaporizers Demo Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit
N2 Building, Surge Tank/Sea Can 1d 26-Doc-28 27-Doc-28 D0660 Demo N2 Building, Surge Tank / Sea Can 1d 26-Doc-28 27-Doc-28 Vaportars 2d 27-Doc-28 29-Doc-28 29-Doc-28 D0660 Vaportzers Demo 2d 27-Doc-28 29-Doc-28 D0670 Demo Bullets (Ehnane, Propane, Butane) 4d 29-Doc-28 04-Jan-29 D0680 Cold Box Demo 5d 04-Jan-29 11-Jan-29 D0680 Cold Box Demo 5d 04-Jan-29 11-Jan-29 D0680 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0680 Demo Mole Sieve Unit 1d 12-Jan-29 15-Jan-29 D0710 Demo MR CompWater Tim Coolers 1d 16-Jan-29 15-Jan-29 D0720 Demo MR CompWater Tim Coolers 1d 16-Jan-29 16-Jan-29 D0732 Demo MR CompWater Tim Coolers 1d 16-Jan-29 16-Jan-29 D0732 Demo MR CompWater Tim Coolers 1d 16-Jan-29 16-Jan-29 D0732 Demo R Light Standards 1d 16-Jan-29 16-Ja	Demo N2 Building, Surge Tank / S Vaporizers Demo Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit
D0640 Demo N2 Building, Surge Tank / Sea Can 1d 26-Dec-28 27-Dec-28 Metorizers 2d 27-Dec-28 29-Dec-28 D0660 Vaporizers Demo 2d 27-Dec-28 29-Dec-28 D0670 Demo Builets (Ethane, Propane, Butane) 4d 29-Dec-28 64-Jan-29 D0670 Demo Builets (Ethane, Propane, Butane) 4d 29-Dec-28 64-Jan-29 D0680 Cold Box Demo 5d 64-Jan-29 11-Jan-29 D0680 Cold Box Demo 5d 64-Jan-29 11-Jan-29 D0680 Demo Field Gas Comp & Coolers 1d 11-Jan-29 12-Jan-29 D0690 Demo Field Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Field Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Peluge Building & FW System 1d 15-Jan-29 07-Feb-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-	Vaporizers Demo Vaporizers Demo Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit Demo Feed Gas Comp & Cooler
Vaporizers 2d 27-Dec-28 29-Dec-28 D0660 Vaporizers Demo 2d 27-Dec-28 29-Dec-28 Builds (Ehane, Propane, Butane) 4d 29-Dec-28 04-Jan-29 D0670 Demo Builets (Ehane, Propane, Butane) 4d 29-Dec-28 04-Jan-29 D0680 Cold Box Demo 5d 04-Jan-29 11-Jan-29 D0680 Cold Box Demo 5d 04-Jan-29 12-Jan-29 D0680 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0690 Demo Mole Sieve Unit 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Deluge Building & FW System 1d 15-Jan-29 15-Jan-29 D0732 Demo Fead Gas L-Demobilization Work & Final Cleanup 2d 25-Apr-29 23-May-29 D0734 Light Standards 1d 15-Jan-29 15-Jan-29 07-Feb-29 <	Vaporizers Demo Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit Demo Feed Gas Comp & Cooler
D0600 VapDlete Unit 20 247-08-28 29-08-28 04-Jan-29 D0670 Demo Bullets (Ethane, Propane, Butane) 4d 29-Dec-28 04-Jan-29 D0680 Cold Box 5d 04-Jan-29 11-Jan-29 D0680 Cold Box 5d 04-Jan-29 11-Jan-29 D0680 Cold Box 5d 04-Jan-29 11-Jan-29 D0680 Cold Box Edit Stow Unit 1d 11-Jan-29 12-Jan-29 D0690 Demo Mole Sieve Unit 1d 12-Jan-29 15-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0731 Light Standards 1d 15-Jan-29 15-Jan-29 D0732 Demo Feed Gas Comp & Coolers 1d 15-Jan-29 17-Jan-29 D0733 NORMs Abateremt Contingency 15d 17-Ja	Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit Demo Feed Gas Comp & Cooler
Edubis (Edular, Hopink, Bullar) 44 22-Dec/28 04-Jan-29 D0670 Demo Bullets (Ethane, Propane, Butane) 6d 04-Jan-29 11-Jan-29 D0680 Cold Box 5d 04-Jan-29 11-Jan-29 D0680 Cold Box 5d 04-Jan-29 11-Jan-29 D0680 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0690 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Deluge Building & FW System 1d 15-Jan-29 16-Jan-29 D0732 Demo Deluge Building & FW System 1d 16-Jan-29 17-Jan-29 D0732 Demo Deluge Building & FW System 1d 16-Jan-29 17-Jan-29 D0734 Light Shadards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29	Demo Bullets (Ethane, Propane, Cold Box Demo Demo Mole Sieve Unit Demo Feed Gas Comp & Cooler
Cold Box	Cold Box Demo Demo Mole Sieve Unit Demo Feed Gas Comp & Cooler
D0680 Cold Box Demo 5d 04-Jan-29 11-Jan-29 Moto Sive Unit 1d 11-Jan-29 12-Jan-29 D0690 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo MR System 1d 15-Jan-29 15-Jan-29 D0732 Demo MR Subatement Contingency 1d 15-Jan-29 15-Jan-29 D0734 Light Standards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 Base Plant-inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 <t< td=""><td>I Cold Box Demo I Demo Mole Sieve Unit V I Demo Feed Gas Comp & Cooler</td></t<>	I Cold Box Demo I Demo Mole Sieve Unit V I Demo Feed Gas Comp & Cooler
Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0690 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 D0700 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp.Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Deluge Building & FW System 1d 15-Jan-29 17-Jan-29 D0734 Light Standards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 2	↓ Demo Mole Sieve Unit ↓ Demo Feed Gas Comp & Cooler
D0690 Demo Mole Sieve Unit 1d 11-Jan-29 12-Jan-29 Feed Gss Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 MR Comp/Water Trim Coolers 1d 12-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Demo Luge Building & FW System 1d 15-Jan-29 16-Jan-29 D0732 Demo Demo Demo Luge Building & FW System 1d 15-Jan-29 17-Jan-29 D0734 Light Standards 1d 15-Jan-29 17-Jan-29 07-Feb-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 23-May-29 D0800 LNG Tank Disposal - Demobilization Work & Final Cleanup/Demob 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 BecER 1121d 03-Mar-26 11-Sep-30 112-Jan-29 112-Jan-29 112-Jan-29 112-Jan-29	∣ Demo Mole Sieve Unit ▼ ∣ Demo Feed Gas Comp & Coolei
Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 MR Comp/Water Trim Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 D0720 Demo MR Comp/Water Trim Coolers 1d 15-Jan-29 D0732 Demo Deluge Building & FW System 1d 15-Jan-29 D0734 Light Standards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 D0800 LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Disposal - Demobilization Work & Final Cleanup/Demob 20d 25-Apr-29 23-May-29 BCER 1121d 03-Mar-26 11-Sep-30 11-Sep-30 Facility Permits (FEI Scope) 1121d 03-Mar-26 15-Mar-28 Regatification 246d 02-Jul-26 16-Jul-26 RGT112180.1 Prepare Notification to Surrounding	▼ I Demo Feed Gas Comp & Cooler
D0710 Demo Feed Gas Comp & Coolers 1d 12-Jan-29 15-Jan-29 MR CompWater Trim Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR CompWater Trim Coolers 1d 15-Jan-29 15-Jan-29 D0732 Demo Deluge Building & FW System 1d 15-Jan-29 16-Jan-29 D0734 Light Standards 1d 16-Jan-29 17-Feb-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 Base Plant - inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 BCER 1121d 03-Mar-26 11-Sep-30 Facility Permits (FEI Scope) 1121d 03-Mar-26 11-Sep-30 Facility Permit Amendment- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Rgasification 24dd 02-Jul-26 29-Jul-27 16-Jul-26 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neigh	Demo Feed Gas Comp & Coolei
MR CompWater time Coolers 1d 15-Jan-29 15-Jan-29 D0720 Demo MR CompWater Trim Coolers 1d 15-Jan-29 15-Jan-29 D0120 Demo MR CompWater Trim Coolers 1d 15-Jan-29 15-Jan-29 D0132 Demo Deluge Building & FW System 1d 15-Jan-29 07-Feb-29 D0734 Light Standards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 Base Plant - inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LING Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 D0800 LING Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 Regulatory Permits (FEI Scope) 1121d 03-Mar-26 11-Sep-30 BCER 1121d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 15-Mar-28 Regasification 246d 02-Jul-26 15-Jun-28 Regasification to Surrounding Neighbors/Righ	
D0/20 Definition Compliance of the Complianc	Demo MR CompWater Trim Cod
Detugs Duilting of W System 1/d 1/3/2 0/1/2 0/	
D0734 Light Standards 1d 16-Jan-29 17-Jan-29 D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 Base Plant - inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 BcER 1121d 03-Mar-26 11-Sep-30 BCER 1121d 03-Mar-26 11-Sep-30 Facility Permit Amendent- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 16-Jul-26 16-Jul-26 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 30-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	Demo Deluge Building & FW Sys
D0736 NORMs Abatement Contingency 15d 17-Jan-29 07-Feb-29 Base Plant - inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 Regulatory Permits (FEI Scope) 1121d 03-Mar-26 11-Sep-30 BCER 1121d 03-Mar-26 11-Sep-30 Facility Permit Amendment- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 30-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	Light Standards
Base Plant - inc LNG Tank Disposal - Demobilization Work & Final Cleanup 20d 25-Apr-29 23-May-29 D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 Regulatory Permits (FEI Scope) 1121d 03-Mar-26 11-Sep-30 BCER 1121d 03-Mar-26 15-Mar-28 Regasification 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26 30-Jul-26	NORMs Abatement Contingence
D0800 LNG Tank Demo Final Cleanup/Demob 20d 25-Apr-29 23-May-29 Regulatory Permits (FEI Scope) 1121d 03-Mar-26 11-Sep-30 BCER 1121d 03-Mar-26 15-Mar-28 Facility Permit Amendment- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	\mathbf{w}
Regulatory Permits (FEl Scope) 1121d 03-Mar-26 11-Sep-30 BCER Facility Permit Amendment- Oil and Gas Activities Act Facility Permit Amendment- Oil and Gas Activities Act Regasification Regasification RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26 30-Jul-26	📱 LNG Tank Demo Final Clear
BCER 1121d 03-Mar-26 11-Sep-30 Facility Permit Amendment- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	
Facility Permit Amendment- Oil and Gas Activities Act 503d 03-Mar-26 15-Mar-28 Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	
Regasification 246d 02-Jul-26 29-Jun-27 RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 02-Jul-26 16-Jul-26 RGT112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	· · · · · · · · · · · · · · · · · · ·
RGT112180.1 Prepare Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 02-Jul-26 16-Jul-26 16-Jul-26 16-Jul-26 16-Jul-26 16-Jul-26 16-Jul-26	V
RG 1112180.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 16-Jul-26 30-Jul-26	action to Surrounding Neighbors/Right Holders (by PC
	onsultation Period/Receive Responses
RG1112180.3 Public Consultation Period/Receive Responses 20d 30-Jul-26 28-Aug-26	Summary to Support Application
RG1112180.4 Prepare Summary to Support Application 10d 28-Aug-26 14-Sep-26	are Equility Dermit Application (Deliverable(a) (Eirst Natio
RG1112182 Prepare Facility Permit Application (Deliverable(s) / First Nation Consultation) 20d 06-Nov-26 04-Dec-26	are rapility remit Application (Deliverable(S)/1 list Naturnation
RG1112184 Submit Application 3d 07-Dec-26 09-Dec-26	Povew/Poriod
RG1112186 Review Period 135d 10-Dec-26 29-Jun-27	A Receive Approval/Dermit - BCER Eacility Permit Ameri
RG1112188 Receive Approval/Permit - BCER Facility Permit Amendment 0d 29-Jun-27	
Base Plant & Tank Demolition 1/bd 25-Jun-27 15-Iviar-28 RGT612182 Prenare Facility Permit Amendment 60d 25- Jun-27 22-Sen-27	Prepare Facility Permit Amendment
RGT612184 Submit Application 3d 17-Sep-27 22-Sep-27	Submit Application
RGT612186 Review Period 116d 22-Sep-27 15-Mar-28	Review Period
RGT612188 Receive Approval/Permit Od 15-Mar-28	Receive Approval/Permit
ING Tark Expansion 401d 03-Mar-26 12-Oct-27	
RGT412182.1 Prepare Notification to Surrounding Neighbors/Right Holders 10d 03-Mar-26 16-Mar-26	ation to Surrounding Neighbors/Right Holders
RGT412182.2 Mail Notification to Surrounding Neighbors/Right Holders (by BCER) 10d 17-Mar-26 30-Mar-26	n to Surrounding Neighbors/Right Holders (by BCER)
RGT412182.3 Public Consultation Period, Receive Responses & Prepare Summary to Support Al 40d 31-Mar-26 27-May-26	ultation Period, Receive Responses & Prepare Summa
Remaining Level of Effort Actual Work Critical Remaining Work Data Date:26-Jul-24	Checked Approved

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Fage. 57 01 75		



' ID	Activity Name	Rem.	Start	Finish	020		20	021		20	22		2023	3	20)24		202	5
		Duration			Q	Q	QQ	Q	Q	QQ	Q	QQ	Q	QQ	QQ	Q	QQ	Q	2 Q
RGT412182	Prepare Facility Permit Amendment Application	62d	18-Jan-27	15-Apr-27															
RGT412184	SubmitApplication	3d	16-Apr-27	20-Apr-27											1				
RGT412186	Review Period	120d	21-Apr-27	12-Oct-27											1				1
RGT412188	Receive Approval/Permit	0d		12-Oct-27											1				
Waste Discharge A	Authorization - EM Act; Oil & Gas Waste Regulation	291d	16-Dec-26	25-Feb-28															
Regasification		195d	16-Dec-26	01-Oct-27															
RGT616723	Prepare Waste Discharge Authorization Application	40d	16-Dec-26	19-Feb-27															1
RGT616724	SubmitApplication	3d	16-Feb-27	19-Feb-27						-				: : :	1				1
RGT616725	Review Period	155d	19-Feb-27	01-Oct-27															
RGT616726	Receive Approval/Permit - Waste Discharge Authorization	0d		01-Oct-27						: ; ;									
Base Plant & Tan	k Demolition *TB Deleted*	203d	29-Apr-27	25-Feb-28										1					1
RGT616727	Prepare Waste Discharge Authorization Application TBC	40d	29-Apr-27	25-Jun-27						1				1	1				1
RGT616728	SubmitApplication	3d	25-Jun-27	30-Jun-27						-	· · ·			: : :					1
RGT616729	Review Period	160d	30-Jun-27	25-Feb-28						1									1
RGT616730	Receive Approval/Permit - Waste Discharge Authorization	0d		25-Feb-28															
LNG Tank Expans	ion	203d	28-Apr-27	23-Feb-28															
RGT412282	Prepare Waste Discharge Authorization Application *Dur. TBC*	40d	28-Apr-27	23-Jun-27															
RGT412284	SubmitApplication	3d	24-Jun-27	28-Jun-27							· · ·			1					1
RGT412286	Review Period	160d	29-Jun-27	23-Feb-28															1
RGT412288	Receive Approval/Permit - Waste Discharge Authorization	0d		23-Feb-28															
Facility NOI - Oil ar	nd Gas Activities Act	21d	09-Jul-27	09-Aug-27															
Regasification		20d	09-Jul-27	09-Aug-27						-				1	1				1
RGA16290	Relevant IFC Drawings; P&IDs, Schematic	5d	09-Jul-27	16-Jul-27							· · ·								
RGA16291	Plot Plan	5d	09-Jul-27	16-Jul-27										-					
RGA16292	Project Description	5d	09-Jul-27	16-Jul-27															
RGA16293	Stamped Record Drawings (Post-NOI submission)	5d	09-Jul-27	16-Jul-27															
RGA16294	Email notification 2 weeks prior to construction start	5d	09-Jul-27	16-Jul-27						-	· · ·			: : :	1				: : :
RGA16288	Prepare Notice of Intent Document	5d	16-Jul-27	23-Jul-27						-				1					1
RGA16289	Review/Approval Period - BCER NOI	10d	23-Jul-27	09-Aug-27										1	1				
LNG Tank Expans	ion	20d	12-Jul-27	09-Aug-27										1					
RGA16297	Relevant IFC Drawings; P&IDs, Schematic	5d	12-Jul-27	19-Jul-27						 - - -									I I I
RGA16298	Plot Plan	5d	12-Jul-27	19-Jul-27						-				1					1
RGA16299	Project Description	5d	12-Jul-27	19-Jul-27						-	· · ·			: : :					1
RGA16300	Stamped Record Drawings (Post-NOI submission)	5d	12-Jul-27	19-Jul-27															1
RGA16301	Email notification 2 weeks prior to construction start	5d	12-Jul-27	19-Jul-27															
RGA16295	Prepare Notice of Intent Document	5d	19-Jul-27	23-Jul-27															
RGA16296	Review/Approval Period - BCER NOI	10d	26-Jul-27	09-Aug-27						1				1					1
INGER - Leave to (203d	02- lun-27	20_Mar_28										1					-
Regasification		30d	02-Jun-27	14-Jul-27															
A10854	Deliverable(s) as per PCTT (See Activity Notebook)	20d	02-Jun-27	29-Jun-27			-						1 1 1 1						1
RGB816390	Prepare and Send Leave to Construct Notification	5d	23-Jun-27	29-Jun-27															
RGB816391	Review Period - LtC	10d	30-Jun-27	14-Jul-27						:			1 1 1 1 1			1 1 1 1 1 1 1			1
Base Plant & Tan	k Demolition	30d	15-Feb-28	29-Mar-28						: : :			1		1				-
					<u>.</u>											<u> </u>			<u> </u>
- Domoiniu I				_												Date		<u> </u>	
 Remaining Lev 				Data Date:2	6-Jul	-24								ŀ	10	~ 04		tier	

Actual Level of Effort

Remaining Work Critical Secondary

Page: 38 of 73

19-Aug-24 IFI

			1	SOLARIS
				MANAGEMENT
		202	6	2027 2028 2029 2030 2031
)	0	202	0 0 0	
	~	<u> </u>	4	Prepare Facility Permit Amendment Application
				I Submit Application
			-	Review Period
		1	-	Receive Approval/Permit
				Prepare Waste Discharge Authorization Application
		1	1	Submit Application
			1	Review Period
				Receive Approval/Permit - Waste Discharge Author
			i i	
			1	Prepare Waste Discharge Authorization Application T
		1	1 1 1	Submit Application
		1	1	Review Period
		1	1	♦ Receive Approval/Permit - Waste Discharge
			!	· · · · · · · · · · · · · · · · · · ·
			1	Prepare Waste Discharge Authorization Application *
				I Submit Application
			1	Review Period
			!	Receive Approval/Permit - Waste Discharge /
				<u> </u>
				Relevant IFC Drawings: P&IDs. Schematic
		-	1	Plot Plan
			1	Project Description
				Stamped Record Drawings (Post-NOI submission)
		-	1 1 1	Email notification 2 weeks prior to construction start
		1	: : :	Prepare Notice of Intent Document
		1	: : :	Review/Approval Period - BCER NOI
		1	1	\blacksquare
			1 1 1	Relevant IFC Drawings; P&IDs, Schematic
		1	: : :	ı) Plot Plan
		1	: : :	I Project Description
		1	1 1 1	I Stamped Record Drawings (Post-NOI submission)
		-	: : :	Email notification 2 weeks prior to construction start
		+-	: : :	Prepare Notice of Intent Document
		1	: : :	Review/Approval Period - BCER NOI
			1	
		1	1	Deliverable(s) as per PCTT (See Activity Notebook)
		+ -		Prepare and Send Leave to Construct Notification
		1	1	Review Period - LtC
		1	1 1 1	₩
_				

Revision	Checked	Approved



/ity ID	Activity Name	Rem.	Start	Finish	020		202	!1	2	2022		2023	3	20	024		202	.5
		Duration			Q	Q	Q Q	QQ	QC	QQ	QQ	QC	Q Q	QC	2 Q	QC	Q	QQ
A10856	Deliverable(s) as per PCTT (See Activity Notebook)	20d	15-Feb-28	15-Mar-28														
RGB816393	Prepare and Send Leave to Construct Notification	5d	08-Mar-28	15-Mar-28														
RGB816394	Review Period	10d	15-Mar-28	29-Mar-28														
LNG Tank Expansi	ion	30d	14-Sep-27	26-Oct-27									1	1				
A10857	Deliverable(s) as per PCTT (See Activity Notebook)	20d	14-Sep-27	12-Oct-27				-										
RGB816395	Prepare and Send Leave to Construct Notification	5d	05-Oct-27	12-Oct-27				-										
RGB816396	Review Period	10d	13-Oct-27	26-Oct-27														
LNGFR - Leave to C	Operate	942d	18-Nov-26	11-Sep-30														
Regasification		30d	18-Nov-26	07-Jan-27														
A10855	Deliverable(s) as per PCTT (See Activity Notebook)	200	18-INOV-26	16-Dec-26	-													1
RGB816389	Prepare and Send Leave to Operate Notification	50	09-Dec-26	16-Dec-26														
RGB816392	Review Period	10d	16-Dec-26	07-Jan-27													- L J .	
LNG Tank Expansi	ion Deliverable(e) as per DCTT (See Activity Netabael()	30d	29-Jul-30	11-Sep-30				-					-					-
A10009	Deliverable(s) as per PCTT (See Activity Notebook)	200	29-Jul-30	27-Aug-30	-													
RGB010399	Prepare and Send Leave to Operate Notification		20-Aug-30	27-Aug-30	-													
RGB816400	Review Period	100	27-Aug-30	11-Sep-30														
Metro Vancouver		445d	02-Dec-26	19-Sep-28								4						·
Waste Discharge -	Metro Vancouver Bylaw 1082	435d	16-Dec-26	19-Sep-28										1				
RGT108182	Prepare Waste Permit Application	2000 20d	16-Dec-26	24-Jan-27				-										
RGT108184	Submit Application	.3d	21-Jan-27	26-Jan-27														-
RGT108186	Review Period	245d	26- lan-27	20 Jan-28														
RGT108188	Receive Approval/Permit - Waste Discharge	0d	20 0011 21	24 Jan-28														
Base Plant & Tank		3284	12_May_27	05-Sen-28										1				
RGT608182	Prepare Waste Permit Application TBC	80d	12-May-27	03-Sep-27														
RGT608184	Submit Application	3d	07-Sep-27	09-Sep-27										-				
RGT608186	Review Period	245d	10-Sep-27	05-Sep-28	-								1					
RGT608188	Receive Approval/Permit - Waste Discharge TBC	0d	10 000 27	05-Sep-28											- - - + - + - + - + - + + - + + + + + + + + + + + + +			
ING Tank Expansi	ion	328d	27_May_27	10-Sep-28				-										
RGT408182	Prepare Waste Permit Application	80d	27-May-27	20-Sep-27														
RGT408184	Submit Application	3d	21-Sep-27	23-Sep-27														
RGT408186	Review Period	245d	24-Sep-27	19-Sep-28														1
RGT408188	Receive Approval/Permit - Waste Discharge	Od	21 000 21	10-Sen-28													- L - J .	
Waste Discharge P	Permit (Air Emission) - AOM Bylaw No. 1082, 2008	445d	02-Dec-26	10-Sep-28								· · ·						1
Regasification		268d	02-Dec-20	10-Jan-28				-										
RGT108282	Prepare Waste Dsicharge Permit Application	20d	02-Dec-26	07-Jan-27														
RGT108284	Submit Application	3d	07-Jan-27	12-Jan-27										1				
RGT108286	Review Period	245d	12-Jan-27	10-Jan-28							+							
RGT108288	Receive Approval/Permit - MV Waste Dsicharge Permit (AIR)	Od		10-Jan-28					1					1				1
Base Plan & Tank	Demolition	328d	12-Mav-27	05-Sep-28														1
RGT408252	Prepare Waste Dsicharge Permit Application	80d	12-May-27	03-Sep-27														
RGT408254	Submit Application	3d	07-Sep-27	09-Sep-27														
RGT408255	Review Period	245d	10-Sep-27	05-Sep-28														
RGT408256	Receive Approval/Permit - MV Waste Discharge Permit (AIR)	0d	1	05-Sep-28				1	1					1				
				I	Ľ			-					<u>'</u>	1	Date			
Remaining Leve	el of Effort Actual Work Critical Remaining Work			Data Date:2	6-Ju	I-24							┝	19 - Au	10-24		IFI	
Actual Level of I	Effort Effort Remaining Work E			Page: 39	of 73	3							ŀ		3-7			





y ID	Activity Name	Rem.	Start	Finish)20		2021	1	2	2022		2023		2024	1	2	.025	
		Duration			QC	Q	QC	Q	Q	QQ	QQ	QQ	QQ	Q	QQ	QC	2 Q	Ċ
LNG Tank Expans	sion *To be deleted if not needed*	328d	27-May-27	19-Sep-28														
RGT408282	Prepare Waste Discharge Permit Application	80d	27-May-27	20-Sep-27														
RGT408284	SubmitApplication	3d	21-Sep-27	23-Sep-27	· · ·												, , , , , , , , , , , , , , , , , , ,	_
RGT408286	Review Period	245d	24-Sep-27	19-Sep-28														
RGT408288	Receive Approval/Permit - MV Waste Discharge Permit (AIR)	Od		19-Sep-28														
City of Delta		364d	07-May-26	26-Oct-27														
Soil Deposit and R	Removal Permit - Bylaw No. 7221, 2016	364d	07-May-26	26-Oct-27				:							1			
Ground Improvem		103d	07-May-26	02-Oct-26														-
RG1306282	Prepare Soil Deposit and Removal Permit Application (See Act. Notebook)	20d	07-May-26	04-Jun-26											-			
RGT306284	Submit Application	3d	05-Jun-26	09-Jun-26														
RGT306286	Review Period	80d	10-Jun-26	02-Oct-26														
RGT306288	Receive Approval/Permit - Soil Deposit	0d		02-Oct-26														
Regasification		100d	26-Jan-27	17-Jun-27	·									-				
RGT106782	Prepare Soil Deposit and Removal Permit Application	20d	26-Jan-27	23-Feb-27														
RGT106784	SubmitApplication	3d	19-Feb-27	23-Feb-27														
RGT106786	Review Period	80d	24-Feb-27	17-Jun-27														
RGT106788	Receive Approval/Permit - Delta City Soil Deposit/Removal	0d		17-Jun-27														
Base Plant & Tan	k Demolition *TB Deleted*	103d	28-May-27	26-Oct-27														
RGT606782	Prepare Soil Deposit and Removal Permit Application	20d	28-May-27	25-Jun-27														
RGT606784	SubmitApplication	3d	25-Jun-27	30-Jun-27														
RGT606786	Review Period	80d	30-Jun-27	26-Oct-27														
RGT606788	Receive Approval/Permit	0d		26-Oct-27				:										
LNG Tank Expans	sion	103d	01-Feb-27	28-Jun-27	· · ·												· · ·	
RGT406782	Prepare Soil Deposit and Removal Permit Application	20d	01-Feb-27	01-Mar-27														
RGT406784	SubmitApplication	3d	02-Mar-27	04-Mar-27														
RGT406786	Review Period	80d	05-Mar-27	28-Jun-27				:										
RGT406788	Receive Approval/Permit	0d		28-Jun-27				:										
Transport Canad	da	279d	05-Nov-26	22-Dec-27											-			
Aeronautical Clear	rance - Canadian Aviation Regulations (CARs)	194d	18-Jan-27	25-Oct-27				- p							·			
Regasification		128d	22-Apr-27	25-Oct-27														
RGT616706	Prepare Aeronautical Clearance Permit Application	5d	22-Apr-27	28-Apr-27														
RGT616707	SubmitApplication	3d	29-Apr-27	03-May-27				: : :										
RGT616708	Review Period	120d	04-May-27	25-Oct-27														
RGT616709	Receive Approval/Permit - Transport Canada (Aeronautical)	0d		25-Oct-27													1 1	
LNG Tank Expans	sion	128d	18-Jan-27	20-Jul-27														
RGT616711	Prepare Aeronautical Clearance Permit Application	5d	18-Jan-27	22-Jan-27				:										
RGT616712	SubmitApplication	3d	25-Jan-27	27-Jan-27														
RGT616713	Review Period	120d	28-Jan-27	20-Jul-27														
RGT616714	Receive Approval/Permit	0d		20-Jul-27														
Navigable Waters	Act Approval - Canadian Navigable Waters Act	279d	05-Nov-26	22-Dec-27				:										
Base Plant		183d	31-Mar-27	22-Dec-27				:						. 1 . 1				
RGT614282	Prepare Navigable Water Permit Application	60d	31-Mar-27	25-Jun-27														
RGT614284	SubmitApplication	3d	25-Jun-27	30-Jun-27														_
RGT614286	Review Period	120d	30-Jun-27	22-Dec-27										1 1			1 1	_
Remaining Los	vel of Effort Actual Work Critical Remaining Work					<u> </u>								De	ate	$\overline{}$		=
				Data Date:20	o-Jul-	24							19-4	Aug-2	24	IF	-1	
				Page: 40	OT / 3													





tivit	y ID	Activity Name	_Rem.	Start	Finish	020)		202	21		202	22		2023		20)24		20	25
			Duration				Q Q	Q	Q	Q	ຊໄດ	Q	QQ	Q	QQ	Q	QC	i Q (2 Q	Q	Q
	RGT614288	Receive Approval/Permit	0d		22-Dec-27								1		1						
	LNG Tank		183d	05-Nov-26	04-Aug-27								-		1 1 1	1					1
	RGT616719	Prepare Navigable Water Permit Application	60d	05-Nov-26	05-Feb-27								-		1 1 1						1
	RGT616720	Submit Application	3d	08-Feb-27	10-Feb-27										, , , , , , , , , , , , , , , , , , , ,			1 1 1 1 			
	RGT616721	Review Period	120d	11-Feb-27	04-Aug-27										1						
	RGT616722	Receive Approval/Permit	0d		04-Aug-27										1						
	Ministry of Enviro	nment	103d	07-May-26	02-Oct-26										1						
	Site Profile Submiss	ion (TBD) - EM Act; Contaminated Sites Regulation	103d	07-May-26	02-Oct-26																
	Ground Improvemen	nt Dreneys FM Demeit Application (Fermer/Feren Attached Deliverables)	103d	07-May-26	02-Oct-26																
	RG1309182	Prepare Ew Permit Application (Forms/Fees, Attached Deliverables)	200	07-IVIAy-20	04-Jun-26	_									1						1
	RG1309184		30	05-Jun-26	09-Jun-26								-		1 1 1						-
	RG1309186	Review Period	800	10-Jun-26	02-Oct-26	_	-						-		1 1 1	1					1
	RGT309188	Receive Approval/Permit - Environment Management	0d		02-Oct-26								-		1 . 1						
	Auxiliary Systems ((Utility Rack & Equipment) Tie-in to T3 Tank	932d	18-Jan-27	18-Oct-30																
	EPC (Detailed - Co	ontractor)	932d	18-Jan-27	18-Oct-30																
ſ	General		932d	18-Jan-27	18-Oct-30		-								1						1
	G136		Ud	18-Jan-27	00.04 07	_							-		1						1
	G166	HAZOP/SIL/LOPA	Ud		26-May-27								-		1 1 1						1
	G266	60% Model Review	0d		30-Jun-27																
	E15991	Construction Permit Approval	0d	16-Jul-27											1 1 1		-				1
	G176	90% Model Review	0d		27-Aug-27										1						
	E16001	Ground Improvement Completed	0d		30-Sep-27		-								1						1
	G226	Construction Mobilization to Site	0d	05-Oct-27									-		1						
	G196	Final Document Class 2 Control Estimate Completed	0d		09-Nov-27																
	G186	Issue for Construction (IFC)	0d		24-Nov-27								-		1 1 1	1					
	E18631	Pipe Fabrication Complete	0d		29-May-28								-		1 1 1	1					
	G236	Last Piece of Mechanical Equipment Delivered to Site	0d		06-Jun-28								-		1 . 1						
	G216	Project Mechanical Completion	0d		20-Aug-30										. I		-				1
	G246	Commissioning & Startup Completed	0d		18-Oct-30										1						
	G256	Turnover to Operations - LNG Tank	0d		18-Oct-30															1	
	Engineering & Desig	jn	215d	18-Jan-27	24-Nov-27		-								1						1
	General		120d	19-May-27	09-Nov-27								-		1						
	E16111	HAZOP	5d	19-May-27	26-May-27								-		1						
	E16121	HAZOP Closeout	20d	27-May-27	23-Jun-27										,						
	G126	60% Model Review	5d	24-Jun-27	30-Jun-27								-		1 1 1						
	G206	90% Model Review	5d	23-Aug-27	27-Aug-27								-								1
	E19901	Prepare Class 2 Control Estimate	30d	28-Sep-27	09-Nov-27		-								1						
	Milestones		195d	12-Feb-27	24-Nov-27										1						
	Process	Simulation IFO	145d	12-Feb-27	13-Sep-27																
	E 18051	Simulation - IFC			12-FeD-27										1		-				1
	E18061		Ua		12-Feb-2/					÷					1 1 1						1
	E18/11	Hydraulic Calculations/Line Sizing - IFC	Ud		15-Mar-27		:						-	1	1 1 1	1					1
	E18/21	Process Flow Diagram - IFH	Ud		15-Mar-27	ł									, , ,	1					1
_																\neg		Date		\square	
_	Remaining Level	I OT ETTOT			Data Date:2	6-J	ul-2	24								1	19-Au	g-24		İFI	
-	Actual Level of E	TTOIT EFFE Remaining Work FFE Critical Secondary			Page: 41	of	73									F		<u> </u>		+	





Activit	y ID	Activity Name	Rem.	Start	Finish D2	20 2021 2022 2023	2024 2025	2026 2027 2028	2029 2	2030 2031
			Duration		Ţ					
	E18691	Utilities Requirement - IFH	0d		29-Mar-27			♦ Utilities Requirement -	IFH	
	E15191	P&ID - IFH	0d		21-Apr-27			◆ P&ID - IFH		
	E18761	Loss Management Philosophy - IFH	0d		21-Apr-27			Loss Management Pl	nilosóphy - IFH	
	E18821	Control Narrative - IFH	0d		21-Apr-27			 Control Narrative - IFI 	H	
	E18671	Material Selection Diagram - IFH	Od		27-Apr-27			 Material Selection Dia 	igram - IFH	
	E18741	Utility Flow Diagram - IFH	0d		27-Apr-27			 Utility Flow Diagram - 	IFH	
	E18771	Operating Philosophy - IFH	0d		18-May-27			 Operating Philosoph 	y-IFH	
	E18801	Pressure Relief Summary - IFH	0d		18-May-27			♦ Pressure Relief Sum	mary - IFH	
	E19831	Vendor P&ID - IFH	0d		18-May-27			◆ Vendor P&ID - IFH		
	E18731	Process Flow Diagram - IFC	0d		08-Jul-27			Process Flow Diag	ram - IFC	
	E16021	P&ID - IFC	Od		29-Jul-27			♦ P&ID - IFC		
	E18781	Loss Management Philosophy - IFC	0d		29-Jul-27			♦ Loss Managemer	it Philosophy - IF	C
	E19861	Control Narrative - IFC	0d		29-Jul-27			♦ Control Narrative -	- IFC	
	E16031	C&SU Procedure - IFC	0d		06-Aug-27			♦ C&SU Procedure	-IFC	
	E18701	Utilities Requirement - IFC	0d		06-Aug-27			♦ Utilities Requirem	ent - IFC	
	E18681	Material Selection Diagram - IFC	0d		20-Aug-27			♦ Material Selection	ı Diagram - IFC	
	E18751	Utility Flow Diagram - IFC	0d		20-Aug-27			♦ Utility Flow Diagra	am - IFC	
	E19851	Vendor P&ID - IFC	0d		20-Aug-27			♦ Vendor P&ID - IF	C	
	E18791	Operating Philosophy - IFC	0d		13-Sep-27			 Operating Philos 	ophy - IFC	
	E18811	Pressure Relief Summary - IFC	Od		13-Sep-27			♦ Pressure Relief:	Summary - IFC	
	Civiil/Structural/	Architectural	185d	01-Mar-27	24-Nov-27			· · · · · · · · · · · · · · · · · · ·		
	E18831	Civil Engineering Design Criteria - IFC	0d		01-Mar-27			♦ Civil Engineering Desig	n Criteria - IFC	
	E18841	Structural Engineering Design Criteria - IFC	0d		01-Mar-27			♦ Structural Engineering I	Design Criteria -	IFC
	E18851	Piperack Structural Capacity Report - IFC	0d		01-Mar-27			 Piperack Structural Car 	pacity Report - IF	C
	E18881	C/S/A Standard Drawings & Details - IFC	Od		01-Mar-27			♦ C/S/A Standard Drawin	gs & Details - IF(C
	E18891	C/S/A Specifications - IFC	Od		01-Mar-27			 C/S/A Specifications - IF 	C	
	E15231	Structural Steel Location Plans - IFC	0d		27-Sep-27			♦ Structural Steel	Location Plans -	IFC
	E15241	Building Drawings - IFC	0d		12-Oct-27			 Building Drawin 	gs - IFC	
	E15261	Foundation Details & Drawings - IFC	0d		12-Oct-27			 Foundation De 	tails & Drawings	- IFC
	E15281	Earthworks/Civil Drawings - IFC	0d		12-Oct-27			♦ Earthworks/Civil	il Drawings - IFC	
	E18871	Structural Steel MTO for Estimating	0d		12-Oct-27			 \$tructural Steel 	MTO for Estimat	ing
	E18861	MIscellaneous Steel - Support Details - IFC	0d		24-Nov-27			♦ Miscellaneous	s Steel - Support	Details - IFC
	Mechanical		105d	01-Mar-27	29-Jul-27			v		
	E19871	Equipment Specifications - IFC	0d		01-Mar-27			 Equipment Specification 	ns - IFC	
	E15301	Equipment List - IFH	0d		21-Apr-27			♦ Equipment List - IFH		
	E18901	Fire Fighting Equipment List - IFH	0d		21-Apr-27			Fire Fighting Equipment	ent List - IFH	
	E18921	Spare Parts Listing	0d		04-May-27			♦ Spare Parts Listing		
	E17741	Equipment List - IFC	0d		29-Jul-27			♦ Equipment List - IF	=C	
	E18911	Fire Fighting Equipment List - IFC	0d		29-Jul-27			♦ Fire Fighting Equip	oment List - IFC	
	Piping Engineeri	ing	135d	01-Mar-27	13-Sep-27			T		
	E18941	Piping Specifications - IFC	Od		01-Mar-27			 Piping Specifications - II 	FC	
	E15521	LDT - IFH	0d		21-Apr-27			◆ LDT - IFH		
							Date	Revision	Checkod	Approved
	Remaining Lev	vel of Effort Actual Work Critical Remaining Work			Data Date:26-	Jul-24	-Aug-24 IFI			Αμισνέα
	Actual Level of	Effort Effort Remaining Work E			Page: 42 of	73	<u> </u>			

SOLARIS



Activi	ty ID	Activity Name	Rem.	Start	Finish D2	0 2021	1	2022 20	023	2024	2025	2026	2027	2028	2029	2030	2031
			Duration						QQ		Q Q Q Q						
	E16051	LDT-IFC	0d		29-Jul-27								♦ LL)] - IFC			
	E15561	Stress Analysis Reports - IFC	0d		13-Sep-27								•	stress Analysis	Reports - IFC		
	E18931	SP Item List - IFC	0d		13-Sep-27								• • •	SP Item List - II	-C		
	Piping Design		160d	01-Mar-27	19-Oct-27									nnort Dotoile		· · · · · · · · · ·	
	E19031	Pipe Support Details - IFC	0d		01-Mar-27									tondord Drow	linge & Detaile	IEC	
	E19041	Piping Standard Drawings & Details - IFC	0d		01-Mar-27										villigs & Details	- 16 C	
	E19841	Plot Plan - IFH	0d		21-Apr-27												
	E18951	Tie-in List - IFH	0d		18-May-27								♦ He-Ir	I LISI: - IFFI			
	E19051	Fire Protection & Safety Equipment Layout - IFH	0d		18-May-27		· · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·		♦ Fire	Protection & S	atety Equipme	nt Layout	- 11- 11
	E19081	Plot Plan - IFC	0d		29-Jul-27								♦ Pic	ot Plan - IFC			
	E15591	Equipment Location Plan - IFC	0d		20-Aug-27								♦E	quipment Loca	ation Plan - IHC		
	E19021	Utility Location Plan - IFC	0d		20-Aug-27								♦U	tility Location F	lan - I⊢C		
	E18961	Piping Tie-in List - IFC	0d		13-Sep-27								•	Piping Tie-in Lis	st - IFC		
	E19011	Equipment General Arrangement Drawings - IFC	0d		13-Sep-27						· · · ·			-quipment Ge	neral Arrangen	nent Drav	vings - IFC
	E15551	Piping (PVF, Insulation Piping SP Items) MTO for Estimating	0d		12-Oct-27								•	Piping (PVF, Ir	hsulation Piping	sP Item	s) MTO for I
	E18971	General Arrangement Key Plan - IFC	0d		12-Oct-27								•	GeneralArrar	igement Key P	lan - IFC	
	E18981	Model Key Plan - IFC	0d		12-Oct-27								•	Model Key Pla	an - IFC		
	E18991	Construction Isos - IFC	0d		12-Oct-27								•	Construction I	sos - IFC		
	E19061	Fire Protection & Safety Equipment Layout - IFC	0d		12-Oct-27									Fire Protection	n & Safety Equ	pment La	ayout - IFC
	E19001	Piping General Arrangement Drawings - IFC	0d		19-Oct-27								•	Piping Gener	alArrangemen	t Drawing	js-IFC
	Electrical		175d	01-Mar-27	09-Nov-27												
	E19181	Electrical Standard Drawings & Details - IFC	0d		01-Mar-27									al Standard Dr	awings & Deta	lls - IFC	
	E19191	Electrical Specifications - IFC	0d		01-Mar-27									al Specification	ls-IFC		
	E19261	Grounding/Cathodic Protection Details - IFC	0d		01-Mar-27						1 1 1 1 1 1 1 1 1		♦ Ground	ing/Cathodic I	Protection Deta	lls - l⊢C	
	E19121	Area Classification Diagram - IFH	0d		18-May-27								♦ Area	Classification	Diagram - IFH		
	E19101	Single Line Diagram - IFC	0d		27-Aug-27								♦ S	ingle Line Dia	gram - IFC		
	E19111	Cable Schedule - IFC	0d		27-Aug-27								◆ C	able Scheduk	e - IFC		
	E19091	Electrical Load List - IFC	0d		13-Sep-27								• E	Electrical Load	List - IFC		
	E19131	Area Classification Diagram - IFC	0d		13-Sep-27								/ •	Area Classifica	tion Diagram -	IFC	
	E19251	Tie-In List - IFC	0d		13-Sep-27								•	lie-In List - IFC			
	E19431	ETAP/Arc Flash - IFC	0d		13-Sep-27								• E	TAP/Arc Flas	h - IFC		
	E19141	24V Distribution Wiring Schematics - IFC	0d		27-Sep-27								•	24V Distributio	n Wiring Sche	natics - If	−C
	E19151	120V Schematics - IFC	0d		27-Sep-27								•	120V Schema	itics - IFC		
	E19161	600V/480V Schematics - IFC	0d		27-Sep-27						1 1 1 1 1 1 1 1 1		· · · · · · · · · · · · · · · · · · ·	600V/480V So	chematics - IFC		
	E19171	4160V Schematics - IFC	0d		27-Sep-27								•	4160V Schem	atics - IFC		
	E15751	Issue Electrical (Cable/Tray/Conduit/Junction Box) MTO for Estimating	0d		12-Oct-27								•	Issue Electrica	al (Cable/Tray/(Conduit/J	unction Box)
	E19211	Junction Box Layout - IFC	0d		12-Oct-27								•	Junction Box	Layout - IFC		
	E19221	Contactor Panel Layout - IFC	0d		12-Oct-27									Contactor Par	nel Layout - IF(
	E19231	Control Panel Layout - IFC	0d		12-Oct-27						1 1 1 1 1 1 1 1 1	1 1 1	•	Control Panel	Layout - IFC		
	E19241	Distribution Panel Drawings - IFC	0d		12-Oct-27								•	Distribution Pa	anel Drawings	IFC	
	E19281	Electrical Equipment Layout - IFC	0d		12-Oct-27									Electrical Equi	ipment Layout	- IFC	
	E19291	New Substation Layout - IFC	0d		12-Oct-27								•	New Substation	on Layout - IFC		
	- Domokolis - I									Date		Re	vision		Checked	A	
		Effort Critical Remaining Work			Data Date:26-	Jul-24			1	9-Aug-24	IFI						<u></u>
	Actual Level of				Page: 43 of	15											

SOLARIS

M A N A G E M E N T CONSULTANTS INC.



Vity ID	Activity Name	Rem.	Start	Finish	020	2021		2022	2023	3	20)24	20	025
		Duration			QQ	QQC	Q		QC	λQ	QQ			i Q
E19201	Input/Output Drawings - IFC	Od		09-Nov-27										
E19271	Cable, Raceway, Lighting & Grounding Layout - IFC	Od		09-Nov-27										
E19301	Cable Tray Layout - IFC	0d		09-Nov-27										
E19891	Heat Tracing Layout - IFC	0d		09-Nov-27										
Instrumentation		175d	01-Mar-27	09-Nov-27										÷
E19401	Standard Drawings & Details - IFC	Od		01-Mar-27										
E19411	I&C Specifications - IFC	0d		01-Mar-27										
E15901	Shut Down Key - IFH	0d		21-Apr-27										-
E15921	Instrument Index - IFH	Od		18-May-27										
E15941	Alarm Set Point Table - IFH	Od		18-May-27										
E19341	Metering Schematics - IFH	Od		18-May-27										
E19361	Fire & Gas Detection Drawings - IFH	0d		18-May-27										
E19381	PCS/PLC/SIS/SCADA Architecture Diagram - IFH	0d		18-May-27					 					
E16081	Shut Down Key - IFC	0d		29-Jul-27	-									
E16101	Alarm Set Point Table - IFC	Od		13-Sep-27										:
E17761	Instrument Index - IFC	0d		13-Sep-27										
E19311	Svstem Block Diagram - IFC	0d		13-Sep-27										
E19331	I&C Tie-In Drawing - IFC	0d		13-Sep-27					 					
E19351	Metering Schematics - IEC	0d 0d		13-Sep-27										
F19371	Fire & Gas Detection Drawings - IFC	0d 0d		13-Sep-27										
E10071	PCS/PLC/SIS/SCADA Architecture Diagram - IEC	0d		13-Sen-27	-									
E15061		Dd Od		12-0ct-27	-									
E10301		00		00 Nov 27	· · · · · · · ·				 					
Brocoss		00 165d	19 Jan 07	12 Sop 27										
Simulation		20d	18-Jan-27	12-Feb-27										
E17811	Simulation - IFC	20d	18-Jan-27	12-Feb-27					· · ·					
Heat & Material	Balance	20d	18-Jan-27	12-Feb-27										
E17821	Heat & Material Balance - IFC	20d	18-Jan-27	12-Feb-27										
Material Selection	on Diagram	110d	16-Mar-27	20-Aug-27										:
E17831	Material Selection Diagram - IFH	30d	16-Mar-27	27-Apr-27										
E17841	Material Selection Diagram - IFC	30d	09-Jul-27	20-Aug-27					· · ·					
Equipment Data	Sheet	10d	16-Mar-27	29-Mar-27		· · ·			 					
E17961	BOG Compressor Sizing Calculations - IFP	10d	16-Mar-27	29-Mar-27										
E17971	LNG Loading Pump Sizing Calculations - IFP	10d	16-Mar-27	29-Mar-27										
Utilities		120d	16-Feb-27	06-Aug-27										:
E17791	Utilities Requirement - IFH	30d	16-Feb-27	29-Mar-27										
E17801	Utilities Requirement - IFC	30d	24-Jun-27	06-Aug-27		 								
Hydraulic Calcu	lations	20d	16-Feb-27	15-Mar-27					· · ·					
E17981	Hydraulic Calculations/Line Sizing - IFC	20d	16-Feb-27	15-Mar-27										
		130d	16-Feb-27	20-Aug-27										
E17771	Process Flow Diagram Development - IFH	20d	16-Feb-27	15-Mar-27					· · ·			1 1 1 1 1 1 1 1		
E17911	Utility Flow Diagram Development - IFH	30d	16-Mar-27	27-Apr-27	:				 					
E17791	Process Flow Diagram Development - IFC	10d	24-Jun-27	08-10-27		1 1			i i		, i .			i.

	Remaining Level of Life
-	Actual Level of Effort

Remaining Work Critical Secondary

condary

Data Date:26-Jul-24 Page: 44 of 73 19-Aug-24 IFI

SOLARIS
MANAGEMENT CONSULTANTS INC.
2026 2027 2028 2029 2030 2031
♦ Input/Output Drawings - IFC
Caple; Raceway, Lignting & Grounding Layout - Cable Travel accert unco
♦ Standard Drawings & Details - IFC
♦ I&C Specifications - IFC
♦ Shut Down Key - IFH
♦ Instrument Index - IFH
♦ Alarm Set Point Table - IFH
♦ Metering Schematics - IFH
♦ Fire & Gas Detection Drawings - IFH
PCS/PLC/SIS/SCADA Architecture Diagram - IFH
♦ Shut Down Key - IFC
♦ Alarm Set Point Table - IFC
Instrument Index - IFC
♦ System Block Diagram - IFC
♦ I&C Tie-In Drawing - IFC
Metering Schematics - IFC
Fire & Gas Detection Drawings - IFC
♦ PCS/PLC/SIS/SCADA Architecture Diagram - IFC
♦ Instrumentation & Controls MTO
♦ Instrument Layout + IFC
Simulation - IFC
□ Heat & Material Balance - IFC
Material Selection Diagram - IFH
🔲 Material Selection Diagram - IFC
♥ ■ BOG Compressor Sizing Calculations - IFP
LNG Loading Pump Sizing Calculations - IFP
Utilities Requirement - IFH
🗖 Utilities Requirement - IFC
\blacksquare
□ Hydraulic Calculations/Line Sizing - IFC
Process Flow Diagram Development - IFH
📋 Utility Flow Diagram Development - IFH
Process Flow Diagram Development - IFC

Revision	Checked	Approved



ctivity ID	Activity Name	Rem. Duration	Start	Finish	20	2021	2022 2023			
F17921	Utility Flow Diagram Development - IEC	30d	09-Jul-27	20-Aug-27					Utility Flow Di	agram Development - IFC
Piping & Instrum	ent Diagram (P&ID)	105d	23-Mar-27	20-Aug-27					—	
E15181	P&ID Development IFH	20d	23-Mar-27	21-Apr-27					P&ID Developme	ntIFH
F17931	Vendor P&ID - IFH	20d	21-Apr-27	18-May-27					Vendor P&ID - IF	Ή
E16011	P&ID Development IFC	15d	09-Jul-27	29-Jul-27					P&ID Develop	mentIFC
F17941	Vendor P&ID - IFC	.30d	09-10-27	20-Aug-27					Vendor P&ID	- IFC
	Sheet	10d	16-Eeb-27	01_Mar_27						
E17951	Instruments Sizing Calculations - IFP	10d	16-Feb-27	01-Mar-27					Instruments Sizing	Calculations - IFP
Loss Managemer	nt Philosophy	100d	09-Mar-27	29-Jul-27					· · · · · · · · · · · · · · · · · · ·	
E17851	Loss Management Philosophy - IFH	30d	09-Mar-27	21-Apr-27			L J L		📋 Loss Managemei	nt Philosophy - IFH
E17861	Loss Management Philosophy - IFC	30d	17-Jun-27	29-Jul-27					📋 Loss Manage	ment Philosophy - IFC
Operating Philos	ophy	100d	21-Apr-27	13-Sep-27						
E17871	Operating Philosophy - IFH	20d	21-Apr-27	18-May-27					Operating Philos	ophy-IFH
E17881	Operating Philosophy - IFC	30d	30-Jul-27	13-Sep-27					🗖 Operating P	hilosophy - IFC
Pressure Relief	Summary	100d	21-Apr-27	13-Sep-27						
E17891	Pressure Relief Summary - IFH	20d	21-Apr-27	18-May-27					Pressure Relief S	Summary - IFH
E17901	Pressure Relief Summary - IFC	30d	30-Jul-27	13-Sep-27					🔲 Pressure Re	lief Summary - IFC
Control Narrative		95d	23-Mar-27	06-Aug-27					• • • • • • • • • • • • • • • • • • •	
E18641	Prepare Control Narrative - IFH	20d	23-Mar-27	21-Apr-27					Prepare Control 1	Varrative - IFH
E19691	Prepare Control Narrative - IFC	30d	17-Jun-27	29-Jul-27					Prepare Continuity	ol Narrative - IFC
E15201	Prepare C&SU Procedure - IFC	30d	24-Jun-27	06-Aug-27					🔲 Prepare C&S	U Procedure - IFC
Civil/Structural/Ar	chitectural	215d	18-Jan-27	24-Nov-27					-	
Design Criteria		30d	18-Jan-27	01-Mar-27						
E17991	Civil Engineering Design Criteria - IFC	30d	18-Jan-27	01-Mar-27					🗖 Civil Engineering D	esign Criteria - IFC
E18001	Structural Engineering Design Criteria - IFC	30d	18-Jan-27	01-Mar-27					Structural Engineer	ing Design Criteria - IFC
Modeling		150d	18-Jan-27	20-Aug-27						
G156	C/S/A Modeling to 60%	104d	18-Jan-27	15-Jun-27					C/S/A Modeling	to 60%
E15211	C/S/A Modeling to 90%	35d	02-Jul-27	20-Aug-27					🗖 C/\$/A Modelir	ng to 90%
Sitework		30d	30-Aug-27	12-Oct-27					₩	
E15271	Earthworks/Civil Drawings - IFC	30d	30-Aug-27	12-Oct-27					Earthworks	/Civil Drawings - IFC
Foundation		30d	30-Aug-27	12-Oct-27						
E15251	Foundation Details & Drawings - IFC	30d	30-Aug-27	12-Oct-27						Details & Drawings - IFC
Structural Steel		215d	18-Jan-27	24-Nov-27						
E18041	Piperack Structural Capacity Report - IFC	30d	18-Jan-27	01-Mar-27						
E15221	Structural Steel Drawings - IFC	10d	14-Sep-27	27-Sep-27	_					
E18621	Miscellaneous Steel - Support Details - IFC	30d	13-Oct-27	24-Nov-27						eous Steel - Support Details - IFC
Buildings		30d	30-Aug-27	12-Oct-27						
E18021	Building Drawings - IFC	30d	30-Aug-27	12-Oct-27						
Material Take Off	Structural Stack MTO fair Estimation	30d	30-Aug-27	12-Oct-27						teel MTO for Estimating
E18051	Structural Steel MIO for Estimating	300	30-Aug-27	12-Uct-27						
Standard Drawing	gs & Details	30d	18-Jan-27	01-Mar-27					C/S/A Standard Dra	wings & Details - IFC
E 10011	C/S/A Standard Drawings & Details - IFC	300	10-Jan-27	01-Mar-27						
F18031	C/S/A Specifications - IEC	30d	18-Jan-27	01-Mar-27					C/S/A Specifications	s-IFC
Requisition		45d	18-lan-27	22-Mar-27						
		450	10-5411-21							
Remaining Lev	el of Effort Actual Work Critical Remaining Work			Data Date:2	6-Jul-2	4			Revision	Checked Approved
Actual Level of	Effort Remaining Work Critical Secondary			Page: 45	of 73			IS-Aug-24		

SOLARIS



Image: Note: State in the state in the																CON	SULTANTS	INC.	
	ity ID	Activity Name	Rem. Duration	Start	Finish	20	202		2023	3 2		2025	2026	2027		028	2029	2030	2031
File 2014/C-2014 COUNT-ROG Names - Person Streets & Spectra Streets 2014 L.Sub / 19 Person / 19 File 2014/C-2014 COUNT-Streets 01 05-48-472 224-4927 File 2014/C-2014 COUNT-Streets 00 05-48-472 224-4927 File 2014/C-2014 COUNT-Streets 00 05-48-472 224-4927 File 2014/C-2014 COUNT-Streets 00 05-48-272 224-4927 File 2014/C-2014 COUNT-Streets 00 05-48-272 224-4927 File 2014/C-2014 COUNT-Streets 00 15-48-27 224-4927 File 2014/C-2014 COUNT-Streets 00 15-48-27 224-4927 File 2014/C-2014 COUNT-Streets 00 00-48-27 224-4927 File COUNT-Streets 00 00-48-27 224-4927 15-48-27 224-4927 File COUNT-Streets 00 00-48-27 224-4927 15-48-27 224-4927 15-48-27 224-4927	20014-C-5301 · F	BOG Building	45d	18- lan-27	22_Mar_27														
Provide Status Control Option Opti	E15821	20014-C-5301 - BOG Building - Prepare IFP Data Sheets & Specifications	20d	18-Jan-27	12-Feb-27									2001	4-C-5301	I - BOG I	Building - Pre	pare IFP	^o Data Sheets
E1371 2010 4 - 2011 - POC Bulking - Revery Weard Date 100 29 Aug 27 22 Aug 27 E1371 2010 4 - 2017 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2010 - 2017 Aug 2017 A	E19721	20014-C-5301 - BOG Building - Receive Vendor Info	0d	09-Mar-27										♦ 200	14-¢-530	1 - BOG	Building - Re	ceive Ve	endor Info
EVEN 2011 - C-2011	E19731	20014-C-5301 - BOG Building - Review Vendor Data	10d	09-Mar-27	22-Mar-27		-							1 200	14-¢-530	01 - BOG	Building - Re	eview Ve	ndor Data
Bit Add State Concernence Cold Bit Add State Concernence Cold Bit Add State Concernence Cold Concernence Concernence	F19741	20014-C-5301 - BOG Building - Code 1 Vendor Data	0d		22-Mar-27				· · · · · · · · · · · · · · · · · · ·					♦ 200	14-C-530	01 - BOC	Building - Co	ode 1 Ve	ndor Data
E1731 2014 - 6 2012	20014-C-5302: 0	Galvanized Steel	20d	18-Jan-27	12-Feb-27												Ū		
Source Active	E17331	20014-C-5302: Galvanized Steel - Prepare IFP Data Sheets & Specifications	20d	18-Jan-27	12-Feb-27									2001	4-C-5302	2: Galvar	nized Steel - F	Prepare I	FP Data Shee
E17371 20014 C_SS30. Concrete: Prepare IPP Data Streets & Specifications 200 9 dual xx 7 12 feb X7 Bitdet SXXE III devide: Fig Add XX Fill	20014-C-5303: (Concrete	20d	18-Jan-27	12-Feb-27									₩					
Status Diff Diff <thdiff< th=""> Diff Diff <</thdiff<>	E17371	20014-C-5303 - Concrete - Prepare IFP Data Sheets & Specifications	20d	18-Jan-27	12-Feb-27									2001	4-C 5303	3 - Concr	ete - Prepare	IFP Dat	ta Sheets & S
ET731 2004 C-6330X - F1 Materials - Propert FP Data Sheek & Specificature 704 15.4.4.77 7.9.4.677 Externate 1505 15.4.4.77 7.9.4.677 17.4.4.77 7.9.4.677 Externate 1505 0.5.4.4.77 7.9.4.677 17.4.4.77 E1507 Faugment List - FH 300 0.6.4.4.77 7.1.4.4.77 E1507 Faugment List - FH 300 0.6.4.4.77 7.1.4.4.77 E1507 Equipment List - FH 300 1.6.4.4.77 7.4.4.77 E1507 Equipment Sector List - FC 300 1.5.4.4.77 6.4.4.6.72 1.4.4.77 E1601 Equipment Sector List - FC 300 1.5.4.4.77 0.4.4.6.72 1.5.4.8.72 0.4.4.6.72 E1611 Specifications - FC 300 1.5.4.8.72 0.4.4.6.72 1.5.4.8.72 0.4.4.6.72 E1511 Specifications - FC 300 1.5.4.8.72 0.4.4.6.72 1.5.4.8.72 0.4.4.6.72 E1511 Specifications - FC 1.5.4.8.72 0.4.4.6.72 0.4.4.6.72 1.4.4.6.72 E1511 Specifications - FC 1.5.4.8.72 0.4.4.6.72 0.4.4.6.72	20014-C-53XX: I	Fill Materials	20d	18-Jan-27	12-Feb-27														
Interacted 1056 IB-Jan 27 29-Jul 27 Engineerit 0.00 0.044-27 21-Jul 27 Segments FEG 0.00 21-Jul 27 20-Jul 27 Segments FEG 0.00 21-Jul 27 20-Jul 27 E15311 2014-45.00 0.00 21-Jul 27 20-Jul 27 20-Jul 27 E15311 2014-45.00 0.00 21-Jul 27 20-Jul 27 20-Jul 24-Stort 10-Stort Partice	E17431	20014-C-53XX - Fill Materials - Prepare IFP Data Sheets & Specifications	20d	18-Jan-27	12-Feb-27									2001	4-C + 53X)	K - Fill Ma	aterials - Prep	are IFP	Data Sheets 8
Elevent Lit FM Bool Bookstard Zit Apr 27 Zit Apr 27 ET3381 The Fighting Equipment Lit - FH 300 De Aur 27 Zit Apr 27 Zit Apr 27 ET3381 The Fighting Equipment Lit - FC 301 To Aur 27 Zit Apr 27	Mechanical		135d	18-Jan-27	29-Jul-27									V					
E1301 E1001mm1Lbs-inf 0.00 0-400-2 21-40-27 E1301 E1001mm1Lbs-inf 0.00 0-400-2 21-40-27 E1301 E1001mm1Lbs-inf 0.00 17-Jun-27 20-Jul-27 20-Jul-27 E1301 E1001mm1Lbs-inf 0.00 17-Jun-27 20-Jul-27 20-Jul-27 20-Jul-27 E1401 Space Fars 756 18-Jun-27 20-Jul-27 20-Jul-27 20-Jul-27 E1301 Space Fars 656 0.00-400-27 20-Jul-27 0-40-40-27 11-40-47 11-40-47	Equipment List		100d	09-Mar-27	29-Jul-27									Éa	uinment l	ist - IFH			
E1333 CPF Enging 12-phremit.LE-TPC 300 09-087/7 27-04/27 28-04/27 E13731 Edupment.LE-TPC 304 17-Jun-27 28-04/27 24-04/27 E14701 Equipment.LE-TPC 304 17-Jun-27 28-04/27 Edupment.LE-TPC ImF Enging Equipment.LE-TPC ImF Enging Engine ImF Enging Enginenc ImF Enging Engine Enging	E15291		300	09-Mar-27	21-Apr-27										- Eighting		ontlict IEH		
E1/37 Caliptent ILL® - IC 300 1/LUR2/1 28/Jul/2/ E18401 Free Fighting Equipment Like - IFC 560 16-Jun/2/1 28/Jul/2/ Spens Print Free Fighting Equipment Like - IFC 560 16-Jun/2/1 28/Jul/2/ E19701 Equipment Specifications - IFC 16-Jun/2/1 28/Jul/2/ 16-Jun/2/1 28/Jul/2/ E10311 2001-4/Mol10 - Cyre Pampa - Review Vendor Tobia 0 21-Apc/27 0-Maty/27 28/Jul/2/ 1/Jul/2/1 28/Jul/2/1 1/Jul/2/1 28/Jul/2/1 1/Jul/2/1 28/Jul/2/1 1/Jul/2/1 28/Jul/2/1 1/Jul/2/1 28/Jul/2/1 28/Jul/2/1 28/Jul/2/	E18391		300	09-Mar-27	21-Apr-27											nt lict			
E18301 FMF Signing Cappment L1 + IC 3001 1 / Jun / 2 304 / Jun / 2 304 / Jun / 2 304 / Jun / 2 Jun / 2 <t< td=""><td>E17731</td><td></td><td>30d</td><td>17-Jun-27</td><td>29-Jul-27</td><td>_</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>in mont list</td><td></td><td></td></t<>	E17731		30d	17-Jun-27	29-Jul-27	_											in mont list		
Specifications - IF C 76d 19-Jun - 27 01-Max - 27	E18401	Fire Fighting Equipment List - IFC	30d	17-Jun-27	29-Jul-27										rile rign	iung ⊏qu	ipment List -	ГC	
E H0/101 E H0/101	Spare Parts	Environment On a filtration of IEO	75d	18-Jan-27	04-May-27									Equi	nment Sn	ecificatio	ns - IFC		
E19411 Spall Parts Larg 013 21/44/32 04/48/32 Requisition 653 02/46/27 30/14/45101 Cyn Pumps - Prepare RFP Data Sheeta Specifications 200 20014/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Info 00 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Data 100 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Data 100 21/46/27 30/14/45101 Cyn Pumps - Revete Vendor Data 100 10/14/45102 20014/4/5101 Cyn Pumps - Revete Vendor Data 100 20/14/45101 Cyn Pumps - Revete Vendor Data 100 20/14/45101 Cyn Pumps - Revete Vendor Data 100 20/14/45102 20/14/45102 20/14/45102 20/14/45102 20/14/45102 20/14/45102 20/14/45102 20/14/45102 20/14/45102 <t< td=""><td>E19701</td><td>Equipment Specifications - IFC</td><td>300</td><td>18-Jan-27</td><td>01-Mar-27</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>are Parts</td><td>e listina</td><td>///J - // O</td><td></td><td></td></t<>	E19701	Equipment Specifications - IFC	300	18-Jan-27	01-Mar-27										are Parts	e listina	///J - // O		
Instantion Obd United and the second	E18411	Spare Parts Listing		21-Apr-27	04-Iviay-27														
E15311 20014M-5101 - Cyro Pumpa - Proper 6/FP Data Sheets & Specifications 20d 02.Mar-27 29.Mar-27 E15321 20014M-5101 - Cyro Pumpa - Receive Vendor Info 0d 21.Apr-27 0d-May-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Info 0d 21.Apr-27 0d-May-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 0d 0d-May-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 0d-May-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 17.Jun-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 17.Jun-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 21.Apr-27 E15331 20014M-5101 - Cyro Pumpa - Receive Vendor Data 0d 21.Apr-27 E15331 20014M-5102 - BOC Compressor - Prepare RFP Data Sheets & Specifications 20d 02.Mar-27 20014M-5102 - BOC Compressor - Receive Vendor Data 0d 21.Apr-27 04.May-27 E16331 20014M-5102 - BOC Compressor - Receive Vendor Data 0d 17.Jun-27 E16401 20014M-5102 - BOC Compressor - Receive Vendor Data 0d 14.Ap	20014-M-5101 (85d	02-Mar-27	30-Jun-27														
E15321 20014-M5101 - Oyno Pumps - Receive Vendor Info 0d 21-Apr-27 E15331 20014-M5101 - Oyno Pumps - Receive Vendor Data 10d 21-Apr-27 04-May-27 E15331 20014-M5101 - Oyno Pumps - Receive Vendor Data 0d 0d - M4-May-27 04-May-27 E15331 20014-M5101 - Oyno Pumps - Receive Vendor Data 0d 0d - M4-May-27 04-May-27 E15351 20014-M5101 - Oyno Pumps - Receive Vendor Data Books 0d 10f - 1/-un-27 30-Jun-27 E15371 20014-M5101 - Oyno Pumps - Receive Vendor Data Books 0d 0d - 02-May-27 30-Jun-27 20014-M5101 - Oyno Pumps - Issue Approval For Payment 0d 0d - 02-May-27 30-Jun-27 20014-M5102 - BOG Compressor - Review Vendor Data 0d 10d 21-Apr-27 10-May-27 216331 20014-M5102 - BOG Compressor - Review Vendor Data 0d 12-Apr-27 14-May-27 E16331 20014-M5102 - BOG Compressor - Review Vendor Data Books 0d 12-Apr-27 14-May-27 E16331 20014-M5102 - BOG Compressor - Review Vendor Data Books 0d 12-Apr-27 14-May-27 E16411 20014-M5102 - BOG Compressor - Review Vendor Data Books 0d 17-Jun-27	E15311	20014-M-5101 - Cyro Pumps - Prepare RFP Data Sheets & Specifications	20d	02-Mar-27	29-Mar-27									∎ 200	14-M-510	01 - Cyrc	o Pumps - Pre	pare RF	P Data Shee
E13331 20014-M-5101 - Cyro Pumps - Review Vendor Data 10d 21-Apr-27 04-May-27 E13341 20014-M-5101 - Cyro Pumps - Code 1 Vendor Data 0d 04-May-27 E13351 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 0d 17-Jun-27 E13351 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 0d 17-Jun-27 E13351 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 10d 17-Jun-27 E13351 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 10d 17-Jun-27 E13371 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 10d 17-Jun-27 20014-M-5102 - BOG Compressor - Perpere RFP Data Sheets & Specifications 20d 0d 04-Mar-27 E163371 20014-M-5102 - BOG Compressor - Review Vendor Inta 0d 21-Apr-27 E16331 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 0d 0d+Mar-27 E16331 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 17-Jun-27 29-Mar-27 E16411 20014-M-5102 - BOG Compressor - Review Vendor Data Books 0d 17-Jun-27 30-Jun-27 E16421 20014-M-5102 - BOG Compressor - Review Vendor Data Books	E15321	20014-M-5101 - Cyro Pumps - Receive Vendor Info	0d	21-Apr-27										♦ 20	014-M-51	101 - Cyr	o Pumps - R	eceive Ve	endor Info
E15341 20014-M5101 - Cyro Pumps - Code 1 Vendor Data 0d 04-May-27 E15351 20014-M5101 - Cyro Pumps - Receive Vendor Data Books 0d 17-Jun-27 E15351 20014-M5101 - Cyro Pumps - Receive Vendor Data Books 0d 17-Jun-27 E15351 20014-M5101 - Cyro Pumps - Receive Vendor Data Books 0d 17-Jun-27 E15371 20014-M5101 - Cyro Pumps - Receive Vendor Data Books 0d 0d 30-Jun-27 20014-M5102 - BOC Compressor - Prepare RFP Data Sheets & Specifications 0d 21-Apr-27 29-Mar-27 E16331 20014-M5102 - BOC Compressor - Receive Vendor Data 0d 21-Apr-27 29-Mar-27 E16391 20014-M5102 - BOC Compressor - Receive Vendor Data 0d 21-Apr-27 29-Mar-27 E16311 20014-M5102 - BOC Compressor - Receive Vendor Data 0d 21-Apr-27 04-May-27 E16411 20014-M5102 - BOC Compressor - Receive Vendor Data Books 0d 17-Jun-27 30-Jun-27 E16421 20014-M5102 - BOC Compressor - Receive Vendor Data Books 0d 17-Jun-27 30-Jun-27 E16421 20014-M5102 - BOC Compressor - Receive Vendor Data Books 0d 17-Jun-27 30-Jun-27 E1	E15331	20014-M-5101 - Cyro Pumps - Review Vendor Data	10d	21-Apr-27	04-May-27									1 20	014-M-5	101 - Cyi	ro Pumps - R	eview Ve	endor Data
E15361 20014-M-5101 - Oyro Pumps - Receive Vendor Data Books 0d 17Jun-27 E15361 20014-M-5101 - Oyro Pumps - Receive Vendor Data Books 10d 17Jun-27 30Jun-27 E15371 20014-M-5101 - Oyro Pumps - Receive Vendor Data Books 10d 17Jun-27 30Jun-27 E15371 20014-M-5101 - Oyro Pumps - Receive Vendor Data Sheets & Specifications 20d 2.0014-M-5102 - BOG Compressor - Prepare RFP Data Sheets & Specifications 20d 2.0014-M-5102 - BOG Compressor - Prepare RFP Data Sheets & Specifications 20d 2.1-Apr-27 2.0014-M-5102 - BOG Compressor - Receive Vendor Data 10d 21-Apr-27 2.0014-M-5102 - BOG Compressor - Receive Vendor Info 12.2014-M-5102 - BOG Compressor - Receive Vendor Data 10d 21-Apr-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data 10d 21-Apr-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data 10d 17Jun-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data 10d 17Jun-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data 10d 17Jun-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data Books 10d 17Jun-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data Books 10d 17Jun-27 12.2014-M-5102 - BOG Compressor - Receive Vendor Data Books 10d 17Jun-27 12.2014-M-5102 - BOG	F15341	20014-M-5101 - Cyro Pumps - Code 1 Vendor Data	0d		04-May-27									♦ 20	014-M-5 ⁻	101 - Cy	ro Pumps - C	ode 1 Ve	endor Data
E15361 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 E15371 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 20014-M-5101 - Cyro Pumps - Review Vendor Data Books 20014-M-5102 - BOG Compressor - Prepare RFP Data Sheets & Specifications 20d 02-Mar-27 29-Mar-27 E16371 20014-M-5102 - BOG Compressor - Review Vendor Info 0d 21-Apr-27 20-Mar-27 29-Mar-27 E16381 20014-M-5102 - BOG Compressor - Review Vendor Info 0d 21-Apr-27 04-May-27 E16401 20014-M-5102 - BOG Compressor - Review Vendor Data 10d 21-Apr-27 04-May-27 E16411 20014-M-5102 - BOG Compressor - Review Vendor Data Books 0d 17-Jun-27 0-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 0-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 0-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 30-Jun-27 30	E15351	20014-M-5101 - Cyro Pumps - Receive Vendor Data Books	D0	17-Jun-27	••••y=:									♦ 2	0014-M-	5101 - C	yro Pumps -	Receive	Vendor Data
E1001 1001 Hold 10 Cyro Pumps - Issue Approval For Payment 0d 30.Jun.27 20014M-S102 #006 Compressor - Prepare RFP Data Sheets & Specifications 20d 0.24Mar.27 29.Mar.27 E16371 20014M-S102 - BOG Compressor - Prepare RFP Data Sheets & Specifications 20d 0.24Mar.27 29.Mar.27 E16371 20014M-S102 - BOG Compressor - Receive Vendor Info 0d 21.Apr.27 0.4May.27 E16391 20014M-S102 - BOG Compressor - Receive Vendor Data 10d 21.Apr.27 0.4May.27 E16411 20014-M-S102 - BOG Compressor - Receive Vendor Data 0d 0.4-May.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 17.Jun.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 17.Jun.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 17.Jun.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 30.Jun.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 30.Jun.27 E16421 20014-M-S102 - BOG Compressor - Review Vendor Data Books 0d 30.Jun.27 E16421 20014-M-S102 - BOG Compressor	E15361	20014-M-5101 - Cyro Pumps - Review Vendor Data Books	10d	17-lun-27	30- lun-27									1	20014-M-	-5101 - 0	- Cyro Pumps -	Review	Vendor Data
Color Humbolings Color Humbolings <td< td=""><td>E15371</td><td>20014-M-5101 - Cyro Pumps - Issue Approval For Payment</td><td>D01</td><td></td><td>30- Jun-27</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>20014-M-</td><td>-5101 - 0</td><td>Vro Pumps -</td><td>Issue Ac</td><td>pproval For Pa</td></td<>	E15371	20014-M-5101 - Cyro Pumps - Issue Approval For Payment	D01		30- Jun-27										20014-M-	-5101 - 0	Vro Pumps -	Issue Ac	pproval For Pa
Obside 12:0014-M-5102 - BOG Compressor - Prepare RFP Data Sheets & Specifications E16371 20014-M-5102 - BOG Compressor - Receive Vendor Info 0d 21-Apr-27 29-Mia-27 E16381 20014-M-5102 - BOG Compressor - Receive Vendor Info 0d 21-Apr-27 04-Mia-27 E16391 20014-M-5102 - BOG Compressor - Receive Vendor Info 0d 21-Apr-27 04-Mia-27 E16391 20014-M-5102 - BOG Compressor - Code 1 Vendor Data 0d 04-May-27 04-May-27 E16411 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 04-May-27 04-May-27 E16421 20014-M-5102 - BOG Compressor - Review Vendor Data Books 0d 17-Jun-27 30-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 E16431 20014-M-5102 - BOG Compressor - Issue Approval For Payment 0d 30-Jun-27 30-Jun-27 Stress 10d 30-Aug-27 13-Sep-27 13-Sep-27 13-Sep-27 E16421 Stress Analysis Reports - IFC 30d 90-Mar-27 21-Apr-27<	20014-M-5102: F	20014-IN-0101 - Cylo 1 umps - issue Approvant of Layment	85d	02 Mar 27	30-Jun 27														
E16381 20014-M-5102 - BOG Compressor - Receive Vendor Info 0d 21-Apr-27 - - - - 20014-M-5102 - BOG Compressor - Receive Vendor Data 10d 21-Apr-27 04-May-27 - - 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 04-May-27 04-May-27 04-May-27 - 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 04-May-27 04-May-27 - 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 04-May-27 04-May-27 - 20014-M-5102 - BOG Compressor - Review Vendor Data 20014-M-5102 - BOG Compressor - Review Vendor Data 0d 04-May-27 - 20014-M-5102 - BOG Compressor - Review Vendor Data 20014-M-5102 - BOG Compressor - Review Vendor Data<	E16371	20014-M-5102 - BOG Compressor - Prepare REP Data Sheets & Specifications	20d	02-Mar-27	29-Mar-27									200	14-M-510	02 - BOC	G Compresso	r - Prepa	are RFP Data
E1630 20014-M-5102 BOG Compressor - Review Vendor Data 10d 214-pr-27 04-May-27 E16401 20014-M-5102 BOG Compressor - Review Vendor Data 0d 04-May-27 04-May-27 E16411 20014-M-5102 BOG Compressor - Review Vendor Data Books 0d 17-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 0d 17-Jun-27 E16421 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 17-Jun-27 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 10d 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 30-Jun-27 10d 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 30-Jun-27 10d 30-Jun-27 10d 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 30-Jun-27 10d 30-Jun-27 10d 30-Jun-27 20014-M-5102 BOG Compressor - Review Vendor Data Books 10d 30-Jun-27 13-Sep-27 13-Sep-27 13-Sep-27 13-Sep-27 13-Sep-27 13-	E16381	20014-M-5102 - BOG Compressor - Receive Vendor Info	 0d	21-Apr-27										♦ 20	014 ⁴ M-51	102 - ВО	G Compress	or - Rece	eive Vendor In
E16401 20014-M-5102 - BOG Compressor - Code 1 Vendor Data 0d 04-May-27 E16401 20014-M-5102 - BOG Compressor - Code 1 Vendor Data 0d 04-May-27 E16411 20014-M-5102 - BOG Compressor - Receive Vendor Data Books 0d 17-Jun-27 E16421 20014-M-5102 - BOG Compressor - Receive Vendor Data Books 0d 17-Jun-27 E16421 20014-M-5102 - BOG Compressor - Receive Vendor Data Books 10d 17-Jun-27 E16431 20014-M-5102 - BOG Compressor - Receive Vendor Data Books 10d 17-Jun-27 Bitter 1 30-Jun-27 30-Jun-27 30-Jun-27 Piping Engineering 10d 30-Aug-27 13-Sep-27 Stress 10d 30-Aug-27 13-Sep-27 E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 29-Jul-27 E16041 LDT Development - IFH 30d 09-Mar-27 29-Jul-27 E16041 LDT Development - IFC 30d 30-Jun-27 29-Jul-27 Specialty Items 30d 30-Jul-27 13-Sep-27 10-Development - IFC E16041 LDT Development - IFC 30d 30-Jul-27 29-Jul-27 Dat	E16391	20014-M-5102 - BOG Compressor - Review Vendor Data	10d	21-Apr-27	04-May-27									1 20	014-M-5 ⁻	102 - BO	G Compress	or - Revi	iew Vendor Da
E16411 20014-M-5102 - BOG Compressor - Receive Vendor Data Books 0d 17.Jun-27 30.Jun-27 E16421 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17.Jun-27 30.Jun-27 E16431 20014-M-5102 - BOG Compressor - Review Vendor Data Books 10d 17.Jun-27 30.Jun-27 Piping Engineering 184d 18-Jan-27 08-Oct-27 13.Sep-27 Stress 10d 30-Aug-27 13.Sep-27 E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 13.Sep-27 E18421 DD Development - IFH 30d 09-Mar-27 29-Jul-27 E15511 LDT Development - IFH 30d 09-Mar-27 29-Jul-27 E16041 LDT Development - IFC 30d 30-Jun-27 13-Sep-27 Speciatly Items 30d 30-Jun-27 29-Jul-27 LDT Development - IFH LDT Development - IFH E16041 LDT Development - IFC 30d 30-Jul-27 13-Sep-27 1-D LDT Development - IFH LDT Development - IFC Speciatly Items 30d 30-Jul-27 13-Sep-27 1-D LDT Development - IFC D LDT Develo	E16401	20014-M-5102 - BOG Compressor - Code 1 Vendor Data	0d	217.0121	04-May-27									♦ 20	014-M-5 ⁻	102 - BO	G Compress	or - Cod	e 1 Vendor Da
Lot 11 Lot 11	E16411	20014-M-5102 - BOG Compressor - Receive Vendor Data Books	D0	17lun-27	or may 21									♦ 2	0014-M-	5102 - B	OG Compres	ssor - Re	ceive Vendor
E 1042 I 20014-Mr5102 - BOG Compressor - Issue Approval For Payment 0d 0d 30-Jun-27 F16431 20014-Mr5102 - BOG Compressor - Issue Approval For Payment 0d 0d 30-Jun-27 Piping Engineering 10d 30-Aug-27 13-Sep-27 13-Sep-27 E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 13-Sep-27 E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 29-Jul-27 E15511 LDT Development - IFH 30d 09-Mar-27 29-Jul-27 E16041 LDT Development - IFC 30d 17-Jun-27 29-Jul-27 Specialty Items 30d 30-Jul-27 13-Sep-27 Actual Level of Effort Actual Work Critical Remaining Work Critical Remaining Work Critical Remaining Work Actual Level of Effort Remaining Work Critical Secondary Date Revision Checked Approved	E16421	20014-M-5102 - BOG Compressor - Review Vendor Data Books	10d	17-Jun-27	30- lun-27	-								1	20014-M-	-5102 - E	3OG Compre	ssor - Re	eview Vendor
Date Remaining Level of Effort Out	E16431	20014-M-5102 - BOG Compressor - Issue Approval For Payment	D01	11-0011-27	30-Jun-27										20014-M-	-5102 - E	3OG Compre	ssor - Iss	sue Approval I
Stress 10d 30-Aug-27 13-Sep-27 E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 13-Sep-27 Line Designation Table (LDT) 100d 09-Mar-27 29-Jul-27 E15511 LDT Development - IFH 30d 09-Mar-27 29-Jul-27 E16041 LDT Development - IFC 30d 17-Jun-27 29-Jul-27 Specialty Items 30d 30-Jul-27 13-Sep-27 Market	Pining Engineerin	a	1944	19 Jan 27	09 Oct 27										-				
E18421 Stress Analysis Reports - IFC 10d 30-Aug-27 13-Sep-27 Line Designation Table (LDT) 100d 09-Mar-27 29-Jul-27 E15511 LDT Development - IFH 30d 09-Mar-27 21-Apr-27 E16041 LDT Development - IFC 30d 17-Jun-27 29-Jul-27 Specialty Items 30d 30-Jul-27 13-Sep-27 Page: 46 of 73 Date Revision Chicked Approved	Stress	9	104u	30-Aug-27	13-Sep-27														
Line Designation Table (LDT) 100d 09-Mar-27 29-Jul-27 E15511 LDT Development - IFH 30d 09-Mar-27 21-Apr-27 E16041 LDT Development - IFC 30d 17-Jun-27 29-Jul-27 Specialty Items 30d 30-Jul-27 13-Sep-27 Remaining Level of Effort Actual Work Critical Remaining Work Data Date:26-Jul-24 Actual Level of Effort Ensing Work Critical Secondary Checked Approved	E18421	Stress Analysis Reports - IFC	10d	30-Aug-27	13-Sep-27			· · · · · · · · · · · · · · · · · · ·							Stress	Analysis	Reports - IFC		
E15511 LDT Development - IFH Date LDT Development - IFH LDT Development - IFH E16041 LDT Development - IFC 30d 17-Jun-27 29-Jul-27 13-Sep-27 Specialty Items 30d 30-Jul-27 13-Sep-27 13-Sep-27 14-DT Development - IFC Date Revision Checked Approved Remaining Level of Effort Actual Work Critical Remaining Work Data Date:26-Jul-24 Date Revision Checked Approved Actual Level of Effort Remaining Work Critical Secondary Page: 46 of 73 Page: 46 of 73 IFI IFI IFI	Line Designation	Table (LDT)	100d	09-Mar-27	29-Jul-27									V					
E16041 LDT Development - IFC Specialty Items 30d 30d </td <td>E15511</td> <td>LDT Development - IFH</td> <td>30d</td> <td>09-Mar-27</td> <td>21-Apr-27</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>🗖 LD</td> <td>T Develo</td> <td>pment - I</td> <td>IFH</td> <td></td> <td></td>	E15511	LDT Development - IFH	30d	09-Mar-27	21-Apr-27									🗖 LD	T Develo	pment - I	IFH		
Specialty Items 30d 30-Jul-27 13-Sep-27 Image: Actual Work	E16041	LDT Development - IFC	30d	17-Jun-27	29-Jul-27										LDT Dev	/elopmer	nt - IFC		
 Remaining Level of Effort Actual Work Critical Remaining Work Critical Remaining Work Critical Secondary Date Date Remaining Work Critical Secondary Critical Secondary 	Specialty Items		30d	30-Jul-27	13-Sep-27														
Image: Actual Work Image: Ac																			
Actual Level of Effort Critical Secondary Page: 46 of 73	🖛 Remaining Lev	el of Effort - Actual Work - Critical Remaining Work			Data Date:2	6-Jul-2	24				Date			Revision			Checked		Approved
	Actual Level of	Effort Effort Remaining Work Critical Secondary			Page: 46	of 73	-			19-Au	g-24								

SOLARIS MANAGEMENT

2	FORTIS BC"

Activity	' ID	Activity Name	Rem. Duration	Start	Finish	020	2021	2022	2023 2	2024	2025	2026	20		2028	2029	2030	2031
	F18431	SP Items List - IEC	30d	30-Jul-27	13-Sep-27									SP Iter	ns List - IFC			
	Specifications		30d	18-Jan-27	01-Mar-27									_				
	E19881	Piping Specifications - IFC	30d	18-Jan-27	01-Mar-27								🔲 Pi	ping Specifi	cations - IF	C		
	Requisition		74d	24-Jun-27	08-Oct-27													
	20014-L-5201 - S	Stainless Steel Pipe	60d	24-Jun-27	20-Sep-27				· · · · · · · · · · · · · · · · · · ·		· · · ·					 		
	E17031	20014-L-5201 - Stainless Steel Pipe - Prepare RFP Data Sheets & Specifications	20d	24-Jun-27	22-Jul-27									🛛 20014-L	-5201 - Sta	inless Steel	Pipe - Pre	pare RFP1
	E17041	20014-L-5201 - Stainless Steel Pipe - Receive Vendor Info	0d	07-Sep-27										♦ 20014	-L-5201 - S	tainless Ste	el Pipe - Re	eceive Ven
	E17051	20014-L-5201 - Stainless Steel Pipe - Review Vendor Data	10d	07-Sep-27	20-Sep-27									2001 4	-L-5201 - S	Stainless Ste	el Pipe - R	eview Ven
	20014-L-5202 - 0	Carbon Steel Pipe	60d	24-Jun-27	20-Sep-27										5000 0			
	E16971	20014-L-5202 - Carbon Steel Pipe - Prepare RFP Data Sheets & Specifications	20d	24-Jun-27	22-Jul-27									20014-L	-5202 - Ca	rbon Steel F	ripe - Prep	are RFP D
	E16981	20014-L-5202 - Carbon Steel Pipe - Receive Vendor Info	0d	07-Sep-27										♦ 20014	L-5202 - C	arbon Stee	Pipe - Reo	ceive Vend
	E16991	20014-L-5202 - Carbon Steel Pipe - Review Vendor Data	10d	07-Sep-27	20-Sep-27									20014	-L-5202 - C	Carbon Stee	l Pipe - Re	view Vend
	20014-L-5203 - S	Stainless Steel Fittings	60d	24-Jun-27	20-Sep-27										F000 04-		Filling F	
	E17171	20014-L-5203 - Stainless Steel Fittings - Prepare RFP Data Sheets & Specification	1 20d	24-Jun-27	22-Jul-27									20014-L	-5203 - Sta	iniess Steel	Fittings - F	repare RF
	E17181	20014-L-5203 - Stainless Steel Fittings - Receive Vendor Info	0d	07-Sep-27										♦ 20014	-L-5203 - S	tainless Ste	el Fittings -	Receive v
	E17191	20014-L-5203 - Stainless Steel Fittings - Review Vendor Data	10d	07-Sep-27	20-Sep-27									2 0014	-L-5203 - S	stainless Ste	el Fittings ·	- Review Ve
	20014-L-5204 - (Carbon Steel Fittings	60d	24-Jun-27	20-Sep-27									20014	5204 Co	rhan Staal F	ittingo Dr	
	E20131	20014-L-5204 - Carbon Steel Fittings - Prepare RFP Data Sheets & Specifications	s 20d	24-Jun-27	22-Jul-27									20014-L	-5204 - Ca		illings - Pr	
	E20141	20014-L-5204 - Carbon Steel Fittings - Receive Vendor Info	0d	07-Sep-27										• 20014	-L-5204 - C	arbon Stee	r⊢nungs-r	Receive ve
	E20151	20014-L-5204 - Carbon Steel Fittings - Review Vendor Data	10d	07-Sep-27	20-Sep-27						· · · ·			0 20014	-L-5204 - C	arbon Stee	I Fittings - I	Review Vei
	20014-L-5205 - S	Stainless Steel Valves	74d	24-Jun-27	08-Oct-27									20014	-5205 - Sta	inless Stepl	Valves - P	Prenare RE
	E15431	20014-L-5205 - Stainless Steel Valves - Prepare RFP Data Sheets & Specification	: 20d	24-Jun-27	22-Jul-27									20014-L		Il licss Dieel		
	E15441	20014-L-5205 - Stainless Steel Valves - Receive Vendor Info	0d	27-Sep-27										◆ 2001 ²	-L-5205 - 3			
	E15451	20014-L-5205 - Stainless Steel Valves - Review Vendor Data	10d	27-Sep-27	08-Oct-27									2001	1-L-5∠05 -	Stainless St	eel valves	- Review v
	20014-L-5206 - 0	Carbon Steel Valves	60d	24-Jun-27	20-Sep-27						- L J		÷	20014-1	-5206 - Ca	rhon Steel \	/alves - Pro	enare REP
	E20281	20014-L-5206 - Carbon Steel Valves - Prepare RFP Data Sheets & Specifications	200	24-Jun-27	22-JUF27									■ 20014-L		orbon Stoo		
	E20291	20014-L-5206 - Carbon Steel Valves - Receive Vendor Info	0d	07-Sep-27														
	E20301	20014-L-5206 - Carbon Steel Valves - Review Vendor Data	10d	07-Sep-27	20-Sep-27									20014	-L-5∡06 - (arbon Stee	i vaives - F	veriew ver
	20014-5207 - Cr	yogenic Pipe Supports	60d	24-Jun-27	20-Sep-27									20014-5	207 - Civo	aenic Pine S	Supports - I	Prenare RI
	E17091	20014-5207 - Cryogenic Pipe Supports - Prepare RFP Data Sheets & Specificatio	200	24-Jun-27	ZZ-JUFZ/										5207 Cn	ocionic Dino	Supports	Pocolvo
	E17101	20014-5207 - Cryogenic Pipe Supports - Receive Vendor Info	0d	07-Sep-27											5207 - Ciy	ogenio Pipe	Supporto	
	E1/111	20014-5207 - Cryogenic Pipe Supports - Review Vendor Data	10d	07-Sep-27	20-Sep-27									<u> </u>	-5207 - CI	ogenic ripe	Supports	- Neview v
	20014-L-5208 - 1	nsulation	60d	24-Jun-27	20-Sep-27									20014-1	-5208 - Ins	ulation - Pre	nare RFP	Data Shee
	E17231	20014-L-5200 - Insulation - Flepale RFF Data Sheets & Specifications	200	24-Juli-27	ZZ-JUFZ7	_								▲ 20014	-5208 - Ir	sulation - R	eceive Ver	ndor Info
	E17201	20014-L-5200 - Insulation - Receive Vendor Into	00 10d	07-Sep-27	20 Can 27										_ _5208 _	sulation - R	eview Ven	ndor Data
	EI/2/I	20014-L-5208 - Insulation - Review Vendor Data	100	07-Sep-27	20-Sep-27													
	Modeling		190d	18-Jan-27	19-Uct-27													
	G146	Piping Modeling to 60%	100d	18-Jan-27	15-Jun-27									Piping Mc	deling to 60)%		
	E15531	Piping Modeling to 90%	35d	02-Jul-27	20-Aug-27									Piping I	/lodeling to	90%		
	Tie-In		100d	21-Apr-27	13-Sep-27								-					
	E18441	Tie-In List - IFH	20d	21-Apr-27	18-May-27									Tie-In List	IFH			
	E18451	Piping Tie-In List - IFC	30d	30-Jul-27	13-Sep-27									Piping	Tie-In List -	IFC		
	Key Plan		30d	30-Aua-27	12-Oct-27									₩				
										··· · · !		· · ·	· · · ·		····			
-	Remaining Leve	el of Effort - Actual Work - Critical Remaining Work			Data Date:	26-,101	-24			Date			Revision			Checked	Ар	proved
-	Actual Level of	Effort Effort Remaining Work E			Page: 47	7 of 73	_ T		19-A	ug-24	IFI							
						-												

SOLARIS

M A N A G E M E N T CONSULTANTS INC.



r ID	Activity Name	Rem.	Start	Finish	020		202	1	20	022		2023		202	24	2	2025
		Duration			Q	QC		Q Q	QQ		ג Q	QQ	1Q(QQ	QQ	Q	ຉຉຉ
E18461	General Arrangement Key Plan - IFC	30d	30-Aug-27	12-Oct-27													
E18481	Model Key Plan - IFC	30d	30-Aug-27	12-Oct-27													
Piping General A	Arrangements/Isos	35d	30-Aug-27	19-Oct-27								· · ·					
E15581	Construction Isos - IFC	30d	30-Aug-27	12-Oct-27											1		
E18471	Equipment General Arrangement Drawings - IFC	10d	30-Aug-27	13-Sep-27								1 1 1 1 1 1					
E15571	Piping General Arrangement Drawings - IFC	15d	28-Sep-27	19-Oct-27													
Location Plan		105d	23-Mar-27	20-Aug-27	i i i					1						1	
E19711	Plot Plan - IFH	20d	23-Mar-27	21-Apr-27								1 1 1 1 1 1					
E19071	Plot Plan - IFC	15d	09-Jul-27	29-Jul-27					-								
E16061	Equipment Location Plan - IFC	15d	30-Jul-27	20-Aug-27													
E18401	Litility Station Location Plan - IFC	15d	30- Jul-27	20-Aug-27													
Ding Support		204	19 Jon 27	20-Aug-27	+		- +					1					
E18501	Pine Sunnort Details - IFC	30d	18- Jan-27	01-Mar-27													
Standard Drawing		204	10-Jan 27	01-Mar 27													
E18511	Pining Standard Drawings & Details - IFC	30d	18-Jan-27	01-Mar-27											2 I 1 I		
Material Take Off		204	20 Aug 27	12 Oct 27					-								
F15541	Pining (PVF Insulation Pining SP Items) MTO for Estimating	30d	30-Aug-27	12-00-27													
Eiro Protection		1204	00 / lug 27	12 Oct 27													
F18521	Fire Protection & Safety Equipment Layout - IEH	20d	21-Apr-27	12-00-27					1			· · ·					
E18531	Fire Protection & Safety Equipment Layout IFC	20d	217 (pi 27	12 Oct 27													
Electrical Engine		300 2054	10 Jan 07	12-00-27								1 1 1 1 1 1					
Lectrical Enginee	ering & Design	2050	18-Jan-27	12 Sop 27	+		- +										
E18171	Electrical Load List - IEC	30d	30-Jul-27	13-Sep-27													
Single Line Diag		20d	30 Jul 27	27 Aug 27								· · ·					
F18381	Single Line Diagram - IEC	20d	30-Jul-27	27-Aug-27													
ETAD/Arc Elash		20d	30 Jul 27	13 Son 27													
F19421	ETAP/Arc Elash - IEC	30d	30-Jul-27	13-Sep-27										•			
Cable		20d	30 Jul 27	27 Aug 27													
E18191	Cable Schedule - IEC	20d	30-Jul-27	27-Aug-27													
Area Classificati	ion	100d	21_Apr_27	13-Sep-27													
F15721	Area Classification Diagram - IEH	20d	21-Apr-27	18-May-27													
E18371	Area Classification Diagram - IFC	304	30- lul-27	13-Sen-27			- +					· · · · · · · · ·					
L 1037 I	Area Classification Diagrann- II C	JOU		13-3ep-27													
	241/ Distribution Wiring Schematics	20d	30-Aug-27	27-Sep-27													
E10291		200	20 Aug 27	27-Sep-27													
E18301		200	30-Aug-27	27-Sep-27	-										2 I 1 I		
E18311	600V/480V Schematics - IFC	20d	30-Aug-27	27-Sep-27													
E18321	4160V Schematics - IFC	20d	30-Aug-27	27-Sep-27													
Standard Drawin	ngs & Details	30d	18-Jan-27	01-Mar-27													
E18351	Electrical Standard Drawings & Details - IFC	30d	18-Jan-27	01-Mar-27													
Specifications		30d	18-Jan-27	01-Mar-27					-			· · ·					
E18361	Electrical Specifications - IFC	30d	18-Jan-27	01-Mar-27													
Input/Output Drav	wings	30d	28-Sep-27	09-Nov-27											2 I 1 I 1 I		
E18341	Input/Output Drawings - IFC	30d	28-Sep-27	09-Nov-27													
Junction Box & F	Panel Drawings/Layout	30d	30-Aug-27	12-Oct-27				-					1				
E18261	Junction Box Layout - IFC	30d	30-Aug-27	12-Oct-27													
														<u></u>)ate	$\overline{-}$	
Remaining Lev				Data Date:2	6-Jul	-24							1	<u></u>	-24		-1
Actual Level of	Effort Effort Remaining Work Effort Critical Secondary			Page: 48	of 73	3							<u> </u>	<u> </u>		-+	

SOLARIS MANAGEMENT CONSULTANTS INC 2026 2027 2028 2029 2030 2031 Q</t Model Key Plan - IFC **T** Construction Isos - IFC Equipment General Arrangement Drawings - IFC Piping General Arrangement Drawings - IFC Plot Plan - IFH Plot Plan - IFC Equipment Location Plan - IFC Utility Station Location Plan - IFC Pipe Support Details - IFC Piping Standard Drawings & Details - IFC Piping (PVF, Insulation Piping SP Items) MTO for E Fire Protection & Safety Equipment Layout - IFH Fire Protection & Safety Equipment Layout - IFC Electrical Load List - IFC ₩ Single Line Diagram - IFC T ETAP/Arc Flash - IFC \mathbf{T} Cable Schedule - IFC Area Classification Diagram - IFH Area Classification Diagram - IFC W □ 24V Distribution Wiring Schematics - IFC □ 120V Schematics - IFC □ 600V/480V Schematics - IFC □ 4160V Schematics - IFC Electrical Standard Drawings & Details - IFC Electrical Specifications - IFC -Input/Output Drawings - IFC * Junction Box Layout - IFC Revision Checked Approved



ivity ID		Activity Name	Rem.	Start	Finish	020		202	1		2022		202	23	2	024		20	25	-
			Duration			QQ	2 Q	Q	Q	Q	QQ	QC	QQ	QQ	QC	2Q	Q	QQ	Q	Q
	E18271	Contactor Panel Layout - IFC	30d	30-Aug-27	12-Oct-27												-			
	E18281	Control Panel Layout - IFC	30d	30-Aug-27	12-Oct-27			1						1				1	1 1 1 1 1 1	
	E18331	Distribution Panel Drawings - IFC	30d	30-Aug-27	12-Oct-27			1						1				1	1 1 1 1 1 1	
	Tie-In		30d	30-Jul-27	13-Sep-27			1						:				1	1 1 1 1 1 1	
	E18241	Tie-In List - IFC	30d	30-Jul-27	13-Sep-27			1						1				: : :	1 1 1 1 1 1	
	Grounding/Cathod	lic Protection	30d	18-Jan-27	01-Mar-27								!					·	· · · · · · · · ·	
	E18201	Grounding/Cathodic Protection Details - IFC	30d	18-Jan-27	01-Mar-27									1				1		
	Heat Trace	Heat Tracing Leveut JEC	15d	20-Oct-27	09-Nov-27													1		
	E 10201		150	20-00-27	09-100-27									1				1		i
	F20121	Electrical Modeling to 60%	104d	18-Jan-27	20-Aug-27 15-Jun-27									1				1	, , , , , ,	
	E15731	Electrical Modeling to 90%	35d	02-10-27	20-Aug-27											· - ¦		·	 	
	Material Take Off		30d	30-Aug-27	12-Oct-27				-					1	1			1		
	E15741	Electrical (Cable/Trav/Conduit/Junction Box) MTO for Estimating	30d	30-Aug-27	12-Oct-27				-					1		1		1		
	Electrical Layouts		50d	30-Aug-27	09-Nov-27			1						-				1		į
	E18221	Electrical Equipment Layout - IFC	30d	30-Aug-27	12-Oct-27			1	-					:				1		
	E18231	New Substation Layout - IFC	30d	30-Aug-27	12-Oct-27		-													
	E18181	Cable Tray Layout - IFC	15d	20-Oct-27	09-Nov-27									:						
	E18211	Cable, Raceway, Lighting & Grounding Layout - IFC	15d	20-Oct-27	09-Nov-27			1						1				1	1 1 1 1 1 1	
	Requisition		80d	24-Jun-27	19-Oct-27									1				1	1 1 1 1 1 1	ļ
	_20014-E-5500: El	ectrical Building	80d	24-Jun-27	19-Oct-27														, , , , , , , , , , , , , , , , , , ,	
	E20381	20014-E-5500 - Electrical Building - Prepare RFP Data Sheets & Specifications	20d	24-Jun-27	22-Jul-27				-					-				1		
	E20391	20014-E-5500 - Electrical Building - Receive Vendor Info	0d	09-Aug-27														1		i
	E20401	20014-E-5500 - Electrical Building - Review Vendor Data	10d	09-Aug-27	20-Aug-27													1		
	E20411	20014-E-5500 - Electrical Building - Code 1 Vendor Data	0d		20-Aug-27			1						1				1	, , , , , ,	
	E20421	20014-E-5500 - Electrical Building - Receive Vendor Data Books	0d	05-Oct-27															1 1 1 1 1 1	
	E20431	20014-E-5500 - Electrical Building - Review Vendor Data Books	10d	05-Oct-27	19-Oct-27				-					1		1		1		
	E20441	20014-E-5500 - Electrical Building - Issue Approval for Payment	0d		19-Oct-27			1										1 1 1	· · ·	
	20014-E-5500: 5k	V Variable Speed Drive	80d	24-Jun-27	19-Oct-27			1						1				1 1 1	1 1 1 1 1 1	
	E16491	20014-E-5500 - 5kV Variable Speed Drive - Prepare RFP Data Sheets & Specifica	20d	24-Jun-27	22-Jul-27													1		
	E16501	20014-E-5500 - 5kV Variable Speed Drive - Receive Vendor Info	0d	09-Aug-27		ļ		!										·		
	E16511	20014-E-5500 - 5kV Variable Speed Drive - Review Vendor Data	10d	09-Aug-27	20-Aug-27									1						
	E16521	20014-E-5500 - 5kV Variable Speed Drive - Code 1 Vendor Data	0d		20-Aug-27			1										1		
	E16531	20014-E-5500 - 5kV Variable Speed Drive - Receive Vendor Data Books	0d	05-Oct-27				1	-					-				: : :	1 1 1 1 1 1	
	E16541	20014-E-5500 - 5kV Variable Speed Drive - Review Vendor Data Books	10d	05-Oct-27	19-Oct-27			1	-					: : :	1		1	1 1 1	1 1 1 1 1 1	
	E16551	20014-E-5500 - 5kV Variable Speed Drive - Issue Approval for Payment	0d		19-Oct-27			, , , , , , , , , , , , , , , , , , ,								,		, , ,		
	20014-E-5501: Ur	ninterruptable Power Supply (UPS)	80d	24-Jun-27	19-Oct-27									1				1		1
	E16731	20014-E-5501 - Uninterruptable Power Supply - Prepare RFP Data Sheets & Spe	20d	24-Jun-27	22-Jul-27			1						1				1	1 1 1 1 1 1	1
	E16741	20014-E-5501 - Uninterruptable Power Supply - Receive Vendor Info	0d	09-Aug-27				1						1 1 1				: : :	1 1 1 1 1 1	ļ
	E16751	20014-E-5501 - Uninterruptable Power Supply - Review Vendor Data	10d	09-Aug-27	20-Aug-27			1	-					1				1		
	E16761	20014-E-5501 - Uninterruptable Power Supply - Code 1 Vendor Data	0d		20-Aug-27															
	E16771	20014-E-5501 - Uninterruptable Power Supply - Receive Vendor Data Books	0d	05-Oct-27				1	-					1				1		
	E16781	20014-E-5501 - Uninterruptable Power Supply - Review Vendor Data Books	10d	05-Oct-27	19-Oct-27									1	1		1	1		

Remaining Level of Effort	Actual Work	Critical Remaining Work	Deta Deta 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 49 01 75		

SOLARIS MANAGEMENT CONSULTANTS INC 2026 2027 2028 2029 2030 2031 Q Control Panel Layout - IFC Distribution Panel Drawings - IFC 77 Tie-In List - IFC 77 Grounding/Cathodic Protection Details - IFC Heat Tracing Layout - IFC Electrical Modeling to 60% Electrical Modeling to 90% -Electrical (Cable/Tray/Conduit/Junction Box) MTO Electrical Equipment Layout - IFC New Substation Layout - IFC Cable Tray Layout - IFC Cable, Raceway, Lighting & Grounding Layout 20014-E-5500 - Electrical Building - Prepare RFP Da ◆ 20014-E-5500 - Electrical Building - Receive Vendor 20014-E-5500 - Electrical Building - Review Vendor ◆ 20014-E-5500 - Electrical Building - Code 1 Vendor ◆ 20014-E-5500 - Electrical Building - Receive Vend 20014-E-5500 - Electrical Building - Review Vend ◆ 20014-E-5500 - Electrical Building - Issue Approv 20014-E-5500 - 5kV Variable Speed Drive - Prepare ◆ 20014-E-5500 - 5kV Variable Speed Drive - Receiv 20014-E-5500 - 5kV Variable Speed Drive - Review ◆ 20014-E-5500 - 5kV Variable Speed Drive - Code ◆ 20014-E-5500 - 5kV Variable Speed Drive - Rece 20014-E-5500 - 5kV Variable Speed Drive - Revi ◆ 20014-E-5500 - 5kV Variable Speed Drive - Issu ų ir vieta s 20014-E-5501 - Uninterruptable Power Supply - Pre ◆ 20014-E-5501 - Uninterruptable Power Supply - Re 20014-E-5501 - Uninterruptable Power Supply - Re ◆ 20014-E-5501 Uninterruptable Power Supply - Co ◆ 20014-E-5501 - Uninterruptable Power Supply -20014-E-5501 - Uninterruptable Power Supply -

Revision	Checked	Approved



E16791 20014-E-5502: Tr E16851 E16861 E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5501 - Uninterruptable Power Supply - Issue Approval for Payment ansformers 20014-E-5502 - Transformers - Prepare RFP Data Sheets & Specifications 20014-E-5502 - Transformers - Receive Vendor Info 20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data Books	Ouration Od 80d 20d 0d 10d	24-Jun-27 24-Jun-27	19-Oct-27 <u>19-Oct-27</u> 22- Jul-27	QQ	QQQ	QQ	QQC		Q	QQ	QQ	QQQ	QQC			QQQ 20014-E-	QQ 5501 - l	a a a a Ininterrupt	고 Q Q able Pov	QQQC /er Supply
E16791 20014-E-5502: Tr E16851 E16861 E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5501 - Uninterruptable Power Supply - Issue Approval for Payment ansformers 20014-E-5502 - Transformers - Prepare RFP Data Sheets & Specifications 20014-E-5502 - Transformers - Receive Vendor Info 20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data	0d 80d 20d 0d 10d	24-Jun-27 24-Jun-27	19-Oct-27 <u>19-Oct-27</u> 22- Jul-27								1 1 1 1 1 1 1 1					20014-E-	5501 - L	Ininterrupt	able Pov	ver Supply
20014-E-5502: Tr E16851 E16861 E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	ansformers 20014-E-5502 - Transformers - Prepare RFP Data Sheets & Specifications 20014-E-5502 - Transformers - Receive Vendor Info 20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data	80d 20d 0d 10d	24-Jun-27 24-Jun-27	19-Oct-27	1 1	1 1												i i i			1 1 1
E16851 E16861 E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Prepare RFP Data Sneets & Specifications 20014-E-5502 - Transformers - Receive Vendor Info 20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data Books	0d 10d	24-Jun-27														14-E-550)2 - Trar	sformers .	Prenar	BEP Data
E16801 E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Receive Vendor Inio 20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data Books	10d		ZZ-JUFZ/			· · · · · · · · · · ·					: : :- : :				▲ 2		02 - Tra	nsformers	- Receiv	- Vendor Ir
E16871 E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Review Vendor Data 20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data Books	100	09-Aug-27	00 4 07								1 1 1 1 1 1 1 1				▼ <u>-</u> 1 2	014-E-55	02 11a 02 Tra	neformere		
E16881 E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Code 1 Vendor Data 20014-E-5502 - Transformers - Receive Vendor Data Books	0.1	09-Aug-27	20-Aug-27													014-E-55	02 11a	neformere		
E16891 E16901 E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Receive Vendor Data Books	Ud		20-Aug-27	-											◆ ∠		02 1 II a	ranoformo		
E16901 E16911 20014-E-5503: Sv		0d	05-Oct-27									· · · · · · · · · · · · · · · · · · ·					40014-E-0	502 - 1	ansionne	S-Rece	
E16911 20014-E-5503: Sv	20014-E-5502 - Transformers - Review Vendor Data Books	10d	05-Oct-27	19-Oct-27		· +	· · · · · · · · · · ·					1 1 1 1 1 1					20014-E-	5502 -	ransforme	rs - Revi	ew vendor
20014-E-5503: Sv	20014-E-5502 - Transformers - Issue Approval for Payment	0d		19-Oct-27								1 1 1 1 1 1 1 1		1			20014-E-	5502 -	ransforme	rs - Issue	Approvan
	witchgear & LV MCC	80d	24-Jun-27	19-Oct-27								1 1 1 1 1 1 1 1						13 - Swi	chaear & I	VMCC	- Pronaro F
E16611	20014-E-5503 - Switchgear & LV MCC - Prepare RFP Data Sneets & Specification	200	24-Jun-27	22-JUI-27															tobacar 8		Perceive '
E16621	20014-E-5503 - Switchgear & LV MCC - Receive Vendor Info	0d	09-Aug-27													● ∠		03-30			
E16631	20014-E-5503 - Switchgear & LV MCC - Review Vendor Data	10d	09-Aug-27	20-Aug-27								 				U 2	014-E-55	03 - 50	licngear &		- Review v
E16641	20014-E-5503 - Switchgear & LV MCC - Code 1 Vendor Data	0d		20-Aug-27												♦ 2	014-E-55	03 + SW	ttcngear &		- Code 1 v
E16651	20014-E-5503 - Switchgear & LV MCC - Receive Vendor Data Books	0d	05-Oct-27									1 1 1 1 1 1 1 1					20014-E-5	503 - S	witchgear	& LV MC	C - Receive
E16661	20014-E-5503 - Switchgear & LV MCC- Review Vendor Data Books	10d	05-Oct-27	19-Oct-27												0	20014-E-	5503 - 3	Switchgear	& LV MC	C-Review
E16671	20014-E-5503 - Switchgear & LV MCC - Issue Approval for Payment	0d		19-Oct-27												•	20014-E-	5503 - 3	Switchgear	& LV MC	C - Issue A
20014-E-5504: El	ectrical Bulks	80d	24-Jun-27	19-Oct-27								 									
E15601	20014-E-5504 - Electrical Bulks - Prepare RFP Data Sheets & Specifications	20d	24-Jun-27	22-Jul-27												2	J14-E-550)4 - Eleo	trical Bulks	- Prepa	e RFP Dat
E15611	20014-E-5504 - Electrical Bulks - Receive Vendor Info	0d	09-Aug-27									1 1 1 1 1 1		1		♦ 20	014-E-55	04 - Ele	ctrical Bulk	s - Recei	ve Vendor I
E15621	20014-E-5504 - Electrical Bulks - Review Vendor Data	10d	09-Aug-27	20-Aug-27								1 1 1 1 1 1				l 2	014-E-55	04 - Ele	ctrical Bulk	s - Revie	w Vendor [
E15631	20014-E-5504 - Electrical Bulks - Code 1 Vendor Data	0d		20-Aug-27												♦ 2	014-E-55	04 - Ele	ctrical Bulk	s - Code	1 Vendor [
E15641	20014-E-5504 - Electrical Bulks - Receive Vendor Data Books	0d	05-Oct-27													•	20014-E-5	5504 - E	lectrical Bu	lks - Rec	eive Vendc
E15651	20014-E-5504 - Electrical Bulks - Review Vendor Data Books	10d	05-Oct-27	19-Oct-27									1 1		1 1	P	20014-E-	5504 - I	lectrical B	ılks - Re	view Vendc
E15661	20014-E-5504 - Electrical Bulks - Issue Approval for Payment	0d		19-Oct-27												•	20014-E-	5504 - E	Electrical B	ulks - Iss	le Approva
Instrumentation En	gineering & Design	205d	18-Jan-27	09-Nov-27								1 1 1 1 1 1 1 1									
Instrument Index		100d	21-Apr-27	13-Sep-27								1 1 1 1 1 1									
E15911	Instrument Index - IFH	20d	21-Apr-27	18-May-27								1 1 1 1 1				Instr	iment Inde	x-⊩H			
E17751	Instrument Index - IFC	30d	30-Jul-27	13-Sep-27													nstrument	Index -	IFC		
System Block Dia	gram	30d	30-Jul-27	13-Sep-27												-	votom Plo				
E18061	System Block Diagram - IFC	30d	30-Jul-27	13-Sep-27								1 1 1 1 1 1		1			ystem Bio	ск рад	ram-⊪⊂		
Instrument Layout	Instrument such IFO	15d	20-Oct-27	09-Nov-27								1 1 1 1 1 1 1 1					Instrumo	ntlavo	IT IFC		
E 1807 1	Instrument Layout - IFC	150	20-00-27	09-INOV-27		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·					1 1 1 1 1 4			- +						
	I&C Tie in Drawing - IEC	30d	30-Jul-27	13-Sep-27													SC Tie-in E	Drawing	- IFC		
L 1000 I		1004	21 Apr 27	12 Sop 27																	
F18091	Metering Schematics - IFH	20d	21-Apr-27	18-May-27												Mete	ring Scher	natics -	IFH		
E18101	Metering Schematics - IFC	200 30d	30- Jul-27	13-Sen-27								1 1 1 1 1 1		1			/leterina S	chemati	cs - IFC		
Fire & Gas Detect		100d	21 Apr 27	13-Sep-27								: : 									
F18111	Fire & Gas Detection Drawings - IEH	20d	21-Apr-27	18-May-27												🛛 Fire	Gas Det	ection D	rawings -	FH	
 F18121	Fire & Gas Detection Drawings - IEC	 30d	30-Jul-27	13-Sep-27													ire & Gas	Detecti	on Drawing	ış - IFC	
Architectural Diac	Iram	100d	21_Apr_27	13-Sen-27																-	
E18131	PCS/PLC/SIS/SCADA Architecture Diagram - IFH	20d	21-Apr-27	18-Mav-27												PCS	PLC/SIS/S	SCADA	Architectur	e Diagra	m-IFH
E18141	PCS/PLC/SIS/SCADA Architecture Diagram - IFC	30d	30-Jul-27	13-Sep-27								1 1 1 1					CS/PLC/S	SIS/SCA	DAArchite	cture Di	ıgram - IFC
		1	1	· ·	<u>l: :</u>	1 1	: : :		1 1 1		<u></u> г)ate	<u> </u>		<u> </u>	vision		<u>: :</u> 	Checked	<u> </u>	
 Remaining Leve 	el of Effort Actual Work Critical Remaining Work			Data Date:2	6-Jul-2	4					ں میں <u>م</u> _0	1-24	IFI		<u>л</u>				Checked	<u> </u>	wpioved

SOLARIS



tivity ID	Activity Name	Rem.	Start	Finish	2020 2021 2022 2023 2024 2025	2026 2027 2028 2029 2030 2031
		Duration				
Standard Draw	wings & Details	30d	18-Jan-27	01-Mar-27		
E18151	Standard Drawings & Details - IFC	30d	18-Jan-27	01-Mar-27		
Specifications		30d	18-Jan-27	01-Mar-27		
E18161	I&C Specifications - IFC	30d	18-Jan-27	01-Mar-27		
Alarm/Set Poir	nt Table	100d	21-Apr-27	13-Sep-27		
E15931	Alarm Set Point Table - IFH	20d	21-Apr-27	18-May-27		
E16091	Alarm Set Point Table - IFC	30d	30-Jul-27	13-Sep-27		Aarm Set Point lable - IHC
Shutdown Key	y	90d	23-Mar-27	29-Jul-27		
E15891	Shut Down Key - IFH	20d	23-Mar-27	21-Apr-27		Shut Down Key - IFH
E16071	Shut Down Key - IFC	30d	17-Jun-27	29-Jul-27		📋 Shut Down Key - IFC
Material Take	Off	30d	30-Aug-27	12-Oct-27		
E15951	Instrumentation & Controls MTO	30d	30-Aug-27	12-Oct-27		Instrumentation & Controls MTO
Programming		10d	14-Sep-27	27-Sep-27		
E15881	Programming Package	10d	14-Sep-27	27-Sep-27		Programming Package
Requisition		140d	01-Feb-27	20-Aug-27		
20014-J-5600	0 - ESDV	45d	01-Feb-27	06-Apr-27		
E15761	20014-J-5600 - ESDV - Prepare Instrument Data Sheet - RFP	20d	01-Feb-27	01-Mar-27		
E19681	20014-J-5600 - ESDV - Receive Vendor Info	0d	23-Mar-27			♦ 20014-J-5600 - ESDV - Receive Vendor Info
E19751	20014-J-5600 - ESDV - Review Vendor Data	10d	23-Mar-27	06-Apr-27		0 20014-J-5600 - ESDV - Review Vendor Data
E19761	20014-J-5600 - ESDV - Code 1 Vendor Data	0d		06-Apr-27		♦ 20014-J-5600 - ESDV - Code 1 Vendor Data
20014-J-5601	1: Orifice Plate Meter	40d	24-Jun-27	20-Aug-27		
E17611	20014-J-5601: Orifice Plate Meter - Prepare Instrument Data Sheet - RFP	15d	24-Jun-27	15-Jul-27		20014-J-5601: Orifice Plate Meter - Prepare Instrun
E19661	20014-J-5601: Orifice Plate Meter - Receive Vendor Info	0d	09-Aug-27			♦ 20014-J-5601: Orifice Plate Meter - Receive Vendo
E19811	20014-J-5601: Orifice Plate Meter - Review Vendor Data	10d	09-Aug-27	20-Aug-27		20014-J-5601: Orifice Plate Meter - Review Vendo
F19821	20014-J-5601: Orifice Plate Meter - Code 1 Vendor Data	0d		20-Aug-27		♦ 20014-J-5601: Orifice Plate Meter - Code 1 Vendo
20014-1-5602	2 - Flow Control Valve	45d	01-Eeb-27	06-Apr-27		
E17491	20014-J-5602 - Flow Control Valve - Prepare Instrument Data Sheet - RFP	20d	01-Feb-27	01-Mar-27		20014-J-5602 - Flow Control Valve - Prepare Instrument
E19651	20014-J-5602 - Flow Control Valve - Receive Vendor Info	0d	23-Mar-27			♦ 20014-J-5602 - Flow Control Valve - Receive Vendor In
E19771	20014- I-5602 - Flow Control Valve - Review Vendor Data	10d	23-Mar-27	06-Apr-27		1 20014-J-5602 - Flow Control Valve - Review Vendor Da
E10771	20014 L5602 Flow Control Valve Code 1 Vender Data	Dd		06 Apr 27	╞╶╴┊╌╴╞╶╴┊╴╴╡╴╴╞╴╴╞╴╴┊╴╴┊╴╴╞╶╴┊╴╴╞╶╴┊╴╴╞╶╴╡╴╴┊╸╴┊╸╴╞╴╴┊╴╴┊	▲ 20014-1-5602 - Flow Control Valve - Code 1 Vendor Date
20014 15602		40d	04 Jun 07	20 Aug 27		
E17551	20014- I-5602 - PSV/s - Prepare Instrument Data Sheet - REP	400 15d	24-Jun-27	20-Aug-27		20014-J-5602 - PSV's - Prepare Instrument Data S
E17601	20014 L 5602 PSV/a Pacalya Vander Infa	04	00 Aug 27	10 00127		▲ 20014-J-5602 - PSV's - Receive Vendor Info
E19071		00	09-Aug-27	00 4.07		20014-L5602-PSV/s-Review Vendor Data
E19791	20014-J-5602 - PSV s - Review Vendor Data	100	09-Aug-27	20-Aug-27		20014 15602 PSV/a Code 1\/onder Data
E19801	20014-J-5602 - PSV's - Code 1 Vendor Data	Ûd		20-Aug-27		
20014-J-56XX	X: Fire/Gas Detedtion System	20d	24-Jun-27	22-Jul-27		R 20014 56VV Eiro/Cap Detection System Prope
E17671	20014-J-56XX - Fire/Gas Detection System - Prepare Instrument Data Sheet - RF	20d	24-Jun-27	22-Jul-27		
Procurement	(Analistic et an	359d	18-Jan-27	27-Jun-28		
	· ACCILIECTURA	1800	18-Jan-27	05-Oct-27		
F15831	20014-C-5301 - BOG Building - Refresh/Renegotiate	20d	18-Jan-27	12-Feb-27		20014-C-5301 - BOG Building - Refresh/Renegotiate
E159/1		204	18-lon 27	12-Fob 27		□ 20014-C-5301 - BOG Building - Develop MRP
		20u	10-Jd11-21	12-1-60-27		▲ 20014-C-5301 + BOC Building - Issue PO to Vendor
E15851				12-rep-2/		
E19441	20014-C-5301 - BOG Building - Manufacture	155d	16-Feb-27	27-Sep-27		

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	Revision	Checked	Approved
	Remaining Work		Data Date.20-Jul-24	19-Aug-24	IFI		
			Page. 51 01 75				

SOLARIS



Activi	ty ID	Activity Name	Rem.	Start	Finish D2	20 2021 2022	2023 2024	2025 2026	2027 2028	2029	2030 2031
	E15861	20014 C 5301 ROC Building Ready to Ship	Od		27 Sop 27				<u> </u>	Q Q Q Q Q 1 - BOG Building	Q Q Q Q Q Q Q - Ready to Shin
	E15001	20014-C-5301 - BOG Building - Ready to Ship	DU Od		27-Sep-27				◆ 20014-C-530	1 - BOG Building	- FTA to Site
	E13071	20014-0-3301 - BOG Building - ETA to Site	b0	05 Oct 27	04-00-27				◆ 2000 1 0 0000	ceive at Site - BO	G Buildina
	20014 C 5202: C		DU 204	19 Jan 27	10 Eab 07						
	F17341	20014-C-5302 - Galvanized Steel - Refresh/Renegotiate	20d	18-Jan-27	12-Feb-27				20014-C-5302 - Galv	anized Steel - Re	fresh/Renegotiate
	E17351	20014-C-5302 - Galvanized Steel - Develop MRP	20d	18- Jan-27	12-Feb-27					anized Steel - De	velop MRP
	E17361	20014-C-5302 - Calvanized Steel - Issue PO to Vendor	0d	10-041-27	12-Feb-27				▲ 20014-C-5302 - Galv	anized Steel - Issu	ue PO to Vendor
	20014-C-5303: C		40d	18 Jan 27	16 Mar 27						
	E17381	20014-C-5303 - Concrete - Refresh/Renegotiate	20d	18-Jan-27	12-Feb-27				20014-C+5303 - Con	rete - Refresh/Re	enegotiate
	E17391	20014-C-5303 - Concrete - Develop MRP	20d	18-Jan-27	12-Feb-27				20014-C+5303 - Con	rete - Develop M	RP
	F17401	20014-C-5303 - Concrete - Issue PO to Vendor	b0		12-Feb-27				♦ 20014-C+5303 + Con	rete - Issue PO to	o Vendor
	E19631	20014-C-5303 - Concrete - Manufacture	15d	16-Feb-27	08-Mar-27				20014-C-5303 - Cor	crete - Manufactu	ıre
	E13001	20014-C-5303 - Concrete - Ready to Shin	Dd.	101 05 27	08-Mar-27				▲ 20014-C-5303 - Cor	crete - Ready to	Ship
	E17411	20014-C-5303 - Concrete - FTA to Site	b0 Od		15-Mar-27				♦ 20014-C-5303 - Col	crete - FTA to Site	9
	E10091	W/PS 002 Receive at Site Concrete & Equindation Materials	Dd	16 Mar 27					♦ WBS-002 Receive a	t Site Concrete &	Foundation Materia
	E19901	WBS-002 Receive at Site Concrete & Foundation Materials	b0	10-Mar 27			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		♦ WBS-002 Receive a	t Site Structural S	eel Materials
	E 1999 1	WBS-002 Receive at Site Structural Steel Materials	Ud	10-Mar-27	40 Mar 07						
	E17441	20014-C-53XX - Fill Materials - Refresh/Repeantiate	20d	18-Jan-27	10-Mar-27				20014-C-53XX - Fill N	laterials - Refresh	/Renegotiate
	E17451	20014-C-53XX - Fill Materials - Develop MRP	20d	18- Jan-27	12-Feb-27				20014-C-53XX - Fill N	laterials - Develor	MRP
	E17461	20014-C-53XX - Fill Materials - Beve PO to Vendor	0d	10-041-27	12-Feb-27				▲ 20014-C+53XX - Fill N	/aterials - Issue P	O to Vendor
	E10641		154	16 Ech 27	09 Mar 27				■ 20014-C-53XX - Fill	Materials - Manuf	acture
		20014-C-53XX - Fill Materials - Manufacture	DGI	10-Feb-27	00-Ivial-27				▲ 20014-C-53XX - Fill	Materials - Ready	to Shin
	E1/4/1		bU		08-Iviar-27				◆ 20014-C-53XX - Eil	Materials - ETA to	Site
	E1/481	20014-C-53XX - Fill Materials - ETA to Site	Ud	40.14 07	15-Mar-27					t Sito Fill Motoriale	Sile
	E19971	WBS-002 Receive at Site Fill Materials	Ud	16-Mar-27							
	Mechanical		315d	02-Mar-27	07-Jun-28						
	E15381	20014-M-5101 - Cyro Pump - Refresh/Renegotiation	20d	02-Mar-27	29-Mar-27				20014-M-5101 - Cy	ro Pump - Refrest	n/Renegotiation
	E15391	20014-M-5101 - Cyro Pump - Develop MRP	20d	02-Mar-27	29-Mar-27				20014-M-5101 - Cy	ro Pump - Develo	p MRP
	E15401	20014-M-5101 - Cyro Pump - Issue PO to Vendor	b0	02 11151 21	29-Mar-27				♦ 20014-M-5101 - Cy	ro Pump - Issue F	O to Vendor
	E19461	20014-M-5101 - Cyro Pump - Manufacture	285d	30-Mar-27	23-May-28				2001	4-M-5101 - Cyro	Pump - Manufactur
	E18551	20014-M-5101 - Cyro Pump - Testing & Inspection	2000 2d	24-May-28	25-May-28				1 2001	4-M-5101 - Cvro	Pump - Testina & In
	E15411	20014-M-5101 - Cyro Pump - Ready to Shin	2d 0d	24 May 20	30-May-28				♦ 2001	4-M-5101 - Cvro	Pump - Ready to S
	E15471	20014-M-5101 - Cyro Pump - ETA to Site	Dd		06- lun-28				◆ 2001	4-M-5101 - Cvro	Pump - ETA to Site
	E10421		Dd	07 Jup 28	00-0011-20				◆ WBS	002 Receive at \$	Site - Crvo Pumo
	20014 M 5102: R	VDS-002 Necelve at Site - Ci yo Fullip	2154	07-Juli-20	07 Jup 29						
	E16441	20014-M-5102 - BOG Compressor - Refresh/Renegotiation	20d	02-Mar-27	29-Mar-27				20014-M-5102 - BC	G Compressor -	Refresh/Renegotia
	E16451	20014-M-5102 - BOG Compressor - Develop MRP	20d	02-Mar-27	29-Mar-27				20014-M-5102 - BC	G Compressor - I	Develop MRP
	E16461	20014-M-5102 - BOG Compressor - Issue PO to Vendor	0d		29-Mar-27				◆ 20014-M-5102 - BC	G Compressor -	lssue PO to Vendoi
	E10101	20014-M-5102 - BOG Compressor - Manufacture	285d	30-Mar-27	23-May-28				2001	4-M-5102 - BOG	Compressor - Man
	E19541	20014 M 5102 BOG Compressor Tosting & Inspection	2000 2d	24 May 28	25-May-20				2001	4-M-5102 - BOG	Compressor - Testi
	E16471	20014 M 5102 - BOG Compressor - Ready to Shin	20 0d	24-11/ay-20	20 May 29				▲ 2001	4-M-5102 - BOG	Compressor - Rea
	E 1047 1	20014 M 5102 - DOG Compressor - Ready to Ship	۵U ۲۵							14-M-5102 - BOC	Compressor - ETA
		20014-IVE-3102 - BOG Compressor - ETATO Site	Ua	07 1	00-JUN-28					1 - 10 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 -	
	E20751	VVBS-UUZ Receive at Site - BUG Compressor	Ud	07-Jun-28							
	Remaining Lev	rel of Effort Actual Work Eritical Remaining Work			Data Date:26-	Jul-24	Date		Revision	Checked	Approved
	Actual Level of	Effort Effort Remaining Work E			Page: 52 of	73	19-Aug-24	IFI			
			1		-			1		1	1

SOLARIS



Piping						QIQIQIQ	+ O + O + O	$+ (\gamma + (\gamma + i))$	$\gamma + \alpha + \alpha +$	α α α α α	$(\alpha + \alpha)$		1 1 1 1 1 1		N H / N H / N H / 			01010
. iping		249d	24_ lun_27	27- Jun-28													QQ	
20014-L-5201: S	Stainless Steel Pipe	2490 20d	24-Jun-27	22-Jul-27								•						
E17061	20014-L-5201 - Stainless Steel Pipe - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27							*	20	014-L-	5201 - S	tainless Ste	el Pipe	- Refre	sh/Rer
E17071	20014-L-5201 - Stainless Steel Pipe - Develop MRP	20d	24-Jun-27	22-Jul-27								∎ 20	014-L-	5201 - S	tainless Ste	eel Pipe	- Deve	lop MR
E17081	20014-L-5201 - Stainless Steel Pipe - Issue PO to Vendor	0d		22-Jul-27								♦ 20	014-L-	5201 - S	tainless Ste	eel Pipe	- Issue	PO to
20014-L-5202: 0	Carbon Steel Pipe	20d	24-Jun-27	22-Jul-27								₩						
E17001	20014-L-5202 - Carbon Steel Pipe - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								a 20	014-L-	5202 - C	arbon Stee	el Pipe -	Refree	h/Ren
E17011	20014-L-5202 - Carbon Steel Pipe - Develop MRP	20d	24-Jun-27	22-Jul-27								20	014-L-	5202 - C	arbon Stee	el Pipe -	Deve	p MRF
E17021	20014-L-5202 - Carbon Steel Pipe - Issue PO to Vendor	0d		22-Jul-27								♦ 20	014-L-	5202 - C	arbon Stee	el Pipe -	Issue	PO to \
20014-L-5203: S	Stainless Steel Fittings	125d	24-Jun-27	23-Dec-27								-	•					
E17201	20014-L-5203 - Stainless Steel Fittings - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20	014-L-	5203 - S	tainless Ste	el Fittin	igs - Re	efresh/l
E17211	20014-L-5203 - Stainless Steel Fittings - Develop MRP	20d	24-Jun-27	22-Jul-27								1 20	014-L-	5203 - S	tainless Ste	eel Fittin	igs - De	velop
E17221	20014-L-5203 - Stainless Steel Fittings - Issue PO to Vendor	0d		22-Jul-27								♦ 20	014-L-	5203 - S	tainless Ste	eel Fittin	igs - Iss	ue PO
E19511	20014-L-5203 - Stainless Steel Fittings - Manufacture	100d	23-Jul-27	15-Dec-27									200	14-L-520	3 - Stainle	ss Steel	Fitting	ა - Man
E17231	20014-L-5203 - Stainless Steel Fittings - Ready to Ship	0d		15-Dec-27									200	14-L-520	3 - Stainle	ss Steel	Fitting	s - Rea
E17241	20014-L-5203 - Stainless Steel Fittings - ETA to Site	0d		22-Dec-27									♦ 200	14-L-520)3 - Stainle	ss Steel	l Fittinģ	s - ETA
E20601	WBS-002 Receive at Site - Stainless Steel Fittings	0d	23-Dec-27										♦ WB	S-002 Re	eceive at S	ite - Stai	inless	Steel Fi
20014-L-5204: 0	Carbon Steel Fittings	125d	24-Jun-27	23-Dec-27									₹					
E20161	20014-L-5204 - Carbon Steel Fittings - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20	014-L-	5204 - C	arbon Stee	Fitting	s - Ref	resh/R
E20171	20014-L-5204 - Carbon Steel Fittings - Develop MRP	20d	24-Jun-27	22-Jul-27								20	014-L-	5204 - C	arbon Stee	Fitting	s - De	/elop N
E20181	20014-L-5204 - Carbon Steel Fittings - Issue PO to Vendor	0d		22-Jul-27								◆ 20	014-L-	5204 - C	arbon Stee	Fitting	s - Issi	ie PO I
E20211	20014-L-5204 - Carbon Steel Fittings - Manufacture	100d	23-Jul-27	15-Dec-27									200	14-L-520	4 - Carbor	n Steel F	Tittings	- Manı
E20191	20014-L-5204 - Carbon Steel Fittings - Ready to Ship	0d		15-Dec-27									200	14-L-520	4 - Carbor	n Steel F	ittings	- Read
E20201	20014-L-5204 - Carbon Steel Fittings - ETA to Site	0d		22-Dec-27									♦ 200	14-L-520)4 - Carbo	n Steel I	Fitting\$	- ETA
E20611	WBS-002 Receive at Site - Carbon Steel Fittings	0d	23-Dec-27										♦ WB	S-002 Re	eceive at S	ite - Car	rbon S	eel Fitt
20014-L-5205: S	Stainless Steel Valves	235d	15-Jul-27	27-Jun-28								-		▼				
E15461	20014-L-5205 - Stainless Steel Valves - Refresh/Renegotiate	20d	15-Jul-27	12-Aug-27								∎ 2	0014-L	-5205 - 8	Stainless St	eel Valv	/es - R	efresh/
E15471	20014-L-5205 - Stainless Steel Valves - Develop MRP	20d	15-Jul-27	12-Aug-27						1 1		2 🛛	0014-L	-5205 - 5	Stainless S	eel Valv	/es - D	evelop
E15481	20014-L-5205 - Stainless Steel Valves - Issue PO to Vendor	0d		12-Aug-27								♦ 2	0014-L	-5205 - 5	Stainless St	eel Valv	/es - Is	sue PC
E19491	20014-L-5205 - Stainless Steel Valves - Manufacture	200d	13-Aug-27	05-Jun-28										20014	L-5205 - S	Stainless	s Steel	Valves
E15491	20014-L-5205 - Stainless Steel Valves - Ready to Ship	0d		05-Jun-28										20014	L-5205 - S	Stainless	s Steel	Valves
E15501	20014-L-5205 - Stainless Steel Valves - ETA to Site	0d		26-Jun-28										• 20014	-L-5205 -	Stainles	s Stee	Valves
E20011	WBS-002 Receive at Site - Stainless Steel Valves	0d	27-Jun-28								 			🔶 WB\$-	002 Recei	ve at Sit	e - Sta	nless S
20014-L-5206: 0	Carbon Steel Valves	235d	24-Jun-27	07-Jun-28								-						
E20221	20014-L-5206 - Carbon Steel Valves - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								□ 20	014-L-	5206 - C	arbon Stee	el Valves	s - Refr	esh/Re
E20231	20014-L-5206 - Carbon Steel Valves - Develop MRP	20d	24-Jun-27	22-Jul-27								□ 20	014-L-	5206 - C	arbon Stee	el Valves	s - Dev	elop M
E20241	20014-L-5206 - Carbon Steel Valves - Issue PO to Vendor	0d		22-Jul-27								♦ 20	014-L-	5206 - C	arbon Stee	el Valves	s - Issu	e PO to
E20271	20014-L-5206 - Carbon Steel Valves - Manufacture	200d	23-Jul-27	15-May-28										20014-	L-5206 - C	arbon S	Steel Va	ilves - I
E20251	20014-L-5206 - Carbon Steel Valves - Ready to Ship	0d		15-May-28									•	20014-	L-5206 - C	arbon S	Steel Va	ılves - I
E20261	20014-L-5206 - Carbon Steel Valves - ETA to Site	0d		06-Jun-28										20014	L-5206 - 0	Carbon	Steel V	alves -
E20621	WBS-002 Receive at Site - Carbon Steel Valves	0d	07-Jun-28											WBS-0	02 Receiv	e at Site	e - Carl	on Ste
20014-L-5207: 0	Cryogenic Pipe Supports	100d	24-Jun-27	18-Nov-27									7					
E17121	20014-L-5207 - Cryogenic Pipe Supports - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20	014-L-	5207 - C	ryogenic F	'ipe Sup	ports	Refres

SOLARIS



Activ	ty ID	Activity Name	Rem.	Start	Finish	20 202	1	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031
			Duration				QQQ	Q Q Q						QQQ			QQ
	E17131	20014-L-5207 - Cryogenic Pipe Supports - Develop MRP	20d	24-Jun-27	22-Jul-27								20014	-L-5207 - C	ryogenic Pip	e Supports - D	evelop
	E17141	20014-L-5207 - Cryogenic Pipe Supports - Issue PO to Vendor	0d		22-Jul-27			· · · ·					♦ 20014	-L-5207 - C	Cryogenic Pip	e Supports - Is	sue PO
	E19501	20014-L-5207 - Cryogenic Pipe Supports - Manufacture	75d	23-Jul-27	09-Nov-27								— 20	014-L-5207	7 - Cryogenic	: Pipe Supports	, - Mani
	E17151	20014-L-5207 - Cryogenic Pipe Supports - Ready to Ship	0d		09-Nov-27								◆ 200	014-L-5207	7 - Cryogenic	: Pipe Supports	, - Read
	E17161	20014-L-5207 - Cryogenic Pipe Supports - ETA to Site	0d		17-Nov-27								♦ 20	014-L-520	7 - Cryogenir	: Pipe Supports	3 - ETA I
	E20591	WBS-002 Receive at Site - Cryogenic Pipe Supports	0d	18-Nov-27									♦ WI	3S-002 Re	ceive at Site	- Cryogenic Pip	e Supp
	20014-L-5208: Ins	ulation	75d	24-Jun-27	13-Oct-27												
	E17281	20014-L-5208 - Insulation - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20014	L-5208 - Ir	sulation - Re	sfresh/Renegot	iate
	E17291	20014-L-5208 - Insulation - Develop MRP	20d	24-Jun-27	22-Jul-27								20014	L-5208 - Ir	າsulation - De	velop MRP	
	E17301	20014-L-5208 - Insulation - Issue PO to Vendor	0d		22-Jul-27		1						♦ 20014	L-5208 - Ir	sulation - lss	sue PO to Vend	or
	E19521	20014-L-5208 - Insulation - Manufacture	50d	23-Jul-27	04-Oct-27			· · ·					2 00	14-L-5208	- Insulation -	Manufacture	
	E17311	20014-L-5208 - Insulation - Ready to Ship	0d		04-Oct-27								♦ 200	14-L-5208	- Insulation -	Ready to Ship	
	E17321	20014-L-5208 - Insulation - ETA to Site	0d		12-Oct-27								◆ 200	14-L-5208	- Insulation -	ETA to Site	
	E20071	WBS-002 Receive at Site Insulation	0d	13-Oct-27									♦ WB	S-002 Rec	eive at Site Ir	sulation	
	Electrical		230d	24-Jun-27	31-May-28	· · · · · · · · · · · · · · · · · · ·				· · · · · · · · · · · · · · · · · · ·							T
	20014-E-5500 - El	ectrical Building	195d	24-Jun-27	11-Apr-28								- 000114				
	E20311	20014-E-5500 - Electrical Building - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20014	E-5500 - E		ling - Refresh/F	kenego
	E20321	20014-E-5500 - Electrical Building - Develop MRP	20d	24-Jun-27	22-Jul-27								20014	E-5500 - E		ling - Develop I	VIRP
	E20331	20014-E-5500 - Electrical Building - Issue PO to Vendor	0d		22-Jul-27			 					♦ 20014	E-5500 - E		ling - Issue PO	to Vend
	E20371	20014-E-5500 - Electrical Building - Manufacture	150d	23-Jul-27	03-Mar-28									20014-E-	5500 - Electri	cal Building - M	anufad
	E20361	20014-E-5500 - Electrical Building - Testing & Inspection	2d	06-Mar-28	07-Mar-28									20014-E-	5500 - Electr	ical Building - Te	esting 8
	E20341	20014-E-5500 - Electrical Building - Ready to Ship	Od		10-Mar-28		1						•	20014-E-	5500 - Electr	ical Building - R	leady to
	E20351	20014-E-5500 - Electrical Building - ETA to Site	0d		10-Apr-28									♦ 20014-E	-5500 - Elec	trical Building - I	ETAto
	E20671	WBS-002 Receive at Site - Electrical Building	0d	11-Apr-28										WBS-00	2 Receive at	Site - Electrical	Buildin
	20014-E-5500 - 5k	V Variable Speed Drive	150d	24-Jun-27	04-Feb-28								V				
	E16561	20014-E-5500 - 5kV Variable Speed Drive - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20014	E-5500 - 5	kv Variable	Speed Drive - H	(efresh)
	E16571	20014-E-5500 - 5kV Variable Speed Drive - Develop MRP	20d	24-Jun-27	22-Jul-27								20014	E-5500 - 5	kV Variable	Speed Drive - C)evelop
	E16581	20014-E-5500 - 5kV Variable Speed Drive - Issue PO to Vendor	Od		22-Jul-27		1	· · · ·					♦ 20014	-E-5500 - 5	kV Variable	Speed Drive - Is	ssue P0
	E19531	20014-E-5500 - 5kV Variable Speed Drive - Manufacture	105d	23-Jul-27	22-Dec-27								2	0014-E-55	00 - 5kV Vari	able Speed Dri	ve - Ma
	E18581	20014-E-5500 - 5kV Variable Speed Drive - Testing & Inspection	2d	23-Dec-27	24-Dec-27		1						12	0014-E-55	00 - 5kV Vari	able Speed Dri	ve - Tes
	E16591	20014-E-5500 - 5kV Variable Speed Drive - Ready to Ship	0d		06-Jan-28								• 2	20014-E-55	500 - 5kV Var	iable Speed Dr	ive - Re
	E16601	20014-E-5500 - 5kV Variable Speed Drive - ETA to Site	0d		03-Feb-28								•	20014-E-5	500 - 5kV Va	iriable Speed D	Jrive - E
	E20631	WBS-002 Receive at Site - 5kV Variable Speed Drive	0d	04-Feb-28									•	WBS-002 I	Receive at \$	ite - 5kV Variab	le Spee
	20014-E-5501 - Ur	ninterruptable Power Supply	175d	24-Jun-27	13-Mar-28			 				ļ					· · · · · ·
	E16801	20014-E-5501 - Uninterruptable Power Supply - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27								20014	-E-5501 - L	Jninterruptab	le Power Supp	ly - Ref
	E16811	20014-E-5501 - Uninterruptable Power Supply - Develop MRP	20d	24-Jun-27	22-Jul-27								20014	-E-5501 - L	Jninterruptab	le Power Supp	ly - Dev
	E16821	20014-E-5501 - Uninterruptable Power Supply - Issue PO to Vendor	Od		22-Jul-27								◆ 20014	E-5501 - L	Jninterruptab	le Power Supp	ly - Issu
	E19541	20014-E-5501 - Uninterruptable Power Supply - Manufacture	120d	23-Jul-27	20-Jan-28		1							20014-E-5	501 - Uninter	ruptable Power	Suppl
	E18591	20014-E-5501 - Uninterruptable Power Supply - Testing & Inspection	2d	21-Jan-28	24-Jan-28									20014-E-5	501 - Uninter	ruptable Power	r Suppl
	E16831	20014-E-5501 - Uninterruptable Power Supply - Ready to Ship	0d		27-Jan-28								•	20014-E-5	501 - Uninter	ruptable Powe	r Suppl
	E16841	20014-E-5501 - Uninterruptable Power Supply - ETA to Site	0d		10-Mar-28								•	20014-E-	5501 - Unint	erruptable Pow	/er Sup
	E20651	WBS-002 Receive at Site - Uninterruptable Power Supply	0d	13-Mar-28									•	WBS-002	2 Receive at	Site - Uninterru	ptable F
	20014-E-5502 - Tra	ansformers	230d	24-Jun-27	31-May-28		1										
		al of Effort Actual Work Critical Remaining Work			Data Data 20	1.1.24				Date			Revision		Checked		/ed
	Actual Level of F					-JUI-24 f 73			1	9-Aug-24	IFI						
					r aye. 54 0	1.1.5											

SOLARIS



Activ	ity ID	Activity Name	Rem. Duration	Start	Finish 02									203		1
	E16921	20014-E-5502 - Transformers - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27							I-E-5502 -	Transformers	- Refre	sh/Renego	tiate
	E16931	20014-E-5502 - Transformers - Develop MRP	20d	24-Jun-27	22-Jul-27					+	20014	I-E-5502 -	Transformers	- Deve	lop MRP	
	E16941	20014-E-5502 - Transformers - Issue PO to Vendor	0d		22-Jul-27						♦ 20014	I-E-5502 -	Transformers	- Issue	PO to Venc	dor
	E19551	20014-E-5502 - Transformers - Manufacture	200d	23-Jul-27	15-May-28							<u> </u>	-E-5502 - Tra	ansform	iers - Manuf	iactu
	E18601	20014-E-5502 - Transformers - Testing & Inspection	2d	16-May-28	17-May-28							2001/	I-E-5502 - Tra	ansform	iers - Testinç	g & I
	E16951	20014-E-5502 - Transformers - Ready to Ship	0d		23-May-28							♦ 2001 [,]	4-E-5502 - Tr	ansform	iers - Ready	y to
	E16961	20014-E-5502 - Transformers - ETA to Site	0d		30-May-28							♦ 2001	4-E-5502 - Tr	ansforn	ners - ETA to	o Sit
	E20661	WBS-002 Receive at Site - Transformers	0d	31-May-28								♦ WBS	002 Receive	at Site	- Transform	ers
	20014-E-5503 -	Switchgear & LV MCC	165d	24-Jun-27	28-Feb-28							/				
	E16681	20014-E-5503 - Switchgear & LV MCC - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27						20014	I-E-5503 -	Switchgear &	LV MC	C - Refresh	ı/Re
	E16691	20014-E-5503 - Switchgear & LV MCC - Develop MRP	20d	24-Jun-27	22-Jul-27						□ 20014	I-E-5503 -	Switchgear &	LV MC	C - Develop	э МF
	E16701	20014-E-5503 - Switchgear & LV MCC - Issue PO to Vendor	Od		22-Jul-27						♦ 20014	I-E-5503 -	Switchgear &	LVMC	C - Issue PC	O to
	E19561	20014-E-5503 - Switchgear & LV MCC - Manufacture	130d	23-Jul-27	03-Feb-28							20014-E-	5503 - Switch	ngear &	LV MCC - N	∕lan
	E18611	20014-E-5503 - Switchgear & LV MCC - Testing & Inspection	2d	04-Feb-28	07-Feb-28							20014-E	5503 - Switcl	ngear &	LV MCC - T	lesti
	E16711	20014-E-5503 - Switchgear & LV MCC - Ready to Ship	0d		10-Feb-28						•	20014-E	5503 - Switcl	ngear &	. LV MCC - F	Rea
	E16721	20014-E-5503 - Switchgear & LV MCC - ETA to Site	0d		25-Feb-28							▶ 20014-E	-5503 - Switc	hgear &	× LV MCC - I	ETA
	E20641	WBS-002 Receive at Site - Switchgear & LV MCC	0d	28-Feb-28								• WBS-00	2 Receive at	Site - S\	vitchgear &	LV
	20014-E-5504 -	Electrical Bulks	110d	24-Jun-27	02-Dec-27						—					
	E15671	20014-E-5504 - Electrical Bulks - Refresh/Renegotiate	20d	24-Jun-27	22-Jul-27						20014	I-E-5504 -	Electrical Bul	ks - Ref	resh/Reneg	jotia
	E15681	20014-E-5504 - Electrical Bulks - Develop MRP	20d	24-Jun-27	22-Jul-27						20014	I-E-5504 -	Electrical Bul	ks - De∖	/elop MRP	
	E15691	20014-E-5504 - Electrical Bulks - Issue PO to Vendor	Od		22-Jul-27						♦ 20014	I-E-5504 -	Electrical Bul	ks - Issi	ie PO to Ver	ndo
	E19571	20014-E-5504 - Electrical Bulks - Manufacture	80d	23-Jul-27	17-Nov-27						2)014-E-55	04 - Electrica	Bulks -	Manufactur	re
	E15701	20014-E-5504 - Electrical Bulks - Ready to Ship	0d		17-Nov-27						♦ 20)014-E-55	04 - Electrica	Bulks -	Ready to S	hip
	E15711	20014-E-5504 - Electrical Bulks - ETA to Site	0d		01-Dec-27						◆ 2	0014-E-55	04 - Electrica	l Bulks	- ETA to Site	
	E20091	WBS-002 Receive at Site - Electrical Bulks	0d	02-Dec-27							● V	VBS-002 F	eceive at Site	- Elect	rical Bulks	
	Instrumentation		330d	01-Feb-27	31-May-28						V					
	20014-J-5600: E	ESDV	60d	01-Feb-27	28-Apr-27											
	E15771	20014-J-5600 - ESDV - Refresh/Renegotiate	20d	01-Feb-27	01-Mar-27						□ 20014-J+5	300 - ESD	V - Refresh/R	enegot	ate	
	E15781	20014-J-5600 - ESDV - Develop MRP	20d	01-Feb-27	01-Mar-27						□ 20014-J-5	300 - ESD	V - Develop N	1RP		
	E15791	20014-J-5600 - ESDV - Issue PO to Vendor	Od		01-Mar-27						◆ 20014-J-5	300 - ESD	V - Issue PO	o Vend	or	
	E19591	20014-J-5600 - ESDV - Manufacture	35d	02-Mar-27	20-Apr-27						📋 20014-J-	5600 - ES	DV - Manufao	ture		
	E15801	20014-J-5600 - ESDV - Ready to Ship	0d		20-Apr-27						◆ 20014-J-	5600 - ES	DV - Ready to	o Ship		1 1 1 1 1 1
	E15811	20014-J-5600 - ESDV - ETA to Site	0d		27-Apr-27						◆ 20014-J	5600 - ES	DV - ETA to S	ite		
	E20051	WBS-002 Receive at Site - ESDV	Od	28-Apr-27							◆ WBS-00	2 Receive	at Site - ESD	/		
	20014-J-5601 O	rifice Plate Meter	65d	17-Jun-27	21-Sep-27											
	E17621	20014-J-5601: Orifice Plate Meter - Refresh/Renegotiate	20d	17-Jun-27	15-Jul-27		· · · · · · · · · · · · · · · · · · ·				20014	-J-5601: C	Prifice Plate IV	leter - H	etresh/Rene	ego
	E17631	20014-J-5601: Orifice Plate Meter - Develop MRP	20d	17-Jun-27	15-Jul-27						20012	-J-5601: C	Prifice Plate IV	leter - L	evelop MRH	
	E17641	20014-J-5601: Orifice Plate Meter - Issue PO to Vendor	0d		15-Jul-27						♦ 20014	-J-5601: C	Prifice Plate M	leter - Is	sue PO to V	/end
	E19621	20014-J-5601: Orifice Plate Meter - Manufacture	40d	16-Jul-27	13-Sep-27						2 00	14-J-5601	Orifice Plate	Meter -	Manufactur	re
	E17651	20014-J-5601: Orifice Plate Meter - Ready to Ship	Od		13-Sep-27						◆ 200	14-J-5601	Orifice Plate	Meter -	Ready to S	hip
	E17661	20014-J-5601: Orifice Plate Meter - ETA to Site	0d		20-Sep-27						♠ 200	14-J-5601	: Orifice Plate	Meter	ETA to Site	
	E20701	WBS-002 Receive at Site - Orifice Plate Meter	0d	21-Sep-27							♦ WB	S-002 Rec	eive at Site -	Orifice I	Plate Meter	
	20014-J-5602: F	Flow Control Valve	165d	01-Feb-27	28-Sep-27											
	Remaining Le	evel of Effort - Actual Work - Critical Remaining Work			Data Data:26	lul-24		Date		R	evision		Checke	d	Approved	
	Actual Level o	of Effort Effort Remaining Work Critical Secondary			Page: 55 of	73	ļ	19-Aug-24	IFI				_			
		-														

SOLARIS



Activ	<i>i</i> ity ID	Activity Name	Rem.	Start	Finish	20 2021 2022 2023	2024	2025	2026	2027	2028	2029	2030	2031
	E47504	20014 5000 Else Orstell/chie Defect/Deressiste		04 Eab 07	04 Mar 07									
	E17501	20014-J-5602 - Flow Control Valve - Refresh/Renegotiate	20d	01-Feb-27	01-Mar-27					20014-	1-5602 - Flow	Control Val	/e - Neilesii /e - Develor	
	E17511	20014-J-5602 - Flow Control Valve - Develop MRP	200	01-Feb-27	01-Mar-27					▲ 20014-	1-5602 - Flow	Control Val		O to Vendor
	E17521	20014-J-5602 - Flow Control Valve - Issue PO to vendor		00 14 07	01-Mar-27					▼ 20014-		Elow Con	trol Valve - N	Manufacturir
	E19601	20014-J-5602 - Flow Control Valve - Manufacturing	140d	02-Mar-27	20-Sep-27									Zoody to Sh
	E17531	20014-J-5602 - Flow Control Valve - Ready to Ship	Ud		20-Sep-27						20014-3-3002		trol Valve - r	
	E1/541	20014-J-5602 - Flow Control Valve - E IA to Site	0d		27-Sep-27									
	E20681	WBS-002 Receive at Site - Flow Control Valve	0d	28-Sep-27						•		ceive al Sile		
	20014-J-5602: PS	20014 LEG02 DEV/a Datash/Danagatista	235d	17-Jun-27	31-May-28		·			- 20	014-1-5602-	PS\/'s - Refr	esh/Renea	otiate
	E17501	20014-J-5002 - PSV's - Reliesi/Reliegoliale	200	17-Jun-27	15-Jul-27					– 20	014-1-5602-	PSV/s - Dev	elon MRP	54440
	E17571	20014-J-5602 - PSV's - Develop MRP	200	17-Jun-27	15-JUE-27					■ 20	014-1-5602-		PO to Ver	odor
	E17581	20014-J-5602 - PSV's - Issue PO to Vendor	Ud	40.1.1.07	15-Jul-27					▼ 20	2001			Ifacturo
	E19611	20014-J-5602 - PSV's - Manufacture	200d	16-Jul-27	08-May-28						▲ 2001			
	E17591	20014-J-5602 - PSV's - Ready to Ship	0d		08-May-28						◆ 20012			y to Ship
	E17601	20014-J-5602 - PSV's - ETA to Site	Od		30-May-28						♦ 2001	4-J-30U2 - F	->vs-⊨IA	
	E20691	WBS-002 Receive at Site - PSVs	Od	31-May-28							♦ WBS	+002 Receiv	e at Site - P	SVS
	20014-J-56XX: Fi	re/Gas Detection System	110d	24-Jun-27	02-Dec-27							Fire/Cas Do	taction Svet	on Pofros
	E17681	20014-J-56XX: Fire/Gas Detection System - Refresh/Renegotiate	200	24-Jun-27	22-JUE27						014-3-3077		tection Syst	
	E17691	20014-J-56XX: Fire/Gas Detection System - Develop MRP	20d	24-Jun-27	22-Jul-27						014-J-3077	Fire/Cas De	tection Syst	
	E17701	20014-J-56XX: Fire/Gas Detection System - Issue PO to Vendor	0d		22-Jul-27					• 20	014-J-30AA	rile/Gas De	Detection	Suctors M
	E19581	20014-J-56XX: Fire/Gas Detection System - Manufacture	85d	23-Jul-27	24-Nov-27						20014-J-50	XX: Fire/Ga		System - Ivia
	E17711	20014-J-56XX: Fire/Gas Detection System - Ready to Ship	Od		24-Nov-27						20014-J-50	XX: Fire/Ga	sDetection	System - Re
	E17721	20014-J-56XX: Fire/Gas Detection System - ETA to Site	0d		01-Dec-27						20014-J-56	XX: Fire/Ga	s Detection	System - El
	E20711	WBS-002 Receive at Site - Fire/Gas Detection System	0d	02-Dec-27							WBS-002 F	Receive at S	te - ⊢ıre/Ga	s Detection :
	Fabrication		139d	03-Nov-27	29-May-28					V				
	Structural Steel	WPC 002 Structural Steel Entrination	90d	02-Dec-27	17-Apr-28						WBS-(02 - Structu	ral Steel Fa	brication
	E 1994 I		900	02-Det-27	17-Api-28									brioddori
	Carbon Steel		139d 130d	03-Nov-27	29-May-28									
	E19921	WBS-002 - CS Pipe Fabrication	130d	03-Nov-27	15-May-28						WBS	002 - CS Pi	pe Fabricati	on
	Stainless Steel	· ·	139d	03-Nov-27	29-May-28					V				
	E19931	WBS-002 - SS Pipe Fabrication	139d	03-Nov-27	29-May-28						WBS	-002 - SS P	pe Fabricat	ion
	Construction		700d	20-Oct-27	20-Aug-30									
	Civil		60d	20-Oct-27	20-Jan-28							8 40.000		
	E19951	WBS-002 Siteworks	60d	20-Oct-27	20-Jan-28						VVBS-002	Sileworks		
	Concrete & Found		70d	03-Nov-27	17-Feb-28							2 Concrete	8. Eoundatio	ne
	E19961	WBS-002 Concrete & Foundations	70d	03-Nov-27	17-Feb-28					L				113
	E20021	W/BS-002 Structural Steel	90d	04-Feb-28	13-Jun-28							-002 Struct	ural Steel	
	Ruildings		900 65d	19 Ech 29	13-Juli-20									
	E20101	WBS-002 BOG Building Frection	30d	18-Feb-28	03-Apr-28						WBS-0	02 BOG Bu	ilding Erectio	on
	E20821	WBS-002 Electrical Building Erection	30d	11-Apr-28	23-May-28						WBS	002 Electric	al Building E	Erection
	Equipment		24	07-Jun-28	08-lun-28						•			
	E20791	WBS-002 Cryo Pump Installation	2d	07-Jun-28	08-Jun-28						I WBS	002 Cryo F	ump Install	ation
	E20801	WBS-002 BOG Compressor Installation	2d	07-Jun-28	08-Jun-28						I WBS	002 BOG (Compressor	r Installation
			24	01 041120										
	Remaining Lev	el of Effort Actual Work Critical Remaining Work			Data Dato:3	6- lul-24	Date			Revision		Check	ed A	Approved
	Actual Level of	Effort Effort Remaining Work E			Page: 56	of 73	19-Aug-24	IFI						
		- ,	1				1	1					1	

SOLARIS



Activ	vity ID	Activity Name	Rem.	Start	Finish	020		202	21	2	2022	202	23	2024		202	5
			Duration			Q	QC	QQ	QQ			QQ	QQ	QQQ	QQ	Q	QQ
	Piping		110d	18-Feb-28	26-Jul-28												
	E20031	WBS-002 Piping Installation	110d	18-Feb-28	26-Jul-28				-				1			1 1 1 1 1 1	1
	Electrical		105d	20-Nov-29	25-Apr-30								-				1
	E20041	WBS-002 Electrical Installation	105d	20-Nov-29	25-Apr-30												
	Instrumentatiion &	Controls	100d	04-Dec-29	02-May-30								1				1
	E20061	WBS-002 Install Instrumentation	1000	04-Dec-29	02-May-30								-				1
		Tic into LNC Tonk Crado Back (Dining and EIC)	20d	11-Jul-30	08-Aug-30								1			· · ·	1
			200 05d	04 Apr 20	00-Aug-30								-				
	F20081	WBS-002 Install Insulation	95d	04-Apr-30	20-Aug-30												
	Commissioning & S		50d	08-Aug-30	18-Oct-30								1				1
	E15971	WBS-002 Pre-Commissioning/Tie-ins	45d	08-Aug-30	11-Oct-30								-				1
	F19911	WBS-002 Commissioning	45d	15-Aug-30	18-Oct-30								-			· ·	1
	E15081	WBS-002 Start Lin	DO+ DO	107 kg 00	18-Oct-30	-											1
	Cround Improvem	wb5-002 Start Op	2274	07 101 26	20 Oct 27				·								
		ent & Early Works - Ling Tank Area	327U	07-Jul-20	29-00-27								1			1 1 1 1 1 1	1
	Geotech Work/De		1400	07-Jul-20									-				
	GI1500030	Tank Area Ground Improvement Detailed Engineering/Geotech Inv.	140d	07-Jul-26	01-Feb-27												1
	Construction		187d	02-Feb-27	29-Oct-27												
	LNG Tank Area Civil	Work	187d	02-Feb-27	29-Oct-27			- +!-									
	GR4.1		50	02-Feb-27	08-FeD-27	-							1			· · ·	
	GR3.2	Install 1.0 m diameter stone column 30 m length	50d	09-Feb-27	21-Apr-27	-							-			· ·	
	GR3.3	Install 1.0 m wide CSM panels to 12 m depth [Within LNG Tank Foundation Footpr	75d	05-Apr-27	20-Jul-27												
	GR3.4	Install Wickdrains in between CSM panels to 12 m	20d	21-Jul-27	18-Aug-27								1				
	GR4.2	Supply and install 1.5 m of sand fill	30d	19-Aug-27	30-Sep-27											1 1 1 1 1 1	
	GR4.3	Supply and install 1 m thick gravel load bearing pad	10d	01-Oct-27	15-Oct-27								-				
	GR5	DeMobilization (FEI Requirements / Approach)	10d	18-Oct-27	29-Oct-27								-				
	LNG Storage Tank		1065d	29-Jun-26	18-Oct-30												
	EPC (Contractor)		1065d	29-Jun-26	18-Oct-30								1			· · ·	
	Detailed Engineerin	g	264d	30-Jun-26	23-Jul-27											1 1 1 1 1 1	
	MILESTONES		203d	30-Jun-26	28-Apr-27								-				
			38d	30-Jun-26	24-Aug-26												
	MS1000	Client: CONTRACTAWARD	Ud	30-Jun-26		_											
	MS1000.1	Client: Kick-off Meeting	0d	06-Jul-26									-				
	MS1010	Client: Tank Relief Valve Process Flow Rates IFD	0d		06-Jul-26											; ; ;;-	
	MS1020	Client: Seismic Time Histories IFD	0d		06-Jul-26												
	MS1030	Client: Design Data (inc. Min NOL & Overfill Protection margin) IFD	0d		06-Jul-26								1				
	MS1040	Client: Confirm LNG Tank Data Sheets	0d		06-Jul-26								1			1 1 1 1 1 1	1
	MS1050	Client: Provide Plot Plan, Gen. Arrangement Dwgs / Model File - Tank Grade Rack	0d	07-Jul-26									-				
	MS1060	Client: In-Tank Pump Data Sheets & Typical Dimensional Drawings	0d		10-Jul-26											· · ·	
	MS1070	Client: Process Flows to enable PSV sizing	0d		17-Jul-26											;;	
	MS1080	Client: Tank P&ID's IFD	0d		17-Jul-26								-			1 1 1 1 1 1	1
	MS1090	Client: Approval of CB&I Design Methodology	0d		31-Jul-26				-				1			1 I 1 I 1 I	1
	MS1100	Client: Final Rely Upon Geotechnical and Interpretive Report	0d		31-Jul-26											· · ·	
			vu		01 001 20				1				i				_;]

Remaining Level of Effort	Actual Work	Critical Remaining Work	Deta Deta 26 Jul 24	Date	L
			Data Date.20-Jul-24	19-Aug-24	IFI
			Fage. 57 01 75		[



Revision	Checked	Approved



Acti	vity ID	Activity Name	Rem.	Start	Finish	2021	2022 2023	2024	2025	2026	2027	2028	2029	2030 2031
			Duration											
	MS1110	Client: Piping Material Specs IFD	0d		31-Jul-26		· · · · · · · · · · · · · · · · · · ·			♦ Cliei	nt: Piping M	aterial Specs	s IFD	
	MS1120	Client: Approval of CB&I's Tank Orientation Drawing	0d		17-Aug-26					♦ Clie	nt: Approva	l of CB&I's Ta	ank Orientatio	n Drawing
	MS1130	Client: Approval of CB&I's Battery Limit Sketch	0d		17-Aug-26					♦ Clie	nt: Approva	lof CB&l's B	attery Limit Sk	ketch
	MS1140	Client: Valve Data Sheets IFP	0d		24-Aug-26					♦ Clie	ent: Valve D	ata Sheets II	FP	
	MS1150	Client: In-Tank Pump Data Sheets & Detailed Dimensional Drawings IFD	0d		24-Aug-26					♦ Clie	ent: In-Tank	Pump Data	Sheets & Deta	ailed Dimensional Drav
	MS1160	Client: Control Valve Data Sheets IFD	0d		24-Aug-26					♦ Clie	ent: Control	Valve Data S	Sheets IFD	
	ENGINEERING I	MILESTONES	155d	09-Sep-26	28-Apr-27								(
	MS2000	Engineering: Model Review 1 (30%)	0d	09-Sep-26						♦ En	gineering: l	vlodel Revie	<i>N</i> 1 (30%)	
	MS3000	Engineering: Issue Adv Bill / MRP for 9% Nickel Materials	0d		11-Sep-26					♦ En	gineering: I	ssue Adv Bill	/MRP for 9%	Nickel Materials
	MS3010	Engineering: Issue MTO / MRP for Galv. CS & Low Temp Pipe	0d		07-Oct-26					♦E	ngineering:	Issue MTO /	MRP for Gal	/ CS & Low Temp Pipe
	MS3020	Engineering: Issue Adv Bill / MRP For LNG CS Plates	0d		09-Oct-26					♦E	ngineering:	lssue Adv Bi	II / MRP For L	NG CS Plates
	MS2010	Engineering: Model Review 2 (60%)	0d	18-Jan-27							Engineer	ng: Model R	eview 2 (60%))
	TTK296114	Start Preparing Gate 4 Deliverables	0d		28-Apr-27						♦ Start I	Preparing Ga	ate 4 Deliverat	bles
	MS2020	Engineering: Model Review 3 (90%)	0d	28-Apr-27							🔶 Engin	eering: Mode	el Review 3 (9	0%)
	PROJECT MANA	GEMENT	80d	07-Jul-26	29-Oct-26					—				
	PM1755	Project Management Plans & Project Schedule	80d	07-Jul-26	29-Oct-26					F 💻 F	Project Man	agement Pla	ns & Project S	Schedule
	461 PIPING		229d	07-Jul-26	09-Jun-27								TLO	
	ENPI5040	Calcs: Modeling & Dwgs - Tank Grade Rack Piping	150d	07-Jul-26	16-Feb-27						Calcs: IV		Ngs - Iank Gra	ade Rack Piping
	ENPI0010	Specs: Piping & Valve Material Specifications	73d	21-Jul-26	03-Nov-26						specs: Pipin	g & vaive M	aterial Specific	cations
	ENP10020	Lists: Piping & Manual Valve List	65d	21-Jul-26	22-Oct-26						ists: Piping	& Manual Va	live List	
	ENPI0030	Model: Piping Modeling	150d	16-Sep-26	27-Apr-27				· · · · · · · · · · · · · · · · · · ·		Mode	Piping Mod	leling	
	ENP10040	Calcs: Pipe Stress Analysis	85d	23-Sep-26	29-Jan-27						Calcs: Pi	pe Stress An	alysis	
	ENP10080	MTO/Req: Galv. CS, Low Temp. & SS Pipe/Fittings/Flanges (Advanced)	7d	29-Sep-26	07-Oct-26					I N	TO/Req: G	alv. CS, Low	Temp. & SS F	Pipe/Fittings/Flanges (A
	ENP10090	MTO/Req: CS & SS Manual Valves	8d	23-Oct-26	03-Nov-26					0	/ITO/Req: C	S & SS Mar	ual Valves	
	ENPI5050	MTO/MRQ: Piping & Component Matls - Tank Grade Rack	10d	12-Jan-27	25-Jan-27) MTO/MR	Q: Piping & (Component M	latls - Tank Grade Rac
	ENPI0050	Dwgs: Piping Iso's & Area Dwgs	33d	01-Feb-27	18-Mar-27						Dwgs: I	Piping Iso's 8	Area Dwgs	
	ENPI0110	MTO/Req: Gaskets & Fasteners	10d	01-Feb-27	12-Feb-27						MTO/Re	q: Gaskets &	& Fasteners	
	ENPI0120	MTO/Req: CS & SS Cold Shoes & Misc. Pipe Supports	10d	19-Mar-27	01-Apr-27						MTO/F	Req: CS & SS	Cold Shoes	& Misc. Pipe Supports
	ENPI0060	Dwgs: Shop Generate Spool Dwgs	37d	19-Mar-27	11-May-27						📕 Dwgs	Shop Gen	erate Spool D	wgs
	ENPI0070	Subct Pkg: Engineering Docs for Module Yard Fab Pkg	27d	03-May-27	09-Jun-27						📕 Sub	dt Pkg: Engir	neering Docs f	for Module Yard Fab P
	454 CONCRETE		222d	07-Jul-26	31-May-27									
	ENCI1620	Calcs: Calcs, Modeling & Dwgs - Tank Grade Rack Fdns	55d	07-Jul-26	23-Sep-26						alcs: Calcs,	Modeling & [Dwgs - Tank G	Frade Rack Fdns
	ENCI0010	Specs: Concrete Tank Design Methodolgy	25d	04-Aug-26	08-Sep-26					Sp 📕	ecs: Concr	ete Tank Des	sign Methodol	ду
	ENCI0020	Specs: Concrete Construction & Material Specs	20d	18-Aug-26	15-Sep-26					∎ Sp	ecs: Concr	ete Construc	ction & Materia	al Specs
	ENCI0030	Calcs: Design Specification for Analysis & Preliminary Pre-stress Design	25d	18-Aug-26	22-Sep-26					∎ Ca	alcs: Design	Specification	n for Analysis a	& Preliminary Pre-stres
	ENCI0040	Calcs: Concrete Design Analyses	155d	09-Sep-26	27-Apr-27						Calcs	Concrete D	esign Analyse	\$
	ENCI0050	Subct Pkg: Early Engineering Docs for Concrete Subct	5d	23-Sep-26	29-Sep-26					I S	ubct Pkg: E	arly Enginee	ring Docs for (Concrete Subct
	ENCI1630	Subct Pkg: Engineering Docs for Subct Bids - Tank Grade Rack Fdns	10d	24-Sep-26	07-Oct-26	1				۱ S	ubct Pkg: E	ngineering C	ocs for Subct	Bids - Tank Grade Ra
	ENCI0060	Dwgs: Concrete Slab Design & Dwgs	67d	16-Oct-26	27-Jan-27						Dwgs: Co	oncrete Slab	Design & Dw	gs
	ENCI0070	Dwgs: Concrete Wall Design & Dwgs	75d	29-Oct-26	22-Feb-27						Dwgs: C	oncrete Wa	ll Design & Dw	vgs
	ENCI0080	MTO/Req: Cryogenic Reinforcing Steel / Krybar	15d	28-Jan-27	18-Feb-27						MTO/Re	q: Cryogenia	c Reinforcing S	Steel / Krybar
	ENCI0090	Dwgs: Concrete Roof Design & Dwgs	58d	09-Mar-27	31-May-27						🛑 Dwg	s: Concrete l	Roof Design 8	& Dwgs
	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·				• • • • • • • • • • •								
	Remaining Lev	vel of Effort Actual Work Critical Remaining Work			Data Date:2	6-Jul-24		Date		Re	evision		Checke	ed Approved

Actual Level of Effort

Remaining Work
 Critical Secondary

Page: 58 of 73

19-Aug-24 IFI

SOLARIS

Revision	Checked	Approved


471 ELECTRICAL ENPI5060 ENEL0010 ENEL0020 ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060	L Calcs: Calcs, Modeling & Dwgs - Tank Grade Rack E&I Dwgs: Electrical Design & Requisitioning Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Tank Only)	Duration 217d 150d	07-Jul-26	21-May-27			QQ	QQ		2 Q	QQ	QC	2 Q
471 ELECTRICAL ENPI5060 ENEL0010 ENEL0020 ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060	L Calcs: Calcs, Modeling & Dwgs - Tank Grade Rack E&l Dwgs: Electrical Design & Requisitioning Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Tank Only)	217d 150d	07-Jul-26	21-May-27									
ENPI5060 ENEL0010 ENEL0020 ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060	Calcs: Calcs, Modeling & Dwgs - Tank Grade Rack E&I Dwgs: Electrical Design & Requisitioning Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Tank Only)	150d							1				1 1 1 1
ENEL0010 ENEL0020 ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060	Dwgs: Electrical Design & Requisitioning Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Tank Only)		07-Jui-26	16-Feb-27	_								
ENEL0020 ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060	Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Tank Only)	188d	18-Aug-26	21-May-27									
ENEL0030 ENEL0040 ENPI5070 ENEL0050 ENEL0060		75d	01-Sep-26	17-Dec-26			 		· · ·				· · ·
ENEL0040 ENPI5070 ENEL0050 ENEL0060	Model: Electrical Modeling	150d	16-Sep-26	27-Apr-27									
ENPI5070 ENEL0050 ENEL0060	Specs: Electrical Specs & Data Sheets	83d	30-Sep-26	03-Feb-27									
ENEL0050 ENEL0060	MTO/MRQ: E&I Component Matts - Tank Grade Rack	10d	12-Jan-27	25-Jan-27									
ENEL0060	MTO/Req: Lighting Fixtures / Lamps / Aviation Lights	10d	21-Jan-27	03-Feb-27									
	MTO/Req: Cable Ladders / Conduit	5d	11-Feb-27	18-Feb-27									
ENEL0070	MTO/Req: Power & Instrument Cables	5d	05-Mar-27	11-Mar-27					· · · · · · · · · · · · · · · · · · ·		1 1		
ENEL0080	MTO/Req: Power & Instrument Cable Glands	5d	05-Mar-27	11-Mar-27									
ENEL0090	MTO/Req: Termination Kits, Outlets, Plugs & Switches	5d	05-Mar-27	11-Mar-27									
ENEL0100	MTO/Req: Tank Grounding & Lightning System	4d	12-Mar-27	17-Mar-27									
ENEL0110	MTO/Req: E&I Junction Boxes & Enclosures	10d	19-Mar-27	01-Apr-27									
ENEL2030	Subct Pkg: Engineering Docs for E&I Subcontract Bid	11d	19-Mar-27	05-Apr-27			 						
454 PLATE STRU	ICTURES	162d	14-Jul-26	11-Mar-27									
ENTK0010	Calcs:Steel Tank Design & Detail Engineering	135d	14-Jul-26	01-Feb-27									
ENTK0020	Specs: Steel Tank Material Specifications	15d	14-Jul-26	04-Aug-26									
ENTK0030	Dwgs: Steel Tank GA & Component Drawings	142d	28-Jul-26	25-Feb-27									
ENTK0040	Specs: PRV's/VRV's, Pump Tube Mat'l, RTD's Specs & Data Sheets	76d	26-Aug-26	14-Dec-26			 - 		· · · · · · · · · · · · · · · · · · ·				
ENTK0050	Adv Bill/Req: Plates (CS, 9%Ni, Aluminum)	32d	26-Aug-26	09-Oct-26									
ENTK0060	MTO/Req: Tank's Misc. Components	132d	26-Aug-26	11-Mar-27									
ENTK0070	MTO/Req: Weld Material	32d	26-Aug-26	09-Oct-26									
ENTK0080	Adv Bill/Req: PRV's/VRV's & Nozzles/Appurt.	55d	10-Sep-26	27-Nov-26									
ENTK0090	Adv Bill/Reg: Pump Tube & Internal Piping	92d	23-Oct-26	11-Mar-27					· · · · · · · · · · · · · · · · · · ·				
ENTK0100	MTO/Reg: Insulation (Foamglas, Deck Blanket, Resilient Blanket, Perlite)	65d	06-Nov-26	16-Feb-27									
ENTK100	Adv Bill/Reg: Isolators	10d	28-Jan-27	10-Feb-27									
481 INSTRUMENT	IT & CONTROL	190d	20-Jul-26	27-Apr-27									
ENIC0010	Specs: Instrumentation & Control Specs & Data Sheets	148d	20-Jul-26	25-Feb-27									
ENIC0020	Dwgs: Instrumentation & Control	128d	25-Aug-26	04-Mar-27	L		 		L J 1 1 1 1 1 1				- L J I I I I
ENIC0030	MTO/Reg: Control Valves	35d	25-Aug-26	14-Oct-26									
ENIC0040	Model: Instrumentation & Control Modeling	150d	16-Sep-26	27-Apr-27									
ENIC0050	MTO/Reg: Tank Gauging System	33d	11-Dec-26	03-Feb-27									
ENIC0060	MTO/Reg: Flow Elements	5d	21-Jan-27	27-Jan-27									
ENIC0070	MTO/Reg: Pressure Gauges & Indicators	5d	28-Jan-27	03-Feb-27			 						
ENIC0080	MTO/Reg: Transmitters	10d	28-Jan-27	10-Feb-27				1		1			
ENIC0090	MTO/Reg: Fire & Gas Detection System	16d	28-Jan-27	18-Feb-27									
ENIC0100	MTO/Reg: Manual Call Point	10d	28-lan-27	10-Foh-27									
ENIC0110	MTO/Reg: Process Connection Material	204	20-0ai-27	25_Eab 27									
	MTO/Pag: Instrument Stands & Supports	200 5d	20-Jan-27	11 Mar 07			 		· L J				- L J - L J
	MTO/Pag: Instrument Stanus & Supports	DU Ed	12 Mar 27	10 Mar 07						1			
		DC								1			
432 STRUCTURA	AL CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR	240d	05-Aug-26	23-Jul-27	1 1	1 1				1			1 1

Actual Level of Effort

Remaining Work Critical Secondary

Data Date:26-Jul-24 Page: 59 of 73

19-Aug-24 lifi SOLARIS MANAGEMENT

CONSULTANTS INC 2026 2027 2029 2030 2031 2028 📩 Calcs: Calcs, Modeling & Dwgs - Tank Grade Rack E&I Dwgs Electrical Design & Requisitioning Calcs: Foundation Heating Design, Specs & Dwgs (2.0bcf Model: Electrical Modeling: Specs: Electrical Specs & Data Sheets MTO/MRQ: E&I Component Matls - Tank Grade Rack MTO/Red: Lighting Fixtures / Lamps / Aviation Lights | MTO/Reg: Cable Ladders / Conduit | MTO/Reg: Power & Instrument Cables | MTO/Req: Power & Instrument Cable Glands | MTO/Req: Termination Kits, Outlets, Plugs & Switches | MTO/Req: Tank Grounding & Lightning System MTO/Req: E& Junction Boxes & Enclosures Subct Pkg: Engineering Docs for E&I Subcontract Bid Calcs:Steel Tank Design & Detail Engineering Specs: Steel Tank Material Specifications Dwgs: Steel Tank GA & Component Drawings Specs: PRV's/VRV's, Pump Tube Mat'l, RTD's Specs & Data Adv Bill/Req: Plates (CS, 9%Ni, Aluminum) MTO/Reg: Tank's Misc. Components MTO/Req: Weld Material Adv Bill/Reg: PRV's/VRV's & Nozzles/Appurt. Adv Bill/Req: Pump Tube & Internal Piping MTO/Reg: Insulation (Foamglas, Deck Blanket, Resilient B Adv Bill/Reg: Isolators Specs: Instrumentation & Control Specs & Data Sheets Dwgs: Instrumentation & Control MTO/Reg: Control Valves Model: Instrumentation & Control Modeling MTO/Req: Tank Gauging System MTO/Reg: Flow Elements MTO/Reg: Pressure Gauges & Indicators MTO/Req: Transmitters MTO/Reg: Fire & Gas Detection System MTO/Reg: Manual Call Point MTO/Reg: Process Connection Material | MTO/Req: Instrument Stands & Supports | MTO/Req: Inclinometer

Revision	Checked	Approved



Activ	ty ID	Activity Name	Rem.	Start	Finish	2021 2022	2023	2024 2025	2026	2027 2028	2029	2030 2031
			Duration									
	ENST0010	Specs: Structural Steel Mat'l, Galvanizing, Grouting	40d	05-Aug-26	30-Sep-26					s: Structural Steel Mat	1, Galvanizing, Gr	routing
	ENST0020	Specs: Structural Steel Module Execution Plan	30d	05-Aug-26	16-Sep-26				Spec	s: Structural Steel Moc	ule Execution Pla	in
	ENST2050	Calcs: Modeling & Dwgs - Tank Grade Rack Structural	80d	09-Sep-26	08-Jan-27					alcs: Modeling & Dwg	s - Tank Grade Ra	ack Structural
	ENST0030	Calcs: Structural Design & Modeling - Pump Platform	150d	17-Sep-26	28-Apr-27					Calcs: Structural De	sign & Modeling -	Pump Platform
	ENST0040	Calcs: PSV Platform, Instrument Platform, Pipe Guides, Stair Tower, Roof Walkway	115d	17-Sep-26	09-Mar-27					Calcs: PSV Platform,	Instrument Platfor	m, Pipe Guides, S
	ENST0050	Calcs: Pump Platform IFA	115d	17-Sep-26	09-Mar-27					Calcs: Pump Platform	IFA	
	ENST2060	MTO/MRQ: Pkg #3 Structural Steel - Tank Grade Rack	10d	11-Jan-27	22-Jan-27	1			o N	/ITO/MRQ: Pkg #3 Str	uctural Steel - Tan	k Grade Rack
	ENST0060	Dwgs: PSV Platform, Instrument Platform, Pipe Guides, Stair Tower, Roof Walkway	70d	19-Jan-27	28-Apr-27					Dwgs: PSV Platforn	n, Instrument Platf	orm, Pipe Guides,
	ENST0070	Dwgs: Pump Platform - Main Steel	70d	19-Jan-27	28-Apr-27	1				Dwgs: Pump Platfor	m - Main Steel	
	ENST2020	MTO/MRQ: Pkg #1 Structural Steel - Pump Modules	15d	10-Mar-27	30-Mar-27	1				MTO/MRQ: Pkg #1 \$	Structural Steel - F	Pump Modules
	ENST0080	Subct Pkg: Structural Detailing Shop Dwgs & HCBI Review - Pump Modules	60d	29-Apr-27	23-Jul-27					Subct Pkg: Struc	tural Detailing Sho	op Dwgs & HCBI F
	ENST0090	MTO/MRQ: Pkg #2 Structural Steel - Platforms, Stairways, Walkways & Handrail	16d	29-Apr-27	20-May-27					MTO/MRQ: Pkg #2	Structural Steel -	Platforms, Stairwa
	441 MECHANICA	L	20d	21-Jan-27	18-Feb-27				W			
	ENME0010	Specs: Equipment Specs & Data Sheets	10d	21-Jan-27	03-Feb-27				0 \$	Specs: Equipment Spe	cs & Data Sheets	S
	ENME0020	MTO/Req: Dry Chem System	10d	04-Feb-27	18-Feb-27					MTO/Req: Dry Chem	System	
	ENME0030	MTO/Req: Jib Crane	10d	04-Feb-27	18-Feb-27				0	MTO/Req: Jib Crane		
	Regulatory Permits	(Contractor Scope)	318d	05-Nov-26	23-Feb-28							
	BCER		203d	28-Apr-27	23-Feb-28							
	Short Term Water	r Use Permit Water Sustainability Act	203d	28-Apr-27	23-Feb-28	4				Prenare Short Te	m Water I Ise Anr	olication
	RG1616731	Prepare Short lerm water Use Application	400	28-Apr-27	23-Jun-27							Jieauon
	RG1616732		30	24-Jun-27	28-Jun-27						l Dariad *Dur TBC	*
	RG1616733	Review Period *Dur. IBC*	160d	29-Jun-27	23-Feb-28						enou Dui. IBC	Chart Tarm \M/dtar
	RGT616734	Receive Approval/Permit - Short Term Water Use	0d		23-Feb-28						-pprova/Permit-	Short lenn water
	City Of Delta MD	mit Delte Bulau No. 6022	173d	18-Jan-27	23-Sep-27							
	RGT406182	Prepare Highway Use and Inspection Permits Application	20d	28-Apr-27	26-May-27					Prepare Highway	Jse and Inspectic	on Permits Applicat
	RGT406184	Submit Application	3d	27-May-27	31-May-27	-				Submit Application		
	RGT406186	Review Period	80d	01- lun-27	23-Sen-27					Review Period		
	RGT406188	Receive Approval/Permit *Driving Ground Improvement*	000	01 001 27	23-Sen-27					Receive Approx	val/Permit *Drivir	ng Ground Improv
	Buildings Permit	- Delta Building/Plumbing Bylaw (No. 6060)	334	28-Apr-27	14- lun-27					-		
	RGT406382	Prepare Building Permit Application	10d	28-Apr-27	11-May-27					Prepare Building P	ermit Application	
	RGT406384	Submit Application	3d	12-May-27	14-May-27					Submit Application		
	RGT406386	Review Period	20d	17-May-27	14-Jun-27					Review Period		
	RGT406388	Receive Approval/Permit	0d	,	14-Jun-27					Receive Approval	Permit	
	Temporary Buildi	ngs Permit - Delta Building/Plumbing Bylaw (No. 6060)	33d	28-Apr-27	14-Jun-27							
	RGT616767	Prepare Building Permit Application	10d	28-Apr-27	11-May-27					Prepare Building P	ermit Application	
	RGT616770	Submit Application	3d	12-May-27	14-May-27					Submit Application		
	RGT616768	Review Period	20d	17-May-27	14-Jun-27				·	Review Period		
	RGT616769	Receive Approval/Permit	0d		14-Jun-27					♦ Receive Approval	Permit	
	Plumbing Permit	- Delta Building/Plumbing Bylaw (No. 6060)	43d	15-Jun-27	16-Aug-27					••		
	RGT616735	Prepare Plumbing Permit Application	20d	15-Jun-27	13-Jul-27					Prepare Plumbin	g Permit Applicatio	on
	RGT616736	SubmitApplication	3d	14-Jul-27	16-Jul-27	1				Submit Applicatio	n	
	RGT616737	Review Period	20d	19-Jul-27	16-Aug-27					Review Period		
	111 1					<u> </u>		<u>, , , , , , , , ,</u>	<u> </u>			
	Remaining Lev	el of Effort Actual Work Critical Remaining Work			Data Date:2	6-Jul-24	-	Date	Revis	ion	Checked	Approved

	0
_	Actual Level of Effort

Remaining Work Critical Secondary

Page: 60 of 73

19-Aug-24 IFI SOLARIS

Revision	Checked	Approved



		Duration				2021	2022	2023	2024	2023		20	2021	20	20 2	2023	2030	2031
		Duration			Q Q						Q Q Q	QQ						2 Q Q Q
RGT616738	Receive Approval/Permit - Plumbing Permit	Od		16-Aug-27									•	Receive	Approval/Pe	ermit - Pium	oing Perr	nt
Sprinkler Permit -	Delta Building/Plumbing Bylaw (No. 6060)	43d	15-Jun-27	16-Aug-27										robaro S	nrinklor Dor	mit Applicat	ion	
RG1616739		200	15-Jun-27	13-Ju-27											plinkier rei			
RG1616740		30	14-Jul-27	16-Jul-27									3					
RG1616741	Review Period	20d	19-Jul-27	16-Aug-27														
RGT616742	Receive Approval/Permit - Sprinkler Permit	0d		16-Aug-27									•	Receive	Approval/Pe	ermit - Sprin	kier Pern	nt
Occupancy Permit	t - Delta Building/Plumbing Bylaw (No. 6060)	43d	15-Jun-27	16-Aug-27												Jormit Annli	oction	
RGT616743	Prepare Occupancy Permit Application	20d	15-Jun-27	13-Jul-27										repare C	ccupancy r	-emir Appi	auon	
RGT616744	SubmitApplication	3d	14-Jul-27	16-Jul-27							· · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	3	upmitAp	plication			
RGT616745	Review Period	20d	19-Jul-27	16-Aug-27	_									Review F	eriod			
RGT616746	Receive Approval/Permit - Occupancy Permit	Od		16-Aug-27									•	Receive	Approval/Pe	ermit - Occu	pancy Po	ərmit
Development Pern	nit (Commercial/Industrial) - Delta Official Community Plan Bylaws	143d	18-Jan-27	11-Aug-27														
RGT406482	Prepare Development Permit (Commercial/Industrial) Application	60d	18-Jan-27	13-Apr-27	_								Prep	are Deve	elopment Pe	ermit (Comi	nercial/In	dustrial) Ap
RGT406484	Submit Application	3d	14-Apr-27	16-Apr-27								· · ·	I Subi	nitApplic	ation			
RGT406486	Review Period	80d	19-Apr-27	11-Aug-27										Review F	eriod			
RGT406488	Receive Approval/Permit	0d		11-Aug-27									•	Receive	\pproval/Pe	ermit		
Demolition Permit	- Delta Official Community Plan Bylaws	78d	02-Mar-27	21-Jun-27														
RGT406682	Prepare Demolition Permit Application	15d	02-Mar-27	22-Mar-27									Prepa	are Demo	lition Permi	t Applicatio	ו	
RGT406684	SubmitApplication	3d	23-Mar-27	25-Mar-27									Subn	nit Applica	ition			
RGT406686	Review Period	60d	26-Mar-27	21-Jun-27								+ + 	📕 Re	eview Pe	iod			
RGT406688	Receive Approval/Permit	0d		21-Jun-27									♦ Re	eceive Ap	proval/Perr	nit		
TSBC - Safety Stan	Idards Act	55d	22-Jun-27	09-Sep-27														
Boiler and/or Pres	ssure Vessel Registration / Approval	35d	22-Jun-27	11-Aug-27														
RGT415182	Prepare Boiler and/or Pressure Vessel Registration Application	22d	22-Jun-27	23-Jul-27										repare E	oiler and/or	Pressure	/essel Re	gistration i
RGT415184	Submit Application	3d	23-Jul-27	27-Jul-27	1 1							· · ·	1 \$	Submit Ap	plication		1 1	
RGT415186	Review Period	10d	28-Jul-27	11-Aug-27										Review F	eriod			
RGT415188	Receive Approval/Permit	0d		11-Aug-27									•	Receive /	Approval/Pe	ermit		
Pressure Piping R	Registration / Approval	35d	22-Jun-27	11-Aug-27														
RGT415282	Prepare Pressure Piping Registration Application	22d	22-Jun-27	23-Jul-27									F	repare F	ressure Pip	oing Registr	ation App	lication
RGT415284	SubmitApplication	3d	23-Jul-27	27-Jul-27								+! 1 1 1 1 1 1	1 5	SubmitAp	plication			
RGT415286	Review Period	10d	28-Jul-27	11-Aug-27										Review F	eriod			
RGT415288	Receive Approval/Permit	0d		11-Aug-27									•	Receive /	Approval/Pe	ermit		
Installation permit	s - Electrical System	35d	22-Jun-27	11-Aug-27														
RGT415382	Prepare Electrical System Installation Permit Application	22d	22-Jun-27	23-Jul-27									 F	repare E	lectrical Sys	stem Install	ation Perr	mit Applicat
RGT415384	Submit Application	3d	23-Jul-27	27-Jul-27								4! 1 1 1 1 1 1	1 5	Submit Ap	plication			
RGT415386	Review Period	10d	28-Jul-27	11-Aug-27										Review F	eriod			
RGT415388	Receive Approval/Permit	Od		11-Aug-27									•	Receive /	Approval/Pe	ermit		
Installation permit	ts - Refrigeration System	35d	22-Jun-27	11-Aug-27														
RGT415482	Prepare Refrigeration Installation Permit Application	22d	22-Jun-27	23-Jul-27									 F	repare F	efrigeration	n Installatior	Permit A	plication
RGT415484	Submit Application	3d	23-Jul-27	27-Jul-27								+	18	Submit Ap	plication			
RGT415486	Review Period	10d	28-Jul-27	11-Aua-27										Review F	eriod			
RGT415488	Receive Approval/Permit	Dd		11-Aug-27									•	Receive /	Approval/Pe	ermit		
	s - Gas	354	22_ lun 27	11_Aug 27														
	0 - Va0	300	22-JUN-27	11- <i>i</i> -uy-27							1	1 1	, v , v ;				i i	

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data:26 Jul 24	Date	
Actual evel of Effort	Bemaining Work		Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 61 01 75	Í	

SOLARIS

Revision	Checked	Approved



Activity ID Activity Name	Rem. Duration	Start	Finish		2024 202	
RGT415582 Prepare Gas Installation Permit Application	22d	22-Jun-27	23-Jul-27			Prepare Gas Installation Permit Application
RGT415584 Submit Application	3d	23-Jul-27	27-Jul-27			I Submit Application
RGT415586 Review Period	10d	28-Jul-27	11-Aug-27			Review Period
RGT415588 Receive Approval/Permit	0d		11-Aug-27			♦ Receive Approval/Permit
Operation permits - Electrical System	35d	21-Jul-27	09-Sep-27			
RGT415682 Prepare Electrical System Operation Permit Application	22d	21-Jul-27	20-Aug-27	7		Prepare Electrical System Operation Permit Applica
RGT415684 Submit Application	3d	23-Aug-27	25-Aug-27			Submit Application
RGT415686 Review Period	10d	26-Aug-27	09-Sep-27			Review Period
RGT415688 Receive Approval/Permit	0d		09-Sep-27			♦ Receive Approval/Permit
Operation permits - Refrigenation System	35d	21-Jul-27	09-Sep-27			₩
RGT415782 Prepare Electrical System Operation Permit Application	22d	21-Jul-27	20-Aug-27			Prepare Electrical System Operation Permit Application
RGT415784 Submit Application	3d	23-Aug-27	25-Aug-27			Submit Application
RGT415786 Review Period	10d	26-Aug-27	09-Sep-27			Review Period
RGT415788 Receive Approval/Permit	Od		09-Sep-27			♦ Receive Approval/Permit
Operation permits (Boiler, Pressure Vessels)	35d	21-Jul-27	09-Sep-27			
RGT415882 Prepare Boiler & Pressure Vessels Operation Permit Application	22d	21-Jul-27	20-Aug-27			Prepare Boiler & Pressure Vessels Operation Perm
RGT415884 Submit Application	3d	23-Aug-27	25-Aug-27			Submit Application
RGT415886 Review Period	10d	26-Aug-27	09-Sep-27			Review Period
RGT415888 Receive Approval/Permit	Od		09-Sep-27			♦ Receive Approval/Permit
Port of Vancouver	25d	23-Aug-27	27-Sep-27			₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩
Notice to Shipping - Canada Marine Act	25d	23-Aug-27	27-Sep-27		· · · · · · · · · · · · · · · · · · ·	$\overline{\mathbf{A}}$
RGT413184 Notice to Shipping Registration	5d	23-Aug-27	27-Aug-27			
RGT413188 Receive Info/Confirmation	20d	30-Aug-27	27-Sep-27			Receive Info/Confirmation
BC One Call	4d	15-Oct-27	20-Oct-27			▼
BC One Call Registration	4d	15-Oct-27	20-Oct-27			■ I BC One Call Registration
DCT401104 DC One Call Registration	24	19-0ct-27	15-0ct-27		· · · · · · · · · · · · · · · · · · ·	
	Su	10-0ct-27	20-00-27			
Highway Use Permit - Transportation Act	85d	30-Mar-27	29-Jul-27			
RGT410182 Prepare Highway Use Permit Application	20d	30-Mar-27	28-Apr-27			Prepare Highway Use Permit Application
RGT410184 Submit Application	3d	28-Apr-27	30-Apr-27			I Submit Application
RGT410186 Review Period	62d	03-Mav-27	29-Jul-27			💼 Review Period
RGT410188 Receive Approval/Permit	0d	y	29-Jul-27			♦ Receive Approval/Permit
Work Safe BC - Worker Compensation Act/OHS Regulation	6d	09-Aug-27	16-Aug-27			\mathbf{V}
Guidelines 20.3-2 Qualified coordinators	6d	09-Aug-27	16-Aug-27			▼
RGT416182 Prepare Application of Guidelines 20.3-2 Qualified coordinators	1d	09-Aug-27	09-Aug-27			I Prepare Application of Guidelines 20.3-2 Qualified c
RGT416184 Submit Application	3d	10-Aug-27	12-Aug-27			Submit Application
RGT416186 Review Period	2d	13-Aug-27	16-Aug-27			I Review Period
RGT416188 Receive Approval/Permit	Od		16-Aug-27			♦ Receive Approval/Permit
G20.2(1)/20.2.1(1) Notice of Project, Section 20.2(1) of the OHS Regulation	6d	09-Aug-27	16-Aug-27			\blacksquare
RGT416282 Prepare Work Safe BC Application - Notice of Project	1d	09-Aug-27	09-Aug-27			Prepare Work Safe BC Application - Notice of Project
RGT416284 Submit Application	3d	10-Aug-27	12-Aug-27			Submit Application
RGT416286 Review Period	2d	13-Aug-27	16-Aug-27			I Review Period
RGT416288 Receive Approval/Permit	0d		16-Aug-27			♦ Receive Approval/Permit
Remaining Level of Effort Actual Work Critical Remaining Work			Data Date:	 26lul-24	Date	Revision Checked Approved
Actual Level of Effort Remaining Work Critical Secondary			Page: 62	of 73	19-Aug-24 IFI	

SOLARIS



ity ID	Activity Name	Rem.	Start	Finish	20 2021 2022 2023	2024	2025	2026 2027	2028 2029	2030	2031
		Duration									
30M33 Permit - 0	OHS Regulation Sections 19	6d	09-Aug-27	16-Aug-27					Nork Safe BC An	nlication - 30	M33 Permit
RG1410382		10	09-Aug-27	09-Aug-27				I Submit	Annlication		
RG1416384	Submit Application	30	10-Aug-27	12-Aug-27							
RG1416386	Review Period	20	13-Aug-27	16-Aug-27							
RG1416388	Receive Approval/Permit	0d		16-Aug-27					eApprovarPermit		
CBSA/CRA/Other	f Nilk antaida af DO	18d	09-Jul-27	04-Aug-27							
RGT/05182	Prenare Modular Units Built Outside of BC Permit Application	180 10d	09-Jul-27	04-Aug-27				Prepare	Modular Units Built	Outside of I	BC Permit Ar
PCT405184		34	23 Jul 27	23-50F27				I Submit	Application		
DCT405104		50 Ed	23-Jul-27						Period		
RG1405186		DC	28-JUI-27	04-Aug-27					Annroval/Permit		
RG1405188	Receive Approval/Permit	Ud		04-Aug-27					аррючангенны		
Nav Canada	imant Email Notification	135d	19-Apr-27	29-Oct-27							
Email Notificati		15d	19-Apr-27	07-May-27							
RGT404184	Prepare and Send Notification Email	5d	19-Apr-27	26-Apr-27				Prepare ar	d Send Notification	Email	
RGT404188	Receive Confirmation Response	10d	26-Apr-27	07-May-27				Receive C	onfirmation Respon	se	
	am - Grane	128d	28-Apr-27	20-Oct-27							
RGT411182	Prepare Land Use Crane Permit Application	5d	28-Apr-27	04-May-27				I Prepare La	and Use Crane Perr	mit Applicatio	on
RGT411184	Submit Application	3d	05-May-27	07-May-27				Submit Ap	olication		
RGT411186	Review Period *Chk duration with Andrew/Matvia*	120d	10-May-27	29-Oct-27				Rev	ew Period *Chk du	uration with A	Andrew/Mat
PGT/11188		0d	10-May-21	20-0ct-27				A Rec	eive Approval/Permi	it	
Transport Canada		1024	OF Nev OC					· · · · · · · · · · · · · · · · · · ·			
Aeronautical Cle	a parance - Canadian Aviation Regulations (CARs)	128d	18- lan-27	20- Jul 27							
RGT414182	Prepare Aeronautical Clearance Permit Application	5d	18-Jan-27	22-Jan-27				Prepare Aeror	nautical Clearance F	Permit Applic	ation
RGT414184	Submit Application	3d	25-Jan-27	27-Jan-27				Submit Applica	ation		
RGT414186	Review Period	120d	28- Jan-27	20- JuL27					Period		
PGT/1/188	Pocoivo Approval/Pormit	0d	20 0011 27	20 Jul 27				A Receive	Approval/Permit		
Neurigeble Weter	The cerve Approval Fernic	1024	OF New OC								
RGT414282	Prenare Navigable Water Permit Application	60d	05-Nov-26	04-Aug-27				Prepare Navi	able Water Permit	Application	
RGT/11/28/	Submit Application	34	08-Eeb-27	10-Eeb-27				I Submit Applic	ation		
DOT414204		1004	11 Eab 27						Period		
RG1414200		1200	11-Feb-27	04-Aug-27					Approval/Permit		
RG1414288	Receive Approval/Permit			04-Aug-27							
Procurement		625d	29-Jun-26	11-Jan-29							
STRUCTURAL N	MATERIAL	371d	29-Jun-26	04-Jan-28							
TRP296222.1	LNG Tank Proc. Hold Release (after receiving ph2 EAC)	0d	20 0011 20	29-Jun-26				♦ LNG Tank Proc. Hold	Release (after rece	iving ph2 EA	AC)
PRST1080	ROS: To Customs Clearance: Pkg #1 Structural Steel - Pump Platform Modules	0d		15-Oct-27					: To Customs Cleara	ance: Pkg #	1 Structural
PRST2080	ROS: Structural Steel Pkg #2 Platforms Stairs Walkways & Handrail (Shop)	DQ		07-Dec-27				♦ RC	S: Structural Steel	Pkg #2 Platfo	orms, Stairs,
PRST3080	ROS: Structural Steel Pkg #3 - Tank Grade Rack	Dd		15-Dec-27				A RC)S: Structural Steel I	Pkg #3 - Tan	k Grade Ra
EBM01020	POS: Dessive Structural Steel at Eab Vard	00 60							OS: Receive Structu	ural Steel at I	Fab Yard
			00 Mar 07	02 Date 07							
	ROS: Nelson Studs (SS)	2050 0d	02-iviar-27	23-Dec-27				 ROS: Nelsor 	Studs (\$S)		
MQ4140	POS: Nolcon Stude (OS)	- DU - LO		02 Mar 27				▲ ROS Nelson	Studs (CS)		
IVIO4110								▼ ROO. Nobol	S Embeds		
MS4120	RUS Shop: US Embeds	Ûd		08-Mar-27							
Domaining Law	up of Effort Actual Work Critical Domaining Work					Date		Revision	Chec	ked	Approved
Remaining Lev	VELOT ETTOTE TOTAL ACTUAL WORK THE CITCLE REMAINING WORK			Data Dato 2	6- lul-24						

Actual Level of Effort

Data Date:26-Jul-24 Page: 63 of 73

SOLARIS



Activ	ty ID	Activity Name	Rem.	Start	Finish	20 2021 2022	2 2023	2024	2025	2026 2027 2028	2029	2030 2031
			Duration						QQQ			
	MS4130	ROS Shop: Capping Material	0d		09-Mar-27					◆ ROS Shop: Capping N	/laterial	
	MS4140	ROS: Weld Material (CS)	0d		11-Mar-27						,5)	
	MS4150	ROS Shop: Temp. Erection Matls (Roof Framing & Doorsheet Stiffening)	0d		26-Mar-27	· · · · · · · · · · · · · · · · · · ·			; ;	♦ ROS Snop: lemp. Ere	ection Matis (Roc	or Framing & Doors
	MS4160	ROS: Foamglas	0d		07-Apr-27					♦ RUS: Hoamglas		
	MS4170	ROS Shop: Nozzles & Apperts	0d		14-Apr-27					♦ ROS Shop: Nozzles &	& Apperts	
	MS4180	ROS: Resilient Blanket	0d		28-Apr-27					♦ ROS: Resilient Blank	et	
	MS4190	ROS: Fiberglass Deck Blanket	0d		19-May-27					♦ ROS: Fiberglass De	eck Blanket	
	MS4200	ROS: Aluminum Plates - Suspended Deck	0d		25-May-27					◆ ROS: Aluminum Plat	tes - Suspended	Deck
	MS4210	ROS Shop: Plates CS (Roof Plates)	0d		27-May-27					♦ ROS Shop: Plates 0	CS (Roof Plates)	
	MS4220	ROS Shop: Plates CS (Compression Bar)	0d		27-May-27					♦ ROS Shop: Plates (CS (Compressio	n Bar)
	MS4230	ROS Shop: Plates CS (Shell Plates)	0d		27-May-27				1 1 1 1 1 1 1 1 1	♦ ROS Shop: Plates (CS (Shell Plates)	
	MS4240	ROS Shop: Plates CS (Bottom Plates)	0d		27-May-27					♦ ROS Shop: Plates C	CS (Bottom Plate	es)
	MS4250	ROS: Weld Material (Aluminum)	0d		10-Jun-27					♦ ROS Weld Materia	ıl (Aluminum)	
	MS4260	ROS Shop: Internal Piping	0d		10-Jun-27				· · · ·	♦ ROS Shop: Interna	l Piping	
	MS4280	ROS Shop: Pump Tubes	Od		17-Jun-27					♦ ROS Shop: Pump ⁻	Tubes	
	MS4270	ROS Shop: Misc. Plate, Bar, Angle (Stiffeners, Door Sheets, Embeds, Framing)	Od		17-Jun-27					♦ ROS Shop: Misc. P	late, Bar, Angle (Stiffeners, Door S
	MS4290	ROS: Cryogenic Reinforcing Steel / Krybar	0d		25-Jun-27					♦ ROS: Cryogenic R	einforcing Steel	/ Krybar
	MS4300	ROS: Weld Material (9% Nickel)	0d		13-Jul-27					♦ ROS: Weld Materi	ial (9% Nickel)	
	MS4310	ROS: Perlite	0d		09-Aug-27					♦ ROS: Perlite		
	MS4320	ROS: Tank RTD's	0d		13-Aug-27					♦ ROS: Tank RTD's	S	
	MS4330	ROS: Isolators	0d		20-Aug-27					♦ ROS: Isolators		
	MS4340	ROS Shop: Roof Embeds	Od		03-Nov-27					♦ ROS \$hop: R	oof Embeds	
	MS4350	ROS Shop: Internal Stairways, Platforms & Pipe Supports	Od		24-Nov-27					🔶 ROS Shop: Ir	nternal Stairways	, Platforms & Pipe
	MS4360	ROS: 9% Ni Plates	Od		16-Dec-27					♦ ROS: 9% Ni	Plates	
	MS4370	ROS: PRV's & VRV's	0d		23-Dec-27					♦ ROS: PRV's	& VRV's	
	MECHANICAL M	ATERIAL	5d	04-Feb-28	11-Feb-28					▼		
	MS4000	ROS: Jib Crane	0d	0110020	04-Feb-28					♦ ROS: Jib C	Crane	
	MS4010	ROS: Dry Chem System	0d		11-Feb-28					♦ ROS: Dry (Chem System	
	PIPING MATERIA		234d	10-Mar-27	18-Feb-28							
	MS4400	ROS: Field (Small Bore) CS & SS Pipe, Fittings & Flanges	0d		10-Mar-27					♦ ROS: Field (Small Bore	e) CS & SS Pipe	, Fittings & Flange
	MS4410	ROS Shop: (Large Bore) CS & SS Pipe, Fittings	0d		21-Apr-27					♦ ROS Shop: (Large B	ore) CS & SS Pi	pe, Fittings
	MS4420	ROS: CS & SS Fasteners	0d		02-Jun-27					♦ ROS CS & SS Fas	teners	
	MS4430	ROS: Hose Connections	0d		13-Aug-27					♦ ROS: Hose Conr	nections	
	MS4440	ROS: Spiral Wound / Garlok / Neoprene Gaskets	0d		20-Sep-27					♦ ROS: Spiral Wo	ound / Garlok / N	eoprene Gaskets
	MS4450	ROS: Cold Shoes & Misc. Pipe Supports	0d		06-Oct-27				1 1 1 1 1 1 1 1 1	♦ ROS: Cold Sho	bes & Misc. Pipe	Supports
	MS4460	ROS: Manual Ball Valves	0d		21-Jan-28				1 1 1 1 1 1 1 1 1 1 1	♦ ROS: Manu	ual Ball Valves	
	MS4470	ROS: Manual Gate/Globe/Check Valves	0d		21-Jan-28					🔶 ROS: Manu	ual Gate/Globe/0	Check Valves
	MS4480	ROS: Manual Butterfly Valves	0d		18-Feb-28					♦ RÓS: Man	ual Butterfly Val	/es
	MS4490	ROS: Piping & Component Matts - Tank Grade Rack	0d		18-Feb-28				k	♦ ROS: Pipir	ng & Componen	t Matls - Tank Grac
		TERIAL	48d	28-Jun-27	07-Sep-27					•••		
	MS4510	ROS: Lighting Fixtures & Lamps	0d		28-Jun-27					♦ ROS: Lighting Fixt	ures & Lamps	
	MS4500	ROS: Cable Ladders & Conduit	0d		05-Jul-27					♦ RO\$: Cable Ladde	ers & Conduit	
	18.								· · ·		01	Δ
	Remaining Lev	el of Effort Actual Work Critical Remaining Work			Data Date:26	-Jul-24	10		IFI	Revision		Approved
	Actual Level of Effort Effort Remaining Work Critical Secondary				Page: 64 o	f 73	15	, wy ∠ -r			1	

SOLARIS



Activ	ity ID	Activity Name	Rem.	Start	Finish	20 202	21	2022	2023	2024	2025	202	26	2027	2028	2029	203	30 2031
	MS4E20	POSt Tank Crounding & Lightning System			15 101 07										Q Q Q Q Tank Grou	QQQQ nding & Light	QQ ning Sv	QQQQQQ vstem
	MS4520	ROS: Tank Grounding & Lightning System	bU							·					Termination	n Kits Outlets		& Switches
	MS4530	ROS: lermination Kits, Outlets, Plugs & Switches	Ud		22-JUF-27											nnis, Ouici onent Matle	, Tank	Grade Back
	MS4570	RUS: E&I Component Matis - Iank Grade Rack	Ud		23-Aug-27										S. Dowor &	Instrument C		Glade Nack
	MS4540	RUS: Power & Instrument Cables	Ud		24-Aug-27													
	MS4550	ROS: Junction Boxes & Enclosures	0d		01-Sep-27			· · · ·									losures	ز
	MS4560	ROS: Power Cable Glands	0d		07-Sep-27				· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·				● RU	S. Power C	able Glands		
	INSTRUMENTATIO	ON MATERIAL	127d	21-May-27	24-Nov-27										low Elemen	ite		
	MS4600	RUS: Flow Elements	Ud		21-May-27	-												
	MS4610	RUS: Pressure Gauges	Ud		16-Jun-27										Inclinamata	r r		
	MS4620		Ud		23-Jun-27											n No	torial	
	MS4630	ROS: Process Connection Material	0d		28-Jun-27				· · · · · · · · · · · · · · · · · · ·						Monuel Co		lenai	
	MS4640	ROS: Manual Call Point	0d		15-Jul-27										ManuarCa			
	MS4650	ROS: Instrument Stands & Supports	0d		22-Jul-27											Stands & SL	ipports	
	MS4660	ROS: Transmitters	0d		26-Jul-27									♦ ROS	: Iransmitte	rs		
	MS4670	ROS: Fire & Gas Detection	0d		07-Oct-27									♦ RC)S: Fire & C	as Detection		
	MS4680	ROS: Tank Gauging System	0d		29-Oct-27									♦R	OS: Tank G	auging Syste	m	
	MS4690	ROS: Control Valves	0d		24-Nov-27									♦F	ROS: Contro	ol Valves		
	SUBCONTRACT E	BID, EVALUATE & AWARD	591d	19-Aug-26	11-Jan-29								V			7		
	Civil Concrete Wo		220d	30-Sep-26	20-Aug-27									ubot: Pid/	Aword/ Cub	mittala: Civil (onorat	to Works
	SUBC1000	Subct: Bid/ Award/ Submittals: Civil Concrete Works	80d	30-Sep-26	29-Jan-27										Award Cub			
	MS3030	Construction: Award Subcontract for Tank Grade Rack Fdns	0d		04-Dec-26				· · · · ·	· · · · · · · · · · · · · · · · · · ·				nstruction:	Award Sub		ank Gr	
	SUBC2220	Subct: Civil Sub Generate & HCBI Review Slab, Wall & Roof Construction Dwgs	s T 140d	01-Feb-27	20-Aug-27									Sub	ct: Civil Sud	Generate &	нсығ	(eview Slab, vv
	Tower Crane		80d	19-Aug-26	11-Dec-26									ot: Did/ Aver	rd. Towor (rano		
	SUBC2080	Subct: Bid/ Award: Tower Crane	40d	19-Aug-26	15-Oct-26										ubmittala 9		Tour	or Cropp
	SUBC2110	Subct: Sub Submittals & HCBI Review - Tower Crane	40d	16-Oct-26	11-Dec-26									DCL: SUD S	uomiliais &		v – IOWe	ercrane
	Tank Grade Rack	Fdns	80d	08-Oct-26	08-Feb-27								Suk	hct: Bid/ Av	vard: Tank (Srade Rack I	Edne	
	SUBC2090	Subct: Bid/ Award: Tank Grade Rack Fons	40a	08-Oct-26	04-Dec-26									Subat: Sub	Submittala			rada Baak Edn
	SUBC2160	Subct: Sub Submittals & HCBI Review - Grade Rack Fons	40d	07-Dec-26	08-Feb-27										Subinimais		ew - Gi	aue nack run
	Site Setup Electri	cal Subat: Bid/Award: Site Satur Electrical	60d	19-Jan-27	14-Apr-27									Subct Bid	Award Site	Setun Elect	rical	
	SUBC2070	Subct: Bid/ Award: Sile Selup Electrical	300	19-Jan-27	02-Iviar-27									Subct: Si	ih Submitta			Site Setup Fler
	SUBC2200	Subci: Sub Submittais & HCBI Review - Site Setup Electrical	300	03-IVIar-27	14-Apr-27			· · · ·	· · · ·	·								
		Subst: Bid/Award: F&I	80d	26-Nov-27	27-Mar-28										Subct: Bid	Award: F&I		
	SURC2120	Subet: Sub Submittale & HCRI Paview - E&I		31, lan 28	20-041-20 27-Mar 29									T	Subct S	Sub Submitta	s&HC	Bl Review - मि
	SUDCZ 120	Subci. Sub Submittais & TICDI Neview - Ext	400 75d	17 Eab 27														
	SUBC2170	Subct: Bid/Award: NDF	40d	17-Feb-27	14-Apr-27									Subct: Bi	d/Award: N	DE		
	SUBC2190	Subct: Sub Submittals & HCRI Review - NDF	35d	15-Apr-27	03- Jun-27			· · · · ·		· · · · · · · · · · · · · · · · · · ·				Subct:	Sub Submit	tals & HCBI I	Review	- NDE
	Surveyor		60d	31_Aug_27	25-Nov-27									-				
	SUBC2020	Subct: Bid/Award: Survevor	30d	31-Aug-27	13-Oct-27									📕 📕	ıbct: Bid/Aw	ard: Survey	b r	
	SUBC2210	Subct: Sub Submittals & HCBI Review - Surveyor	30d	14-Oct-27	25-Nov-27									– S	Subct: Sub \$	Submittals & I	HCBI R	(eview - Surve)
	Heavy Haul		80d	10-May-28	31-Aug-28									_				,
	SUBC2060	Subct: Bid/Award: Heavy Haul	40d	10-May-28	06-Jul-28					$ \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1} \frac{1}{1}$					📕 Sub	ct: Bid/Awarc	: Heav	/y Haul
	SUBC2100	Subct: Sub Submittals & HCBI Review - Heavy Haul	40d	07-Jul-28	31-Aua-28										🗖 Su	ibct: Sub Sub	mittals	& HCBI Reviev
	Piping & Insulation	on	120d	02-Jun-28	22-Nov-28										-			
							. I			Date		<u> </u>	Revisio			Checke	d	Approved
					Data Date:2	6-Jul-24			F	19-Aug-24	IFI							
					Page: 65	UT / 3			F									

SOLARIS



	01	FODTIC	Tilbu	ry Island I	LNG Facility	Expansion (3 BCF)	SOLARIS
		FORIIS BC ^{**}	TLSE	Sub-Proje	cts Integrated	l Schedule-Detailed	
							CONSULTANTS INC.
Activity ID		Activity Name	Rem. Duration	Start	Finish 220		
	SUBC2030	Subct: Bid/ Award: Piping	40d	02-Jun-28	28-Jul-28		
	SUBC2050	Subct: Bid/Award: Pipe Insulation	40d	31-Jul-28	25-Sep-28		Subct: Bid/Award: Pipe Insulation
	SUBC2130	Subct: Sub Submittals & HCBI Review - Piping	40d	31-Jul-28	25-Sep-28		📕 Subct: Sub Submittals & H¢BI Rev
	SUBC2140	Subct: Sub Submittals & HCBI Review - Pipe Insulation	40d	26-Sep-28	22-Nov-28		📕 Subct: Sub Submittals & HCBI R
	Scaffolding		70d	26-Sep-28	11-Jan-29		
	SUBC2150	Subct: Bid/Award: Scaffolding	40d	26-Sep-28	22-Nov-28		Subct: Bid/ Award: Scaffolding
	SUBC2180	Subct: Sub Submittals & HCBI Review - Scaffolding	30d	23-Nov-28	11-Jan-29		Subct: Sub Submittals & HCBI
	13 LNG TANK MA	ITERIALS	318d	14-Sep-26	23-Dec-27		
	RFQ to PO		233d	14-Sep-26	23-Aug-27		REO to PO: 9%Ni Plates
	PRIK0100	RFQ to PO: 9%NI Plates	080 05d	14-Sep-26	18-Dec-26		
	PRTK0140	RFQ to PO: Aluminum Plates	350	08-Oct-26	27-NOV-26		
	PRIK0070	RFQ to PO: CS Plates	330	13-Oct-26	27-INOV-20		
	PRIK0410	RFQ to PO: Misc. Plates, Bars, Angles	35d	13-Oct-26	01-Dec-26		
	PRTK0620	RFQ to PO: Weld Material (CS)	350	13-Uct-26	01-Dec-26		\square REO to P(). Pump Tubes
	PRTK0290	RFQ to PO: Pump lubes	350	06-Nov-26	04-Jan-27		BEO to DO: Conning Material
	PRIK0/10	RFQ to PO: Capping Material	35d	23-Nov-26	18-Jan-27		
	PRIK0380	RFQ to PO: PRV's & VRV's	70d	30-Nov-26	16-Mar-27		
	PRIK0530	RFQ to PO: Nozzles & Appurt.	32d	30-Nov-26	20-Jan-27		\square DEO to DO: Noleon Stude
	PRIK0560	RFQ to PO: Nelson Studs	350	30-Nov-26	25-Jan-27		PEO to PO: Wold Material (0% Ni)
	PRIK0650	RFQ to PO: Weld Material (9% NI)	350	02-Dec-26	27-Jan-27		= PEO to PO: Tomp. Eraction Material
	PRIK0590	RFQ to PO: lemp. Erection Material	35d	14-Jan-27	04-Mar-27		PEO to PO: Wold Material (Aluminum)
	PRIK0680	RFQ to PO: Weld Material (Aluminum)	45d	14-Jan-27	18-Mar-27		
	PRIK0350	RFQ to PO: lank RTD's	35d	26-Jan-27	16-Mar-27		
	PRTK0320	RFQ to PO: Internal Piping	38d	12-Mar-27	05-May-27		
	PRTK0440	RFQ to PO: Internal Stairways, Platforms & Pipe Supports	63d	25-May-27	23-Aug-27		
		Supplier Load Time: CS Plates	260d	30-Nov-26	16-Dec-27		Supplier Lead Time: CS Plates
		Supplier Lead Time: Aluminum Plates	900	30-Nov-20	14-Api-27		Supplier Lead Time: Aluminum Plates
		Supplier Lead Time: Authinfull Flates	1224	02 Doc 26	$02 \ \text{lup} \ 27$		Supplier Lead Time: Misc. Plates. Bars. Angles
		Supplier Lead Time: Wold Motorial (CS)	60d	02-Dec-20	02-Juli-27		Supplier/Lead Time: Weld Material (CS)
		Supplier Lead Time: Weld Waterial (CS)	2004	02-Dec-20	12 Oct 27		Supplier Lead Time: 9%Ni Plates
		Supplier Lead Time: 97011 Flates	2000 95d	21-Dec-20	05 May 27		
		Supplier Lead Time: Capping Material	204	10 Jan 27	16 Ecb 27		Supplier Lead Time: Capping Material
		Supplier Lead Time: Capping Material Supplier Lead Time: Nozzles & Appurt	200 43d	19-Jan-27	10-Feb-27		Supplier Lead Time: Nozzles & Appurt
		Supplier Lead Time: Nolcon Stude	430	21-Jan-27	23-1Vidi-27		Supplier Lead Time: Nelson Studs
		Supplier Lead Time: Weld Meterial (0% Ni)	200 110d	20-Jan 27	23-Feb-27		Supplier Lead Time: Weld Material (9% Ni)
		Supplier Lead Time: Tomp Erection Meterial	11d	20-Jan-27	10 Mar 27		Supplier Lead Time: Temp. Frection Material
	DDTKUSED		054	17_Mar 27	30_ lul 27		
			DCF	$\frac{17 \text{ Nor } 07}{17 \text{ Nor } 07}$			Supplier Load Time: PRV/s & V/R//s
		Supplier Lead Time: Mold Material (Aluminum)	1900	17-IVIAF-27			
		Supplier Lead Time: vveld Material (Aluminum)	43d	19-IVIAR-27	19-IVIAy-27		
		Supplier Lead Time: Internal Piping	20d	06-IVIAy-27	03-JUN-27		
	PK1K0450	Supplier Lead Time: Internal Stalfways, Platforms & Pipe Supports	50d	24-Aug-27	U3-INOV-27		
	Remaining Lev	vel of Effort Actual Work Critical Remaining Work			Data Date:26-Ju	II-24	Revision Checked Approved

Actual Level of Effort

Remaining Work
 Critical Secondary

Page: 66 of 73



Activity	ID	Activity Name	Rem. Duration	Start	Finish							
	Delivery		215d	17-Feb-27	23-Dec-27							
	PRTK0730	Delivery: Capping Material	15d	17-Feb-27	09-Mar-27					Delivery Capping Mat	erial	
	PRTK0580	Delivery: Nelson Studs	5d	24-Feb-27	02-Mar-27					I Delivery: Nelson Studs		
	PRTK0640	Delivery: Weld Material (CS)	5d	05-Mar-27	11-Mar-27					I Delivery Weld Materia	I (CS)	
	PRTK0610	Delivery: Temp. Erection Material	5d	22-Mar-27	26-Mar-27					I Delivery: Temp. Erection	on Material	
	PRTK0550	Delivery: Nozzles & Appurt.	15d	24-Mar-27	14-Apr-27					Delivery: Nozzles & A	ppurt.	
	PRTK0160	Delivery: Aluminum Plates	37d	01-Apr-27	25-May-27					🔲 Delivery: Aluminum	Plates	
	PRTK0090	Delivery: CS Plates	30d	15-Apr-27	27-May-27					Delivery: CS Plates		
	PRTK0310	Delivery: Pump Tubes	30d	06-May-27	17-Jun-27					Delivery: Pump Tub	bes	
	PRTK0120	Delivery: 9%Ni Plates (1st Shipment to Everett Shop)	45d	20-May-27	23-Jul-27					💼 🛛 Delivery: 9%Ni Pla	ates (1st Shipme	nt to Everett Shop
	PRTK0700	Delivery: Weld Material (Aluminum)	15d	20-May-27	10-Jun-27					Delivery: Weld Mate	erial (Aluminum)	
	PRTK0430	Delivery: Misc. Plates, Bars, Angles	11d	03-Jun-27	17-Jun-27					Delivery: Misc. Plate	es, Bars, Angles	
	PRTK0340	Delivery: Internal Piping	5d	04-Jun-27	10-Jun-27					I Delivery: Internal Pi	ping	
	PRTK0670	Delivery: Weld Material (9% Ni)	5d	07-Jul-27	13-Jul-27					👔 Delivery: Weld Ma	terial (9% Ni)	
	PRTK0370	Delivery: Tank RTD's	8d	03-Aug-27	13-Aug-27					🛽 Delivery: Tank R	īD's	
	PRTK0130	Delivery: 9%Ni Plates (2nd Shipment to Site)	45d	14-Oct-27	16-Dec-27					🔲 Delivery: 9%	Ni Plates (2nd S	hipment to Site)
	PRTK0460	Delivery: Internal Stairways, Platforms & Pipe Supports	14d	04-Nov-27	24-Nov-27					🛛 Delivery: Inter	nal Stairways, P	latforms & Pipe Su
	PRTK0400	Delivery: PRV's & VRV's	5d	17-Dec-27	23-Dec-27					Delivery: PR	V's & VRV's	
	15 PIPING MATER	IALS	335d	08-Oct-26	18-Feb-28							
	RFQ to PO		192d	08-Oct-26	20-Jul-27							
	PRPI0010	RFQ to PO: Field (Small Bore) CS & SS Pipe, Fittings & Flanges	88d	08-Oct-26	19-Feb-27					RFQ to PO: Field (Sma	Ill Bore) CS & SS	Pipe, Fittings & Fl
	PRPI0040	RFQ to PO: Process Piping (Large Bore) CS & SS Pipe, Fittings & Flanges	83d	08-Oct-26	11-Feb-27					RFQ to PO: Process Pi	ping (Large Bore	e) CS & SS Pipe, F
	PRPI0070	RFQ to PO: Manual Butterfly Valves	51d	13-Nov-26	01-Feb-27					RFQ to PO: Manual Bu	tterfly Valves	
	PRPI0100	RFQ to PO: Manual Ball Valves	51d	13-Nov-26	01-Feb-27					RFQ to PO: Manual Ba	l Valves	
	PRPI0130	RFQ to PO: Manual Gate/Globe/Check Valves	51d	13-Nov-26	01-Feb-27					RFQ to PO: Manual Ga	te/Globe/Check	Valves
	PRPI1990	RFQ to PO: Piping & Component Matls - Tank Grade Rack	55d	26-Jan-27	14-Apr-27					RFQ to PO: Piping &	Component Mat	lls - Tank Grade Ra
	PRPI0190	RFQ to PO: Spiral Wound / Garlok / Neoprene Gaskets	108d	16-Feb-27	20-Jul-27					RFQ to PO: Spira	l Wound / Garlol	k / Neoprene Gasl
	PRPI0220	RFQ to PO: CS & SS Fasteners	40d	16-Feb-27	13-Apr-27					RFQ to PO: CS & SS	Fasteners	
	PRPI0250	RFQ to PO: Hose Connections	75d	16-Feb-27	02-Jun-27					RFQ to PO: Hose (Connections	
	PRPI0160	RFQ to PO: Cold Shoes & Misc. Pipe Supports	40d	05-Apr-27	31-May-27					RFQ to PO: Cold \$	hoes & Misc. Pip	e Supports
	Supply		255d	07-Dec-26	16-Dec-27				•			
	PRP10020	Supplier Lead Time: Field (Small Bore) CS & SS Pipe, Fittings & Flanges	56d	07-Dec-26	03-Mar-27					Supplier Lead Time: H	eld (Small Bore)	CS & SS Pipe, Fitt
	PRP10050	Supplier Lead Time: Process Piping (Large Bore) CS & SS Pipe, Fittings & Flanges	85d	07-Dec-26	14-Apr-27						Process Piping (L	_arge Bore) CS &
	PRP10080	Supplier Lead Time: Manual Butterfly Valves	220d	02-Feb-27	16-Dec-27					Supplier Lea	id lime: Manual	Butterny Valves
	PRPI0110	Supplier Lead Time: Manual Ball Valves	200d	02-Feb-27	18-Nov-27					Supplier Lead	I lime: Manual E	all valves
	PRPI0140	Supplier Lead Time: Manual Gate/Globe/Check Valves	200d	02-Feb-27	18-Nov-27					Supplier Lead	I lime: Manual C	Sate/Globe/Check
	PRPI0230	Supplier Lead Time: CS & SS Fasteners	20d	14-Apr-27	11-May-27					Supplier Lead Time:	CS & SS Faster	ners
	PRPI2000	Supplier Lead Time: Receive Material & Fabricate - Piping & Component Matls - Ta	168d	15-Apr-27	14-Dec-27					Supplier Lea	a lime: Receive	Material & Fabrica
	PRPI0170	Supplier Lead Time: Cold Shoes & Misc. Pipe Supports	75d	01-Jun-27	16-Sep-27					Supplier Lead T	Ime: Cold Shoes	s & Misc. Pipe Sup
	PRPI0260	Supplier Lead Time: Hose Connections	35d	03-Jun-27	22-Jul-27					Supplier Lead Tim	e: Hose Cohnec	ctions
	PRPI0200	Supplier Lead Time: Spiral Wound / Garlok / Neoprene Gaskets	28d	21-Jul-27	30-Aug-27					Supplier Lead Ti	me: Spiral Woun	id / Garlok / Neopr
	Delivery		239d	04-Mar-27	18-Feb-28							
		1						· · · · ·		•••		
	💶 Remaining Leve	el of Effort			Data Data 2	6 1.1. 24	Da	te	Re	vision	Checked	Approved

Actual Level of Effort

Remaining Work Critical Secondary

19-Aug-24 IFI

SOLARIS

Revision	Checked	Approved



Activ	ity ID	Activity Name	Rem.	Start	Finish)20 202´	1 202	2 2023	20	024	2025	2026	2027	2028	2	029 2	2030	2031
			Duration							QQ								
	PRPI0030	Delivery: Field (Small Bore) CS & SS Pipe, Fittings & Flanges	5d	04-Mar-27	10-Mar-27									y Field (S	mali Bore) CS & SS PI (arma arma)	be,⊢ittin	Igs & Fland
	PRPI0060	Delivery: Process Piping (Large Bore) CS & SS Pipe, Fittings & Flanges	5d	15-Apr-27	21-Apr-27									ery: Proce	SS Piping ((Large Bore)	υ δαδ	5 гре, ги
	PRPI0240	Delivery: CS & SS Fasteners	15d	12-May-27	02-Jun-27						·					eners		
	PRPI0270	Delivery: Hose Connections	15d	23-Jul-27	13-Aug-27									Pelivery: Ho	ose Conne	ections		
	PRPI0210	Delivery: Spiral Wound / Garlok / Neoprene Gaskets	14d	31-Aug-27	20-Sep-27									Delivery: S	Spiral Wou	Ind / Garlok /	Neopre	ine Gaske
	PRPI0180	Delivery: Cold Shoes & Misc. Pipe Supports	14d	17-Sep-27	06-Oct-27									Delivery: (Cold Shoe	es & Misc. Pip	e Supp	orts
	PRPI0120	Delivery: Manual Ball Valves	40d	19-Nov-27	21-Jan-28									🛑 Delive	ery: Manua	al Ball Valves		
	PRPI0150	Delivery: Manual Gate/Globe/Check Valves	40d	19-Nov-27	21-Jan-28									🛑 Delive	ery: Manua	al Gate/Glob	e/Check	(Valves
	PRPI2010	Delivery: Piping & Component Matls - Tank Grade Rack	42d	15-Dec-27	18-Feb-28									🔲 Deliv	ery: Pipin	g & Compon	ent Matl	s - Tank G
	PRP10090	Delivery: Manual Butterfly Valves	40d	17-Dec-27	18-Feb-28									🗖 Deliv	ery: Manı	ual Butterfly ∖	alves	
	03 CONCRETE N	IATERIALS	263d	13-Oct-26	03-Nov-27							•		7				
	RFQ to PO		192d	13-Oct-26	22-Jul-27													
	PRTK0470	RFQ to PO: CS Bottom Embeds	25d	13-Oct-26	17-Nov-26					1 1 1 1 1 1	1 1 1 1 1 1 1 1 1		RFQ to P): CS Botto	om Embec	ds		
	PRTK0010	RFQ to PO: Isolators	11d	11-Feb-27	26-Feb-27								∎ RFQ to	PO: Isola	tors			
	PRTK0040	RFQ to PO: Cryogenic Rebar	11d	19-Feb-27	05-Mar-27								∎ RFQ t	o PO: Cryo	genic Rel	bar		
	PRTK0500	RFQ to PO: CS Roof Embeds	42d	25-May-27	22-Jul-27								📕 R	FQ to PO:	CS Roof	Embeds		
	Supply		220d	18-Nov-26	07-Oct-27								V					
	PRTK0480	Supplier Lead Time: CS Bottom Embeds	53d	18-Nov-26	08-Feb-27						· · · · · · · · · · · · · · · · · · ·		💼 Supplie	r Lead Tim	e: CS Bot	tom Embeds		
	PRTK0020	Supplier Lead Time: Isolators	100d	01-Mar-27	21-Jul-27								s 🛑 S	upplier Lea	ad Time: Is	olators		
	PRTK0050	Supplier Lead Time & Fabrication - Cryogenic Rebar	70d	08-Mar-27	15-Jun-27								💼 Su	oplier Leac	l Time & F	abrication - (Cryogen	iic Rebar
	PRTK0510	Supplier Lead Time: CS Roof Embeds	53d	23-Jul-27	07-Oct-27									Supplier L	ead Time	: CS Roof Er	nbeds	
	Delivery		185d	09-Feb-27	03-Nov-27									7				
	PRTK0490	Delivery: CS Bottom Embeds	19d	09-Feb-27	08-Mar-27								Delive	y CS Bott	om Embe	ds		
	PRTK0060	Delivery: Cryogenic Rebar	8d	16-Jun-27	25-Jun-27								l De	livery: Cryo	ogenic Re	bar		
	PRTK3380	Delivery: Isolators	21d	22-Jul-27	20-Aug-27								I 🛛	Delivery: Iso	olators			
	PRTK0520	Delivery: CS Roof Embeds	18d	08-Oct-27	03-Nov-27									Delivery:	CS Roof	Embeds		
	17 INSTRUMENT	& CONTROL MATERIALS	275d	15-Oct-26	24-Nov-27								V	▼				
	RFQ to PO		138d	15-Oct-26	07-May-27						· · · · · · · · · · · · · · · · · · ·							
	PRIC0130	RFQ to PO: Control Valves	45d	15-Oct-26	17-Dec-26									O: Control	Valves			
	PRIC0040	RFQ to PO: Flow Elements	45d	28-Jan-27	01-Apr-27								RFQ	to PO: Flov	w Elemen	ts		
	PRIC0070	RFQ to PO: Pressure Gauges & Indicators	45d	04-Feb-27	09-Apr-27								🔲 RFQ	to PO: Pre	ssure Ga	uges & Indic	ators	
	PRIC0220	RFQ to PO: Tank Gauging System	55d	04-Feb-27	23-Apr-27								🔲 RFQ	to PO: Tar	nk Gaugin	ig System		
	PRIC0100	RFQ to PO: Transmitters	45d	11-Feb-27	16-Apr-27								🔲 RFQ	to PO: Tra	nsmitters			
	PRIC0280	RFQ to PO: Manual Call Point	45d	11-Feb-27	16-Apr-27								🔲 RFQ	to PO: Ma	nual Call I	Point		
	PRIC0250	RFQ to PO: Fire & Gas Detection	55d	19-Feb-27	07-May-27								🔲 RFC	to PO: Fir	re & Gas I	Detection		
	PRIC0190	RFQ to PO: Process Connection Material	35d	26-Feb-27	16-Apr-27								🔲 RFQ	to PO: Pro	ocess Cor	nnection Mate	erial	
	PRIC0160	RFQ to PO: Instrument Stands & Supports	35d	12-Mar-27	30-Apr-27								🗖 RFC	to PO: Ins	strument S	Stands & Sup	ports	
	PRIC0010	RFQ to PO: Inclinometer	35d	19-Mar-27	07-May-27								📕 RFC	to PO: Inc	clinometer	-		
	Supply		193d	18-Dec-26	30-Sep-27													
	PRIC0140	Supplier Lead Time: Control Valves	190d	18-Dec-26	27-Sep-27									Supplier L	ead Time	: Control Val	es	
	PRIC0050	Supplier Lead Time: Flow Elements	30d	05-Apr-27	14-May-27								Sup	plier Lead	Time: Flov	v Elements		
	PRIC0080	Supplier Lead Time: Pressure Gauges & Indicators	32d	, 12-Apr-27	26-May-27								🗖 Sup	plier Lead	Time: Pre	ssure Gauge	es & Ind	icators
	PRIC0110	Supplier Lead Time: Transmitters	64d	19-Apr-27	19-Jul-27								— S	upplier Lea	nd Time: Ti	ransmitters		
										Date			Revision			Checked	An	proved
	Remaining Lev				Data Date:26	j-Jul-24			19 - Au	g-24	IFI					5	14, 1	
	Actual Level of	r Εποπ Emaining Work Emaining Work Critical Secondary			Page: 68 (ot 73				-								

SOLARIS



Activ	ity ID	Activity Name	Rem. Duration	Start	Finish	20 202	1 2022	2023	2024	2025	2026	2027	2028	2029	2030 2031	
	PPIC0200	Supplier Load Time: Process Connection Material	364	10 Apr 27	08 Jup 27								oplier Lead Tin	al al al al al al a ne: Process Cor	viection Material	Q
			500 57d	19-Apt-27									pplier Lead Ti	me: Manual Cal	Point	
	PRIC0230	Supplier Lead Time: Tank Gauging System	100d	26_Apr-27	16-Sep-27								Supplier Lead	I Time: Tank Ga	uging System	
	PRIC0170	Supplier Lead Time: Instrument Stands & Supports	42d	03-May-27	30- Jun-27							🗖 Su	polier Lead Ti	me: Instrument	Stands & Supports	;
	PRICODO	Supplier Lead Time: Institutient Statius & Supports	42u	10 May 27	00 Jun 27							Sur	pplier Lead Tin	ne: Inclinometer		
		Supplier Lead Time: Fire & Cas Detection	22u	10-May-27	20 Sop 27						$ \frac{1}{1}$ $ \frac{1}{1}$ $ \frac{1}{1}$ $ \frac{1}{1}$ $ -$		Supplier Lead	d Time: Fire & G	as Detection	
	PRICU200	Supplier Lead Time. Fire & Gas Delection	1224	10-May-27	30-3ep-27											
	PRIC0060	Delivery: Flow Elements	5d	17-May-27	24-N0V-27 21-May-27							ı Deliv	▼ /ery: Flow Ele	ments		
	PRIC0090	Delivery: Pressure Gauges & Indicators	15d	27-May-27	16-Jun-27							🛛 Del	ivery: Pressur	e Gauges & Inc	icators	
	PRIC0210	Delivery: Process Connection Material	10d	09- lun-27	28- Jun-27							п De	livery: Proces	s Connection M	aterial	
	PRIC0030		10d	10- lup-27	23- Jun-27							∎ De	liverv: Inclinon	neter		
		Delivery: Instrument Stands & Supports	15d		20-5011-27							∎ De	eliverv: Instrun	nent Stands & S	upports	
			54		15 Jul 27							I De	elivery: Manua	al Call Point		
	PRIC0300		DC Ed	09-Jul-27	15-Ju-27								elivery: Transr	nitters		
	PRICU120	Delivery: Transmillers	DC	20-Jul-27	20-JUF27								Delivery: Tar	k Gauging Svs	em	
	PRICU240	Delivery: lank Gauging System	300	17-Sep-27	29-Oct-27									htrol Valves		
	PRICU150	Delivery: Control valves	40d	28-Sep-27	24-NoV-27								Delivery. Co	& Cas Dotoctio	n	
	PRIC0270	Delivery: Fire & Gas Detection	50	01-Oct-27	07-Oct-27						_					
	07 INSULATION M	AIERIALS	1/5d	23-Nov-26	09-Aug-27											
	PRTK0170	REQ to PQ: Foamulas Blocks	35d	23-Nov-26	18-Jan-27							RFQ to I	PO: Foamglas	Blocks		
	PRTK0230	REQ to PO' Resilient Blanket	50d	21-Dec-26	09-Mar-27				J L	L J L		🗖 RFQ to	PO: Resilien	t Blanket		
	PRTK0200	REQ to PO: Fiberglass Deck Blanket	35d	17-Feb-27	07-Apr-27							RFQ	to PO: Fibergl	ass Deck Blank	et	
	PRTK0260	REO to PO: Perlite	35d	10-Mar-27	28-Apr-27							RFQ	to PO: Perlite			
	Supply		130d	19-lan-27	23- Jul-27											
	PRTK0180	Supplier Lead Time: Foamglas Blocks	50d	19-Jan-27	30-Mar-27							🔲 Suppl	ier Lead Time:	Foamglas Bloc	ks	
	PRTK0240	Supplier Lead Time: Resilient Blanket	25d	10-Mar-27	14-Apr-27							📋 Supp	lier Lead Time	: Resilient Blank	et	
	PRTK0210	Supplier Lead Time: Fiberglass Deck Blanket	25d	08-Apr-27	12-Mav-27							🔲 Sup	plier Lead Tim	e: Fiberglass De	eck Blanket	
	PRTK0270	Supplier Lead Time: Perlite	60d	29-Apr-27	23-Jul-27							📩 Sı	upplier Lead T	īme: Perlite		
	Delivery		90d	31-Mar-27	09-Aug-27											
	PRTK0190	Delivery: Foamglas Blocks	5d	31-Mar-27	07-Apr-27							I Delive	ery: Foamglas	Blocks		
	PRTK0250	Delivery: Resilient Blanket	10d	15-Apr-27	28-Apr-27							1 Deliv	ery: Resilient I	Blanket		
	PRTK0220	Delivery: Fiberglass Deck Blanket	5d	13-May-27	19-May-27							I Deliv	/ery: Fiberglas	ss Deck Blanket		
	PRTK0280	Delivery: Perlite	10d	26-Jul-27	09-Aug-27							I D	elivery: Perlite			
	05 STRUCTURAL	MATERIALS	225d	25-Jan-27	15-Dec-27								-			
	RFQ to PO		112d	25-Jan-27	05-Jul-27											
	PRST3090	RFQ to PO: Plg #3 Structural Steel - Tank Grade Rack	55d	25-Jan-27	13-Apr-27							🔲 RFQ	to PO: Plg #3	Structural Steel	- Tank Grade Racl	۲
	PRST0010	RFQ to PO: Pkg #1 Structural Steel - Pump Platform Modules	30d	31-Mar-27	12-May-27							🔲 RFC	to PO: Pkg #	1 Structural Ste	el - Pump Platform	M
	PRST0040	RFQ to PO: Structural Steel - Pkg #2 Platforms, Stairs, Walkways & Handrai	30d	21-May-27	05-Jul-27							🔲 RF	Q to PO: Stru	ictural Steel - Pk	g #2 Platforms, Sta	airs
	Supply		147d	14-Apr-27	12-Nov-27								7		<u> </u>	_
	PRST3100	Supplier Lead Time: Pkg #3 Structural Steel - Tank Grade Rack	140d	14-Apr-27	02-Nov-27				· · · · · · · · · · · · · · · · · · ·	1 1 1 1 1 1 1 1 1			Supplier Lea	ad Ime: Pkg #3	structural Steel - T	an
	PRST0020	Supplier Lead Time: Pkg #1 Structural Steel - Pump Platform Modules	90d	13-May-27	21-Sep-27								Supplier Lead	I lime: Pkg #1 S	tructural Steel - Pu	m
	PRST0050	Supplier Lead Time: Structural Steel - Pkg #2 Platforms, Stairs, Walkways & Handr	۲ 90d	06-Jul-27	12-Nov-27								Supplier Lea	ad Time: Structu	ral Steel - Pkg #2 F	۶la
	Delivery		59d	22-Sep-27	15-Dec-27											
									D - 4-			au dialian-		Ola a alta d	A	
	Remaining Leve	el of Effort			Data Date:	26-Jul-24		-			R	evision		Checked	Approved	

-	Actual Level of Effort

Remaining Work Critical Secondary

Page: 69 of 73

19-Aug-24 IFI

SOLARIS

Revision	Checked	Approved



Activ	ty ID	Activity Name	Rem.	Start	Finish)20	2021	2022	2023	2024	2025	2026	2027	20)28	2029	2030	2031
	_		Duration			Q												
	PRST0030	Delivery: Pkg #1 Structural Steel - Pump Platform Modules	17d	22-Sep-27	15-Oct-27										y: Pkg #1	Structura	I Steel - F	ump Plattor
	PRST3110	Delivery: Pkg #3 Structural Steel - Tank Grade Rack	30d	03-Nov-27	15-Dec-27										ery: Pkg	#3 Structu	Iral Steel	- lank Grad
	PRST0060	Delivery: Structural Steel - Pkg #2 Platforms, Stairs, Walkways & Handrail	17d	15-Nov-27	07-Dec-27									Deliv	ery: Struc	tural Stee	I - Pkg #2	Platforms, S
	16 ELECTRICAL	MATERIALS	155d	26-Jan-27	07-Sep-27													
	RFQ to PO	DEO to DOLE SU Common on t Matter Tank Orado Dock	87d	26-Jan-27	31-May-27									to PO∙ I	- &I Comr	onent Ma	tle - Tank	Grade Racl
	PREL1790	PEO to PO. Exil Component Mais - Tank Grade Rack	550	20-Jan-27	14-Api-27	_								to PO-I	iahtina F	ivtures / L	amns / Δ	viation Lights
	PREL0130	RFQ to PO: Lighting Fixtures / Lamps / Aviation Lights	450	04-Feb-27	09-Apr-27										able li a	Inders & C	anips / A	
	PREL0040	RFQ to PO: Cable Ladders & Condult	450	19-Feb-27	23-Apr-27											Inetrumor		
	PREL0070	RFQ to PO: Power & Instrument Cables	45d	12-Mar-27	14-May-27													Nondo
	PREL0100	RFQ to PO: Power & Instrument Cable Glands	45d	12-Mar-27	14-May-27													
	PREL0160	RFQ to PO: Termination Kits, Outlets, Plugs & Switches	50d	12-Mar-27	21-May-27											on Kils, O	ulleis, Pil	igs & Swiich
	PREL0010	RFQ to PO: Tank Grounding & Lightning System	39d	18-Mar-27	12-May-27										lank Gro	unaing &	Ligntning	System
	PREL0190	RFQ to PO: Junction Boxes & Instrument Enclosures	40d	05-Apr-27	31-May-27									-Q to PO	Junction	Boxes &	Instrume	nt Enclosure
			95d	12-Apr-27	25-Aug-27									upplior	ad Timo:	l iabtina E	vturos /	amps / Avis
	PREL0140	Supplier Lead Time: Lighting Fixtures / Lamps / Aviation Lights	50d	12-Apr-27	21-Jun-27													Latte Teels
	PREL1800	Supplier Lead Time: E&I Component Matts - Tank Grade Rack	60d	15-Apr-27	09-Jul-27									Supplier L				haus-rank (Senduit
	PREL0050	Supplier Lead Time: Cable Ladders & Conduit	35d	26-Apr-27	14-Jun-27									upplier Le	ad lime:		ders & C	onduit
	PREL0020	Supplier Lead Time: Tank Grounding & Lightning System	29d	13-May-27	23-Jun-27								5	upplier Le	ad lime:	Iank Gro	unaing &	Lightning S
	PREL0080	Supplier Lead Time: Power & Instrument Cables	64d	17-May-27	17-Aug-27									Supplier	Lead Im	e: Power (–	& Instrum	ent Cables
	PREL0110	Supplier Lead Time: Power & Instrument Cable Glands	64d	17-May-27	17-Aug-27									Supplier	Lead Tim	e: Power	& Instrum	ent Cable G
	PREL0170	Supplier Lead Time: Termination Kits, Outlets, Plugs & Switches	28d	25-May-27	02-Jul-27									Supplier L	ead Time	Terminati	on Kits, C)utlets, Plugs
	PREL0200	Supplier Lead Time: Junction Boxes & Instrument Enclosures	60d	01-Jun-27	25-Aug-27									Supplier	Lead Tim	e: Junctio	n Boxes	& Instrumen
	Delivery		58d	15-Jun-27	07-Sep-27													
	PREL0060	Delivery: Cable Ladders & Conduit	14d	15-Jun-27	05-Jul-27									Jellvery: C		ders & Co	nauit	
	PREL0150	Delivery: Lighting Fixtures / Lamps / Aviation Lights	5d	22-Jun-27	28-Jun-27									elivery: L	ignting Fi	xtures / La	imps / Av	ation Lights
	PREL0030	Delivery: Tank Grounding & Lightning System	15d	24-Jun-27	15-Jul-27									Delivery:	ank Grou	Inding & L	ightning	System
	PREL0180	Delivery: Termination Kits, Outlets, Plugs & Switches	14d	05-Jul-27	22-Jul-27									Delivery:	lerminatio	n Kits, Ou	itlets, Plu	gs & Switche
	PREL1810	Delivery: E&I Component Matls - Tank Grade Rack	30d	12-Jul-27	23-Aug-27									Delivery:	E&I Con	iponent M	atls - Tan	k Grade Ra
	PREL0090	Delivery: Power & Instrument Cables	5d	18-Aug-27	24-Aug-27								1	Delivery:	Power &	Instrumer	nt Cables	
	PREL0120	Delivery: Power & Instrument Cable Glands	14d	18-Aug-27	07-Sep-27								0	Delivery	: Power &	Instrume	nt Cable	Glands
	PREL0210	Delivery: Junction Boxes & Instrument Enclosures	5d	26-Aug-27	01-Sep-27								1	Delivery	: Junction	Boxes &	Instrume	nt Enclosure
	11 MECHANICAL	EQUIPMENT	243d	19-Feb-27	11-Feb-28													
	RFQ to PO		80d	19-Feb-27	14-Jun-27													
	PREQ0010	RFQ to PO: Jib Crane	80d	19-Feb-27	14-Jun-27													
	PREQ0040	RFQ to PO: Dry Chem System	80d	19-Feb-27	14-Jun-27								R		Dry Che	em Systen	n	
		Constitution of Trace, the Constant	153d	15-Jun-27	28-Jan-28										nnlier I er	d Time: li	h Crane	
	PREQUUZU		153d	15-Jun-27	28-Jan-28													System
	PREQ0050	Supplier Lead Time: Dry Chem System	153d	15-Jun-27	28-Jan-28									Su		a nine. D	iy Chem	System
			10d	31-Jan-28	11-Feb-28									I De	livery lih	Crane		
			UC L	31-Jdf1-20											liverv: Dr	Chem S	vstem	
				31-Jan-28	11-FeD-28												,50011	
	FABRICATION MI	LESTONES	525d	26-Nov-27	17-Iviay-29													
				20-1100-27	17-1viay-29	1 1						1	1 1	• • •		. •	<u> </u>	

Remaining Level of Eff	ort — Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 70 01 75		

SOLARIS

Revision	Checked	Approved



	Ċ	FORTIS BC	Till TLS	Tilbury Island LNG Facility Expansion (3 BCF) TLSE Sub-Projects Integrated Schedule-Detailed Rem. Start Finish 20 2021 2022 2023 2024								SOLARIS MANAGEMENT CONSULTANTS INC.								
Activ	ity ID	Activity Name	Rem Durati	. Start	Finish			2023		2025	2026				2030 2031					
	MS3060	Fab: Start of Pump Platform Module Fabrication	0d	26-Nov-27									♦ Fab: Start c	of Pump Platform	Module Fabricati	ion				
	MS4700	Fab: ROS: Off-Module Pipe Spools (Utility Lines, Risers, Und	er-Module & Misc) 0d		06-Jan-29									Fab: ROS: Of	f-Module Pipe Sp	000				
	MS4710	ROS: Pump Platform Modules & Other Structures	0d		17-May-29									♦ ROS: Pu	mp Platform Mod	ule				
	13 INNER TANK 8	SUSPENDED DECK FABRICATION	200	d 09-Jul-27	28-Feb-28							-								
	FBTK0130	Fab: Inner Tank Stiffeners, Pump Tubes & Internal Piping	200	d 09-Jul-27	28-Feb-28								Fab: Inn	er Tank Stiffener	s, Pump Tubes &	Int				
	FBTK0150	Fab: Suspended Deck Materials	85d	14-Oct-27	21-Jan-28								🛑 Fab: Sus	pended Deck Ma	terials					
	05 PUMP PLATFO	RM FABRICATION	392	18-Oct-27	17-May-29							V								
	FBST1000	Customs: Pkg #1 Structural Steel - Customs Clearance	50d	18-Oct-27	04-Jan-28								Customs:	PKg #1 Structura	al Steel - Customs	s C				
	FBMO0030	Fab: Pump Platform Modules	407	26-Nov-27	15-Mar-29									Fab: Pump	Platform Modules	S				
	FBMO0040	Fab: Off-Module Piping & Structural	260	26-Nov-27	25-Sep-28									ab: Off-Module	Piping & Structura	al				
	FBMO1080	Shipment: Off-Module Piping & Structural	78d	26-Sep-28	25-Dec-28									Shipment: Off	Module Piping &	Str				
	FBMO1060	Shipment: Off-Module Pipe Spools (Utility Lines, Risers, Under	er-Module & Misc) 49d	04-Nov-28	06-Jan-29									Shipment: Off	Module Pipe Spo	Sol				
	FBMO0010	Shipment: Pump Platform Modules	59d	19-Mar-29	17-May-29									Shipmen	: Pump Platform	Mo				
	05 MISC PLATFOR	RMS, STAIRS, WALKWAYS & HANDRAIL FABRICATION	206	17-Mar-28	13-Nov-28															
	FBMO1100	Fab: Structural Platforms, Stairs, Walkways & Handrail	164	17-Mar-28	25-Sep-28									-ab: Structural Pl	attorms, Stairs, vv	air				
	FBMO1090	Shipment: Structural Platforms, Stairs, Walkways & Handrail	42d	25-Sep-28	13-Nov-28									Shipment: Struc	tural Platforms, S	tair				
	FBST2020	ROS: Structural Steel Pkg #2 Platforms, Stairs, Walkways & H	landrail (Site) 0d		13-Nov-28								•	ROS: Structural	Steel Pkg #2 Pla	ttor				
	13 OUTER TANK P		1750	06-Apr-27	15-Dec-27									am Skotoboo P E	mbada					
	FBIK0010	Fab: Outer Bottom Sketches & Embeds	90d	06-Apr-27	20-Jul-27								ab. Ouler boll	UTT SKeiches & E						
	FBTK0110	Fab: Temporary Erection Material	53d	23-Apr-27	09-Jul-27								ab: lemporary							
	FBTK0070	Fab: Outer Tank (Roof Fittings & Nozzles)	20d	04-May-27	01-Jun-27								5. Outer lank (1				
	FBTK0030	Fab: Outer Liner Shell Rings #1 - #18 & Compression Bar	81d	15-Jun-27	17-Sep-27								Fab: Outer Lir	her Shell Rings #	1 - #18 & Compre	ess				
	FBTK0090	Fab: Outer Tank Roof & Supports	83d	16-Jun-27	14-Oct-27								Hab: Outer la	ank Roof & Supp	orts					
	FBTK0020	Fab: CS Wall & Roof Embeds	36d	03-Nov-27	15-Dec-27			ļ					Fab: CS W	/all& Roof Embe	ds					
	Construction	NII FOTONEO	716	25-Oct-27	16-Sep-30															
	MS3160	MILESIONES	/150 0d	26-Oct-27	16-Sep-30								Construction	: Mobilize @ Site						
	MS3050	Construction: Mobilize UCBI Site PMT	b0	20-00t-27									▲ Constructio	n: Mobilize HCBI	Site PMT					
	MS3070	Construction: Nobilize Field Sile Field	b0 b0	10 Mar 28		_							♦ Constru	iction: Start of Ou	ter Tank Steel Bo	otto				
	MS2090	Construction: Start of Outer Tank Steel Boltom	00 0d	10-11101-20	02 Eab 20							• • • • • • • • • • • • • • • • • • • •		▲ Construction	Completion of C	:om				
	MS2000	Construction: Completion of Complession Bar & top Embeds	5 0d	12 Mar 20	02-Feb-29										n: Completion of l	Ro				
	Me2100	Construction: Completion of Poof Diretto	DU רה	13-10181-29	26 100 20									▲ Constru	ction: Completion	.∼ 1 ∩f				
	Me3120	Construction: Completion of Sotting Dump Platform	DU הס		10 Son 20										struction: Complet	tior				
	MS2110	Construction: Completion of Setting Fullip Flation	00 0d	10 Oct 20	19-Sep-29									▲ Con	struction Begin T	Tanl				
	Me2120	Construction: Degin latik Glade Rack Foundations	DU רה	10-00-29	10 Apr 20						· · · · · · · · · · · · · · · · · · ·			• • • • •	◆ Construction	Co				
	NS3150	Construction: Completion of Tank Hydrotest	D0		16 Son 20											tior				
	MS3150	Construction: LING Tank Ready for Cooldown	D0		16-Sep-30											tion				
		Construction Completion - Base Scope	00		16-Sep-30															
	MS1170	Client: Full Access to LNG Tank Foundation Pad Area	01 \ D0	25-0ct-27	10-Sep-30								Client: Full A	cess to LNG Ta	nk Foundation Pa	ad /				
	MS1180	Client: Full Access to Temporary Facility Location	ьо Ь0	25-Oct-27									Client: Full A	cess to Tempora	ary Facility Locatio	on				
	MS1190	Client: Full Access to Remote Lavdown and Warehousing Ara	ea 0d	25-Oct-27									Client: Full A	ccess to Remote	Laydown and W	/are				
	MS4040	Client: MOE & Heavy Haul Route Availability & Clearance	۵۵ مر ۵۵	25-0ct-27									Client: MOF	& Heavy Haul Ro	oute Availability &	Cle				
	MS4020	Client: Dummy Pumps (Sunnlied By Others)	۵۵ ۲۵	20-00-27	04_Mar_30										Client: Dummv F	Jun				
	Remaining Lev	el of Effort 🛛 💻 Actual Work 🛛 📕 Critical Remain	ing Work		Data Date:	26-Jul-24			Date		F	Revision		Checked	Approved					
	Actual Level of	Effort Effort Critical Second	ary		Page: 71	of 73			9-Aug-24	_ IFI 				_	_					



			Tilbu	ry Island I	LNG Facili	ty Expansion (3 BCF)		MALL SO		l I
		FORTIS BC"	TLSE	Sub-Proje	ects Integra	ted Schedule-Detailed		= 30	LAKIS	
		I OITIIO DC						CONSU	A G E M E N T LTANTS INC	
Activ	rity ID	Activity Name	Rem.	Start	Finish	020 2021 2022 202	23 2024 2	2025 2026 2027 2028	2029 20	030 2031
			Duration							
	MS4030	Client: Deliver Client Supplied In-Tank Pumps & Foot Valves	0d		22-Mar-30				• (Client: Deliver Clie
	MS4050	Client: BOP Subcontractor Start of BOG Compressor Bldg & Appurtenances	0d		16-Sep-30					♦ Client: BOP
	MS4060	Client: Install In-Tank Pump Power Cable	0d		16-Sep-30					 Client: Instal
	03 MOBILIZATION		25d	26-Oct-27	01-Dec-27					
	MOB1110	Site Mobilization	25d	26-Oct-27	01-Dec-27			Site Mobilization		
	05 STRUCTURAL	STEEL	198d	16-Oct-28	04-Oct-29					otion Eloyator 9 E
	CNS15110	Install Construction Elevator & Root Perimeter Handrail	/5d	16-Oct-28	08-Mar-29					
	CNST5120	Install Instrument & PSV/ VRV Platforms	60d	26-Jun-29	04-Oct-29					nstrument & PSV
		Concrete: Foundation Bottom Slob	435d	24-Nov-27	13-Sep-29				oundation Bott	om Slah
				24-INOV-27	10-Mar-28				Concrete: Wa	Il ifts 1 thru 10
			1750	04-Jul-28	28-Mar-29					Place Poof Conc
		Concrete: Place Roof Concrete & Plinths	620	28-Mar-29	26-Jun-29					
	CNCO0060	Concrete: Horizontal & Vertical Pre-Stress Tensioning Ring Beam	10d	28-Mar-29	12-Apr-29					
	CNCO1760	Concrete: Horizontal & Vertical Post Tensioning	54d	26-Jun-29	13-Sep-29					
	13 LNG TANK	Crd Assemble Outer Shall 2 Ding	652d	08-Dec-27	30-Jul-30			Grd Assem	le Outer Shell :	3-Ring
	CNTK2300	Gid Assemble Outer Shell 3-Ring	450	00-DeC-27	10-Mar-20				ar Embeds	
		Install Annual Empeas		14-Feb-28	10-Mar-28				tall Outer Shell	Rings #1_#16
	CNTK1100	Install Outer Snell Rings #1-#16	1510	10-Mar-28	27-INOV-28					& Install Roof Line
	CNTK2450	Ground Assemblies & Install Root Liner	1060	10-Mar-28	11-Sep-28			Glou	or Lipor Pottor	
	CNTK2470	Install Outer Liner Bottom	35d	10-Mar-28	11-May-28					
	CNTK2460	Install Root Nozzles	60d	03-Aug-28	14-Nov-28					
	CNTK2440	Install Suspended Deck	65d	11-Sep-28	16-Jan-29				istall Suspende	
	CNTK2370	Install Compression Bar & Top Embeds	30d	27-Nov-28	02-Feb-29				nstall Compres	SION BAR & IOP ER
	CNTK2380	Air Raise Roof & Secure to Compression Bar	22d	02-Feb-29	13-Mar-29				Air Raise Roo	r & Secure to Con
	CNTK2390	Install Leveling Concrete, Cellular Block, Bearing Ring, Inner Annular & TCP	70d	13-Mar-29	21-Jun-29				Install Leve	ling Concrete, Ce
	CNTK2400	Install Inner Tank Shell Rings #1-#13	174d	21-Jun-29	15-Mar-30				ir (istall Inner Iank S
	CNTK2410	Install Secondary & Inner Tank Bottoms	54d	02-Nov-29	04-Feb-30				ins 📕	stall Secondary &
	CNTK2420	Install Inner Ladder, Platforms & Internal Piping / Nozzles	87d	02-Nov-29	22-Mar-30					nstall Inner Ladde
	CNTK2480	Grd Assemble, Install & Hydrotest Pump Tubes	60d	02-Nov-29	04-Mar-30				🛑 G	rd Assemble, Inst
	CNTK2250	Install Inner Shell Doorsheet	7d	11-Mar-30	22-Mar-30				e li	nstall Inner Shell [
	CNTK2430	Hydrotest Inner Tank, Clean & Dry	15d	22-Mar-30	18-Apr-30					Hydrotest Inner T
	CNTK2260	Install TCP & Outer Shell at Doorsheet	10d	23-May-30	10-Jun-30					Install TCP & O
	CNCO1750	Closure: TOC's Outer Tank Steel, Concrete & Pre-Stress	47d	23-May-30	30-Jul-30					Closure: TOC
	CNTK2350	Pneumatic & Vacuum Test of Tank	3d	10-Jun-30	14-Jun-30					Pneumatic & Va
	15 PUMP PLATFO	RM / PIPING	235d	29-Mar-29	23-May-30					
	CNMO1160	Install Shell Guides	65d	29-Mar-29	19-Jul-29				💼 Install She	ell Guides
	CNIN0010	Piping Ground Assemblies	52d	12-Apr-29	12-Jul-29				Piping Gro	ound Assemblies
	CNMO1060	Receive At Site: Modules At MOF & Transport to Tank Site	15d	17-May-29	12-Jun-29				Receive At	Site: Modules At
	CNMO1170	Prep & Install Pump Platform Modules	42d	12-Jul-29	19-Sep-29				Prep &	Install Pump Platf
	CNMO1165	Install Structural Under-Platform	65d	20-Jul-29	05-Nov-29				💼 Install	Structural Under
	CNMO1190	Install Module Inter-Connect Piping	80d	06-Nov-29	09-Apr-30					nstall Module Inte
	CNMO1200	Install Pipe Insulation	41d	14-Mar-30	23-May-30					Install Pipe Insul
	16 / 17 ELECTRIC	AL & I&C	120d	13-Sep-29	25-Apr-30				V	
	Remaining Lev	el of Effort Actual Work Critical Remaining Work			Data Date:?	6lul-24	Date	Revision	Checked	Approved
	Actual Level of	Effort Effort Remaining Work Critical Secondary			Page: 72	of 73	19-Aug-24 II	FI		
1					•		1			



\ctiv	vity ID	Activity Name	Rem.	Start	Finish	020		2	021		20	22	2023		20	24		2025	-	-
			Duration			Q	Q	QQ	Q	QC	Q	QQ	QQ	Q	QQ	QQ	Q	QQ	Q	Q
	CNEI0010	Electrical: Tray, J-Box, Cable, Instruments & Terminations	120d	13-Sep-29	25-Apr-30			-	+ +				-		-			-		Ī
	CNEI5010	Electrical: Tank Lighting	60d	13-Sep-29	10-Jan-30															
	22 TANK GRADE R	łack	207d	06-Sep-29	16-Sep-30															
	CNCI5020	Civil: Demob Crane	20d	06-Sep-29	10-Oct-29								1							
	CNCI5100	Civil: Form & Place Concrete - Grade Rack Fdns Area A	25d	10-Oct-29	22-Nov-29			÷					1		1					
	CNCI5030	Structural: Grade Rack - Lower Level 1A	25d	22-Nov-29	21-Jan-30			-					1		:					
	CNCI5110	Civil: Form & Place Concrete - Grade Rack Fdns Area B	25d	22-Nov-29	21-Jan-30								-	: : :	:					
	CNCI5040	Piping: Grade Rack - Lower Level 1A	20d	21-Jan-30	25-Feb-30								1		:					
	CNCI5060	Structural: Grade Rack - Lower Level 1B	25d	21-Jan-30	05-Mar-30															-
	CNCI5090	Structural: Grade Rack - Upper Level 2A	35d	21-Jan-30	22-Mar-30			÷												
	CNCI5160	Piping: Grade Rack - Lower Level 1B	20d	25-Feb-30	01-Apr-30								-							
	CNCI5050	E&I: Grade Rack - Lower Level	12d	05-Mar-30	26-Mar-30															
	CNCI5150	Structural: Grade Rack - Upper Level 2B	45d	22-Mar-30	10-Jun-30										-					
	CNCI5070	Piping: Grade Rack - Level 2A	29d	01-Apr-30	22-May-30															-
	CNCI5170	Piping: Grade Rack - Level 2B	45d	22-May-30	08-Aug-30								-							
	CNCI5180	E&I: Grade Rack - Upper Level	23d	08-Aug-30	16-Sep-30										1					
	11 EQUIPMENT		168d	04-Oct-29	09-Aug-30			÷												
	CNEQ1220	Install Jib Crane	15d	04-Oct-29	30-Oct-29															
	CNEQ1240	Install Dry Chem System	20d	19-Nov-29	07-Jan-30															-
	CNEQ1230	Install In-Tank Pumps	12d	22-Jul-30	09-Aug-30										-					
	07 INSULATION		71d	18-Apr-30	20-Aug-30										1					
	CNIN1150	Install Shell Resilient Blanket	20d	18-Apr-30	23-May-30			÷							1					
	CNIN1120	Install Perlite in Annular Space	21d	14-Jun-30	22-Jul-30	1 1 1 1			1 1 1 1 1 1		· · · ·							· · · · · · · · · · · · · · · · · · ·		
	CNIN1130	Install Fiberglass Blanket on Deck & Nozzle Piping	17d	22-Jul-30	20-Aug-30			-						: : :	: : :					
	N2 PURGE		15d	20-Aug-30	13-Sep-30															
	PCOM0010	N2 Purge	15d	20-Aug-30	13-Sep-30										-					
	Commissioning		40d	22-Aug-30	18-Oct-30			÷					1		1					
	TTK500030	Commissioning/Startup	40d	22-Aug-30	18-Oct-30			÷												

Remaining Level of Effort	Actual Work	Critical Remaining Work	Data Data 26 Jul 24	Date	
			Data Date.20-Jul-24	19-Aug-24	IFI
			Page. 73 01 73		

		1	1			1	1	/III/		CL NO	5	C	L	A	F	ENT ENT	5				
_	20	26	_		20	27			20	20	Ur.	ist	20	20	413	2	030	_		203,	1
	20	20	0	0	20	21	0	0	20	20	0	0	20	29	0			0		03	
		Q	9	Q	Q	Q	9	Q	Q	Q	Q	Q	Q		Q		Elec	ctric	al:	Tray	. J-
															1	Fle	ctric	al: 1	anl	ې Lic	nhtir
										1					:			,			
		 -														civil: C	emc	b d	Crai	ne	
																Civil:	For	n 8	Pla	ace	Co
																Str	uctu	ral:	Gr	ade	Ra
																Civ	/il: Fo	orm	&	Plac	жd
																P	ipina	I: G	rad	le R	ack
															, , ,		truct	ura	ıl C	irac	le R
															: : :		Struc	tur	al (Gra	de l
															: : :		Dinin		Cro	do	
															1 1 1		≓oı. /	ny. Ora			
															: : :		=œl. (516	ide	Ra	CK -
		; ; ;													; ; ;		5	ruc	tura	al: G	rac
															: : :		Pip	oing	: G	rad	e R
															1	I		Pip	ing	Gr	ade
															1			E	&I: (Gra	de I
										: : :				١				_	1		- - - -
1		· · ·								 			4			Install	Jib (Cra	ne		
																Ins	tall D	ry	Che	em	Sys
															1			Ins	tall	n-T	ank
															1						
															1 1 1		Ins	tall	Sh	ell F	Resi
													4		1 1 1			nst	all F	Perli	te ir
															: : :			Ins	tall	Fib	erg
																	₩				
															: : :			N2	2 P	urge	e
															1 1 1						
															1 1 1		: -		on	nmķ	SSIO
			R	evis	sion							+	0	Che	cke	d		App	orov	ed	
												+									
												- 1									

Appendix I
VALIDATION ESTIMATING CONTINGENCY REPORT

FILED CONFIDENTIALLY

Appendix J VALIDATION ESTIMATING ESCALATION REPORT

FILED CONFIDENTIALLY

Appendix K FINANCIAL SCHEDULES (PREFERRED ALTERNATIVE)

REFER TO LIVE SPREADSHEET MODEL

Provided in electronic format only

FILED CONFIDENTIALLY

Appendix L NOTIFICATION LETTER TO POTENTIALLY AFFECTED INDIGENOUS GROUPS



December 19, 2023

RE: FORTISBC TILBURY LNG STORAGE EXPANSION PROJECT UPDATE

FortisBC would like to update ______ regarding the regulatory review of the Tilbury LNG Storage Expansion Project (TLSE). In particular, one component of the Tilbury Phase 2 LNG Expansion Project, which includes the proposed addition of 142,400 m3 of Liquified Natural Gas (LNG) storage to the existing Tilbury LNG facility, is being reviewed as part of two simultaneous regulatory processes: (1) a regulated utility review process undertaken by the British Columbia Utilities Commission (BCUC); and (2) a provincially-led environmental assessment process. FortisBC provides an update regarding review process below.

Regulated Utility Review Process

As a regulated utility, FortisBC requires a Certificate of Public Convenience and Necessity (CPCN) from the BCUC for major projects that may affect rates paid by FortisBC customers. On December 29, 2020, FortisBC filed an Application for a CPCN for the Tilbury Liquefied Natural Gas (LNG) Storage Expansion Project with the BCUC to seek approval for the construction of a new LNG storage tank that will provide an additional backup source of natural gas for our customers in the Lower Mainland in the event of a gas supply disruption. If the CPCN is approved by the BCUC, we estimate that construction of the storage tank could begin as early as 2025 with projected completion by 2029.

On March 23, 2023, the BCUC released a decision adjourning the proceedings to permit FortisBC additional time to address specific comments raised by the BCUC and interveners in relation to its Application. The BCUC, among other things, requires further information on project alternatives and gas disruption risks throughout FortisBC's gas distribution system. FortisBC is addressing these comments and currently plans to file additional supplementary evidence in support of the planned LNG storage expansion in September, 2024. We will continue to provide you with updates on the process as well as a copy of the supplementary evidence to ______ concurrently with filing to the BCUC.

Environmental Review Process

FortisBC has continued to engage with ______ regarding the Tilbury Phase 2 LNG Expansion project as it proceeds through the environmental assessment process led by the British Columbia Environmental Assessment Office and the Impact Assessment Agency of Canada. The Tilbury Phase 2 LNG Expansion encompasses a larger expansion of the Tilbury site, including both the new LNG storage tank which is part of the current CPCN process as well as additional LNG production capacity. As indicated above, the new LNG storage tank being assessed by the BCUC in the CPCN process is required to provide improved resiliency for FortisBC's gas customers in the event of a gas supply disruption.

As the next step in the environmental process, FortisBC expects to submit the Draft Environmental Assessment Application in September 2024 and will continue to engage with ______ to discuss comments and feedback throughout the environmental assessment.



FEI will continue to synchronize consultation activities for both the Tilbury Phase 2 LNG Expansion Project and the TLSE Project to ensure engagement is robust, efficient, and transparent, and help to reduce the overall burden placed on Indigenous nations by removing the duplicative efforts that would otherwise be required to review each process separately.

If you have any questions regarding the Tilbury Project or the BCUC process, please contact me at <u>courtney.hodson@fortisbc.com</u> or by phone at 604-592-7603.

Sincerely,

Courtney Hodson Manager, Community & Indigenous Relations FortisBC

cc. Ian Finke, Director LNG Operations, FortisBC

Appendix M DRAFT ORDERS



Suite 410, 900 Howe Street Vancouver, BC Canada V6Z 2N3 bcuc.com P: 604.660.4700TF: 1.800.663.1385

ORDER NUMBER G-xx-xx

IN THE MATTER OF the Utilities Commission Act, RSBC 1996, Chapter 473

and

FortisBC Energy Inc. Application for a Certificate of Public Convenience and Necessity for the Tilbury Liquefied Natural Gas Storage Expansion Project – Supplemental Evidence

BEFORE:

[Panel Chair] Commissioner Commissioner

on Date

ORDER

WHEREAS:

- A. On October 24, 2024, FortisBC Energy Inc. (FEI) filed its Supplemental Evidence (Supplemental Evidence) together with FEI's 2024 Resiliency Plan with the British Columbia Utilities Commission (BCUC) as part of its Application for a Certificate of Public Convenience and Necessity (CPCN) for the Tilbury Liquefied Natural Gas (LNG) Storage Expansion (TLSE) Project filed on December 29, 2020 (together, TLSE Application), in accordance with the BCUC Decision and Order G-62-23;
- B. The TLSE Application requests approval of, among other things, a CPCN pursuant to sections 45 and 46 of the *Utilities Commission Act* (UCA) for the TLSE Project;
- C. By Decision and Order G-62-23, dated March 23, 2023, the BCUC adjourned the TLSE Application proceeding (Adjournment Decision), pending the filing of evidence described in the Adjournment Decision;
- D. FEI states that the Supplemental Evidence and the 2024 Resiliency Plan reinforce that the TLSE Project is in the public interest and should be approved as proposed, including the associated deferral account and depreciation and salvage rate approvals requested in the Application;
- E. FEI requests that Appendices D, E, G, I, J, and K to the Supplemental Evidence be held confidential due to the operationally sensitive and commercially sensitive nature of the information. FEI further requests that redacted information in the 2024 Resiliency Plan and related Appendices RP 1, RP 2 and RP 4 be held confidential on a restricted basis due to the highly sensitive security related nature of the information and that it be available only to the BCUC (together, Confidential Information); and

F. The BCUC has commenced its review of the Supplemental Evidence and 2024 Resiliency Plan and finds that the establishment of a regulatory timetable to restart the proceeding is warranted.

NOW THEREFORE the BCUC orders as follows:

- 1. The written public hearing for the review of the TLSE Application is restarted and a regulatory timetable for the review of the Supplemental Evidence and 2024 Resiliency Plan is established as set out in Appendix A to this order.
- 2. FEI is to provide a copy of this Supplemental Evidence, the 2024 Resiliency Plan, and this order, electronically where possible, on or before Friday, December 8, 2024, to all registered interveners in the TLSE Application proceeding.
- 3. FEI is to publish the public versions of the Supplemental Evidence and 2024 Resiliency Plan, and a copy of this order, on its website at <u>www.fortisbc.com</u> as soon as practicable, but no later than Friday, December 6, 2024.
- 4. FEI is to provide confirmation to the BCUC that it has complied with Directives 2 and 3 of this order by Friday, December 13, 2024.
- 5. Appendices D, E, G, I, J, and K to the Supplemental Evidence will be held confidential until determined otherwise by the BCUC.
- 6. The redacted portions of the 2024 Resiliency Plan and related Appendices will be held by the BCUC on a restricted confidential basis, accessible to the BCUC only.

DATED at the City of Vancouver, in the Province of British Columbia, this (XX) day of (Month Year).

BY ORDER

(X. X. last name) Commissioner

Attachment

FortisBC Energy Inc.

Application for a Certificate of Public Convenience and Necessity

for the Tilbury Liquefied Natural Gas Storage Expansion Project – Supplemental Evidence

REGULATORY TIMETABLE

Action	Date (2024)
FEI provides notice of Supplemental Evidence and this order	Friday, December 6
FEI provides confirmation of compliance with notice requirements	Friday, December 13
Action	Date (2025)
Technical Workshop on Exponent Inc. Risk and Consequence Determination Method*	Wednesday, January 29
BCUC Information Request (IR) No. 1	Thursday, February 20
Intervener IR No. 1	Thursday, February 27
FEI response to IRs No. 1	Thursday, April 3
BCUC IR No. 2	Thursday, April 17
Intervener IR No. 2	Thursday, April 24
FEI Response to IRs No. 2	Thursday, May 15
Letters of comment deadline	Thursday, May 22
FEI final argument	Thursday, June 12
Intervener final arguments	Thursday, July 3
FEI reply argument	Thursday, July 24

* The BCUC will conduct the workshop in person, commencing at 9am and will take place at [location]. FEI and registered interveners must register and provide a list of attendees by emailing the Commission Secretary at <u>commission.secretary@bcuc.com</u> to confirm their attendance at the workshop by Monday, January 27, 2025. Parties who cannot participate in person may submit a written request to the BCUC to attend virtually with an explanation supporting such a request, in confidence if necessary, by emailing <u>commission.secretary@bcuc.com</u> by the above-noted workshop registration deadline.



Suite 410, 900 Howe Street Vancouver, BC Canada V6Z 2N3 bcuc.com P: 604.660.4700
TF: 1.800.663.1385
F: 604.660.1102

ORDER NUMBER

C-<mark>xx-xx</mark>

IN THE MATTER OF the Utilities Commission Act, RSBC 1996, Chapter 473

and

FortisBC Energy Inc. Application for Approval of a Certificate of Public Convenience and Necessity for the Tilbury Liquefied Natural Gas Storage Expansion Project

BEFORE:

[Panel Chair] Commissioner Commissioner

on <mark>Date</mark>

ORDER

WHEREAS:

- A. On October 24, 2024, FortisBC Energy Inc. (FEI) filed its Supplemental Evidence (Supplemental Evidence) together with FEI's 2024 Resiliency Plan with the British Columbia Utilities Commission (BCUC) as part of its Application for a Certificate of Public Convenience and Necessity (CPCN) for the Tilbury Liquefied Natural Gas (LNG) Storage Expansion (TLSE) Project filed on December 29, 2020 (together, TLSE Application), in accordance with the BCUC Decision and Order G-62-23;
- B. The TLSE Application requests approval of, among other things, a CPCN pursuant to sections 45 and 46 of the *Utilities Commission Act* (UCA) for the TLSE Project which includes the following:
 - i. Construction and operation of a 3 billion cubic feet (Bcf) LNG storage tank;
 - ii. Construction and operation of 800 million cubic feet per day (MMcf/day) of regasification capacity;
 - iii. Construction or modification and operation of any of the necessary auxiliary systems, including items such as utility pipe racks, in-tank pumps, piping, and connections to the sendout gas pipeline; and
 - iv. Demolition of the above-ground portion of the Tilbury Base Plant LNG storage tank and liquefaction facilities;
- C. FEI also seeks BCUC approval, pursuant to sections 59 to 61 of the UCA of the following:

- The non-rate base TLSE Application and Preliminary Stage Development Costs deferral account, attracting a weighted average cost of capital return until the account enters rate base. FEI proposes to transfer the balance in the deferral account to rate base on January 1 of the year following BCUC approval of the Application and commence amortization over a three-year period thereafter;
- ii. The non-rate base TLSE Foreign Exchange (FX) Mark to Market deferral account, with no financing return, to capture the mark-to-market valuation of any foreign currency forward contracts entered into related to construction of the TLSE Project; and
- Depreciation and net salvage rates of 1.67 percent and 0.67 percent, respectively, for the new 3 Bcf LNG storage tank;
- D. By Decision and Order G-62-23 dated March 23, 2023, the BCUC adjourned the TLSE Application proceeding (Adjournment Decision), pending the filing of evidence described in the Adjournment Decision;
- E. FEI states that the Supplemental Evidence and the 2024 Resiliency Plan reinforces that the TLSE Project is in the public interest and should be approved as proposed, including the associated deferral account and depreciation and salvage rate approvals requested in the Application;
- F. In the proceeding, FEI has made various requests in its filings that certain information in the TLSE Application, responses to information requests (IRs), Rebuttal Evidence, Supplemental Evidence and the 2024 Resiliency Plan be held confidential due to operational sensitivity, security sensitivity or commercial sensitivity (together, Confidential Information);
- G. By Order G-XX-24 dated ###, the BCUC restarted the proceeding and established a regulatory timetable for the review of the Supplemental Evidence and 2024 Resiliency Plan, which consisted of public notice, a technical workshop and two rounds of IRs, followed by written arguments. The BCUC also ordered that certain portions of the Supplemental Evidence and 2024 Resiliency Plan be held confidential; and
- H. The BCUC has considered the TLSE Application, evidence and submissions in this proceeding and finds that the following determinations are warranted.

NOW THEREFORE pursuant to sections 45, 46 and 59 to 61 of the UCA, the BCUC orders as follows:

- 1. A CPCN is granted to FEI for the TLSE Project, which as described further in the TLSE Application and Supplemental Evidence, consists of the following:
 - a. construction and operation of a 3 billion cubic feet (Bcf) LNG storage tank;
 - b. construction and operation of 800 million cubic feet per day (MMcf/day) of regasification capacity;
 - c. construction or modification and operation of any of the necessary auxiliary systems, including items such as utility pipe racks, in-tank pumps, piping, and connections to the sendout gas pipeline; and
 - d. demolition of the above-ground portion of the Tilbury Base Plant LNG storage tank and liquefaction facilities.

- 2. FEI is approved to establish the following deferral accounts:
 - A non-rate base TLSE Application and Preliminary Stage Development Costs deferral account, attracting a weighted average cost of capital (WACC) return until the account enters rate base.
 FEI is also approved to transfer the balance in the deferral account to rate base on January 1 of the year following the date of this Decision and commence amortization over a three-year period thereafter.
 - b. A non-rate base TLSE FX Mark to Market deferral account, with no financing return, to capture the mark-to-market valuation of any foreign currency forward contracts entered into related to construction of the TLSE Project.
- 3. FEI is also approved to use the following for the new 3 Bcf LNG storage tank:
 - a. A depreciation rate of 1.67 percent.
 - b. A net salvage rate of 0.67 percent.
- 4. FEI is directed to file with the BCUC the following reports:
 - a. Within 30 days of the finalization of the construction contract, a Contract Finalization Report;
 - Within 30 days of the end of each semi-annual reporting period, starting after the submission of the Contract Finalization Report and ending upon the filing of the Final Report, Semi-Annual Progress Reports;
 - c. As soon as practicable but no longer than 30 days upon the identification of a material change including any significant delays or material cost variances, a Material Change Report (which may be filed as part of the Semi-Annual Progress Report where time permits); and
 - d. Within six months of the final in-service date, a Final Report.
- 5. FEI must comply with all other directives and determinations outlined in the decision accompanying this order.
- 6. The BCUC will continue to hold confidential the Confidential Information unless determined otherwise by the BCUC.

DATED at the City of Vancouver, in the Province of British Columbia, this (XX) day of (Month Year).

BY ORDER

(X. X. last name) Commissioner

Attachment (Yes? No?)